# Bernstein eigenvarieties

# Christophe Breuil and Yiwen Ding

# Abstract

We construct parabolic analogues of (global) eigenvarieties, of patched eigenvarieties and of (local) trianguline varieties, that we call respectively Bernstein eigenvarieties, patched Bernstein eigenvarieties, and Bernstein paraboline varieties. We study the geometry of these rigid analytic spaces, in particular (generalizing results of Breuil-Hellmann-Schraen) we show that their local geometry can be described by certain algebraic schemes related to the generalized Grothendieck-Springer resolution. We deduce several local-global compatibility results, including a classicality result (with no trianguline assumption at p), and new cases towards the locally analytic socle conjecture of Breuil in the non-trianguline case.

# Contents

1	Introduction				
<b>2</b>	Pre	Preliminaries			
	2.1	Gener	al notation	14	
	2.2	<i>p</i> -adic	differential equations over Bernstein components	15	
	2.3	Potent	tially crystalline representations	16	
3	Bernstein eigenvarieties				
	3.1	Abstra	act construction	18	
		3.1.1	Notation and setup	19	
		3.1.2	$(\Omega, \lambda)$ -part of Jacquet-Emerton modules	20	
		3.1.3	Local Bernstein eigenvarieties	27	
	3.2	Bernst	cein eigenvarieties	31	
		3.2.1	<i>p</i> -adic automorphic representations	31	
		3.2.2	Bernstein eigenvarieties: construction and first properties	32	
		3.2.3	Density of classical points	35	
		3.2.4	Galois representations	38	
		3.2.5	Locally analytic socle and companion points	42	
	3.3	Patche	ed Bernstein eigenvarieties	44	

4	Bernstein paraboline varieties					
	4.1	Deformations of $(\varphi, \Gamma)$ -modules				
		4.1.1 Deformations of certain irreducible $(\varphi, \Gamma)$ -modules	50			
		4.1.2 Deformations of type $\Omega$	56			
	4.2	Bernstein paraboline varieties	61			
	4.3	Potentially crystalline deformation spaces	71			
<b>5</b>	$\mathbf{The}$	The geometry of some schemes related to generalized Springer resolutions				
	5.1	Preliminaries	80			
	5.2	Analysis of the global geometry				
	5.3	Analysis of the local geometry	86			
	5.4	Characteristic cycles	88			
6	Loc	Local models for the Bernstein paraboline varieties				
	6.1	Almost de Rham $B_{dR}$ -representations $\ldots \ldots \ldots$	93			
	6.2	$(\varphi, \Gamma)$ -modules of type $\Omega$ over $\mathcal{R}_{E,L}[\frac{1}{t}]$	95			
	6.3	$(\varphi, \Gamma)$ -modules of type $\Omega$ over $\mathcal{R}_{E,L}$	102			
	6.4	Galois representations and Bernstein paraboline varieties	105			
	6.5	Galois cycles	109			
7	App	Applications 1				
	7.1	Automorphy cycles	111			
		7.1.1 Representation theoretic preliminaries	111			
		7.1.2 Cycles on patched Bernstein eigenvarieties	115			
	7.2	Companion constituents	120			
A	App	Appendix 13				
	A.1	$\Omega$ -filtration in families	130			
	A.2	Characteristic cycles of parabolic Verma modules	135			
	A.3	Bruhat intervals of length 2	141			
	A.4	Errata to [19]	143			

# 1 Introduction

Let p be a prime number. The study of p-adic eigenvarieties is an important and fruitful theme in arithmetic geometry. This paper is motivated by the role that eigenvarieties play in the study of local-global compatibility problems in the (p-adic) Langlands program (e.g. see [19] or [37]). The "classical" theory of eigenvarieties has the restriction that one can only see finite slope padic automorphic forms or representations, or, in terms of Galois representations, trianguline representations. In order to extend the theory to the *non-trianguline* case, we construct in this paper a parabolic version of eigenvarieties, that we call Bernstein eigenvarieties. Note that there was already some work in that direction, cf. [53], [68], [2] (see after Theorem 1.4 below for a brief comparison with [53]). These spaces parametrize certain p-adic automorphic (resp. Galois) representations which are not of finite slope (resp. not trianguline). Following the strategy and methods in the series of articles [17], [18], [19] (which themselves are based on many previous results by other people), we study and use the geometry of these Bernstein eigenvarieties and of their patched and (local) Galois avatars to obtain various local-global compatibility results in the non-trianguline case.

Before stating our main results, we briefly give the global setup of the paper (with some simplifications for convenience). Let  $n \geq 2$  an integer,  $F^+$  a totally real number field and F a totally imaginary quadratic extension of  $F^+$  such that all places of  $F^+$  dividing p split in F. We fix a unitary algebraic group G over  $F^+$  which becomes  $\operatorname{GL}_n$  over F and such that  $G(F^+ \otimes_{\mathbb{Q}} \mathbb{R})$  is compact and G is split at all places above p. Let  $U^p = \prod_{v \nmid \infty, p} U_v$  be a compact open subgroup of  $G(\mathbb{A}_{F^+}^{\infty,p})$ , and E be a sufficiently large finite extension of  $\mathbb{Q}_p$ . Put

$$\widehat{S}(U^p, E) := \{ f : G(F^+) \setminus G(\mathbb{A}^{\infty}_{F^+}) / U^p \to E, \ f \text{ continuous} \},\$$

which is a Banach space for the supremum norm and is equipped with a continuous (unitary) action of  $G(F_p^+) := G(F^+ \otimes_{\mathbb{Q}} \mathbb{Q}_p)$  by right translation on functions. For simplicity, we assume in this introduction that p is inert in  $F^+$ , and we fix a place  $\wp$  of F dividing p. We have then  $G(F_p^+) \cong \operatorname{GL}_n(F_p^+) \cong \operatorname{GL}_n(F_{\wp})$  and  $\widehat{S}(U^p, E)$  is a unitary Banach space representation of  $\operatorname{GL}_n(F_{\wp})$ . We recall that the locally algebraic vectors  $\widehat{S}(U^p, E)^{\operatorname{lalg}}$  (for the action of  $\operatorname{GL}_n(F_{\wp})$ ) admit a description in terms of classical automorphic representations of G (see for example Proposition 3.2.1 (1)). The space  $\widehat{S}(U^p, E)$  is also equipped with a faithful action of a certain commutative global Hecke algebra  $\mathbb{T}(U^p)$  over  $\mathcal{O}_E$  (the ring of integers of E) which is generated by sufficiently many prime-to-p Hecke operators. With some more assumptions (on G, F, etc., see § 3.2.4 and § 3.3), one can show that

- $\mathbb{T}(U^p)$  is isomorphic to a finite product of complete noetherian  $\mathcal{O}_E$ -algebras  $\mathbb{T}(U^p)_{\overline{\rho}}$ , indexed by some *n*-dimensional continuous representations  $\overline{\rho}$  of  $\operatorname{Gal}_F$  over the residue field  $k_E$  of  $\mathcal{O}_E$ ;
- for each  $\overline{\rho}$  such that  $\mathbb{T}(U^p)_{\overline{\rho}} \neq 0$ , there is a surjective morphism  $R_{\overline{\rho},S} \twoheadrightarrow \mathbb{T}(U^p)_{\overline{\rho}}$ , where  $R_{\overline{\rho},S}$  is the universal deformation ring of certain deformations of  $\overline{\rho}$  (see § 3.3).

These properties are not necessary for our construction of Bernstein eigenvarieties, but for convenience we assume they hold in the introduction. With respect to  $\mathbb{T}(U^p) \cong \prod_{\overline{\rho}} \mathbb{T}(U^p)_{\overline{\rho}}$ , we have a decomposition  $\widehat{S}(U^p, E) \cong \bigoplus_{\overline{\rho}} \widehat{S}(U^p, E)_{\overline{\rho}}$  of  $\operatorname{GL}_n(F_{\wp})$ -representations. We fix henceforth  $\overline{\rho}$  such that  $\mathbb{T}(U^p)_{\overline{\rho}} \neq 0$ , and thus  $\widehat{S}(U^p, E)_{\overline{\rho}} \neq 0$ . Let  $\mathfrak{m}_{\rho} \subset \mathbb{T}(U^p)_{\overline{\rho}}[1/p]$  be a maximal ideal and  $\rho$ :  $\operatorname{Gal}_F \to \operatorname{GL}_n(E)$  the associated representation, where  $\operatorname{Gal}_L := \operatorname{Gal}(\overline{L}/L)$  for a field L.

The subspace  $\widehat{S}(U^p, E)_{\overline{\rho}}[\mathfrak{m}_{\rho}]$  of  $\widehat{S}(U^p, E)_{\overline{\rho}}$  annihilated by  $\mathfrak{m}_{\rho}$  is an admissible unitary Banach space representation of  $\operatorname{GL}_n(F_{\wp})$ . The study of its relation with the local Galois representation  $\rho_{\wp} := \rho|_{\operatorname{Gal}_{F_{\wp}}}$ , referred to as local-global compatibility, is one of the main themes in the (*p*-adic) Langlands program. We first state our local-global compatibility results, which are obtained using the aforementioned Bernstein eigenvarieties, and we describe these latter afterwards. For these results, as in [19] we also need the following so-called Taylor-Wiles hypothesis:

# **Hypothesis 1.1.** (1) p > 2;

(2) the field F is unramified over  $F^+$ , F does not contain a non trivial root  $\sqrt[p]{1}$  of 1 and G is quasi-split at all finite places of  $F^+$ ;

- (3)  $U_v$  is hyperspecial when the finite place v of  $F^+$  is inert in F;
- (4)  $\overline{\rho}$  is absolutely irreducible and  $\overline{\rho}(\operatorname{Gal}_{F(\overline{k}/\overline{1})})$  is adequate.

Let  $B \subset \operatorname{GL}_n$  be the Borel subgroup of upper triangular matrices and  $T \subset B$  the subgroup of diagonal matrices. We first have the following classicality result.

**Theorem 1.2** (cf. Theorem 7.2.9). Assume Hypothesis 1.1 and the following hypothesis  $(at p)^1$ :

- (1)  $\rho_{\wp}$  is potentially crystalline with distinct Hodge-Tate weights and is generic (in the sense of  $\frac{9}{2.3}$ );
- (2) there exists a parabolic subgroup  $P \supseteq B$  of  $\operatorname{GL}_n$  such that  $J_P(\widehat{S}(U^p, E)^{\operatorname{an}}_{\overline{\rho}}[\mathfrak{m}_{\rho}])$  has nonzero locally algebraic vectors for  $L^D_P(F_{\wp})$ , where  $J_P(\cdot)$  is Emerton's locally analytic Jacquet functor for P([40]) and  $L^D_P$  is the derived subgroup of the Levi subgroup  $L_P \supseteq T$  of P.

Then  $\widehat{S}(U^p, E)[\mathfrak{m}_{\rho}]^{\text{lalg}} \neq 0$ , i.e.  $\rho$  is associated to a classical automorphic representation of  $G(\mathbb{A}_{F^+})$ .

The theorem in the case P = B was proved in [19]. Indeed, the assumption (2) in Theorem 1.2 when P = B is equivalent to  $J_B(\widehat{S}(U^p, E)^{an}_{\overline{\rho}}[\mathfrak{m}_{\rho}]) \neq 0$ , which means that  $\rho$  appears on the (classical) eigenvariety associated to G with tame level  $U^p$ . The main novelty of Theorem 1.2 is that  $\rho_{\wp}$  is not necessarily trianguline.

Assume that  $\rho_{\wp}$  is generic potentially crystalline and let  $\mathbf{r}(\rho_{\wp})$  be the Weil-Deligne representation associated to  $\rho_{\wp}$ . It admits a decomposition  $\mathbf{r}(\rho_{\wp}) \cong \bigoplus_{i=1}^{r} \mathbf{r}_{i}$  by absolutely irreducible Weil-Deligne representations  $\mathbf{r}_{i}$  (which are distinct as  $\rho_{\wp}$  is generic). Each ordering of the  $\mathbf{r}_{i}$  defines a partial flag  $\mathscr{F}$  on  $\mathbf{r}(\rho_{\wp})$ , and we let  $P \supseteq B$  be the associated parabolic subgroup. We say  $\mathscr{F}$  is a P-filtration on  $\mathbf{r}(\rho_{\wp})$ , or a refinement of  $\rho_{\wp}$ . By Fontaine's theory, the filtration  $\mathscr{F}$  induces a P-filtration  $\mathscr{F}_{\tau}$  on  $D_{\mathrm{dR}}(\rho_{\wp})_{\tau} := D_{\mathrm{dR}}(\rho_{\wp}) \otimes_{(F_{\wp} \otimes_{\mathbb{Q}_{p}} E)} (F_{\wp} \otimes_{F_{\wp}, \tau} E)$  for each  $\tau : F_{\wp} \hookrightarrow E$ . Let Fil $_{\tau}$  be the full flag of the Hodge filtration on  $D_{\mathrm{dR}}(\rho_{\wp})_{\tau}$ . The relative position of the partial flag  $\mathscr{F}_{\tau}$  and the full flag Fil $_{\tau}$  is measured by an element  $w \in \mathscr{W}_{P}^{\max}$ , where  $\mathscr{W}_{P}^{\max}$  is the set of maximal length representatives in the Weyl group  $\mathscr{W} \cong S_{n}$  of GL<sub>n</sub> of the *right* cosets  $\mathscr{W}_{L_{P}} \setminus \mathscr{W}$  (here  $\mathscr{W}_{L_{P}}$  is the Weyl group of  $L_{P}$ ). More precisely, fixing a basis of  $D_{\mathrm{dR}}(\rho_{\wp})_{\tau}$  over E, then  $\mathscr{F}_{\tau}$  (resp. Fil $_{\tau}$ ) corresponds to an element in  $\mathrm{GL}_{n}/P$  (resp.  $\mathrm{GL}_{n}/B$ ), still denoted by  $\mathscr{F}_{\tau}$  (resp. Fil $_{\tau}$ ). There exists a unique  $w_{\tau} \in \mathscr{W}_{P}^{\max}$  such that  $(\mathscr{F}_{\tau}, \mathrm{Fil}_{\tau})$  lies in the GL<sub>n</sub>-orbit of  $(1, w_{\tau}) \in \mathrm{GL}_{n}/P \times \mathrm{GL}_{n}/B$ 

<sup>&</sup>lt;sup>1</sup>We also need some mild assumption on  $\rho_v$  for finitely many finite places  $v \nmid p$  of F that we omit in the introduction, see Theorem 7.2.9.

for the diagonal  $\operatorname{GL}_n$ -action (where  $w_{\tau}$  here denotes a lifting of  $w_{\tau} \in \mathscr{W}$  in  $N_{\operatorname{GL}_n}(T)$ ). We let  $w_{\mathscr{F}} := (w_{\tau}) \in \mathscr{W}_{P,F_{\wp}}^{\max} := \prod_{\tau:F_{\wp} \hookrightarrow E} \mathscr{W}_{P}^{\max}.$ 

On the automorphic side, to any  $w \in \mathscr{W}_{P,F_{\omega}}^{\min}$  (defined as  $\mathscr{W}_{P,F_{\omega}}^{\max}$  but with "maximal length" replaced by "minimal length"), one can associate as in [13, § 6] a topologically irreducible locally  $\mathbb{Q}_p$ -analytic representation  $C(w, \mathscr{F})$  of  $\mathrm{GL}_n(F_{\wp})$  over E. We refer to loc. cit. and § 3.2.5 for its precise definition. We recall that, when w = 1,  $C(1, \mathscr{F})$  is isomorphic to the locally algebraic representation of  $\operatorname{GL}_n(F_{\wp})$  associated to  $\rho_{\wp}$  by the classical (suitably normalized) local Langlands correspondence (and is actually independent of  $\mathscr{F}$ ) and that if  $\widehat{S}(U^p, E)[\mathfrak{m}_{\rho}]^{\text{lalg}} \neq 0$ , then  $\widehat{S}(U^p, E)[\mathfrak{m}_{\rho}]^{\text{lalg}} \cong C(1, \mathscr{F})^{\oplus m}$  for some  $m \in \mathbb{Z}_{\geq 1}$ . Let  $w_0 \in \mathscr{W}_{F_{\wp}} := \prod_{\tau: F_{\wp} \to E} \mathscr{W}$  be the element of maximal length, the following theorem establishes several new cases of [14, Conj. 5.3].

**Theorem 1.3** (cf. Theorem 7.2.20). Assume Hypothesis 1.1,  $\widehat{S}(U^p, E)[\mathfrak{m}_{\rho}]^{\text{lalg}} \neq 0$ , and that  $\rho_{\omega}$ is generic potentially crystalline with distinct Hodge-Tate weights. Assume moreover

(\*) any two factors 
$$\operatorname{GL}_{n_i}$$
 in  $L_P = \begin{pmatrix} \operatorname{GL}_{n_1} & \cdots & 0 \\ \vdots & \ddots & \vdots \\ 0 & \cdots & \operatorname{GL}_{n_r} \end{pmatrix}$  with  $n_i > 1$  (if they exist) are not

adjacent.

Then for  $w \in \mathscr{W}_{P,F_{\wp}}^{\min}$  we have a  $\operatorname{GL}_{n}(F_{\wp})$ -equivariant injection  $C(w,\mathscr{F}) \hookrightarrow \widehat{S}(U^{p},E)^{\operatorname{an}}_{\overline{\rho}}[\mathfrak{m}_{\rho}]$  if and only if  $ww_0 \ge w_{\mathscr{F}}$  for the Bruhat order<sup>2</sup>.

The case P = B was proved in [19] (see also [14], [36] for related work). When  $P \neq B$ , almost nothing was known, except very partial results in [38]. Note that, if there is at most one  $\mathbf{r}_i$  with  $\dim_E \mathbf{r}_i > 1$ , then the assumption (\*) is empty and [14, Conj. 5.3] is proved for any refinement  $\mathscr{F}$  of  $\rho_{\wp}$ . The technical assumption (\*) comes from some properties of the geometry of Bernstein eigenvarieties (see the discussion at the end of this introduction). Without (\*), we still have some partial results towards [14, Conj. 5.3], for instance the "only if" part in the conclusion of Theorem 1.3 holds without any assumption on P.

When P = B, both Theorem 1.2 and Theorem 1.3 followed from an extensive study of eigenvarieties (and of the corresponding patched eigenvarieties and trianguline varieties) in [17], [18], [19]. We follow the same strategy in this work.

Let us start by defining the global Bernstein eigenvalueues. We note that  $\begin{pmatrix} \operatorname{GL}_{n_1} & \cdots & 0 \\ \vdots & \ddots & \vdots \\ 0 & \cdots & \operatorname{GL}_{n_r} \end{pmatrix}$ .

We fix  $\Omega = \prod_{i=1}^{r} \Omega_i$  a cuspidal Bernstein component of  $L_P(F_{\wp}) \cong \prod_{i=1}^{r} \operatorname{GL}_{n_i}(F_{\wp})$ , and denote by  $\mathcal{Z}_{\Omega} = \bigotimes_{i=1}^{r} \mathcal{Z}_{\Omega_{i}}$  the corresponding Bernstein centre (over E). Thus a point of  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}}$ corresponds to an irreducible smooth cuspidal representation of  $L_P(F_{\wp})$ , and we frequently use the associated representation to denote the point. Let  $\mathcal{Z}_0 := Z_{L_P}(\mathcal{O}_{F_{\varphi}})$  (where  $Z_{L_P}$  is the center of  $L_P$ ) and  $\widehat{\mathcal{Z}}_0$  be the rigid space over E parametrizing continuous characters of  $\mathcal{Z}_0$ . We fix a uniformizer  $\varpi$  of  $F_{\wp}$ . For a continuous character  $\chi$  of  $(\mathcal{O}_{F_{\wp}}^{\times})^{\oplus s}$  with  $s \in \mathbb{Z}_{\geq 1}$ , we denote by  $\chi_{\varpi}$  the character of  $(F_{\wp}^{\times})^{\oplus s}$  such that  $\chi_{\varpi}|_{(\mathcal{O}_{F_{\wp}}^{\times})^{\oplus r}} = \chi$  and  $\chi_{\varpi}((\overline{\varpi}^{k_1}, \ldots, \overline{\varpi}^{k_s})) = 1$  for any  $k_1, \ldots, k_s \in \mathbb{Z}$ .

<sup>&</sup>lt;sup>2</sup>Note that for  $w \in \mathscr{W}_{F_{\varphi}}, w \in \mathscr{W}_{P,F_{\varphi}}^{\min}$  is equivalent to  $ww_0 \in \mathscr{W}_{P,F_{\varphi}}^{\max}$ .

Finally we also fix  $\lambda = (\lambda_i)_{1 \le i \le n} = (\lambda_{i,\tau})_{\substack{1 \le i \le n \\ \tau: F_{\varphi} \hookrightarrow E}}$  an integral *P*-dominant weight of  $\operatorname{GL}_n(F_{\varphi})$ . The following theorem summarizes some key features of Bernstein eigenvarieties:

**Theorem 1.4** (cf. § 3.2). For each  $(\Omega, \lambda)$  as above there is a rigid analytic space  $\mathcal{E}_{\Omega,\lambda}(U^p, \overline{\rho})$  over E and an injection of rigid spaces over E

$$\iota_{\Omega,\lambda}: \mathcal{E}_{\Omega,\lambda}(U^p,\overline{\rho}) \hookrightarrow (\operatorname{Spf} \mathbb{T}(U^p)_{\overline{\rho}})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0} \ \big( \hookrightarrow (\operatorname{Spf} R_{\overline{\rho},\mathcal{S}})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0} \big)$$

satisfying the following properties:

(1) the induced morphism  $\mathcal{E}_{\Omega,\lambda}(U^p,\overline{\rho}) \to \widehat{\mathcal{Z}}_0$  is locally finite;

(2) a point  $(\eta, \pi_{L_P}, \chi) \in (\operatorname{Spf} \mathbb{T}(U^p)_{\overline{\rho}})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_0$  lies in  $\mathcal{E}_{\Omega,\lambda}(U^p, \overline{\rho})$  if and only if there exists an injection of locally  $\mathbb{Q}_p$ -analytic representations of  $L_P(F_{\wp})$  (where  $\mathfrak{m}_{\eta}$  denotes the maximal ideal of  $\mathbb{T}(U^p)_{\overline{\rho}}[1/p]$  associated to  $\eta$ ):

$$\pi_{L_P} \otimes_E (\chi_{\varpi} \circ \det_{L_P}) \otimes_E L(\lambda)_P \longleftrightarrow J_P(\widehat{S}(U^p, E)^{\mathrm{an}}_{\overline{\rho}})[\mathfrak{m}_\eta]$$

where  $J_P(\cdot)$  is Emerton's Jacquet functor for P,  $L(\lambda)_P$  is the algebraic representation of  $L_P(F_{\wp})$ of highest weight  $\lambda$  (with respect to  $B \cap L_P$ ), and  $\det_{L_P} : L_P(F_{\wp}) \to Z_{L_P}(F_{\wp})$  is the determinant map;

(3)  $\mathcal{E}_{\Omega,\lambda}(U^p,\overline{\rho})$  is equidimensional of dimension  $[F_{\wp}:\mathbb{Q}_p]r;$ 

(4) the set of classical points, i.e. the points  $(\eta, \pi_{L_P}, \chi) \in \mathcal{E}_{\Omega,\lambda}(U^p, \overline{\rho})$  such that

$$\pi_{L_P} \otimes_E (\chi_{\varpi} \circ \det_{L_P}) \otimes_E L(\lambda)_P \hookrightarrow J_P(\widehat{S}(U^p, E)^{\mathrm{lalg}}_{\overline{\rho}})[\mathfrak{m}_{\eta}],$$

is Zariski-dense in  $\mathcal{E}_{\Omega,\lambda}(U^p,\overline{\rho})$ .

Similar spaces have been constructed in [53] (see also [2] for a construction via overconvergent cohomology, and also [68]), but the new feature in Theorem 1.4 is that we take into account the action of the full Bernstein centre (rather than just the action of  $Z_{L_P}(F_{\wp})$ ), obtained by applying Bushnell-Kutzko's theory of types. This allows to parametrize the full  $L_P(F_{\wp})$ -action, which is particularly important for our applications. When P = B, we have an isomorphism  $\widehat{\mathcal{Z}}_0 \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \cong \widehat{T(F_{\wp})}$  (= continuous characters of  $T(F_{\wp})$ ) and we can show that all (reduced) varieties  $\mathcal{E}_{\Omega,\lambda}(U^p,\overline{\rho})^{\operatorname{red}}$  are isomorphic to the (finite slope) reduced eigenvariety  $\mathcal{E}(U^p,\overline{\rho})^{\operatorname{red}}$ , see Remark 3.2.6.

We next discuss *p*-adic families of Galois representations on  $\mathcal{E}_{\Omega,\lambda}(U^p,\overline{\rho})$ . For a point  $x_i$  of  $(\operatorname{Spec} \mathcal{Z}_{\Omega_i})^{\operatorname{rig}}$   $(i \in \{1, \ldots, r\})$ , we denote by  $\pi_{x_i}$  the associated irreducible supercuspidal representation of  $\operatorname{GL}_{n_i}(F_{\wp})$  over E and we let  $\mathbf{r}_{x_i} := \operatorname{rec}(\pi_{x_i})$  be the associated (absolutely irreducible) Weil representation via the classical local Langlands correspondence (normalized as in [49]). The Weil representation  $\mathbf{r}_{x_i}$  corresponds to a Deligne-Fontaine module  $\operatorname{DF}_{x_i}$  (see for instance [21, Prop. 4.1]), which by Berger's theory [6, Thm. A] corresponds in turn to a *p*-adic differential equation  $\Delta_{x_i}$ , i.e. a  $(\varphi, \Gamma)$ -module of rank  $n_i$  over  $\mathcal{R}_{k(x_i),F_{\wp}}$  which is de Rham of constant Hodge-Tate weight 0 (here  $k(x_i)$  is the residue field at  $x_i$  and  $\mathcal{R}_{k(x_i),F_{\wp}}$  is the Robba ring associated to  $F_{\wp}$  with  $k(x_i)$ -coefficients). Let  $x := (\eta, (x_i), \chi = (\chi_i)) \in \mathcal{E}_{\Omega,\lambda}(U^p, \overline{\rho})$ , the image of  $\eta$  via the injection ( $\operatorname{Spf} \mathbb{T}(U^p)_{\overline{\rho}}$ )^{\operatorname{rig}} \hookrightarrow (\operatorname{Spf} R\_{\overline{\rho},\mathcal{S}})^{\operatorname{rig}} corresponds to a continuous representation  $\rho_x : \operatorname{Gal}(\overline{F}/F) \to \operatorname{GL}_n(k(x))$  over the residue field k(x) of  $\mathcal{E}_{\Omega,\lambda}(U^p, \overline{\rho})$  at x. We put  $\rho_{x,\wp} := \rho_x|_{\operatorname{Gal}_{F_{\wp}}}$ .

**Theorem 1.5** (cf. Theorem 3.2.16, Proposition 3.2.17). Let x as above.

(1) The  $(\varphi, \Gamma)$ -module  $D_{\mathrm{rig}}(\rho_{x,\wp})$  associated to  $\rho_{x,\wp}$  admits a P-filtration Fil<sub>•</sub>  $D_{\mathrm{rig}}(\rho_{x,\wp})$  by saturated  $(\varphi, \Gamma)$ -submodules of  $D_{\mathrm{rig}}(\rho_{x,\wp})$  such that for  $i = 1, \ldots, r$ :

$$\left(\operatorname{gr}_{i} D_{\operatorname{rig}}(\rho_{x,\wp})\right)[1/t] \cong \left(\Delta_{x_{i}}^{\prime} \otimes_{\mathcal{R}_{k(x),F_{\wp}}} \mathcal{R}_{k(x),F_{\wp}}(\chi_{i,\varpi})\right)[1/t],$$
(1.1)

where  $\mathcal{R}_{k(x),F_{\wp}}(\delta)$  denotes the rank one  $(\varphi, \Gamma)$ -module associated to a continuous character  $\delta$  and  $\Delta'_{x_i} := \Delta_{x_i} \otimes_{\mathcal{R}_{k(x),F_{\wp}}} \mathcal{R}_{k(x),F_{\wp}}(\operatorname{unr}(q^{-s_{i-1}+(1-n_i)/2})), q$  being the cardinality of the residue field of  $F_{\wp}, s_{i-1} := r_1 + \cdots + r_{i-1}$  and  $\operatorname{unr}(a)$  denoting the unramified character of  $F_{\wp}^{\times}$  sending any uniformizer to a.

(2) For  $\tau: F_{\wp} \hookrightarrow E$ , the Sen  $\tau$ -weights of  $\rho_{x,\wp}$  are given by  $\{h_{j_i,\tau} + \operatorname{wt}(\chi_i)_{\tau}\}_{\substack{1 \leq i \leq r \\ s_{i-1}+1 \leq j_i \leq s_i}}$ , where  $\operatorname{wt}(\chi')_{\tau}$  denotes the  $\tau$ -weight of  $\chi'$  for a continuous character  $\chi'$  of  $\mathcal{O}_{F_{\wp}}^{\times}$  (or of  $F_{\wp}^{\times}$ ).

We call a filtration on a  $(\varphi, \Gamma)$ -module satisfying the property (1.1) an  $\Omega$ -filtration, and call  $(\{\Delta_{x_i}\}, \{\chi_i\})$  a parameter of the  $\Omega$ -filtration. Theorem 1.5 (1) gives an analogue of the fact that p-adic Galois representations over eigenvarieties are trianguline. The proof of Theorem 1.5 (1) is based on an interpolation result for  $\Omega$ -filtrations in families given in § A.1, which is an analogue of the theory of global triangulation of [60], [67], [4].

We now define rigid analytic spaces which are (local) Galois avatars of Bernstein eigenvarieties and analogues of the trianguline variety of [52], [17] when P = B. They parametrize Galois representations admitting an  $\Omega$ -filtration. Let  $\overline{\rho}_{\wp} := \overline{\rho}|_{\operatorname{Gal}_{F_{\wp}}}$  and  $\mathbf{h} := (\mathbf{h}_i)_{i=1,\dots,n} := (h_{i,\tau})_{\substack{i=1,\dots,n\\ \tau: F_{\wp} \hookrightarrow E}}$ with  $h_{i,\tau} = \lambda_{i,\tau} - i + 1$ . Note that the weight  $\mathbf{h}$  is strictly P-dominant. Let  $R_{\overline{\rho}_{\wp}}$  be the universal framed deformation ring of  $\overline{\rho}_{\wp}$ . Define  $U_{\Omega,\mathbf{h}}(\overline{\rho}_{\wp})$  as the subset of  $(\operatorname{Spf} R_{\overline{\rho}_{\wp}})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0}}$ which consists of the points  $(\varrho, (x_i), (\chi_i))$  such that:

- the parameter  $((x_i), (\chi_i))$  is generic in the sense of § 4.2;
- $D_{\rm rig}(\varrho)$  admits an  $\Omega$ -filtration {Fil<sub>i</sub>  $D_{\rm rig}(\varrho)$ } such that one has embeddings

$$\operatorname{gr}_i D_{\operatorname{rig}}(\varrho) \otimes_{\mathcal{R}_{k(x), F_{\wp}}} \mathcal{R}_{k(x), F_{\wp}}(\chi_{i, \varpi}^{-1}) \hookrightarrow \Delta_{x_i} \otimes_{\mathcal{R}_{k(x), F_{\wp}}} \mathcal{R}_{k(x), F_{\wp}}(z^{\mathbf{h}_{s_i}})$$

where the image has Hodge-Tate weights  $(\mathbf{h}_{s_{i-1}+1}, \ldots, \mathbf{h}_{s_i})$  (here  $z^{\mathbf{h}_{s_i}} := \prod_{\tau: F_{\wp} \hookrightarrow E} \tau(z)^{h_{s_i,\tau}}$ ).

We define  $X_{\Omega,\mathbf{h}}(\overline{\rho}_{\wp})$  to be the (reduced) Zariski-closure of  $U_{\Omega,\mathbf{h}}(\overline{\rho}_{\wp})$  in  $(\operatorname{Spf} R_{\overline{\rho}_{\wp}})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$  and denote by  $\iota_{\Omega,\mathbf{h}}: X_{\Omega,\mathbf{h}}(\overline{\rho}_{\wp}) \hookrightarrow (\operatorname{Spf} R_{\overline{\rho}_{\wp}})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$  the closed embedding. We call  $X_{\Omega,\mathbf{h}}(\overline{\rho}_{\wp})$  a *Bernstein paraboline variety*. When P = B, one can check that  $X_{\Omega,\mathbf{h}}(\overline{\rho}_{\wp})$  is isomorphic to the trianguline variety  $X_{\operatorname{tri}}^{\Box}(\overline{\rho})$  of [17] (see Remark 4.2.4 (1)). It also has similar geometric properties as  $X_{\operatorname{tri}}^{\Box}(\overline{\rho})$  (compare the following theorem with [17, Thm. 2.6]):

**Theorem 1.6** (cf. Theorem 4.2.5). (1) The rigid analytic space  $X_{\Omega,h}(\overline{\rho}_{\wp})$  is equidimensional of dimension

$$n^2 + [F_{\wp}:\mathbb{Q}_p]\Big(\frac{n(n-1)}{2} + r\Big).$$

(2) The set  $U_{\Omega,h}(\overline{\rho}_{\omega})$  is Zariski-open and Zariski-dense in  $X_{\Omega,h}(\overline{\rho}_{\omega})$ .

(3) The rigid analytic space  $U_{\Omega,h}(\overline{\rho}_{\wp})$  is smooth over E, and the natural morphism  $U_{\Omega,h}(\overline{\rho}_{\wp}) \to (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$  is smooth.

Note that, differently from the trianguline case, the proof of Theorem 1.6 crucially uses Kisin's results on potentially crystalline deformation rings ([63]).

Similarly as in [17], the Bernstein paraboline varieties are closely related to a *patched* version of the Bernstein eigenvarieties. Recall that under Hypothesis 1.1, it was constructed in [26] a *p*-adic unitary Banach space representation  $\Pi_{\infty}$  of  $\operatorname{GL}_n(F_{\wp})$  equipped with an action of a certain patched Galois deformation ring  $R_{\infty} \cong R^p_{\infty} \widehat{\otimes}_{\mathcal{O}_E} R_{\overline{\rho}_{\wp}}$  (commuting with  $\operatorname{GL}_n(F_{\wp})$ ). There is an ideal  $\mathfrak{a}$  of  $R_{\infty}$ such that one has a surjection  $R_{\infty}/\mathfrak{a} \twoheadrightarrow R_{\overline{\rho},\mathcal{S}}$  and an isomorphism  $\Pi_{\infty}[\mathfrak{a}] \cong \widehat{S}(U^p, E)_{\overline{\rho}}$ . Applying the construction of Bernstein eigenvarieties with  $\widehat{S}(U^p, E)$  replaced by  $\Pi_{\infty}$ , we can construct an embedding of rigid analytic spaces

$$\iota_{\Omega,\lambda} : \mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho}) \longleftrightarrow (\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0} \cong (\operatorname{Spf} R^{p}_{\infty})^{\operatorname{rig}} \times (\operatorname{Spf} R_{\overline{\rho}_{\wp}})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$$
(1.2)

which satisfies the properties in the following theorem:

**Theorem 1.7** (cf. § 3.3). (1) A point  $(\mathfrak{m}_y, \pi_{L_P}, \chi) \in (\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0}$  lies in  $\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})$  if and only if one has an  $L_P(F_{\wp})$ -equivariant embedding

$$\pi_{L_P} \otimes_E (\chi_{\varpi} \circ \det_{L_P}) \otimes_E L(\lambda)_P \hookrightarrow J_P(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})[\mathfrak{m}_y]$$

where  $\Pi_{\infty}^{R_{\infty}-an}$  denotes the  $R_{\infty}$ -analytic vectors in the sense of [17, § 3.1].

(2) The rigid analytic space  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\bar{\rho})$  is reduced and equidimensional of dimension

$$\dim(\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}} + n^2 + [F_{\wp} : \mathbb{Q}_p] \Big( \frac{n(n-1)}{2} + r \Big).$$

(3) There is a natural morphism of rigid spaces

$$\mathcal{E}_{\Omega,\lambda}(U^p) \longrightarrow \mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho}) \times_{(\operatorname{Spf} R_{\infty})^{\operatorname{rig}}} (\operatorname{Spf} R_{\overline{\rho},\mathcal{S}})^{\operatorname{rig}}$$

which is bijective on points (where the morphism  $(\operatorname{Spf} R_{\overline{\rho},S})^{\operatorname{rig}} \to (\operatorname{Spf} R_{\infty})^{\operatorname{rig}}$  is induced by  $R_{\infty}/\mathfrak{a} \twoheadrightarrow R_{\overline{\rho},S}$ ).

(4) The embedding (1.2) factors through

$$\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho}) \longleftrightarrow (\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}} \times \jmath(X_{\Omega,h}(\overline{\rho}_{\wp}))$$
(1.3)

(where j is a certain shift of the natural embedding, see (3.32)) and induces an isomorphism between  $\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})$  and a union of irreducible components of  $(\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}} \times j(X_{\Omega,h}(\overline{\rho}_{\wp}))$  equipped with the reduced closed rigid subspace structure.

To gain a better understanding of the embedding (1.3), we are led to study the local geometry of Bernstein paraboline varieties. As in the trianguline case (when P = B), it is closely related to certain schemes appearing in parabolic generalizations of Grothendieck's and Springer's resolution of singularities. We now define these schemes. For a closed algebraic subgroup H of  $\operatorname{GL}_n$  (seen over  $F_{\wp}$ ), put  $H_{\wp} := (\operatorname{Res}_{\mathbb{Q}_p}^{F_{\wp}} H) \times_{\operatorname{Spec}_{\mathbb{Q}_p}} \operatorname{Spec} E$ . Likewise, for a Lie subalgebra  $\mathfrak{h}$  of the Lie algebra  $\mathfrak{gl}_n$ over  $F_{\wp}$ , put  $\mathfrak{h}_{\wp} := \mathfrak{h} \otimes_{\mathbb{Q}_p} E$ . Let  $\tilde{\mathfrak{g}}_{\wp}$  (resp.  $\tilde{\mathfrak{g}}_{P,\wp}$ ) be the closed E-subscheme of  $(\operatorname{GL}_{n,\wp}/B_{\wp}) \times \mathfrak{gl}_{n,\wp}$ (resp. of  $(\operatorname{GL}_{n,\wp}/P_{\wp}) \times \mathfrak{gl}_{n,\wp}$ ) defined by:

$$\{(gB_{\wp},\psi)\in (\operatorname{GL}_{n,\wp}/B_{\wp})\times\mathfrak{gl}_{n,\wp} \mid \operatorname{Ad}(g^{-1})\psi\in\mathfrak{b}_{\wp}\}$$

$$\left(\text{resp. } \{(gP_{\wp},\psi)\in (\operatorname{GL}_{n,\wp}/P_{\wp})\times \mathfrak{gl}_{n,\wp} \mid \operatorname{Ad}(g^{-1})\psi\in \mathfrak{r}_{P,\wp}\}\right),$$

where  $\mathfrak{b} \subset \mathfrak{gl}_n$  (resp.  $\mathfrak{r}_P \cong \mathfrak{n}_P \rtimes \mathfrak{z}_{L_P}$ ) is the Lie algebra of B (resp. of the full radical  $N_P \rtimes Z_{L_P}$  of  $P \cong N_P \rtimes L_P$ ). Put

$$X_{P,\wp} := \tilde{\mathfrak{g}}_{P,\wp} \times_{\mathfrak{gl}_{n,\wp}} \tilde{\mathfrak{g}}_{\wp}.$$

We define a morphism  $\pi$  as the composition:

$$\pi: X_{P,\wp} \hookrightarrow (\operatorname{GL}_{n,\wp}/P_{\wp}) \times (\operatorname{GL}_{n,\wp}/B_{\wp}) \times \mathfrak{gl}_{n,\wp} \longrightarrow (\operatorname{GL}_{n,\wp}/P_{\wp}) \times (\operatorname{GL}_{n,\wp}/B_{\wp})$$
$$= \amalg_{w \in \mathscr{W}_{P,F_{\wp}}^{\max}} \operatorname{GL}_{n,\wp}(1,w)(P_{\wp} \times B_{\wp})$$

where the second map is the canonical projection. For  $w \in \mathscr{W}_{P,F_{\wp}}^{\max}$ , let  $U_w := \operatorname{GL}_{n,\wp}(1,w)(P_{\wp} \times B_{\wp})$ and  $X_w$  the reduced closed subscheme of  $X_{P,\wp}$  defined as the Zariski-closure of  $\pi^{-1}(U_w)$ . One can show that  $\{X_w\}_{w \in \mathscr{W}_{P,F_{\wp}}^{\max}}$  are the irreducible components of  $X_{P,\wp}$  and that  $X_{P,\wp}$  is equidimensional of dimension

$$[F_{\wp}:\mathbb{Q}_p]\frac{n(n-1)}{2} + \dim \mathfrak{r}_{P,\wp} = [F_{\wp}:\mathbb{Q}_p]\Big(\frac{n(n-1)}{2} + r\Big) + \dim \mathfrak{n}_{P,\wp}.$$

The following geometric property of  $X_w$  is particularly important for our applications:

**Theorem 1.8** (cf. Theorem 5.3.1). Let  $w \in \mathscr{W}_{P,F_{\wp}}^{\max}$  and  $x = (g_1P_{\wp}, g_2B_{\wp}, 0) \in X_w \hookrightarrow (\operatorname{GL}_{n,\wp}/P_{\wp}) \times (\operatorname{GL}_{n,\wp}/B_{\wp}) \times \mathfrak{gl}_{n,\wp}$ , then the scheme  $X_w$  is unibranch at x.

When P = B, it was showed in [19, Thm. 2.3.6] that the whole scheme  $X_w$  is normal. When  $P \neq B$ , the geometry of  $X_w$  appears seriously more difficult and we don't know if  $X_w$  is normal. The proof of Theorem 1.8 is inspired by a result of Le, Le Hung, Morra and Levin in the setting of deformation rings ([65, Lemma 3.4.8]) and a priori only works for those points  $x \in X_w$  with  $\mathfrak{gl}_{n,\wp}$ -factor zero (as in Theorem 1.8).

Let  $x = (\varrho, (x_i), (\chi_i)) \in X_{\Omega,\mathbf{h}}(\overline{\rho}_{\wp})$  such that  $\varrho$  is almost de Rham (in the sense of [44]) with distinct Sen weights. We fix an  $F_{\wp} \otimes_{\mathbb{Q}_p} k(x)$ -linear isomorphism  $\alpha : F_{\wp} \otimes_{\mathbb{Q}_p} k(x) \xrightarrow{\sim} D_{\mathrm{pdR}}(\varrho) = (B_{\mathrm{pdR}} \otimes_{\mathbb{Q}_p} \varrho)^{\mathrm{Gal}_{F_{\wp}}}$  where  $B_{\mathrm{pdR}}$  is the ring of [44, § 4.3] (see also § 6.1). By an analogue of Theorem 1.5 (1) for the variety  $X_{\Omega,\mathbf{h}}(\overline{\rho}_{\wp})$  instead of  $\mathcal{E}_{\Omega,\lambda}(U^p,\overline{\rho})$ , the  $(\varphi,\Gamma)$ -module  $D_{\mathrm{rig}}(\varrho)[\frac{1}{t}]$  over  $\mathcal{R}_{k(x),F_{\wp}}[\frac{1}{t}]$  admits a *P*-filtration  $\mathscr{F}_{\varrho}$  which induces a *P*-filtration (still denoted)  $\mathscr{F}_{\varrho}$  on  $D_{\mathrm{pdR}}(\varrho) \cong$  $(B_{\mathrm{pdR}} \otimes_{B_{\mathrm{dR}}} W_{\mathrm{dR}}(D_{\mathrm{rig}}(\varrho)[\frac{1}{t}]))^{\mathrm{Gal}_{F_{\wp}}}$  (see for instance [19, Lemma 3.3.5 (ii)] for  $W_{\mathrm{dR}}(-)$ , see also § 6.2). We use Fil\_{\varrho} to denote the complete flag given by the Hodge filtration on  $D_{\mathrm{pdR}}(\varrho)$ . Using the framing  $\alpha$ , the filtration  $\mathscr{F}_{\varrho}$  (resp. Fil\_{\varrho}) corresponds to an element still denoted by  $\mathscr{F}_{\varrho}$  (resp. Fil\_{\varrho}) in  $\mathrm{GL}_{n,\wp}/P_{\wp}$  (resp. in  $\mathrm{GL}_{n,\wp}/B_{\wp}$ ). Finally, let  $\nu_{\mathrm{pdR}}$  be Fontaine's nilpotent operator on  $D_{\mathrm{pdR}}(\varrho)$ . As both  $\mathscr{F}_{\varrho}$  and Fil\_{\varrho} are stabilized by  $\nu_{\mathrm{pdR}}$ , we have

$$y_{\mathrm{pdR}} := (\mathscr{F}_{\varrho}, \mathrm{Fil}_{\varrho}, \nu_{\mathrm{pdR}}) \in X_{P,\wp} \hookrightarrow (\mathrm{GL}_{n,\wp}/P_{\wp}) \times (\mathrm{GL}_{n,\wp}/B_{\wp}) \times \mathfrak{gl}_{n,\wp}.$$
(1.4)

There exists thus  $w_{\mathscr{F}_{\varrho}} \in \mathscr{W}_{P,F_{\wp}}^{\max}$  such that  $y_{\mathrm{pdR}} \in \pi^{-1}(U_{w_{\mathscr{F}_{\varrho}}})$ . In fact,  $w_{\mathscr{F}_{\varrho}}$  is independent of the choice of  $\alpha$  and is equal to the element  $w_{\mathscr{F}}$  defined above Theorem 1.3 if  $\varrho = \rho_{\wp}$  and  $\mathscr{F}_{\varrho} = \mathscr{F}$  (note that we have  $\nu_{\mathrm{pdR}} = 0$  in this case).

By an analogue of Theorem 1.5 (2), one can show that for each  $\tau : F_{\wp} \hookrightarrow E$ , the set of  $\tau$ -Sen weights of  $\varrho$  is given by  $\{k_{x,j,\tau} := h_{j,\tau} + \operatorname{wt}(\chi_{i(j)})_{\tau}\}_{j=1,\ldots,n}$  where  $i(j) \in \{1,\ldots,r\}$  is such that  $s_{i(j)-1} < j \leq s_{i(j)}$ . There exist then a unique element  $w_x = (w_{x,\tau}) \in \mathscr{W}_{P,F_{\wp}}^{\min}$  and a strictly dominant

weight  $\mathbf{h}_x$  (which is in fact given by the (decreasing) Sen weights of  $\varrho$ ) such that  $\mathbf{k}_x = w_x(\mathbf{h}_x)$ where  $\mathbf{k}_x := \{k_{x,j,\tau}\}_{\substack{j=1,\ldots,n\\ \tau: F_{\varphi} \hookrightarrow E}}$ .

By generalizing  $[19, \S 3]$ , we have:

**Theorem 1.9** (cf. § 6.4). Keep the above notation and assume  $\rho$  is de Rham (equivalently  $\nu_{pdR} = 0$ ) and  $\mathscr{F}_{\rho}$  is generic (in the sense of (6.5)).

(1) We have  $w_{\mathscr{F}} \leq w_x w_0$ .

(2) There exists a formal scheme  $X_{\varrho,\mathcal{M}_{\bullet}}^{\Box,w_x}$  over E such that the associated reduced formal scheme  $(X_{\varrho,\mathcal{M}_{\bullet}}^{\Box,w_x})^{\mathrm{red}}$  is formally smooth of dimension  $n^2 + \dim \mathfrak{p}_{\wp}$  over the completion  $\widehat{X}_{w_x,y_{\mathrm{pdR}}}$  of  $X_{w_x}$  at  $y_{\mathrm{pdR}}$ , and formally smooth of dimension  $n^2[F_{\wp}:\mathbb{Q}_p]$  over the completion  $\widehat{X}_{\Omega,h}(\overline{\rho}_{\wp})_x$  of  $X_{\Omega,h}(\overline{\rho}_{\wp})_x$  at x:

$$\widehat{X_{\Omega,h}(\overline{\rho}_{\wp})}_{x} \longleftarrow (X_{\varrho,\mathcal{M}_{\bullet}}^{\Box,w_{x}})^{\mathrm{red}} \longrightarrow \widehat{X}_{w_{x},y_{\mathrm{pdR}}}.$$

In particular,  $X_{\Omega,h}(\overline{\rho}_{\omega})$  is unibranch at the point x.

By Theorem 1.9 (2) and Theorem 1.7 (4), we deduce the following "R = T"-type result:

**Corollary 1.10.** Let  $x = (\mathfrak{m}^p, \varrho, \pi_{L_P}, \chi) \in \mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho}) \hookrightarrow (\operatorname{Spf} R_{\infty}^p)^{\operatorname{rig}} \times (\operatorname{Spf} R_{\overline{\rho}_{\wp}})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0}$ . Assume that  $\mathfrak{m}^p$  is a smooth point of  $(\operatorname{Spf} R_{\infty}^p)^{\operatorname{rig}}$  and that  $\varrho$  is generic potentially crystalline with distinct Hodge-Tate weights. Then the embedding (1.3) induces an isomorphism after taking completions at x.

We now discuss the problem of companion points on Bernstein eigenvarieties (resp. on patched Bernstein eigenvarieties, resp. on Bernstein paraboline varieties), which will be crucial to attack the socle conjecture. Let y be a point of  $(\operatorname{Spf} R_{\overline{\rho},S})^{\operatorname{rig}}$  (resp. of  $(\operatorname{Spf} R_{\infty})^{\operatorname{rig}}$ , resp. of  $(\operatorname{Spf} R_{\overline{\rho}_{\varphi}})^{\operatorname{rig}}$ ), and  $\varrho$  be the  $\operatorname{Gal}_{F_{\varphi}}$ -representation associated to y. We assume  $\varrho$  is generic potentially crystalline with distinct Hodge-Tate weights. We let  $\mathbf{h}$  be the (decreasing) Hodge-Tate weights of  $\varrho$  and  $\lambda = (\lambda_{i,\tau})_{\substack{i=1,\dots,n\\ \tau:F_{\varphi}\leftrightarrow E}}$  with  $\lambda_{i,\tau} = h_{i,\tau} + i - 1$  (so  $\lambda$  is dominant with respect to B). Assume  $\mathbf{r}(\varrho) \cong \bigoplus_{i=1}^{r} \mathbf{r}_{x_i}$ with  $\underline{x} = (x_i) \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}}$ . Note that, as  $\varrho$  is generic, the P-filtration  $\mathbf{r}_{x_1} \subset \mathbf{r}_{x_1} \oplus \mathbf{r}_{x_2} \subset \cdots$ on  $\mathbf{r}(\varrho)$  corresponds to a unique  $\Omega$ -filtration on  $D_{\operatorname{rig}}(\varrho)$ , see § 2.3. Consider the point x := $(y, \underline{x}, 1)$  in  $(\operatorname{Spf} R_{\overline{\rho}, S})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0}$  (resp. in  $(\operatorname{Spf} R_{\infty}^p)^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0})$ . As usual, we denote by  $w \cdot \mu$  the dot action on a weight  $\mu$ .

**Conjecture 1.11.** Let  $w \in \mathscr{W}_{\min,F_{\wp}}^{P}$ , then  $x \in \mathcal{E}_{\Omega,w\cdot\lambda}(U^{p},\overline{\rho})$  (resp.  $x \in \mathcal{E}_{\Omega,w\cdot\lambda}^{\infty}(\overline{\rho})$ , resp.  $x \in X_{\Omega,w(h)}(\overline{\rho}_{\wp})$ ) if and only if  $ww_{0} \geq w_{\mathscr{F}_{\rho}}$ .

As  $w \in \mathscr{W}_{\min,F_{\wp}}^{P}$ ,  $w(\mathbf{h})$  is strictly *P*-dominant and  $w \cdot \lambda$  is *P*-dominant, so the corresponding rigid spaces in Conjecture 1.11 are well defined. The reader who is familiar with companion points in the trianguline case may find the statement of Conjecture 1.11 a little strange. Indeed, for the classical eigenvariety  $\mathcal{E}(U^p, \overline{\rho})$  (the case of the patched eigenvariety or of the trianguline variety being similar), there is a canonical embedding  $\mathcal{E}(U^p, \overline{\rho}) \hookrightarrow (\operatorname{Spf} R_{\overline{\rho}, \mathcal{S}})^{\operatorname{rig}} \times \widehat{T(F_{\wp})}$  and the companion points are the distinct points that lie above a same point  $y \in (\operatorname{Spf} R_{\overline{\rho}, \mathcal{S}})^{\operatorname{rig}}$ . In our case however, as there are different rigid spaces depending on  $(\Omega, \lambda)$ , it seems more convenient to fix the point  $x \in (\operatorname{Spf} R_{\overline{\rho}, \mathcal{S}})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0}}$  and let the Bernstein eigenvarieties (together with the embedding  $\iota_{\Omega, \lambda}$ ) vary. See Remark 3.3.12 and Remark 4.3.10 for more details. By Theorem 1.9 (1) and a study of the relation between potentially crystalline deformation spaces and Bernstein paraboline varieties (cf.  $\S$  4.3), we can prove Conjecture 1.11 for Bernstein paraboline varieties:

**Theorem 1.12** (cf. Corollary 6.4.12). Let  $w \in \mathscr{W}_{\min,F_{\widetilde{v}}}^{P}$ , then  $x \in X_{\Omega,w(h)}(\overline{\rho}_{\wp})$  if and only if  $ww_0 \geq w_{\mathscr{F}_{\rho}}$ .

Using Theorem 1.7 (3) and Theorem 1.7 (4), Theorem 1.12 then implies the "only if" part of Conjecture 1.11.

We now move back to the global applications in Theorem 1.2 and Theorem 1.3, both of which are about irreducible constituents in the socle of  $\widehat{S}(U^p, E)^{\operatorname{an}}[\mathfrak{m}_{\rho}]$ . Let y be the image of  $\mathfrak{m}_{\rho}$  via  $(\operatorname{Spf} R_{\overline{\rho}, \mathcal{S}})^{\operatorname{rig}} \hookrightarrow (\operatorname{Spf} R_{\infty})^{\operatorname{rig}}$ . Let  $\mathfrak{m}_y \subset R_{\infty}[1/p]$  be the associated maximal ideal, then  $\widehat{S}(U^p, E)^{\operatorname{an}}[\mathfrak{m}_{\rho}] \cong \Pi^{R_{\infty}-\operatorname{an}}[\mathfrak{m}_y]$ . In particular, to prove Theorem 1.2 and Theorem 1.3, it suffices to show the same statement with  $\Pi^{R_{\infty}-\operatorname{an}}_{\infty}[\mathfrak{m}_y]$  instead of  $\widehat{S}(U^p, E)^{\operatorname{an}}[\mathfrak{m}_{\rho}]$ . Let  $\mathscr{F}$  be a P-filtration on  $\mathbf{r}(\rho_{\wp})$  as in the discussion above Theorem 1.3. As before let  $\mathbf{h}$  be the (decreasing) Hodge-Tate weights of  $\rho_{\wp}$  and  $\lambda = (\lambda_{i,\tau})$  with  $\lambda_{i,\tau} = h_{i,\tau} + i - 1$ . Let  $\Omega$  be a Bernstein component of  $L_P(F_{\wp})$  and  $\underline{x} = (x_i) \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}}$  such that  $\mathbf{r}_{x_i} \cong \mathbf{r}_i$  for  $i = 1, \cdots, r$  (so  $\Omega$  and  $\underline{x}$  are determined by  $\rho$  and  $\mathscr{F}$ ). We then associate to  $\rho$  and the P-filtration  $\mathscr{F}$  on  $\mathbf{r}(\rho_{\wp})$  a point  $x_{\mathscr{F}} = (y, \underline{x}, 1) \in (\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0}$ . From the locally analytic representation theory, for each  $w \in \mathscr{W}_{P,F_{\wp}}^{\min}$ , we can construct a cycle  $[\mathcal{N}_{w \cdot \lambda, y, \mathscr{F}}] \in Z^{[F^+:\mathbb{Q}]\frac{n(n+1)}{2}}$  (Spec  $\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty}, y}$ ), where  $Z^d(-)$  is the free abelian group generated by the irreducible closed subschemes of codimension d and  $\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty}, y}$  is the completion of  $\mathfrak{X}_{\infty} := (\operatorname{Spf} R_{\infty})^{\operatorname{rig}}$  at the point y, such that:

•  $[\mathcal{N}_{w \cdot \lambda, y, \mathscr{F}}] \neq 0$  if and only if  $C(w, \mathscr{F})$  embeds into  $\Pi_{\infty}^{R_{\infty} - \mathrm{an}}[\mathfrak{m}_{y}]$ .

It is also not difficult to prove the implication  $[\mathcal{N}_{w\cdot\lambda,y,\mathscr{F}}] \neq 0 \Rightarrow x_{\mathscr{F}} \in \mathcal{E}^{\infty}_{\Omega,w\cdot\lambda}(\overline{\rho})$ . It turns out that this implication is in fact an equivalence (which then implies that, in this case, the existence of companion points is equivalent to the existence of companion constituents):

**Proposition 1.13** (cf. Proposition 7.2.3). For  $w \in \mathscr{W}_{P,F_{\wp}}^{\min}$ ,  $[\mathcal{N}_{w \cdot \lambda, y, \mathscr{F}}] \neq 0$  if and only if  $x_{\mathscr{F}} \in \mathcal{E}_{\Omega, w \cdot \lambda}^{\infty}(\overline{\rho})$ .

We first discuss the proof of Theorem 1.2. The assumption (2) in *loc. cit.* guarantees that there exist a parabolic subgroup  $P \supseteq B$  of  $\operatorname{GL}_n$  and a P-filtration  $\mathscr{F}$  on  $\mathbf{r}(\rho_{\wp})$  such that the above associated point  $x_{\mathscr{F}} = (y, \underline{x}, 1) \in (\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_0$  lies in a certain patched Bernstein eigenvariety  $\mathcal{E}_{\Omega, w \cdot \lambda}^{\infty}(\overline{\rho})$  for some  $w \in \mathscr{W}_{P, F_{\wp}}^{\min}$ . Using Proposition 1.13 (and a bit of representation theory), one can deduce  $x_{\mathscr{F}} \in \mathcal{E}_{\Omega, \lambda}^{\infty}(\overline{\rho})$ . Then the classicality follows by applying Proposition 1.13 to w = 1 and using that  $C(1, \mathscr{F})$  is locally algebraic.

We now discuss the proof of Theorem 1.3. We henceforth fix a filtration  $\mathscr{F}$  and write  $x := x_{\mathscr{F}}$ ,  $[\mathcal{N}_{w \cdot \lambda, y}] := [\mathcal{N}_{w \cdot \lambda, y, \mathscr{F}}]$ . Note that, as  $\Pi_{\infty}^{R_{\infty} - \mathrm{an}}[\mathfrak{m}_{y}]^{\mathrm{lalg}} \cong \widehat{S}(U^{p}, E)[\mathfrak{m}_{\rho}]^{\mathrm{lalg}}$  is non-zero and isomorphic to a direct sum of copies of  $C(1, \mathscr{F})$ , we have  $[\mathcal{N}_{\lambda, y}] \neq 0$  and  $x \in \mathcal{E}_{\Omega, \lambda}^{\infty}(\overline{\rho})$ . The "only if" part (with no assumption on P) already follows from Proposition 1.13 and Theorem 1.12. So we need to show  $x \in \mathcal{E}_{\Omega, w \cdot \lambda}^{\infty}(\overline{\rho})$  for  $w \in \mathscr{W}_{P, F_{\wp}}^{\min}$  such that  $ww_{0} \geq w_{\mathscr{F}}$ . Let  $\lg(-)$  denotes the length function on Weyl group. The case where  $\lg(w_{\mathscr{F}}) \geq \lg(w_{0}) - 1$  is not very difficult (and holds for any P). We assume in the sequel  $\lg(w_{\mathscr{F}}) \leq \lg(w_{0}) - 2$ . By induction and some similar arguments as in the proof of the "if" part of Theorem 1.12, one is reduced to showing the following statement: • if  $x \in \mathcal{E}^{\infty}_{\Omega, w'w_0 \cdot \lambda}(\overline{\rho})$  for all  $w' \in \mathscr{W}^{\max}_{P, F_{\wp}}$  such that  $w' > w_{\mathscr{F}}$ , then  $x \in \mathcal{E}^{\infty}_{\Omega, w_{\mathscr{F}} w_0 \cdot \lambda}(\overline{\rho})$ .

Thus, assuming  $x \in \mathcal{E}^{\infty}_{\Omega,w'w_0\cdot\lambda}(\overline{\rho})$  for all  $w' \in \mathscr{W}^{\max}_{P,F_{\wp}}, w' > w_{\mathscr{F}}$ , we need to show  $[\mathcal{N}_{w_{\mathscr{F}}w_0\cdot\lambda,y}] \neq 0$ . One important fact is that the cycles  $[\mathcal{N}_{w\cdot\lambda,y}]$  can be related to cycles coming from irreducible components of certain generalized Steinberg varieties. Let  $Z_{P,\wp}$  be the fibre of  $X_{P,\mathfrak{p}}$  at  $0 \in \mathfrak{z}_{L_{P,\wp}}$ via

$$X_{P,\wp} \longrightarrow \mathfrak{z}_{L_P,\wp}, \ (g_1 P_\wp, g_2 B_\wp, \psi) \longmapsto \operatorname{Ad}(g_1^{-1})\psi$$

where  $\overline{(-)}$  means the natural projection  $\mathfrak{r}_{P,\wp} \twoheadrightarrow \mathfrak{z}_{L_P,\wp}$ . One can show that  $Z_{P,\wp}$  is equidimensional with (reduced) irreducible components given by  $\{Z_w := (Z_{P,\wp} \cap X_w)^{\mathrm{red}}\}_{w \in \mathscr{W}_{P,F_\wp}^{\mathrm{max}}}$ . Let  $y_{\mathrm{pdR}}$  be the point of  $X_{P,\wp}$  associated to  $\rho_{\wp}$  and  $\mathscr{F}$  as in (1.4). As  $\nu_{\mathrm{pdR}} = 0$ ,  $y_{\mathrm{pdR}} \in Z_{P,\wp}$ . Similarly as in Theorem 1.8, one can prove that for  $w \in \mathscr{W}_{P,F_\wp}^{\mathrm{max}}$ , if  $y_{\mathrm{pdR}} \in Z_w$ , then  $Z_w$  is unibranch at  $y_{\mathrm{pdR}}$  (cf. Theorem 5.4.3). Using Theorem 1.9 (2), for  $w \in \mathscr{W}_{P,F_\wp}^{\mathrm{max}}$ , one can then associate to the completion  $\widehat{\mathcal{O}}_{Z_w,y_{\mathrm{pdR}}}$  of  $Z_w$  at  $y_{\mathrm{pdR}}$  a unique irreducible cycle  $[\mathfrak{Z}_{w,y}] \in Z^{[F^+:\mathbb{Q}]\frac{n(n+1)}{2}}(\mathrm{Spec}\,\widehat{\mathcal{O}}_{\mathfrak{X}_\infty,y})$ . We have

•  $\mathfrak{Z}_{w,y} \neq 0$  if and only if  $w \geq w_{\mathscr{F}}$ .

By results on the characteristic cycles associated to generalized Verma modules (that we couldn't really find in the literature and that we prove in  $\S$  A.2), we have the following statements:

- $[\mathcal{N}_{w_{\mathscr{F}}w_0\cdot\lambda,y}]\in\mathbb{Z}_{\geq 0}\mathfrak{Z}_{w_{\mathscr{F}},y};$
- if x is a smooth point of  $\mathcal{E}^{\infty}_{\Omega, ww_0 \cdot \lambda}(\overline{\rho})$  for  $w \in \mathscr{W}^{\min}_{P, F_{\wp}}$ , there is an integer  $m_y \in \mathbb{Z}_{\geq 1}$  such that in  $Z^{[F^+:\mathbb{Q}]\frac{n(n+1)}{2}}(\operatorname{Spec}\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y})$ :

$$\sum_{\substack{w' \in \mathscr{W}_{P,F_{\wp}}^{\min} \\ w \le w' \le w_{\mathscr{F}} w_0}} b_{w,w'}[\mathcal{N}_{w'\cdot\lambda,y}] = m_y \left(\sum_{\substack{w' \in \mathscr{W}_{P,F_{\wp}}^{\min} \\ w \le w' \le w_{\mathscr{F}} w_0}} b_{w,w'}\mathfrak{Z}_{w'w_0,y}\right)$$
(1.5)

where  $b_{w,w'}$  is the multiplicity of the simple  $U(\mathfrak{g}_{\wp})$ -module  $L(w' \cdot 0)$  of highest weight  $w' \cdot 0$ in the parabolic Verma module  $M_P(w \cdot 0)$  of highest weight  $w \cdot 0$ .

If x is not smooth, we also have an equation similar to (1.5) but with the coefficients  $m_y b_{w,w'}$  on the right hand side replaced by certain non-negative integers that we do not know how to control (the left hand side staying unchanged).

Let us assume  $m_y = 1$  in the rest of the argument for simplicity. To use these equations to deduce  $[\mathcal{N}_{w_{\mathscr{F}}w_0\cdot\lambda,y}] \neq 0$ , we are led to three cases (recall we have assumed  $\lg(w_{\mathscr{F}}) \leq \lg(w_0) - 2$ ), with the third case still resisting without a further assumption on P:

Case (1) is exactly the same as in [19] (in particular we are always in this case when P = B): assume that there exist  $w_1, w_2, w \in \mathscr{W}_{P,F_{\wp}}^{\max}$  such that  $w \ge w_{\mathscr{F}}, \lg(w) = \lg(w_{\mathscr{F}}) + 2, \{w_1, w_2\} = [w_{\mathscr{F}}, w] := \{w' \mid w_{\mathscr{F}} < w' < w\}$  and  $\dim \mathfrak{z}_{L_{P,\wp}}^{w_{\mathscr{F}}w^{-1}} = \dim \mathfrak{z}_{L_{P,\wp}} - 2$ . Under these conditions, by a tangent space argument, one can prove that  $\mathcal{E}_{\Omega,w'w_0\cdot\lambda}(\overline{\rho})$  is smooth at x for  $w' \in \{w_1, w_2, w\}$ . We can deduce from (1.5) equalities:

$$\begin{cases} [\mathcal{N}_{w_1w_0\cdot\lambda,y}] + [\mathcal{N}_{w,\mathscr{F}}w_0\cdot\lambda,y] = \mathfrak{Z}_{w_1,y} + \mathfrak{Z}_{w,\mathscr{F}},y \\ [\mathcal{N}_{w_2w_0\cdot\lambda,y}] + [\mathcal{N}_{w,\mathscr{F}}w_0\cdot\lambda,y] = \mathfrak{Z}_{w_2,y} + \mathfrak{Z}_{w,\mathscr{F}},y \\ [\mathcal{N}_{ww_0\cdot\lambda,y}] + [\mathcal{N}_{w_1w_0\cdot\lambda,y}] + [\mathcal{N}_{w_2w_0\cdot\lambda,y}] + [\mathcal{N}_{w,\mathscr{F}}w_0\cdot\lambda,y] = \mathfrak{Z}_{w,y} + \mathfrak{Z}_{w_1,y} + \mathfrak{Z}_{w_2,y} + \mathfrak{Z}_{w,\mathscr{F}},y. \end{cases}$$
(1.6)

Using that  $\mathfrak{Z}_{w'',y}$  can only have non-negative coefficients in  $[\mathcal{N}_{w'w_0\cdot\lambda,y}]$  for  $w', w'' \in \mathscr{W}_{P,F_{\wp}}^{\max}$  and  $[\mathcal{N}_{w_{\mathscr{F}}w_0\cdot\lambda,y}] \in \mathbb{Z}_{\geq 0}\mathfrak{Z}_{w_{\mathscr{F}},y}$ , it is not difficult to deduce from the equalities in (1.6) that  $[\mathcal{N}_{w_{\mathscr{F}}w_0\cdot\lambda,y}] \neq 0^3$ .

Case (2): assume that there exists  $w > w_{\mathscr{F}}$  with  $\lg(w) = \lg(w_{\mathscr{F}}) + 2$  such that there is a unique  $w_1 \in \mathscr{W}_{P,F_{\wp}}^{\max}$  such that  $w_{\mathscr{F}} < w_1 < w$  (in other words, the Bruhat interval  $[w_{\mathscr{F}}, w]$  is not full in the quotient  $\mathscr{W}_{L_P,F_{\wp}} \setminus \mathscr{W}_{F_{\wp}}$ ). In this case, we deduce from (1.5) equalities:

$$\begin{cases} [\mathcal{N}_{w_1w_0\cdot\lambda,y}] + [\mathcal{N}_{w,\mathscr{F}}w_0\cdot\lambda,y] = \mathfrak{Z}_{w_1,y} + \mathfrak{Z}_{w,\mathscr{F}},y\\ [\mathcal{N}_{ww_0\cdot\lambda,y}] + [\mathcal{N}_{w_1w_0\cdot\lambda,y}] = a_0\mathfrak{Z}_{w,y} + a_1\mathfrak{Z}_{w_1,y} \end{cases}$$

for some  $a_i \in \mathbb{Z}_{\geq 0}$ . Though we don't have more control on the  $a_i$ , these equalities are (again) sufficient to imply  $[\mathcal{N}_{w \not\in w_0 \cdot \lambda, y}] \neq 0$ .

Case (3): assume that, for any  $w \leq w_{\mathscr{F}} w_0$  with  $\lg(w) = \lg(w_{\mathscr{F}} w_0) - 2$ , we have  $\dim \mathfrak{z}_{L_P,\wp}^{w_{\mathscr{F}} w_0 w^{-1}} > \dim \mathfrak{z}_{L_P,\wp} - 2$  and there exist  $w_1, w_2 \in \mathscr{W}_{P,F_{\wp}}^{\min}$  such that  $\{w' \mid w < w' < w_{\mathscr{F}} w_0\} = \{w_1, w_2\}$ . The main difference with Case (1) is that we do not know if  $\mathcal{E}_{\Omega,ww_0\cdot\lambda}^{\infty}(\overline{\rho})$  is smooth at the point x (the tangent space argument collapses because of  $\dim \mathfrak{z}_{L_P,\wp}^{w_{\mathscr{F}} w_0 w^{-1}} > \dim \mathfrak{z}_{L_P,\wp} - 2$ ). Consequently, the third equation in (1.6) has to be replaced by an equation of the form (the two others staying unchanged)

$$[\mathcal{N}_{ww_0\cdot\lambda,y}] + [\mathcal{N}_{w_1w_0\cdot\lambda,y}] + [\mathcal{N}_{w_2w_0\cdot\lambda,y}] + [\mathcal{N}_{w_\mathscr{F}w_0\cdot\lambda,y}] = a_0\mathfrak{Z}_{w,y} + a_1\mathfrak{Z}_{w_1,y} + a_2\mathfrak{Z}_{w_2,y} + a_3\mathfrak{Z}_{w_\mathscr{F},y}]$$

for some  $a_i \in \mathbb{Z}_{\geq 0}$ . Without more control on these coefficients  $a_i$ , this equation together with the first two in (1.6) seem not enough to imply  $[\mathcal{N}_{w,\mathscr{F}}w_0\cdot\lambda,y]\neq 0$ .

The assumption (\*) in Theorem 1.3 is there precisely to avoid Case (3) (cf. Proposition A.3.3, see also Remark A.3.5 for an example of Case (3) for  $GL_4$ ).

One may expect other arithmetic applications of Bernstein eigenvarieties. In fact, the results in this work provide a framework to which many arguments for classical eigenvarieties may be adapted (as what we already do in this paper). In a forthcoming work [50] of Yiqin He, Bernstein eigenvarieties are used to establish some local-global compatibility results on simple  $\mathcal{L}$ -invariants for certain GL<sub>n</sub>-representations attached to Zelevinsky's linked segments (which was previously only known in the trianguline case).

Finally, we remark that in his PhD. thesis [55], Shanxiao Huang proves results that parallel Theorem 1.4, Theorem 1.5, Theorem 1.6 and the global analogue of part (4) of Theorem 1.7 (i.e. a version without patched objects).

## Acknowledgement

This work would not exist without the work of Eugen Hellmann and Benjamin Schraen in [17], [18], [19], and it is a pleasure to acknowledge this. The authors thank Tsao-Hsien Chen, Lucas Fresse, Victor Ginzburg, Eugen Hellmann, Wen-Wei Li, Ruochuan Liu, Anne Moreau, Alexandre Pyvovarov, Simon Riche, Peng Shan, Benjamin Schraen, Changjian Su, Julianna Tymoczko, Zhixiang Wu and Liang Xiao for discussions or answers to questions. Special thanks to Liang Xiao for many helpful discussions on § A.1, and to Anne Moreau for many discussions on the singularities of Springer fibers.

 $<sup>^{3}</sup>$ We recommend the interested reader to work it out as an exercise.

C. B. is supported by the C.N.R.S and is a member of the A.N.R. project CLap-CLap ANR-18-CE40-0026. Y. D. is supported by the NSFC Grant No. 8200905010 and No. 8200800065.

Finally, the first named author thanks the second for his work and for his patience.

# 2 Preliminaries

# 2.1 General notation

ŀ

Let L be a finite extension of  $\mathbb{Q}_p$  and E be a finite extension of  $\mathbb{Q}_p$  sufficiently large such that  $\Sigma_L := \{\tau : L \hookrightarrow \overline{\mathbb{Q}_p}\} = \{\tau : L \hookrightarrow E\}$ . For  $\mathbf{k} = (k_\tau)_{\tau \in \Sigma_L} \in \mathbb{Z}^{\oplus |\Sigma_L|}$ , denote by  $z^{\mathbf{k}} := \prod_{\tau \in \Sigma_L} \tau(z)^{k_\tau} : L^{\times} \to E^{\times}$ . Let  $\mathcal{O}_L$ , resp.  $\mathcal{O}_E$  be the ring of integers of L, resp. E,  $k_E$  the residue field of E,  $\varpi_L$  be a uniformizer of  $\mathcal{O}_L$ ,  $q_L := |\mathcal{O}_L/\varpi_L|$  and  $\operatorname{val}_L(x)$  the valuation on  $L^{\times}$  such that  $\operatorname{val}_L(\varpi_L) = 1$ . For a character  $\chi$  of  $\mathcal{O}_L^{\times}$ , denote by  $\chi_{\varpi_L}$  the character of  $L^{\times}$  such that  $\chi_{\varpi_L}|_{\mathcal{O}_L^{\times}} = \chi$  and  $\chi_{\varpi_L}(\varpi_L) = 1$ ; for a character  $\delta$  of  $L^{\times}$ , denote by  $\delta_0 := \delta|_{\mathcal{O}_L^{\times}}$ . We use the convention that the Hodge-Tate weight of the p-adic cyclotomic character if 1. For a group A and  $a \in A$ , we denote by  $\operatorname{unr}(a) : L^{\times} \to A$  the unramified character sending any uniformizer to a.

Let A (resp. X) be an affinoid algebra (resp. a rigid analytic space), we write  $\mathcal{R}_{A,L}$  (resp.  $\mathcal{R}_{X,L}$ ) for the Robba ring associated to L with A-coefficients (resp. with  $\mathcal{O}_X$ -coefficients) (see [60, Def. 6.2.1]), and  $\mathcal{R}_{A,L}(\delta)$  for the  $(\varphi, \Gamma)$ -module of character type over  $\mathcal{R}_{A,L}$  associated to a continuous character  $\delta : L^{\times} \to A^{\times}$  in [60, Const. 6.2.4].

Let  $m \in \mathbb{Z}_{\geq 1}$ ,  $\pi$  be an irreducible smooth admissible representation of  $\operatorname{GL}_m(L)$ , denote by  $\operatorname{rec}(\pi)$  the *F*-semi-simple Weil-Deligne representation associated to  $\pi$  via the local Langlands correspondence normalized as in [49]. We normalize local class field theory by sending a uniformizer to a (lift of the) geometric Frobenius. In this way, we identify characters of the Weil group  $W_L \subset \operatorname{Gal}_L := \operatorname{Gal}(\overline{L}/L)$  and characters of  $L^{\times}$  without further mention. Let  $\chi_{\text{cyc}}$  denote the cyclotomic character of  $\operatorname{Gal}_L$  (and of  $L^{\times}$ ).

Let  $\Omega$  be a cuspidal Bernstein component of  $\operatorname{GL}_m(L)$  ([8]) and  $\pi \in \Omega$ . We put

$$\mu_{\Omega} := \{ \eta : L^{\times} \to E^{\times} \mid \pi \otimes_{E} \eta \circ \det \cong \pi \},$$

$$\mu_{\Omega}^{\text{unr}} := \{ \eta : L^{\times} \to E^{\times} \text{ unramified } \mid \pi \otimes_{E} \eta \circ \det \cong \pi \}.$$
(2.1)

We have  $\mu_{\Omega}^{\text{unr}} \subseteq \mu_{\Omega}$  and it is easy to see that both are finite groups (look at the central characters) and independent of the choice of  $\pi$  in  $\Omega$ . Denote by  $\mathcal{Z}_{\Omega}$  the corresponding Bernstein centre (see § 2.2 below for more details). For a closed point  $x \in \text{Spec } \mathcal{Z}_{\Omega}$ , denote by  $\pi_x$  the associated irreducible cuspidal smooth representation of  $\text{GL}_m(L)$  over k(x),  $\mathbf{r}_x := \text{rec}(\pi_x)$  and  $\Delta_x$  the *p*-adic differential equation associated to  $\mathbf{r}_x$ . Recall from [6] that  $\Delta_x$  is the  $(\varphi, \Gamma)$ -module of rank *m* over  $\mathcal{R}_{k(x),L}$  which is de Rham of constant Hodge-Tate weight 0 such that  $D_{\text{pst}}(\Delta_x)$  (forgetting the Hodge filtration) is isomorphic to the Deligne-Fontaine module associated by Fontaine to  $\mathbf{r}_x$  ([21, Prop. 4.1]). We may use the associated  $\text{GL}_m(L)$ -representation or the associated Weil-Deligne representation or the associated *p*-adic differential equation to denote a closed point of  $\text{Spec } \mathcal{Z}_{\Omega}$ depending on the situation.

Throughout the paper, we denote by B the upper triangular matrices in  $GL_n$  and we will

consider parabolic subgroups P of  $GL_n$  containing B, i.e. of the form

$$\begin{pmatrix} \operatorname{GL}_{n_1} & * & \cdots & * \\ 0 & \operatorname{GL}_{n_2} & \cdots & * \\ \vdots & \vdots & \ddots & * \\ 0 & 0 & \cdots & \operatorname{GL}_{n_r} \end{pmatrix},$$
(2.2)

where  $n_i \in \mathbb{Z}_{\geq 1}$  such that  $\sum_{i=1}^r n_i = n$ . For  $i \in \{1, \ldots, r\}$  we define  $s_i := \sum_{j=1}^i n_j$  and  $s_0 := 0$ . We denote by  $L_P$  the Levi subgroup of P containing the group T of diagonal matrices. An integral weight  $\lambda = (\lambda_1, \ldots, \lambda_n)$  of  $\operatorname{GL}_n$  is called P-dominant (resp. strictly P-dominant) if for  $j = 1, \ldots, r$  with  $n_j > 1$ , and  $s_{j-1} \leq i \leq s_{j-1} + n_j - 1$ , we have  $\lambda_i \geq \lambda_{i+1}$  (resp.  $\lambda_i > \lambda_{i+1}$ ).

We use  $\lg(-)$  to denote the length function on elements in Weyl groups. Let  $\mathscr{W} \cong S_n$  be the Weyl group of  $\operatorname{GL}_n$ , and  $\mathscr{W}_{L_P} \subset \mathscr{W}$  be the Weyl group of  $L_P$ . Denote by  $\mathscr{W}_{\min}^P \subset \mathscr{W}$  (resp.  $\mathscr{W}_{\max}^P$ ) the set of minimal (resp. maximal) length representatives in  $\mathscr{W}$  of the right cosets in  $\mathscr{W}_{L_P} \setminus \mathscr{W}$ . Let  $w_0 \in \mathscr{W}$  be the element of maximal length. Then  $w \in \mathscr{W}_{\min}^P$  if and only if  $ww_0 \in \mathscr{W}_{\max}^P$ . We denote by  $\mathscr{W}_L := \mathscr{W}^{|\Sigma_L|}$  (resp.  $\mathscr{W}_{L_P,L} := \mathscr{W}^{|\Sigma_L|}_{L_P}$ ) which is the Weyl group of  $\operatorname{Res}_{\mathbb{Q}_p}^L \operatorname{GL}_n$  (resp.  $\operatorname{Res}_{\mathbb{Q}_p}^L L_P$ ). Then  $\mathscr{W}_{\min,L}^P = (\mathscr{W}_{\min}^P)^{|\Sigma_L|}$  (resp.  $\mathscr{W}_{\max,L}^P = (\mathscr{W}_{\max}^P)^{|\Sigma_L|}$ ) is the set of minimal (resp. maximal) length representatives in  $\mathscr{W}_L$  of  $\mathscr{W}_{L_P,L} \setminus \mathscr{W}_L$ . Put  $w_{0,L} := (w_0, \ldots, w_0) \in \mathscr{W}_L$  for the element of maximal length in  $\mathscr{W}_L$ . We use "·" to denote the dot action of a Weyl group on the corresponding weight space (cf. [57, Def. 1.8]).

If X is a scheme locally of finite type over E, or a locally noetherian formal scheme over  $\mathcal{O}_E$ whose reduction (modulo an ideal of definition) is locally of finite type over  $k_E$ , we denote by  $X^{\text{rig}}$ the associated rigid analytic space over E. If X is a scheme locally of finite type over E or a rigid analytic space over E, we denote by  $X^{\text{red}}$  the associated reduced Zariski-closed subspace. If x is a point of X, we denote by k(x) the residue field at x,  $\mathcal{O}_{X,x}$  the local ring at x,  $\widehat{\mathcal{O}}_{X,x}$  its  $\mathfrak{m}_{\mathcal{O}_{X,x}}$ -adic completion and  $\widehat{X}_x$  the affine formal scheme Spf  $\widehat{\mathcal{O}}_{X,x}$ . If x is a closed point of X, then  $\widehat{\mathcal{O}}_{X,x}$  is a noetherian complete local k(x)-algebra of residue field k(x).

# 2.2 *p*-adic differential equations over Bernstein components

Let  $m \geq 1$ ,  $\Omega$  be cuspidal type of  $\operatorname{GL}_m(L)$  and  $\mathcal{Z}_{\Omega}$  be the associated Bernstein centre over E (that we recall below). In this section, we construct a "universal" *p*-adic differential equation on  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}}$  that interpolates  $\{\Delta_x\}_{x \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}}}$ .

Let  $\pi$  be an irreducible smooth representation of  $\operatorname{GL}_m(L)$  over E of type  $\Omega$ . We assume that E contains the *m*-th roots of unity. By comparing the central characters, we see that there exists  $m_0|m$  such that  $\mu_{m_0} = \{a \in E^{\times} \mid \pi \otimes_E \operatorname{unr}(a) \circ \det \cong \pi\}$ , where  $\mu_{m_0}$  denotes the group of  $m_0$ -th roots of unity in  $E^{\times}$ . We equip  $E[z, z^{-1}]$  with an action of  $\mu_{m_0}$  by a(z) := az for  $a \in \mu_{m_0}$ . We then have a natural isomorphism

$$\mathcal{Z}_{\Omega} \cong E[z, z^{-1}]^{\mu_{m_0}} \cong E[z^{m_0}, z^{-m_0}]$$

such that the induced map  $\mathbb{G}_m := \operatorname{Spec} E[z, z^{-1}] \to \operatorname{Spec} \mathcal{Z}_\Omega$  sends  $\alpha$  to  $\pi \otimes_E \operatorname{unr}(\alpha) \circ \det$ .

We first construct a  $(\varphi, \Gamma)$ -module on  $\mathbb{G}_m^{\text{rig}}$ . Let  $\Delta$  be the *p*-adic differential equation over  $\mathcal{R}_{E,L}$  associated to  $\pi$  (or equivalently to  $\text{rec}(\pi)$ ). Let

$$A_j := E\langle z_j, t_j \rangle / (z_j t_j - p^{2j}),$$

then the maximal spectrum  $\operatorname{Spm} A_j$  is  $\{z \in \mathbb{G}_m^{\operatorname{rig}} \mid p^{-j} \leq |z|_p \leq p^j\}$  (mapping z to  $p^{-j}z_j$ and  $z^{-1}$  to  $p^{-j}t_j$ ) and  $\{\operatorname{Spm} A_j\}_{j\in\mathbb{Z}_{\geq 0}}$  form an admissible covering of  $\mathbb{G}_m^{\operatorname{rig}}$ . We define  $\Delta_{A_j} :=$  $\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A_j,L}(\operatorname{unr}(z))$ , which is a  $(\varphi, \Gamma)$ -module free of rank *m* over  $\mathcal{R}_{A_j,L}$ . These  $\{\Delta_{A_j}\}_{j\in\mathbb{Z}_{\geq 0}}$ glue to a  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{\mathbb{G}_m^{\operatorname{rig}},L}$ , where  $\mathcal{R}_{\mathbb{G}_m^{\operatorname{rig}},L}$  is defined as in [60, Def. 6.2.1].

Let  $\varsigma_{m_0}$  be a primitive  $m_0$ -th root of unity. Since  $\pi \cong \pi \otimes_E \operatorname{unr}(\varsigma_{m_0}) \circ \operatorname{det}$ , we have  $\Delta \cong \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\operatorname{unr}(\varsigma_{m_0}))$ . Let  $\iota_1 : \Delta \to \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\operatorname{unr}(\varsigma_{m_0}))$  be an isomorphism of  $(\varphi, \Gamma)$ modules. For a continuous character  $\delta$  of  $L^{\times}$ , we also use  $\iota_1$  to denote the induced morphism  $\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta) \to \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\operatorname{unr}(\varsigma_{m_0})) \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta)$ . We put for  $i \in \mathbb{Z}_{\geq 1}$ :

$$\iota_i := \iota_1 \circ \iota_1 \circ \cdots \circ \iota_1 : \Delta \xrightarrow{\iota_1} \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\operatorname{unr}(\varsigma_{m_0})) \xrightarrow{\iota_1} \cdots \xrightarrow{\iota_1} \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\operatorname{unr}(\varsigma_{m_0}^i)).$$

Since  $\operatorname{Hom}_{(\varphi,\Gamma)}(\Delta, \Delta) \cong E$  and E contains all  $m_0$ -th roots of unity, we can multiply  $\iota_1$  by a scalar in  $E^{\times}$  so that  $\iota_{m_0} = \operatorname{id}_{\Delta}$ . Let  $\Delta_{A_j}^i := \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A_j,L}(\operatorname{unr}(\varsigma_{m_0}^i z))$ . The isomorphism  $\iota_i$  induces an isomorphism (still denoted)  $\iota_i : \Delta_{A_j} \xrightarrow{\sim} \Delta_{A_j}^i$  satisfying  $\iota_{m_0} = \operatorname{id}_{\Delta A_j}$ . We fix a basis  $\underline{e}$  of  $\Delta$  over  $\mathcal{R}_{E,L}$ , and still denote by  $\underline{e}$  the corresponding basis  $\underline{e} \otimes 1$  of  $\Delta_{A_j}^i$  over  $\mathcal{R}_{A_j,L}$ . We do not ask that  $\iota_1$  respects  $\underline{e}$  (i.e. sends  $\underline{e}$  to  $\underline{e} \otimes 1$ ), hence the isomorphisms  $\iota_i$  in general do not stabilize  $\underline{e}$ .

It is clear that  $\operatorname{Spm} A_j$  is stable by the induced action of  $\mu_{m_0}$  on  $\mathbb{G}_m^{\operatorname{rig}}$ . The action of  $\mu_{m_0}$  on  $A_j$  induces an action on  $\mathcal{R}_{A_j,L}$ . We equip  $\Delta_{A_j}$  with an  $\mathcal{R}_{A_j,L}$ -semi-linear action of  $\mu_{m_0}$  such that  $\varsigma_{m_0}^i$  acts via

$$\Delta_{A_j} \longrightarrow \Delta^i_{A_j} \xrightarrow{\iota_i^{-1}} \Delta_{A_j}$$

where the first map sends  $v \otimes a \in \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A_j,L}(\operatorname{unr}(z))$  to  $v \otimes \varsigma_{m_0}^i(a) \in \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A_j,L}(\operatorname{unr}(\varsigma_{m_0}^i z))$ . Indeed, one can check that this defines a group action of  $\mu_{m_0}$ , that commutes with the  $(\varphi, \Gamma)$ -action.

Define  $\Delta_{B_j} := \Delta_{A_j}^{\mu_{m_0}}$ . By [7, Prop. 2.2.1] (applied first to  $B = A_j$ ,  $S = \mathcal{R}_{E,L}^{[r,s]}$  and  $G = \mu_{m_0}$ , then letting r, s vary), we can deduce that  $\Delta_{B_j}$  is a  $(\varphi, \Gamma)$ -module free of rank m over  $\mathcal{R}_{B_j,L}$ where  $B_j := A_j^{\mu_{m_0}} \cong E\langle z_j^{m_0}, t_j^{m_0} \rangle / (z_j^{m_0} t_j^{m_0} - p^{2jm_0})$ . The affinoids  $\{\text{Spm } B_j\}_{j \in \mathbb{Z}_{\geq 0}}$  then form an admissible covering of  $(\text{Spec } \mathcal{Z}_{\Omega})^{\text{rig}}$ . Moreover it is easy to see that  $\{\Delta_{B_j}\}_{j \in \mathbb{Z}_{\geq 0}}$  glue to a  $(\varphi, \Gamma)$ module over  $\mathcal{R}_{(\text{Spec } \mathcal{Z}_{\Omega})^{\text{rig}},L}$  that we denote by  $\Delta_{\Omega}$ . One checks that  $\Delta_{\Omega}$  is independent of the choice of  $\pi$  of type  $\Omega$ . It is also clear that for a point  $x \in (\text{Spec } \mathcal{Z}_{\Omega})^{\text{rig}}$  with  $\pi_x$  the associated smooth representation of  $\text{GL}_m(L)$  over k(x), the fibre  $\Delta_x := x^* \Delta_{\Omega}$  is isomorphic to the p-adic differential equation associated to  $\pi_x$ .

# 2.3 Potentially crystalline representations

We recall the structure of potentially crystalline Galois representations.

Let  $\rho$  be an *n*-dimensional potentially crystalline representation of  $\operatorname{Gal}_L$  over E. Let L' be a finite Galois extension of L such that  $\rho|_{\operatorname{Gal}_{L'}}$  is crystalline. Consider the Deligne-Fontaine module associated to  $\rho$ :

$$\mathrm{DF}(\rho) := \big( D_{L'} := (B_{\mathrm{cris}} \otimes_{\mathbb{Q}_p} \rho)^{\mathrm{Gal}_{L'}}, \varphi, \mathrm{Gal}(L'/L) \big),$$

where  $D_{L'} := (B_{\text{cris}} \otimes_{\mathbb{Q}_p} \rho)^{\text{Gal}_{L'}}$  is a finite free  $L'_0 \otimes_{\mathbb{Q}_p} E$ -module of rank  $n, L'_0$  being the maximal unramified subextension of L' (over  $\mathbb{Q}_p$ ), where the  $\varphi$ -action on  $D_{L'}$  is induced from the  $\varphi$ -action on  $B_{\text{cris}}$ , and where the Gal(L'/L)-action on  $D_{L'}$  is the residual action of  $\text{Gal}_L$ . By Fontaine's equivalence of categories as in [21, Prop. 4.1], we can associate to  $\text{DF}(\rho)$  an *n*-dimensional Weil-Deligne representation  $\mathbf{r}(\rho)$  of  $W_L$  over E (and we can recover  $\text{DF}(\rho)$  from  $\mathbf{r}(\rho)$  as in *loc. cit.*). Let P be a parabolic subgroup of  $GL_n$  as in (2.2). Assume that  $\mathbf{r}(\rho)$  admits a filtration

$$\mathscr{F}: \operatorname{Fil}_{\bullet}^{\mathscr{F}} \mathbf{r}(\rho) = \left( 0 = \operatorname{Fil}_{0}^{\mathscr{F}} \mathbf{r}(\rho) \subsetneq \operatorname{Fil}_{1}^{\mathscr{F}} \mathbf{r}(\rho) \subsetneq \cdots \subsetneq \operatorname{Fil}_{r}^{\mathscr{F}} \mathbf{r}(\rho) = \mathbf{r}(\rho) \right)$$

by Weil-Deligne subrepresentations such that  $\dim_E \operatorname{Fil}_i^{\mathscr{F}} \mathbf{r}(\rho) = \sum_{j=1}^i n_j$ . We call such a filtration a *P*-filtration. We call the filtration  $\mathscr{F}$  a minimal filtration if  $\mathbf{r}(\rho)_i := \operatorname{gr}_i^{\mathscr{F}} \mathbf{r}(\rho)$  is an irreducible Weil-Deligne representation for all *i*. We assume that  $\mathscr{F}$  is minimal in the sequel. In this case, the Galois representation  $\rho$  is called generic if  $\operatorname{Hom}(\mathbf{r}(\rho)_i, \mathbf{r}(\rho)_j) = 0$  and  $\operatorname{Hom}(\mathbf{r}(\rho)_i, \mathbf{r}(\rho)_j(1)) = 0$ for all  $i \neq j$  (where Hom is taken in the category of Weil-Deligne representations and (1) means the twist by  $x \in L^{\times} \mapsto \frac{1}{q_L^{\operatorname{val}_L(x)}}$ ). It is easy to see that being generic does not depend on the choice of minimal filtrations on  $\mathbf{r}(\rho)$ , and that if  $\rho$  is generic then  $\mathbf{r}(\rho) \cong \bigoplus_{i=1}^r \mathbf{r}(\rho)_i$ . Let  $\Omega_i$  be the Bernstein component of  $\operatorname{GL}_{n_i}(L)$  such that the smooth irreducible representation corresponding to  $\mathbf{r}(\rho)_i$  via the classical local Langlands correspondence (normalized as in [49]) lies in  $\Omega_i$ . Let  $\Omega := \prod_i \Omega_i$ , which is a Bernstein component of  $L_P(L)$ . The minimal *P*-filtration  $\mathscr{F}$  will also be called an  $\Omega$ -filtration.

The *P*-filtration  $\mathscr{F}$  corresponds to a filtration (still denoted)  $\mathscr{F} = \operatorname{Fil}_{\bullet}^{\mathscr{F}} \operatorname{DF}(\rho)$  (and still called a *P*-filtration) on  $\operatorname{DF}(\rho)$  by Deligne-Fontaine submodules, such that  $\operatorname{Fil}_{i}^{\mathscr{F}} \operatorname{DF}(\rho)$  is associated to  $\operatorname{Fil}_{i}^{\mathscr{F}} \mathbf{r}(\rho)$  via [21, Prop. 4.1]. If  $\rho$  is generic, we have then  $\operatorname{DF}(\rho) \cong \bigoplus_{i=1}^{r} \operatorname{gr}_{i}^{\mathscr{F}} \operatorname{DF}(\rho)$ .

As  $\rho$  is potentially crystalline, it is de Rham, and thus we have  $D_{dR}(\rho) \cong (D_{L'} \otimes_{L'_0} L')^{Gal(L'/L)}$ , which is a free  $L \otimes_{\mathbb{Q}_p} E$ -module of rank n. The P-filtration  $\mathscr{F}$  on  $DF(\rho)$  induces a P-filtration  $\mathscr{F}$  on  $D_{dR}(\rho)$  by free  $L \otimes_{\mathbb{Q}_p} E$ -submodules  $\operatorname{Fil}_i^{\mathscr{F}} D_{dR}(\rho) := (\operatorname{Fil}_i^{\mathscr{F}} D_{L'} \otimes_{L'_0} L')^{Gal(L'/L)}$ . Recall also that  $D_{dR}(\rho)$  is equipped with a natural decreasing Hodge filtration  $\operatorname{Fil}_{\bullet}^H D_{dR}(\rho)$  (induced by the one on  $B_{\operatorname{cris}}$ ) given by (not necessarily free)  $L \otimes_{\mathbb{Q}_p} E$  submodules. We assume that  $\rho$  has distinct Hodge-Tate weights. Hence, for each  $\tau \in \Sigma_L$ , we have a complete flag (with an obvious notation for  $D_{dR}(\rho)_{\tau}$ )

$$0 \subsetneq \operatorname{Fil}_{-h_{n,\tau}}^{H} D_{\mathrm{dR}}(\rho)_{\tau} \subsetneq \operatorname{Fil}_{-h_{n-1,\tau}}^{H} D_{\mathrm{dR}}(\rho)_{\tau} \subsetneq \cdots \subsetneq \operatorname{Fil}_{-h_{1,\tau}}^{H} D_{\mathrm{dR}}(\rho)_{\tau} = D_{\mathrm{dR}}(\rho)_{\tau}$$

where  $h_{i,\tau}$  are the integers such that  $\dim_E \operatorname{gr}_{-h_{i,\tau}}^H D_{\mathrm{dR}}(\rho)_{\tau} = 1$ . Thus the Hodge-Tate weights of  $\rho$  are  $\mathbf{h} = (\mathbf{h}_i)_{i=1,\dots,n} = (h_{1,\tau} > \dots > h_{n,\tau})_{\tau \in \Sigma_L}$ .

We fix a basis of  $D_{\mathrm{dR}}(\rho)_{\tau}$  over E for each  $\tau \in \Sigma_L$ . The filtration  $\mathscr{F}$  (resp. Fil<sup>H</sup>) on  $D_{\mathrm{dR}}(\rho)$  thus corresponds to an E-point of the flag variety  $\operatorname{Res}_{\mathbb{Q}_p}^L \operatorname{GL}_n / \operatorname{Res}_{\mathbb{Q}_p}^L P$  (resp.  $\operatorname{Res}_{\mathbb{Q}_p}^L \operatorname{GL}_n / \operatorname{Res}_{\mathbb{Q}_p}^L B$ ):

$$g_1(\operatorname{Res}_{\mathbb{Q}_p}^L P)(E) = (g_{1,\tau}P(E))_{\tau \in \Sigma_L} \quad (\operatorname{resp.} g_2(\operatorname{Res}_{\mathbb{Q}_p}^L B)(E) = (g_{2,\tau}B(E))_{\tau \in \tau_L})).$$

For each  $\tau \in \Sigma_L$ , there exists thus a unique  $w_{\mathscr{F},\tau} \in \mathscr{W}_{\max}^P$  such that

$$(g_{1,\tau}P(E), g_{2,\tau}B(E)) \in \operatorname{GL}_n(E)(1, w_{\mathscr{F},\tau})(P \times B)(E) \subseteq (\operatorname{GL}_n/P \times \operatorname{GL}_n/B)(E)$$

where we still use  $w_{\mathscr{F},\tau} \in N_{\mathrm{GL}_n}(T)$  to denote a lifting of the corresponding element in  $\mathscr{W}$  (which is traditionally denoted by  $\dot{w}_{\mathscr{F},\tau}$ ). We write  $w_{\mathscr{F}} := (w_{\mathscr{F},\tau})_{\tau \in \Sigma_L} \in \mathscr{W}^{|\Sigma_L|}$ . We call  $(\rho, \mathcal{F})$  non-critical if  $w_{\mathscr{F}} = w_{0,L}$ , or equivalently  $w_{\mathscr{F},\tau} = w_0$  for all  $\tau \in \Sigma_L$ .

Consider the  $(\varphi, \Gamma)$ -module  $D_{\operatorname{rig}}(\rho)$  over  $\mathcal{R}_{E,L}$  associated to  $\rho$  (see [6] and the references therein). Let  $\Delta$  (resp.  $\operatorname{Fil}_i^{\mathscr{F}} \Delta$ ) be the *p*-adic differential equation over  $\mathcal{R}_{E,L}$  associated to  $\operatorname{DF}(\rho)$ (resp. to  $\operatorname{Fil}_i^{\mathscr{F}} \operatorname{DF}(\rho)$ ), or equivalently to  $\mathbf{r}(\rho)$  (resp. to  $\operatorname{Fil}_i^{\mathscr{F}} \mathbf{r}(\rho)$ ). Then  $\mathscr{F} := \operatorname{Fil}_{\bullet}^{\mathscr{F}} \Delta$  gives an increasing filtration on  $\Delta$  by saturated  $(\varphi, \Gamma)$ -submodules. Consider

$$\mathcal{M}(\rho) := D_{\mathrm{rig}}(\rho)[1/t] \cong \Delta[1/t].$$

By inverting t, the filtration  $\mathscr{F}$  on  $\Delta$  induces an increasing filtration (still denoted)  $\mathscr{F} := \{\operatorname{Fil}_{i}^{\mathscr{F}}\mathcal{M}(\rho) := (\operatorname{Fil}_{i}^{\mathscr{F}}\Delta)[1/t]\}$  on  $\mathcal{M}(\rho)$  by  $(\varphi, \Gamma)$ -submodules over  $\mathcal{R}_{E,L}[1/t]$ . Finally, the filtration  $\mathscr{F}$  on  $\mathcal{M}(\rho)$  induces a filtration on  $D_{\operatorname{rig}}(\rho)$ :

$$\mathscr{F} := \{ \operatorname{Fil}_{i}^{\mathscr{F}} D_{\operatorname{rig}}(\rho) := \operatorname{Fil}_{i}^{\mathscr{F}} \mathcal{M}(\rho) \cap D_{\operatorname{rig}}(\rho) \}$$
(2.3)

by saturated  $(\varphi, \Gamma)$ -submodules of  $D_{\text{rig}}(\rho)$ . By Berger's equivalence of categories ([6, Thm. A]), Fil<sup> $\mathscr{F}$ </sup> $D_{\text{rig}}(\rho)$  corresponds to the filtered Deligne-Fontaine module Fil<sup> $\mathscr{F}$ </sup> $\text{DF}(\rho)$  equipped with the induced filtration from the Hodge filtration on  $D_{L'} = (B_{\text{cris}} \otimes_{\mathbb{Q}_p} \rho)^{\text{Gal}_{L'}}$  (coming from the filtration on  $B_{\text{cris}}$ ). Such a filtration will be called an  $\Omega$ -filtration on  $D_{\text{rig}}(\rho)$  (see § 4.1.2 for a definition in a more general setting).

One sees that the Hodge-Tate weights of  $\operatorname{Fil}_{i}^{\mathscr{F}}D_{\operatorname{rig}}(\rho)$  are given by (recall  $s_{i} = \sum_{j=1}^{i} n_{j}$ )  $\{h_{(w_{\mathscr{F},\tau}w_{0})^{-1}(1),\tau},\ldots,h_{(w_{\mathscr{F},\tau}w_{0})^{-1}(s_{i}),\tau}\}_{\tau\in\Sigma_{L}}$ , hence the Hodge-Tate weights of  $\operatorname{gr}_{i}^{\mathscr{F}}D_{\operatorname{rig}}(\rho)$  are

$$(w_{\mathscr{F}}(\mathbf{h})_{s_{i-1}+1}, \dots, w_{\mathscr{F}}(\mathbf{h})_{s_{i}}) = (h_{(w_{\mathscr{F},\tau}w_{0})^{-1}(s_{i-1}+1),\tau}, \dots, h_{(w_{\mathscr{F},\tau}w_{0})^{-1}(s_{i}),\tau})_{\tau \in \Sigma_{L}}$$

(which are decreasing as  $w_{\mathscr{F},\tau}w_0 \in \mathscr{W}_{\min}^P$ ). In particular,  $(\rho,\mathscr{F})$  is non-critical if and only if the Hodge-Tate weights of  $\operatorname{gr}_i^{\mathscr{F}}D_{\operatorname{rig}}(\rho)$  are  $(\mathbf{h}_{s_{i-1}+1},\ldots,\mathbf{h}_{s_i})$  for  $i = 1,\ldots,r$ . Since  $\operatorname{gr}_i^{\mathscr{F}}D_{\operatorname{rig}}(\rho) \subseteq t^{-N}\operatorname{gr}_i^{\mathscr{F}}\Delta$  for N sufficiently large, using [6, Thm. A] and comparing the weights, we have an injection of  $(\varphi,\Gamma)$ -modules over  $\mathcal{R}_{E,L}$  for  $i = 1,\ldots,r$ :

$$\operatorname{gr}_{i}^{\mathscr{F}} D_{\operatorname{rig}}(\rho) \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(z^{-w_{\mathscr{F}}(\mathbf{h})_{s_{i}}}) \hookrightarrow \operatorname{gr}_{i}^{\mathscr{F}} \Delta.$$
 (2.4)

Finally let  $\overline{\rho}$ :  $\operatorname{Gal}_L \to \operatorname{GL}_n(k_E)$  be a continuous representation,  $\xi : I_L \to \operatorname{GL}_n(E)$  be an inertial type (where  $I_L \subset W_L$  denotes the inertial subgroup), and  $\mathbf{h} \in \mathbb{Z}^{n|\Sigma_L|}$  be a strictly dominant weight as above. We denote by  $R_{\overline{\rho}}^{\operatorname{pcr}}(\xi, \mathbf{h})$  the universal potentially crystalline *framed* deformation ring of  $\overline{\rho}$  of inertial type  $\xi$  and of Hodge-Tate weights  $\mathbf{h}$  (cf. [63]).

# 3 Bernstein eigenvarieties

In this section, we construct Bernstein eigenvarieties from p-adic automorphic representations. In § 3.1, we give the general formalism of the construction, which can be applied to any admissible locally analytic representation of (a product of copies of)  $GL_n$ . We then apply in § 3.2 this formalism to p-adic automorphic representations on compact unitary group, to get what we call Bernstein eigenvarieties. We prove basic properties of the latter, like the density of classical points, etc. We also show that the Galois representations associated to points on Bernstein eigenvarieties admit a certain filtration, and address the problem of companion points. Finally, in § 3.3, we apply the general formalism to the "patched" p-adic automorphic representation of [26] to obtain a patched version of Bernstein eigenvarieties (that has a more local flavor).

## 3.1 Abstract construction

This section gives a general formal construction of certain rigid analytic spaces from Emerton's Jacquet modules of locally analytic representations, using Bushnell-Kutzko's theory of types.

#### 3.1.1 Notation and setup

We will assume that the reader has some familiarity with p-adic functional analysis, and we use most of the time without further mention - the various foundational results in [76], [77], [78], [79] and [42].

For a locally  $\mathbb{Q}_p$ -analytic group H, denote by  $\mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(H, E)$  the space of locally  $\mathbb{Q}_p$ -analytic functions on H with values in E and by  $D(H, E) := \mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(H, E)^{\vee}$  its strong dual (the distribution algebra), which is a Fréchet-Stein algebra when H is compact. Denote by  $\mathcal{C}^{\infty}(H, E) \hookrightarrow \mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(H, E)$  the closed subspace of locally constant functions on H with values in E, and set  $D^{\infty}(H, E) := \mathcal{C}^{\infty}(H, E)^{\vee}$ , which is a Hausdorff quotient of D(H, E).

For a topologically finitely generated locally  $\mathbb{Q}_p$ -analytic abelian group Z, denote by  $\widehat{Z}$  the rigid space over E parameterizing locally  $\mathbb{Q}_p$ -analytic characters of Z (cf. [42, Prop. 6.4.5]). By [42, Prop. 6.4.6], there is a natural injection  $D(Z, E) \hookrightarrow \Gamma(\widehat{Z}, \mathcal{O}_{\widehat{Z}})$ . For  $\chi$  a locally  $\mathbb{Q}_p$ -analytic character of Z, we denote by  $\mathfrak{m}_{\chi}$  the associated maximal ideal of E[Z]. If  $Z \cong (L^{\times})^n$  or is a compact open subgroup of  $(L^{\times})^n$  (for L as in § 2.1), and  $\chi$  is a locally  $\mathbb{Q}_p$ -analytic character of Z, we denote by  $\mathfrak{wt}(\chi)$  the weight of  $\chi$  (see [19, Notation]). For instance if  $\chi$  is E-valued, then  $\mathfrak{wt}(\chi) = (\mathfrak{wt}(\chi)_{\tau})_{\tau \in \Sigma_L} = (\mathfrak{wt}(\chi)_{i,\tau})_{i=1,\dots,n} \in (E^n)^{|\Sigma_L|}$ . In this case, for  $\lambda = (\lambda_{i,\tau}) \in (\mathbb{Z}^n)^{|\Sigma_L|}$ , we denote by  $\delta_{\lambda}$  the algebraic character of Z of weight  $\lambda$ .

For a continuous Banach representation  $\Pi$  of a *p*-adic Lie group G, we denote by  $\Pi^{an}$  the locally  $\mathbb{Q}_p$ -analytic subrepresentation of  $\Pi$  and by  $\Pi^{\text{lalg}} \subseteq \Pi^{an}$  the locally  $\mathbb{Q}_p$ -algebraic subrepresentation of  $\Pi$ .

Let  $\mathscr{I}$  be a finite index set. For any  $i \in \mathscr{I}$ , assume we have a finite extension  $F_i$  of  $\mathbb{Q}_p$ . For each  $F_i$ , we fix a uniformizer  $\varpi_i$ , and denote by  $\kappa_{\varpi_i} : F_i^{\times} \to \mathcal{O}_{F_i}^{\times}$  the map sending  $\varpi_i$  to 1 and being the identity on  $\mathcal{O}_{F_i}^{\times}$ .

Let  $G := \prod_{i \in \mathscr{I}} \operatorname{Res}_{\mathbb{Q}_p}^{F_i} \operatorname{GL}_n$  (an algebraic group), and  $G_p := G(\mathbb{Q}_p)$ . For each  $i \in \mathscr{I}$ , we fix a parabolic subgroup  $P_i$  of  $\operatorname{GL}_n$  containing the Borel subgroup B of upper triangular matrices, and let  $L_{P_i}$  the Levi subgroup of  $P_i$  containing the group T of diagonal matrices. Let P := $\prod_{i \in \mathscr{I}} \operatorname{Res}_{\mathbb{Q}_p}^{F_i} P_i \supseteq B_{\mathscr{I}} := \prod_{i \in \mathscr{I}} \operatorname{Res}_{\mathbb{Q}_p}^{F_i} B$ , and  $L_P := \prod_{i \in \mathscr{I}} \operatorname{Res}_{\mathbb{Q}_p}^{F_i} L_{P_i}$  which is the Levi subgroup of P containing  $T_{\mathscr{I}} := \prod_{i \in \mathscr{I}} \operatorname{Res}_{\mathbb{Q}_p}^{F_i} T$ . Let  $N_P$  (resp.  $N_{P_i}$ ) be the unipotent radical of P (resp. of  $P_i$ ),  $P^-$  (resp.  $P_i^-$ ) the parabolic subgroup opposite to P (resp. to  $P_i$ ),  $N_{P^-}$  (resp.  $N_{P_i}^-$ ) the unipotent radical of  $P^-$  (resp. of  $P_i^-$ ),  $Z_{L_P}$  (resp.  $Z_{L_{P_i}}$ ) the centre of  $L_P$  (resp. of  $L_{P_i}$ ), and  $L_P^D$ (resp.  $L_{P_i}^D$ ) the derived subgroup of  $L_P$  (resp.  $L_{P_i}$ ). We have therefore

$$N_P = \prod_{i \in \mathscr{I}} \operatorname{Res}_{\mathbb{Q}_p}^{F_i} N_{P_i}, \quad P^- = \prod_{i \in \mathscr{I}} \operatorname{Res}_{\mathbb{Q}_p}^{F_i} P_i^-, \quad N_{P^-} = \prod_{i \in \mathscr{I}} \operatorname{Res}_{\mathbb{Q}_p}^{F_i} N_{P_i^-},$$
$$Z_{L_P} = \prod_{i \in \mathscr{I}} \operatorname{Res}_{\mathbb{Q}_p}^{F_i} Z_{L_{P_i}}, \quad L_P^D = \prod_{i \in \mathscr{I}} \operatorname{Res}_{\mathbb{Q}_p}^{F_i} L_{P_i}^D.$$

We denote by  $\mathfrak{g}, \mathfrak{b}_{\mathscr{I}}, \mathfrak{p}, \mathfrak{n}_{P}, \mathfrak{l}_{P}, \mathfrak{n}_{P^{-}}, \mathfrak{z}_{L_{P}}, \mathfrak{l}_{P}^{D}$  the Lie algebra over E of  $G, B_{\mathscr{I}}, P, N_{P}, L_{P}, N_{P^{-}}, Z_{L_{P}}, L_{P}^{D}$  respectively. For a Lie algebra  $\mathfrak{h}$  over E, denote by  $U(\mathfrak{h})$  the universal enveloping algebra over E. We define

$$L^0_P := \prod_{i \in \mathscr{I}} L_{P_i}(\mathcal{O}_{F_i}) \subset L_P(\mathbb{Q}_p), \quad Z^0_{L_P} := \prod_{i \in \mathscr{I}} Z_{L_{P_i}}(\mathcal{O}_{F_i}) = L^0_P \cap Z_{L_P}(\mathbb{Q}_p) \subset Z_{L_P}(\mathbb{Q}_p)$$
$$N^0_P := \prod_{i \in \mathscr{I}} N_P(\mathcal{O}_{F_i}) \subset N_P(\mathbb{Q}_p), \quad N^0_{P^-} := \prod_{i \in \mathscr{I}} N_{P^-}(\mathcal{O}_{F_i}) \subset N_{P^-}(\mathbb{Q}_p)$$

and denote by  $\det_{L_P}$  the determinant map  $L_P(\mathbb{Q}_p) \to Z_{L_P}(\mathbb{Q}_p)$ . For each  $i \in \mathscr{I}, L_{P_i}(F_i)$  has the

 $\text{form} \begin{pmatrix} \operatorname{GL}_{n_{i,1}} & 0 & \cdots & 0 \\ 0 & \operatorname{GL}_{n_{i,2}} & \cdots & 0 \\ \vdots & \vdots & \ddots & 0 \\ 0 & 0 & \cdots & \operatorname{GL}_{i,n_{r_i}} \end{pmatrix} \text{for some } r_i \in \mathbb{Z}_{\geq 1} \text{ and integers } n_{i,j} \in \mathbb{Z}_{\geq 1}, 1 \leq j \leq r_i \text{ with} \\ \sum_{j=1}^{r_i} n_{i,j} = n. \text{ Hence } L_P(\mathbb{Q}_p) \cong \prod_{i \in \mathscr{I}} \prod_{j=1}^{r_i} \operatorname{GL}_{n_{i,j}}(F_i). \text{ For each } (i,j) \in \mathscr{I} \times \{1,\ldots,r_i\} \text{ we fix} \\ a \text{ cuspidal Bernstein component } \Omega_{i,j} \text{ for } \operatorname{GL}_{n_{i,j}}(F_i), \text{ and we let } \Omega := \prod_{i \in \mathscr{I}} \prod_{j=1}^{r_i} \Omega_{i,j}. \text{ Let } \mathcal{Z}_{\Omega_{i,j}} \\ \text{ (rem } \mathcal{I}) \text{ be the Permutation control of } \Omega \quad (\text{rem } f \Omega) \text{ curve } F_i \text{ (see S 2.2), we parameters an integer of } \Omega \\ \end{bmatrix}$ 

(resp.  $\mathcal{Z}_{\Omega}$ ) be the Bernstein centre of  $\Omega_{i,j}$  (resp. of  $\Omega$ ) over E (see § 2.2), we have an isomorphism of commutative E-algebras

$$\mathcal{Z}_{\Omega} \cong \bigotimes_{i \in \mathscr{I}} \bigotimes_{j=1}^{r_i} \mathcal{Z}_{\Omega_{i,j}}.$$

Let  $(J_{i,j}, \sigma_{i,j}^0)$  be a maximal simple type of  $\Omega_{i,j}$  (cf. [23, § 6]) such that the compact open subgroup  $J_{i,j}$  is contained in  $\operatorname{GL}_{n_{i,j}}(\mathcal{O}_{F_i})$ . Recall that  $\sigma_{i,j}^0$  is an absolutely irreducible smooth representation of  $J_{i,j}$  over E. Put  $\sigma_{i,j} := \operatorname{Ind}_{J_{i,j}}^{\operatorname{GL}_{n_{i,j}}(\mathcal{O}_{F_i})} \sigma_{i,j}^0$ , which is an absolutely irreducible smooth representation of  $\operatorname{GL}_{n_{i,i}}(\mathcal{O}_{F_i})$  over E (e.g. see the proof of [75, Cor. 6.1]). Let

$$\sigma^0 := \boxtimes_{i \in \mathscr{I}} \boxtimes_{j=1}^{r_i} \sigma_{i,j}^0, \quad J := \prod_{i,j} J_{i,j} \text{ and } \sigma := \boxtimes_{i \in \mathscr{I}} \boxtimes_{j=1}^{r_i} \sigma_{i,j} \cong \operatorname{Ind}_J^{L_P^0} \sigma^0$$

which is an absolutely irreducible smooth representation of  $L_P^0$  over E. Recall we have natural isomorphisms of commutative E-algebras (where "c-ind" denotes the compact induction)

$$\operatorname{End}_{\operatorname{GL}_{n_{i,j}}(F_i)}\left(\operatorname{c-ind}_{\operatorname{GL}_{n_{i,j}}(\mathcal{O}_{F_i})}^{\operatorname{GL}_{n_{i,j}}(F_i)}\sigma_{i,j}\right) \cong \operatorname{End}_{\operatorname{GL}_{n_{i,j}}(F_i)}\left(\operatorname{c-ind}_{J_{i,j}}^{\operatorname{GL}_{n_{i,j}}(F_i)}\sigma_{i,j}^{0}\right) \cong \mathcal{Z}_{\Omega_{i,j}}$$
$$\operatorname{End}_{L_P(\mathbb{Q}_p)}(\operatorname{c-ind}_{L_p^0}^{L_P(\mathbb{Q}_p)}\sigma) \cong \operatorname{End}_{L_P(\mathbb{Q}_p)}(\operatorname{c-ind}_{J}^{L_P(\mathbb{Q}_p)}\sigma) \cong \mathcal{Z}_{\Omega}.$$

Let  $\lambda$  be an integral *P*-dominant weight of *G*, i.e.  $\lambda = (\lambda_i)_{i \in \mathscr{I}} = (\lambda_{i,\tau})_{\substack{i \in \mathscr{I} \\ \tau \in \Sigma_{F_i}}}$  with each weight  $\lambda_{i,\tau}$  of  $\operatorname{GL}_n$  being  $P_i$ -dominant (cf. § 2.1), and  $L(\lambda)_P$  the algebraic representation of  $L_P(\mathbb{Q}_p)$  over E of highest weight  $\lambda$ . If  $\lambda$  is moreover dominant, we denote by  $L(\lambda)$  the algebraic representation of  $G_p$  (over E) of highest weight  $\lambda$ .

#### 3.1.2 $(\Omega, \lambda)$ -part of Jacquet-Emerton modules

Let V be an admissible locally  $\mathbb{Q}_p$ -analytic representation of  $G(\mathbb{Q}_p)$  over E. Using Emerton's locally analytic Jacquet functor and the type theory (à la Bushnell-Kutzko), we associate to V a certain  $\mathcal{Z}_{\Omega} \times Z^0_{L_P}$ -module  $B_{\Omega,\lambda}(V)$ .

First applying Emerton's Jacquet functor  $J_P(-)$  ([40]) to V, we obtain an essentially admissible locally  $\mathbb{Q}_p$ -analytic representation  $J_P(V)$  of  $L_P(\mathbb{Q}_p)$  over E. Let  $\lambda$  be an integral P-dominant weight of G. We then define

$$J_P(V)_{\lambda} := \operatorname{Hom}_{\mathfrak{l}_P^D} \left( L(\lambda)_P, J_P(V) \right) \cong \left( J_P(V) \otimes_E L(\lambda)_P^{\vee} \right)^{\mathfrak{l}_P^D} \cong \varinjlim_H \left( J_P(V) \otimes_E L(\lambda)_P^{\vee} \right)^H$$
(3.1)

where H runs through compact open subgroups of  $L^D_P(\mathbb{Q}_P)$ . This is a closed  $L_P(\mathbb{Q}_p)$ -subrepresentation of  $J_P(V) \otimes_E L(\lambda)_P^{\vee}$  (with the induced topology).

We equip the space of locally analytic functions  $\mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z^0_{L_P}, E)$  with an  $L_P(\mathbb{Q}_p)$ -action given by the regular action of  $Z_{L_P}^0$  on  $\mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z_{L_P}^0, E)$  precomposed with:

$$L_P(\mathbb{Q}_P) \xrightarrow{(\det_{L_P})^{-1}} Z_{L_P}(\mathbb{Q}_p) \cong \prod_{i \in \mathscr{I}} Z_{L_{P_i}}(F_i) \xrightarrow{\prod_i \kappa_{\varpi_i}} \prod_{i \in \mathscr{I}} Z_{L_{P_i}}(\mathcal{O}_{F_i}) = Z_{L_P}^0.$$
(3.2)

Let  $\sigma$  be an irreducible smooth representation of  $L_P^0$  as in § 3.1.1. We put:

$$B_{\sigma,\lambda}(V) := \operatorname{Hom}_{L^0_P} \left( \sigma, J_P(V)_{\lambda} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \operatorname{la}}(Z^0_{L_P}, E) \right) \\ \cong \left( \sigma^{\vee} \otimes_E \left( J_P(V)_{\lambda} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \operatorname{la}}(Z^0_{L_P}, E) \right) \right)^{L^0_P} \quad (3.3)$$

where  $J_P(V)_{\lambda} \widehat{\otimes}_E C^{\mathbb{Q}_p - \operatorname{la}}(Z^0_{L_P}, E)$  is the completion of  $J_P(V)_{\lambda} \otimes_E C^{\mathbb{Q}_p - \operatorname{la}}(Z^0_{L_P}, E)$  equipped with the projective - or equivalently injective - tensor product topology (note that both factors are vector spaces of compact type) and with the diagonal action of  $L_P(\mathbb{Q}_p)$ . We view  $B_{\sigma,\lambda}(V)$  as a closed subspace of  $\sigma^{\vee} \otimes_E J_P(V)_{\lambda} \widehat{\otimes}_E C^{\mathbb{Q}_p - \operatorname{la}}(Z^0_{L_P}, E)$  which is an *E*-vector space of compact type (recall the finite dimensional  $\sigma^{\vee}$  is equipped with the finest locally convex topology). Hence  $B_{\sigma,\lambda}(V)$  is also an *E*-vector space of compact type.

**Remark 3.1.1.** The definition of  $B_{\sigma,\lambda}(V)$  might appear somewhat artificial. The motivation is to construct an object parametrizing  $L_P(\mathbb{Q}_p)$ -subrepresentations of  $J_P(V)_{\lambda}$  that lie in the Bernstein component  $\Omega$  up to twist by continuous characters of  $Z_{L_P}(\mathbb{Q}_p)$  (see Proposition 3.1.8). One may consider removing the factor  $C^{\mathbb{Q}_p-\text{la}}(Z^0_{L_P}, E)$  and using types for  $L^D_P(\mathbb{Q}_p)$  instead of the type  $(\sigma, L^0_P)$  for  $L_P(\mathbb{Q}_p)$ . The resulting object is actually more natural. However, it is not clear to the authors how to use such an object to parametrize the  $L_P(\mathbb{Q}_p)$ -subrepresentations of  $J_P(V)_{\lambda}$ discussed above.

Next, we discuss various group actions on  $B_{\sigma,\lambda}(V)$ .

There is a natural locally  $\mathbb{Q}_p$ -analytic action of  $Z_{L_P}(\mathbb{Q}_p) \times Z_{L_P}^0$  on  $\sigma^{\vee} \otimes_E J_P(V)_{\lambda} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E)$  where  $Z_{L_P}(\mathbb{Q}_p)$  acts on  $J_P(V)_{\lambda}, Z_{L_P}^0$  acts on  $\mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E)$ , and there is no action on  $\sigma^{\vee}$ . It is easy to check that this action of  $Z_{L_P}(\mathbb{Q}_p) \times Z_{L_P}^0$  commutes with the diagonal  $L_P^0$ -action  $(L_P^0$  acting on all 3 factors). So we see that  $B_{\sigma,\lambda}(V)$  inherits a locally  $\mathbb{Q}_p$ -analytic action of  $Z_{L_P}(\mathbb{Q}_p) \times Z_{L_P}^0$ . In order to avoid confusion, we write  $\mathcal{Z}_0$  for  $Z_{L_P}^0$  when it acts on  $\mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E)$  alone and we use the notation  $\Upsilon_0$  for this action. Likewise we write  $\mathcal{Z}_1$  for  $Z_{L_P}(\mathbb{Q}_p)$  when it acts on  $J_P(V)_{\lambda}$  alone and use the notation  $\Upsilon_1$  for this action.

We write  $\Delta_0$  for the action of  $Z_{L_P}(\mathbb{Q}_p)$  on  $B_{\sigma,\lambda}(V)$  induced by the diagonal action of  $Z_{L_P}(\mathbb{Q}_p)$ on  $J_P(V)_{\lambda} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z_{L_P}^0, E)$  (and the trivial action on  $\sigma$ ), i.e.  $\Delta_0$  is given by the action of  $\mathcal{Z}_1 \times \mathcal{Z}_0$ composed with the morphism

$$Z_{L_P}(\mathbb{Q}_p) \xrightarrow{(\mathrm{id},(3.2))} Z_{L_P}(\mathbb{Q}_p) \times Z_{L_P}^0 = \mathcal{Z}_1 \times \mathcal{Z}_0.$$
(3.4)

Denoting by  $\psi_{\sigma}$  the central character of  $\sigma$  (a character of  $Z_{L_P}(\mathbb{Q}_p)$ ), we see that the restriction  $\Delta_0|_{Z_{L_P}^0}$  on  $B_{\sigma,\lambda}(V)$  is given by  $\psi_{\sigma}$ . We deduce hence for any  $z^0 \in Z_{L_P}^0$  (see the  $(\det_{L_P})^{-1}$  in (3.2)):

$$\Upsilon_1(z^0) = \psi_\sigma(z^0) \Upsilon_0\left(\det_{L_P}(z^0)\right). \tag{3.5}$$

With our choice of the uniformizers  $\varpi_i$ , we have a map  $\mathbb{Z} \hookrightarrow F_i^{\times}$ ,  $1 \mapsto \varpi_i$ , that induces a map  $\bigoplus_{i \in \mathscr{I}} \mathbb{Z}^{r_i} \hookrightarrow Z_{L_P}(\mathbb{Q}_p)$ . We denote by  $Z_{\underline{\varpi}}$  its image. Then the action of  $\Delta_0$  is determined by  $\Delta_0|_{Z_{\underline{\varpi}}}$  (since  $\Delta_0|_{Z_{L_P}^0}$  acts via  $\psi_{\sigma}$  and  $Z_{L_P}(\mathbb{Q}_p) \cong Z_{L_P}^0 \times Z_{\underline{\varpi}}$ ). On the other hand, the action of  $\mathcal{Z}_1 \times \mathcal{Z}_0$  restricts to an action of  $Z_{\underline{\varpi}} \times \mathcal{Z}_0$  on  $B_{\sigma,\lambda}(V)$ . Since (3.2) is trivial on  $Z_{\underline{\varpi}}$ , using (3.4) we see that  $\Upsilon_1|_{Z_{\underline{\varpi}}} = \Delta_0|_{Z_{\underline{\varpi}}}$ .

From the natural bijection

$$B_{\sigma,\lambda}(V) \cong \operatorname{Hom}_{L_P(\mathbb{Q}_p)}\left(\operatorname{c-ind}_{L_P^0}^{L_P(\mathbb{Q}_p)}\sigma, J_P(V)_{\lambda}\widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\operatorname{la}}(Z_{L_P}^0, E)\right)$$

we deduce that  $B_{\sigma,\lambda}(V)$  is also equipped with a natural action of  $\mathcal{Z}_{\Omega} \cong \operatorname{End}_{L_P(\mathbb{Q}_p)}(\operatorname{c-ind}_{L_P^0}^{L_P(\mathbb{Q}_p)}\sigma)$ which commutes with the action of  $\mathcal{Z}_1 \times \mathcal{Z}_0$ . Moreover, we have a natural morphism

$$Z_{L_P}(\mathbb{Q}_p) \longrightarrow \operatorname{End}_{L_P(\mathbb{Q}_p)} \left( \operatorname{c-ind}_{L_p^0}^{L_P(\mathbb{Q}_p)} \sigma \right) \cong \mathcal{Z}_{\Omega}$$

and it is easy to see that the  $Z_{L_P}(\mathbb{Q}_p)$ -action on  $B_{\sigma,\lambda}(V)$  induced by this map coincides with the  $\Delta_0$ -action. In particular (from the last assertion in the previous paragraph) the action  $\Upsilon_1|_{Z_{\infty}}$  can be recovered from the  $Z_{\Omega}$ -action. With (3.5), we finally see that the action of the full  $Z_1$  can be read out from the action of  $Z_0 \times Z_{\Omega}$ .

The following lemma is straightforward (using tensor-Hom adjunction).

**Lemma 3.1.2.** Let M be a finite length  $\mathcal{Z}_{\Omega}$ -module over E, then we have

$$\operatorname{Hom}_{\mathcal{Z}_{\Omega}}(M, B_{\sigma, \lambda}(V)) \xrightarrow{\sim} \operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \left( \operatorname{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma \right) \otimes_{\mathcal{Z}_{\Omega}} M, J_{P}(V)_{\lambda} \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\operatorname{la}}(Z_{L_{P}}^{0}, E) \right).$$

**Lemma 3.1.3.** The action of  $Z_{\underline{\omega}} \times \mathcal{Z}_0$  (seen inside  $\mathcal{Z}_1 \times \mathcal{Z}_0$ ) on  $B_{\sigma,\lambda}(V)$  makes it an essentially admissible representation of  $Z_{\underline{\omega}} \times \mathcal{Z}_0$ .

Proof. We first consider  $J_P(V)_{\lambda}$ . Since  $J_P(V)_{\lambda}$  is an essentially admissible representation of  $L_P(\mathbb{Q}_p)$  (by [40, Thm. 4.2.32] and [35, Lemma 2.8]), the topological dual (with the strong topology)  $J_P(V)^{\vee}_{\lambda}$  is a coadmissible module over  $D(H, E) \widehat{\otimes}_E \Gamma(\widehat{Z_{L_P}(\mathbb{Q}_p)}, \mathcal{O}_{\widehat{Z_{L_P}(\mathbb{Q}_p)}})$  for an arbitrary compact open subgroup H of  $L_P(\mathbb{Q}_p)$ . Shrinking H, we can and do assume H has the form  $H \cong H^D \times Z_H$  where  $H^D$  (resp.  $Z_H$ ) is a compact open subgroup of  $L^D_P(\mathbb{Q}_p)$  (resp. of  $Z_{L_P}(\mathbb{Q}_p)$ ). Hence  $D(H, E) \cong D(H^D, E) \widehat{\otimes}_E D(Z_H, E)$ . Since the  $H^D$ -action on  $J_P(V)_{\lambda}$  is smooth, the action of  $D(H, E) \widehat{\otimes}_E \Gamma(\widehat{Z_{L_P}(\mathbb{Q}_p)})$  on  $J_P(V)^{\vee}_{\lambda}$  factors through

$$D^{\infty}(H^D, E)\widehat{\otimes}_E D(Z_H, E)\widehat{\otimes}_E \Gamma(\widehat{Z_{L_P}(\mathbb{Q}_p)}, \mathcal{O}_{\widehat{Z_{L_P}(\mathbb{Q}_p)}}),$$

and further through its quotient

$$D^{\infty}(H^D, E)\widehat{\otimes}_E \Gamma(\widehat{Z_{L_P}(\mathbb{Q}_p)}, \mathcal{O}_{\widehat{Z_{L_P}(\mathbb{Q}_p)}})$$

via the embedding  $D(Z_H, E) \hookrightarrow \Gamma(\widehat{Z_{L_P}(\mathbb{Q}_p)}, \mathcal{O}_{\widehat{Z_{L_P}(\mathbb{Q}_p)}}).$ 

We now consider  $J_P(V)_{\lambda}\widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z_{L_P}^0, E)$ . We denote  $Z_{L_P}^0$  by  $\mathcal{Z}_0$  (resp.  $\mathcal{Z}'_0$ , resp.  $\mathcal{Z}''_0$ ) when it acts on  $J_P(V)_{\lambda}\widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z_{L_P}^0, E)$  by only acting on the term  $\mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z_{L_P}^0, E)$  (resp. by only acting on  $J_P(V)_{\lambda}$ , resp. by acting via  $Z_{L_P}^0 \xrightarrow{(\text{id}, \text{det}_{L_P}^{-1})} \mathcal{Z}'_0 \times \mathcal{Z}_0$ ). In particular the  $\mathcal{Z}'_0$ -action coincides with  $\Upsilon_1|_{Z_{L_P}^0}$ . Note that (3.2) defines a trivial action of  $H^D \times Z_{\underline{\varpi}}$  on  $\mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z_{L_P}^0, E)$ , so the diagonal action of  $H^D \times Z_{\underline{\varpi}}$  is the same as the one given by only acting on  $J_P(V)_{\lambda}$ . It follows from the previous discussion that the strong dual of  $J_P(V)_{\lambda}\widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z_{L_P}^0, E)$  is a coadmissible module over (recall  $\mathcal{Z}_1 \cong Z_{L_P}(\mathbb{Q}_p)$  acts via  $\Upsilon_1$ )

$$\begin{split} D^{\infty}(H^{D},E)\widehat{\otimes}_{E}\Gamma(\widehat{\mathcal{Z}_{1}},\mathcal{O}_{\widehat{\mathcal{Z}_{1}}})\widehat{\otimes}_{E}D(\mathcal{Z}_{0},E) \\ &\cong D^{\infty}(H^{D},E)\widehat{\otimes}_{E}\Gamma(\widehat{\mathcal{Z}_{\underline{\omega}}},\mathcal{O}_{\widehat{\mathcal{Z}_{\underline{\omega}}}})\widehat{\otimes}_{E}D(\mathcal{Z}_{0}',E)\widehat{\otimes}_{E}D(\mathcal{Z}_{0},E). \end{split}$$

Using the group isomorphism

$$\mathcal{Z}_0'' \times \mathcal{Z}_0 \xrightarrow{\sim} \mathcal{Z}_0' \times \mathcal{Z}_0, (a, b) \mapsto (a, b \det_{L_P}(a)^{-1}),$$

we see that the strong dual of  $J_P(V)_{\lambda} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z^0_{L_P}, E)$  is also a coadmissible module over

$$D_0 := D^{\infty}(H^D, E) \widehat{\otimes}_E \Gamma(\widehat{Z_{\underline{\varpi}}}, \mathcal{O}_{\widehat{Z_{\underline{\varpi}}}}) \widehat{\otimes}_E D(\mathcal{Z}_0'', E) \widehat{\otimes}_E D(\mathcal{Z}_0, E).$$

We now finally consider  $B_{\sigma,\lambda}(V)$ . Let  $M := (\sigma^{\vee} \otimes_E J_P(V)_{\lambda} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E))^{\vee}$  (a Fréchet space). We use  $\mathcal{Z}_0, \mathcal{Z}'_0, \mathcal{Z}''_0$  to denote the corresponding induced action of  $Z_{L_P}^0$  on M that acts trivially on  $\sigma^{\vee}$ . Shrinking H if necessary, we assume the H-action on  $\sigma$  is trivial. Using the previous paragraph, we see that M is a coadmissible module over  $D_0$ . Noting that the action of  $\mathcal{Z}''_0$  coincides with  $\Delta_0|_{Z_{L_P}^0}, B_{\sigma,\lambda}(V)$  is by definition a direct summand of:

$$W := \left(\sigma^{\vee} \otimes_E J_P(V)_{\lambda} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z^0_{L_P}, E)\right)^{H^D} [\mathcal{Z}''_0 = \psi_\sigma].$$

Endow  $M \otimes_{D_0,\kappa_{\psi\sigma}} \left( \Gamma(\widehat{Z_{\underline{\omega}}}, \mathcal{O}_{\widehat{Z_{\underline{\omega}}}}) \widehat{\otimes}_E D(\mathcal{Z}_0, E) \right)$  with the locally convex quotient topology from  $M \otimes_E \left( \Gamma(\widehat{Z_{\underline{\omega}}}, \mathcal{O}_{\widehat{Z_{\underline{\omega}}}}) \widehat{\otimes}_E D(\mathcal{Z}_0, E) \right)$  where  $\kappa_{\psi\sigma}$  denotes the projection  $D_0 \twoheadrightarrow \Gamma(\widehat{Z_{\underline{\omega}}}, \mathcal{O}_{\widehat{Z_{\underline{\omega}}}}) \widehat{\otimes}_E D(\mathcal{Z}_0, E)$  which sends  $D^{\infty}(H^D, E)$  to E by moding out by the augmentation ideal, which sends  $D(\mathcal{Z}''_0, E)$  to E via the character  $\psi_{\sigma}$ , and which is the identity for the other factors of  $D_0$ . Then we have an isomorphism

$$W^{\vee} \cong M \widehat{\otimes}_{D_0, \kappa_{\psi_{\sigma}}} \left( \Gamma(\widehat{Z}_{\underline{\varpi}}, \mathcal{O}_{\widehat{Z}_{\underline{\varpi}}}) \widehat{\otimes}_E D(\mathcal{Z}_0, E) \right)$$

where the right hand side is the Hausdorff completion of  $M \otimes_{D_0,\kappa_{\psi_\sigma}} \left( \Gamma(\widehat{Z}_{\underline{\omega}}, \mathcal{O}_{\widehat{Z}_{\underline{\omega}}}) \widehat{\otimes}_E D(\mathcal{Z}_0, E) \right)$ . As M is coadmissible over  $D_0$ , we deduce that  $W^{\vee}$  is coadmissible over  $\Gamma(\widehat{Z}_{\underline{\omega}}, \mathcal{O}_{\widehat{Z}_{\underline{\omega}}}) \widehat{\otimes}_E D(\mathcal{Z}_0, E)$ . Hence W is an essentially admissible locally  $\mathbb{Q}_p$ -analytic representation of  $Z_{\underline{\omega}} \times \mathcal{Z}_0$ , and so is  $B_{\sigma,\lambda}(V)$ .

We use the notation  $\mathfrak{z}_0 := \mathfrak{z}_{L_P}$  to emphasize the action on  $B_{\sigma,\lambda}(V)$  derived from  $\mathcal{Z}_0$ . For an *E*-algebra  $A, \mathfrak{m} \subset A$  an ideal, and an *A*-module M, we denote by  $M[\mathfrak{m}^{\infty}] := \varinjlim_n M[\mathfrak{m}^n]$  the *A*-submodule of M consisting of elements annihilated by  $\mathfrak{m}^n$  for some  $n \ge 0$ .

**Lemma 3.1.4.** Let  $\mathfrak{d}$  be a weight of  $\mathfrak{z}_0$ , then we have

$$B_{\sigma,\lambda}(V)[\mathfrak{z}_0=\mathfrak{d}] = \bigoplus_{\delta,\chi} B_{\sigma,\lambda}(V)[\mathfrak{z}_0=\mathfrak{d}][\mathfrak{m}_{\chi}][\mathfrak{m}_{\delta}^{\infty}] = \bigoplus_{\mathfrak{m}\in \operatorname{Spm} \mathcal{Z}_{\Omega},\chi} B_{\sigma,\lambda}(V)[\mathfrak{z}_0=\mathfrak{d}][\mathfrak{m}_{\chi}][\mathfrak{m}^{\infty}]$$

where  $\delta$  (resp.  $\chi$ ) runs through the smooth characters of  $\Delta_0 \cong Z_{L_P}(\mathbb{Q}_p)$  (resp. through the locally algebraic characters of  $\mathcal{Z}_0 \cong Z_{L_P}^0$  of weight  $\mathfrak{d}$ ) and  $\mathfrak{m}_{\delta} \subset E[\Delta_0]$  (resp.  $\mathfrak{m}_{\chi} \subset E[\mathcal{Z}_0]$ ) is the maximal ideal associated to  $\delta$  (resp.  $\chi$ ). Moreover, each summand in the direct sums is finite dimensional over E.

*Proof.* We have by definition using (3.2):

$$B_{\sigma,\lambda}(V)[\mathfrak{z}_0=\mathfrak{d}] = (J_P(V)_\lambda \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z^0_{L_P}, E)[\mathfrak{z}_{L_P}=\mathfrak{d}] \otimes_E \sigma^{\vee})^{L_P^0} = (J_P(V)_\lambda[\mathfrak{z}_{L_P}=\mathfrak{d}\circ\mathrm{det}_{L_P}]\widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z^0_{L_P}, E)[\mathfrak{z}_{L_P}=\mathfrak{d}] \otimes_E \sigma^{\vee})^{L_P^0}.$$
(3.6)

As  $C^{\mathbb{Q}_p-\text{la}}(Z_{L_P}^0, E)[\mathfrak{z}_{L_P} = \mathfrak{d}] \cong C^{\infty}(Z_{L_P}^0, E)$  is topologically isomorphism to a direct limit of finite dimensional *E*-vector spaces. We deduce by [64, Prop. 1.2] that (3.6) remains unchanged if  $\widehat{\otimes}_E$  is replaced by  $\widehat{\otimes}_E$ . As in Lemma 3.1.3, let *H* be a compact open subgroup of  $L_P^0$  such that *H* acts trivially on  $\sigma$  and  $H \cong H^D \times Z_H$  where  $H^D := H \cap L_P^D$  and  $Z_H := H \cap Z_{L_P}^0$ . By (3.6), we see that  $B_{\sigma,\lambda}(V)[\mathfrak{z}_0 = \mathfrak{d}]$  is a  $\Delta_0$ -equivariant direct summand of

$$(J_P(V)_{\lambda}[\mathfrak{z}_{L_P} = \mathfrak{d} \circ \det_{L_P}] \otimes_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z^0_{L_P}, E)[\mathfrak{z}_{L_P} = \mathfrak{d}] \otimes_E \sigma^{\vee})^H.$$

It thus suffices to prove the statement for the latter. We have

$$(J_P(V)_{\lambda}[\mathfrak{z}_{L_P} = \mathfrak{d} \circ \det_{L_P}] \otimes_E \mathcal{C}^{\mathbb{Q}_p - \operatorname{la}}(Z^0_{L_P}, E)[\mathfrak{z}_{L_P} = \mathfrak{d}] \otimes_E \sigma^{\vee})^H \cong ((J_P(V)_{\lambda} \otimes_E \sigma^{\vee})^{H^D}[\mathfrak{z}_{L_P} = \mathfrak{d} \circ \det_{L_P}] \otimes_E \mathcal{C}^{\mathbb{Q}_p - \operatorname{la}}(Z^0_{L_P}, E)[\mathfrak{z}_{L_P} = \mathfrak{d}])^{Z_H}.$$

By [53, Thm. 4.10], we have

$$(J_P(V)_{\lambda} \otimes_E \sigma^{\vee})^{H^D} [\mathfrak{z}_{L_P} = \mathfrak{d} \circ \det_{L_P}] = \bigoplus_{\delta'} (J_P(V)_{\lambda} \otimes_E \sigma^{\vee})^{H^D} [\mathfrak{z}_{L_P} = \mathfrak{d} \circ \det_{L_P}] [\mathfrak{m}_{\delta'}^{\infty}]$$

where  $\delta'$  runs though the locally algebraic characters of  $\mathcal{Z}_1 \cong Z_{L_P}(\mathbb{Q}_p)$  of weight  $\mathfrak{d}$  (and  $[\mathfrak{m}_{\delta'}^{\infty}]$  is for the corresponding  $\mathcal{Z}_1$ -action). Moreover, by the proof of *loc. cit.* and the same argument as in the proof of [20, Prop. 4.1], each  $(J_P(V)_\lambda \otimes_E \sigma^{\vee})^{H^D}[\mathfrak{z}_{L_P} = \mathfrak{d} \circ \det_{L_P}][\mathfrak{m}_{\delta'}^{\infty}]$  is finite dimensional over E. For such  $\delta'$ , we easily get:

$$\left( \left( J_P(V)_{\lambda} \otimes_E \sigma^{\vee} \right)^{H^D} [\mathfrak{z}_{L_P} = \mathfrak{d} \circ \det_{L_P}] [\mathfrak{m}_{\delta'}^{\infty}] \otimes_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) [\mathfrak{z}_{L_P} = \mathfrak{d}] \right)^{Z_H}$$

$$\cong \bigoplus_{\delta, \chi} \left( \left( J_P(V)_{\lambda} \otimes_E \sigma^{\vee} \right)^{H^D} [\mathfrak{z}_{L_P} = \mathfrak{d} \circ \det_{L_P}] [\mathfrak{m}_{\delta'}^{\infty}] \otimes_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) [\mathfrak{z}_{L_P} = \mathfrak{d}] \right)^{Z_H} [\mathfrak{m}_{\chi}] [\mathfrak{m}_{\delta'}^{\infty}]$$

$$\cong \bigoplus_{\mathfrak{m}, \chi} \left( \left( J_P(V)_{\lambda} \otimes_E \sigma^{\vee} \right)^{H^D} [\mathfrak{z}_{L_P} = \mathfrak{d} \circ \det_{L_P}] [\mathfrak{m}_{\delta'}^{\infty}] \otimes_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) [\mathfrak{z}_{L_P} = \mathfrak{d}] \right)^{Z_H} [\mathfrak{m}_{\chi}] [\mathfrak{m}^{\infty}]$$

$$(3.7)$$

where  $[\mathfrak{m}_{\delta}^{\infty}]$  is for the smooth action of  $\Delta_0 \cong Z_{L_P}(\mathbb{Q}_p)$ ,  $[\mathfrak{m}_{\chi}]$  for the locally algebraic action of  $\mathcal{Z}_0 \cong Z_{L_P}^0$  (with  $\chi$  as in the statement) and  $[\mathfrak{m}^{\infty}]$  for the smooth action of  $\mathcal{Z}_{\Omega}$  ( $\mathfrak{m}$  as in the statement), noting that smooth representations of  $\mathcal{Z}_0$  over E are semi-simple. By unwinding the actions of  $\mathcal{Z}_1$  and  $\Delta_0$ , if a summand in the right hand side of (3.7) is non-zero, then  $\delta'\delta^{-1}$  is a locally algebraic character of weight  $\mathfrak{d} \circ \det_{L_P}$  which is trivial on  $Z_{\Xi}$  and  $\delta'|_{Z_H} = (\chi \circ \det_{L_P})|_{Z_H}$ . We deduce that for each smooth character  $\delta$  of  $\Delta_0$  and each locally algebraic character  $\chi$  of  $\mathcal{Z}_0$  of weight  $\mathfrak{d}$ , there exist at most finitely many  $\delta'$  such that (3.7) is non-zero. Likewise replacing characters  $\delta$  of  $\Delta_0$  by maximal ideals  $\mathfrak{m}$  of  $\mathcal{Z}_{\Omega}$ . The lemma follows.

We discuss the problem of the choice of  $\sigma$  and  $\lambda$ . We first introduce some notation.

Notation 3.1.5. (1) For a continuous character  $\chi$  of  $Z_{L_P}^0$  (resp. of  $Z_{L_{P_i}}(\mathcal{O}_{F_i})$ ), denote by  $\chi_{\underline{\varpi}}$  (resp.  $\chi_{\overline{\varpi}_i}$ ) the character of  $Z_{L_P}(\mathbb{Q}_p)$  (resp. of  $Z_{L_{P_i}}(F_i)$ ) that is trivial on  $Z_{\underline{\varpi}}$  (resp. on  $Z_{\overline{\varpi}_i} = Z_{\underline{\varpi}} \cap Z_{L_{P_i}}(F_i)$ ) and is equal to  $\chi$  on  $Z_{L_P}^0$  (resp. on  $Z_{L_{P_i}}(\mathcal{O}_{F_i})$ ).

(2) For a continuous character  $\delta$  of  $Z_{L_P}(\mathbb{Q}_p)$  (resp. of  $Z_{L_{P_i}}(F_i)$ ), put  $\delta^0 := \delta|_{Z_{L_P}^0}$  (resp.  $\delta^0 := \delta|_{Z_{L_P_i}(\mathcal{O}_{F_i})}$ ),  $\delta^0_{\underline{\varpi}} := (\delta^0)_{\underline{\varpi}}$  (resp.  $\delta^0_{\underline{\varpi}_i} := (\delta^0)_{\underline{\varpi}_i}$ ) and  $\delta^{\mathrm{unr}}_{\underline{\varpi}} := \delta(\delta^0_{\underline{\varpi}})^{-1}$  (resp.  $\delta^{\mathrm{unr}}_{\underline{\varpi}_i} := \delta(\delta^0_{\underline{\varpi}_i})^{-1}$ ). Hence  $\delta^{\mathrm{unr}}_{\underline{\varpi}}$  (resp.  $\delta^{\mathrm{unr}}_{\underline{\varpi}_i}$ ) is an unramified character of  $Z_{L_P}(\mathbb{Q}_p)$  (resp.  $Z_{L_{P_i}}(F_i)$ ) and  $\delta = \delta^{\mathrm{unr}}_{\underline{\varpi}} \delta^0_{\underline{\varpi}}$  (resp.  $\delta = \delta^{\mathrm{unr}}_{\underline{\varpi}_i} \delta^0_{\underline{\varpi}_i}$ ). Let  $\psi$  be a smooth character of  $Z_{L_P}(\mathbb{Q}_p)$  over  $E, \sigma' := \sigma \otimes_E (\psi^0 \circ \det_{L_P})$  and  $\Omega'$  the Bernstein component associated to  $\sigma'$ . We have an  $L_P(\mathbb{Q}_p)$ -equivariant isomorphism

$$\operatorname{c-ind}_{L_P^0}^{L_P(\mathbb{Q}_p)} \sigma' \xrightarrow{\sim} \left( \operatorname{c-ind}_{L_P^0}^{L_P(\mathbb{Q}_p)} \sigma \right) \otimes_E (\psi \circ \det_{L_P}), \ f \mapsto [g \mapsto f(g)\psi^{-1}(\det_{L_P}(g))] \otimes 1.$$
(3.8)

Denote by  $\iota_{\psi}$  the following composition

$$\iota_{\psi} : \operatorname{End}_{L_{P}(\mathbb{Q}_{p})} \left( \operatorname{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma \right) \xrightarrow{\sim} \operatorname{End}_{L_{P}(\mathbb{Q}_{p})} \left( \left( \operatorname{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma \right) \otimes_{E} \left( \psi \circ \det_{L_{P}} \right) \right) \xrightarrow{\sim} \operatorname{End}_{L_{P}(\mathbb{Q}_{p})} \left( \operatorname{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma' \right)$$

where the first map is given by twisting by  $\psi \circ \det_{L_P}$ , and the second map is induced by (3.8). We easily check that  $\iota_{\psi} : \mathcal{Z}_{\Omega} \to \mathcal{Z}_{\Omega'}$  sends  $\pi \in \operatorname{Spec} \mathcal{Z}_{\Omega'}$  to  $\pi \otimes_E (\psi^{-1} \circ \det_{L_P}) \in \operatorname{Spec} \mathcal{Z}_{\Omega}$ . Let  $\lambda'$  be another integral *P*-dominant weight for *G* such that there exists an integral weight  $\mathfrak{d}$  of  $Z_{L_P}(\mathbb{Q}_p)$ with  $\lambda' - \lambda = \mathfrak{d} \circ \det_{L_P}$  (in particular  $L(\lambda)_P|_{L_P^D(\mathbb{Q}_p)} \cong L(\lambda')_P|_{L_P^D(\mathbb{Q}_p)}$ ). Let  $\delta_{\mathfrak{d}}$  be the algebraic character of  $Z_{L_P}(\mathbb{Q}_p)$  (over *E*) of weight  $\mathfrak{d}$ . Recall  $\delta_{\mathfrak{d}} = \delta_{\mathfrak{d},\mathfrak{w}}^{\operatorname{unr}} \delta_{\mathfrak{d},\mathfrak{w}}^{\mathfrak{d}}$  from Notation 3.1.5.

**Lemma 3.1.6.** There is a natural isomorphism of vector spaces of compact type

$$\operatorname{tw}_{\psi,\mathfrak{d}}: B_{\sigma',\lambda'}(V) \xrightarrow{\sim} B_{\sigma,\lambda}(V)$$

satisfying the following compatibility for the  $Z_{\Omega'} \times Z_{\underline{\omega}} \times Z_0$ - and  $Z_{\Omega} \times Z_{\underline{\omega}} \times Z_0$ -actions, where  $v \in B_{\sigma',\lambda'}(V)$ ,  $\alpha \in Z_{\Omega'}$ ,  $\beta \in Z_{\underline{\omega}}$  and  $\gamma \in Z_0$ :

$$\operatorname{tw}_{\psi,\mathfrak{d}}\left((\alpha,\beta,\gamma)\cdot v\right) = (\delta_{\mathfrak{d}}\circ\operatorname{det}_{L_{P}})(\beta)(\delta_{\mathfrak{d}}\psi)^{-1}(\gamma)\left(\left(\iota_{\delta_{\mathfrak{d},\underline{\varpi}}^{\operatorname{unr}}}\left(\iota_{\psi_{\underline{\varpi}}^{0}}^{-1}(\alpha)\right),\beta,\gamma\right)\cdot\operatorname{tw}_{\psi,\mathfrak{d}}(v)\right).$$

*Proof.* We have  $\mathcal{Z}_0$ -equivariant isomorphisms (using (3.8) for the second):

$$\operatorname{tw}_{0}: B_{\sigma',\lambda'}(V) \xrightarrow{\sim} \operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \operatorname{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma', J_{P}(V)_{\lambda} \otimes_{E} (\delta_{\mathfrak{d}}^{-1} \circ \operatorname{det}_{L_{P}}) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\operatorname{la}}(Z_{L_{P}}^{0}, E) \right) \xrightarrow{\sim} \operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \operatorname{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma, J_{P}(V)_{\lambda} \otimes_{E} (\delta_{\mathfrak{d}}^{-1} \circ \operatorname{det}_{L_{P}}) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\operatorname{la}}(Z_{L_{P}}^{0}, E) \otimes_{E} (\psi^{-1} \circ \operatorname{det}_{L_{P}}) \right)$$

such that  $\operatorname{tw}_0((\alpha,\beta)\cdot v) = (\psi \circ \operatorname{det}_{L_P})(\beta) ((\iota_{\psi}^{-1}(\alpha),\beta)\cdot \operatorname{tw}_0(v))$ . We have an isomorphism

$$((\delta_{\mathfrak{d}}^{-1}\psi^{-1})^{0}_{\underline{\varpi}} \circ \det_{L_{P}})\widehat{\otimes}_{E}\mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E) \xrightarrow{\sim} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E)$$

$$1 \otimes f \longmapsto [z \mapsto f(z)(\delta_{\mathfrak{d}}\psi)^{0}(z)]$$

$$(3.9)$$

which is  $L_P(\mathbb{Q}_p)$ -equivariant (but not  $\mathcal{Z}_0$ -equivariant) where  $L_P(\mathbb{Q}_p)$  acts on the left hand side by the diagonal action (with  $L_P(\mathbb{Q}_p)$  acting on  $\mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z^0_{L_p}, E)$  via (3.2)). This isomorphism, together with tw<sub>0</sub>, induce an isomorphism

$$\operatorname{tw}_{1}: B_{\sigma',\lambda'} \xrightarrow{\sim} \operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \operatorname{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma, J_{P}(V)_{\lambda} \otimes_{E} \left( (\delta_{\mathfrak{d}}^{-1}\psi^{-1})_{\underline{\varpi}}^{\operatorname{unr}} \circ \operatorname{det}_{L_{P}} \right) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\operatorname{la}}(Z_{L_{P}}^{0}, E) \right)$$

satisfying  $\operatorname{tw}_1((\alpha,\beta,\gamma)\cdot v) = (\psi \circ \operatorname{det}_{L_P})(\beta)(\delta_{\mathfrak{d}}^{-1}\psi^{-1})(\gamma)((\iota_{\psi}^{-1}(\alpha),\beta,\gamma)\cdot \operatorname{tw}_1(v))$  for  $(\alpha,\beta,\gamma) \in \mathcal{Z}_{\Omega'} \times Z_{\underline{\omega}} \times \mathcal{Z}_0$ . We have isomorphisms:

$$\operatorname{tw}_{2} : \operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \operatorname{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma, J_{P}(V)_{\lambda} \otimes_{E} ((\delta_{\mathfrak{d}}^{-1}\psi^{-1})_{\underline{\varpi}}^{\operatorname{unr}} \circ \det_{L_{P}}) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\operatorname{la}}(Z_{L_{P}}^{0}, E) \right)$$

$$\xrightarrow{\sim} \operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( (\operatorname{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma) \otimes_{E} ((\delta_{\mathfrak{d}}\psi)_{\underline{\varpi}}^{\operatorname{unr}} \circ \det_{L_{P}}), J_{P}(V)_{\lambda} \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\operatorname{la}}(Z_{L_{P}}^{0}, E) \right)$$

$$\xrightarrow{\sim} \operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \operatorname{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma, J_{P}(V)_{\lambda} \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\operatorname{la}}(Z_{L_{P}}^{0}, E) \right)$$

where the second isomorphism is induced by (3.8) (for the unramified character  $(\delta_0 \psi)_{\underline{\omega}}^{\text{unr}}$ ) and tw<sub>2</sub> means the composition. It is straightforward to check that tw<sub>2</sub> is  $\mathcal{Z}_0$ -equivariant and satisfies for  $(\alpha, \beta) \in \mathcal{Z}_\Omega \times Z_{\underline{\omega}}$ :

$$\operatorname{tw}_2((\alpha,\beta)\cdot v) = ((\delta_{\mathfrak{d}}^{-1}\psi^{-1})^{\operatorname{unr}}_{\underline{\varpi}} \circ \operatorname{det}_{L_P})(\beta) \big( (\iota_{(\delta_{\mathfrak{d}}\psi)^{\operatorname{unr}}_{\underline{\varpi}}}(\alpha),\beta) \cdot \operatorname{tw}_2(v) \big).$$

We put  $\operatorname{tw}_{\psi,\mathfrak{d}} := \operatorname{tw}_2 \circ \operatorname{tw}_1$ . Forgetting the actions of  $\mathcal{Z}_{\Omega}$  and  $\mathcal{Z}_{\Omega'}$ , we see that  $\operatorname{tw}_{\psi,\mathfrak{d}}$  is given by the following composition

$$B_{\sigma',\lambda'}(V) \cong \left( (\sigma')^{\vee} \otimes_E J_P(V)_{\lambda} \otimes_E (\delta_{\mathfrak{d}}^{-1} \circ \det_{L_P}) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) \right)^{L_P^0}$$
  
$$\xrightarrow{\sim} \left( \sigma^{\vee} \otimes_E J_P(V)_{\lambda} \otimes_E ((\delta_{\mathfrak{d},\underline{\varpi}}^{\mathrm{unr}})^{-1} \circ \det_{L_P}) \otimes_E ((\delta_{\mathfrak{d},\underline{\varpi}}^0 \psi_{\underline{\varpi}}^0)^{-1} \circ \det_{L_P}) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) \right)^{L_P^0}$$
  
$$\xrightarrow{\sim} \left( \sigma^{\vee} \otimes_E J_P(V)_{\lambda} \otimes_E ((\delta_{\mathfrak{d},\underline{\varpi}}^{\mathrm{unr}})^{-1} \circ \det_{L_P}) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) \right)^{L_P^0}$$
  
$$\cong \left( \sigma^{\vee} \otimes_E J_P(V)_{\lambda} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) \right)^{L_P^0} \cong B_{\sigma,\lambda}(V)$$

where the third isomorphism is induced by (3.9). In particular,  $tw_{\psi,\mathfrak{d}}$  is a topological isomorphism.

Let  $(J'_{i,j}, (\sigma^0_{i,j})')$  be another maximal simple type of  $\Omega_{i,j}$  and  $K'_{i,j}$  be a maximal compact open subgroup of  $\operatorname{GL}_{n_{i,j}}(F_i)$  containing  $J'_{i,j}$ . Define  $(\sigma')^0 := \boxtimes_{i,j}(\sigma^0_{i,j})', \sigma'_{i,j} := \operatorname{c-ind}_{J'_{i,j}}^{K'_{i,j}}(\sigma^0_{i,j})',$  $\sigma' := \boxtimes_{i,j}\sigma'_{i,j}$  and  $K' := \prod_{i,j}K'_{i,j}$  (so that  $\sigma'$  is an absolutely irreducible representation of K' over E). By [23, Cor. 7.6],  $(J', (\sigma')^0)$  is conjugate to  $(J, \sigma^0)$ , i.e. there exist  $h \in L_P(\mathbb{Q}_p)$  such that  $J' = hJh^{-1}$  and a bijection  $\iota_h : \sigma^0 \xrightarrow{\sim} (\sigma^0)'$  satisfying  $\iota_h(\alpha v) = (h\alpha h^{-1})\iota_h(v)$  for  $\alpha \in J, v \in \sigma^0$ . The morphism  $\iota_h$  induces a bijection:

$$\tilde{\iota}_h : \operatorname{c-ind}_J^{L_P(\mathbb{Q}_p)} \sigma^0 \xrightarrow{\sim} \operatorname{c-ind}_{J'}^{L_P(\mathbb{Q}_p)}(\sigma')^0, f \mapsto [g \mapsto \iota_h(f(h^{-1}gh))]$$

satisfying  $\tilde{\iota}_h(gf) = (hgh^{-1})(\tilde{\iota}_h(f))$  for  $g \in L_P(\mathbb{Q}_p)$ . We deduce isomorphisms

$$\mathcal{Z}_{\Omega} \cong \operatorname{End}_{L_{P}(\mathbb{Q}_{p})}(\operatorname{c-ind}_{J}^{L_{P}(\mathbb{Q}_{p})}\sigma^{0}) \cong \operatorname{End}_{L_{P}(\mathbb{Q}_{p})}(\operatorname{c-ind}_{J'}^{L_{P}(\mathbb{Q}_{p})}(\sigma')^{0}).$$

We fix such isomorphisms in the rest of this paragraph. We define:

$$B_{\sigma',\lambda}(V) := \left( (\sigma')^{\vee} \otimes_E J_P(V)_{\lambda} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z^0_{L_P}, E) \right)^{K'}$$

where K' acts diagonally on  $J_P(V)_{\lambda} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z^0_{L_P}, E)$  via the embedding  $K' \hookrightarrow L_P(\mathbb{Q}_p)$  and (3.2). We endow  $B_{\sigma',\lambda}(V)$  with an action of  $\mathcal{Z}_0 \cong Z^0_{L_P}, Z_{\underline{\varpi}}$  and  $\mathcal{Z}_\Omega$  as in § 3.1.2.

**Lemma 3.1.7.** There exists a natural isomorphism of vector spaces of compact type which is equivariant under the action of  $Z_{\Omega} \times Z_{\overline{\omega}} \times Z_0$ :

$$B_{\sigma,\lambda}(V) \cong B_{\sigma',\lambda}(V).$$

*Proof.* We have a  $\mathcal{Z}_{\Omega} \times Z_{\underline{\omega}} \times \mathcal{Z}_{0}$ -equivariant isomorphism

$$B_{\sigma,\lambda}(V) \cong \operatorname{Hom}_J(\sigma^0, J_P(V)_\lambda \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \operatorname{la}}(Z^0_{L_P}, E))$$

and a similar isomorphism for  $B_{\sigma',\lambda}(V)$ . Let  $W := J_P(V)_{\lambda} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z^0_{L_P}, E)$ , one can check that the following diagram commutes

where the top map is given by  $f \mapsto [v \mapsto h(f(\iota_h^{-1}(v)))]$ , and the bottom map is given by  $F \mapsto [w \mapsto h(F(\tilde{\iota}_h^{-1}(w)))]$ . One sees moreover that the top map is a topological isomorphism, that the bottom map is bijective and  $\mathcal{Z}_{\Omega} \times Z_{\underline{\omega}} \times \mathcal{Z}_0$ -equivariant. The lemma follows.  $\Box$ 

By Lemma 3.1.7, the  $\mathcal{Z}_{\Omega} \times Z_{\underline{\omega}} \times \mathcal{Z}_{0}$ -module  $B_{\sigma,\lambda}(V)$  does not depend on the choice  $\sigma$ . We denote hence  $B_{\Omega,\lambda}(V) := B_{\sigma,\lambda}(V)$  in the sequel.

## 3.1.3 Local Bernstein eigenvarieties

We keep the setting of § 3.1.2. We construct certain rigid analytic spaces parametrizing irreducible  $\mathcal{Z}_{\Omega} \times \mathcal{Z}_{0}$ -submodules of  $B_{\Omega,\lambda}(V)$ .

By Lemma 3.1.3 and [42, § 6.4], there exists a coherent sheaf  $\mathcal{M}_{\Omega,\lambda}(V)$  over the rigid analytic space  $\widehat{Z}_{\underline{\omega}} \times \widehat{Z}_0$  such that  $B_{\Omega,\lambda}(V)$  is isomorphic to the global sections of  $\mathcal{M}_{\Omega,\lambda}(V)$ . Moreover,  $\mathcal{M}_{\Omega,\lambda}(V)$  is equipped with an  $\mathcal{O}_{\widehat{Z}_{\underline{\omega}} \times \widehat{Z}_0}$ -linear action of  $\mathcal{Z}_\Omega$ . Using the fact that the action of  $Z_{\underline{\omega}}$ factors through  $\mathcal{Z}_\Omega$  (see before Lemma 3.1.2), we conclude that  $\mathcal{M}_{\Omega,\lambda}(V)$  gives rise to a coherent sheaf, still denoted by  $\mathcal{M}_{\Omega,\lambda}(V)$ , over (Spec  $\mathcal{Z}_\Omega)^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_0$  such that

$$\Gamma\left((\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0}}, \mathcal{M}_{\Omega,\lambda}(V)\right) \cong B_{\Omega,\lambda}(V)^{\vee}.$$
(3.10)

We let  $\operatorname{Supp} \mathcal{M}_{\Omega,\lambda}(V)$  be the Zariski-closed support of  $\mathcal{M}_{\Omega,\lambda}(V)$  (defined by the annihilator of  $\mathcal{M}_{\Omega,\lambda}(V)$ ).

**Proposition 3.1.8.** Let  $x = (\pi_{L_P}, \chi) := ((\pi_{i,j}), (\chi_{i,j})) \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0}$ . Then we have a bijection of k(x)-vector spaces  $(\chi_{\underline{\omega}} \text{ as in Notation } \underline{3.1.5})$ :

$$(x^*\mathcal{M}_{\Omega,\lambda}(V))^{\vee} \cong \operatorname{Hom}_{L_P(\mathbb{Q}_p)}(\pi_{L_P}\otimes_{k(x)}(\chi_{\underline{\varpi}}\circ \det_{L_P})\otimes_E L(\lambda)_P, J_P(V)).$$

In particular,  $(\pi_{L_P}, \chi) \in \operatorname{Supp} \mathcal{M}_{\Omega,\lambda}(V)$  if and only if there is an  $L_P(\mathbb{Q}_p)$ -equivariant embedding

$$\pi_{L_P} \otimes_{k(x)} (\chi_{\underline{\varpi}} \circ \det_{L_P}) \otimes_E L(\lambda)_P \hookrightarrow J_P(V).$$

*Proof.* By definition, we have a bijection

$$(x^*\mathcal{M}_{\Omega,\lambda}(V))^{\vee} \cong B_{\Omega,\lambda}(V)[\mathcal{Z}_{\Omega} = \pi_{L_P}, \mathcal{Z}_0 = \chi].$$

By unwinding  $B_{\Omega,\lambda}(V) \cong B_{\sigma,\lambda}(V)$ , we see that the right hand side is isomorphic to (denoting

by  $\mathfrak{m}_{\pi_{L_P}}$  the maximal ideal of  $\mathcal{Z}_{\Omega}$  corresponding to  $\pi_{L_P}$ )

$$\operatorname{Hom}_{\mathcal{Z}_{\Omega}}\left(\mathcal{Z}_{\Omega}/\mathfrak{m}_{\pi_{L_{P}}},\operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})}\left(\operatorname{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})}\sigma,J_{P}(V)_{\lambda}\otimes_{k(x)}(\chi_{\underline{\varpi}}^{-1}\circ\operatorname{det}_{L_{P}})\right)\right)$$
  
$$\xrightarrow{\sim}\operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})}\left(\pi_{L_{P}},J_{P}(V)_{\lambda}\otimes_{k(x)}(\chi_{\underline{\varpi}}^{-1}\circ\operatorname{det}_{L_{P}})\right)$$
  
$$\cong\operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})}\left(\pi_{L_{P}}\otimes_{k(x)}(\chi_{\underline{\varpi}}\circ\operatorname{det}_{L_{P}}),J_{P}(V)_{\lambda}\right)$$
  
$$\cong\operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})}\left(\pi_{L_{P}}\otimes_{k(x)}(\chi_{\underline{\varpi}}\circ\operatorname{det}_{L_{P}}),J_{P}(V)\otimes_{E}L(\lambda)_{P}^{\vee}\right)$$
  
$$\cong\operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})}\left(\pi_{L_{P}}\otimes_{k(x)}(\chi_{\underline{\varpi}}\circ\operatorname{det}_{L_{P}})\otimes_{E}L(\lambda)_{P},J_{P}(V)\right)$$

where the first isomorphism follows from Lemma 3.1.2, the second is obvious, the third follows from (3.1) and the fact that  $\pi_{L_P} \otimes_E (\chi_{\underline{\varpi}} \circ \det_{L_P})$  is smooth for the  $L_P^D(\mathbb{Q}_p)$ -action, and the fourth is easily induced by the natural map  $L(\lambda)_P \otimes_E L(\lambda)_P^{\vee} \to E$ . The proposition follows.  $\Box$ 

The following proposition easily follows from Lemma 3.1.6:

**Proposition 3.1.9.** With the setting of Lemma 3.1.6, we have an isomorphism of rigid analytic spaces

$$\operatorname{Supp} \mathcal{M}_{\Omega,\lambda}(V) \xrightarrow{\sim} \operatorname{Supp} \mathcal{M}_{\Omega',\lambda'}(V), \ (\pi_{L_P},\chi) \mapsto \big(\pi_{L_P} \otimes_E \big( (\psi^0_{\underline{\varpi}}(\delta^{\operatorname{unr}}_{\mathfrak{d},\underline{\varpi}})^{-1}) \circ \det_{L_P} \big), \chi(\psi^0)^{-1}(\delta^0_{\mathfrak{d}})^{-1} \big).$$

**Remark 3.1.10.** Assume  $P = B_{\mathscr{I}}$ , we have  $\mathcal{Z}_{\Omega} \cong Z_{\underline{\varpi}}$ ,  $L_P(\mathbb{Q}_p) \cong T_{\mathscr{I}}(\mathbb{Q}_p)$  and  $L(\lambda)_P = \delta_{\lambda}$ . Using the isomorphism

$$\iota_{\Omega,\lambda} : (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0} \xrightarrow{\sim} \widehat{Z_{L_P}(\mathbb{Q}_p)} \cong \widehat{T_{\mathscr{I}}(\mathbb{Q}_p)}, \quad (\pi_{L_P}, \chi) \mapsto \pi_{L_P} \chi_{\underline{\varpi}} \delta_{\lambda}$$

we can also view  $\mathcal{M}_{\Omega,\lambda}(V)$  as a coherent sheaf over  $\widehat{T_{\mathscr{I}}(\mathbb{Q}_p)}$  and  $\operatorname{Supp} \mathcal{M}_{\Omega,\lambda}(V)$  as a closed rigid subspace of  $\widehat{T_{\mathscr{I}}(\mathbb{Q}_p)}$ . By Proposition 3.1.9, the resulting rigid subspaces  $\operatorname{Supp} \mathcal{M}_{\Omega,\lambda}(V)$  of  $\widehat{T_{\mathscr{I}}(\mathbb{Q}_p)}$ for different  $(\Omega, \lambda)$  are all the same. Moreover, in this case, one can directly associate to the (essentially admissible) locally analytic representation  $J_P(V)$  of  $T_{\mathscr{I}}(\mathbb{Q}_p)$  a coherent sheaf  $\mathcal{M}(V)$ over  $\widehat{T_{\mathscr{I}}(\mathbb{Q}_p)}$ , without using  $L(\lambda)_P$  and without tensoring by the factor  $\mathcal{C}^{\mathbb{Q}_p-\operatorname{la}}(Z_{L_P}^0, E)$  such that  $\Gamma(\widehat{T_{\mathscr{I}}(\mathbb{Q}_p)}, \mathcal{M}(V)) \cong J_P(V)^{\vee}$ . For each point  $x = \delta$  of  $\widehat{T_{\mathscr{I}}(\mathbb{Q}_p)}$ , the fibre  $(x^*\mathcal{M}(V))^{\vee}$  is naturally isomorphic to  $J_P(V)[Z_{L_P}(\mathbb{Q}_p) = \delta]$ . By Proposition 3.1.8 (and using  $\iota_{\Omega,\lambda}$ ), we see that  $\mathcal{M}(V)$ and  $\mathcal{M}_{\Omega,\lambda}(V)$  have isomorphic fibres (as E-vector spaces) at each point.

Set (see (2.1) for  $\mu_{\Omega_{i,i}}$ ):

$$\mu_{\Omega} := \big\{ (\psi_{i,j})_{\substack{i \in \mathscr{I} \\ j=1,\dots,r_i}} : Z_{L_P}(\mathbb{Q}_p) \to E^{\times} \mid \psi_{i,j} \in \mu_{\Omega_{i,j}} \big\},$$

we define an action of  $\mu_{\Omega}$  on  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$  such that  $\psi = (\psi_{i,j}) \in \mu_{\Omega}$  sends  $((\pi_{i,j}), (\chi_{i,j}))$  to

$$\left(\left(\pi_{i,j}\otimes_E \operatorname{unr}(\psi_{i,j}(\varpi_i))\right), \left(\chi_{i,j}\psi_{i,j}|_{\mathcal{O}_{F_i}^{\times}}\right)\right)$$

By Proposition 3.1.9 (or by Proposition 3.1.8), we have

**Corollary 3.1.11.** For  $x \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$ , we have  $x \in \operatorname{Supp} \mathcal{M}_{\Omega,\lambda}(V)$  if and only if  $\psi(x) \in \operatorname{Supp} \mathcal{M}_{\Omega,\lambda}(V)$  for one (or any)  $\psi \in \mu_{\Omega}$ .

Finally we show that, under certain assumptions, Supp  $\mathcal{M}_{\Omega,\lambda}(V)$  is closely related to Fredholm hypersurfaces. We first unwind a bit the definition of the Jacquet-Emerton modules.

Let H be a compact open subgroup of  $G_p = G(\mathbb{Q}_p)$ , and  $N_H := H \cap N_P(\mathbb{Q}_p)$ . Let  $L_P(\mathbb{Q}_p)^+ = \{z \in L_P(\mathbb{Q}_p) \mid zN_H z^{-1} \subseteq N_H\}$ . Recall that  $V^{N_H}$  is equipped with a natural Hecke action of  $L_P(\mathbb{Q}_p)^+$  given by

$$\pi_z(v) = \frac{1}{|N_H/(zN_H z^{-1})|} \sum_{g \in N_H/(zN_H z^{-1})} gz(v).$$
(3.11)

Then  $J_P(V)$  is defined to be  $(V^{N_H})_{\rm fs}$ , where  $(-)_{\rm fs}$  denotes the finite slope part functor for the action of  $Z_{L_P}(\mathbb{Q}_p)^+ := L_P(\mathbb{Q}_p)^+ \cap Z_{L_P}(\mathbb{Q}_p)$  (cf. [40, § 3.2]).

**Lemma 3.1.12.** We have  $J_P(V)_{\lambda} \cong \left( (V^{N_H} \otimes_E L(\lambda)_P^{\vee})^{\mathfrak{l}_P^D} \right)_{\mathrm{fs}} \cong \left( (V^{N_H} \otimes_E L(\lambda)_P^{\vee})_{\mathrm{fs}} \right)^{\mathfrak{l}_P^D}$ .

Proof. Recall  $J_P(V)_{\lambda} \cong (J_P(V) \otimes_E L(\lambda)_P^{\vee})^{\mathfrak{l}_P^D}$  by (3.1). By [40, Prop. 3.2.9], we have  $J_P(V) \otimes_E L(\lambda)_P^{\vee} \cong (V^{N_H} \otimes_E L(\lambda)_P^{\vee})_{\mathrm{fs}}$ , where the Hecke action of  $z \in Z_{L_P}(\mathbb{Q}_p)^+$  on  $V^{N_H} \otimes_E L(\lambda)_P^{\vee}$  is given by  $\pi_z \otimes z$ . It is clear that the action of  $\mathfrak{l}_P^D$  commutes with  $Z_{L_P}(\mathbb{Q}_p)^+$ . By [40, Prop. 3.2.11], the lemma follows.

We fix  $\sigma$  as in (3.3) and recall that we have an isomorphism  $B_{\Omega,\lambda}(V) \cong B_{\sigma,\lambda}(V)$ .

**Lemma 3.1.13.** We have an isomorphism of locally analytic representations of  $Z_{\underline{\omega}} \times \mathcal{Z}_0$ :

$$B_{\Omega,\lambda}(V) \cong \left( \left( (V^{N_H} \otimes_E L(\lambda)_P^{\vee}) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) \otimes_E \sigma^{\vee} \right)^{L_P^0} \right)_{\mathrm{fs}}$$

where  $Z_{\underline{\alpha}}$  (resp.  $\mathcal{Z}_0$ ) acts on the RHS by its action on  $V^{N_H} \otimes_E L(\lambda)_P^{\vee}$  (resp. on  $\mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z_{L_P}^0, E)$ ).

*Proof.* We have isomorphisms

$$B_{\Omega,\lambda}(V) \cong \left( \left( (V^{N_H} \otimes_E L(\lambda)_P^{\vee})_{P}^{l_P^{D}} \right)_{\mathrm{fs}} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) \otimes_E \sigma^{\vee} \right)^{L_P^0} \\ \cong \left( \left( (V^{N_H} \otimes_E L(\lambda)_P^{\vee})_{P}^{l_P^{D}} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) \otimes_E \sigma^{\vee} \right)_{\mathrm{fs}} \right)^{L_P^0} \\ \cong \left( \left( (V^{N_H} \otimes_E L(\lambda)_P^{\vee})_{P}^{l_P^{D}} \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) \otimes_E \sigma^{\vee} \right)^{L_P^0} \right)_{\mathrm{fs}} \\ \cong \left( \left( (V^{N_H} \otimes_E L(\lambda)_P^{\vee}) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) \otimes_E \sigma^{\vee} \right)^{L_P^0} \right)_{\mathrm{fs}},$$

where the first isomorphism follows from Lemma 3.1.12, the second from [40, Prop. 3.2.9], the third from [40, Prop. 3.2.11], and the last from the fact that  $\mathfrak{l}_P^D$  acts trivially on the factors  $\mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z_{L_P}^0, E)$  and  $\sigma^{\vee}$ .

**Remark 3.1.14.** Let  $z \in Z_{L_P}(\mathbb{Q}_p)^+$  and let  $Y_z$  be the subgroup of  $Z_{L_P}(\mathbb{Q}_p)$  generated by z. Assume  $Y_z Z_{L_P}(\mathbb{Q}_p)^+ = Z_{L_P}(\mathbb{Q}_p)$ . By [40, Prop. 3.2.27], the lemmas also hold with " $(-)_{\mathrm{fs}}$ " (for the whole group  $Z_{L_P}(\mathbb{Q}_p)$ ) replaced by the finite slope part " $(-)_{Y_z-\mathrm{fs}}$ " for  $Y_z$ .

We now take H uniform pro-p in  $G_p$  and  $z \in Z_{\underline{\omega}}$  such that

(1)  $H \cong (N_{P^-}(\mathbb{Q}_p) \cap H) \times (L_P(\mathbb{Q}_p) \cap H) \times (N_P(\mathbb{Q}_p) \cap H) =: N_H^- \times L_H \times N_H$ , and H is normalized by  $L_P^0$ ;

- (2)  $L_H \cong (L_H \cap L_P^D(\mathbb{Q}_p)) \times (L_H \cap Z_{L_P}(\mathbb{Q}_p)) =: L_H^D \times Z_{L_H};$
- (3)  $L_H$  acts trivially on  $\sigma$ ;
- (4)  $z \in Z_{L_P}(\mathbb{Q}_p)^+$  and satisfies  $\cap_n(z^n N_H z^{-n}) = 0$ ,  $N_H^- \subset z N_H^- z^{-1}$  and  $Y_z Z_{L_P}(\mathbb{Q}_p)^+ = Z_{L_P}(\mathbb{Q}_p)$ where  $Y_z$  is the subgroup of  $Z_{L_P}(\mathbb{Q}_p)$  generated by z;
- (5) *H* is normalized by  $z^{-1}N_H z$ .

The existence of H satisfying (1), (2) and (3) is clear. It is also clear that there exists  $z \in Z_{L_P}(\mathbb{Q}_p)^+$  such that (4) holds. By multiplying z by an element in  $Z_{L_P}^0$ , we can take  $z \in Z_{\underline{\omega}}$ . Finally, replacing H by  $H^{p^m}$  for some  $m \geq 1$ , (5) also holds (with the other properties unchanged). As an example, one can take H to be  $\prod_{i \in \mathscr{I}} (1 + \varpi_i^k M_n(\mathcal{O}_{F_i}))$  for k sufficiently large, and  $z = \prod_{i \in \mathscr{I}} z_i$  with

$$z_i := \operatorname{diag}\left(\underbrace{\varpi_i^{r_i-1}, \dots, \varpi_i^{r_i-1}}_{n_{i,1}}, \underbrace{\varpi_i^{r_i-2}, \dots, \varpi_i^{r_i-2}}_{n_{i,2}}, \dots, \underbrace{1, \dots, 1}_{n_{i,r_i}}\right) \in Z_{L_{P_i}}(L_i).$$

By (1) and (5), one can deduce that

$$H' := (zHz^{-1})N_H \cong (zN_H^- z^{-1}) \times L_H \times N_H$$

contains H (note that H' is also an open uniform pro-p subgroup of  $G_p$ ). The following proposition is analogous to [17, Prop. 5.3] (see also [40, Prop. 4.2.36]).

**Proposition 3.1.15.** Assume that  $V|_{H'} \cong C^{\mathbb{Q}_p-\text{la}}(H', E)^{\oplus k}$  for some  $k \in \mathbb{Z}_{\geq 1}$ . There exist an admissible covering of  $\widehat{\mathcal{Z}}_0$  by affinoid opens  $\mathcal{U}_1 \subset \mathcal{U}_2 \subset \cdots \subset \mathcal{U}_h \subset \cdots$  and the following data for any  $h \geq 1$  where  $A_h := \mathcal{O}_{\widehat{\mathcal{Z}}_0}(\mathcal{U}_h)$ :

- a Banach  $A_h$ -module  $M_h$  satisfying the condition (Pr) of [24];
- an  $A_h$ -linear compact operator, denoted by  $z_h$ , on the  $A_h$ -module  $M_h$ ;
- $A_h$ -linear continuous maps  $\begin{cases} \alpha_h : M_h \longrightarrow M_{h+1} \widehat{\otimes}_{A_{h+1}} A_h \\ \beta_h : M_{h+1} \widehat{\otimes}_{A_{h+1}} A_h \longrightarrow M_h \end{cases}$  such that  $\beta_h \circ \alpha_h = z_h$  and  $\alpha_h \circ \beta_h = z_{h+1}$  with  $\beta_h$   $A_h$ -compact;
- a topological  $\mathcal{O}(\widehat{\mathcal{Z}_0})$ -linear isomorphism

$$M := \left( \left( (V^{N_H} \otimes_E L(\lambda)_P^{\vee}) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) \otimes_E \sigma^{\vee} \right)^{L_P^0} \right)^{\vee} \cong \varprojlim_h M_h$$

commuting with the action induced by  $(\pi_z \otimes z) \otimes 1 \otimes 1$  on the LHS and the action of  $(z_h)_{h \ge 1}$ on the RHS.

One can visualize all the above conditions in the following commutative diagram

$$M \longrightarrow \cdots \longrightarrow M_{h+1} \longrightarrow M_{h+1} \otimes_{A_{h+1}} A_h \xrightarrow{\beta_h} M_h \longrightarrow \cdots$$

$$\downarrow \pi_z \qquad \qquad \downarrow z_{h+1} \qquad z_{h+1} \otimes 1_{A_h} \downarrow \qquad \checkmark \alpha_h \qquad \downarrow z_h$$

$$M \longrightarrow \cdots \longrightarrow M_{h+1} \longrightarrow M_{h+1} \otimes_{A_{h+1}} A_h \xrightarrow{\beta_h} M_h \longrightarrow \cdots$$

*Proof.* We put

$$W := \left( (V^{N_H} \otimes_E L(\lambda)_P^{\vee}) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_P}^0, E) \otimes_E \sigma^{\vee} \right)^{L_H}$$
(3.12)

and  $N := W^{\vee}$ . By definition, M is a direct summand of N equivariant under the action of  $\mathcal{Z}_0$  and of  $(\pi_z \otimes z) \otimes 1 \otimes 1$ . Let s := |H'/H|, thus  $V|_H \cong \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(H, E)^{\oplus ks}$ . We have then an  $L_H$ -equivariant isomorphism  $V^{N_H} \cong (\mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(N_H^-, E) \otimes_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(L_H, E))^{\oplus ks}$ . Let  $r := \dim_E L(\lambda)_P^{\vee}$ , we have  $L_H$ equivariant isomorphisms (see for example [35, Lemma 2.19] for the second isomorphism):

$$V^{N_{H}} \otimes_{E} L(\lambda)_{P}^{\vee} \cong \left( \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(N_{H}^{-}, E) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(L_{H}, E) \right)^{\oplus ks} \otimes_{E} L(\lambda)_{P}^{\vee}$$
$$\cong \left( \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(N_{H}^{-}, E) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(L_{H}, E) \right)^{\oplus rks}$$
$$\cong \left( \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(N_{H}^{-}, E) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(L_{H}^{D}, E) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z_{L_{H}}, E) \right)^{\oplus rks}.$$

There exists s' such that  $(\mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z_{L_P}^0, E) \otimes_E \sigma^{\vee})|_{L_H} \cong \mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z_{L_H}, E)^{\oplus s'}$ , where the  $L_H$ -action on the right hand side is induced from the regular  $Z_{L_H}$ -action via  $L_H \xrightarrow{\det_{L_P}^{-1}} Z_{L_H}$ . Let m := krss', we then have

$$W \cong \left( \left( \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(N_H^-, E) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(L_H^D, E) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_H}, E) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_H}, E) \right)^{L_H} \right)^{\oplus m}$$
  
$$\cong \left( \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(N_H^-, E) \widehat{\otimes}_E \left( \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_H}, E) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_H}, E) \right)^{Z_{L_H}} \right)^{\oplus m}$$
  
$$\cong \left( \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(N_H^-, E) \widehat{\otimes}_E \left( \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_H} \times Z_{L_H}, E) \right)^{Z_{L_H}} \right)^{\oplus m},$$

where the  $Z_{L_H}$ -fixed vectors in the last term are for the  $Z_{L_H}$ -action induced from the regular  $Z_{L_H} \times Z_{L_H}$ -action via the morphism  $Z_{L_H} \hookrightarrow Z_{L_H} \times Z_{L_H}$ ,  $a \mapsto (a, \det_{L_P}^{-1}(a))$ . Using the isomorphism  $Z_{L_H} \times Z_{L_H} \xrightarrow{\sim} Z_{L_H} \times Z_{L_H}$ ,  $(a, b) \longmapsto (a, \det_{L_P}^{-1}(a)b)$ , we finally deduce a  $Z_{L_H}$ -equivariant isomorphism:

$$W \cong \left( \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(N_H^-, E) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p - \mathrm{la}}(Z_{L_H}, E) \right)^{\oplus n}$$

where  $Z_{L_H}$  acts on W via  $Z_{L_H} \hookrightarrow Z_0$ , i.e. by only acting on the factor  $\mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z_{L_P}^0, E)$  in (3.12) by the right regular action, and similarly with  $\mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z_{L_H}, E)$  on the right hand side. The proposition follows then from (an easy variation of) the argument in the proof of [17, Prop. 5.3] (with  $\Pi^{N_0}$  of *loc. cit.* replaced by W).

## 3.2 Bernstein eigenvarieties

In this section, we first give our global setup. Then we apply the formalism of § 3.1.3 to construct rigid analytic spaces, called *Bernstein eigenvarieties*, parametrizing certain *p*-adic automorphic representations. We also show some basic properties of Bernstein eigenvarieties.

#### **3.2.1** *p*-adic automorphic representations

We briefly recall the global setting of  $[14, \S 5]$  (which will be the same as ours) and introduce some notation.

We fix embeddings  $\iota_{\infty}: \overline{\mathbb{Q}} \hookrightarrow \mathbb{C}$  and  $\iota_p: \overline{\mathbb{Q}} \hookrightarrow \overline{\mathbb{Q}_p}$ . We let F be a CM field that is a totally imaginary quadratic extension of a totally real field  $F^+$  such that all the places of  $F^+$  dividing p split in F. We let  $G/F^+$  be a unitary group of rank  $n \geq 2$  associated to  $F/F^+$ , i.e. such that  $G \times_{F^+} F \cong \operatorname{GL}_n / F$ . We assume that  $G(F_v^+)$  is compact at all archimedean places v of  $F^+$ . For a finite place v of  $F^+$  such that v splits in F, we choose a place  $\tilde{v}$  of F dividing v. For such places, we have natural isomorphisms  $F_v^+ \cong F_{\tilde{v}}$  and  $i_{\tilde{v}} : G(F_v^+) \xrightarrow{\sim} G(F_{\tilde{v}}) \xrightarrow{\sim} \operatorname{GL}_n(F_{\tilde{v}})$ .

We let  $U^p$  be a compact open subgroup of  $G(\mathbb{A}_{F^+}^{\infty,p})$  of the form  $U^p = \prod_{v \nmid p, \infty} U_v$  where  $U_v$ is a compact open subgroup of  $G(F_v^+)$  which is hyperspecial when v is inert in F. We choose a finite set S of finite places of  $F^+$  containing the set  $S_p$  of places dividing p and the set of places v such that either v is ramified in F or  $U_v$  is not maximal at v. We assume moreover that for all places  $v \notin S$  that are split in F,  $U_v = i_{\widetilde{v}}^{-1}(\operatorname{GL}_n(\mathcal{O}_{F_{\widetilde{v}}}))$ . We let  $\mathbb{T}^S := \varinjlim_I \otimes_{v \in I} \mathbb{T}_v$  where  $\mathbb{T}_v := \mathcal{O}_E[U_v \setminus G(F_v^+)/U_v]$  and I runs through the finite sets of places v of  $F^+$  which are not in S and split in F (recall that E is a sufficiently large finite extension of  $\mathbb{Q}_p$ ). Note that  $\mathbb{T}_v$  is polynomially generated over  $\mathcal{O}_E$  by the operators  $T_{\widetilde{v},i} = \begin{bmatrix} U_v i_{\widetilde{v}}^{-1} \begin{pmatrix} \varpi_{\widetilde{v}} 1_i & 0\\ 0 & 1_{n-i} \end{pmatrix} U_v \end{bmatrix}$  for  $1 \leq i \leq n$ .

We consider the usual spaces of p-adic automorphic forms of level  $U^p$  in that context:

$$\widehat{S}(U^p, E) := \{ f : G(F^+) \setminus G(\mathbb{A}_{F^+}^{\infty}) / U^p \to E, f \text{ continuous} \}, 
\widehat{S}(U^p, \mathcal{O}_E) := \{ f : G(F^+) \setminus G(\mathbb{A}_{F^+}^{\infty}) / U^p \to \mathcal{O}_E, f \text{ continuous} \}.$$

We equip  $\widehat{S}(U^p, E)$  with the norm defined using the unit ball  $\widehat{S}(U^p, \mathcal{O}_E)$ , in particular  $\widehat{S}(U^p, E)$ is a *p*-adic Banach space. This Banach space is also equipped with a natural continuous unitary action of  $G(F^+ \otimes_{\mathbb{Q}} \mathbb{Q}_p) \cong \prod_{v \in S_p} \operatorname{GL}_n(F_{\widetilde{v}})$ , and an action of  $\mathbb{T}^S$  (with each element acting via a continuous operator) that commutes with  $G(F^+ \otimes_{\mathbb{Q}} \mathbb{Q}_p)$ . Note that all these actions preserve  $\widehat{S}(U^p, \mathcal{O}_E)$ . We also have

$$\widehat{S}(U^p, \mathcal{O}_E) \cong \varprojlim_s S(U^p, \mathcal{O}_E/\varpi_E^s) \cong \varprojlim_s \varinjlim_{U_p} S(U^p U_p, \mathcal{O}_E/\varpi_E^s)$$

where  $S(U^p U_p, \mathcal{O}_E / \varpi_E^s)$  denotes the space of functions  $G(F^+) \setminus G(\mathbb{A}_{F^+}^\infty) / (U^p U_p) \to \mathcal{O}_E / \varpi_E^s$ .

For an automorphic representation  $\pi \cong \pi_{\infty} \otimes_{\mathbb{C}} \pi^{\infty} \cong \pi_{\infty} \otimes_{\mathbb{C}} \pi^{\infty,p} \otimes_{\mathbb{C}} \pi_p$  of

$$G(\mathbb{A}_{F^+}) \cong G(F^+ \otimes_{\mathbb{Q}} \mathbb{R}) \times G(\mathbb{A}_{F^+}^{\infty}) \cong G(F^+ \otimes_{\mathbb{Q}} \mathbb{R}) \times G(\mathbb{A}_{F^+}^{\infty,p}) \times G(F^+ \otimes_{\mathbb{Q}} \mathbb{Q}_p),$$

we associate an algebraic representation  $W_{\pi,p}$  of  $G(F^+ \otimes_{\mathbb{Q}} \mathbb{Q}_p)$  over  $\overline{\mathbb{Q}_p}$  as in [14, Prop. 5.1] (and as in the references therein). Recall that we have (for example see [14, Prop. 5.1] for (1) and part (a) in the proof of [16, Lemma 6.1] for (2)):

**Proposition 3.2.1.** (1) We have a  $G(F^+ \otimes_{\mathbb{Q}} \mathbb{Q}_p) \times \mathbb{T}^S$ -equivariant isomorphism

$$\widehat{S}(U^p, E)^{\text{lalg}} \otimes_E \overline{\mathbb{Q}_p} \cong \bigoplus_{\pi} \left( (\pi^{\infty, p})^{U^p} \otimes_{\overline{\mathbb{Q}}} (\pi_p \otimes_{\overline{\mathbb{Q}}} W_{\pi, p}) \right)^{\oplus m(\pi)},$$

where  $\pi$  runs through the automorphic representations of  $G(\mathbb{A}_{F^+})$ .

(2) Assume  $U^p$  is sufficiently small, then for any compact open subgroup H of  $G(F^+ \otimes_{\mathbb{Q}} \mathbb{Q}_p)$ , there exists m such that  $\widehat{S}(U^p, E)|_H \cong \mathcal{C}(H, E)^{\oplus m}$ .

## 3.2.2 Bernstein eigenvarieties: construction and first properties

We apply the construction in § 3.1 to the locally  $\mathbb{Q}_p$ -analytic representation  $\widehat{S}(U^p, E)^{\mathrm{an}}$  of  $G(F^+ \otimes_{\mathbb{Q}} \mathbb{Q}_p) \cong \prod_{v \in S_p} \mathrm{GL}_n(F_{\widetilde{v}}).$ 

We first modify the notation in § 3.1 to be consistent with § 3.2.1 in an obvious way. The index set  $\mathscr{I}$  will be  $S_p$ , and  $G_p$  will be  $G(F^+ \otimes_{\mathbb{Q}} \mathbb{Q}_p) \cong \prod_{v \in S_p} \operatorname{GL}_n(F_{\widetilde{v}})$ . The element  $i \in \mathscr{I}$  will be replaced by  $\widetilde{v}$  everywhere (for example,  $F_i$  in § 3.1 will be  $F_{\widetilde{v}}$  etc.) and we fix a uniformizer  $\varpi_{\widetilde{v}}$  for each  $F_{\widetilde{v}}$ . As in 3.1, we fix a parabolic subgroup  $P_{\widetilde{v}} \supset B$  of  $\operatorname{GL}_n$  for each  $\widetilde{v}$  with a fixed Levi subgroup  $L_{P_{\widetilde{v}}} \supset T$  and denote by  $N_{P_{\widetilde{v}}}$  its nilpotent radical. We fix a cuspidal Bernstein component  $\Omega$  for  $L_P(\mathbb{Q}_p) \cong \prod_{v \in S_p} L_{P_{\widetilde{v}}}(F_{\widetilde{v}}) \cong \prod_{v \in S_p} \prod_{j=1}^{r_{\widetilde{v}}} \operatorname{GL}_{n_{\widetilde{v},j}}(F_{\widetilde{v}})$ , and let  $\sigma \cong \boxtimes_{v \in S_p} \boxtimes_{j=1}^{r_{\widetilde{v}}} \sigma_{\widetilde{v},j}$  be a smooth absolutely irreducible representation of  $L_P^0 := \prod_{v \in S_p} \prod_{j=1}^{r_{\widetilde{v}}} \operatorname{GL}_{n_{\widetilde{v},j}}(\mathcal{O}_{F_{\widetilde{v}}})$  over E associated to  $\Omega$  as in § 3.1. We finally fix an integral P-dominant weight  $\lambda = (\lambda_{\widetilde{v},i,\tau}) \underset{i=1,\ldots,n}{v \in \Sigma_{\widetilde{v}}}$  for  $G_p$  where

$$\Sigma_{\widetilde{v}} := \Sigma_{F_{\widetilde{v}}}$$
. For  $i \in \{1, \ldots, r_{\widetilde{v}}\}$  we set  $s_{\widetilde{v},i} := \sum_{j=1}^{i} n_{\widetilde{v},j}$  and  $s_{\widetilde{v},0} = 0$ .

By the discussion above Proposition 3.1.8,  $B_{\Omega,\lambda}(\widehat{S}(U^p, E)^{\mathrm{an}})^{\vee}$  gives rise to a coherent sheaf  $\mathcal{M}_{\Omega,\lambda}(U^p)$  over  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\mathrm{rig}} \times \widehat{\mathcal{Z}}_0$  where  $\mathcal{Z}_0 \cong Z^0_{L_P}$ . By functoriality,  $B_{\Omega,\lambda}(\widehat{S}(U^p, E)^{\mathrm{an}})$  is naturally equipped with an action of  $\mathbb{T}^S$  that commutes with the action of  $\mathcal{Z}_\Omega \times \mathcal{Z}_0$ . We deduce that  $\mathcal{M}_{\Omega,\lambda}(U^p)$  is equipped with a natural  $\mathcal{O}_{(\operatorname{Spec} \mathcal{Z}_\Omega)^{\mathrm{rig}} \times \widehat{\mathcal{Z}}_0}$ -linear action of  $\mathbb{T}^S$ . For each affinoid open  $U = \operatorname{Spm} R$  of  $(\operatorname{Spec} \mathcal{Z}_\Omega)^{\mathrm{rig}} \times \widehat{\mathcal{Z}}_0$ , the (commutative) R-subalgebra  $A_R$  of  $\operatorname{End}_R(\mathcal{M}_{\Omega,\lambda}(U^p)|_U)$  generated by  $\mathbb{T}^S$  is a finite type R-module. These  $\{\operatorname{Spm} A_R\}$  then glue to a rigid analytic space, denoted by  $\mathcal{E}_{\Omega,\lambda}(U^p)$ , which is finite over  $(\operatorname{Spec} \mathcal{Z}_\Omega)^{\mathrm{rig}} \times \widehat{\mathcal{Z}}_0$ . From the definition of  $\mathcal{E}_{\Omega,\lambda}(U^p)$ , we see that  $\mathcal{M}_{\Omega,\lambda}(U^p)$  is also a coherent sheaf over  $\mathcal{E}_{\Omega,\lambda}(U^p)$ .

The following properties follow easily from the construction, Proposition 3.1.8 and Corollary 3.1.11.

**Proposition 3.2.2.** (1) For a finite extension E' of E, an E'-point of  $\mathcal{E}_{\Omega,\lambda}(U^p)$  can be identified to a triple  $(\eta, \pi_{L_P}, \chi)$  where  $\eta : \mathbb{T}^S \to E'$  is a system of Hecke eigenvalues and  $(\pi_{L_P}, \chi)$  is an E'-point of  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$ .

(2) We have an isomorphism equivariant under the action of  $\mathcal{Z}_{\Omega} \times \mathcal{Z}_{0} \times \mathbb{T}^{S}$ :

$$\Gamma(\mathcal{E}_{\Omega,\lambda}(U^p),\mathcal{M}_{\Omega,\lambda}(U^p)) \cong B_{\Omega,\lambda}(S(U^p,E)^{\mathrm{an}})^{\vee}.$$

Moreover, for  $x = (\eta, \pi_{L_P}, \chi) \in \mathcal{E}_{\Omega,\lambda}(U^p)$ , the above isomorphism induces an isomorphism of k(x)-vector spaces:

$$(x^*\mathcal{M}_{\Omega,\lambda}(U^p))^{\vee} \cong \operatorname{Hom}_{L_P(\mathbb{Q}_p)} \left( \pi_{L_P} \otimes_E \left( \chi_{\underline{\varpi}} \circ \det_{L_P} \right) \otimes_E L(\lambda)_P, J_P(\widehat{S}(U^p, E)^{\operatorname{an}})[\mathbb{T}^S = \eta] \right).$$
(3.13)

(3) For  $\eta : \mathbb{T}^S \to E'$  a system of Hecke eigenvalues and  $(\pi_{L_P}, \chi)$  an E'-point of  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_0$ , the following are equivalent:

- there exists a point  $x \in \mathcal{E}_{\Omega,\lambda}(U^p)$  of parameter  $(\eta, \pi_{L_P}, \chi)$ ;
- there exists a point  $x \in \mathcal{E}_{\Omega,\lambda}(U^p)$  of parameter  $(\eta, \psi(\pi_{L_P}, \chi))$  for any  $\psi \in \mu_{\Omega}$  (see the discussion above Corollary 3.1.11 for the action of  $\mu_{\Omega}$  on  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0})$ ;
- the vector space on the right hand side of (3.13) is non-zero.

By Proposition 3.1.9, we have

**Proposition 3.2.3.** With the notation of Proposition 3.1.9, we have an isomorphism of rigid spaces

$$\mathcal{E}_{\Omega,\lambda}(U^p) \xrightarrow{\sim} \mathcal{E}_{\Omega',\lambda'}(U^p), (\eta, \pi_{L_P}, \chi) \mapsto \Big(\eta, \pi_{L_P} \otimes_E \big( (\psi^0_{\underline{\varpi}}(\delta^{\mathrm{unr}}_{\underline{\vartheta},\underline{\varpi}})^{-1}) \circ \det_{L_P} \big), \chi(\psi^0_{\underline{\varpi}})^{-1}(\delta^0_{\underline{\vartheta},\underline{\varpi}})^{-1} \Big).$$

Let  $z \in Z_{\underline{\omega}} \subset Z_{L_P}(\mathbb{Q}_p)$  be as in the discussion above Proposition 3.1.15. We define  $\kappa_z$  as the composition

$$\kappa_z: \mathcal{E}_{\Omega,\lambda}(U^p) \longrightarrow (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_0 \longrightarrow (\operatorname{Spec} E[Z_{\underline{\varpi}}])^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_0 \longrightarrow \mathbb{G}_m^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_0$$
(3.14)

where the last two morphisms are induced by  $E[Y_z] \hookrightarrow E[Z_{\underline{\omega}}] \to \mathcal{Z}_{\Omega}$  (recall  $Y_z$  is the subgroup of  $Z_{\underline{\omega}}$  generated by z). It follows from [40, Prop. 3.2.23] and Proposition 3.1.15 (see also the proof of [17, Lemma 3.10]) that  $(\kappa_z)_* \mathcal{M}_{\Omega,\lambda}(U^p)$  is a coherent sheaf over  $\mathbb{G}_m^{\mathrm{rig}} \times \widehat{\mathcal{Z}}_0$  and  $\kappa_z$  is finite. We denote by  $Z_z(U^p)$  its scheme-theoretic support in  $\mathbb{G}_m^{\mathrm{rig}} \times \widehat{\mathcal{Z}}_0$ . Note that the first morphism in (3.14) factors through the scheme-theoretic support of  $\mathcal{M}_{\Omega,\lambda}(U^p)$  in  $(\mathrm{Spec} \, \mathcal{Z}_\Omega)^{\mathrm{rig}} \times \widehat{\mathcal{Z}}_0$ , and  $\kappa_z$ factors through  $Z_z(U^p)$ . We define  $\kappa$  as the composition (the second map being the canonical projection)

$$\kappa: \mathcal{E}_{\Omega,\lambda}(U^p) \to (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0} \twoheadrightarrow \widehat{\mathcal{Z}_0}$$

which obviously factors through  $\kappa_z$ . By exactly the same argument as in the proofs of [17, Lemma 3.10] and [17, Prop. 3.11] (with [17, Prop. 5.3] replaced by Proposition 3.1.15), we have:

**Proposition 3.2.4.** (1) The rigid space  $Z_z(U^p) \hookrightarrow \mathbb{G}_m^{\operatorname{rig}} \times \widehat{Z}_0$  is a Fredholm hypersurface of  $\mathbb{G}_m^{\operatorname{rig}} \times \widehat{Z}_0$  (cf. [17, § 3.3]). Moreover, there exists an admissible covering  $\{U'_i\}$  of  $Z_z(U^p)$  by affinoids  $U'_i$  such that the composition

$$g: Z_z(U^p) \hookrightarrow \mathbb{G}_m^{\mathrm{rig}} \times \widehat{\mathcal{Z}_0} \twoheadrightarrow \widehat{\mathcal{Z}_0}$$

induces a finite surjective morphism from  $U'_i$  to an affinoid open  $W_i$  of  $\widehat{\mathcal{Z}}_0$ , and such that  $U'_i$ is a connected component of  $g^{-1}(W_i)$ . For each i,  $\Gamma(U'_i, (\kappa_z)_*\mathcal{M}_{\Omega,\lambda}(U^p))$  is a finitely generated projective  $\mathcal{O}_{\widehat{\mathcal{Z}}_0}(W_i)$ -module.

(2) There exists an admissible covering  $\{U_i\}$  of  $\mathcal{E}_{\Omega,\lambda}(U^p)$  by affinoids  $U_i$  such that

- there exists an affinoid open  $W_i$  of  $\widehat{\mathcal{Z}}_0$  satisfying that  $\kappa$  is a finite surjective morphism from each irreducible component of  $U_i$  to  $W_i$ ;
- $\mathcal{O}_{\mathcal{E}_{\Omega,\lambda}(U^p)}(U_i)$  is isomorphic to an  $\mathcal{O}_{\widehat{\mathcal{Z}}_0}(W_i)$ -algebra of endomorphisms of a finitely generated projective  $\mathcal{O}_{\widehat{\mathcal{Z}}_0}(W_i)$ -module.

We also have as in [17, Cor. 3.12], [17, Cor. 3.13] and [18, Lemma 3.8] by the same arguments:

**Corollary 3.2.5.** (1) The rigid space  $\mathcal{E}_{\Omega,\lambda}(U^p)$  is nested ([3, Def. 7.2.10]), equidimensional of dimension  $\sum_{v \in S_p} ([F_{\widetilde{v}} : \mathbb{Q}_p]r_{\widetilde{v}})$ , and has no embedded component.

(2) The morphism  $\kappa_z$  is finite and the image of an irreducible component of  $\mathcal{E}_{\Omega,\lambda}(U^p)$  is an irreducible component of  $Z_z(U^p)$ . The image of an irreducible component of  $\mathcal{E}_{\Omega,\lambda}(U^p)$  by  $\kappa$  is a Zariski-open of  $\widehat{\mathcal{Z}}_0$ .

(3) The coherent sheaf  $\mathcal{M}_{\Omega,\lambda}(U^p)$  is Cohen-Macaulay over  $\mathcal{E}_{\Omega,\lambda}(U^p)$ .

**Remark 3.2.6.** Assume  $P = B_p := \prod_{v \in S_p} B(F_{\widetilde{v}})$ , and let  $T_p := \prod_{v \in S_p} T(F_{\widetilde{v}})$ . Consider the composition (cf. Remark 3.1.10)

$$\mathcal{E}_{\Omega,\lambda}(U^p) \longrightarrow (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_0 \xrightarrow{\iota_{\Omega,\lambda}} \widehat{T}_p.$$

Using Proposition 3.2.3, the Bernstein eigenvarieties  $\mathcal{E}_{\Omega,\lambda}(U^p)$  equipped with the above morphism over  $\widehat{T_p}$  are all isomorphic for different  $(\Omega, \lambda)$ . Moreover by Remark 3.1.10 and (the same argument as in the proof of) [3, Prop. 7.2.8], one can show that  $\mathcal{E}_{\Omega,\lambda}(U^p)^{\text{red}}$  is isomorphic to the standard reduced eigenvariety  $\mathcal{E}(U^p)^{\text{red}}$  constructed directly from  $J_{B_p}(\widehat{S}(U^p, E)^{\text{an}})$  (see for example [14, § 7]).

## 3.2.3 Density of classical points

We prove Theorem 3.2.11 below.

**Definition 3.2.7.** Let  $x = (\eta_x, \pi_{x,L_P}, \chi_x)$  be a point in  $\mathcal{E}_{\Omega,\lambda}(U^p)$ .

(1) We call x classical if

$$\operatorname{Hom}_{L_P(\mathbb{Q}_p)}\left(\pi_{x,L_P}\otimes_E \left((\chi_x)_{\underline{\varpi}}\circ \det_{L_P}\right)\otimes_E L(\lambda)_P, J_P(\widehat{S}(U^p,E)^{\operatorname{lalg}})[\mathbb{T}^S=\eta_x]\right)\neq 0.$$
(3.15)

- (2) We call x very classical if
- $\chi_x$  is locally algebraic and the weight  $\lambda^x := (\operatorname{wt}(\chi_x) \circ \operatorname{det}_{L_P}) + \lambda$  is dominant;
- any irreducible constituent of the locally analytic parabolic induction

$$\left(\operatorname{Ind}_{P^{-}(\mathbb{Q}_{p})}^{G(\mathbb{Q}_{p})}\left(\pi_{x,L_{P}}\otimes_{E}\left((\chi_{x})_{\underline{\varpi}}\circ\det_{L_{P}}\right)\otimes_{E}L(\lambda)_{P}\right)\right)^{\operatorname{an}}$$
(3.16)

which is not locally algebraic, does not admit a  $G(\mathbb{Q}_p)$ -invariant  $\mathcal{O}_E$ -lattice.

**Lemma 3.2.8.** A very classical point  $x = (\eta_x, \pi_{x,L_P}, \chi_x)$  is classical.

*Proof.* We write  $(\chi_x)_{\underline{\varpi}}$  in the form  $(\chi_x)_{\underline{\varpi}}^{\infty} \delta_{\mathrm{wt}(\chi_x)}$  where  $(\chi_x)_{\underline{\varpi}}^{\infty}$  is a smooth character of  $Z_{L_P}(\mathbb{Q}_p)$  (recall from § 3.1.1 that  $\delta_{\mathrm{wt}(\chi_x)}$  is the algebraic character of  $Z_{L_P}(\mathbb{Q}_p)$  of weight  $\mathrm{wt}(\chi_x)$ ). By Proposition 3.2.2 (3) we have

$$\operatorname{Hom}_{L_P(\mathbb{Q}_p)}\left(\pi_{x,L_P}\otimes_E (\chi_x)_{\underline{\varpi}} \circ \det_{L_P} \otimes_E L(\lambda)_P, J_P(\widehat{S}(U^p,E)^{\operatorname{an}})[\mathbb{T}^S = \eta_x]\right) \neq 0.$$

By [14, Thm. 4.3] (the notation of which we freely use), any non-zero element in the above vector space induces a non-zero morphism of  $G(\mathbb{Q}_p)$ -representations (recall  $\mathfrak{g}, \mathfrak{p}^-$  denote the Lie algebra of  $G(\mathbb{Q}_p), P^-(\mathbb{Q}_p)$  respectively, and  $\delta_P$  denotes the modulus character of  $P(\mathbb{Q}_p)$ )

$$\mathcal{F}_{P^{-}}^{G}\left(\left(\mathcal{U}(\mathfrak{g})\otimes_{\mathcal{U}(\mathfrak{p}^{-})}L^{-}(-\lambda^{x})_{P}\right)^{\vee},\pi_{x,L_{P}}\otimes_{E}\delta_{P}^{-1}\otimes_{E}\left((\chi_{x})_{\underline{\varpi}}^{\infty}\circ\det_{L_{P}}\right)\right)\\ \longrightarrow \widehat{S}(U^{p},E)^{\mathrm{an}}[\mathbb{T}^{S}=\eta_{x}]\hookrightarrow\widehat{S}(U^{p},E)^{\mathrm{an}}.$$

By the results of [74], the representation on the left hand side has the following properties:

• it has the same irreducible constituents as the representation (3.16);

• there is a  $G(\mathbb{Q}_p)$ -equivariant surjection (where  $(\operatorname{Ind}_{P^-(\mathbb{Q}_p)}^{G(\mathbb{Q}_p)}(-))^{\infty}$  is the smooth parabolic induction)

$$\mathcal{F}_{P^{-}}^{G}\left(\left(\mathrm{U}(\mathfrak{g})\otimes_{\mathrm{U}(\mathfrak{p}^{-})}L^{-}(-\lambda^{x})_{P}\right)^{\vee},\pi_{x,L_{P}}\otimes_{E}\delta_{P}^{-1}\otimes_{E}\left((\chi_{x})_{\underline{\varpi}}^{\infty}\circ\det_{L_{P}}\right)\right)$$
$$\longrightarrow\mathcal{F}_{P^{-}}^{G_{p}}\left(L^{-}(-\lambda^{x}),\pi_{x,L_{P}}\otimes_{E}\delta_{P}^{-1}\otimes_{E}\left((\chi_{x})_{\underline{\varpi}}^{\infty}\circ\det_{L_{P}}\right)\right)$$
$$\cong\left(\operatorname{Ind}_{P^{-}(\mathbb{Q}_{P})}^{G(\mathbb{Q}_{P})}\left(\pi_{x,L_{P}}\otimes_{E}\delta_{P}^{-1}\otimes_{E}\left((\chi_{x})_{\underline{\varpi}}^{\infty}\circ\det_{L_{P}}\right)\right)\right)^{\infty}\otimes_{E}L(\lambda^{x})$$

and any irreducible constituent of the kernel is *not* locally algebraic.

The lemma then follows easily by definition.

We have the following numerical classicality criterion.

Proposition 3.2.9 (Numerical classicality). Let

$$x = (\eta_x, \pi_{x, L_P}, \chi_x) = \left(\eta_x, \otimes_{v \in S_P} \pi_{x, \widetilde{v}}, \otimes_{v \in S_P} \chi_{x, \widetilde{v}}\right) \in \mathcal{E}_{\Omega, \lambda}(U^p)$$

such that  $\chi_x$  is locally algebraic and  $\lambda^x = (\operatorname{wt}(\chi_x) \circ \operatorname{det}_{L_P}) + \lambda$  is dominant. Assume that for all  $v \in S_p$  and  $k \in \{1, \ldots, r_{\widetilde{v}}\}$ , we have

$$\sum_{j=1}^{k} \operatorname{val}_{\widetilde{v}}(\omega_{\pi_{x,\widetilde{v},j}}(\varpi_{\widetilde{v}})) < \sum_{j=1}^{k} (s_{\widetilde{v},j} + s_{\widetilde{v},j-1} - s_{\widetilde{v},k}) - \sum_{\tau \in \Sigma_{\widetilde{v}}} \sum_{j=1}^{s_{\widetilde{v},k}} \lambda_{\widetilde{v},j,\tau} + \inf_{\tau \in \Sigma_{\widetilde{v}}} \left\{ \operatorname{wt}(\chi_{x})_{\widetilde{v},k,\tau} - \operatorname{wt}(\chi_{x})_{\widetilde{v},k+1,\tau} + \lambda_{\widetilde{v},s_{\widetilde{v},k},\tau} - \lambda_{\widetilde{v},s_{\widetilde{v},k+1},\tau} + 1 \right\}, \quad (3.17)$$

where  $\operatorname{val}_{\widetilde{v}}$  denotes the p-adic valuation normalized by sending  $\varpi_{\widetilde{v}}$  to 1,  $\operatorname{wt}(\chi_x)_{\widetilde{v},k',\tau'} := \operatorname{wt}(\chi_{x,\widetilde{v}})_{k',\tau'}$ , and where  $\omega_{\pi_{x,\widetilde{v},j}}$  denotes the central character of  $\pi_{x,\widetilde{v},j}$ . Then the point x is very classical.

*Proof.* By [74, Thm. (i),(ii)], the representation  $\left(\operatorname{Ind}_{P^{-}(\mathbb{Q}_{p})}^{G(\mathbb{Q}_{p})}(\pi_{x,L_{P}}\otimes_{E}((\chi_{x})_{\underline{\varpi}}\circ\det_{L_{P}})\otimes_{E}L(\lambda)_{P})\right)^{\operatorname{an}}$  admits a Jordan-Holder filtration with graded pieces of the form:

$$\mathcal{F}_{P^{-}}^{G}(L^{-}(-w\cdot\lambda^{x}),\pi_{x,L_{P}}\otimes_{E}((\chi_{x})_{\underline{\varpi}}^{\infty}\circ\det_{L_{P}})\otimes_{E}\delta_{P}^{-1})$$

where  $L^{-}(-w \cdot \lambda^{x})$  runs through irreducible constituents of  $U(\mathfrak{g}) \otimes_{U(\mathfrak{p}^{-})} L^{-}(-\lambda^{x})_{P}$ . In particular, each  $w \cdot \lambda^{x}$  is *P*-dominant. Assume that there exists an irreducible constituent *V* of  $\left(\operatorname{Ind}_{P^{-}(\mathbb{Q}_{p})}^{G(\mathbb{Q}_{p})} \pi_{x,L_{P}} \otimes_{E} ((\chi_{x})_{\underline{\varpi}} \circ \det_{L_{P}}) \otimes_{E} L(\lambda)_{P}\right)^{\operatorname{an}}$  such that *V* admits a  $G(\mathbb{Q}_{p})$ -invariant lattice and *V* is not locally algebraic. We deduce that there exists

$$1 \neq w = (w_{\widetilde{v}})_{v \in S_p} = (w_{\widetilde{v},\tau})_{v \in S_p} \in \mathscr{W}^{|\Sigma_p|} \cong \prod_{v \in S_p} \mathscr{W}_{F_{\widetilde{v}}} =: \mathscr{W}_F$$

such that  $w \cdot \lambda^x$  is *P*-dominant and *V* is a constituent of  $\mathcal{F}_{P^-}^{G_p}(L^-(-w \cdot \lambda^x), \pi_{x,L_P} \otimes_E ((\chi_x)_{\underline{\omega}}^{\infty} \circ \det_{L_P}) \otimes_E \delta_P^{-1})$ . We let  $Q_{\widetilde{v}} \supset P_{\widetilde{v}}$  be the maximal parabolic of  $\operatorname{GL}_n$  such that  $w \cdot \lambda$  is  $\prod_{v \in S_p} \operatorname{Res}_{\mathbb{Q}_p}^{F_{\widetilde{v}}} Q_{\widetilde{v}}^{-1}$  dominant, i.e.  $(w \cdot \lambda^x)_{\widetilde{v},\tau}$  is  $Q_{\widetilde{v}}$ -dominant for all  $v \in S_p$  and  $\tau \in \Sigma_{\widetilde{v}}$ . We set  $Q := \prod_{v \in S_p} \operatorname{Res}_{\mathbb{Q}_p}^{F_{\widetilde{v}}} Q_{\widetilde{v}}$ . We have by [74, Thm. (iii)]

$$\mathcal{F}_{P^{-}}^{G}\left(L^{-}(-w\cdot\lambda^{x}),\pi_{x,L_{P}}\otimes_{E}\left((\chi_{x})_{\underline{\varpi}}^{\infty}\circ\det_{L_{P}}\right)\otimes_{E}\delta_{P}^{-1}\right)$$
$$\cong\mathcal{F}_{Q^{-}}^{G}\left(L^{-}(-w\cdot\lambda^{x}),\left(\operatorname{Ind}_{L_{Q}(\mathbb{Q}_{p})\cap P^{-}(\mathbb{Q}_{p})}^{L_{Q}(\mathbb{Q}_{p})}(\pi_{x,L_{P}}\otimes_{E}\left((\chi_{x})_{\underline{\varpi}}^{\infty}\circ\det_{L_{P}}\right)\otimes_{E}\delta_{P}^{-1})\right)^{\infty}\right).$$
By [74, Thm. (i),(ii),(iv)], there exists an irreducible constituent  $\pi_{L_Q}$  of the smooth representation  $\left(\operatorname{Ind}_{L_Q(\mathbb{Q}_p)\cap P^-(\mathbb{Q}_p)}^{L_Q(\mathbb{Q}_p)}(\pi_{x,L_P}\otimes_E((\chi_x)_{\underline{\varpi}}^{\infty}\circ\det_{L_P})\otimes_E\delta_P^{-1})\right)^{\infty}$  such that

$$V \cong \mathcal{F}_{Q^{-}}^{G} \left( L^{-}(-w \cdot \lambda^{x}), \pi_{L_{Q}} \right)$$
  
$$\longleftrightarrow \mathcal{F}_{Q^{-}}^{G} \left( L^{-}(-w \cdot \lambda^{x}), \left( \operatorname{Ind}_{L_{Q}(\mathbb{Q}_{p}) \cap P^{-}(\mathbb{Q}_{p})}^{L_{Q}(\mathbb{Q}_{p})}(\pi_{x,L_{P}} \otimes_{E} ((\chi_{x})_{\underline{\varpi}}^{\infty} \circ \det_{L_{P}}) \otimes_{E} \delta_{P}^{-1}) \right)^{\infty} \right).$$

Since V admits a  $G(\mathbb{Q}_p)$ -invariant  $\mathcal{O}_E$ -lattice, we see by [13, Cor. 3.5] that for any  $z \in Z_{L_Q}(\mathbb{Q}_p)^+ \subset Z_{L_P}(\mathbb{Q}_p)^+$  we have (where  $\omega_{\pi_{L_Q}}$  denotes the central character of  $\pi_{L_Q}$ )

$$\operatorname{val}_{p}\left(\delta_{w \cdot \lambda^{x}}(z)\delta_{P}^{-1}(z)\omega_{\pi_{L_{Q}}}(z)\right) \geq 0.$$
(3.18)

By [35, Lemma 3.18], we easily deduce (noting that  $\beta_a$  in *loc. cit.* is independent of the choice of  $\lambda$ ) that for each  $v \in S_p$  such that  $w_{\widetilde{v}} \neq 1$  there exists  $k \in \{1, \ldots, r_{\widetilde{v}}\}$  such that  $z := (\underbrace{\varpi_{\widetilde{v}}, \ldots, \varpi_{\widetilde{v}}}_{s_{\widetilde{v},k}}, 1, \ldots, 1) \in Z_{L_{Q_{\widetilde{v}}}}(F_{\widetilde{v}})^+ \subset Z_{L_Q}(\mathbb{Q}_p)^+$  and

$$\begin{aligned} \operatorname{val}_{\widetilde{v}} \left( \delta_{w \cdot \lambda^{x} - \lambda^{x}}(z) \right) &\leq \sum_{\substack{\tau \in \Sigma_{\widetilde{v}} \\ w_{\widetilde{v}, \tau} \neq 1}} \left( \lambda_{\widetilde{v}, s_{\widetilde{v}, k+1}, \tau}^{x} - \lambda_{\widetilde{v}, s_{\widetilde{v}, k}, \tau}^{x} - 1 \right) \\ &= \sum_{\substack{\tau \in \Sigma_{\widetilde{v}} \\ w_{\widetilde{v}, \tau} \neq 1}} \left( \operatorname{wt}(\chi_{x})_{\widetilde{v}, k+1, \tau} - \operatorname{wt}(\chi_{x})_{\widetilde{v}, k, \tau} + \lambda_{\widetilde{v}, s_{\widetilde{v}, k+1}, \tau} - \lambda_{\widetilde{v}, s_{\widetilde{v}, k}, \tau} - 1 \right). \end{aligned}$$

Together with (3.18), we deduce

$$\operatorname{val}_{\widetilde{v}}\left(\delta_{\lambda^{x}}(z)\delta_{P}^{-1}(z)\omega_{\pi_{L_{Q}}}(z)\right) \geq \sum_{\substack{\tau \in \Sigma_{\widetilde{v}}\\ w_{\widetilde{v},\tau} \neq 1}} \left(\operatorname{wt}(\chi_{x})_{\widetilde{v},k,\tau} - \operatorname{wt}(\chi_{x})_{\widetilde{v},k+1,\tau} + \lambda_{\widetilde{v},s_{\widetilde{v},k},\tau} - \lambda_{\widetilde{v},s_{\widetilde{v},k+1},\tau} + 1\right).$$

We compute:

$$\operatorname{val}_{\widetilde{v}}\left(\delta_{P}^{-1}(z)\right) = \sum_{j=1}^{k} \left(s_{\widetilde{v},k} - s_{\widetilde{v},j} - s_{\widetilde{v},j-1}\right)$$
$$\operatorname{val}_{\widetilde{v}}\left(\delta_{\lambda^{x}}(z)\right) = \sum_{\tau \in \Sigma_{\widetilde{v}}} \sum_{j=1}^{s_{\widetilde{v},k}} \lambda_{\widetilde{v},j,\tau}^{x} = \sum_{\tau \in \Sigma_{\widetilde{v}}} \sum_{j=1}^{s_{\widetilde{v},k}} \lambda_{\widetilde{v},j,\tau} + \sum_{\tau \in \Sigma_{\widetilde{v}}} \sum_{j=1}^{k} \left(n_{\widetilde{v},j} \operatorname{wt}(\chi_{x})_{\widetilde{v},j,\tau}\right)$$
$$\operatorname{val}_{\widetilde{v}}\left(\omega_{\pi_{L_{Q}}}(z)\right) = \sum_{j=1}^{k} \operatorname{val}_{\widetilde{v}}\left(\omega_{\pi_{\widetilde{v},j}}(\varpi_{\widetilde{v}})\right) - \sum_{\tau \in \Sigma_{\widetilde{v}}} \sum_{j=1}^{k} \left(n_{\widetilde{v},j} \operatorname{wt}(\chi_{x})_{\widetilde{v},j,\tau}\right).$$

Hence we deduce

$$\sum_{j=1}^{k} \operatorname{val}_{\widetilde{v}} \left( \omega_{\pi_{\widetilde{v},j}}(\varpi_{\widetilde{v}}) \right) \geq \sum_{j=1}^{k} \left( s_{\widetilde{v},j} + s_{\widetilde{v},j-1} - s_{\widetilde{v},k} \right) - \sum_{\tau \in \Sigma_{\widetilde{v}}} \sum_{j=1}^{s_{\widetilde{v},k}} \lambda_{\widetilde{v},j,\tau} + \lambda_{\widetilde{v},s_{\widetilde{v},k},\tau} - \lambda_{\widetilde{v},s_{\widetilde{v},k+1},\tau} + 1 \right),$$
$$+ \sum_{\substack{\tau \in \Sigma_{\widetilde{v}} \\ w_{\widetilde{v},\tau} \neq 1}} \left( \operatorname{wt}(\chi_{x})_{\widetilde{v},k,\tau} - \operatorname{wt}(\chi_{x})_{\widetilde{v},k+1,\tau} + \lambda_{\widetilde{v},s_{\widetilde{v},k},\tau} - \lambda_{\widetilde{v},s_{\widetilde{v},k+1},\tau} + 1 \right),$$

which contradicts (3.17) noting that  $\operatorname{wt}(\chi_x)_{\tilde{v},k,\tau} - \operatorname{wt}(\chi_x)_{\tilde{v},k+1,\tau} + \lambda_{\tilde{v},s_{\tilde{v},k},\tau} - \lambda_{\tilde{v},s_{\tilde{v},k+1},\tau} + 1 \ge 1$  for all  $\tau$  as  $\lambda^x$  is dominant and that the set  $\{\tau \in \Sigma_{\tilde{v}} \mid w_{\tilde{v},\tau} \neq 1\}$  is non-empty since  $w_{\tilde{v}} \neq 1$ . The proposition follows.

**Remark 3.2.10.** (1) Similar results (but in the setting of overconvergent cohomology) were obtained in [2].

(2) Note that all the terms on the right hand side of (3.17) except  $\operatorname{wt}(\chi_x)_{\tilde{v},k,\tau} - \operatorname{wt}(\chi_x)_{\tilde{v},k+1,\tau}$ are constants for points in  $\mathcal{E}_{\Omega,\lambda}(U^p)$ .

(3) Recall there is a finite morphism of E-algebras  $E[Z_{\underline{\omega}}] \to \mathcal{Z}_{\Omega}$  such that the associated morphism Spec  $\mathcal{Z}_{\Omega} \to \text{Spec } E[Z_{\underline{\omega}}]$  sends a point  $\pi_{x,L_P} \in \text{Spec } \mathcal{Z}_{\Omega}$  to the character  $z \mapsto \omega_{\pi_{x,L_P}}(z)$ . We have hence finite morphisms (see also (3.14)):

$$\mathcal{E}_{\Omega,\lambda}(U^p) \longrightarrow (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_0 \longrightarrow (\operatorname{Spec} E[Z_{\underline{\varpi}}])^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_0.$$

Note that the criterion in (3.17) only uses the information of the image of x in  $(\operatorname{Spec} E[Z_{\overline{\omega}}])^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0}$ .

The following theorem follows from the classicality criterion (Proposition 3.2.9) by the same argument as in the proof of [17, Thm. 3.19] (see also [28, § 6.4.5]).

**Theorem 3.2.11.** The set of very classical points is Zariski-dense in  $\mathcal{E}_{\Omega,\lambda}(U^p)$ . Moreover, for any point  $x = (\eta, \pi_{L_P}, \chi) \in \mathcal{E}_{\Omega,\lambda}(U^p)$  with  $\chi$  locally algebraic, and for any admissible open  $U \subseteq$  $\mathcal{E}_{\Omega,\lambda}(U^p)$  containing x, there exists an admissible open  $V \subseteq U$  containing x such that the set of very classical points in V is Zariski-dense in V.

#### **3.2.4** Galois representations

We study families of Galois representations on  $\mathcal{E}_{\Omega,\lambda}(U^p)$ . In particular, we show that the associated  $(\varphi, \Gamma)$ -modules admit a special kind of filtration.

We now assume that that G is quasi-split at all finite places and that  $F/F^+$  is unramified at all finite places. By [48, Thm. 2.3], to an automorphic representation  $\pi$  of G, one can associate an *n*-dimensional continuous essentially self-dual representation  $\rho_{\pi}$  of Gal<sub>F</sub> over E (enlarging E if necessary). In fact, if  $\pi^{U^p} \neq 0$ , the  $\mathbb{T}^S$ -action on  $\pi^{U^p}$  is given by a system of eigenvalues  $\eta_{\pi} : \mathbb{T}^S \to E$ . The representation  $\rho_{\pi}$  is unramified outside S. And for any  $v \notin S$  that splits in F, the characteristic polynomial of  $\rho_{\pi}(\operatorname{Frob}_{\widetilde{v}})$  (where  $\operatorname{Frob}_{\widetilde{v}}$  is a geometric Frobenius at  $\widetilde{v}$ ), is given by

$$X^{n} + \dots + (-1)^{j} (N\widetilde{v})^{\frac{j(j-1)}{2}} \eta_{\pi}(T_{\widetilde{v}}^{(j)}) X^{n-j} + \dots + (-1)^{n} (N\widetilde{v})^{\frac{n(n-1)}{2}} \eta_{\pi}(T_{\widetilde{v}}^{(n)}),$$
(3.19)

where  $N\tilde{v}$  is the cardinality of the residue field at  $\tilde{v}$ . Using Proposition 3.2.1, to all classical points  $x = (\eta_x, \pi_{x,L_P}, \chi_x) \in \mathcal{E}_{\Omega,\lambda}(U^p)$  (Definition 3.2.7 (1)), we can associate a continuous representation  $\rho_x$  of Gal<sub>F</sub> that is unramified outside S and satisfies (3.19) (with  $\eta_\pi$  replaced by  $\eta_x$ ). Denote by Gal<sup>S</sup><sub>F</sub> the Galois group of the maximal extension of F that is unramified outside S. Put

$$\mathcal{O}(\mathcal{E}_{\Omega,\lambda}(U^p))^0 := \{ f \in \mathcal{O}(\mathcal{E}_{\Omega,\lambda})(U^p) \mid |f(x)|_p \le 1 \text{ for all } x \in \mathcal{E}_{\Omega,\lambda}(U^p) \}.$$

Using that  $\mathbb{T}^S$  preserves  $\widehat{S}(U^p, \mathcal{O}_E)$ , it is easy to see that the natural morphism  $\mathbb{T}^S \to \mathcal{O}(\mathcal{E}_{\Omega,\lambda}(U^p))$ has image in  $\mathcal{O}(\mathcal{E}_{\Omega,\lambda}(U^p))^0$ . Using the density of classical points (Theorem 3.2.11) and [28, Prop. 7.1.1], we deduce

**Proposition 3.2.12.** There exists a unique n-dimensional continuous pseudocharacter

$$\mathscr{T}: \operatorname{Gal}_F^S \longrightarrow \mathcal{O}(\mathcal{E}_{\Omega,\lambda}(U^p))^0$$

such that the evaluation of  $\mathscr{T}$  at any classical point x of  $\mathcal{E}_{\Omega,\lambda}(U^p)$  coincides with  $Trace(\rho_x)$ .

By [83, Thm. 1(2)], we deduce

**Corollary 3.2.13.** For each point  $x = (\eta_x, \pi_{x,L_P}, \chi_x) \in \mathcal{E}_{\Omega,\lambda}(U^p)$ , there exists a (unique) semisimple continuous representation  $\rho_x$  of  $\operatorname{Gal}_F$  over k(x) which is unramified outside S and such that  $\operatorname{Trace}(\rho_x(\operatorname{Frob}_{\widetilde{v}})) = \eta_x(T_{\widetilde{v},1})$  for any  $v \notin S$  split in F.

Next, we study the behaviour of the restriction of the Galois representations  $\{\rho_x\}_{x\in\mathcal{E}_{\Omega,\lambda}(U^p)}$  at *p*-adic places.

Let  $x = (\eta_x, \pi_{x,L_P}, \chi_x) \in \mathcal{E}_{\Omega,\lambda}(U^p)$ . Recall  $\pi_{x,L_P} = \boxtimes_{v \in S_p} \pi_{x,L_{P_{\widetilde{v}}}} = \boxtimes_{v \in S_p} (\boxtimes_{i=1}^{r_{\widetilde{v}}} \pi_{x,\widetilde{v},i})$ . Let  $\mathbf{r}_{x,\widetilde{v},i} := \operatorname{rec}(\pi_{x,\widetilde{v},i})(\frac{1-n_{\widetilde{v},i}}{2}-s_{\widetilde{v},i-1})$ , an irreducible Weil-Deligne representation of  $W_{F_{\widetilde{v}}}$  of dimension  $n_{\widetilde{v},i}$ .

Assume first that the point x is classical. By Definition 3.15 and [17, Thm. 4.3], there exists a non-zero  $G_p$ -equivariant morphism (recalling  $\lambda^x = (\operatorname{wt}(\chi_x) \circ \operatorname{det}_{L_p}) + \lambda)$ 

$$\left(\operatorname{Ind}_{P^{-}(\mathbb{Q}_{p})}^{G(\mathbb{Q}_{p})}\pi_{x,L_{P}}\otimes_{k(x)}\left((\chi_{x})_{\underline{\varpi}}^{\infty}\circ\det_{L_{P}}\right)\otimes_{E}\delta_{P}^{-1}\right)^{\infty}\otimes_{E}L(\lambda^{x})\longrightarrow\widehat{S}(U^{p},E)^{\operatorname{lalg}}[\mathbb{T}^{S}=\eta_{x}] \quad (3.20)$$

where  $(\chi_x)_{\underline{\varpi}}^{\infty} = \boxtimes_{v \in S_p} (\chi_{x,\widetilde{v}})_{\overline{\varpi}_{\widetilde{v}}}^{\infty} = \boxtimes_{v \in S_p} \boxtimes_{i=1}^{r_{\widetilde{v}}} (\chi_{x,\widetilde{v},i})_{\overline{\varpi}_{\widetilde{v}}}^{\infty}$  denotes the smooth character of  $Z_{L_P}(\mathbb{Q}_p)$ such that  $(\chi_x)_{\underline{\varpi}}^{\infty} \delta_{\mathrm{wt}(\chi_x)} = (\chi_x)_{\underline{\varpi}}$ . By Proposition 3.2.1 (1), there exists a classical automorphic representation  $\pi = \pi_{\infty} \boxtimes \pi^{\infty,p} \boxtimes (\boxtimes_{v \in S_p} \pi_{\widetilde{v}})$  such that  $\eta_{\pi} = \eta_x$ , and a  $\mathrm{GL}_n(F_{\widetilde{v}})$ -equivariant surjection

$$\left(\operatorname{Ind}_{P_{\widetilde{v}}^{-}(F_{\widetilde{v}})}^{\operatorname{GL}_{n}(F_{\widetilde{v}})}\pi_{x,L_{P_{\widetilde{v}}}}\otimes_{k(x)}\left((\chi_{x,\widetilde{v}})_{\varpi_{\widetilde{v}}}^{\infty}\circ\det_{L_{P_{\widetilde{v}}}}\right)\otimes_{E}\delta_{P_{\widetilde{v}}}^{-1}\right)^{\infty}\longrightarrow\pi_{\widetilde{v}}.$$

We have then (see for example [80, Thm. 1.2(b)])

$$\operatorname{rec}(\pi_{\widetilde{v}})|_{W_{F_{\widetilde{v}}}} = \bigoplus_{i=1}^{r_{\widetilde{v}}} \Big( \operatorname{rec}(\pi_{x,\widetilde{v},i}) \Big( \frac{n-n_{\widetilde{v},i}}{2} - s_{\widetilde{v},i-1} \Big) \otimes_{k(x)} \operatorname{rec}((\chi_{x,\widetilde{v},i})_{\varpi_{\widetilde{v}}}^{\infty}) \Big).$$
(3.21)

Denote by  $\mathbf{r}_{x,\tilde{v}}$  the Weil-Deligne representation associated to (the Deligne-Fontaine module of)  $\rho_{x,\tilde{v}}$ , and  $\mathbf{r}_{x,\tilde{v}}^{ss}$  its *F*-semi-simplification. By the local-global compatibility in classical local Langlands correspondence for  $\ell = p$  (e.g. [25]), we have  $\mathbf{r}_{x,\tilde{v}}^{ss} \cong \operatorname{rec}(\pi_{\tilde{v}})(\frac{1-n}{2})$ . We deduce hence

$$\mathbf{r}_{x,\widetilde{v}}^{\mathrm{ss}}|_{W_{F_{\widetilde{v}}}} \cong \bigoplus_{i=1}^{r_{\widetilde{v}}} \big( \mathbf{r}_{x,\widetilde{v},i} \otimes_{k(x)} \operatorname{rec}((\chi_{x,\widetilde{v},i})_{\varpi_{\widetilde{v}}}^{\infty}) \big).$$

We call the point x generic if, for  $\epsilon = 0, 1$  and for all  $i \neq i'$  and  $v \in S_p$  we have

$$\mathbf{r}_{x,\widetilde{v},i}\otimes_{k(x)}\operatorname{rec}((\chi_{x,\widetilde{v},i})_{\varpi_{\widetilde{v}}}^{\infty})\cong\mathbf{r}_{x,\widetilde{v},i'}(\epsilon)\otimes_{k(x)}\operatorname{rec}((\chi_{x,\widetilde{v},i'})_{\varpi_{\widetilde{v}}}^{\infty})$$

or equivalently

$$\pi_{x,\widetilde{v},i} \otimes_{k(x)} \left( \left( \operatorname{unr}(q_{\widetilde{v}}^{s_{\widetilde{v},i-1}-(1-n_{\widetilde{v},i})/2})(\chi_{x,\widetilde{v},i})_{\varpi_{\widetilde{v}}}^{\infty} \right) \circ \det_{L_{P_{\widetilde{v}}}} \right) \\ \cong \pi_{x,\widetilde{v},i'} \otimes_{k(x)} \left( \left( \operatorname{unr}(q_{\widetilde{v}}^{s_{\widetilde{v},i'-1}-(1-n_{\widetilde{v},i'})/2+\epsilon})(\chi_{x,\widetilde{v},i'})_{\varpi_{\widetilde{v}}}^{\infty} \right) \circ \det_{L_{P_{\widetilde{v}}}} \right). \quad (3.22)$$

Assume that x is generic, it is easy to see that  $\rho_{x,\tilde{v}}$  is potentially crystalline and generic for all  $v \in S_p$  (see § 2.3), hence we have (here there is no need to take semi-simplification and restriction to  $W_{F_{\tilde{v}}}$ )

$$\mathbf{r}_{x,\widetilde{v}} \cong \bigoplus_{i=1}^{r_{\widetilde{v}}} \left( \mathbf{r}_{x,\widetilde{v},i} \otimes_{k(x)} \operatorname{rec}((\chi_{x,\widetilde{v},i})_{\varpi_{\widetilde{v}}}^{\infty}) \right).$$
(3.23)

We obtain thus a  $P_{\tilde{v}}$ -filtration  $\mathscr{F}_{\tilde{v}}$  on  $\mathbf{r}_{x,\tilde{v}}$  (§ 2.3) given by

$$\operatorname{Fil}_{i}^{\mathscr{F}_{\widetilde{v}}} \mathbf{r}_{x,\widetilde{v}} := \bigoplus_{j=1}^{i} \left( \mathbf{r}_{x,\widetilde{v},j} \otimes_{k(x)} \operatorname{rec}((\chi_{x,\widetilde{v},j})_{\varpi_{\widetilde{v}}}^{\infty}) \right), \quad i \in \{0,\ldots,r_{\widetilde{v}}\}.$$
(3.24)

Recall also that  $\rho_{x,\tilde{v}}$  is de Rham of Hodge-Tate weights

$$\mathbf{h}_{x,\widetilde{v}} := (\mathbf{h}_{x,\widetilde{v},i})_{1 \le i \le n} := \{h_{x,\widetilde{v},i,\tau} := \lambda_{\widetilde{v},i,\tau}^x - i + 1\}_{\substack{\tau \in \Sigma_{\widetilde{v}} \\ 1 \le i \le n}}.$$

As in § 2.3, we associate to  $(\rho_{x,\tilde{v}}, \mathscr{F}_{\tilde{v}})$  an element  $w_{x,\mathscr{F}_{\tilde{v}}} \in \mathscr{W}_{F_{\tilde{v}}} \cong \mathscr{W}^{\oplus |\Sigma_{\tilde{v}}|}$  (denoted by  $w_{\mathscr{F}}$  in *loc. cit.*). Let  $\Delta_{x,\tilde{v}}$  be the *p*-adic differential equation associated to  $\mathbf{r}_{x,\tilde{v}}$  and  $\Delta_{x,\tilde{v},i}$  the *p*-adic differential equation associated to  $\mathbf{r}_{x,\tilde{v}}$  and  $\Delta_{x,\tilde{v},i}$  the *p*-adic differential equation associated to  $\mathbf{r}_{x,\tilde{v},i}$  (§ 2.2). We have  $D_{\mathrm{rig}}(\rho_{x,\tilde{v}})[1/t] \cong \Delta_{x,\tilde{v}}[1/t]$  and by (3.23):

$$\Delta_{x,\widetilde{v}} \cong \bigoplus_{i=1}^{r_{\widetilde{v}}} \left( \Delta_{x,\widetilde{v},i} \otimes_{\mathcal{R}_{k(x),F_{\widetilde{v}}}} \mathcal{R}_{k(x),F_{\widetilde{v}}}((\chi_{x,\widetilde{v},i})_{\varpi_{\widetilde{v}}}^{\infty}) \right).$$

As discussed in § 2.3, the  $P_{\tilde{v}}$ -filtration  $\mathscr{F}_{\tilde{v}}$  on  $\mathbf{r}_{x,\tilde{v}}$  induces a  $P_{\tilde{v}}$ -filtration on  $\Delta_{x,\tilde{v}}$  which further induces a  $P_{\tilde{v}}$ -filtration (still denoted)  $\mathscr{F}_{\tilde{v}}$  on  $D_{\mathrm{rig}}(\rho_{x,\tilde{v}})$ . Recall also that, if  $\mathscr{F}_{\tilde{v}}$  is non-critical, then  $\mathrm{gr}_{i}^{\mathscr{F}_{\tilde{v}}} D_{\mathrm{rig}}(\rho_{x,\tilde{v}})$  is de Rham of Hodge-Tate weights  $\{h_{x,\tilde{v},j,\tau}\}_{\substack{\tau \in \Sigma_{\tilde{v}} \\ s_{\tilde{v},i-1} < j \leq s_{\tilde{v},i}}$ . Let

$$\mathbf{h}_{\widetilde{v}} := (\mathbf{h}_{\widetilde{v},i})_{1 \le i \le n} = (h_{\widetilde{v},i,\tau} := \lambda_{\widetilde{v},i,\tau} - i + 1)_{\substack{\tau \in \Sigma_{\widetilde{v}} \\ 1 \le i \le n}},$$
(3.25)

and note that  $\mathbf{h}_{\widetilde{v}}$  is strictly  $P_{\widetilde{v}}$ -dominant. Suppose that  $\mathscr{F}_{\widetilde{v}}$  is non-critical, we have then injections for  $i \in \{1, \ldots, r_{\widetilde{v}}\}$  (cf. (2.4)):

$$\operatorname{gr}_{i}^{\mathscr{F}_{\widetilde{v}}} D_{\operatorname{rig}}(\rho_{x,\widetilde{v}}) \otimes_{\mathcal{R}_{k(x),F_{\widetilde{v}}}} \mathcal{R}_{k(x),F_{\widetilde{v}}}\big((\chi_{x,\widetilde{v},i})_{\varpi_{\widetilde{v}}}^{-1} z^{-\mathbf{h}_{\widetilde{v},s_{i}}}\big) \longleftrightarrow \Delta_{x,\widetilde{v},i}.$$

**Lemma 3.2.14.** Let  $x = (\eta_x, \pi_{x,L_P}, \chi_x) \in \mathcal{E}_{\Omega,\lambda}(U^p)$  be a classical point. Assume

$$\operatorname{val}_{\widetilde{v}}(\omega_{\pi_{x,\widetilde{v},i}}(\varpi_{\widetilde{v}})) - \operatorname{val}_{\widetilde{v}}(\omega_{\pi_{x,\widetilde{v},i'}}(\varpi_{\widetilde{v}})) \\ \neq \sum_{\tau \in \Sigma_{\widetilde{v}}} \left(\operatorname{wt}(\chi_{x})_{\widetilde{v},i,\tau} - \operatorname{wt}(\chi_{x})_{\widetilde{v},i',\tau}\right) + [F_{\widetilde{v}}:\mathbb{Q}_{p}](s_{\widetilde{v},i} - s_{\widetilde{v},i'} + \epsilon)$$

for  $\epsilon = 0, 1$  and for all  $i \neq i'$ , then  $\rho_{x,\tilde{v}}$  is generic. If moreover (3.17) holds then  $(\rho_{x,\tilde{v}}, \mathscr{F}_{\tilde{v}})$  is non-critical.

*Proof.* The first part of the lemma is straightforward to check. By the proof of Proposition 3.2.9, if (3.17) holds, then for any  $1 \neq w_{\tilde{v}} \in (\mathscr{W}_{\min}^{P_{\tilde{v}}})^{\oplus |\Sigma_{\tilde{v}}|}$ , there exists  $z \in Z_{L_{P_{\tilde{v}}}}(F_{\tilde{v}})^+$  such that

$$\delta_{w_{\widetilde{v}}\cdot\lambda_{\widetilde{v}}^{x}}(z)\delta_{P_{\widetilde{v}}}^{-1}(z)\omega_{\pi_{x,L_{P_{\widetilde{v}}}}}(z)\notin\mathcal{O}_{k(x)}.$$

By [13, Thm. 7.6], we deduce  $w_{x,\mathscr{F}_{\widetilde{v}}} = w_{0,F_{\widetilde{v}}}$  (noting that the  $w^{\text{alg}}$  of *loc. cit.* is  $w_{x,\mathscr{F}_{\widetilde{v}}}w_{0,F_{\widetilde{v}}}$  in our case). Hence  $(\rho_{x,\widetilde{v}},\mathscr{F}_{\widetilde{v}})$  is non-critical by definition (§ 2.3).

We call a classical point  $x \in \mathcal{E}_{\Omega,\lambda}(U^p)$  non-critical if  $(\rho_{x,\tilde{v}}, \mathscr{F}_{\tilde{v}})$  is non-critical for all  $v \in S_p$ . By the same argument as in the proof of [17, Thm. 3.9] and using Proposition 3.2.9, Lemma 3.2.14, we have the following strengthening of Theorem 3.2.11: **Theorem 3.2.15.** The set of very classical non-critical generic points is Zariski-dense in  $\mathcal{E}_{\Omega,\lambda}(U^p)$ . Moreover, for any point  $x = (\eta, \pi_{L_P}, \chi) \in \mathcal{E}_{\Omega,\lambda}(U^p)$  with  $\chi$  locally algebraic, and for any admissible open  $U \subseteq \mathcal{E}_{\Omega,\lambda}(U^p)$  containing x, there exists an admissible open  $V \subseteq U$  containing x such that the set of very classical non-critical generic points in V is Zariski-dense in V.

The following theorem is an analogue of the statement "Galois representations on eigenvarieties are trianguline".

**Theorem 3.2.16.** Let  $x = (\eta_x, \pi_{x,L_P}, \chi_x)$  be a point of  $\mathcal{E}_{\Omega,\lambda}(U^p)$ . Then for  $v \in S_p$ ,  $D_{\mathrm{rig}}(\rho_{x,\widetilde{v}})$ admits a  $P_{\widetilde{v}}$ -filtration  $\mathscr{F}_{\widetilde{v}} = \mathrm{Fil}_{\bullet}^{\mathscr{F}_{\widetilde{v}}} D_{\mathrm{rig}}(\rho_{x,\widetilde{v}})$  of saturated  $(\varphi, \Gamma)$ -submodules of  $D_{\mathrm{rig}}(\rho_{x,\widetilde{v}})$  such that

$$\big(\operatorname{gr}_{i}^{\mathscr{F}_{\widetilde{v}}}D_{\operatorname{rig}}(\rho_{x,\widetilde{v}})\big)[1/t] \cong \big(\Delta_{x,\widetilde{v},i}\otimes_{\mathcal{R}_{k(x),F_{\widetilde{v}}}}\mathcal{R}_{k(x),F_{\widetilde{v}}}((\chi_{x,\widetilde{v},i})_{\varpi_{\widetilde{v}}})\big)[1/t].$$

*Proof.* Since  $\mathcal{E}_{\Omega,\lambda}(U^p)^{\text{red}}$  is nested, by [3, Lemma 7.2.9] and Theorem 3.2.15, there exists an irreducible affinoid neighbourhood U of x in  $\mathcal{E}_{\Omega,\lambda}(U^p)^{\text{red}}$  such that the set C(U) of very classical generic non-critical points in U is Zariski-dense in U. By pulling-back the universal character of  $\mathcal{Z}_0$  over  $\widehat{\mathcal{Z}}_0$  via the composition

$$U \hookrightarrow \mathcal{E}_{\Omega,\lambda}(U^p)^{\mathrm{red}} \hookrightarrow \mathcal{E}_{\Omega,\lambda}(U^p) \to (\operatorname{Spec} \mathcal{Z}_\Omega)^{\mathrm{rig}} \times \widehat{\mathcal{Z}_0} \twoheadrightarrow \widehat{\mathcal{Z}_0},$$

we obtain a continuous character  $\chi_U = \boxtimes_{v \in S_p} \boxtimes_{i=1}^{r_{\widetilde{v}}} \chi_{U,\widetilde{v},i} : Z^0_{L_P} \to \mathcal{O}(U)^{\times}$ . For  $v \in S_p$  and  $1 \leq i \leq r_{\widetilde{v}}$ , by pulling-back the "universal" *p*-adic differential equation over  $(\operatorname{Spec} \mathcal{Z}_{\Omega_{\widetilde{v},i}})^{\operatorname{rig}}$  constructed in § 2.2 via the composition

$$U \hookrightarrow \mathcal{E}_{\Omega,\lambda}(U^p) \to (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0} \twoheadrightarrow (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \simeq \prod_{v \in S_p} \prod_{i=1}^{r_{\widetilde{v}}} (\operatorname{Spec} \mathcal{Z}_{\Omega_{\widetilde{v},i}})^{\operatorname{rig}} \twoheadrightarrow (\operatorname{Spec} \mathcal{Z}_{\Omega_{\widetilde{v},i}})^{\operatorname{rig}},$$

we obtain a  $(\varphi, \Gamma)$ -module  $\Delta'_{U,\tilde{v},i}$  over  $\mathcal{R}_{U,F_{\tilde{v}}}$ . We let

$$\Delta_{U,\widetilde{v},i} := \Delta'_{U,\widetilde{v},i} \otimes_{\mathcal{R}_{E,F_{\widetilde{v}}}} \mathcal{R}_{E,F_{\widetilde{v}}} \big( \operatorname{unr}(q_{\widetilde{v}}^{s_{\widetilde{v},i-1}-(1-n_{\widetilde{v},i})/2}) \big).$$

Thus for each point x in U, the evaluation  $x^*\Delta_{U,\tilde{v},i}$  is isomorphic to  $\Delta_{x,\tilde{v},i}$ . Applying [3, Lemma 7.8.11], we obtain a rigid analytic space  $\widetilde{U}$  with a finite dominant (surjective) morphism  $g: \widetilde{U} \to U$  and a locally free  $\mathcal{O}(\widetilde{U})$ -module  $\rho_{\widetilde{U}}$  of rank n equipped with a continuous  $\operatorname{Gal}_F^S$ -action such that  $g^{-1}(C(U))$  is Zariski-dense in  $\widetilde{U}$  and  $\rho_{\widetilde{U}}|_x = \rho_{g(x)}$  for all  $x \in \widetilde{U}$ . Applying Corollary A.1.2 and Corollary A.1.3 (1) to (noting that, for  $x \in \widetilde{U}^{\mathrm{red}}$ ,  $\Delta_i|_x = \Delta_{g(x),\tilde{v},i}$  and  $\delta_i|_x = \chi_{g(x),\tilde{v},i}$ )

$$\left\{X = \widetilde{U}^{\mathrm{red}}, M = D_{\mathrm{rig}}(\rho_{\widetilde{U}^{\mathrm{red}},\widetilde{v}}), \Delta_i := g^* \Delta_{U,\widetilde{v},i}|_{\widetilde{U}^{\mathrm{red}}}, \delta_i := (g^* \chi_{U,\widetilde{v},i}) z^{\mathbf{h}_{\widetilde{v},s_i}}|_{\widetilde{U}^{\mathrm{red}}}\right\},$$

the theorem follows.

**Proposition 3.2.17.** Let  $x = (\eta_x, \pi_{x,L_P}, \chi_x)$  be a point of  $\mathcal{E}_{\Omega,\lambda}(U^p)$ . Then for  $v \in S_p$ , and  $\tau \in \Sigma_{\widetilde{v}}$ , the Sen  $\tau$ -weights of  $\rho_{x,\widetilde{v}}$  are given by  $\{h_{j_i,\tau} + \operatorname{wt}(\chi_{x,\widetilde{v},i})_{\tau}\}_{\substack{1 \leq i \leq r \\ s_{i-1}+1 \leq j_i \leq s_i}}$ .

*Proof.* We use the notation in the proof of Theorem 3.2.16. The statement holds for very classical non-critical generic points, hence for points in  $g^{-1}(C(U)) \subset \widetilde{U}$ . Since the Sen  $\tau$ -weights are analytic functions on  $\widetilde{U}$  (see for example [60, Def. 6.2.11]), we deduce the statement holds for any point in  $\widetilde{U}$ . The proposition follows.

Let  $\overline{\rho}$  be a continuous representation of  $\operatorname{Gal}_F$  of dimension n over  $k_E$  such that  $\overline{\rho}(c \cdot c) \cong \overline{\rho}^{\vee} \otimes \overline{\chi_{\operatorname{cyc}}}^{1-n}$  (where  $\operatorname{Gal}(F/F^+) = \{1, c\}$ ) and  $\overline{\rho}$  is unramified outside S. To  $\overline{\rho}$ , we associate a maximal ideal  $\mathfrak{m}_{\overline{\rho}}$  of  $\mathbb{T}^S$  of residue field  $k_E$  such that, denoting  $\eta_{\overline{\rho}} : \mathbb{T}^S/\mathfrak{m}_{\overline{\rho}} \twoheadrightarrow k_E$  the corresponding morphism, the characteristic polynomial of  $\overline{\rho}(\operatorname{Frob}_{\widetilde{v}})$  is given by (3.19) with  $\eta_{\pi}$  replaced by  $\eta_{\overline{\rho}}$  for all  $v \notin S$  that splits in F. The representation  $\overline{\rho}$  is called  $U^p$ -modular if the localization  $S(U^p, \mathcal{O}_E/\varpi_E^s)_{\overline{\rho}} := S(U^p, \mathcal{O}_E/\varpi_E^s)_{\mathfrak{m}_{\overline{\rho}}}$  is non-zero (by e.g. [16, Lemma 6.5]) there exist only finitely many  $\overline{\rho}$  such that  $\overline{\rho}$  is  $U^p$ -modular). We define

$$\widehat{S}(U^p, \mathcal{O}_E)_{\overline{\rho}} := \varprojlim_s S(U^p, \mathcal{O}_E/\varpi_E^s)_{\overline{\rho}} \quad \text{and} \quad \widehat{S}(U^p, E)_{\overline{\rho}} := \widehat{S}(U^p, \mathcal{O}_E)_{\overline{\rho}} \otimes_{\mathcal{O}_E} E.$$

We construct  $\mathcal{E}_{\Omega,\lambda}(U^p,\overline{\rho})$  from  $\widehat{S}(U^p,E)_{\overline{\rho}}$  exactly as we construct  $\mathcal{E}_{\Omega,\lambda}(U^p)$ . Since  $\widehat{S}(U^p,E)_{\overline{\rho}}$  is a direct summand of  $\widehat{S}(U^p,E)$  (equivariant under the action of  $G(F \otimes_{\mathbb{Q}} \mathbb{Q}_p) \times \mathbb{T}^S$ ), one easily sees that all the previous results hold with  $\mathcal{E}_{\Omega,\lambda}(U^p)$  replaced by  $\mathcal{E}_{\Omega,\lambda}(U^p,\overline{\rho})$ .

#### 3.2.5 Locally analytic socle and companion points

We recall the locally analytic socle conjecture of [13], [14] for generic potentially crystalline representations and discuss its relation with companion points on Bernstein eigenvarieties.

To any *n*-dimensional continuous representation  $\rho$  of  $\operatorname{Gal}_F$  over E we associate a maximal ideal  $\mathfrak{m}_{\rho}$  of  $\mathbb{T}^S \otimes_{\mathcal{O}_E} E$  of residue field E such that, denoting  $\eta_{\rho} : \mathbb{T}^S \twoheadrightarrow E$  the corresponding morphism, the characteristic polynomial of  $\rho(\operatorname{Frob}_{\widetilde{v}})$  is given by (3.19) with  $\eta_{\pi}$  replaced by  $\eta_{\rho}$  for all  $v \notin S$  that splits in F. Let  $\rho$  be such a representation and assume that  $\widehat{S}(U^p, E)[\mathfrak{m}_{\rho}] = \widehat{S}(U^p, E)[\mathbb{T}^S = \eta_{\rho}]$  is non-zero. For  $v \in S_p$ , assume also that  $\rho_{\widetilde{v}}$  is generic potentially crystalline (see § 2.3) of Hodge-Tate weights  $(h_{\widetilde{v},1,\tau} > h_{\widetilde{v},2,\tau} > \cdots > h_{\widetilde{v},n,\tau})_{\tau \in \Sigma_{\widetilde{v}}}$ . Let  $\mathscr{F}_{\widetilde{v}}$  be a minimal parabolic filtration of  $\mathbf{r}(\rho_{\widetilde{v}})$ , and  $P_{\widetilde{v}}$  be the associated parabolic subgroup of  $\operatorname{GL}_n$  (cf. § 2.3). We use the notation  $n_{\widetilde{v},i}, r_{\widetilde{v}}$  and  $s_{\widetilde{v},i}$  of § 3.2.2. We let  $\pi_{L_{P_{\widetilde{v}}}} := \boxtimes_{i=1}^{r_{\widetilde{v}}} \pi_{\widetilde{v},i}$  be the smooth irreducible representation of  $L_{P_{\widetilde{v}}}(F_{\widetilde{v}})$  over E such that  $\operatorname{rec}(\pi_{\widetilde{v},i})((1-n_{\widetilde{v},i})/2-s_{\widetilde{v},i}) \cong \operatorname{gr}_{i}^{\mathscr{F}_{\widetilde{v}}} \mathbf{r}(\rho)$ . For  $\tau \in \Sigma_{\widetilde{v}}$  and  $j = 1, \ldots, n$ , let  $\lambda_{\widetilde{v},\tau,j} = h_{\widetilde{v},\tau,j} + j - 1$ , then  $\lambda_{\widetilde{v}} := (\lambda_{\widetilde{v},1,\tau}, \ldots, \lambda_{\widetilde{v},n,\tau})_{\tau \in \Sigma_{\widetilde{v}}}$  is a dominant weight of  $\operatorname{Res}_{\mathbb{Q}_p}^{F_{\widetilde{v}}} \operatorname{GL}_n$  (with respect to  $\operatorname{Res}_{\mathbb{Q}_p}^{F_{\widetilde{v}}} B$ ). For  $w_{\widetilde{v}} \in \mathscr{W}_{\min,F_{\widetilde{v}}}^{P_{\widetilde{v}}}$ , consider the following locally  $\mathbb{Q}_p$ -analytic representation of  $\operatorname{GL}_n(F_{\widetilde{v}})$  over E:

$$C(w_{\widetilde{v}},\mathscr{F}_{\widetilde{v}}) := \mathcal{F}_{P_{\widetilde{v}}^{-}(F_{\widetilde{v}})}^{\mathrm{GL}_{n}(F_{\widetilde{v}})} \big( L^{-}(-w_{\widetilde{v}} \cdot \lambda_{\widetilde{v}}), \pi_{L_{P_{\widetilde{v}}}} \otimes_{E} \delta_{P_{\widetilde{v}}}^{-1} \big).$$

It is topologically irreducible by [74, Thm. (iv)]) (indeed, since  $\rho_{\widetilde{v}}$  is generic, the smooth induction  $(\operatorname{Ind}_{P_{\widetilde{v}}(F_{\widetilde{v}})\cap L_Q(F_{\widetilde{v}})}^{L_Q(F_{\widetilde{v}})}\pi_{L_{P_{\widetilde{v}}}}\otimes_E \delta_{P_{\widetilde{v}}}^{-1})^{\infty}$  is irreducible for any parabolic  $Q \supset P_{\widetilde{v}}$ ). Note that we have  $C(1, \mathscr{F}_{\widetilde{v}}) \cong \pi_{\widetilde{v}} \otimes_E L(\lambda_{\widetilde{v}})$  where  $\pi_{\widetilde{v}} := (\operatorname{Ind}_{P_{\widetilde{v}}}^{\operatorname{GL}_n} \pi_{L_{P_{\widetilde{v}}}} \otimes_E \delta_{P_{\widetilde{v}}}^{-1})^{\infty}$  is the smooth irreducible representation of  $\operatorname{GL}_n(F_{\widetilde{v}})$  such that  $\operatorname{rec}(\pi_{\widetilde{v}})(\frac{1-n}{2}) \cong r(\rho_{\widetilde{v}})$ . As discussed in § 2.3, to the filtration  $\mathscr{F}_{\widetilde{v}}$ , we associate an element  $w_{\mathscr{F}_{\widetilde{v}}} \in \mathscr{W}_{\max,F_{\widetilde{v}}}^{P_{\widetilde{v}}}$ . The following conjecture is a special case of [14, Conj. 5.3]:

**Conjecture 3.2.18.** For  $v \in S_p$ , let  $w_{\widetilde{v}} \in \mathscr{W}_{\min,F_{\widetilde{v}}}^{P_{\widetilde{v}}}$ . Then  $\widehat{\otimes}_{v \in S_p} C(w_{\widetilde{v}},\mathscr{F}_{\widetilde{v}})$  is a subrepresentation of  $\widehat{S}(U^p, E)[\mathfrak{m}_{\rho}]$  if and only if  $w_{\widetilde{v}} \leq w_{\mathscr{F}_{\widetilde{v}}} w_{0,F_{\widetilde{v}}}$  for all  $v \in S_p$ .

Conjecture 3.2.18 in the crystalline case has been proved in [19] under Taylor-Wiles hypothesis. However, almost nothing (except for some very partial results in [38]) was known when  $\rho_{\tilde{v}}$  is *not* trianguline. As already mentioned in § 1, one main motivation for this paper is to prove Conjecture 3.2.18 for certain parabolic  $P_{\tilde{v}}$  (under the Taylor-Wiles hypothesis 3.3.1) by using the Bernstein eigenvarieties of § 3.2.2.

We now state a weaker version of Conjecture 3.2.18 which is formulated in terms of companion points on Bernstein eigenvarieties.

For  $w = (w_{\widetilde{v}})_{v \in S_p} \in \mathscr{W}_{\min}^P \cong \prod_{v \in S_p} \mathscr{W}_{\min, F_{\widetilde{v}}}^{P_{\widetilde{v}}}$ , the weight  $w \cdot \lambda$  is *P*-dominant where  $\lambda := (\lambda_{\widetilde{v}})_{v \in S_p}$ . Let  $\Omega_{\mathscr{F}}$  be the Bernstein component of  $\pi_{L_P} := \boxtimes_{v \in S_p} \pi_{L_{P_{\widetilde{v}}}}$ .

Conjecture 3.2.19. We have

$$\left(\eta_{\rho}, \boxtimes_{v \in S_p} \pi_{L_{P_{\widetilde{w}}}}, 1\right) \in \mathcal{E}_{\Omega_{\mathscr{F}}, w \cdot \lambda}(U^p) \tag{3.26}$$

if and only if  $w_{\widetilde{v}} \leq w_{\mathscr{F}_{\widetilde{v}}} w_{0,F_{\widetilde{v}}}$  for all  $v \in S_p$ .

Lemma 3.2.20. (1) Conjecture 3.2.18 implies Conjecture 3.2.19.

(2) The "only if" part of Conjecture 3.2.19 implies the "only if" part of Conjecture 3.2.18.

*Proof.* (1) Suppose Conjecture 3.2.18 holds. By (3.13), (3.26) is equivalent to

$$\operatorname{Hom}_{L_P(\mathbb{Q}_p)}\left(\pi_{L_P}\otimes_E L(w\cdot\lambda)_P, J_P(\widehat{S}(U^p, E)^{\operatorname{an}})[\mathfrak{m}_\rho]\right) \neq 0.$$
(3.27)

If  $w_{\widetilde{v}} \leq w_{\mathscr{F}_{\widetilde{v}}} w_{0,F_{\widetilde{v}}}$  for all  $v \in S_p$ , then by assumption

$$\mathcal{F}_{P^{-}}^{G_{p}}(L^{-}(-w\cdot\lambda),\pi_{L_{P}}\otimes_{E}\delta_{P}^{-1})\cong\widehat{\otimes}_{v\in S_{p}}C(w_{\widetilde{v}},\mathscr{F}_{\widetilde{v}})\hookrightarrow\widehat{S}(U^{p},E)[\mathfrak{m}_{\rho}].$$

By [14, Thm. 4.3], this implies (3.27). Conversely, suppose (3.27) holds. By [14, Thm. 4.3] and [74, Thm.] (and the fact that the  $\rho_{\tilde{v}}$  are all generic), one deduces there exists  $w' = (w'_{\tilde{v}})$  with  $w'_{\tilde{v}} \geq w_{\tilde{v}}$  for all  $v \in S_p$  such that one has an embedding

$$\mathcal{F}_{P^{-}}^{G_{p}}(L^{-}(-w'\cdot\lambda),\pi_{L_{P}}\otimes_{E}\delta_{P}^{-1})\cong\widehat{\otimes}_{v\in S_{p}}C(w'_{\widetilde{v}},\mathscr{F}_{\widetilde{v}})\hookrightarrow\widehat{S}(U^{p},E)[\mathfrak{m}_{\rho}].$$

By Conjecture 3.2.18, this implies  $w_{\widetilde{v}} \leq w_{\mathscr{F}_{\widetilde{v}}} w_{0,F_{\widetilde{v}}}$ , and hence  $w_{\widetilde{v}} \leq w_{\mathscr{F}_{\widetilde{v}}} w_{0,F_{\widetilde{v}}}$ .

(2) Suppose the "only if" part of Conjecture 3.2.19 holds. For  $w = (w_{\tilde{v}})$ , if one has an embedding  $\widehat{\otimes}_{v \in S_p} C(w_{\tilde{v}}, \mathscr{F}_{\tilde{v}}) \hookrightarrow \widehat{S}(U^p, E)[\mathfrak{m}_{\rho}]$ , then as in the proof of (1), we deduce using [14, Thm. 4.3] that (3.27) holds and hence (3.26) holds. This implies  $w_{\tilde{v}} \leq w_{\mathscr{F}_{\tilde{v}}} w_{0,F_{\tilde{v}}}$ .

**Remark 3.2.21.** Suppose  $\widehat{S}(U^p, E)^{\text{lalg}}[\mathfrak{m}_{\rho}] \neq 0$  (for  $\rho_{\widetilde{v}}$  generic potentially crystalline with distinct Hodge-Tate weights for all  $v \in S_p$ ). Then using [25], we have an embedding

$$\boxtimes_{v \in S_p} \pi_{\widetilde{v}} \hookrightarrow \widehat{S}(U^p, E)^{\text{lalg}}[\mathfrak{m}_{\rho}].$$

To each minimal parabolic filtration  $\mathscr{F}$  (that can be viewed as an analogue of a refinement in the crystalline case), we can associate (using [14, Thm. 4.3]) a classical point

$$(\eta_{\rho}, \boxtimes_{v \in S_p} \pi_{L_{P_{\widetilde{v}}}}, 1) \in \mathcal{E}_{\Omega_{\mathscr{F}}, \lambda}(U^p)$$

When  $w_{\mathscr{F}} \neq w_{0,F} := (w_{0,F_{\widetilde{v}}})_{v \in S_p}$ , Conjecture 3.2.19 then predicts the existence of some nonclassical points on the Bernstein eigenvarieties such that the associated Galois representation is still isomorphic to  $\rho$ . Such points are referred to as companion points of the classical point.

# 3.3 Patched Bernstein eigenvarieties

As in [17, § 3] for the "usual" eigenvarieties, we construct patched Bernstein eigenvarieties by applying the formalism in § 3.1 to the patched *p*-adic automorphic representations of [26]. We give some basic properties of patched Bernstein eigenvarieties, and show that they have a close relation with certain (purely local) "paraboline" varieties (on the Galois side) that will be constructed in § 4.1.

Let  $\overline{\rho}$  be a  $U^p$ -modular continuous representation of  $\operatorname{Gal}_F$  over  $k_E$  (see the end of § 3.2.4). We assume henceforth the following so-called Taylor-Wiles hypothesis (see [19, § 1]):

# Hypothesis 3.3.1. (1) p > 2;

(2) the field F is unramified over  $F^+$ , F does not contain a non trivial root  $\sqrt[p]{1}$  of 1 and G is quasi-split at all finite places of  $F^+$ ;

(3)  $U_v$  is hyperspecial when the finite place v of  $F^+$  is inert in F;

(4)  $\overline{\rho}$  is absolutely irreducible and  $\overline{\rho}(\operatorname{Gal}_{F(\frac{p}{1})})$  is adequate.

The following functor

$$\left\{ \begin{array}{c} \text{Local artinian } \mathcal{O}_E\text{-algebras} \\ \text{of residue field } k_E \end{array} \right\} \to \{\text{Sets}\}, \quad A \mapsto \left\{ \begin{array}{c} \text{Deformations } \rho_A \text{ of } \overline{\rho} \text{ over } A \\ \text{such that } \rho_A \text{ unramified outside } S \\ \text{and } \rho_A(c \cdot c) \cong \rho_A^{\vee} \otimes \chi_{\text{cyc}}^{1-n} \end{array} \right\} / \sim$$

is pro-representable by a complete local Noetherian  $\mathcal{O}_E$ -algebra of residue field  $k_E$ , denoted by  $R_{\overline{\rho},S}$ . For  $s \in \mathbb{Z}_{\geq 1}$ , and a compact open subgroup  $U_p \subset \prod_{v \in S_p} \operatorname{GL}_n(\mathcal{O}_{F_{\overline{v}}})$ , let  $\mathbb{T}(U^p U_p, \mathcal{O}_E / \varpi_E^s)_{\overline{\rho}}$  be the image of  $\mathbb{T}^S$  in the  $\mathcal{O}_E / \varpi_E^s$ -algebra of endomorphisms of  $S(U^p U_p, \mathcal{O}_E / \varpi_E^s)_{\overline{\rho}}$ . Put

$$\mathbb{T}(U^p)_{\overline{\rho}} := \varprojlim_s \varprojlim_{U_p} \mathbb{T}(U^p U_p, \mathcal{O}_E / \varpi_E^s)_{\overline{\rho}},$$

which is a complete local  $\mathcal{O}_E$ -algebra. By [84, Prop. 6.7], there is a natural surjective morphism of  $\mathcal{O}_E$ -algebras (hence  $\mathbb{T}(U^p)_{\overline{\rho}}$  is also Noetherian)

$$R_{\overline{\rho},S} \longrightarrow \mathbb{T}(U^p)_{\overline{\rho}}$$

The rigid space  $\mathcal{E}_{\Omega,\lambda}(U^p,\overline{\rho})$  is then a closed subspace of the rigid spaces

$$(\operatorname{Spf} \mathbb{T}(U^p)_{\overline{\rho}})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0} \hookrightarrow (\operatorname{Spf} R_{\overline{\rho},S})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0}.$$

Indeed  $B_{\Omega,\lambda}(\widehat{S}(U^p, E)_{\overline{\rho}}^{\mathbb{Q}_p-\mathrm{an}})^{\vee}$  gives rise to a coherent sheaf over  $(\operatorname{Spf} R_{\overline{\rho},S})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0}$ , whose Zariski-support is exactly  $\mathcal{E}_{\Omega,\lambda}(U^p, \overline{\rho})$ .

We let  $R_{\overline{\rho}_{\widetilde{v}}}$  be the maximal reduced and  $\mathbb{Z}_p$ -flat quotient of the *framed* local deformation ring of  $\overline{\rho}_{\widetilde{v}}$  (which was denoted by  $R_{\overline{\rho}_{\widetilde{v}}}^{\Box}$  in [17]) and we put

$$R^{\mathrm{loc}} := \widehat{\otimes}_{v \in S} R_{\overline{\rho}_{\widetilde{v}}}, \ R_{\overline{\rho}^p} := \widehat{\otimes}_{v \in S \setminus S_p} R_{\overline{\rho}_{\widetilde{v}}}, R_{\overline{\rho}_p} := \widehat{\otimes}_{v \in S_p} R_{\overline{\rho}_{\widetilde{v}}}, \ R_{\infty} := R^{\mathrm{loc}}[[x_1, \dots, x_g]], \ R_{\infty}^p := R_{\overline{\rho}^p}[[x_1, \dots, x_g]],$$

where  $g \geq 1$  is an integer which will be fixed below. We let  $S_{\infty} := \mathcal{O}_E[[y_1, \ldots, y_t]]$  where  $t = g + [F^+ : \mathbb{Q}] \frac{n(n-1)}{2} + |S|n^2$  and  $\mathfrak{a} := (y_1, \ldots, y_t) \subset S_{\infty}$ . Shrinking  $U^p$  (and S), we can and do assume

$$G(F) \cap (hU^pK_ph^{-1}) = \{1\}$$
 for all  $h \in G(\mathbb{A}_{F^+}^\infty)$ 

where  $K_p := \prod_{v \in S_p} K_v := \prod_{v \in S_p} i_{\tilde{v}}^{-1}(\operatorname{GL}_n(\mathcal{O}_{F_{\tilde{v}}}))$ . Then the action of  $R_{\bar{\rho},S}$  on  $\widehat{S}(U^p, E)_{\bar{\rho}}$  factors through a quotient  $R_{\bar{\rho},S} \twoheadrightarrow R_{\bar{\rho},S}$ , where  $R_{\bar{\rho},S}$  denotes the universal deformation ring of the deformation problem:

$$\mathcal{S} = (F/F^+, S, \widetilde{S}, \mathcal{O}_E, \overline{\rho}, \chi_{\text{cyc}}^{1-n} \delta_{F/F^+}^n, \{R_{\overline{\rho}_{\widetilde{v}}}\}_{v \in S}).$$

By [17, Thm. 3.5] (generalizing [26]), there exist an integer  $g \ge 1$  and

- (1) a continuous  $R_{\infty}$ -admissible unitary representation  $\Pi_{\infty}$  of  $G_p$  over E together with a  $G_p$ stable and  $R_{\infty}$ -stable unit ball  $\Pi^0_{\infty} \subset \Pi_{\infty}$ ;
- (2) a morphism of local  $\mathcal{O}_E$ -algebras  $S_{\infty} \to R_{\infty}$  such that  $M_{\infty} := \operatorname{Hom}_{\mathcal{O}_E}(\Pi_{\infty}^0, \mathcal{O}_E)$  is a finite projective  $S_{\infty}[[K_p]]$ -module;
- (3) a surjection  $R_{\infty}/\mathfrak{a}R_{\infty} \twoheadrightarrow R_{\overline{\rho},S}$  and a compatible  $G_p$ -equivariant isomorphism  $\Pi_{\infty}[\mathfrak{a}] \cong \widehat{S}(U^p, E)_{\overline{\rho}}$ .

Denote by  $\Pi_{\infty}^{R_{\infty}-\mathrm{an}}$  the subrepresentation of  $G_p$  of locally  $R_{\infty}$ -analytic vectors of  $\Pi_{\infty}$  (cf. [17, § 3.1]. Let  $\Omega$ ,  $\sigma$ ,  $\lambda$  be as in § 3.2.2 and consider (with  $J_P(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\lambda}$  defined as in (3.1)):

$$B_{\Omega,\lambda}(\Pi_{\infty}^{R_{\infty}-\mathrm{an}}) := B_{\sigma,\lambda}(\Pi_{\infty}^{R_{\infty}-\mathrm{an}}) = \mathrm{Hom}_{K_{p}}\left(\sigma, J_{P}(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\lambda}\widehat{\otimes}_{E}\mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z_{L_{P}}^{0}, E)\right)$$

By [17, Prop. 3.4] and an easy variation of the proof of Lemma 3.1.3, we see that  $B_{\Omega,\lambda}(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})^{\vee}$ is a coadmissible module over  $\mathcal{O}((\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times \widehat{Z}_{\underline{\omega}} \times \widehat{Z}_{0})$  which corresponds to a coherent sheaf  $\mathcal{M}_{\Omega,\lambda}^{\infty,0}$  over  $(\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times \widehat{Z}_{\underline{\omega}} \times \widehat{Z}_{0}$ . Taking into account the  $\mathcal{Z}_{\Omega}$ -action,  $\mathcal{M}_{\Omega,\lambda}^{\infty,0}$  gives rise to a coherent sheaf  $\mathcal{M}_{\Omega,\lambda}^{\infty}$  over  $(\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$ , such that

$$\Gamma\left((\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}, \mathcal{M}_{\Omega,\lambda}^{\infty}\right) \cong B_{\Omega,\lambda}(\Pi_{\infty}^{R_{\infty}-\operatorname{an}})^{\vee}.$$
(3.28)

We let

$$\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho}) \hookrightarrow (\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$$
(3.29)

be the Zariski-closed support of  $\mathcal{M}_{\Omega,\lambda}^{\infty}$ , and call  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  a patched Bernstein eigenvariety.

Let  $\mathcal{W}_{\infty} := (\operatorname{Spf} S_{\infty})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$ , and  $z \in Z_{\underline{\omega}}$  be as in the discussion above Proposition 3.1.15. As in (3.14) denote by  $\kappa_{z}$  the following composition

$$\kappa_{z}: \mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho}) \hookrightarrow (\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0}} \to (\operatorname{Spf} S_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} E[Z_{\underline{\varpi}}])^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0}} \to (\operatorname{Spf} S_{\infty})^{\operatorname{rig}} \times \mathbb{G}_{m}^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0}} \cong \mathcal{W}_{\infty} \times \mathbb{G}_{m}^{\operatorname{rig}}$$

where the second and third maps come from  $E[Y_z] \hookrightarrow E[Z_{\overline{\omega}}] \to \mathcal{Z}_{\Omega}$ . Denote by  $\omega$  the composition:

$$\omega: \mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho}) \xrightarrow{\kappa_z} \mathcal{W}_{\infty} \times \mathbb{G}^{\mathrm{rig}}_m \longrightarrow \mathcal{W}_{\infty}.$$

By an easy generalization of Proposition 3.1.15 to the case  $\widetilde{G}_p := G_p \times \mathbb{Z}_p^q$ ,  $V = \Pi_{\infty}^{R_{\infty}-an} = \Pi_{\infty}^{S_{\infty}-an}$ (adding the extra factor  $\mathbb{Z}_p^q$  everywhere in the proof), one can show that  $B_{\Omega,\lambda}(\Pi_{\infty}^{R_{\infty}-an})^{\vee}$  is a coherent sheaf over  $\mathcal{W}_{\infty} \times \mathbb{G}_m^{\mathrm{rig}}$ . We denote by  $Z_z(\overline{\rho}) \hookrightarrow \mathcal{W}_{\infty} \times \mathbb{G}_m^{\mathrm{rig}}$  its Zariski-closed support. We have the following analogue of Proposition 3.2.4, which is proven by the same argument as in the proofs of [17, Lemma 3.10] and [17, Prop. 3.11]. **Proposition 3.3.2.** (1) The rigid variety  $Z_z(\overline{\rho})$  is a Fredholm hypersurface in  $\mathcal{W}_{\infty} \times \mathbb{G}_m^{\mathrm{rig}}$ . Moreover, there exists an admissible covering  $\{U'_i\}$  of  $Z_z(\overline{\rho})$  by affinoids  $U'_i$  such that the morphism

$$g: Z_z(\overline{
ho}) \hookrightarrow \mathcal{W}_\infty imes \mathbb{G}_m^{\mathrm{rig}} \twoheadrightarrow \mathcal{W}_\infty$$

induces a surjective finite morphism from  $U'_i$  to an affinoid open  $W_i$  of  $\mathcal{W}_{\infty}$ , and such that  $U'_i$  is a connected component of  $g^{-1}(W_i)$ . For each i,  $\Gamma(U'_i, (\kappa_z)_*\mathcal{M}^{\infty}_{\Omega,\lambda})$  is a finitely generated projective  $\mathcal{O}_{\mathcal{W}_{\infty}}(W_i)$ -module.

- (2) There exists an admissible covering  $\{U_i\}$  of  $\mathcal{E}^{\infty}_{\Omega,\lambda}(\bar{\rho})$  by affinoids  $U_i$  such that
- there exists an affinoid open  $W_i$  of  $\mathcal{W}_{\infty}$  satisfying that  $\omega$  is a finite surjective morphism from each irreducible component of  $U_i$  to  $W_i$ ;
- $\mathcal{O}_{\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})}(U_i)$  is isomorphic to an  $\mathcal{O}_{\mathcal{W}_{\infty}}(W_i)$ -algebra of endomorphisms of a finitely generated projective  $\mathcal{O}_{\mathcal{W}_{\infty}}(W_i)$ -module.

We deduce the following analogue of Corollary 3.2.5 by the same arguments as in the proofs of [17, Cor. 3.12], [17, Cor. 3.13] and [18, Lemma 3.8]:

**Corollary 3.3.3.** (1) The rigid analytic space  $\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})$  is equidimensional of dimension

$$g + |S|n^{2} + \sum_{v \in S_{p}} \left( [F_{\widetilde{v}} : \mathbb{Q}_{p}] \left( \frac{n(n-1)}{2} + r_{\widetilde{v}} \right) \right)$$

and has no embedded component.

(2) The morphism  $\kappa_z$  is finite and the image of an irreducible component of  $\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})$  is an irreducible component of  $Z_z(\overline{\rho})$ . The image of an irreducible component of  $\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})$  under  $\omega$  is a Zariski-open of  $\mathcal{W}_{\infty}$ .

(3) The coherent sheaf  $\mathcal{M}^{\infty}_{\Omega,\lambda}$  is Cohen-Macaulay over  $\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})$ .

We say that a point  $x = (y, \pi_{x,L_P}, \chi_x) \in \mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho}) \hookrightarrow (\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$  is classical if (3.15) is satisfied with  $\widehat{S}(U^p, E)^{\operatorname{lalg}}$  replaced by  $(\Pi_{\infty}^{R_{\infty}-\operatorname{an}})^{\operatorname{lalg}}$  and  $[\mathbb{T}^S = \eta_x]$  replaced by  $[\mathfrak{m}_y]$ , where  $\mathfrak{m}_y$  is the maximal ideal of  $R_{\infty}[\frac{1}{p}]$  associated to y. We say x is very classical if the conditions in Definition 3.2.7 (2) hold (these conditions only concern  $\pi_{x,L_P}$  and  $\chi_x$ ). For  $v \in S$ , denote by  $\rho_{x,\widetilde{v}}$  the  $\operatorname{Gal}_{F_{\widetilde{v}}}$ -representation associated to x via the image of  $y \in (\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \cong$  $\prod_{v \in S_p} (\operatorname{Spf} R_{\overline{\rho}_{\widetilde{v}}})^{\operatorname{rig}} \times \mathbb{U}^g$  in  $(\operatorname{Spf} R_{\overline{\rho}_{\widetilde{v}}})^{\operatorname{rig}}$  where  $\mathbb{U}$  denotes the open unit ball in  $\mathbb{A}^1$ . Let  $\mathbf{r}_{x,\widetilde{v}}$  be the Weil-Deligne representation associated to  $\rho_{x,\widetilde{v}}, \mathbf{r}_{x,\widetilde{v},i} := \operatorname{rec}(\pi_{x,\widetilde{v},i})(\frac{1-n_{\widetilde{v},i}}{2} - s_{\widetilde{v},i-1})$ , and  $(\chi_{x,\widetilde{v},i})_{\varpi_{\widetilde{v}}}^{\infty}$ be the smooth character of  $F_{\widetilde{v}}^{\times}$  associated to  $\chi_{x,\widetilde{v}}$  as in the discussion below (3.20). A classical point x is called generic if (3.22) holds for  $\epsilon = 0, 1$  and  $i \neq i'$ .

**Proposition 3.3.4.** Let  $(y, \pi_{x,L_P}, \chi_x) \in \mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})$  be a generic classical point. For all  $v \in S_p$ ,  $\rho_{x,\widetilde{v}}$  is potentially crystalline of Hodge-Tate weights  $\{\lambda^x_{\widetilde{v},i,\tau} - i + 1\}_{\substack{\tau \in \Sigma_{\widetilde{v}} \\ 1 \leq i \leq n}}$ . Moreover, we have

$$\mathbf{r}_{x,\widetilde{v}} \cong \bigoplus_{i=1}^{r_{\widetilde{v}}} \left( \mathbf{r}_{x,\widetilde{v},i} \otimes_E \operatorname{rec}((\chi_{x,\widetilde{v},i})_{\varpi_{\widetilde{v}}}^{\infty}) \right).$$
(3.30)

*Proof.* The proposition follows from the results in [26, § 4] by similar arguments as in the proof of [17, Prop. 3.16]. Since x is classical, we have a non-zero morphism (recall  $\lambda^x = \lambda + (\operatorname{wt}(\chi_x) \circ \operatorname{det}_{L_P}))$ 

$$\left(\operatorname{Ind}_{P^{-}(\mathbb{Q}_{p})}^{G_{p}}\left(\pi_{x,L_{P}}\otimes_{E}\left((\chi_{x})_{\underline{\varpi}}^{\infty}\circ\det_{L_{P}}\right)\otimes_{E}\delta_{P}^{-1}\right)\right)^{\infty}\otimes_{E}L(\lambda^{x})\longrightarrow(\Pi_{\infty}^{R_{\infty}-\operatorname{an}})^{\operatorname{lalg}}[\mathfrak{m}_{y}].$$
(3.31)

Since x is generic, the representation on the left hand side is absolutely irreducible and hence (3.31) is injective. For  $v \in S_p$ , let  $G_{\tilde{v}} := \operatorname{GL}_n(F_{\tilde{v}})$ , and  $\Omega_{G_{\tilde{v}}}$  be the Bernstein component of

$$\pi_{x,\widetilde{v}} := \left( \operatorname{Ind}_{P_{\widetilde{v}}^{-}(F_{\widetilde{v}})}^{G_{\widetilde{v}}} \left( \pi_{x,L_{P_{\widetilde{v}}}} \otimes_{E} \left( (\chi_{x,\widetilde{v}})_{\varpi_{\widetilde{v}}}^{\infty} \circ \det_{L_{P_{\widetilde{v}}}} \right) \otimes_{E} \delta_{P_{\widetilde{v}}}^{-1} \right) \right)^{\infty}$$

(do not confuse  $\Omega_{G_{\widetilde{v}}}$  with the Bernstein component  $\Omega_{\widetilde{v}}$  of  $L_{P_{\widetilde{v}}}(F_{\widetilde{v}})$ ). By [26, Thm. 3.7] and [26, Cor. 3.12], there exists a smooth irreducible representation  $\sigma_{\widetilde{v}}$  of  $K_{\widetilde{v}} := \operatorname{GL}_n(\mathcal{O}_{F_{\widetilde{v}}})$  such that  $\operatorname{End}_{G_{\widetilde{v}}}(\operatorname{c-ind}_{K_{\widetilde{v}}}^{G_{\widetilde{v}}}\sigma_{\widetilde{v}}) \cong \mathcal{Z}_{\Omega_{G_{\widetilde{v}}}}$  and  $\pi_{x,\widetilde{v}} \cong (\operatorname{c-ind}_{K_{\widetilde{v}}}^{G_{\widetilde{v}}}\sigma_{\widetilde{v}}) \otimes_{\mathcal{Z}_{\Omega_{G_{\widetilde{v}}}},\theta_{x,\widetilde{v}}} E$ , where  $\theta_{x,\widetilde{v}} : \mathcal{Z}_{\Omega_{G_{\widetilde{v}}}} \to E$  denotes the character corresponding to  $\pi_{x,\widetilde{v}}$ . Let  $K_p := \prod_{v \in S_p} K_{\widetilde{v}}, \sigma_{K_p} := \boxtimes_{v \in S_p} \sigma_{K_{\widetilde{v}}}, \Omega_{G_p} := \prod_{v \in S_p} \Omega_{G_{\widetilde{v}}},$ and

$$\pi_x := \boxtimes_{v \in S_p} \pi_{x,\widetilde{v}} \cong \left(\operatorname{c-ind}_{K_p}^{G_p} \sigma_{K_p}\right) \otimes_{\mathcal{Z}_{\Omega_{G_p}}, \theta_x} E \cong \left(\operatorname{Ind}_{P^-(\mathbb{Q}_p)}^{G_p} \left(\pi_{x,L_P} \otimes_E \left((\chi_x)_{\underline{\varpi}}^{\infty} \circ \det_{L_P}\right) \otimes_E \delta_P^{-1}\right)\right)^{\infty}$$

(where  $\theta_x : \mathcal{Z}_{\Omega_{G_p}} \to E$  denotes the character corresponding to  $\pi_x$ ). By Frobenius reciprocity, we have

$$\Pi_{\infty}(\sigma_{K_p},\lambda^x) := \operatorname{Hom}_{K_p}\left(\sigma_{K_p},\Pi_{\infty}\otimes_E L(\lambda^x)^{\vee}\right) \cong \operatorname{Hom}_{G_p}\left(\operatorname{c-ind}_{K_p}^{G_p}\sigma_{K_p},\Pi_{\infty}\otimes_E L(\lambda^x)^{\vee}\right).$$

By the injection (3.31), we deduce that  $\Pi_{\infty}(\sigma_{K_p}, \lambda^x)[\mathfrak{m}_y] \neq 0$  and that there exists a non-zero subspace of  $\Pi_{\infty}(\sigma_{K_p}, \lambda^x)[\mathfrak{m}_y]$  on which  $\mathcal{Z}_{\Omega_{G_p}}$  acts via  $\theta_x$ . Let  $\xi_{\widetilde{v}}$  be the inertial type associated to  $\sigma_{\widetilde{v}}, \mathbf{h}_{\widetilde{v}}^x := (h_{\widetilde{v},i,\tau}^x)_{\substack{i=1,...,n\\\tau\in\Sigma_{\widetilde{v}}}}$  with  $h_{\widetilde{v},i,\tau}^x = \lambda_{\widetilde{v},i,\tau}^x - i + 1$ . By [26, Lemma 4.17 (1)], the action of  $\widehat{\otimes}_{v\in S_p}R_{\overline{\rho}_{\widetilde{v}}}$  on  $\Pi_{\infty}(\sigma_{K_p}, \lambda^x)$  factors through  $\widehat{\otimes}_{v\in S_p}R_{\overline{\rho}_{\widetilde{v}}}^{\mathrm{pcr}}(\xi_{\widetilde{v}}, \mathbf{h}_{\widetilde{v}}^x)$ . The first part of the proposition follows. By [26, Thm. 4.1], there is a morphism  $\mathcal{Z}_{\Omega_{G_{\widetilde{v}}}} \to R_{\overline{\rho}_{\widetilde{v}}}^{\mathrm{pcr}}(\xi_{\widetilde{v}}, \lambda_{\widetilde{v}}^x)[1/p]$  that interpolates the local Langlands correspondence (with the same normalization as in § 2.1). By [26, Lemma 4.17 (2)], the action of  $\mathcal{Z}_{\Omega_{G_p}}$  on  $\Pi_{\infty}(\sigma_{K_p}, \lambda^x)$  factors through  $\mathcal{Z}_{\Omega_{G_p}} \to (\widehat{\otimes}_{v\in S_p}R_{\overline{\rho}_{\widetilde{v}}}^{\mathrm{pcr}}(\xi_{\widetilde{v}}, \lambda_{\widetilde{v}}^x))[1/p]$ . We deduce  $\mathbf{r}_{x,\widetilde{v}} \cong \operatorname{rec}(\pi_{x,\widetilde{v}})((1-n)/2)$ , and the second part follows then from (3.21).

For a generic classical point  $x \in \mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$ , we use (3.30) to define a  $P_{\tilde{v}}$ -filtration  $\mathscr{F}_{\tilde{v}}$  on  $\mathbf{r}_{x,\tilde{v}}$  as in (3.24). We call x non-critical if  $(\rho_{x,\tilde{v}},\mathscr{F}_{\tilde{v}})$  is non-critical for all  $v \in S_p$ . Using Proposition 3.2.9, Lemma 3.2.14, and by the same argument as in the proof of [17, Thm. 3.9], we have the following analogue of Theorem 3.2.15:

**Theorem 3.3.5.** The set of very classical non-critical generic points is Zariski-dense in  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$ . Moreover, for any point  $x = (y, \pi_{L_P}, \chi) \in \mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  with  $\chi$  locally algebraic, and for any admissible open  $U \subseteq \mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  containing x, there exists an admissible open  $V \subseteq U$  containing x such that the set of very classical non-critical generic points in V is Zariski-dense in V.

**Proposition 3.3.6.** The rigid space  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  is reduced.

Proof. The proposition follows by an easy variation of the proof of [17, Cor. 3.20]. We briefly indicate below the changes. We define verbatim the  $R_{\infty}$ -module  $\Sigma$  of *loc. cit.* In [17], it was a finite length  $T_p$ -module on which  $T_p^0 := \prod_{v \in S_p} T(\mathcal{O}_{F_{\tilde{v}}})$  acts by the character  $\delta_{\lambda}$  of *loc. cit.* In our case, it becomes a finite length  $\mathcal{Z}_{\Omega}$ -module on which  $\mathcal{Z}_0 \cong Z_{L_p}^0$  acts by an algebraic character. Similarly as in *loc. cit.*, we are reduced to show that the  $R_{\infty}$ -action and the  $\mathcal{Z}_{\Omega}$ -action on  $\Sigma$  are both semi-simple. To obtain an analogue of [17, (3.13)], we use Lemma 3.1.2 and [14, Thm. 4.3]. The " $\mathcal{H}$ " in the proof of [17, Cor. 3.20] has to be replaced by a Bernstein component  $\Omega_{G_p} = \boxtimes_{v|p} \Omega_{G_{\tilde{v}}}$ of  $G_p$  similarly as the one appearing in the proof of Proposition 3.3.4, and "ind $_{K_p}^{G_p}$ 1" is replaced by c-ind $_{K_p}^{G_p} \sigma_{K_p}$  where  $\sigma_{K_p} = \bigotimes_v \sigma_{\tilde{v}}$  is the  $K_p$ -representation associated to  $\Omega_{G_p}$  as in the proof of Proposition 3.3.4. Finally, " $R_{\overline{\rho}_p}^{\Box,\mathbf{k}-\mathrm{cr}}$ " has to be replaced by  $\widehat{\otimes}_{v\in S_p} R_{\overline{\rho}_{\tilde{v}}}^{\mathrm{pcr}}(\xi_{\tilde{v}}, \mathbf{h}_{\tilde{v}}^x)$  (as in the proof of Proposition 3.3.4).

**Remark 3.3.7.** Assume  $P = B_p$  where  $B_p$  is as in Remark 3.2.6. Using the isomorphism  $\iota_{\Omega,\lambda}$  in Remark 3.1.10, we can view  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  as a (reduced) closed rigid analytic subspace of  $(\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times \widehat{T_p}$  $(T_p \text{ as in Remark 3.2.6})$ , which is independent of the choice of  $(\Omega, \lambda)$  by Proposition 3.2.3 (trivially generalized to the patched case). By construction, the rigid analytic space  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  has the same points as the patched eigenvariety  $X_p(\overline{\rho})$  of [17]. Using Proposition 3.3.6 and [17, Cor. 3.20], we actually obtain  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho}) \cong X_p(\overline{\rho})$ .

**Proposition 3.3.8.** Let  $(\operatorname{Spf} S_{\infty})^{\operatorname{rig}} \to \operatorname{Spec} E$  be the morphism corresponding to the quotient by the ideal  $\mathfrak{a}$ . We have a natural morphism of rigid analytic spaces

$$\mathcal{E}_{\Omega,\lambda}(U^p,\overline{\rho})\longrightarrow \mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})\times_{(\operatorname{Spf} S_{\infty})^{\operatorname{rig}}} \operatorname{Spec} E$$

which is bijective on points.

*Proof.* By the same argument as in the proof of [17, Thm. 4.2], we have

$$J_P(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})[\mathfrak{a}] \cong J_P(\Pi_{\infty}^{R_{\infty}-\mathrm{an}}[\mathfrak{a}]) \cong J_P(\widehat{S}(U^p, E)_{\overline{\rho}}^{\mathrm{an}}).$$

Hence  $B_{\Omega,\lambda}(\Pi^{R_{\infty}-\mathrm{an}}_{\infty})[\mathfrak{a}] \cong B_{\Omega,\lambda}(\widehat{S}(U^p, E)^{\mathrm{an}}_{\overline{\rho}})$ . The proposition follows.

For  $v \in S_p$ , let  $j_{\tilde{v}}$  denote the isomorphism

$$\mathfrak{z}_{\widetilde{v}}: (\operatorname{Spec} \mathcal{Z}_{\Omega_{\widetilde{v}}})^{\operatorname{rig}} \xrightarrow{\sim} (\operatorname{Spec} \mathcal{Z}_{\Omega_{\widetilde{v}}})^{\operatorname{rig}}$$

such that  $\pi_{j_{\widetilde{v}}(\underline{x})_i} \cong \pi_{x_i} \otimes_E \operatorname{unr} \left(q_{\widetilde{v}}^{s_{\widetilde{v},i-1}-\frac{1-n_{\widetilde{v},i}}{2}}\right) \circ \det$  for  $\underline{x} = (x_i) \in (\operatorname{Spec} \mathcal{Z}_{\Omega_{\widetilde{v}}})^{\operatorname{rig}}$ . Let  $j := (j_{\widetilde{v}})_{v \in S_p} :$  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \to (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}}$ . Denote by  $\mathcal{Z}_{0,\widetilde{v}} := Z_{L_{P_{\widetilde{v}}}}(\mathcal{O}_{F_{\widetilde{v}}})$ , thus  $\widehat{\mathcal{Z}}_0 \cong \prod_{v \in S_p} \widehat{\mathcal{Z}}_{0,\widetilde{v}}$ . Consider the composition:

$$\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho}) \longleftrightarrow (\operatorname{Spf} R_{\infty}^{p})^{\operatorname{rig}} \times \prod_{v \in S_{p}} \left( (\operatorname{Spf} R_{\overline{\rho}_{\widetilde{v}}})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega_{\widetilde{v}}})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,\widetilde{v}}} \right) \\
\xrightarrow{\mathcal{I}} (\operatorname{Spf} R_{\infty}^{p})^{\operatorname{rig}} \times \prod_{v \in S_{p}} \left( (\operatorname{Spf} R_{\overline{\rho}_{\widetilde{v}}})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega_{\widetilde{v}}})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,\widetilde{v}}} \right) \quad (3.32)$$

where the second map (still denoted by j) is the identity on the factors other than  $\prod_{v} (\operatorname{Spec} \mathcal{Z}_{\Omega_{\widetilde{v}}})^{\operatorname{rig}}$ and is j on  $\prod_{v} (\operatorname{Spec} \mathcal{Z}_{\Omega_{\widetilde{v}}})^{\operatorname{rig}} \cong (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}}$ . Let  $\mathbf{h} = (\mathbf{h}_{\widetilde{v}})_{v \in S_{p}}$  be associated to  $\lambda$  as in (3.25). In the next section (see § 4.2 below), we construct a reduced closed subspace  $X_{\Omega_{\widetilde{v}},\mathbf{h}_{\widetilde{v}}}(\overline{\rho}_{\widetilde{v}})$  of  $(\operatorname{Spf} R_{\overline{\rho}_{\widetilde{v}}})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega_{\widetilde{v}}})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,\widetilde{v}}}$ . We let  $X_{\Omega,\mathbf{h}}(\overline{\rho}_{p}) := \prod_{v \in S_{p}} X_{\Omega_{\widetilde{v}},\mathbf{h}_{\widetilde{v}}}(\overline{\rho}_{\widetilde{v}})$ .

**Theorem 3.3.9.** The composition in (3.32) factors through  $(\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}} \times X_{\Omega,h}(\overline{\rho}_p)$ , and induces an isomorphism between  $\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})$  and a union of irreducible components of  $(\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}} \times X_{\Omega,h}(\overline{\rho}_p)$ equipped with the reduced closed rigid subspace structure.

*Proof.* By Proposition 3.3.4, and the construction of  $\{X_{\Omega_{\widetilde{v}},\mathbf{h}_{\widetilde{v}}}(\overline{\rho}_{\widetilde{v}})\}$  in § 4.2, one sees that all the generic classical points of  $\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})$  are sent to

$$(\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}} \times \prod_{v \in S_p} U_{\Omega, \mathbf{h}}(\overline{\rho}_{\widetilde{v}}) \hookrightarrow (\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}} \times X_{\Omega, \mathbf{h}}(\overline{\rho}_p)$$

where we refer to the discussion above Proposition 4.2.2 for the rigid analytic space  $U_{\Omega,\mathbf{h}}(\overline{\rho}_{\widetilde{v}})$ . The first part then follows from the density of generic classical points (Theorem 3.3.5). The second part follows from Proposition 3.3.6, from the fact  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  is closed in the right hand side of (3.32), hence in  $(\operatorname{Spf} R_{\infty}^p)^{\operatorname{rig}} \times X_{\Omega,\mathbf{h}}(\overline{\rho}_p)$ , and from the fact that both rigid spaces  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  and  $(\operatorname{Spf} R_{\infty}^p)^{\operatorname{rig}} \times X_{\Omega,\mathbf{h}}(\overline{\rho}_p)$  have the same dimension (by Corollary 3.3.3 for the first and Proposition 4.2.5 (1) for  $X_{\Omega_{\widetilde{v}},\mathbf{h}_{\widetilde{v}}}(\overline{\rho}_{\widetilde{v}})$  together with dim $(\operatorname{Spf} R_{\infty}^p)^{\operatorname{rig}} = g + \sum_{v \in S \setminus S_p} \dim(\operatorname{Spf} R_{\overline{\rho}_{\widetilde{v}}})^{\operatorname{rig}} = g + n^2 |S \setminus S_p|)$ .

**Remark 3.3.10.** Let  $\mathfrak{X}^p$  be an irreducible component of  $(\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}}$ . Call an irreducible component  $X_p$  of  $X_{\Omega,h}(\overline{\rho}_p)$   $\mathfrak{X}^p$ -automorphic if  $\mathfrak{X}^p \times X_p$  is contained in the image of (3.32). One may expect that  $X_p$  is  $\mathfrak{X}^p$ -automorphic if and only if  $X_p$  contains a generic potentially crystalline point with distinct Hodge-Tate weights. We refer to [17, § 3.6] for related discussions in the trianguline case.

We finally discuss the problem of companion constituents and companion points in the patched setting. Let  $\mathfrak{m}$  be a maximal ideal of  $R_{\infty}[1/p]$  such that  $\Pi_{\infty}[\mathfrak{m}] \neq 0$ . For  $v \in S_p$ , we assume that the  $\operatorname{Gal}_{F_{v}}$ -representation  $\rho_{\tilde{v}}$  associated to  $\mathfrak{m}$  (as above Proposition 3.3.4) is generic potentially crystalline with distinct Hodge-Tate weights. We use the notation in § 3.2.5.

**Conjecture 3.3.11.** For  $v \in S_p$ , let  $w = (w_{\widetilde{v}}) \in \mathscr{W}_{\min}^P = \prod_{v \in S_p} \mathscr{W}_{\min, F_{\widetilde{v}}}^{P_{\widetilde{v}}}$ .

(1) The representation  $\widehat{\otimes}_{v \in S_p} C(w_{\widetilde{v}}, \mathscr{F}_{\widetilde{v}})$  is a subrepresentation of  $\Pi_{\infty}^{R_{\infty}-\mathrm{an}}[\mathfrak{m}]$  if and only if  $w_{\widetilde{v}} \leq w_{\mathscr{F}_{\widetilde{v}}} w_0$  for all  $v \in S_p$ .

(2) The point  $(\mathfrak{m}, \boxtimes_{v \in S_p} \pi_{L_{P_{\widetilde{v}}}}, 1) \in (\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$  lies in  $\mathcal{E}^{\infty}_{\Omega, w \cdot \lambda}(\overline{\rho})$  if and only if  $w_{\widetilde{v}} \leq w_{\mathscr{F}_{\widetilde{v}}} w_{0}$  for all  $v \in S_{p}$ .

**Remark 3.3.12.** The point  $(\mathfrak{m}, \boxtimes_{v \in S_p} \pi_{L_{P_{\overline{v}}}}, 1) \in \mathcal{E}_{\Omega, w \cdot \lambda}^{\infty}(\overline{\rho})$  of Conjecture 3.3.11 (2) may be referred to as "a companion point of  $x = (\mathfrak{m}, \boxtimes_{v \in S_p} \pi_{L_{P_{\overline{v}}}}, 1)$  seen in  $\mathcal{E}_{\Omega, \lambda}^{\infty}(\overline{\rho})$ ". In the case of the patched eigenvariety  $X_p(\overline{\rho})$  of [19], there is a canonical embedding  $X_p(\overline{\rho}) \hookrightarrow (\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times \widehat{T_p}$  and the companion points are the distinct points that lie above a same point  $y \in (\operatorname{Spf} R_{\infty})^{\operatorname{rig}}$ . In our case however, as there are different rigid spaces depending on  $(\Omega, \lambda)$ , it seems more convenient to fix the point  $x \in (\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0}$  and let the Bernstein patched eigenvarieties (together with the embedding into  $(\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_0})$  vary. See also Remark 4.3.10.

By the same argument, we have as in Lemma 3.2.20:

Lemma 3.3.13. (1) Conjecture 3.3.11 (1) implies Conjecture 3.3.11 (2).

(2) The "only if" part of Conjecture 3.3.11 (2) implies the "only if" part of Conjecture 3.3.11 (1).

Finally, using the isomorphism  $\Pi_{\infty}[\mathfrak{a}] \cong \widehat{S}(U^p, E)_{\overline{\rho}}$  (resp. using Proposition 3.3.8), one easily deduces:

Lemma 3.3.14. Conjecture 3.3.11 (1) (resp. Conjecture 3.3.11 (2)) implies Conjecture 3.2.18 (resp. Conjecture 3.2.19).

# 4 Bernstein paraboline varieties

We now move to the Galois side. In § 4.1, we study certain paraboline deformations of  $(\varphi, \Gamma)$ modules which admit an  $\Omega$ -filtration (where  $\Omega$  is a cuspidal Bernstein component as in § 3.1.1). In § 4.2, we construct and study what we call Bernstein paraboline varieties (analogous to the trianguline variety of [17, § 2.2] when P = B). Finally in § 4.3, we study the relation between Bernstein paraboline varieties and potentially crystalline deformation spaces, and show the existence of local companion points. We frequently denote a point in a Bernstein component by its associated Weil-Deligne representation.

# 4.1 Deformations of $(\varphi, \Gamma)$ -modules

We prove various results on deformations of  $(\varphi, \Gamma)$ -modules D that admit an  $\Omega$ -filtration (where  $\Omega$  is a cuspidal Bernstein component as in § 3.1.1). We first study in § 4.1.1 deformations of irreducible constituents of D. By combining the results in § 4.1.1 with results of Chenevier on paraboline deformations, we study deformations of type  $\Omega$  in § 4.1.2 (which are special cases of paraboline deformations).

### **4.1.1** Deformations of certain irreducible $(\varphi, \Gamma)$ -modules

We study deformations of certain irreducible ( $\varphi, \Gamma$ )-modules which are de Rham up to twist. The results in this section will be used in our study of deformations of type  $\Omega$  in § 4.1.2.

We let L be a finite extension of  $\mathbb{Q}_p$  and we use the notation in § 2.1. We begin with some useful facts on extensions of p-adic differential equations. Let  $\Delta$  be an irreducible  $(\varphi, \Gamma)$ -module of rank k over  $\mathcal{R}_{E,L}$ , de Rham of constant Hodge-Tate weight 0.

**Lemma 4.1.1.** (1) Let M be a  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{E,L}$ . Assume that M admits an increasing filtration Fil<sub>•</sub> M by  $(\varphi, \Gamma)$ -submodules such that the graded pieces are all isomorphic to  $\Delta$ . Let N be a saturated  $(\varphi, \Gamma)$ -submodule of M. Then both N and M/N admit a filtration by  $(\varphi, \Gamma)$ -submodules such that the graded pieces are all isomorphic to  $\Delta$ .

(2) Let  $M_1$ ,  $M_2$  be  $(\varphi, \Gamma)$ -modules over  $\mathcal{R}_{E,L}$  which both admit an increasing filtration by  $(\varphi, \Gamma)$ -submodules such that the graded pieces are all isomorphic to  $\Delta$ . Let  $f : M_1 \to M_2$  be a morphism of  $(\varphi, \Gamma)$ -modules, then  $\operatorname{Im}(f)$  is saturated in  $M_2$ .

(3) Let M be as in (1), and  $N_1$ ,  $N_2$  be two saturated  $(\varphi, \Gamma)$ -submodules of M. Then  $N_1 + N_2$  is also saturated in M.

Proof. (1) We endow N with the induced filtration from M and M/N with the quotient filtration, and want to prove that all graded pieces are  $\Delta$ . By induction on the rank of M, we only need to show the statement in the case N is irreducible. By dévissage on Fil<sub>•</sub>M, we have  $\operatorname{Hom}_{(\varphi,\Gamma)}(N,\Delta) \neq 0$ . Hence we have an injection  $N \hookrightarrow \Delta$ , and so N is de Rham. By considering the Sen weights and using the fact N is saturated, we see that N is of constant Hodge-Tate weights 0. We deduce that N is actually isomorphic to  $\Delta$ . Let  $i \in \mathbb{Z}$  such that the injection  $j : N \hookrightarrow M$  has image contained in Fil<sub>i</sub> M but not in Fil<sub>i-1</sub> M. The non-zero composition  $N \xrightarrow{j} \operatorname{Fil}_i M \twoheadrightarrow \operatorname{gr}_i M \cong \Delta$  has to be an isomorphism, and gives a splitting Fil<sub>i</sub>  $M \cong N \oplus \operatorname{Fil}_{i-1} M$ . The filtration Fil<sub>•</sub> then induces a filtration on M/N such that all the graded pieces are isomorphic to  $\Delta$ . (2) Using (1), we are reduced to the case where f is injective. Then by induction on the rank of  $M_2$ , we are reduced to the case where  $M_1 \cong \Delta$ . But in this case, by the argument at the end of (1), there exists i such that  $\operatorname{Fil}_i M_2 \cong \operatorname{Im}(f) \oplus \operatorname{Fil}_{i-1} M_2$ . In particular  $\operatorname{Im}(f)$  is saturated.

(3) We have that  $N_1 \cap N_2$  is saturated in M. By (1) and (2), we see that the image of the composition

$$(N_1 + N_2)/(N_1 \cap N_2) \cong N_1/(N_1 \cap N_2) \oplus N_2/(N_1 \cap N_2) \hookrightarrow N/(N_1 \cap N_2)$$

is saturated. (3) follows.

We let  $\operatorname{Art}(E)$  denote the category of local artinian *E*-algebras of residue field *E*. Let *D* be a  $(\varphi, \Gamma)$ -module of rank *k* over  $\mathcal{R}_{E,L}$ . Denote by  $F_D$  the functor of deformations of *D* on  $A \in \operatorname{Art}(E)$ . Suppose that there exist a continuous character  $\delta : L^{\times} \to E^{\times}$  such that one has an embedding of  $(\varphi, \Gamma)$ -modules

$$D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta^{-1}) \hookrightarrow \Delta$$

(which implies that D is irreducible). We consider the following functor

$$F_D^0 : \operatorname{Art}(E) \longrightarrow \{\operatorname{sets}\}, \ A \longmapsto \{(D_A, \pi_A, \delta_A)\} / \sim$$

where  $D_A$  is a  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{A,L}$ ,  $\pi_{A,1} : D_A \otimes_A E \xrightarrow{\sim} D$ ,  $\delta_A : \mathcal{O}_L^{\times} \to A^{\times}$  such that  $\delta_A \equiv \delta$ (mod  $\mathfrak{m}_A$ ) ( $\mathfrak{m}_A$  is the maximal ideal of A), and there is an injection of  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{A,L}$ :

$$D_A \hookrightarrow \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta_A). \tag{4.1}$$

For  $\tau \in \Sigma_L$ , let  $h_{1,\tau}$  be the maximal  $\tau$ -Hodge-Tate weight of  $D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta^{-1})$  (thus  $h_{1,\tau} \geq 0$ ), and put  $\mathbf{h}_1 := (h_{1,\tau})_{\tau \in \Sigma_L}$ . By comparing the Hodge-Tate weights of  $\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}$  and  $D_A \otimes_{\mathcal{R}_{A,L}} \mathcal{R}_{A,L}(\delta_A^{-1})$ , and using [6, Thm. A], we see that the existence of the injection (4.1) is equivalent to the existence of an injection

$$\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(z^{\mathbf{h}_1} \delta_A) \hookrightarrow D_A.$$

$$(4.2)$$

Indeed, both are equivalent to the existence of an isomorphism of  $(\varphi, \Gamma)$ -modules over  $\mathcal{R}_{A,L}[1/t]$ :  $D_A[1/t] \cong \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta_A)[1/t].$ 

**Lemma 4.1.2.**  $F_D^0$  is a subfunctor of  $F_D$ .

*Proof.* Let  $A \in \operatorname{Art}(E)$ , and  $(D_A, \pi_A, \delta_A) \in F_D^0(A)$ . It is enough to show that  $\delta_A$  is uniquely determined by  $D_A$ . Suppose we have another  $\delta'_A$  such that  $\delta'_A \equiv \delta_A \pmod{\mathfrak{m}_A}$  and  $D_A \hookrightarrow \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta'_A)$ . This map, together with (4.2), induce

$$\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(z^{\mathbf{h}_1} \delta_A) \longleftrightarrow D_A \longleftrightarrow \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta'_A).$$

Hence we have  $\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L} \hookrightarrow \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta'_A \delta^{-1}_A z^{-\mathbf{h}_1})$ . Since  $\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}$  is de Rham, so is  $\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta'_A \delta^{-1}_A z^{-\mathbf{h}_1})$  (using that both have the same rank over  $\mathcal{R}_{E,L}$ ). By looking at the Sen weights and using  $\delta'_A \equiv \delta_A \pmod{\mathfrak{m}_A}$ , we deduce that  $\delta'_A \delta^{-1}_A$  is smooth. Then by comparing the Hodge-Tate weights (and using [6, Thm. A]), we obtain an isomorphism

$$\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L} \cong \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta'_A \delta_A^{-1}).$$
(4.3)

Let  $(\Delta \otimes_{\mathcal{R}_{E},L} \Delta^{\vee})^{0} := (\Delta \otimes_{\mathcal{R}_{E},L} \Delta^{\vee})/\mathcal{R}_{E,L}$ , then  $\Delta \otimes_{\mathcal{R}_{E},L} \Delta^{\vee} \cong (\Delta \otimes_{\mathcal{R}_{E},L} \Delta^{\vee})^{0} \oplus \mathcal{R}_{E,L}$  and  $H^{0}_{(\varphi,\Gamma)}((\Delta \otimes_{\mathcal{R}_{E},L} \Delta^{\vee})^{0}) = 0$ . We have isomorphisms

$$H^{0}_{(\varphi,\Gamma)} \left( \Delta^{\vee} \otimes_{\mathcal{R}_{E,L}} \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta'_{A} \delta^{-1}_{A}) \right) \cong H^{0}_{(\varphi,\Gamma)} \left( \mathcal{R}_{A,L}(\delta'_{A} \delta^{-1}_{A}) \right) \oplus H^{0}_{(\varphi,\Gamma)} \left( \left( \Delta^{\vee} \otimes_{\mathcal{R}_{E,L}} \Delta \right)^{0} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta'_{A} \delta^{-1}_{A}) \right) \cong H^{0}_{(\varphi,\Gamma)} \left( \mathcal{R}_{A,L}(\delta'_{A} \delta^{-1}_{A}) \right)$$
(4.4)

where the second isomorphism follows from  $H^0_{(\varphi,\Gamma)}((\Delta^{\vee}\otimes_{\mathcal{R}_{E,L}}\Delta)^0) = 0$  and an easy dévissage using  $\delta'_A \delta^{-1}_A \equiv 1 \pmod{\mathfrak{m}_A}$ . From (4.3) and (4.4), we deduce an embedding  $A \hookrightarrow H^0_{(\varphi,\Gamma)}(\mathcal{R}_{A,L}(\delta'_A \delta^{-1}_A)))$ , hence an injection  $\mathcal{R}_{A,L} \hookrightarrow \mathcal{R}_{A,L}(\delta'_A \delta^{-1}_A)$  that has to be an isomorphism by comparing the Hodge-Tate weights (recall  $\delta'_A \delta^{-1}_A$  is smooth). By [3, Prop. 2.3.1], we obtain  $\delta'_A \delta^{-1}_A = 1$ , which concludes the proof.

**Proposition 4.1.3.** The functor  $F_D^0$  is relatively representable over  $F_D$ .

*Proof.* As in [3, Prop. 2.3.9], the proposition follows from the following three properties that we will prove.

(1) If  $A \to A'$  is a morphism in  $\operatorname{Art}(E)$  and  $(D_A, \pi_A, \delta_A) \in F_D^0(A)$ , then  $(D_A \otimes_A A', \pi_A \otimes_A A') \in F_D^0(A')$ .

(2) Let  $A \hookrightarrow A'$  be an injection in  $\operatorname{Art}(E)$ ,  $(D_A, \pi_A) \in F_D(A)$ , and assume  $(D_A \otimes_A A', \pi_A \otimes_A A') \in F_D^0(A') \hookrightarrow F_D(A')$  (Lemma 4.1.2), then  $(D_A, \pi_A) \in F_D^0(A)$ .

(3) For A and A' in Art(E), if  $(D_A, \pi_A, \delta_A) \in F_D^0(A)$  and  $(D_{A'}, \pi_{A'}, \delta_{A'}) \in F_D^0(A')$ , then for  $B = A \times_E A'$  we have  $(D_B := D_A \times_D D_{A'}, \pi_B := \pi_A \circ \text{pr} = \pi_{A'} \circ \text{pr}') \in F_D^0(B)$  where  $\text{pr} : B \twoheadrightarrow A$ ,  $\text{pr}' : B \twoheadrightarrow A'$ .

The properties (1) and (3) are straightforward to verify. We prove (2). By (the proof of) Lemma 4.1.2, there is a unique continuous character  $\delta_{A'}: L^{\times} \to (A')^{\times}$  such that  $D_A \otimes_A A' \to \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A',L}(\delta_{A'})$  and  $\delta_{A'} \equiv \delta \pmod{\mathfrak{m}_A}$ . Let M be the saturated closure of  $D_A$  in  $\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A',L}(\delta_{A'})$  (see [3, § 2.2.3]). Since  $\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A',L}(\delta_{A'})$  admits a filtration with graded pieces all isomorphic to  $\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta)$ , so does M by Lemma 4.1.1 (1) (twisting by  $\mathcal{R}_{E,L}(\delta)$ ). For  $x \in \mathfrak{m}_A$ , consider the morphism  $x: M \to M$  given by multiplying by x. By Lemma 4.1.1 (2), we know xM is saturated in M and hence by induction  $\mathfrak{m}_A M$  is saturated in M. We deduce  $M \otimes_A E \cong M/\mathfrak{m}_A M$  is a  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{E,L}$  (in particular is free of finite type over  $\mathcal{R}_{E,L}$ ). Using the isomorphism  $D_A[1/t] \cong M[1/t]$ , and  $D_A[1/t] \otimes_A E \cong D[1/t]$ , we see  $M \otimes_A E$  is of rank k over  $\mathcal{R}_{E,L}$ . From Lemma 4.1.1 (1), we deduce  $M \otimes_A E \cong \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta)$ .

Consider the following  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{E,L}$ :

$$Q := \Delta \otimes_{\mathcal{R}_{E,L}} \Delta^{\vee} \cong \mathcal{R}_{E,L} \oplus (\Delta \otimes_{\mathcal{R}_{E,L}} \Delta^{\vee})^{0}$$

$$Q_{A} := (\Delta^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta^{-1})) \otimes_{\mathcal{R}_{E,L}} M$$

$$Q_{A'} := (\Delta^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta^{-1})) \otimes_{\mathcal{R}_{E,L}} (\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A',L}(\delta_{A'}))$$

$$\cong \mathcal{R}_{A',L}(\delta_{A'}\delta^{-1}) \oplus ((\Delta \otimes_{\mathcal{R}_{E,L}} \Delta^{\vee})^{0} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A',L}(\delta_{A'}\delta^{-1})).$$

We have  $Q_A \hookrightarrow Q_{A'}$  and both  $Q_{A'}$ ,  $Q_A$  are isomorphic to a successive extension of Q. We apply the functor F defined right above [3, Lemma 2.3.8]. By [3, Lemma 2.3.8] and  $\operatorname{Hom}_{(\varphi,\Gamma)}(\mathcal{R}_{E,L}, (\Delta \otimes_{\mathcal{R}_{E,L}} \Delta^{\vee})^0) = 0$ , we see  $F(Q) \cong E$  and

$$F(Q_{A'}) \cong F(\mathcal{R}_{A',L}(\delta_{A'}\delta^{-1})) \cong A'.$$

$$(4.5)$$

By the left exactness of the functor F and an obvious dévissage, we deduce

$$\dim_E F(Q_A) \le \dim_E A. \tag{4.6}$$

Consider the exact sequence

$$0 \to F(Q_A) \to F(Q_{A'}) \to F(Q_{A'}/Q_A).$$

We have

$$Q_{A'}/Q_A \cong \left(\Delta^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta^{-1})\right) \otimes_{\mathcal{R}_{E,L}} \left( \left(\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A',L}(\delta_{A'})\right)/M \right)$$

which, by Lemma 4.1.1 (1) applied to  $(\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A',L}(\delta_{A'}))/M$ , is also isomorphic to a successive extension of Q. By dévissage, we deduce  $\dim_E F(Q_{A'}/Q_A) \leq \dim_E(A'/A)$  over E. This, together with (4.5), (4.6) and an easy dimension counting, imply (4.6) is in fact an equality. Consider now

$$0 \to F(\mathfrak{m}_A Q_A) \to F(Q_A) \to F(Q). \tag{4.7}$$

By dévissage, we have again  $\dim_E F(\mathfrak{m}_A Q_A) \leq \dim_E \mathfrak{m}_A$ . Using  $\dim_E F(Q_A) = \dim_E A$ , we deduce that the right morphism is surjective (and  $\dim_E F(\mathfrak{m}_A Q_A) = \dim_E \mathfrak{m}_A$ ). Noting that (4.7) is a sequence of A-modules (with  $\mathfrak{m}_A$  acting by 0 on F(Q)) and considering the A-submodule of  $F(Q_A)$  generated by a lifting of a generator of  $F(Q) \cong E$ , we easily deduce  $A \xrightarrow{\sim} F(Q_A)$ .

Consider the  $(\varphi, \Gamma)$ -submodule  $Q_A^0$  of  $Q_A$  generated by  $F(Q_A)$ . We claim it is a rank one  $(\varphi, \Gamma)$ module over  $\mathcal{R}_{A,L}$ . Let  $Q_{A'}^0$  be the  $(\varphi, \Gamma)$ -submodule of  $Q_{A'}$  generated by  $F(Q_{A'})$ . Since we have  $F(\mathcal{R}_{A',L}(\delta_{A'}\delta^{-1})) \xrightarrow{\sim} F(Q_{A'})$ , we see  $Q_{A'}^0$  is also the  $(\varphi, \Gamma)$ -submodule of  $\mathcal{R}_{A',L}(\delta_{A'}\delta^{-1})$  generated
by  $F(Q_{A'})$ . Since  $\mathcal{R}_{A',L}(\delta_{A'}\delta^{-1})$  has a filtration with all graded pieces isomorphic to  $\mathcal{R}_{E,L}$ , by
dévissage and [3, Lemma 2.3.8 (ii)] any strict (saturated)  $(\varphi, \Gamma)$ -submodule C of  $\mathcal{R}_{A',L}(\delta_{A'}\delta^{-1})$ is such that dim<sub>E</sub>  $F(C) < \dim_E A'$ . As dim<sub>E</sub>  $F(Q_{A'}^0) = \dim_E F(Q_{A'}) = \dim_E A'$  by (4.5), we
deduce  $Q_{A'}^0 \xrightarrow{\sim} \mathcal{R}_{A',L}(\delta_{A'}\delta^{-1})$ . We also see that the natural morphism  $\mathcal{R}_{A',L} \otimes_{A'} F(Q_{A'}) \to Q_{A'}^0$ is an isomorphism. Now consider

$$\mathcal{R}_{A,L} \otimes_A F(Q_A) \hookrightarrow \mathcal{R}_{A,L} \otimes_A F(Q_{A'}) \cong \mathcal{R}_{A',L} \otimes_{A'} F(Q_{A'}) \xrightarrow{\sim} Q_{A'}^0 \hookrightarrow Q_{A'}.$$

The composition is injective and factors through  $Q_A$ . We deduce then  $\mathcal{R}_{A,L} \otimes_A F(Q_A) \xrightarrow{\sim} Q_A^0$ and hence the latter is a  $(\varphi, \Gamma)$ -module of rank 1 over  $\mathcal{R}_{A,L}$  as  $A \cong F(Q_A)$ .

Let  $\varepsilon_A : L^{\times} \to A^{\times}$  be the continuous character such that  $Q_A^0 \cong \mathcal{R}_{A,L}(\varepsilon_A)$ . By [3, Lemma 2.3.8 (1)] and the fact  $Q_A^0$  is generated by  $F(Q_A^0)$ , it is not difficult to see  $\varepsilon_A \equiv 1 \pmod{\mathfrak{m}_A}$ . Twisting by  $\Delta$  on both sides, the injection  $\mathcal{R}_{A,L}(\varepsilon_A) \hookrightarrow Q_A$  induces a morphism

$$\iota: \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\varepsilon_A) \longrightarrow M \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta^{-1}).$$

We prove that  $\iota$  is an isomorphism. It is sufficient to show it is surjective since both source and target have the same rank over  $\mathcal{R}_{E,L}$ . One easily checks that the morphism  $\mathcal{R}_{A,L}(\varepsilon_A) \hookrightarrow Q_A$  factors as

$$\mathcal{R}_{A,L}(\varepsilon_A) \hookrightarrow \Delta^{\vee} \otimes_{\mathcal{R}_{E,L}} \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\varepsilon_A) \to \Delta^{\vee} \otimes_{\mathcal{R}_{E,L}} M \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta^{-1}) \cong Q_A$$

where the second map is induced by  $\iota$  tensored with  $\Delta^{\vee}$ . In particular,  $\mathcal{R}_{A,L}(\varepsilon_A) \hookrightarrow Q_A$  factors through  $\mathcal{R}_{A,L}(\varepsilon_A) \hookrightarrow \Delta^{\vee} \otimes_{\mathcal{R}_{E,L}} \operatorname{Im}(\iota)$ . Since  $\operatorname{Im}(\iota) \cong (\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\varepsilon_A))/\operatorname{Ker}(\iota)$ , by Lemma 4.1.1 (1) it admits a filtration with graded pieces all isomorphic to  $\Delta$ . If  $\iota$  is not surjective, the multiplicity of  $\Delta$  in this filtration on  $\operatorname{Im}(\iota)$  is strictly smaller than the multiplicity of  $\Delta$  in the filtration of  $M \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta^{-1})$ , which is  $\dim_E A$  using the above equality of ranks over  $\mathcal{R}_{E,L}$ . Applying the functor F and using again a dévissage, we have in that case  $\dim_E F(\operatorname{Im}(\iota) \otimes_{\mathcal{R}_{E,L}} \Delta^{\vee}) < \dim_E A = \dim_E F(\mathcal{R}_{A,L}(\varepsilon_A))$ , which contradicts  $\mathcal{R}_{A,L}(\varepsilon_A) \hookrightarrow \operatorname{Im}(\iota) \otimes_{\mathcal{R}_{E,L}} \Delta^{\vee}$ . We deduce thus  $M \cong \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta \varepsilon_A)$ . Since  $D_A \hookrightarrow M \cong \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta \varepsilon_A)$ , we have  $D_A \in F_D^0(A)$ . This finishes the proof.

Now suppose moreover D has distinct Sen weights, hence  $D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta^{-1})$  has distinct Hodge-Tate weights. Twisting  $\delta$  by some algebraic character of  $L^{\times}$ , we can and do assume that the Hodge-Tate weights of  $D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta^{-1})$  are given by  $\mathbf{h} = (h_{1,\tau} > h_{2,\tau} \cdots > h_{k,\tau} = 0)_{\tau \in \Sigma_L}$ .

**Proposition 4.1.4.** The functor  $F_D^0$  is formally smooth of dimension  $1 + [L:\mathbb{Q}_p](1 + \frac{k(k-1)}{2})$ .

Proof. Let  $A \to A/I$  be as surjection in  $\operatorname{Art}(E)$  with  $I^2 = 0$ . We show the natural map  $F_D^0(A) \to F_D^0(A/I)$  is surjective. Let  $(D_{A/I}, \pi_{A/I}, \delta_{A/I}) \in F_D^0(A/I)$ . Let  $\delta_A : L^{\times} \to A^{\times}$  be a continuous character such that  $\delta_A \equiv \delta_{A/I} \pmod{I}$ . We have by definition an embedding

$$D_{A/I} \otimes_{\mathcal{R}_{A/I,L}} \mathcal{R}_{A/I,L}(\delta_{A/I}^{-1}) \hookrightarrow \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A/I,L}.$$

We choose a basis  $\underline{e}$  of  $D_{\mathrm{dR}}(\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L})$  over  $L \otimes_{\mathbb{Q}_p} A$  (note that the latter is a free  $L \otimes_{\mathbb{Q}_p} A$ module), and denote by  $\underline{e}_{A/I}$  the image of  $\underline{e}$  in  $D_{\mathrm{dR}}(\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A/I,L})$ . In the basis  $\underline{e}_{A/I}$ , the Hodge filtration on  $D_{\mathrm{dR}}(D_{A/I} \otimes_{\mathcal{R}_{A/I,L}} \mathcal{R}_{A/I,L}(\delta_{A/I}^{-1})) \cong D_{\mathrm{dR}}(\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A/I,L})$  induces an increasing filtration by free  $L \otimes_{\mathbb{Q}_p} A/I$ -submodules:

$$0 \neq \operatorname{Fil}_{-\mathbf{h}_{k}} D_{\mathrm{dR}} \left( \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A/I,L} \right) \subsetneq \cdots \subsetneq \operatorname{Fil}_{-\mathbf{h}_{1}} D_{\mathrm{dR}} \left( \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A/I,L} \right) = D_{\mathrm{dR}} \left( \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A/I,L} \right) \quad (4.8)$$

which then corresponds to an element  $\nu_{A/I} \in (\operatorname{Res}_{\mathbb{Q}_p}^L(\operatorname{GL}_k/B))(A/I)$ . Since the flag variety is smooth (hence formally smooth), we can choose a lifting  $\nu_A \in (\operatorname{Res}_{\mathbb{Q}_p}^L(\operatorname{GL}_k/B))(A)$  of  $\nu_{A/I}$ . Then  $\nu_A$  gives an increasing filtration by free  $L \otimes_{\mathbb{Q}_p} A$ -submodules in  $D_{\mathrm{dR}}(\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L})$ , to which we associate the Hodge filtration (still denoted by  $\nu_A$ ) on  $D_{\mathrm{dR}}(\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L})$  defined by (4.8) with A/I replaced by A. By [6, Thm. A], the filtered Deligne-Fontaine module  $(D_{\mathrm{pst}}(\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}), \nu_A)$ corresponds to a  $(\varphi, \Gamma)$ -submodule  $M_A$  of  $\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}$ . Then we see that  $D_A := M_A \otimes_{\mathcal{R}_{A,L}}$  $\mathcal{R}_{A,L}(\delta_A)$  satisfies  $D_A \equiv D_{A/I} \pmod{I}$  and  $D_A \hookrightarrow \Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta_A)$ . Hence  $F_D^0$  is formally smooth.

We next compute the dimension of the *E*-vector space  $F_D^0(E[\varepsilon/\varepsilon^2])$ . Recall that  $F_D^0(E[\varepsilon/\varepsilon^2]) \hookrightarrow F_D(E[\varepsilon/\varepsilon^2])$  and that  $F_D(E[\varepsilon/\varepsilon^2])$  is identified with  $\operatorname{Ext}^1_{(\varphi,\Gamma)}(D,D) \cong \operatorname{Ext}^1_{(\varphi,\Gamma)}(D_0,D_0)$ , where we put  $D_0 := D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta^{-1}) \hookrightarrow \Delta$ . Consider the following morphisms

$$\operatorname{Ext}^{1}_{(\varphi,\Gamma)}(D_{0}, D_{0}) \longrightarrow \operatorname{Ext}^{1}_{(\varphi,\Gamma)}(D_{0}, \Delta),$$
(4.9)

$$\operatorname{Ext}^{1}_{(\varphi,\Gamma)}(\Delta,\Delta) \longrightarrow \operatorname{Ext}^{1}_{(\varphi,\Gamma)}(D_{0},\Delta).$$
(4.10)

For a  $(\varphi, \Gamma)$ -module D' over  $\mathcal{R}_{E,L}$ , denote by  $W_{\mathrm{dR}^+}(D')$  the  $B^+_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} E$ -representation of  $\mathrm{Gal}_L$ associated to D' (see for example [5, Prop. 2.2.6(2)]). For  $\tau \in \Sigma_L$ , let  $B^+_{\mathrm{dR},\tau,E} := B^+_{\mathrm{dR}} \otimes_{L,\tau} E$ . We have

$$W_{\mathrm{dR}}^+(\Delta \otimes_{\mathcal{R}_{E,L}} D_0^{\vee})/W_{\mathrm{dR}}^+(\Delta \otimes_{\mathcal{R}_{E,L}} \Delta^{\vee}) \cong \bigoplus_{\tau \in \Sigma_L} \bigoplus_{i=1}^{\kappa} (t^{-h_{i,\tau}} B_{\mathrm{dR},\tau,E}^+/B_{\mathrm{dR},\tau,E}^+)^{\oplus k}.$$

Using [15, Lemma 5.1.1] (which easily generalizes to finite extensions of  $\mathbb{Q}_p$ ), we get

$$H^{0}_{(\varphi,\Gamma)}\Big(\big(\Delta \otimes_{\mathcal{R}_{E,L}} D^{\vee}_{0}\big)/\big(\Delta \otimes_{\mathcal{R}_{E,L}} \Delta^{\vee}\big)\Big) = 0$$

and we deduce that the morphism (4.10) is injective. Consider then

$$\operatorname{Hom}(L^{\times}, E) \longrightarrow \operatorname{Ext}^{1}_{(\varphi, \Gamma)}(\Delta, \Delta) \longrightarrow \operatorname{Ext}^{1}_{(\varphi, \Gamma)}(D_{0}, \Delta)$$

where the first map sends  $\psi \in \operatorname{Hom}(L^{\times}, E)$  to  $\Delta \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E[\varepsilon]/\varepsilon^2}(1 + \psi\varepsilon)$ . We denote by V the image of the composition and by  $[D_{\psi}]$  the element in V associated to  $\psi$ . Then it is not difficult to see that  $[D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E[\varepsilon]/\varepsilon^2}(\delta^{-1}(1 + \psi\varepsilon))]$  is sent to  $[D_{\psi}]$  via (4.9) (up to non-zero scalars). This implies V is contained in the image of (4.9). By definition,  $[D_{E[\varepsilon]/\varepsilon^2}] \in F_D^0(E[\varepsilon]/\varepsilon^2)$  if and only if  $[D_{E[\varepsilon]/\varepsilon^2} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta^{-1})]$  lies in the preimage of V via (4.9). We compute the dimension of the kernel of (4.9). We have

$$W_{\mathrm{dR}}^+(\Delta \otimes_{\mathcal{R}_{E,L}} D_0^{\vee})/W_{\mathrm{dR}}^+(D_0 \otimes_{\mathcal{R}_{E,L}} D_0^{\vee}) \cong \bigoplus_{\tau \in \Sigma_L} \bigoplus_{i=1}^k \bigoplus_{j=1}^k t^{-h_{i,\tau}} B_{\mathrm{dR},\tau,E}^+/t^{h_{j,\tau}-h_{i,\tau}}$$

By [15, Lemma 5.1.1], we deduce then

$$\dim_E H^0_{(\varphi,\Gamma)} \left( (\Delta \otimes_{\mathcal{R}_{E,L}} D_0^{\vee}) / (D_0 \otimes_{\mathcal{R}_{E,L}} D_0^{\vee}) \right) = \dim_E H^0 \left( \operatorname{Gal}_L, W^+_{\mathrm{dR}} (\Delta \otimes_{\mathcal{R}_{E,L}} D_0^{\vee}) / W^+_{\mathrm{dR}} (D_0 \otimes_{\mathcal{R}_{E,L}} D_0^{\vee}) \right) = \sum_{\tau \in \Sigma_L} \sum_{h_{j,\tau} > h_{i,\tau}} 1 = \sum_{\tau \in \Sigma_L} \frac{k(k-1)}{2} = \frac{k(k-1)}{2} [L : \mathbb{Q}_p].$$

Using  $\operatorname{Ext}^{0}_{(\varphi,\Gamma)}(D_{0}, D_{0}) \cong \operatorname{Ext}^{0}_{(\varphi,\Gamma)}(D_{0}, \Delta)$ , we see that the kernel of (4.9) is isomorphic to the above vector space and hence has dimension  $\frac{k(k-1)}{2}[L:\mathbb{Q}_{p}]$  over E. As  $\dim_{E} V = [L:\mathbb{Q}_{p}] + 1$ , the proposition follows.

At last, we discuss some relations between  $F_D^0$  and de Rham deformations. Denote by  $F_D^{\text{tw,dR}}$ the functor  $\operatorname{Art}(E) \to \{\text{sets}\}$  sending A to the isomorphism class of  $(D_A, \pi_A, \chi_A)$  where  $(D_A, \pi_A) \in F_D(A), \chi_A : \mathcal{O}_L^{\times} \to A^{\times}$  such that  $\chi_A \equiv 1 \pmod{\mathfrak{m}_A}$  and  $D_A \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta^{-1}\chi_{A,\varpi_L})$  is de Rham. As D is irreducible and de Rham,  $D \cong D_{\text{rig}}(V) \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\psi)$  for a certain de Rham  $\operatorname{Gal}_{L^-}$ representation V and a smooth character  $\psi$  of  $L^{\times}$ . Let  $R_D^{\mathrm{dR}}$  be the universal deformation ring of de Rham deformations of D on  $\operatorname{Art}(E)$ , and  $R_V^{\mathrm{dR}}$  the universal deformation ring of de Rham deformations of V on  $\operatorname{Art}(E)$ . Recall that  $R_V^{\mathrm{dR}}$  is isomorphic to the completion at V of the generic fibre of the universal potentially semi-stable deformation ring of the modulo p reduction of (a lattice in) V, cf. [62, § 2.3]. The functor  $D_{\mathrm{rig}}(-) \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\psi)$  induces then an isomorphism  $R_V^{\mathrm{dR}} \xrightarrow{\sim} R_D^{\mathrm{dR}}$ . Let  $(\widehat{\mathcal{O}_L^{\times}})_1$  be the completion of  $\widehat{\mathcal{O}_L^{\times}}$  at the trivial character. One directly checks that the E-formal scheme  $R_D^{\mathrm{dR}} \widehat{\otimes}_E(\widehat{\mathcal{O}_L^{\times}})_1$  (pro-)represents the functor  $F_D^{\mathrm{tw,dR}}$ . In particular, using [63, Thm. 3.3.8], we see that  $F_D^{\mathrm{tw,dR}}$  is formally smooth of dimension  $1 + [L: \mathbb{Q}_p](1 + \frac{k(k-1)}{2})$ .

When  $(D_A, \pi_A, \delta_A) \in F_D^0(A)$ , we have that  $D_A \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta_A^{-1})$  is de Rham, and thus  $(D_A, \pi_A, \chi_A) \in F_D^{\mathrm{tw,dR}}(A)$  where  $\chi_A := (\delta_A^{-1}\delta)|_{\mathcal{O}_L^{\times}}$ .

**Lemma 4.1.5.** The morphism  $F_D^0 \to F_D^{\text{tw,dR}}$ ,  $(D_A, \pi_A, \delta_A) \mapsto (D_A, \pi_A, \chi_A)$  is an isomorphism.

*Proof.* As both functors are formally smooth of dimension  $1 + [L:\mathbb{Q}_p](1 + \frac{k(k-1)}{2})$ , we only need to show  $F_D^0(E[\varepsilon]/\varepsilon^2) \to F_D^{\text{tw,dR}}(E[\varepsilon]/\varepsilon^2)$  is injective. But this is clear.

#### **4.1.2** Deformations of type $\Omega$

In this section, we study the universal deformation functor for certain paraboline deformations of  $(\varphi, \Gamma)$ -modules which admit an  $\Omega$ -filtration (where  $\Omega$  is a cuspidal Bernstein component as in § 3.1.1). In particular, we show that, under a genericness assumption, this functor is prorepresentable and formally smooth.

Let  $r \in \mathbb{Z}_{\geq 1}$ . For  $1 \leq i \leq r$ , let  $n_i \in \mathbb{Z}_{\geq 1}$  with  $\sum_{i=1}^r n_i = n$ . For  $1 \leq i \leq r$ , let  $\Omega_i$  be a cuspidal type for  $\operatorname{GL}_{n_i}(L)$  and  $\mathcal{Z}_{\Omega_i}$  the associated Bernstein centre over E. Recall that for each E-point  $x_i$  of Spec  $\mathcal{Z}_{\Omega_i}$ , we have a smooth irreducible cuspidal representation  $\pi_{x_i}$  of  $\operatorname{GL}_{n_i}(L)$  over E, an F-semi-simple Weil-Deligne representation  $\mathbf{r}_{x_i} := \operatorname{rec}(\pi_{x_i})$ , and a  $(\varphi, \Gamma)$ -module  $\Delta_{x_i}$  of rank  $n_i$  over  $\mathcal{R}_{E,L}$ , de Rham of constant Hodge-Tate weight 0 (see § 2.1). Let  $P \subseteq \operatorname{GL}_n$  be the parabolic subgroup as in (2.2). We let  $\Omega := (\Omega_i)_{i=1,\ldots,r}$  and  $\mathcal{Z}_{\Omega} := \otimes_{i=1}^r \mathcal{Z}_{\Omega_i}$ . We let  $\mathcal{Z}_{0,L} := Z_{L_P}(\mathcal{O}_L)$  (to be consistent with the notation in § 3).

In this paragraph we fix a  $(\varphi, \Gamma)$ -module D of rank n over  $\mathcal{R}_{E,L}$ .

**Definition 4.1.6.** (1) We say that D admits an  $\Omega$ -filtration  $\mathscr{F}$  if D admits an increasing filtration by  $(\varphi, \Gamma)$ -submodules  $0 = \operatorname{Fil}_0 D \subsetneq \operatorname{Fil}_1 D \subsetneq \cdots \subsetneq \operatorname{Fil}_r D = D$  such that, for  $i = 1, \ldots, r$ :

- $\operatorname{gr}_i D$  is a  $(\varphi, \Gamma)$ -module of rank  $n_i$ ;
- there exist an *E*-point  $x_i \in \operatorname{Spec} \mathcal{Z}_{\Omega_i}$  and a continuous character  $\delta_i : L^{\times} \to E^{\times}$  such that one has an embedding  $\operatorname{gr}_i D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i^{-1}) \hookrightarrow \Delta_{x_i}$ .

(2) Let  $\mathscr{F}, \underline{x} = (x_i), \ \delta = \boxtimes_{i=1}^r \delta_i$  be as in (1), we call the corresponding point  $(\underline{x}, \delta)$  in  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{Z_{L_P}(L)}$  a parameter of the  $\Omega$ -filtration  $\mathscr{F}$  if, for each  $\tau \in \Sigma_L$ , 0 is a  $\tau$ -Hodge-Tate weight (hence is the minimal  $\tau$ -Hodge-Tate weight) of  $\operatorname{gr}_i D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i^{-1})$ .

(3) Let  $\mathscr{F}$  be as in (1). We call  $(\underline{x}, \chi = \boxtimes_{i=1}^{r} \chi_{i}) \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}}$  a parameter of the  $\Omega$ -filtration  $\mathscr{F}$  in  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}}$  if  $(\underline{x}, \chi_{\varpi_{L}} = \boxtimes_{i=1}^{r} \chi_{i, \varpi_{L}})$  is a parameter of  $\mathscr{F}$  in  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}}$  if  $(\underline{x}, \chi_{\varpi_{L}} = \boxtimes_{i=1}^{r} \chi_{i, \varpi_{L}})$  is a parameter of  $\mathscr{F}$  in  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{\Omega,L}}$ .

**Remark 4.1.7.** (1) Let  $\mathscr{F}$ ,  $\underline{x}$ ,  $\delta$  be as in Definition 4.1.6 (1). We can twist each  $\delta_i$  by a certain algebraic character of  $L^{\times}$  so that  $(\underline{x}, \delta)$  is parameter of  $\mathscr{F}$ .

(2) For convenience, we may use these two kinds of parameters depending on the situation. Note that the parameters of  $\mathscr{F}$  (either in  $(\operatorname{Spec} \mathbb{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathbb{Z}_{L_P}(L)}$  or in  $(\operatorname{Spec} \mathbb{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathbb{Z}_{0,L}})$  are in general not unique (see Lemma 4.1.9 below).

**Example 4.1.8.** (1) By Theorem 3.2.16 (using the notation there), for any point  $x \in \mathcal{E}_{\Omega,\lambda}(U^p)$ ,  $D_{\mathrm{rig}}(\rho_{x,\tilde{v}})$  admits an  $\Omega_{\tilde{v}}$ -filtration. This is our main motivation to study  $(\varphi, \Gamma)$ -modules with  $\Omega$ -filtrations.

(2) Let  $\rho$  be as in § 2.3 and use the notation of loc. cit. Let  $\underline{x} \in (\operatorname{Spec} Z_{\Omega})^{\operatorname{rig}}$  be the point such that  $\Delta_{x_i} \cong \operatorname{gr}_i^{\mathscr{F}} \Delta$  for  $i = 1, \ldots, r$ . Then by (2.4) (and comparing the Hodge-Tate weights), we see that  $\mathscr{F}$  in (2.3) is an  $\Omega$ -filtration of parameter ( $\underline{x}, \delta = \boxtimes_{i=1}^r \delta_i := \boxtimes_{i=1}^r z^{w_{\mathscr{F}}(h)_{s_i}}$ )  $\in (\operatorname{Spec} Z_{\Omega})^{\operatorname{rig}} \times \widehat{Z_{L_P}(L)}$ . Let  $\underline{x}' = (x'_i)$  be such that  $\Delta_{x'_i} \cong \Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L} (\operatorname{unr}(\varpi_L^{w_{\mathscr{F}}(h)_{s_i}}))$ , then  $(\underline{x}', \delta^0 = \delta|_{\mathcal{Z}_{0,L}}) \in (\operatorname{Spec} Z_{\Omega})^{\operatorname{rig}} \times \widehat{Z_{0,L}}$  is a parameter of  $\mathscr{F}$ .

**Lemma 4.1.9.** Let  $\mathscr{F}$  be an  $\Omega$ -filtration of D.

(1) Let  $(\underline{x}, \delta) \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{Z_{L_P}(L)}$  be a parameter of  $\mathscr{F}$ , then all parameters of  $\mathscr{F}$  in  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{Z_{L_P}(L)}$  are of the form  $(\underline{x}', \delta')$  such that, for  $i = 1, \dots, r$ ,  $r_{x'_i} \cong r_{x_i} \otimes_E \operatorname{unr}(\alpha_i)$  and  $\delta'_i = \delta_i \operatorname{unr}(\alpha_i^{-1})\eta_i$  for some  $\alpha_i \in \overline{E}^{\times}$  and  $\eta_i \in \mu_{\Omega_i}$ .

(2) Let  $(\underline{x}, \chi) \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}}$  be a parameter of  $\mathscr{F}$ , then all parameters of  $\mathscr{F}$  in  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}}$  are of the form  $(\underline{x}', \chi')$  such that, for  $i = 1, \cdots, r$ ,  $r_{x'_i} \cong r_{x_i} \otimes_E \operatorname{unr}(\eta_i(\varpi_L))$  and  $\chi'_i = \chi_i \eta_i|_{\mathcal{O}_r^{\times}}$  for some  $\eta_i \in \mu_{\Omega_i}$ .

*Proof.* (2) is an easy consequence of (1). We prove (1). Let  $(\underline{x}', \delta') \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}}$  be another parameter of  $\mathscr{F}$ . By definition, we have injections

$$\operatorname{gr}_i D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i^{-1}) \longrightarrow \Delta_{x_i}$$

$$(4.11)$$

$$\operatorname{gr}_{i} D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}((\delta_{i}')^{-1}) \quad \longleftrightarrow \quad \Delta_{x_{i}'} \cong \Delta_{x_{i}} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\operatorname{unr}(\alpha_{i})).$$
(4.12)

From (4.12), we deduce

$$\operatorname{gr}_{i} D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}((\delta_{i}')^{-1}\operatorname{unr}(\alpha_{i})) \cong \operatorname{gr}_{i} D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_{i}^{-1}) \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_{i}(\delta_{i}')^{-1}\operatorname{unr}(\alpha_{i}^{-1})) \hookrightarrow \Delta_{x_{i}}.$$

Since both  $\operatorname{gr}_i D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i^{-1})$  and  $\operatorname{gr}_i D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}((\delta_i')^{-1}\operatorname{unr}(\alpha_i))$  are de Rham and have 0 as the minimal  $\tau$ -Hodge-Tate weight for all  $\tau \in \Sigma_L$ , we deduce  $\delta_i(\delta_i')^{-1}$  is smooth. Using (4.2) (applied to (4.11) and (4.12)), we see that there exists  $N \in \mathbb{Z}_{\geq 0}$  sufficiently large such that

$$\Delta_{x_i} \hookrightarrow t^{-N} \operatorname{gr}_i D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i^{-1}) \hookrightarrow t^{-N} \Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i^{-1}\delta_i'\operatorname{unr}(\alpha_i)).$$

Using [6, Thm. A] and comparing the Hodge-Tate weights, we deduce  $\Delta_{x_i} \cong \Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i^{-1}\delta_i'\operatorname{unr}(\alpha_i))$ , hence  $\mathbf{r}_{x_i} \cong \mathbf{r}_{x_i} \otimes_E (\delta_i^{-1}\delta_i'\operatorname{unr}(\alpha_i))$ , implying  $\delta_i^{-1}\delta_i'\operatorname{unr}(\alpha_i) \in \mu_{\Omega_i}$ . This concludes the proof.

**Remark 4.1.10.** In particular,  $\mathscr{F}$  only has finitely many parameters in Spec  $\mathcal{Z}_{\Omega} \times \widehat{\mathcal{Z}_{0,L}}$ .

Let  $(\underline{x}, \delta) \in \operatorname{Spec} \mathcal{Z}_{\Omega} \times \widehat{Z_{L_P}(L)}$  be a parameter of the  $\Omega$ -filtration  $\mathscr{F}$ . We call  $\mathscr{F}$  generic if the following condition is satisfied:

for 
$$i \neq j$$
, if  $\mathbf{r}_{x_j} \cong \mathbf{r}_{x_i} \otimes_E \eta$  for some smooth character  $\eta$  of  $L^{\times}$ ,  
then  $\delta_i \delta_j^{-1} \eta \neq z^{-\mathbf{k}}$  and  $\delta_i \delta_j^{-1} \eta \neq \operatorname{unr}(q_L^{-1}) z^{\mathbf{k}}$  for any  $\mathbf{k} \in \mathbb{Z}_{>0}^{|\Sigma_L|}$ . (4.13)

By Lemma 4.1.9, this definition is independent of the choice of the parameter of  $\mathscr{F}$ .

**Example 4.1.11.** Let D be as in Example 4.1.8 (2), one directly checks that the  $\Omega$ -filtration  $\mathscr{F}$  is generic if  $\rho$  is generic in the sense of § 2.3.

**Lemma 4.1.12.** Assume that the  $\Omega$ -filtration  $\mathscr{F}$  on D is generic, then we have

$$\operatorname{Hom}_{(\varphi,\Gamma)}(\operatorname{gr}^{i} D, \operatorname{gr}^{j} D) = \operatorname{Ext}_{(\varphi,\Gamma)}^{2}(\operatorname{gr}^{i} D, \operatorname{gr}^{j} D) = 0$$

$$(4.14)$$

for  $i \neq j, i, j \in \{1, ..., r\}$ .

*Proof.* We prove the statement for Hom, the proof for  $\operatorname{Ext}^2$  being similar using  $\operatorname{Ext}^2_{(\varphi,\Gamma)}(\operatorname{gr}^i D, \operatorname{gr}^j D) = H^2_{(\varphi,\Gamma)}((\operatorname{gr}^i D)^{\vee} \otimes_{\mathcal{R}_{E,L}} \operatorname{gr}^j D)$  and Tate duality ([66, Thm. 1.2 (2)]). We

let  $(\underline{x}, \delta) \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{Z_{L_P}(L)}$  be a parameter of  $\mathscr{F}$ . Suppose we have a non-zero morphism  $f : \operatorname{gr}^i D \to \operatorname{gr}^j D$  for  $i \neq j$ . Let  $N \in \mathbb{Z}_{\geq 0}$  be sufficiently large, then f induces a non-zero hence injective morphism (see (4.2))

$$t^{N}\Delta_{x_{i}} \to \operatorname{gr}^{i} D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_{i}^{-1}) \to \operatorname{gr}^{j} D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_{i}^{-1}) \to \Delta_{x_{j}} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_{i}^{-1}\delta_{j}).$$
(4.15)

Consider the induced injective morphism  $\Delta_{x_i} \to \Delta_{x_j} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i^{-1}\delta_j z^{-\mathbf{N}})$  where  $\mathbf{N} := (N)_{\tau \in \Sigma_L}$ . Since both the source and target are irreducible, the morphism (and hence (4.15)) becomes an isomorphism inverting t. Since the left hand side of (4.15) is de Rham, so is  $\Delta_{x_j} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i^{-1}\delta_j)$ . This implies (e.g. by considering the Sen weights) that  $\delta_i^{-1}\delta_j$  is locally algebraic, say of the form  $\eta z^{\mathbf{k}}$  for a smooth character  $\eta$  of  $L^{\times}$  and some  $\mathbf{k} \in \mathbb{Z}^{|\Sigma_L|}$ . By using [6, Thm. A] and comparing the Hodge-Tate weights, we see that (4.15) induces  $\Delta_{x_i} \xrightarrow{\sim} \Delta_{x_j} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\eta)$  (so  $\mathbf{r}_{x_i} \cong \mathbf{r}_{x_j} \otimes_E \eta$ ).

We next show  $\mathbf{k} \in \mathbb{Z}_{\leq 0}^{|\Sigma_L|}$ . By definition, 0 is the minimal  $\tau$ -Hodge-Tate weight of  $\operatorname{gr}^i D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i^{-1})$  for all  $\tau$ , while  $\operatorname{wt}(\delta_i^{-1}\delta_j)_{\tau}$  is the minimal  $\tau$ -Hodge-Tate weight of

$$\operatorname{gr}^{j} D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_{i}^{-1}) \cong \operatorname{gr}^{j} D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_{j}^{-1}) \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_{j}\delta_{i}^{-1}).$$

The second (injective) morphism in (4.15) then implies  $\operatorname{wt}(\delta_i^{-1}\delta_j)_{\tau} \leq 0$  for all  $\tau$ , hence  $\mathbf{k} \in \mathbb{Z}_{\leq 0}^{|\Sigma_L|}$ . But  $\delta_i^{-1}\delta_j\eta^{-1} = z^{\mathbf{k}}$  with  $\mathbf{k} \in \mathbb{Z}_{\leq 0}^{|\Sigma_L|}$  contradicts the genericity (4.13) of  $\mathscr{F}$ .

**Corollary 4.1.13.** Assume that D admits a generic  $\Omega$ -filtration  $\mathscr{F}$ , and let  $(\underline{x}, \delta) \in \operatorname{Spec} \mathcal{Z}_{\Omega} \times \widehat{Z_{L_P}(L)}$  be a parameter of  $\mathscr{F}$ . Then D has a unique  $\Omega$ -filtration of parameter  $(\underline{x}, \delta)$ .

*Proof.* Suppose D has two  $\Omega$ -filtrations Fil<sub>•</sub>, Fil<sub>•</sub> of parameter  $(\underline{x}, \delta)$ . By Lemma 4.1.12 and a standard dévissage, we easily deduce  $\operatorname{Hom}_{(\varphi,\Gamma)}(\operatorname{Fil}_{r-1} D, \operatorname{gr}'_r D) = 0$  and hence the injection  $\operatorname{Fil}_{r-1} D \hookrightarrow D$  induces an isomorphism  $\operatorname{Fil}_{r-1} D \xrightarrow{\sim} \operatorname{Fil}'_{r-1} D$ . We go on replacing D by  $\operatorname{Fil}_{r-1} D =$  $\operatorname{Fil}'_{r-1} D$ , and we see that  $\operatorname{Fil}_i D \hookrightarrow D$  induces an isomorphism  $\operatorname{Fil}_i D \xrightarrow{\sim} \operatorname{Fil}'_i D$  for  $i = 1, \ldots, r$ .  $\Box$ 

Assume that D admits a (uniquely determined) generic  $\Omega$ -filtration  $\mathscr{F}$  and let  $F_{D,\mathscr{F}}$  denote the functor  $\operatorname{Art}(E) \to {\operatorname{Sets}}$  which sends  $A \in \operatorname{Art}(E)$  to the set of isomorphism classes

$$F_{D,\mathscr{F}}(A) = \{(D_A, \pi_A, \mathscr{F}_A)\} / \sim$$

where (the isomorphisms being defined in an obvious way)

- (1)  $D_A$  is a  $(\varphi, \Gamma)$ -module of rank *n* over  $\mathcal{R}_{A,L}$  with  $\pi_A : D_A \otimes_A E \xrightarrow{\sim} D$ ;
- (2)  $\mathscr{F}_A = \operatorname{Fil}_{\bullet} D_A$  is an increasing filtration by  $(\varphi, \Gamma)$ -submodules over  $\mathcal{R}_{A,L}$  on  $D_A$  such that  $\operatorname{Fil}_i D_A, i \in \{0, \ldots, r\}$  is a direct summand of  $D_A$  as  $\mathcal{R}_{A,L}$ -modules and  $\pi_A(\operatorname{Fil}_i D_A) = \operatorname{Fil}_i D$ .

**Lemma 4.1.14.** Assume  $\operatorname{Hom}_{(\varphi,\Gamma)}(\operatorname{gr}_i D, \operatorname{gr}_j D) = 0$  for  $i \neq j$ . Then  $F_{D,\mathscr{F}}$  is a subfunctor of  $F_D$ .

*Proof.* By the assumption and a dévissage similar to the one for Lemma 4.1.13, we easily deduce that, if Fil<sub>•</sub>  $D_A$  and Fil'<sub>•</sub>  $D_A$  are two filtrations on  $(D_A, \pi_A) \in F_D(A)$ , then they have to be equal. The lemma follows.

Denote by  $\operatorname{End}_{\mathscr{F}}(D) := \{f \in \operatorname{End}_{\mathcal{R}_{E,L}}(D) \mid f(\operatorname{Fil}_i D) \subset \operatorname{Fil}_i D, \forall i\}$ , which is equipped with a natural  $(\varphi, \Gamma)$ -action as in the discussion below [29, Rem. 3.5]. Recall the following result:

**Proposition 4.1.15** ([29, Prop. 3.6 (2), (3)]). (1) There is a natural isomorphism of E-vector spaces  $F_{D,\mathscr{F}}(E[\varepsilon/\varepsilon^2]) \xrightarrow{\sim} H^1_{(\varphi,\Gamma)}(\operatorname{End}_{\mathscr{F}}(D))$  and

$$\dim_E F_{D,\mathscr{F}}(E[\varepsilon/\varepsilon^2]) = \dim_E H^0_{(\varphi,\Gamma)}(\operatorname{End}_{\mathscr{F}}(D)) + \dim_E H^2_{(\varphi,\Gamma)}(\operatorname{End}_{\mathscr{F}}(D)) + [L:\mathbb{Q}_p] \sum_{i \leq j} n_i n_j.$$

(2) Assume  $H^2_{(\varphi,\Gamma)}(\operatorname{End}_{\mathscr{F}}(D)) = 0$ , then the functor  $F_{D,\mathscr{F}}$  is formally smooth of dimension  $\dim_E H^0_{(\varphi,\Gamma)}(\operatorname{End}_{\mathscr{F}}(D)) + [L:\mathbb{Q}_p] \sum_{i\leq j} n_i n_j.$ 

We have a natural morphism

$$F_{D,\mathscr{F}} \longrightarrow \prod_{i} F_{\operatorname{gr}_i D}$$

sending  $(D_A, \pi_A, \mathscr{F}_A)$  to  $(\operatorname{gr}_i D_A, \pi_A|_{\operatorname{gr}_i D_A})_{i=1,\dots,r}$ .

**Proposition 4.1.16** ([29, Prop. 3.7]). Assume  $H^2_{(\varphi,\Gamma)}(\operatorname{Hom}_{\mathcal{R}_{E,L}}(D/\operatorname{Fil}_i D, \operatorname{gr}_i D)) = 0$  for all i, then the morphism  $F_{D,\mathscr{F}} \to \prod_i F_{\operatorname{gr}_i D}$  is formally smooth.

Let  $(\underline{x}, \delta) \in \operatorname{Spec} \mathcal{Z}_{\Omega} \times \widehat{Z_{L_P}(L)}$  be a parameter of  $\mathscr{F}$ , and let  $F_{D,\mathscr{F}}^0$  be the functor  $\operatorname{Art}(E) \to {\operatorname{Sets}}$  sending  $A \in \operatorname{Art}(E)$  to the set of isomorphism classes

$$F_{D,\mathscr{F}}^0(A) = \{(D_A, \pi_A, \mathscr{F}_A, \delta_A)\}/\sim$$

where (the isomorphisms being again defined in an obvious way)

- (1)  $(D_A, \pi_A, \mathscr{F}_A) \in F_{D,\mathscr{F}}(A);$
- (2)  $\delta_A = (\delta_{A,i})_{i=1,\dots,r}$  where  $\delta_{A,i} : L^{\times} \to A^{\times}$  is a continuous character such that  $\delta_{A,i} \equiv \delta_i \pmod{\mathfrak{m}_A}$  and there exists an injection of  $(\varphi, \Gamma)$ -modules over  $\mathcal{R}_{A,L}$ :

$$\operatorname{gr}_i D_A \hookrightarrow \Delta_i \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta_{A,i}).$$

By definition, we have  $F_{D,\mathscr{F}}^0 \cong F_{D,\mathscr{F}} \times_{\prod_i F_{\operatorname{gr}_i D}} \prod_i F_{\operatorname{gr}_i D}^0$ .

**Proposition 4.1.17.** (1) Assume  $\operatorname{Hom}_{(\varphi,\Gamma)}(\operatorname{gr}_i D, \operatorname{gr}_j D) = 0$  for  $i \neq j$ , then the functor  $F_{D,\mathscr{F}}^0$  is a subfunctor of  $F_D$ .

(2) Assume  $H^2_{(\varphi,\Gamma)}(\operatorname{Hom}_{\mathcal{R}_{E,L}}(D/\operatorname{Fil}_i D, \operatorname{gr}_i D)) = 0$  and  $\operatorname{gr}_i D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i^{-1})$  has distinct Hodge-Tate weights for  $i = 1, \ldots, r$ . Then  $F^0_{D,\mathscr{F}}$  is formally smooth of dimension

$$\dim_E H^2_{(\varphi,\Gamma)}(\operatorname{End}_{\mathscr{F}}(D)) + \dim_E H^0_{(\varphi,\Gamma)}(\operatorname{End}_{\mathscr{F}}(D)) + [L:\mathbb{Q}_p]\big(\frac{n(n-1)}{2} + r\big).$$

*Proof.* (1) By Lemma 4.1.2  $F_{D,\mathscr{F}}^0$  is a subfunctor of  $F_{D,\mathscr{F}}$ , and by Lemma 4.1.14  $F_{D,\mathscr{F}}$  is a subfunctor of  $F_D$ .

(2) From Proposition 4.1.16, we deduce that  $F_{D,\mathscr{F}}^0$  is formally smooth over  $\prod_i F_{\mathrm{gr}_i D}^0$  and hence is formally smooth by Proposition 4.1.4. We have then

$$\begin{split} \dim_E F_{D,\mathscr{F}}^0(E[\varepsilon/\varepsilon^2]) &= \dim_E F_{D,\mathscr{F}}(E[\varepsilon/\varepsilon^2]) - \sum_i \dim_E F_{\mathrm{gr}_i \, D}(E[\varepsilon/\varepsilon^2]) + \sum_i \dim_E F_{\mathrm{gr}_i \, D}^0(E[\varepsilon/\varepsilon^2]) \\ &= \dim_E H_{(\varphi,\Gamma)}^0(\mathrm{End}_{\mathscr{F}}(D)) + \dim_E H_{(\varphi,\Gamma)}^2(\mathrm{End}_{\mathscr{F}}(D)) + [L:\mathbb{Q}_p] \Big(\sum_{i < j} n_i n_j \\ &+ \sum_i \Big(\frac{n_i(n_i - 1)}{2} + 1\Big)\Big) \\ &= \dim_E H_{(\varphi,\Gamma)}^0(\mathrm{End}_{\mathscr{F}}(D)) + \dim_E H_{(\varphi,\Gamma)}^2(\mathrm{End}_{\mathscr{F}}(D)) + [L:\mathbb{Q}_p] \Big(\frac{n(n-1)}{2} + r\Big), \end{split}$$

where the second equality follows from Proposition 4.1.15 (1), Proposition 4.1.4 and the standard fact that  $\dim_E F_{\operatorname{gr}_i D}(E[\varepsilon/\varepsilon^2]) = 1 + [L:\mathbb{Q}_p]n_i^2$  (noting that  $\operatorname{gr}_i D$  is irreducible).

Let  $\rho : \operatorname{Gal}_L \to \operatorname{GL}_n(E)$  be a continuous group morphism and V the associated representation of  $\operatorname{Gal}_L$  over E. We let  $F_\rho$  (resp.  $F_V$ ) denote the deformation functor of  $\rho$  (resp. V) over  $\operatorname{Art}(E)$ . So  $F_\rho$  can be viewed as the framed deformation functor of V over  $\operatorname{Art}(E)$ . Assume  $D \cong D_{\operatorname{rig}}(V)$ , then we have  $F_D \cong F_V$ . Let  $F^0_{\rho,\mathscr{F}} := F^0_{D,\mathscr{F}} \times_{F_V} F_\rho$ . Recall that  $F_\rho$  is pro-representable and is formally smooth over  $F_V$  of relative dimension  $n^2 - \dim_E H^0_{(\varphi,\Gamma)}(\operatorname{End}_{\mathcal{R}_{E,L}}(D))$ .

**Corollary 4.1.18.** (1) Assume  $\operatorname{Hom}_{(\varphi,\Gamma)}(\operatorname{gr}_i D, \operatorname{gr}_j D) = 0$  for  $i \neq j$ , then  $F^0_{\rho,\mathscr{F}}$  is a subfunctor of  $F_{\rho}$  and is pro-representable.

(2) Assume  $H^2_{(\varphi,\Gamma)}(\operatorname{Hom}_{\mathcal{R}_{E,L}}(D/\operatorname{Fil}_i D, \operatorname{gr}_i D)) = 0$  and  $\operatorname{gr}_i D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i^{-1})$  has distinct Hodge-Tate weights for  $i = 1, \ldots, r$ . Then  $F^0_{\rho,\mathscr{F}}$  is formally smooth of dimension

$$\dim_E H^2_{(\varphi,\Gamma)}(\operatorname{End}_{\mathscr{F}}(D)) + \dim_E H^0_{(\varphi,\Gamma)}(\operatorname{End}_{\mathscr{F}}(D)) - \dim_E H^0_{(\varphi,\Gamma)}(\operatorname{End}_{\mathcal{R}_{E,L}}(D)) + n^2 + [L:\mathbb{Q}_p] \Big(\frac{n(n-1)}{2} + r\Big). \quad (4.16)$$

*Proof.* By Proposition 4.1.17 (1),  $F_{\rho,\mathscr{F}}^0$  is a subfunctor of  $F_{\rho}$ . By Lemma 4.1.14,  $F_{\rho,\mathscr{F}} := F_{D,\mathscr{F}} \times_{F_D}$  $F_{\rho}$  is a subfunctor of (the pro-representable functor)  $F_{\rho}$ . Together with [29, Prop. 3.4], we can deduce that  $F_{\rho,\mathscr{F}}$  is pro-representable (noting that we only need to show that

$$F_{\rho,\mathscr{F}}(A' \times_A A'') \longrightarrow F_{\rho,\mathscr{F}}(A') \times_{F_{\rho,\mathscr{F}}(A)} F_{\rho,\mathscr{F}}(A'')$$

is surjective whenever  $A'' \to A$  is surjective). It then follows from Proposition 4.1.3 that  $F^0_{\rho,\mathscr{F}}$ is pro-representable. By base change,  $F^0_{\rho,\mathscr{F}}$  is formally smooth over  $F^0_{D,\mathscr{F}}$  of relative dimension  $n^2 - \dim_E H^0_{(\varphi,\Gamma)}(\operatorname{End}_{\mathcal{R}_{E,L}}(D))$ . Together with Proposition 4.1.17 (2), (2) follows.

**Remark 4.1.19.** Assume that  $(D, \mathscr{F})$  satisfies the properties in (4.14) and that  $\operatorname{gr}_i D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i^{-1})$  has distinct Hodge-Tate weights for  $i = 1, \ldots, r$ . Then using a dévissage, one easily deduces that the assumptions in Corollary 4.1.18 (1) and (2) are satisfied, and that the terms in the first line of (4.16) are all zero. So in this case  $F^0_{\rho,\mathscr{F}}$  is (pro-representable) formally smooth of dimension  $n^2 + [L:\mathbb{Q}_p](\frac{n(n-1)}{2} + r)$ .

# 4.2 Bernstein paraboline varieties

By generalizing results in [17, § 2.2] on the trianguline variety, we construct and study a certain local Galois deformation space of a modulo p Galois representation which consists of Galois representations admitting an  $\Omega$ -filtration.

The following lemma follows easily from [60, Thm.]. We formulate it here since we will frequently use it.

**Lemma 4.2.1.** Let X be a reduced rigid analytic space over E and D a  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{X,L}$ . Assume that, for i = 0, 1, 2, there exists  $d_i \in \mathbb{Z}_{\geq 0}$  such that for all  $x \in X$ ,  $\dim_{k(x)} H^i_{(\varphi,\Gamma)}(D_x) = d_i$ where  $D_x := x^*D$ . Then  $H^i_{(\varphi,\Gamma)}(D)$  is a locally free sheaf of rank  $d_i$  over X and for any morphism  $f: Y \to X$  of rigid spaces, we have  $H^i_{(\varphi,\Gamma)}(f^*D) \cong H^i_{(\varphi,\Gamma)}(D) \otimes_{\mathcal{O}_X} \mathcal{O}_Y$ .

Proof. By [60, Thm. 4.4.5 (1)] and [60, Thm.],  $H^i_{(\varphi,\Gamma)}(D)$  is a coherent sheaf over X, and there exists a complex  $[C^0 \xrightarrow{d^0} C^1 \xrightarrow{d^1} C^2]$  of locally free sheaves of finite type over X such that, for  $f: Y \to X, H^i_{(\varphi,\Gamma)}(f^*D)$  is isomorphic to the *i*-th cohomology of the complex

$$[C^0 \otimes_{\mathcal{O}_X} \mathcal{O}_Y \to C^1 \otimes_{\mathcal{O}_X} \mathcal{O}_Y \to C^2 \otimes_{\mathcal{O}_X} \mathcal{O}_Y].$$

In particular, we have  $H^2_{(\varphi,\Gamma)}(D) \otimes_{\mathcal{O}_X} \mathcal{O}_Y \cong H^2_{(\varphi,\Gamma)}(f^*D)$ . Applying this to points of X, we deduce that  $H^2_{(\varphi,\Gamma)}(D) \otimes_{\mathcal{O}_X} k(x) \cong H^2_{(\varphi,\Gamma)}(D_x)$  has constant dimension  $d_2$  for all  $x \in X$ . Since X is reduced (and  $H^2_{(\varphi,\Gamma)}(D)$  is coherent), this implies that  $H^2_{(\varphi,\Gamma)}(D)$  is locally free of rank  $d_2$ . We deduce that Ker $(d^1)$  is also locally free of finite type (as the kernel of a surjection between two locally free  $\mathcal{O}_X$ modules of finite type is locally free of finite type), and thus  $H^1_{(\varphi,\Gamma)}(D) \otimes_{\mathcal{O}_X} \mathcal{O}_Y \cong H^1_{(\varphi,\Gamma)}(f^*D)$ . Repeating the above argument, we obtain that  $H^1_{(\varphi,\Gamma)}(D)$  is locally free of rank  $d_1$ , that  $\operatorname{Im}(d^0)$ is locally free, and then again that  $H^0_{(\varphi,\Gamma)}(D)$  is locally free of rank  $d_0$  and  $H^0_{(\varphi,\Gamma)}(D) \otimes_{\mathcal{O}_X} \mathcal{O}_Y \cong$  $H^0_{(\varphi,\Gamma)}(f^*D)$ .

We keep the setting of the beginning of § 4.1.2 and fix  $\mathbf{h} = (\mathbf{h}_i)_{i=1,...,n} = (h_{i,\tau})_{\substack{i=1,...,n \\ \tau \in \Sigma_L}} \in \mathbb{Z}^{\oplus n|\Sigma_L|}$ strictly *P*-dominant. Let  $(\mathbf{r}_i) \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{an}}$ , we say that a character  $\delta$  of  $Z_{L_P}(L)$  is generic for  $(\Omega, \mathbf{h})$  if the following condition is satisfied:

if there exist  $i \neq j$  such that  $\mathbf{r}_j = \mathbf{r}_i \otimes_E \eta$  for some smooth character  $\eta$  of  $L^{\times}$  (noting there are finitely many choices of  $\eta$ ), then  $\delta_i \delta_j^{-1} \eta z^{\mathbf{h}_{s_i} - \mathbf{h}_{s_j}} \neq z^{-\mathbf{k}}$  and  $\delta_i \delta_j^{-1} \eta z^{\mathbf{h}_{s_i} - \mathbf{h}_{s_j}} \neq \operatorname{unr}(q_L^{-1}) z^{\mathbf{k}}$  for all  $\mathbf{k} = (k_{\sigma})_{\sigma \in \Sigma_L} \in \mathbb{Z}_{\geq 0}^{|\Sigma_L|}$ .

The set of such points is Zariski open and Zariski dense in  $\widehat{Z_{L_P}(L)}$ . For each  $\mathbf{r}_i$ , we have a natural finite morphism  $\mathbb{G}_m \to \operatorname{Spec} \mathcal{Z}_{\Omega_i}$ ,  $\alpha \mapsto \mathbf{r}_i \otimes_E \operatorname{unr}(\alpha)$  (note that  $\mathcal{Z}_{\Omega_i} \cong E[x, x^{-1}]^{\mu_{\Omega_i}^{\operatorname{unr}}}$ ). We have and fix an isomorphism (depending on the choice of  $\varpi_L$ )  $\mathbb{G}_m^{\operatorname{rig}} \times \widehat{\mathcal{O}_L^{\times}} \xrightarrow{\sim} \widehat{L^{\times}}$ ,  $(a, \chi) \mapsto \operatorname{unr}(a)\chi_{\varpi_L}$ . We define

$$\mathscr{Z} := (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}},$$

 $\iota_{\mathbf{r}_i}$  as the composition  $\iota_{\mathbf{r}_i}: \widehat{L^{\times}} \xrightarrow{\sim} \mathbb{G}_m^{\mathrm{rig}} \times \widehat{\mathcal{O}_L^{\times}} \to (\operatorname{Spec} \mathcal{Z}_{\Omega_i})^{\mathrm{rig}} \times \widehat{\mathcal{O}_L^{\times}}$ , and  $\iota_{\underline{\mathbf{r}}}$  as the composition:

$$\iota_{\underline{\mathbf{r}}}:\widehat{Z_{L_P}(L)} \xrightarrow{\sim} \prod_{i=1}^r \widehat{L^{\times}} \xrightarrow{\sim} \prod_{i=1}^r (\mathbb{G}_m^{\operatorname{rig}} \times \widehat{\mathcal{O}_L^{\times}}) \xrightarrow{(\iota_{r_i})} \prod_{i=1}^r \left( (\operatorname{Spec} \mathcal{Z}_{\Omega_i})^{\operatorname{rig}} \times \widehat{\mathcal{O}_L^{\times}} \right) \cong \mathscr{Z}.$$

A point of  $\mathscr{Z}$  is called generic if its preimage in  $\widehat{Z_{L_P}(L)}$  is generic for  $(\Omega, \mathbf{h})$ . One can check that this notion is independent of the choice of  $\{\mathbf{r}_i\}$ . Denote by  $\mathscr{Z}^{\text{gen}} \subset \mathscr{Z}$  the set of points that are generic. One can also check that if  $(\underline{x}, \chi) \in \mathscr{Z}^{\text{gen}}$  then  $(\underline{x}, \chi_{\varpi_L})$  satisfies the condition in (4.13). Any affinoid open in  $\mathscr{Z}$  can only have finitely many points that are not generic (since the same holds for  $\widehat{Z_{L_P}(L)}$ ). Hence  $\mathscr{Z}^{\text{gen}}$  is Zariski-open and Zariski-dense in  $\mathscr{Z}$ .

Let  $\overline{\rho}$ :  $\operatorname{Gal}_L \to \operatorname{GL}_n(k_E)$  be a continuous group morphism. Let  $U_{\Omega,\mathbf{h}}(\overline{\rho})$  be the subset of  $(\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}} \times \mathscr{Z}$  of the points  $(\rho, \underline{x}, \chi)$  such that

- (1)  $(\underline{x}, \chi) \in \mathscr{Z}^{\text{gen}};$
- (2)  $D_{\rm rig}(\rho)$  admits an  $\Omega$ -filtration  $\mathscr{F} = \{ {\rm Fil}_i D_{\rm rig}(\rho) \}$  such that

$$\operatorname{gr}_{i} D_{\operatorname{rig}}(\rho) \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x),L}(\chi_{i,\varpi_{L}}^{-1}) \hookrightarrow \Delta_{x_{i}} \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x),L}(z^{\mathbf{h}_{s_{i}}})$$
(4.17)

and the image has Hodge-Tate weights  $(\mathbf{h}_{s_{i-1}+1}, \ldots, \mathbf{h}_{s_i})$ .

We define  $X_{\Omega,\mathbf{h}}(\overline{\rho})$  to be the Zariski-closure of  $U_{\Omega,\mathbf{h}}(\overline{\rho})$  in  $(\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}} \times \mathscr{Z}$ . By definition  $X_{\Omega,\mathbf{h}}(\overline{\rho})$  is reduced and we have a natural morphism

$$\omega: X_{\Omega,\mathbf{h}}(\overline{\rho}) \longrightarrow \mathscr{Z}$$

We define an action of  $\mu_{\Omega} = \{\psi = (\psi_i) : Z_{L_P}(L) \to E^{\times} \mid \psi_i \in \mu_{\Omega_i}\}$  on  $\mathscr{Z}$  such that  $\psi = (\psi_i) \in \mu_{\Omega}$ sends  $((\mathbf{r}_i), (\chi_i))$  to  $((\mathbf{r}_i \otimes_E \operatorname{unr}(\psi_i(\varpi_L))), (\chi_i \psi_i^0))$ . It induces an action of  $\mu_{\Omega}$  on  $(\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}} \times \mathscr{Z}$  by acting trivially on  $(\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}}$ . By Lemma 4.1.9 (2),  $U_{\Omega,\mathbf{h}}(\overline{\rho})$  is preserved by the action of  $\mu_{\Omega}$ . We then deduce that  $X_{\Omega,\mathbf{h}}(\overline{\rho})$  is also preserved by  $\mu_{\Omega}$ :

**Proposition 4.2.2.** A point  $(\rho, (\mathbf{r}_i), (\chi_i)) \in (\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}} \times \mathscr{Z}$  lies in  $X_{\Omega, \mathbf{h}}(\overline{\rho})$  if and only if the point  $(\rho, (\mathbf{r}_i \otimes_E \operatorname{unr}(\psi_i(\varpi_L))), (\chi_i \psi_i^0))$  lies in  $X_{\Omega, \mathbf{h}}(\overline{\rho})$  for all  $\psi = (\psi_i) \in \mu_{\Omega}$ .

Let  $\psi = (\psi_i)$  be a smooth character of  $Z_{L_P}(L)$ , and  $\Omega$  be the Bernstein component  $\{\pi \otimes_E \psi\}_{\pi \in \Omega}$ . Let  $\mathbf{h}' = (\mathbf{h}'_i)_{i=1,\dots,n} = (h'_{i,\tau})_{\substack{i=1,\dots,n \\ \tau \in \Sigma_L}} \in \mathbb{Z}^{\oplus n |\Sigma_L|}$  be strictly *P*-dominant such that  $\mathbf{h}' - \mathbf{h} = \mathfrak{d} \circ \det_{L_P}$  for a weight  $\mathfrak{d} = (\mathfrak{d}_i)_{i=1,\dots,r}$  of  $\mathfrak{z}_{L_P,L}$ . The condition (4.17) is then equivalent to

$$\operatorname{gr}_{i} D_{\operatorname{rig}}(\rho) \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x),L} \left( \chi_{i,\varpi_{L}}^{-1} \delta_{\mathfrak{d}_{i},\varpi_{L}}^{0} \psi_{i,\varpi_{L}}^{0} \right) \\ \longleftrightarrow \Delta_{x_{i}} \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x),L} \left( \psi_{i,\varpi_{L}}^{0} \operatorname{unr}(\delta_{\mathfrak{d}_{i}}^{-1}(\varpi_{L})) \right) \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x),L}(z^{\mathbf{h}'_{s_{i}}}).$$

The isomorphism

$$(\operatorname{Spf} R_{\overline{\rho}}) \times \mathscr{Z} \xrightarrow{\sim} (\operatorname{Spf} R_{\overline{\rho}}) \times \mathscr{Z}, \ \left(\rho, (\mathbf{r}_i), (\chi_i)\right) \mapsto \left(\rho, (\mathbf{r}_i \otimes_E (\psi_{i, \varpi_L}^0(\delta^{\operatorname{unr}}_{\mathfrak{d}_i, \varpi_L})^{-1}), (\chi_i(\delta^0_{\mathfrak{d}_i}\psi_i^0)^{-1})\right)$$

sends bijectively  $U_{\Omega,\mathbf{h}}(\overline{\rho})$  to  $U_{\Omega',\mathbf{h}'}(\overline{\rho})$ . We then deduce (compare with Proposition 3.2.3):

**Proposition 4.2.3.** We have an isomorphism:

$$X_{\Omega,\boldsymbol{h}}(\overline{\rho}) \xrightarrow{\sim} X_{\Omega',\boldsymbol{h}'}(\overline{\rho}), \ \left(\rho,(\boldsymbol{r}_i),(\chi_i)\right) \longmapsto \left(\rho,(\boldsymbol{r}_i \otimes (\psi_i^0)_{\varpi_L}(\delta_{\mathfrak{d}_i,\varpi_L}^{\mathrm{unr}})^{-1}),(\chi_i(\delta_{\mathfrak{d}_i}^0\psi_i^0)^{-1})\right).$$

**Remark 4.2.4.** (1) Assume P = B, hence  $L_P = Z_{L_P} = T$ . Using the isomorphism (where  $\boxtimes_{i=1}^r \pi_{x_i}$  is the smooth character of  $T(L) = L_P(L)$  associated to  $\underline{x}$ ):

$$\iota_{\Omega,h}: \mathscr{Z} \xrightarrow{\sim} \widehat{T(L)}, \ (\underline{x}, \chi) \mapsto (\boxtimes_{i=1}^r \pi_{x_i}) \chi z^h,$$

we view  $X_{\Omega,h}(\overline{\rho})$  as a closed rigid subspace of  $(\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}} \times \widehat{T(L)}$  via the following morphism, that we still denote by  $\iota_{\Omega,h}$ :

$$\iota_{\Omega,h}: X_{\Omega,h}(\overline{\rho}) \longrightarrow (\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}} \times \mathscr{Z} \xrightarrow{\operatorname{id} \times \iota_{\Omega,h}} (\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}} \times \widehat{T(L)}.$$

Such a closed rigid subspace is in fact independent of the choice of  $(\Omega, \mathbf{h})$  by Proposition 4.2.3. By assumption, the injection in (4.17) is actually an isomorphism. We then deduce that  $U_{\Omega,\mathbf{h}}(\overline{\rho}) \subset$  $(\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}} \times \widehat{T(L)}$  coincides with  $U_{\operatorname{tri}}^{\Box}(\overline{r})^{\operatorname{reg}}$  in [17, § 2.2], hence  $X_{\Omega,\mathbf{h}}(\overline{\rho})$  coincides with the trianguline variety  $X_{\operatorname{tri}}^{\Box}(\overline{r})$  of loc. cit.

(2) By definition, for  $(\rho, \underline{x}, \chi) \in U_{\Omega,h}(\overline{\rho})$ ,  $(\underline{x}, \boxtimes_{i=1}^{r}(\chi_{i,\varpi_{L}}z^{\mathbf{h}_{s_{i}}})) \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{L_{P}}(L)}$  is a (generic) parameter of the  $\Omega$ -filtration  $\mathscr{F}$  on  $D_{\operatorname{rig}}(\rho)$ . We will show in Corollary 4.2.6 below that, for any point  $(\rho, \underline{x}, \chi) \in X_{\Omega,h}(\overline{\rho})$ ,  $D_{\operatorname{rig}}(\rho)$  is naturally equipped with an  $\Omega$ -filtration  $\mathscr{F}$ . However,  $(\underline{x}, (\chi_{i,\varpi_{L}}z^{\mathbf{h}_{s_{i}}}))$  is not forcedly in general a parameter of  $\mathscr{F}$ . This phenomenon is closely related to the problem of (local) companion points (see Remark 4.3.10 and Example 4.1.8).

The following theorem, generalizing [17, Thm. 2.6], is the main result of this section.

**Theorem 4.2.5.** (1) The rigid analytic space  $X_{\Omega,h}(\overline{\rho})$  is equidimensional of dimension

$$n^2 + [L:\mathbb{Q}_p]\left(\frac{n(n-1)}{2} + r\right)$$

(2) The set  $U_{\Omega,\mathbf{h}}(\overline{\rho})$  is Zariski-open and Zariski-dense in  $X_{\Omega,\mathbf{h}}(\overline{\rho})$ .

(3) The rigid space  $U_{\Omega,h}(\overline{\rho})$  is smooth over E, and the morphism  $\omega|_{U_{\Omega,h}(\overline{\rho})} : U_{\Omega,h}(\overline{\rho}) \to \mathscr{Z}$  is smooth.

By Theorem 4.2.5, Corollary A.1.2 and Corollary A.1.3 (1) (applied to  $X = X_{\Omega,\mathbf{h}}(\overline{\rho})$ ), we get: **Corollary 4.2.6.** Let  $x = (\rho, \underline{x}, \chi) \in X_{\Omega,\mathbf{h}}(\overline{\rho})$ , then  $D_{\mathrm{rig}}(\rho)$  admits an  $\Omega$ -filtration  $\mathscr{F} = \{\mathrm{Fil}_i D_{\mathrm{rig}}(\rho)\}$  such that, for all  $i = 1, \ldots, r$ ,

$$\operatorname{gr}_i D_{\operatorname{rig}}(\rho) \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x),L}(\chi_{i,\varpi_L}^{-1}) \Big[\frac{1}{t}\Big] \cong \Delta_{x_i} \Big[\frac{1}{t}\Big].$$

In the rest of this section, we prove Theorem 4.2.5 by generalizing the proof of [17, Thm. 2.6]. One difference is that, instead of having a smooth rigid space (the  $S(\bar{r})$  of *loc. cit.*) mapping onto the whole  $U_{\Omega,\mathbf{h}}(\bar{\rho})$  ( $U_{\text{tri}}(\bar{r})^{\text{reg}}$  of *loc. cit.*), we only have smooth rigid spaces mapping onto certain open subspaces of  $U_{\Omega,\mathbf{h}}(\bar{\rho})$  that cover  $U_{\Omega,\mathbf{h}}(\bar{\rho})$ .

We start with the construction of some auxiliary smooth rigid analytic spaces. For a reduced rigid space X, we denote by  $\mathcal{O}_X^+$  the subsheaf of  $\mathcal{O}_X$  of functions of norm less than 1. For  $i = 1, \ldots, r$ , let  $\alpha_i \in E^{\times}, \overline{\varrho}_i : \operatorname{Gal}_L \to \operatorname{GL}_{n_i}(k_E)$  be a continuous representation and let  $\xi_i$  be the (cuspidal) inertial type associated to  $\Omega_i$ . We consider the following functor:

$$X \longmapsto \{\rho_X, \varrho_{i,X}, \chi_{i,X}, \operatorname{Fil}_{\bullet}, \nu_i\} / \sim$$
(4.18)

where (the isomorphisms being defined in an obvious way)

(1) X is a reduced rigid analytic space over E;

- (2)  $\rho_X : \operatorname{Gal}_L \to \operatorname{GL}_n(\mathcal{O}_X^+)$  (resp.  $\varrho_{i,X} : \operatorname{Gal}_L \to \operatorname{GL}_{n_i}(\mathcal{O}_X^+)$  for  $i = 1, \ldots, r)$  is a continuous morphism such that, for all  $x \in X$ , the reduction modulo the maximal ideal of  $\mathcal{O}_{k(x)}$  of  $\rho_x := \rho_X \otimes_{\mathcal{O}_Y^+} \mathcal{O}_{k(x)}$  (resp. of  $\varrho_{i,x} := \varrho_{i,X} \otimes_{\mathcal{O}_Y^+} \mathcal{O}_{k(x)}$ ) is equal to  $\overline{\rho}$  (resp.  $\overline{\varrho}_i$ );
- (3)  $\rho_{i,x}$  is de Rham of Hodge-Tate weights  $\mathbf{h}^i := (\mathbf{h}_{s_{i-1}+1}, \ldots, \mathbf{h}_{s_i})$  and of type  $\xi_i$  for all  $x \in X$ ;
- (4)  $\chi_{i,X} : \mathcal{O}_L^{\times} \to \mathcal{O}_X^{\times}$  is a continuous morphism such that, for all points  $x \in X$ ,  $(\{\mathbf{r}(\varrho_{i,x}) \otimes_E unr(\alpha_i)\}, \{\chi_{i,x}\}) \in \mathscr{Z}^{\text{gen}}$  where  $\chi_{i,x} = x^*\chi_{i,X}$  and  $\mathbf{r}(\varrho_{i,x})$  is the Weil-Deligne representation associated to  $\varrho_{i,x}$ ;
- (5) Fil<sub>•</sub>  $D_{\text{rig}}(\rho_X) = \left(0 = \text{Fil}_0 D_{\text{rig}}(\rho_X) \subsetneq \text{Fil}_1 D_{\text{rig}}(\rho_X) \subsetneq \cdots \subsetneq \text{Fil}_r D_{\text{rig}}(\rho_X) = D_{\text{rig}}(\rho_X)\right)$  is an increasing filtration on  $D_{\text{rig}}(\rho_X)$  by  $(\varphi, \Gamma)$ -submodules over  $\mathcal{R}_{X,L}$  which are direct summands of  $D_{\text{rig}}(\rho_X)$  as  $\mathcal{R}_{X,L}$ -modules;
- (6)  $\nu_i : \operatorname{gr}_i D_{\operatorname{rig}}(\rho_X) \xrightarrow{\sim} D_{\operatorname{rig}}(\varrho_{i,X}) \otimes_{\mathcal{R}_{X,L}} \mathcal{R}_{X,L}((\chi_{i,X})_{\varpi_L}\operatorname{unr}(\alpha_i))$  is an isomorphism of  $(\varphi, \Gamma)$ modules over  $\mathcal{R}_{X,L}$ .

**Proposition 4.2.7.** The functor in (4.18) is representable by a smooth reduced rigid analytic space over E of dimension  $(\sum_{i=1}^{r} n_i^2 + n^2) + [L:\mathbb{Q}_p](\frac{n(n-1)}{2} + r)$ , that we denote by  $\mathcal{S}^0_{\overline{\rho}}(\Omega, \boldsymbol{h}, \underline{\alpha}, \{\overline{\varrho}_i\})$ .

*Proof.* For  $i = 1, \ldots, r$ , consider

$$(\operatorname{Spf} R^{\operatorname{pcr}}_{\overline{\varrho}_i}(\xi_i, \mathbf{h}^i))^{\operatorname{rig}} \to (\operatorname{Spec} \mathcal{Z}_{\Omega_i})^{\operatorname{rig}} \xrightarrow{\operatorname{tw}_{\alpha_i}} (\operatorname{Spec} \mathcal{Z}_{\Omega_i})^{\operatorname{rig}}$$
(4.19)

where the first morphism is induced by the morphism in [26, Thm. 4.1] (see also [26, Prop. 4.3]), and the second morphism sends  $\mathbf{r}_i$  to  $\mathbf{r}_i \otimes_E \operatorname{unr}(\alpha_i)$ . The morphism (4.19) is given pointwise by  $\varrho_i \mapsto \mathbf{r}(\varrho_i) \otimes_E \operatorname{unr}(\alpha_i)$ . Taking their product (with the identity map on  $\widehat{\mathcal{Z}}_{0,L}$ ), we define

$$\eta_{\underline{\alpha},\{\overline{\varrho}_i\}}: \prod_{i=1}^r (\operatorname{Spf} R^{\operatorname{pcr}}_{\overline{\varrho}_i}(\xi_i, \mathbf{h}^i))^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}} \longrightarrow (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}} \cong \mathscr{Z}.$$
(4.20)

Let  $\mathcal{U} := \eta_{\underline{\alpha},\{\overline{\varrho}_i\}}^{-1}(\mathscr{Z}^{\text{gen}})$ , which is Zariski-open and Zariski-dense in  $\prod_{i=1}^r (\operatorname{Spf} R_{\overline{\varrho}_i}^{\operatorname{pcr}}(\xi_i, \mathbf{h}^i))^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}}$ . Indeed, the Zariski-density follows from the fact that any affinoid open in the left hand side of (4.20) contains points with  $\widehat{\mathcal{Z}_{0,L}}$ -entry  $\chi = \boxtimes \chi_i$  satisfying  $\operatorname{wt}(\chi_i(\chi_j)^{-1})_{\tau} \notin \mathbb{Z}$  for all  $i \neq j$  and  $\tau \in \Sigma_L$ , and such points are sent to generic points via  $\eta_{\underline{\alpha},\{\overline{\varrho}_i\}}$ . Let  $\varrho_i^{\operatorname{univ}}$  be the universal Galois deformation over  $(\operatorname{Spf} R_{\overline{\varrho}_i}^{\operatorname{pcr}}(\xi_i, \mathbf{h}^i))^{\operatorname{rig}}$  and  $\varrho_{i,\mathcal{U}}^{\operatorname{univ}}$  be its pull-back over  $\mathcal{U}$  via the composition

$$\mathcal{U} \hookrightarrow \prod_{i=1}^{r} (\operatorname{Spf} R^{\operatorname{pcr}}_{\overline{\varrho}_{i}}(\xi_{i}, \mathbf{h}^{i}))^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}} \twoheadrightarrow (\operatorname{Spf} R^{\operatorname{pcr}}_{\overline{\varrho}_{i}}(\xi_{i}, \mathbf{h}^{i}))^{\operatorname{rig}}.$$

Likewise, let  $\chi_{i,\mathcal{U}}^{\text{univ}}$  be the pull-back over  $\mathcal{U}$  of the universal character over  $\widehat{\mathcal{O}_L^{\times}}$  via the composition

$$\mathcal{U} \hookrightarrow \prod_{i=1}^r (\operatorname{Spf} R^{\operatorname{pcr}}_{\overline{\varrho}_i}(\xi_i, \mathbf{h}^i))^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}} \twoheadrightarrow \widehat{\mathcal{Z}_{0,L}} \cong (\widehat{\mathcal{O}_L^{\times}})^r \xrightarrow{\operatorname{pr}_i} \widehat{\mathcal{O}_L^{\times}}.$$

Put

$$\mathcal{D}_{i,\mathcal{U}} := D_{\mathrm{rig}}(\varrho_{i,\mathcal{U}}^{\mathrm{univ}}) \otimes_{\mathcal{R}_{\mathcal{U},L}} \mathcal{R}_{\mathcal{U},L}\big((\chi_{i,\mathcal{U}}^{\mathrm{univ}})_{\varpi_L} \mathrm{unr}(\alpha_i)\big)$$

and  $\mathcal{D}_{i,z} := z^* \mathcal{D}_{i,\mathcal{U}}$  for each  $z \in \mathcal{U}$ . Since  $\xi_i$  is a cuspidal inertial type,  $\mathcal{D}_{i,z}$  is irreducible for all z. Using the fact that  $\eta_{\underline{\alpha}, \{\overline{\varrho}_i\}}(z) \in \mathscr{Z}^{\text{gen}}$ , one can calculate:

$$\dim_{k(z)} H^s_{(\varphi,\Gamma)}(\mathcal{D}_{i,z} \otimes_{\mathcal{R}_{k(z),L}} \mathcal{D}^{\vee}_{i,z}) = \begin{cases} 1 & s = 0\\ n_i^2[L:\mathbb{Q}_p] + 1 & s = 1\\ 0 & s = 2 \end{cases}$$

and by Lemma 4.1.12, for  $i \neq j$ :

$$\dim_{k(z)} H^s_{(\varphi,\Gamma)}(\mathcal{D}_{i,z} \otimes_{\mathcal{R}_{k(z),L}} \mathcal{D}^{\vee}_{j,z}) = \begin{cases} 0 & s = 0\\ n_i n_j [L : \mathbb{Q}_p] & s = 1\\ 0 & s = 2 \end{cases}$$

We deduce by Lemma 4.2.1 that for any rigid analytic space  $\mathcal{U}'$  over  $\mathcal{U}$ , all  $H^s_{(\varphi,\Gamma)}(\mathcal{D}_{i,\mathcal{U}'}\otimes_{\mathcal{R}_{\mathcal{U}',L}}\mathcal{D}_{j,\mathcal{U}'}^{\vee})$  are locally free over  $\mathcal{U}'$ , where  $\mathcal{D}_{i',\mathcal{U}'}$  denotes the base change of  $\mathcal{D}_{i',\mathcal{U}}$  over  $\mathcal{U}'$ .

Now we let  $\mathcal{U}_1 := \mathcal{U}, \ \mathcal{C}_{1,\mathcal{U}_1} := \mathcal{D}_{1,\mathcal{U}_1}$  and  $\nu_1 : \mathcal{C}_{1,\mathcal{U}_1} \xrightarrow{\sim} \mathcal{D}_{1,\mathcal{U}_1}$  be an isomorphism. Let  $\mathcal{U}_2 \to \mathcal{U}_1$  be the vector bundle of rank  $n_1 n_2 [L : \mathbb{Q}_p]$  associated to the locally free  $\mathcal{O}_{\mathcal{U}_1}$ -module  $\operatorname{Ext}^1_{(\varphi,\Gamma)}(\mathcal{D}_{2,\mathcal{U}_1},\mathcal{C}_{1,\mathcal{U}_1})$  (see also the proof of [30, Thm. 3.3] and [52, Thm. 2.4]). We have a universal extension of  $(\varphi,\Gamma)$ -modules over  $\mathcal{R}_{\mathcal{U}_2,L}$ :

$$0 \to \mathcal{C}_{1,\mathcal{U}_2} \to \mathcal{C}_{2,\mathcal{U}_2} \xrightarrow{\nu_2} \mathcal{D}_{2,\mathcal{U}_2} \to 0.$$

By similar arguments as in the previous paragraph and a dévissage, we prove that the  $\mathcal{O}_{\mathcal{U}_2}$ -module  $H^1_{(\varphi,\Gamma)}(\mathcal{C}_{2,\mathcal{U}_2} \otimes_{\mathcal{R}_{\mathcal{U}_2,L}} \mathcal{D}_{3,\mathcal{U}_2})$  is locally free of rank  $[L:\mathbb{Q}_p](n_1+n_2)n_3$ . We let  $\mathcal{U}_3 \to \mathcal{U}_2$  be the vector bundle associated to  $H^1_{(\varphi,\Gamma)}(\mathcal{C}_{2,\mathcal{U}_2} \otimes_{\mathcal{R}_{\mathcal{U}_2,L}} \mathcal{D}^{\vee}_{3,\mathcal{U}_2})$ . By induction, we finally obtain a sequence of rigid analytic spaces

$$\mathcal{V} := \mathcal{U}_r \to \mathcal{U}_{r-1} \to \cdots \to \mathcal{U}_1 = \mathcal{U}$$

such that

- (1) for  $i \geq 2$ ,  $\mathcal{U}_i$  is a vector bundle of rank  $[L:\mathbb{Q}_p](\sum_{j=1}^{i-1} n_j)n_i$  over  $\mathcal{U}_{i-1}$ ;
- (2) there is a (universal)  $(\varphi, \Gamma)$ -module  $\mathcal{C}_{i,\mathcal{U}_i}$  over  $\mathcal{R}_{\mathcal{U}_i,L}$  equipped with an increasing filtration  $(0 = \operatorname{Fil}_0 \subsetneq \operatorname{Fil}_1 \subsetneq \cdots \subsetneq \operatorname{Fil}_i = \mathcal{C}_{i,\mathcal{U}_i})$  given by  $(\varphi, \Gamma)$ -submodules over  $\mathcal{R}_{\mathcal{U}_i,L}$  which are direct summands of  $\mathcal{C}_{i,\mathcal{U}_i}$  as  $\mathcal{R}_{\mathcal{U}_i,L}$ -modules;
- (3) there are isomorphisms of  $(\varphi, \Gamma)$ -modules over  $\mathcal{R}_{\mathcal{U}_i, L}$ :  $\nu_j : \operatorname{gr}_j \mathcal{C}_{i, \mathcal{U}_i} \xrightarrow{\sim} \mathcal{D}_{j, \mathcal{U}_i}$  for  $j \leq i$ .

By construction and [63, Thm. 3.3.8],  $\mathcal{V}$  is smooth of dimension (recall  $\mathfrak{n}_P$  is the Lie algebra over E of the nilpotent radical  $N_P$  of P)

$$\dim \mathcal{U} + [L:\mathbb{Q}_p] \dim_E \mathfrak{n}_P = \sum_{i=1}^r \left( n_i^2 + [L:\mathbb{Q}_p] \frac{n_i(n_i-1)}{2} \right) + [L:\mathbb{Q}_p]r + [L:\mathbb{Q}_p] \dim_E \mathfrak{n}_P.$$

Now we apply the same argument as in the proof of [17, Thm. 2.6] for the representability of  $S^{\Box}(\bar{r})$ of *loc. cit.* Let  $\mathcal{V}^{\text{adm}}$  be the maximal adic open of  $\mathcal{V}$  such that there exist a vector bundle  $\mathcal{M}$  over  $\mathcal{V}^{\text{adm}}$  and a continuous morphism  $\rho_{\mathcal{V}^{\text{adm}}} : \text{Gal}_L \to \text{Aut}_{\mathcal{O}_{\mathcal{V}^{\text{adm}}}}(\mathcal{M})$  satisfying  $D_{\text{rig}}(\rho_{\mathcal{V}^{\text{adm}}}) \cong \mathcal{D}_{\mathcal{V}^{\text{adm}}}$ (cf. [51, Thm. 1.3]). Note that  $\mathcal{V}^{\text{adm}}$  is also a rigid space by [56, (1.1.11)]. Let  $\mathcal{V}^{\text{adm},\Box}$  be the  $\operatorname{GL}_n$ -torsor of the trivialization of the vector bundle  $\mathcal{M}$ . Let  $\mathcal{V}_0^{\operatorname{adm},\square} \subset \mathcal{V}^{\operatorname{adm},\square}$  be the admissible open of points such that  $\rho_{\mathcal{V}_0^{\operatorname{adm},\square}} := \rho_{\mathcal{V}^{\operatorname{adm},\square}}|_{\mathcal{V}_0^{\operatorname{adm},\square}}$  has image in  $\Gamma(\mathcal{V}_0^{\operatorname{adm},\square}, \mathcal{O}_{\mathcal{V}_0^{\operatorname{adm},\square}}^+)$ . Finally let  $\mathcal{S}_{\overline{\rho}}^0(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\})$  be the admissible open of  $\mathcal{V}_0^{\operatorname{adm},\square}$  such that the reduction modulo the maximal ideal of  $\mathcal{O}_{k(x)}$  of  $\rho_{\mathcal{V}_0^{\operatorname{adm},\square}} \otimes_{\mathcal{O}_{\mathcal{V}_0^{\operatorname{adm},\square}}^+} \mathcal{O}_{k(x)}$  is equal to  $\overline{\rho}$  for  $x \in \mathcal{S}_{\overline{\rho}}^0(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\})$ . We see that  $\mathcal{S}_{\overline{\rho}}^0(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\})$  is smooth of dimension

$$\sum_{i=1}^{r} \left( n_i^2 + [L:\mathbb{Q}_p] \frac{n_i(n_i-1)}{2} \right) + [L:\mathbb{Q}_p]r + [L:\mathbb{Q}_p] \dim_E \mathfrak{n}_P + n^2$$
$$= \left( \sum_{i=1}^{r} n_i^2 + n^2 \right) + [L:\mathbb{Q}_p] \frac{n(n-1)}{2} + [L:\mathbb{Q}_p]r.$$

It is then formal to check that  $\mathcal{S}^{0}_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\})$  represents the functor (4.18).

We write  $\mathcal{S}^{0}_{\overline{\rho}} := \mathcal{S}^{0}_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\})$  for simplicity. Consider the composition:

$$\mathcal{S}^{0}_{\overline{\rho}} \longrightarrow \left(\prod_{i=1}^{r} (\operatorname{Spf} R^{\operatorname{pcr}}_{\overline{\rho}}(\xi_{i}, \mathbf{h}^{i}))^{\operatorname{rig}}\right) \times \widehat{\mathcal{Z}_{0,L}} \xrightarrow{\eta_{\underline{\alpha}, \{\overline{\varrho}_{i}\}}} (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}} \cong \mathscr{Z}.$$

where the first map follows from the construction of  $S^0_{\overline{\rho}}$  (note that this map is smooth). Let  $\Delta_{\Omega_i}$ be the universal *p*-adic differential equation over (Spec  $Z_{\Omega_i}$ )<sup>rig</sup> constructed in § 2.2 and  $\chi_i^{\text{univ}}$  be the universal character of  $\mathcal{O}_L^{\times}$  over  $\widehat{\mathcal{O}_L^{\times}}$ . We let  $\chi_{i,S^0_{\overline{\rho}}}^{\text{univ}}$ ,  $\Delta_{\Omega_i,S^0_{\overline{\rho}}}$  be the pull-back of  $\chi_i^{\text{univ}}$ ,  $\Delta_{\Omega_i}$  over  $S^0_{\overline{\rho}}$ . Let  $\rho_{S^0}^{\text{univ}}$  be the universal Gal<sub>L</sub>-representation over  $S^0_{\overline{\rho}}$ . Similarly as in the proof of Proposition 4.2.9 below (using Lemma 4.2.1 and a direct computation of the cohomology of  $(\varphi, \Gamma)$ -modules over Robba rings with coefficients in finite extensions of E), we can show that

$$\operatorname{Hom}_{(\varphi,\Gamma)}\left(\operatorname{gr}_{i} D_{\operatorname{rig}}(\rho_{\mathcal{S}^{0}_{\overline{\rho}}}^{\operatorname{univ}}) \otimes_{\mathcal{R}_{\mathcal{S}^{0}_{\overline{\rho}},L}} \mathcal{R}_{\mathcal{S}^{0}_{\overline{\rho}},L}\left((\chi_{i,\mathcal{S}^{0}_{\overline{\rho}}}^{\operatorname{univ}})_{\varpi_{L}}^{-1}\right), \Delta_{\Omega_{i},\mathcal{S}^{0}_{\overline{\rho}}} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(z^{\mathbf{h}_{s_{i}}})\right)$$
(4.21)

is locally free of rank 1 over  $S^0_{\overline{\rho}}$  for any  $i = 1, \ldots, r$ . We let  $S_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\})$  be the fibre product over  $S^0_{\overline{\rho}}$  for all i of the  $\mathbb{G}_m^{\text{rig}}$ -torsors trivializing the invertible modules in (4.21), which is hence a  $(\mathbb{G}_m^{\text{rig}})^r$ -torsor over  $S^0_{\overline{\rho}}$  and a reduced rigid analytic space over E of dimension

$$\left(r + \sum_{i=1}^{r} n_i^2 + n^2\right) + [L:\mathbb{Q}_p] \left(\frac{n(n-1)}{2} + r\right).$$
 (4.22)

We have a natural commutative diagram

The existence of the upper horizontal morphism follows from the construction of  $S_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\})$ , and it is also clear that the composition of the upper horizontal with the right vertical morphism has image included in  $X_{\Omega,\mathbf{h}}(\overline{\rho})$ , from which we obtain the left vertical morphism  $\kappa_{\alpha,\{\overline{\varrho}_i\}}$ . Let  $\rho^{\text{univ}}$  be the universal framed Galois deformation of  $\overline{\rho}$  over  $(\text{Spf } R_{\overline{\rho}})^{\text{rig}}$ . And we let  $\rho^{\text{univ}}_{X_{\Omega,\mathbf{h}}(\overline{\rho})}$ ,  $\Delta_{\Omega_i,X_{\Omega,\mathbf{h}}(\overline{\rho})}, \chi^{\text{univ}}_{i,X_{\Omega,\mathbf{h}}(\overline{\rho})}$  be the pull-back of  $\rho^{\text{univ}}, \Delta_{\Omega_i}, \chi^{\text{univ}}_i$  over  $X_{\Omega,\mathbf{h}}(\overline{\rho})$ . Applying Corollary A.1.2 in the Appendix, we see that there exist a projective birational morphism

$$f: \widetilde{X}_{\Omega,\mathbf{h}}(\overline{\rho}) \longrightarrow X_{\Omega,\mathbf{h}}(\overline{\rho}),$$

a Zariski closed subset  $Z \subset \widetilde{X}_{\Omega,\mathbf{h}}(\overline{\rho})$  disjoint from  $f^{-1}(U_{\Omega,\mathbf{h}}(\overline{\rho}))$  and line bundles  $\mathcal{L}_i$  over  $\mathscr{U} := \widetilde{X}_{\Omega,\mathbf{h}}(\overline{\rho}) \setminus Z$  such that  $D_{\mathscr{U}} := f^* D_{\mathrm{rig}}(\rho_{X_{\Omega,\mathbf{h}}(\overline{\rho})}^{\mathrm{univ}})|_{\mathscr{U}}$  admits an increasing filtration Fil<sub>•</sub>  $D_{\mathscr{U}}$  by  $(\varphi, \Gamma)$ -submodules over  $\mathcal{R}_{\mathscr{U},L}$  such that

- (1) Fil<sub>i</sub>  $D_{\mathscr{U}}$  are direct summands of  $D_{\mathscr{U}}$  as  $\mathcal{R}_{\mathscr{U},L}$ -modules;
- (2) one has embeddings  $\operatorname{gr}_i D_{\mathscr{U}} \otimes \mathcal{L}_i \hookrightarrow \Delta_{\Omega_i, \mathscr{U}} \otimes_{\mathcal{R}_{\mathscr{U}, L}} \mathcal{R}_{\mathscr{U}, L}((\chi_{i, \mathscr{U}})_{\varpi_L} z^{\mathbf{h}_{s_i}})$  where  $\Delta_{\Omega_i, \mathscr{U}}, \chi_{i, \mathscr{U}}$ are the pull-backs of  $\Delta_{\Omega_i, X_{\Omega, \mathbf{h}}(\overline{\rho})}, \chi_{i, X_{\Omega, \mathbf{h}}(\overline{\rho})}^{\operatorname{univ}}$  over  $\mathscr{U}$  respectively;
- (3) for all  $x \in \mathscr{U}$ , the above embedding restrict to injections  $(\operatorname{gr}_i D_{\mathscr{U}})_x \hookrightarrow \Delta_{\Omega_i, f(x)} \otimes_{\mathcal{R}_{k(x), L}} \mathcal{R}_{k(x), L}(\chi_{i, f(x), \varpi_L} z^{\mathbf{h}_{s_i}})$  and  $(\operatorname{gr}_i D_{\mathscr{U}})_x \otimes_{\mathcal{R}_{k(x), L}} \mathcal{R}_{k(x), L}(\chi_{i, f(x), \varpi_L}^{-1})$  is de Rham of Hodge-Tate weights  $\mathbf{h}^i = (\mathbf{h}_{s_{i-1}+1}, \ldots, \mathbf{h}_{s_i})$ .

Note that, by the proof of Theorem A.1.1, we can and do assume that f factors through a surjective birational morphism  $\widetilde{X}_{\Omega,\mathbf{h}}(\overline{\rho}) \twoheadrightarrow X_{\Omega,\mathbf{h}}(\overline{\rho})'$  where  $X_{\Omega,\mathbf{h}}(\overline{\rho})'$  denotes the normalization of  $X_{\Omega,\mathbf{h}}(\overline{\rho})$ .

Let  $\mathscr{U}_0$  be the preimage of  $\mathscr{Z}^{\text{gen}}$  via the natural composition

$$\mathscr{U} \longrightarrow \widetilde{X}_{\Omega,\mathbf{h}}(\overline{\rho}) \longrightarrow X_{\Omega,\mathbf{h}}(\overline{\rho}) \stackrel{\omega}{\longrightarrow} \mathscr{Z}.$$

Note that  $\mathscr{U}_0$  is Zariski-open in  $\mathscr{U}$  hence also in  $\widetilde{X}_{\Omega,\mathbf{h}}(\overline{\rho})$ . It is also clear that  $f^{-1}(U_{\Omega,\mathbf{h}}(\overline{\rho})) \subset \mathscr{U}_0$ (in fact  $U_{\Omega,\mathbf{h}}(\overline{\rho})$  is equal to the set of rigid analytic points of  $f(\mathscr{U}_0)$ ). Let  $\mathscr{U}_1$  be the fibre product over  $\mathscr{U}$  for all i of the  $\mathbb{G}_m^{\text{rig}}$ -torsors trivializing the line bundle  $\mathcal{L}_i$ . Let  $x \in \mathscr{U}$ , and  $\alpha_i \in E^{\times}$ (enlarging E if necessary) for  $i = 1, \ldots, r$  such that  $\operatorname{gr}_i(D_{\mathscr{U}})_x \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\chi_{i,f(x),\varpi_L}^{-1}\operatorname{unr}(\alpha_i^{-1}))$ is étale. Let  $\mathscr{U}_1(\underline{\alpha})^{\operatorname{adm}}$  be the maximal adic open subset of  $\mathscr{U}_1$  such that there exist a vector bundle  $\varrho_{i,\mathscr{U}_1(\underline{\alpha})^{\operatorname{adm}}}$  over  $\mathscr{U}_1(\underline{\alpha})^{\operatorname{adm}}$  and a continuous morphism  $\operatorname{Gal}_L \to \operatorname{Aut}_{\mathcal{O}_{\mathscr{U}_1(\underline{\alpha})^{\operatorname{adm}}}}(\varrho_{i,\mathscr{U}_1(\underline{\alpha})^{\operatorname{adm}}})$ satisfying (cf. [51, Thm. 1.3]):

$$D_{\mathrm{rig}}(\varrho_{i,\mathscr{U}_{1}(\underline{\alpha})^{\mathrm{adm}}}) \cong \left( \operatorname{gr}_{i} D_{\mathscr{U}_{1}} \otimes_{\mathcal{R}_{\mathscr{U}_{1},L}} \mathcal{R}_{\mathscr{U}_{1},L}(\chi_{i,\mathscr{U}_{1},\varpi_{L}}^{-1} \operatorname{unr}(\alpha_{i}^{-1})) \right) |_{\mathscr{U}_{1}(\underline{\alpha})^{\mathrm{adm}}}$$

Since f is projective and  $X_{\Omega,\mathbf{h}}(\overline{\rho})$  is quasi-separated,  $\widetilde{X}_{\Omega,\mathbf{h}}(\overline{\rho})$  is quasi-separated. We then deduce that both  $\mathscr{U}_1$  and  $\mathscr{U}_1(\underline{\alpha})^{\mathrm{adm}}$  are also quasi-separated, and hence that  $\mathscr{U}_1(\underline{\alpha})^{\mathrm{adm}}$  is a rigid analytic space by [56, (1.1.11)]. It is also clear that any preimage of x lies in  $\mathscr{U}_1(\underline{\alpha})^{\mathrm{adm}}$ . Let  $\mathscr{U}_1(\underline{\alpha})^{\mathrm{adm},\square}$ be the fibre product over  $\mathscr{U}_1(\underline{\alpha})^{\mathrm{adm}}$  for all i of the  $\mathrm{GL}_{n_i}$ -torsors of the trivialization of  $\varrho_{i,\mathscr{U}_1(\underline{\alpha})^{\mathrm{adm}}}$ as  $\mathcal{O}_{\mathscr{U}_1(\underline{\alpha})^{\mathrm{adm}}}$ -module. Let  $\mathscr{U}(\underline{\alpha})^{\mathrm{adm},\square} \subset \mathscr{U}_1(\underline{\alpha})^{\mathrm{adm},\square}$  be the admissible open subset of points such that

$$\varrho_{i,\mathscr{U}_{1}(\underline{\alpha})^{\mathrm{adm},\Box}}:\mathrm{Gal}_{L}\longrightarrow\mathrm{GL}_{n_{i}}\left(\Gamma(\mathscr{U}(\underline{\alpha})^{\mathrm{adm},\Box},\mathcal{O}_{\mathscr{U}_{1}(\underline{\alpha})^{\mathrm{adm},\Box}})\right)$$

has image in  $\Gamma(\mathscr{U}(\underline{\alpha})^{\mathrm{adm},\square}, \mathcal{O}^+_{\mathscr{U}_1(\underline{\alpha})^{\mathrm{adm},\square}})$ . Finally, fix some continuous  $\overline{\varrho}_i : \mathrm{Gal}_L \to \mathrm{GL}_{n_i}(k_E)$ and let  $\mathscr{U}(\underline{\alpha}, \{\overline{\varrho}_i\})^{\mathrm{adm},\square} \subset \mathscr{U}(\underline{\alpha})^{\mathrm{adm},\square}$  be the open locus such that, for  $z \in \mathscr{U}(\underline{\alpha}, \{\overline{\varrho}_i\})^{\mathrm{adm},\square}$ ,  $\mathcal{Q}_{i,\mathscr{U}(\underline{\alpha},\{\overline{\varrho}_i\})^{\mathrm{adm},\Box}} \otimes \mathcal{O}_{k(z)}$  has reduction (modulo the maximal ideal of  $\mathcal{O}_{k(z)}$ ) equal to  $\overline{\varrho}_i$  for all i. From the universal property of  $\mathcal{S}_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\})$ , we obtain a natural morphism

$$\mathscr{U}(\underline{\alpha}, \{\overline{\varrho}_i\})^{\mathrm{adm}, \Box} \longrightarrow \mathcal{S}_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\}).$$

Moreover, using the universal property of  $(\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}}$  and the "universal" property of  $(\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}}$  as in [26, Prop. 4.3], we see that the following diagram commutes

By (4.23) and the fact that  $X_{\Omega,\mathbf{h}}(\overline{\rho})$  is a reduced closed subspace of  $(\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}} \times \mathscr{Z}$ , we deduce a commutative diagram:

$$\mathscr{U}(\underline{\alpha}, \{\overline{\varrho}_i\})^{\mathrm{adm}, \Box} \longrightarrow \mathcal{S}_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\})$$

$$\overset{\kappa_{\underline{\alpha}, \{\overline{\varrho}_i\}} \downarrow}{X_{\Omega, \mathbf{h}}(\overline{\rho}).}$$

$$(4.24)$$

By Corollary A.1.2, there exists a Zariski-open and Zariski-dense subset  $\mathscr{V} \subset X_{\Omega,\mathbf{h}}(\overline{\rho})$  disjoint from f(Z) such that the morphism f induces an isomorphism  $f^{-1}(\mathscr{V}) \xrightarrow{\sim} \mathscr{V}$ . Consider

$$\mathscr{V}(\underline{\alpha}, \{\overline{\varrho}_i\})^{\mathrm{adm}, \Box} := f^{-1}(\mathscr{V}) \times_{\mathscr{U}} \mathscr{U}(\underline{\alpha}, \{\overline{\varrho}_i\})^{\mathrm{adm}, \Box}.$$

Note that  $\mathscr{V}(\underline{\alpha}, \{\overline{\varrho}_i\})^{\mathrm{adm}, \Box}$  can also be constructed from  $f^{-1}(\mathscr{V})$  in the same way  $\mathscr{U}(\underline{\alpha}, \{\overline{\varrho}_i\})^{\mathrm{adm}, \Box}$  was constructed from  $\mathcal{U}$ . Using the same argument as in the first paragraph on page 1598 of [17], one can prove an isomorphism

$$\mathscr{V}(\underline{\alpha}, \{\overline{\varrho}_i\})^{\mathrm{adm}, \Box} \xrightarrow{\sim} \kappa_{\underline{\alpha}, \{\overline{\varrho}_i\}}^{-1}(\mathscr{V})$$

hence by (4.22),  $\mathscr{V}(\underline{\alpha}, \{\overline{\varrho}_i\})^{\mathrm{adm}, \Box}$  is smooth of dimension  $(r + \sum_{i=1}^r n_i^2 + n^2) + [L : \mathbb{Q}_p](\frac{n(n-1)}{2} + r)$ . By [59, Thm. 0.2], for any  $x \in \mathscr{V}$ , there exist an admissible open neighbourhood  $\mathscr{V}_x$  of x in  $\mathscr{V}$  and  $\alpha_i, \overline{\varrho}_i$  such that  $\mathscr{V}_x$  is contained in the image of  $\mathscr{V}(\underline{\alpha}, \{\overline{\varrho}_i\})^{\mathrm{adm}, \Box} \to \mathscr{V}$  (for the corresponding  $\underline{\alpha}, \{\overline{\varrho}_i\}$ ). As the morphism  $\mathscr{V}(\underline{\alpha}, \{\overline{\varrho}_i\})^{\mathrm{adm}, \Box} \to \mathscr{V}$  is smooth of relative dimension  $r + \sum_{i=1}^r n_i^2$ , we deduce that  $\mathscr{V}$  is equidimensional of dimension  $n^2 + [L : \mathbb{Q}_p](\frac{n(n-1)}{2} + r)$ . Since  $\mathscr{V}$  is Zariski dense in  $X_{\Omega,\mathbf{h}}(\overline{\rho})$ , (1) of Theorem 4.2.5 follows.

**Proposition 4.2.8.** The morphism  $\kappa_{\underline{\alpha}, \{\varrho_i\}} : S_{\overline{\rho}}(\Omega, h, \underline{\alpha}, \{\overline{\varrho}_i\}) \to X_{\Omega, h}(\overline{\rho})$  is smooth.

Proof. Let  $x \in S_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\}), y := \kappa_{\underline{\alpha}, \{\overline{\varrho}_i\}}(x), B := \widehat{\mathcal{O}}_{S_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\}), x}, A := \widehat{\mathcal{O}}_{X_{\Omega, \mathbf{h}}(\overline{\rho}), y}$ . Let  $N := \sum_{i=1}^r n_i^2 + r$ , it is enough to show that there exist  $x_1, \ldots, x_N \in B$  such that  $B \cong A[[x_1, \ldots, x_N]]$ . Let  $\rho_x : \operatorname{Gal}_L \to \operatorname{GL}_n(k(x))$  be the image of x in  $(\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}}$ . Recall we have a natural isomorphism  $\widehat{\mathcal{O}}_{(\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}}, \rho_x} \cong R_{\rho_x}$  where  $R_{\rho_x}$  is the universal framed deformation ring of  $\rho_x$ . Let  $\mathscr{F}$  be the associated filtration on  $D_{\operatorname{rig}}(\rho_x)$  (as in (4.18)), and denote by  $R_{\rho_x, \mathscr{F}}^0$  the local complete k(x)-algebra which (pro-)represents the functor  $F_{\rho_x, \mathscr{F}}^0$  (see Corollary 4.1.18 (1)). By Corollary 4.1.18 and Remark 4.1.19 (note that, as x is sent to  $\mathscr{X}^{\operatorname{gen}}$ , the hypothesis there are satisfied for  $\mathscr{F}$  by Lemma

4.1.12),  $R^0_{\rho_x,\mathscr{F}}$  is a quotient of  $R_{\rho_x}$  and is formally smooth of dimension  $n^2 + [L:\mathbb{Q}_p](\frac{n(n-1)}{2} + r)$ . We have a natural morphism

$$R^0_{\rho_x,\mathscr{F}} \longrightarrow B.$$
 (4.25)

Indeed, for an ideal  $I \subset \mathfrak{m}_B$  (the maximal ideal of B) with  $\dim_{k(x)} B/I < \infty$ , by the construction of  $S_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \underline{\rho})$  and by Lemma 4.1.5, we have a deformation of  $(\rho_x, \mathscr{F})$  over B/I lying in  $F^0_{\rho_x, \mathscr{F}}(B/I)$ . By the universal property of  $R^0_{\rho_x, \mathscr{F}}$ , this gives a natural morphism  $R^0_{\rho_x, \mathscr{F}} \to B/I$ . Taking the projective limit over all ideals I, we obtain (4.25). Let  $\varrho_{i,x}$  be the image of x in  $(\operatorname{Spf} R^{\operatorname{pcr}}_{\overline{\varrho_i}}(\xi_i, \mathbf{h}^i))^{\operatorname{rig}}$ . Using the fact that  $\widehat{\mathcal{O}}_{(\operatorname{Spf} R^{\operatorname{pcr}}_{\overline{\varrho_i}}(\xi_i, \mathbf{h}^i))^{\operatorname{rig}}, \varrho_{i,x}}$  (pro-)represents the functor of framed de Rham deformations of  $\varrho_{i,x}$  and using Lemma 4.1.5, one can show that the tangent map  $(\mathfrak{m}_B/\mathfrak{m}_B^2)^{\vee} \to F^0_{\rho_x, \mathscr{F}}(k(x)[\varepsilon]/\varepsilon^2)$  of (4.25) is surjective (we leave the details to the reader). Together with the fact that both B and  $R^0_{\rho_x, \mathscr{F}}$  are formally smooth, we deduce that (4.25) is formally smooth of relative dimension N. There exist thus  $x_1, \ldots, x_N \in B$ , such that

$$B \cong R^0_{\rho_x,\mathscr{F}}[[x_1,\ldots,x_N]].$$

Since  $R^0_{\rho_x,\mathscr{F}}$  is a quotient of  $R_{\rho_x}$ , we deduce from the above isomorphism a surjective morphism  $R_{\rho_x}[[x_1,\ldots,x_N]] \to B$ . The morphism  $X_{\Omega,\mathbf{h}}(\overline{\rho}) \to (\operatorname{Spf} R_{\overline{\rho}})^{\operatorname{rig}}$  induces a morphism  $R_{\rho_x} \to A$ . Using the commutative diagram (4.23), we see that the morphism  $R_{\rho_x} \to B$  factors through A. To sum up, we have obtained a surjective morphism

$$A[[x_1,\ldots,x_N]] \longrightarrow B.$$

Since dim  $A = \dim B - N$  and B is formally smooth, the above morphism is an isomorphism if A is integral. But this follows from exactly the same argument as in the first paragraph on page 1599 of [17] with  $X_{\text{tri}}^{\Box}(\overline{r}), U^{\Box}, S^{\Box}(\overline{r})$  of *loc. cit.* replaced by  $X_{\Omega,\mathbf{h}}(\overline{\rho}), \mathscr{U}(\underline{\alpha}, \{\overline{\varrho_i}\})^{\text{adm},\Box}$  and  $S_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho_i}\})$  respectively.  $\Box$ 

We use Proposition 4.2.8 to prove (2) of Theorem 4.2.5 (following the strategy in the proof of [17, Thm. 2.6]). We also need to use adic spaces. By [56, Prop. 1.7.8],  $\operatorname{Im}(\kappa_{\alpha,\{\overline{\varrho},\}})$  is an adic open subset of (the adic space associated to)  $X_{\Omega,\mathbf{h}}(\overline{\rho})$  with rigid analytic points contained in  $U_{\Omega,\mathbf{h}}(\overline{\rho})$ . Letting  $\underline{\alpha}$ ,  $\{\overline{\varrho}_i\}$  vary, the union of the  $\operatorname{Im}(\kappa_{\underline{\alpha},\{\overline{\varrho}_i\}})$  is also an adic open subset U of  $X_{\Omega,\mathbf{h}}(\overline{\rho})$ . But it is easy to see that any point of the rigid space  $U_{\Omega,\mathbf{h}}(\overline{\rho})$  lies in  $\mathrm{Im}(\kappa_{\alpha,\{\overline{\rho}_i\}})$  for some  $\underline{\alpha}$  and  $\{\overline{\varrho}_i\}$ , hence  $U_{\Omega,\mathbf{h}}(\overline{\rho})$  coincides with the rigid analytic points of the adic open subset U of  $X_{\Omega,\mathbf{h}}(\overline{\rho})$ . We show that U is a Zariski-constructible subset (see [17, Lemma 2.13]). Since  $\mathscr{U}_0$  is Zariski-open in  $X_{\Omega,\mathbf{h}}(\overline{\rho})$  and f is projective, we see by [52, Lemma 2.14] that the set  $f(\mathscr{U}_0)$ is Zariski-constructible in (the adic space associated to)  $X_{\Omega,\mathbf{h}}(\overline{\rho})$ . We claim  $U = f(\mathscr{U}_0)$ . Indeed, both sets have the same rigid analytic points, i.e. those in  $U_{\Omega,\mathbf{h}}(\overline{\rho})$ . Using [52, Lemma 2.15], the inclusion of adic spaces  $\kappa_{\underline{\alpha},\{\overline{\varrho}_i\}}^{-1}(f(\mathscr{U}_0)) \subseteq S_{\overline{\rho}}(\Omega,\mathbf{h},\underline{\alpha},\{\overline{\varrho}_i\})$  is an isomorphism, which implies  $\operatorname{Im}(\kappa_{\underline{\alpha},\{\overline{\varrho}_i\}}) \subseteq f(\mathscr{U}_0)$  (as subsets of the adic space associated to  $X_{\Omega,\mathbf{h}}(\overline{\rho})$ ) and hence  $U \subset f(\mathscr{U}_0)$ . On the other hand, using (4.24) and the fact that the image of  $\mathscr{U}(\underline{\alpha}, \{\overline{\varrho}_i\})^{\mathrm{adm}, \Box}$  in  $\mathscr{U}_0$  forms an adic open covering of  $\mathscr{U}_0$  when  $\underline{\alpha}, \{\overline{\varrho}_i\}$  vary, we can deduce  $f(\mathscr{U}_0) \subset U$ . Hence we see that the adic open subset U of  $X_{\Omega,\mathbf{h}}(\overline{\rho})$  is Zariski-constructible. It then follows from [52, Lemma 2.13] that U is Zariski-open in the adic space associated to  $X_{\Omega,\mathbf{h}}(\overline{\rho})$ , hence a fortiori  $U_{\Omega,\mathbf{h}}(\overline{\rho})$  is Zariski-open in the rigid space  $X_{\Omega,\mathbf{h}}(\bar{\rho})$ . This concludes the proof of (2) of Theorem 4.2.5.

Finally, by [17, Lemma 5.8] (applied to  $S_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\}) \to U_{\Omega,\mathbf{h}}(\overline{\rho}) \to \mathscr{Z})$ , to show (3) of Theorem 4.2.5, i.e. the smoothness of the morphism  $\omega : U_{\Omega,\mathbf{h}}(\overline{\rho}) \to \mathscr{Z}$ , it is sufficient to show that the morphism  $S_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\}) \to \mathscr{Z}$  is smooth. From the construction of  $S_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\})$ , this is a consequence of the following proposition. **Proposition 4.2.9.** The morphism  $\eta$  :  $(\operatorname{Spf} R_{\overline{\varrho}_i}^{\operatorname{pcr}}(\xi_i, h^i))^{\operatorname{rig}} \to (\operatorname{Spec} \mathcal{Z}_{\Omega_i})^{\operatorname{rig}}$  of [26, Prop. 4.3] which sends a deformation  $\varrho_i$  to  $r(\varrho_i)$  is smooth.

Proof. Since both source and target of  $\eta$  are smooth, we only need to show that the tangent map  $d\eta_x : T_{(\text{Spf } R_{\overline{\varrho}_i}^{\text{pcr}}(\xi_i, \mathbf{h}^i))^{\text{rig}}, x} \to T_{(\text{Spec } \mathcal{Z}_{\Omega_i})^{\text{rig}}, \eta(x)}$  is surjective for any  $x \in (\text{Spf } R_{\overline{\varrho}_i}^{\text{pcr}}(\xi_i, \mathbf{h}^i))^{\text{rig}},$ where  $T_{Y,z}$  denotes the tangent space of Y at z for a point z of a rigid space Y. For  $x \in (\text{Spf } R_{\overline{\varrho}_i}^{\text{pcr}}(\xi_i, \mathbf{h}^i))^{\text{rig}},$  let  $\varrho_{i,x}$  be the associated  $\text{Gal}_L$ -representation and  $\Delta_{i,\eta(x)}$  be the p-adic differential equation associated to  $\eta(x)$ . For a  $(\varphi, \Gamma)$ -module D, we put  $D(\mathbf{h}_{s_i}) := D \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(z^{\mathbf{h}_{s_i}}).$ One easily computes for any point x

$$\dim_{k(x)} H^{i}_{(\varphi,\Gamma)} \left( D_{\operatorname{rig}}(\varrho_{i,x})^{\vee} \otimes_{\mathcal{R}_{k(x),L}} \Delta_{i,\eta(x)}(\mathbf{h}_{s_{i}}) \right) = \begin{cases} 1 & i = 0\\ n_{i}^{2}[L:\mathbb{Q}_{p}] + 1 & i = 1\\ 0 & i = 2 \end{cases}$$

By Lemma 4.2.1, we deduce in particular that  $\operatorname{Hom}_{(\varphi,\Gamma)}(D_{\operatorname{rig}}(\varrho_i^{\operatorname{univ}}), \eta^* \Delta_{\Omega_i}(\mathbf{h}_{s_i}))$  is an invertible sheaf over  $(\operatorname{Spf} R_{\overline{\varrho}_i}^{\operatorname{pcr}}(\xi_i, \mathbf{h}^i))^{\operatorname{rig}}$ . Now fix  $x \in (\operatorname{Spf} R_{\overline{\varrho}_i}^{\operatorname{pcr}}(\xi_i, \mathbf{h}^i))^{\operatorname{rig}}$  and let

$$\psi \in \operatorname{Spec} k(x)[\varepsilon]/\varepsilon^2 \longrightarrow (\operatorname{Spec} \mathcal{Z}_{\Omega_i})^{\operatorname{rig}}$$

be an element in  $T_{(\text{Spec } \mathcal{Z}_{\Omega_i})^{\text{rig}},\eta(x)}$ . Let  $\Delta_{\psi} := \psi^* \Delta_{\Omega_i}$ . By the proof of Proposition 4.1.4, there exists a deformation  $\widetilde{\varrho}_{i,x}$  of  $\varrho_{i,x}$  over  $k(x)[\varepsilon]/\varepsilon^2$  such that one has an embedding of  $(\varphi, \Gamma)$ -modules over  $\mathcal{R}_{\text{Spec } k(x)[\varepsilon]/\varepsilon^2,L}$ :

$$j: D_{\mathrm{rig}}(\widetilde{\varrho}_{i,x}) \hookrightarrow \Delta_{\psi}(\mathbf{h}_{s_i}).$$

As  $\Delta_{\psi}$  is de Rham, so is  $\tilde{\varrho}_{i,x}$  (recall they have the same rank over  $\mathcal{R}_{k(x),L}$ ). Thus  $\tilde{\varrho}_{i,x}$  corresponds to an element

$$\psi' : \operatorname{Spec} k(x)[\varepsilon]/\varepsilon^2 \longrightarrow (\operatorname{Spf} R^{\operatorname{pcr}}_{\overline{\varrho}_i}(\xi_i, \mathbf{h}^i))^{\operatorname{rig}}$$

in the tangent space at x. Let us prove that  $d\eta_x$  maps  $\psi'$  to  $\psi$ , or equivalently  $\eta \circ \psi' = \psi$ . Consider  $\Delta_{\eta \circ \psi'} := (\eta \circ \psi')^* \Delta_{\Omega_i}(\mathbf{h}_{s_i})$ . Using Lemma 4.2.1, a local generator of the invertible sheaf  $\operatorname{Hom}_{(\varphi, \Gamma)}(D_{\operatorname{rig}}(\varrho_i^{\operatorname{univ}}), \eta^* \Delta_{\Omega_i}(\mathbf{h}_{s_i}))$  induces by pull-back via  $\psi'$  a morphism

$$\iota: D_{\mathrm{rig}}(\widetilde{\varrho}_{i,x}) \longrightarrow (\eta \circ \psi')^* \Delta_{\Omega_i}(\mathbf{h}_{s_i})$$

which is a generator of the  $k(x)[\varepsilon]/\varepsilon^2$ -module  $\operatorname{Hom}_{(\varphi,\Gamma)}(D_{\operatorname{rig}}(\tilde{\varrho}_{i,x}), \Delta_{\eta\circ\psi'})$ . Since  $D_{\operatorname{rig}}(\varrho_{i,x})$  is irreducible, it is not difficult to see that  $\iota$  has to be injective. From the two injections  $\jmath$ ,  $\iota$ , by comparing the Hodge-Tate weights and using [6, Thm. A], we can deduce  $\Delta_{\eta\circ\psi'} \cong \Delta_{\psi}$ . By the discussion in § 2.2, we see that there exists a bijection  $T_{(\operatorname{Spec} Z_{\Omega_i})^{\operatorname{rig}},\eta(x)} \xrightarrow{\sim} k(x), f \mapsto a_f$  such that  $f^*\Delta_{\Omega_i} \cong \Delta_{i,\eta(x)} \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x)[\varepsilon]/\varepsilon^2,L}(\operatorname{unr}(1+a_f\varepsilon))$ . Hence, for  $* \in \{\psi, \eta \circ \psi'\}$ , there is  $a_* \in k(x)$  such that  $\Delta_* \cong \Delta_{i,\eta(x)} \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x)[\varepsilon]/\varepsilon^2,L}(\operatorname{unr}(1+a_*\varepsilon))$ . By the proof of Lemma 4.1.2,  $\Delta_{\psi} \cong \Delta_{\eta\circ\psi'}$  implies  $\operatorname{unr}(1+a_{\psi}\varepsilon) = \operatorname{unr}(1+a_{\eta\circ\psi'}\varepsilon)$ , which further implies  $a_{\psi} = a_{\eta\circ\psi'}$  and hence  $\eta \circ \psi' = \psi$ . The proposition follows.  $\Box$ 

We end this paragraph by the following proposition on Sen weights which will be used in  $\S$  6.4.

**Proposition 4.2.10.** Let  $x = (\rho, \underline{x}, \chi) \in X_{\Omega, h}(\overline{\rho})$ . Then for  $\tau \in \Sigma_L$ , the set  $\{ \operatorname{wt}(\chi_i)_{\tau} + h_{j_i, \tau} \mid i = 1, \ldots, r, j_i = s_{i-1} + 1, \ldots, s_i \}$  is the set of the Sen  $\tau$ -weights of  $\rho$ .

Proof. Since  $U_{\Omega,\mathbf{h}}(\overline{\rho})$  is Zariski-open and Zariski-dense in  $X_{\Omega,\mathbf{h}}(\overline{\rho})$  and the Sen  $\tau$ -weights are analytic functions on  $X_{\Omega,\mathbf{h}}(\overline{\rho})$  (cf. [60, Def. 6.2.11]), we only need to prove the statement for points in  $U_{\Omega,\mathbf{h}}(\overline{\rho})$ . Since any point of  $U_{\Omega,\mathbf{h}}(\overline{\rho})$  lies in the image of  $\kappa(\underline{\alpha}, \{\overline{\varrho}_i\})$  for certain  $\underline{\alpha}, \{\overline{\varrho}_i\}$ , using the commutative diagram (4.23) we are reduced to prove the statement for points in  $S_{\overline{\rho}}(\Omega, \mathbf{h}, \underline{\alpha}, \{\overline{\varrho}_i\})$ . For such a point  $x = (\rho, \underline{\varrho}, \chi, \operatorname{Fil}_{\bullet}, \underline{\nu})$ , by definition, the filtration  $\operatorname{Fil}_{\bullet} D_{\operatorname{rig}}(\rho)$  satisfies  $\operatorname{gr}_i D_{\operatorname{rig}}(\rho) \cong$  $D_{\operatorname{rig}}(\varrho_{i,x}) \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x),L}(\chi_i \operatorname{unr}(\alpha_i))$ . We see that  $\{\operatorname{wt}(\chi_i)_{\tau} + h_{j,\tau}\}_{j=s_{i-1}+1,\ldots,s_i}$  is the set of Sen  $\tau$ -weight of  $\operatorname{gr}_i D_{\operatorname{rig}}(\rho)$ . The proposition follows.

#### 4.3 Potentially crystalline deformation spaces

We study a variant of potentially crystalline deformation spaces, and we show it admits a cell decomposition with respect to Schubert cells of  $\operatorname{GL}_n/P$ . By studying the embeddings of the cells into our (various) Bernstein paraboline varieties, we prove the existence of local companion points on the Bernstein paraboline varieties for generic potentially crystalline representations.

We first recall some facts on inertial types. Let  $d \in \mathbb{Z}_{\geq 1}$ ,  $\xi : I_L \to \operatorname{GL}_d(E)$  a cuspidal inertial type and  $\mathbf{r}_d$  an absolutely irreducible Weil-Deligne representation over E such that  $\mathbf{r}_d|_{I_L} \cong \xi$ . Assume E contains all d-th roots of unity  $\mu_d$ . Let L' be a finite extension of L such that the action of  $I_L$  on  $\xi$  factors through the inertia subgroup  $I(L'/L) \subset \operatorname{Gal}(L'/L)$ . Using  $\mathbf{r}_d \cong \mathbf{r}_d \otimes_E \operatorname{unr}(\alpha) \Rightarrow$  $\wedge^d \mathbf{r}_d \otimes_E \operatorname{unr}(\alpha^d) = \wedge^d \mathbf{r}_d$ , there exists  $d_0|d$  such that  $\{\alpha \in E^{\times} \mid \mathbf{r}_d \cong \mathbf{r}_d \otimes_E \operatorname{unr}(\alpha)\} = \mu_{d_0}$ . Note that  $d_0$  only depends on the inertial type  $\xi$ .

Let  $\xi_0 \subseteq \xi$  be an absolutely irreducible subrepresentation, and fix  $F \in W_L$  a lifting of the arithmetic Frobenius. We denote by  $F(\xi_0)$  the conjugate of  $\xi_0$  by F (so it is an absolutely irreducible representation of I(L'/L)). The following lemma follows from [26, Lemma 4.4].

**Lemma 4.3.1.** The integer  $d_0$  is the minimal positive integer such that  $F^{d_0}(\xi_0) \subseteq \xi_0$ . Moreover, we have  $\xi \cong \bigoplus_{i=0}^{d_0-1} F^i(\xi_0)$  as I(L'/L)-representations and the  $F^i(\xi_0)$ ,  $i = 0, \ldots, d_0 - 1$  are pairwise non-isomorphic.

We fix a basis  $\underline{e} = (\underline{e}_0, \dots, \underline{e}_{d_0-1})$  of  $\xi$  such that  $\underline{e}_i$  is a basis of  $\xi_i := F^i(\xi_0)$ . Let  $\mathbf{r}_1, \mathbf{r}_2$  be two Weil-Deligne representations of inertial type  $\xi$  and fix I(L'/L)-equivariant isomorphisms for  $i \in \{1, 2\}$ :

$$E\underline{e} = \xi \xrightarrow{\sim} \mathbf{r}_i. \tag{4.26}$$

Consider the operators  $\mathbf{r}_i(F)$  acting on (the underlying vector space of)  $\xi$  via the above isomorphisms. Then  $\mathbf{r}_2(F) \circ \mathbf{r}_1(F)^{-1}$ :  $\xi \to \xi$  is I(L'/L)-equivariant, hence  $\mathbf{r}_2(F) \circ \mathbf{r}_1(F)^{-1}$  preserves  $\xi_i$  for  $i = 0, \ldots, d_0 - 1$  and is equal to a scalar  $\alpha_i \in E^{\times}$  when restricted to  $\xi_i$  (note that  $\alpha_i$  depends on the choices of the isomorphisms (4.26)). Let  $\alpha := \prod_{i \in \mathbb{Z}/d_0} \alpha_i$ , we have  $\mathbf{r}_2(F^{d_0}) \circ \mathbf{r}_1(F^{d_0})^{-1} = \alpha$  on each  $\xi_i$  hence on  $\xi$ . Moreover  $\alpha$  is independent of the choice of the isomorphisms (4.26).

**Lemma 4.3.2.** Let  $\beta \in E$  be a  $d_0$ -th root of  $\alpha$  (enlarging E if necessary), then  $\mathbf{r}_2 \cong \mathbf{r}_1 \otimes_E \operatorname{unr}(\beta)$ .

*Proof.* Since  $\mathbf{r}_2$  and  $\mathbf{r}_1$  have the same cuspidal inertial type, there exists  $\beta'$  such that  $\mathbf{r}_2 \cong \mathbf{r}_1 \otimes_E \operatorname{unr}(\beta')$ . Note that  $\beta'$  is unique up to multiplication by an element in  $\mu_{d_0}$ . It follows that  $\mathbf{r}_2(F^{d_0}) = \mathbf{r}_1(F^{d_0})(\beta')^{d_0}$ , hence  $(\beta')^{d_0} = \alpha$ .

We now go back to the setting of § 4.2. We let  $\Omega_0$  be the Bernstein component of  $\operatorname{GL}_n(L)$ associated to the cuspidal Bernstein component  $\Omega$  of  $L_P(L)$ , i.e.  $\Omega_0$  is the Bernstein component with cuspidal type  $(L_P(L), \pi_{L_P} \cong \boxtimes_{i=1}^r \pi_i)$ . Let  $\mathcal{Z}_{\Omega_0}$  be the centre of  $\Omega_0$ , and  $\xi_i$  be the inertial type of  $\Omega_i$  for  $i = 0, \ldots, r$  (note that  $\xi_0$  is different here from the  $\xi_0$  of Lemma 4.3.1). We have thus  $\xi_0 \cong \bigoplus_{i=1}^r \xi_i$ . Moreover, by [33, Prop. 4.1] (see also [26, § 3.6]), there is a one-to-one correspondence between closed points of Spec  $\mathcal{Z}_{\Omega_0}$  and semi-simple Weil-Deligne representations **r** of inertial type  $\xi_0$  (so N = 0 on **r** and  $\mathbf{r} \cong \bigoplus_{i=1}^r \mathbf{r}_i$  for certain absolutely irreducible Weil-Deligne representations  $\mathbf{r}_i$  of inertial type  $\xi_i$ ). Let  $\mathcal{W}_{\Omega} := \{w \in S_r \mid \Omega_i = \Omega_{w^{-1}(i)}, \forall i = 1, \ldots, r\}$ . By [33, Prop. 2.1], there is a natural isomorphism of *E*-algebras

$$\mathcal{Z}_{\Omega}^{\mathscr{W}_{\Omega}} \xrightarrow{\sim} \mathcal{Z}_{\Omega_{0}}.$$

On closed points, the corresponding map  $\operatorname{Spec} \mathcal{Z}_{\Omega} \to (\operatorname{Spec} \mathcal{Z}_{\Omega})/\mathcal{W}_{\Omega} \cong \operatorname{Spec} \mathcal{Z}_{\Omega_0}$  sends the *r*-tuple  $(\mathbf{r}_1, \ldots, \mathbf{r}_r)$  to  $\bigoplus_{i=1}^r \mathbf{r}_i$ .

Let  $\{\tilde{\xi}_1, \ldots, \tilde{\xi}_s\}$  be the set of isomorphic classes of  $\{\xi_1, \ldots, \xi_r\}$ , and for  $j \in \{1, \ldots, s\}$  let  $m_j := |\{i \mid \xi_i \cong \tilde{\xi}_j\}|$ . Thus  $\sum_{j=1}^s m_j = r$ ,  $\mathscr{W}_{\Omega} \cong \prod_{j=1}^s S_{m_j}$ , and we have an isomorphism  $\xi_0 \cong \bigoplus_{j=1}^s \tilde{\xi}_j^{\oplus m_j}$ . By Lemma 4.3.1, we have a decomposition  $\tilde{\xi}_j \cong \bigoplus_{k \in \mathbb{Z}/d_j \mathbb{Z}} \tilde{\xi}_{j,k}$  for some integer  $d_j \ge 1$  dividing  $\dim_E \tilde{\xi}_j$ , where  $\tilde{\xi}_{j,k} := F^k(\tilde{\xi}_j)$  for  $j = 1, \ldots, s$  and  $k \in \mathbb{Z}/d_j \mathbb{Z}$  are pairwise nonisomorphic absolutely irreducible representations of I(L'/L) over E (note that distinct inertial types do not have any common irreducible constituents). Let  $f_j := \dim_E \tilde{\xi}_{j,0}$  ( $= \dim_E \tilde{\xi}_{j,k}$  for all  $k \in \mathbb{Z}/d_j \mathbb{Z}$ ). Hence  $\sum_{j=1}^s m_j d_j f_j = n$ ,  $\sum_{j=1}^s m_j = r$ ,  $d_j f_j = n_i$  if  $\xi_i \cong \tilde{\xi}_j$  and we have a decomposition

$$\xi_0 \cong \bigoplus_{j=1}^{\circ} \Big( \bigoplus_{k \in \mathbb{Z}/d_j \mathbb{Z}} \Big( \underbrace{\widetilde{\xi}_{j,k} \oplus \dots \oplus \widetilde{\xi}_{j,k}}_{m_j} \Big) \Big).$$
(4.27)

In order to describe a Weil representation of inertial type  $\xi_0$  in a more concrete way, we now fix a basis of  $\xi_0$  with respect to the decomposition (4.27):

$$\underline{e} = (\underline{e}_{j,k})_{\substack{j=1,\dots,s\\k\in\mathbb{Z}/d_j\mathbb{Z}}} = (\underline{e}_{j,k,i})_{\substack{j=1,\dots,s\\k\in\mathbb{Z}/d_j\mathbb{Z}\\i=1,\dots,m_j}}$$

where each  $(\underline{e}_{j,k,i})$  for  $i = 1, \ldots, m_j$  means a choice of a basis on the  $f_j$ -dimensional *E*-vector space  $\tilde{\xi}_{j,k}$ . We choose these basis so that the following conditions are satisfied:

**Condition 4.3.3.** (1) For  $1 \le i, i' \le m_j$ , the *E*-linear map  $\tilde{\xi}_{j,k} \to \tilde{\xi}_{j,k}$  sending the basis  $\underline{e}_{j,k,i'}$  to the basis  $\underline{e}_{j,k,i'}$  is I(L'/L)-equivariant.

(2) The E-linear map  $\iota_{j,k,i}: \widetilde{\xi}_{j,k} \to \widetilde{\xi}_{j,k+1}$  sending the basis  $\underline{e}_{j,k,i}$  to the basis  $\underline{e}_{j,k+1,i}$  satisfies

$$\iota_{j,k,i}(gv) = (FgF^{-1})\iota_{j,k,i}(v)$$

for all  $v \in \widetilde{\xi}_{j,k}$  and  $g \in I(L'/L)$ .

Condition (1) is equivalent to the fact that, for any element  $g \in I(L'/L)$ , the matrices of gin the basis  $\underline{e}_{j,k,i}$  and  $\underline{e}_{j,k,i'}$  are the same. Condition (2) is equivalent to the fact that, for any element  $g \in I(L'/L)$ , the matrix of g in the basis  $\underline{e}_{j,k,i}$  is equal to the matrix of  $FgF^{-1}$  in the basis  $\underline{e}_{j,k+1,i}$ .

We fix a semi-simple Weil-Deligne representation  $\mathbf{r}_0 \cong \bigoplus_{j=1}^s \tilde{\mathbf{r}}_j^{\oplus m_j}$  such that  $\tilde{\mathbf{r}}_j$  is of inertial type  $\tilde{\xi}_j$  for all j. So  $\mathbf{r}_0$  is of inertial type  $\xi_0$  and N = 0 on  $\mathbf{r}_0$ . We fix an isomorphism  $\iota_0$  of I(L'/L)-representations:

$$\iota_0: \langle \underline{e} \rangle = \xi_0 \xrightarrow{\sim} \mathbf{r}_0$$
where the notation  $\langle \underline{e} \rangle$  means the *E*-vector spaces generated by the basis  $\underline{e}$ . The *F*-action on  $\mathbf{r}_0$  then gives an endomorphism  $\mathbf{r}_0(F)$  on  $\xi_0$  which sends  $\tilde{\xi}_{j,k}^{\oplus m_j}$  to  $\tilde{\xi}_{j,k+1}^{\oplus m_j}$ . Modifying the isomorphism  $\iota_0$  if necessary and by condition (2) in our choice of the basis  $\underline{e}$  of  $\xi_0$ , we can and do assume

$$\underline{e}_{j,k+1,i} = \mathbf{r}_0(F)\underline{e}_{j,k,i} \text{ for } j = 1, \dots, s \text{ and } k = 0, \dots, d_j - 2.$$

$$(4.28)$$

Let  $\mathbf{r}$  be an arbitrary Weil-Deligne representation of inertial type  $\xi_0$  (with N possibly non-zero on  $\mathbf{r}$ ), and fix again an I(L'/L)-equivariant isomorphism  $\langle \underline{e} \rangle = \xi_0 \xrightarrow{\sim} \mathbf{r}$ . The F-action on  $\mathbf{r}$  gives another endomorphism  $\mathbf{r}(F)$  on  $\xi_0$  sending  $\tilde{\xi}_{j,k}^{\oplus m_j}$  to  $\tilde{\xi}_{j,k+1}^{\oplus m_j}$ . The endomorphism  $\mathbf{r}(F) \circ \mathbf{r}_0(F)^{-1}$ : of  $\xi_0$  is I(L'/L)-equivariant, hence preserves each  $\tilde{\xi}_{j,k}^{\oplus m_j}$  for  $j = 1, \ldots, s$  and  $k \in \mathbb{Z}/d_j\mathbb{Z}$ . Since  $\operatorname{End}_{I(L'/L)}(\tilde{\xi}_{j,k}) \cong E$ , the restriction of  $\mathbf{r}(F) \circ \mathbf{r}_0(F)^{-1}$  to  $\tilde{\xi}_{j,k}^{\oplus m_j}$  is given in the basis  $(\underline{e}_{j,k,i})_{i=1,\ldots,m_j}$ by a matrix  $A_{j,k}$  which lies in the image  $\operatorname{GL}_{m_j}(E\operatorname{Eld}_{f_j})$  of

$$\operatorname{GL}_{m_j}(E) \longleftrightarrow \operatorname{GL}_{m_j f_j}(E), \ (a_{uv})_{1 \le u, v \le m_j} \mapsto (a_{uv} \operatorname{Id}_{f_j})_{1 \le u, v \le m_j}$$
(4.29)

where  $\operatorname{Id}_{f_j} \in \operatorname{GL}_{f_j}(E)$  is the identity matrix. Moreover, for  $j = 1, \ldots, s$  and  $k = 0, \ldots, d_j - 2$ , we see by (4.28) that  $A_{j,k}$  is actually the matrix of the morphism  $\mathbf{r}(F) : \tilde{\xi}_{j,k}^{\oplus m_j} \to \tilde{\xi}_{j,k+1}^{\oplus m_j}$  in the basis  $\underline{e}_{j,k}$  and  $\underline{e}_{j,k+1}$ . Hence the matrix of the endomorphism  $\mathbf{r}(F) \circ \mathbf{r}_0(F)^{-1}$  in the basis  $\underline{e}$  is

$$A = \operatorname{diag}\left(\{A_{j,k}\}_{\substack{j=1,\dots,s\\k\in\mathbb{Z}/d_j\mathbb{Z}}}\right) \in \operatorname{GL}_n(E).$$

$$(4.30)$$

The converse also holds: given a matrix  $A' = \text{diag}(\{A'_{j,k}\})$  as in (4.30) with  $A'_{j,k} \in \text{GL}_{m_j}(E\text{Id}_{f_j})$ , one can associate a Weil-Deligne representation of inertial type  $\xi_0$  with N = 0 by letting F act on the basis  $\underline{e}$  by  $A' \circ \mathbf{r}_0(F)$ .

Now let  $B_{j,k}$  be the preimage of  $A_{j,k}$  via (4.29) and put  $B_j := B_{j,0}B_{j,1}\cdots B_{j,d_j-1} \in \operatorname{GL}_{m_j}(E)$ . Using again (4.28), we see that the image  $A_j$  of  $B_j$  via (4.29) is the matrix of  $\mathbf{r}(F^{d_j}) \circ \mathbf{r}_0(F^{-d_j})|_{\tilde{\xi}_{j,d_j-1}^{\oplus m_j}}$  in the basis  $\underline{e}_{j,d_j-1}$  of  $\tilde{\xi}_{j,d_j-1}^{\oplus m_j}$ . Since both  $\mathbf{r}(F^{d_j})$  and  $\mathbf{r}_0(F^{-d_j})$  preserve  $\tilde{\xi}_{j,k}^{\oplus m_j}$  for all  $k = 0, \ldots, d_j - 1$ ), the conjugacy class of  $A_j$  is independent of the choice of the basis of  $\tilde{\xi}_{j,d_j-1}^{\oplus m_j}$ . We call  $\mathbf{r}$  generic if  $\mathbf{r}^{\mathrm{ss}} \cong \bigoplus_{i=1}^r \mathbf{r}_i$  satisfies  $\mathbf{r}_i \neq \mathbf{r}_{i'}$  and  $\mathbf{r}_i \neq \mathbf{r}_{i'} \otimes_E \operatorname{unr}(q_L)$  for all  $i \neq i'$ . In particular, if  $\mathbf{r}$  is generic, then N = 0 on  $\mathbf{r}$  and  $\mathbf{r} \cong \mathbf{r}^{\mathrm{ss}}$ . Note that all  $\mathbf{r}$  of inertial type  $\xi_0$  are generic if  $\Omega_i \neq \Omega_{i'}$  for all  $i \neq i'$ .

**Lemma 4.3.4.** With the above notation,  $\mathbf{r}$  is generic if and only if, for any  $j = 1, \ldots, s$ , the eigenvalues  $\alpha_{j,1}, \ldots, \alpha_{j,m_j}$  of  $B_j$  satisfy  $\alpha_{j,i} \neq \alpha_{j,i'}$ , and  $\alpha_{j,i} \neq \alpha_{j,i'} q_L^{d_j}$  for  $i \neq i'$ .

*Proof.* Assume **r** is generic, in particular  $\mathbf{r} \cong \bigoplus_{j=1}^{s} \bigoplus_{i=1}^{m_j} \mathbf{r}_{j,i}$ . Let  $\beta_{j,i} \in E^{\times}$  be such that  $\mathbf{r}_{j,i} \cong \tilde{\mathbf{r}}_j \otimes_E$ unr $(\beta_{j,i})$ . We then deduce that there exists a basis  $\underline{e}'_{j,d_j-1}$  of  $\tilde{\xi}_{j,d_j-1}^{\oplus m_j}$  such that the corresponding matrix of the operator  $\mathbf{r}(F^{d_j}) \circ \mathbf{r}_0(F^{-d_j}) |_{\tilde{\xi}_{j,d_j-1}^{\oplus m_j}}$  is equal to

$$\operatorname{diag}(\underbrace{\beta_{j,1}^{d_j},\ldots,\beta_{j,1}^{d_j}}_{f_j},\ldots,\underbrace{\beta_{j,m_j}^{d_j},\ldots,\beta_{j,m_j}^{d_j}}_{f_j}).$$

Thus  $\{\alpha_{j,1}, \ldots, \alpha_{j,m_j}\} = \{\beta_{j,1}^{d_j}, \ldots, \beta_{j,m_j}^{d_j}\}$ , and the "only if" part then follows from definition of genericity. Now assume  $\alpha_{j,1}, \ldots, \alpha_{j,m_j}$  satisfy the conditions in the lemma, in particular, are

distinct. By comparing dimensions, we easily see that, for each  $\alpha_{j,i}$ , the subspace  $\mathbf{r}_{j,i,d_j-1}$  of  $\tilde{\xi}_{j,d_j-1}^{\oplus m_j}$  on which  $\mathbf{r}(F^{d_j}) \circ \mathbf{r}_0(F^{-d_j})$  acts via  $\alpha_{j,i}$  is isomorphic to  $\tilde{\xi}_{j,d_j-1}$ . Any subrepresentation  $\tilde{\xi}_{j,d_j-1}$  in  $\tilde{\xi}_{j,d_j-1}^{\oplus m_j}$  is preserved by  $\mathbf{r}_0(F^{-d_j})$ . We deduce then  $\mathbf{r}_{j,i,d_j-1}$  is preserved by  $\mathbf{r}(F^{d_j})$ . Under the  $\mathbf{r}(F)$ -action,  $\mathbf{r}_{j,i,d_j-1}$  then generates an irreducible Weil-Deligne subrepresentation  $\mathbf{r}_{j,i}$  of  $\mathbf{r}$  of inertial type  $\tilde{\xi}_j$ . By Lemma 4.3.2,  $\mathbf{r}_{j,i} \cong \tilde{\mathbf{r}}_j \otimes_E \operatorname{unr}(\beta_{j,i})$  for any  $d_j$ -th root  $\beta_{j,i}$  of  $\alpha_{j,i}$ . By the conditions on  $\{\alpha_{j,i}\}$ , we see that  $\mathbf{r} \cong \bigoplus_{j=1}^s \bigoplus_{i=1}^{m_j} \mathbf{r}_{j,i}$  and that  $\mathbf{r}$  is generic.

Fix  $\mathbf{h} = (h_{i,\tau})_{\substack{i=1,\dots,n\\\tau\in\Sigma_L}} \in \mathbb{Z}^{n[L:\mathbb{Q}_p]}$  with  $h_{i,\tau} > h_{i+1,\tau}$ . Consider  $\mathfrak{X}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h}) := (\operatorname{Spf} R_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h}))^{\mathrm{rig}}$ .

By [26, Thm. 4.1], there is a natural morphism

$$\mathfrak{X}^{\mathrm{pcr}}_{\overline{\rho}}(\xi_0, \mathbf{h}) \longrightarrow (\operatorname{Spec} \mathcal{Z}_{\Omega_0})^{\mathrm{rig}} \cong (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\mathrm{rig}} / \mathscr{W}_{\Omega}$$

which, pointwise, sends  $\rho$  to the semi-simple Weil-Deligne representation associated to  $\rho$ . Let

$$\widetilde{\mathfrak{X}}^{\mathrm{pcr}}_{\overline{\rho}}(\xi_0, \mathbf{h}) := \mathfrak{X}^{\mathrm{pcr}}_{\overline{\rho}}(\xi_0, \mathbf{h}) \times_{(\operatorname{Spec} \mathcal{Z}_{\Omega_0})^{\mathrm{rig}}} (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\mathrm{rig}}$$

so a point of  $\widetilde{\mathfrak{X}}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$  is of the form  $(\rho, (\mathbf{r}_i))$  with  $\mathbf{r}(\rho)^{\mathrm{ss}} \cong \bigoplus_{i=1}^r \mathbf{r}_i$ . In particular, the *r*-tuple  $(\mathbf{r}_i)$  induces an  $\Omega$ -filtration on  $D_{\mathrm{rig}}(\rho)$ . Let  $U_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$  (resp.  $\widetilde{U}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$ ) be the set of points  $\rho \in \mathfrak{X}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$  (resp.  $(\rho, (\mathbf{r}_i)) \in \widetilde{\mathfrak{X}}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$ ) such that  $\mathbf{r}(\rho)$  is generic.

**Proposition 4.3.5.** The set  $U_{\overline{\rho}}^{\text{pcr}}(\xi_0, h)$  is Zariski-open and Zariski-dense in  $\mathfrak{X}_{\overline{\rho}}^{\text{pcr}}(\xi_0, h)$ .

Proof. Let U be an arbitrary (non-empty) connected admissible affinoid open of  $\mathfrak{X}_{\rho}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$ . A major part of the proof is to show  $U_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h}) \cap U$  is Zariski-open in U. We will show that a certain torsor  $\widetilde{U}$  over U admits a smooth morphism to  $\left(\prod_{\substack{j=1,...,s\\k\in\mathbb{Z}/d_j\mathbb{Z}}} \mathrm{GL}_{m_j}\right) \times \mathrm{Res}_{\mathbb{Q}_p}^L(\mathrm{GL}_n/B)$  (cf. (4.36)), where, roughly speaking, the morphism to the first factor sends  $\rho$  to the matrices  $\{B_{j,k}\}$  associated to  $\mathbf{r}(\rho)$  as above, and the morphism to the second factor sends  $\rho$  to the Hodge filtration on  $D_{\mathrm{dR}}(\rho)$ . The torsor that we will use comes from choices of basis on the corresponding objects.

Recall that by [63] we have the following data:

- (1) a rank *n* locally free  $L'_0 \otimes_{\mathbb{Q}_p} \mathcal{O}_U \cong \prod_{\tau \in \Sigma_{L'_0}} \mathcal{O}_U$ -module  $\mathrm{DF}_U \cong \prod_{\tau \in \Sigma_{L'_0}} \mathrm{DF}_{U,\tau}$  equipped with a semi-linear action of  $(\mathrm{Gal}(L'/L), \varphi)$ ;
- (2) a decreasing filtration  $\operatorname{Fil}^{\bullet} = \prod_{\tau \in \Sigma_L} \operatorname{Fil}_{\tau}^{\bullet}$  on the rank *n* locally free  $L \otimes_{\mathbb{Q}_p} \mathcal{O}_U \cong \prod_{\tau \in \Sigma_L} \mathcal{O}_U$ module  $\mathcal{D}_U := (\operatorname{DF}_U \otimes_{L'_0} L')^{\operatorname{Gal}(L'/L)} \cong \prod_{\tau \in \Sigma_L} \mathcal{D}_{U,\tau}$  by  $L \otimes_{\mathbb{Q}_p} \mathcal{O}_U$ -submodules which are direct summands of  $\mathcal{D}_U$  as  $\mathcal{O}_U$ -module such that (letting  $-h_{0,\tau} = -\infty$  and  $-h_{n+1,\tau} = +\infty$ )

$$\operatorname{rk}_{\mathcal{O}_U}\operatorname{Fil}^i_{\tau}\mathcal{D}_{U,\tau} = n - j + 1, \text{ for } -h_{j-1,\tau} < i \leq -h_{j,\tau}.$$

Moreover, for any point  $x \in U$ , the specialization of  $(DF_U, \mathcal{D}_U)$  at x is equal to  $(D_{pst}(\rho_x), D_{dR}(\rho_x))$ where  $\rho_x$  is the associated Gal<sub>L</sub>-representation. Shrinking U, we can and do assume that  $DF_U$ and  $\mathcal{D}_U$  are free over  $\mathcal{O}_U$ . We fix an embedding  $\tau_0 : L_0 \hookrightarrow E$  and put  $\tau_i := \tau_0 \circ \text{Frob}^{-i}$  where Frob is the (absolute arithmetic) Frobenius on the Witt vectors. Then we have a decomposition

$$L'_0 \otimes_{\mathbb{Q}_p} E \xrightarrow{\sim} \bigoplus_{i \in \mathbb{Z}/[L_0:\mathbb{Q}_p]\mathbb{Z}} \left( \bigoplus_{\substack{\tau \in \Sigma_{L'_0} \\ \tau|_{L_0} = \tau_i}} E \right).$$

and a corresponding decomposition  $DF_U \cong \bigoplus_{i \in \mathbb{Z}/[L_0:\mathbb{Q}_p]\mathbb{Z}} DF_{U,i}$ , where  $DF_{U,i}$  is free over  $L'_0 \otimes_{L_0,\tau_i}$  $\mathcal{O}_U$  and preserved by  $\operatorname{Gal}(L'/L)$ . Shrinking U and enlarging E if necessary (and using  $H^i(H, M) = 0$  for all i > 0, finite groups H and H-modules M over  $\mathbb{Q}$ ), we can and do assume that there is a semi-linear  $\operatorname{Gal}(L'/L)$ -representation  $V_0$ , free of rank n over  $L'_0 \otimes_{L_0,\tau_0} E$ , such that we have a  $\operatorname{Gal}(L'/L)$ -equivariant isomorphism

$$V_0 \otimes_E \mathcal{O}_U \xrightarrow{\sim} \mathrm{DF}_{U,0}$$
. (4.31)

For  $i \in \mathbb{Z}/[L_0 : \mathbb{Q}_p]\mathbb{Z}$ , let  $\operatorname{Fr}^i : L'_0 \otimes_{L_0,\tau_0} E \to L'_0 \otimes_{L_0,\tau_0 \circ \operatorname{Frob}^{-i}} E$  be the isomorphism of algebras sending  $a \otimes b$  to  $\operatorname{Frob}^i(a) \otimes b$ . The morphism  $\operatorname{Fr}^i$  is  $\operatorname{Gal}(L'/L)$ -equivariant, where  $\operatorname{Gal}(L'/L)$  acts on both sides via its natural action on the factor  $L'_0$ . Put

$$V_i := V_0 \otimes_{L'_0 \otimes_{L_0,\tau_0} E, \operatorname{Fr}^i} (L'_0 \otimes_{L_0,\tau_i} E), \qquad (4.32)$$

which is a semi-linear  $\operatorname{Gal}(L'/L)$ -representation over  $L'_0 \otimes_{L_0,\tau_i} E$ . The action of  $\varphi^i$  on  $\operatorname{DF}_U$  sends  $\operatorname{DF}_{U,0}$  to  $\operatorname{DF}_{U,i}$ . For  $i = 0, \ldots, [L_0 : \mathbb{Q}_p] - 1$ , the composition  $V_0 \otimes_E \mathcal{O}_U \xrightarrow{\sim} \operatorname{DF}_{U,0} \xrightarrow{\varphi^i} \operatorname{DF}_{U,i}$ induces an  $L'_0 \otimes_{L_0,\tau_i} \mathcal{O}_U$ -semilinear  $\operatorname{Gal}(L'/L)$ -equivariant isomorphism

$$V_i \otimes_E \mathcal{O}_U \xrightarrow{\sim} \mathrm{DF}_{U,i}$$
. (4.33)

For each  $\tau \in \Sigma_{L'_0}$  with  $\tau|_{L_0} = \tau_0$ , we have  $V_{0,\tau} \cong \xi_0$  as I(L'/L)-representation over E. The basis  $\underline{e}$  of  $\xi_0$  (that we previously fixed) then gives a basis  $\underline{e}_{\tau} = (e_{\tau,1}, \ldots, e_{\tau,n})$  of  $V_{0,\tau}$  satisfying Condition 4.3.3 for the I(L'/L)-action. Then  $(\underline{e}_{\tau})_{\tau|_{L_0}=\tau_0}$  form a basis of  $V_0$  over E. Choosing the basis  $(\underline{e}_{\tau})_{\tau|_{L_0}=\tau_0}$  is the same as choosing a basis of  $V_0$ , formed by bases of  $V_{0,\tau}$  for each  $\tau$  on which the I(L'/L)-action satisfies Condition 4.3.3. For  $i = 1, \ldots, n$ , let  $\tilde{e}_i := (e_{\tau,i})_{\tau|_{L_0}=\tau_0} \in V_0$ . Then  $\underline{\tilde{e}} = (\tilde{e}_1, \ldots, \tilde{e}_n)$  is a basis of  $V_0$  over  $L'_0 \otimes_{L_0,\tau_0} E$ , which also gives a basis of  $V_i$  over  $L'_0 \otimes_{L_0,\tau_i} E$  by  $- \otimes 1$  (see (4.32)). For  $i \in \mathbb{Z}/[L_0 : \mathbb{Q}_p]\mathbb{Z}$ , there exists thus  $P_i \in \mathrm{GL}_n(L' \otimes_{L_0,\tau_i} E)$  such that  $(\mathfrak{e}_1, \ldots, \mathfrak{e}_n) := (\tilde{e}_1, \ldots, \tilde{e}_n)P_i$  is a basis of  $(V_i \otimes_{L'_0} L')^{\mathrm{Gal}(L'/L)}$  over  $L \otimes_{L_0,\tau_i} E$ . For  $\tau \in \Sigma_L$  and  $i \in \mathbb{Z}/[L_0 : \mathbb{Q}_p]\mathbb{Z}$  such that  $\tau|_{L_0} = \tau_i$ , we let  $\underline{\mathfrak{e}}_{\tau} := (\mathfrak{e}_{1,\tau}, \ldots, \mathfrak{e}_{n,\tau})$  be the  $\tau$ -factor of  $\underline{\mathfrak{e}}$ , which is a basis of  $((V_i \otimes_{L'_0} L')^{\mathrm{Gal}(L'/L)})_{\tau}$  over E.

Let  $\mathscr{G}$  be the affine subgroup  $\operatorname{Aut}_{\operatorname{Gal}(L'/L)}(V_0)$  of  $\operatorname{Aut}(V_0) \cong \operatorname{Res}_{L_0}^{L'_0} \operatorname{GL}_n \times_{\operatorname{Spec} L_0,\tau_0} \operatorname{Spec} E$  where "Aut" means  $L'_0 \otimes_{L_0,\tau_0} E$ -linear bijections. Note that  $\mathscr{G}$  is smooth by Cartier's theorem. Let  $\widetilde{U}$  be the  $\mathscr{G}$ -torsor over U of  $\operatorname{Gal}(L'/L)$ -equivariant isomorphisms in (4.31). Define

$$\mathrm{DF}_{\widetilde{U}} = \bigoplus_{i \in \mathbb{Z}/[L_0:\mathbb{Q}_p]\mathbb{Z}} \mathrm{DF}_{\widetilde{U},i} := \bigoplus_{i \in \mathbb{Z}/[L_0:\mathbb{Q}_p]\mathbb{Z}} \mathrm{DF}_{U,i} \otimes_{\mathcal{O}_U} \mathcal{O}_{\widetilde{U}}$$

equipped with the action of  $(\varphi, \operatorname{Gal}(L'/L))$  by extension of scalars. By definition, we have a  $\operatorname{Gal}(L'/L)$ -equivariant isomorphism of  $\mathcal{O}_{\widetilde{U}}$ -modules  $V_0 \otimes_E \mathcal{O}_{\widetilde{U}} \xrightarrow{\sim} \operatorname{DF}_{\widetilde{U},0}$ , which induces  $\operatorname{Gal}(L'/L)$ -equivariant isomorphisms  $V_i \otimes_E \mathcal{O}_{\widetilde{U}} \xrightarrow{\sim} \operatorname{DF}_{\widetilde{U},i}$  for  $i \in \mathbb{Z}/[L_0 : \mathbb{Q}_p]\mathbb{Z}$  similarly as for (4.33). These isomorphisms then induce isomorphisms

$$(V_i \otimes_{L'_0} L')^{\operatorname{Gal}(L'/L)} \otimes_E \mathcal{O}_{\widetilde{U}} \xrightarrow{\sim} (\mathrm{DF}_{\widetilde{U},i} \otimes_{L'_0} L')^{\operatorname{Gal}(L/L)}$$
(4.34)

(use that the left hand side of (4.34) is a direct summand of  $(V_i \otimes_{L'_0} L') \otimes_E \mathcal{O}_{\widetilde{U}} \cong DF_{\widetilde{U},i} \otimes_{L'_0} L$  to see that (4.34) is an isomorphism). Let  $\mathcal{D}_{\widetilde{U}} := \mathcal{D}_U \otimes_{\mathcal{O}_U} \mathcal{O}_{\widetilde{U}} \cong (DF_{\widetilde{U}} \otimes_{L'_0} L')^{\operatorname{Gal}(L'/L)}$ , and  $\mathcal{D}_{\widetilde{U},\tau}$  be its  $\tau$ -factor for  $\tau \in \Sigma_L$ . Using the isomorphism (4.34), we obtain a basis  $\underline{\mathfrak{e}}_{\tau} \otimes 1$  of  $\mathcal{D}_{\widetilde{U},\tau}$  over  $\mathcal{O}_{\widetilde{U}}$  for all

 $\tau \in \Sigma_L$ . We denote by  $\operatorname{Fil}_{\tau}^{\bullet}$  the filtration on  $\mathcal{D}_{\widetilde{U},\tau}$  induced by the corresponding filtration on  $\mathcal{D}_{U,\tau}$  by extension of scalars. With respect to the basis  $\underline{\mathfrak{e}}_{\tau} \otimes 1$ ,  $\operatorname{Fil}_{\tau}^{\bullet}$  gives a flag  $\mathscr{F}_{\widetilde{U},\tau} \in (\operatorname{GL}_n/B)(\mathcal{O}_{\widetilde{U}})$ . Taking all embeddings  $\tau \in \Sigma_{\tau}$ , we thus obtain a morphism

$$\widetilde{U} \longrightarrow \operatorname{Res}_{\mathbb{Q}_p}^L(\operatorname{GL}_n/B).$$
 (4.35)

We let  $F \in W_L$  act on  $DF_{\widetilde{U},0}$  via  $\varphi^{-[L_0:\mathbb{Q}_p]} \circ \overline{F}$ , where  $\overline{F}$  denotes the image of F in Gal(L'/L). We have a decomposition

$$\mathrm{DF}_{\widetilde{U},0} \cong \bigoplus_{\substack{\tau \in \Sigma_{L'_0} \\ \tau \mid_{L_0} = \tau_0}} \mathrm{DF}_{\widetilde{U},\tau}$$

and each  $\mathrm{DF}_{\widetilde{U},\tau}$  is preserved by F and I(L'/L). For any  $\tau \in \Sigma_{L'_0}$  such that  $\tau|_{L_0} = \tau_0$ ,  $\mathrm{DF}_{\widetilde{U},\tau}$ equipped with F and the action of I(L'/L) gives a Weil-Deligne representation of  $W_L$  over  $\mathcal{O}_{\widetilde{U}}$ (which factors through  $W_L/I_{L'}$  and is independent of the choice of  $\tau$ ). Using (4.31), the action of F on  $\mathrm{DF}_{\widetilde{U},\tau}$  induces an operator denoted by  $\mathbf{r}_{\widetilde{U}}(F)$  on  $V_{0,\tau} \otimes_E \mathcal{O}_{\widetilde{U}}$ . On the other hand, we have the operator  $\mathbf{r}_0(F)$  on  $V_{0,\tau} \cong \xi_0$  given as in the discussion above (4.28), and we still denote by  $\mathbf{r}_0(F)$  its extension of scalars on  $V_{0,\tau} \otimes_E \mathcal{O}_{\widetilde{U}}$ . Similarly as in the discussion above (4.29) and using

$$\operatorname{Hom}_{I(L'/L)}\left(\widetilde{\xi}_{j,k}\otimes_{E}\mathcal{O}_{\widetilde{U}},\widetilde{\xi}_{j',k'}\otimes_{E}\mathcal{O}_{\widetilde{U}}\right) = \begin{cases} \mathcal{O}_{\widetilde{U}} & j=j', k=k'\\ 0 & \text{otherwise,} \end{cases}$$

we see that the operator  $\mathbf{r}_{\widetilde{U}}(F) \circ \mathbf{r}_0(F)^{-1}$  corresponds to a matrix  $A_{\widetilde{U}} = \operatorname{diag}(\{A_{\widetilde{U},j,k}\}_{\substack{j=1,\ldots,s\\k\in\mathbb{Z}/d_j\mathbb{Z}}})$ where  $A_{\widetilde{U},j,k}$  is the matrix of the  $\mathcal{O}_{\widetilde{U}}$ -linear endomorphism

$$\mathbf{r}_{\widetilde{U}}(F) \circ \mathbf{r}_{0}(F)^{-1} : \widetilde{\xi}_{j,k}^{\oplus m_{j}} \otimes_{E} \mathcal{O}_{\widetilde{U}} \xrightarrow{\sim} \widetilde{\xi}_{j,k}^{\oplus m_{j}} \otimes_{E} \mathcal{O}_{\widetilde{U}}$$

and is the image of a matrix  $B_{\widetilde{U},j,k} \in \mathrm{GL}_{m_j}(\mathcal{O}_{\widetilde{U}})$  via the morphism  $\mathrm{GL}_{m_j}(\mathcal{O}_{\widetilde{U}}) \hookrightarrow \mathrm{GL}_{m_jf_j}(\mathcal{O}_U)$ ,  $(a_{uv}) \mapsto (a_{uv}\mathrm{I}_{f_j})$ . The matrices  $\{B_{\widetilde{U},j,k}\}$  give rise to a morphism

$$\widetilde{U} \longrightarrow \prod_{\substack{j=1,\dots,s\\k \in \mathbb{Z}/d_j\mathbb{Z}}} \operatorname{GL}_{m_j}$$

Together with (4.35), we finally obtain a morphism

$$f: \widetilde{U} \longrightarrow \left(\prod_{\substack{j=1,\dots,s\\k \in \mathbb{Z}/d_j\mathbb{Z}}} \operatorname{GL}_{m_j}\right) \times \operatorname{Res}_{\mathbb{Q}_p}^L(\operatorname{GL}_n/B).$$
(4.36)

We prove that f is smooth. Since both source and target are smooth over E (using [63, Thm. 3.3.8] for the source), we only need to show that the tangent map of f at any point of  $\tilde{U}$  is surjective. Let  $x \in \tilde{U}$  and  $f(x) = (\{B_{x,j,k}\}, \{\mathscr{F}_{x,\tau}\})$ . Let u be an element in the tangent space of the right of (4.36) at f(x), and denote the corresponding element by:

$$(\{\widetilde{B}_{x,j,k}\}, \{\widetilde{\mathscr{F}}_{x,\tau}\}) \in \left( \left(\prod_{\substack{j=1,\dots,s\\k\in\mathbb{Z}/d_j\mathbb{Z}}} \operatorname{GL}_{m_j}\right) \times \operatorname{Res}_{\mathbb{Q}_p}^L(\operatorname{GL}_n/B) \right) (k(x)[\varepsilon]/\varepsilon^2).$$

Let  $\widetilde{A} := \operatorname{diag}(\{\widetilde{A}_{x,j,k}\}_{\substack{j=1,\dots,s\\k\in\mathbb{Z}/d_j\mathbb{Z}}}) \in \operatorname{GL}_n(k(x)[\varepsilon]/\varepsilon^2)$  where  $\widetilde{A}_{x,j,k}$  is the image of  $\widetilde{B}_{x,j,k}$  via  $\operatorname{GL}_{m_j}(k(x)[\varepsilon]/\varepsilon^2) \hookrightarrow \operatorname{GL}_{m_jf_j}(k(x)[\varepsilon]/\varepsilon^2), (a_{uv}) \mapsto (a_{uv}\mathrm{I}_{f_j})$ . We use  $\widetilde{A}$  to construct a Deligne-Fontaine module  $\widetilde{\mathrm{DF}}_x$  over  $L'_0 \otimes_{\mathbb{Q}_p} k(x)[\varepsilon]/\varepsilon^2$ . Let  $\widetilde{\mathrm{DF}}_x := (\bigoplus_{i\in\mathbb{Z}/[L_0:\mathbb{Q}_p]\mathbb{Z}}V_i) \otimes_E k(x)[\varepsilon]/\varepsilon^2$ , we have a decomposition  $\widetilde{\mathrm{DF}}_x \cong \bigoplus_{\tau\in\Sigma_{L'_0}} \widetilde{\mathrm{DF}}_{x,\tau}$ . To get a operator  $\varphi$  on the whole  $\widetilde{\mathrm{DF}}_x$ , we only need to construct  $\varphi^i : \widetilde{\mathrm{DF}}_\tau \to \widetilde{\mathrm{DF}}_{\tau\circ\mathrm{Frob}^{-i}}$  for one  $\tau \in \Sigma_{L'_0}$  and for all  $i = 1, \dots, [L'_0 : \mathbb{Q}_p]$ . Recall we have fixed a basis  $\underline{e}_\tau$  of each  $\widetilde{\mathrm{DF}}_\tau$ . For  $i = 1, \dots, [L'_0 : L_0]$ , let  $M_i \in \mathrm{GL}_n(E)$  such that  $\overline{F}(e_{\tau,1}, \dots, e_{\tau,n}) = (e_{\tau\circ\mathrm{Frob}_{L_0}^{-i}, 1}, \dots, e_{\tau\circ\mathrm{Frob}_{L_0}^{-i}, n})M_i$  and put:

$$\varphi^{[L_0:\mathbb{Q}_p]i}:\widetilde{\mathrm{DF}}_{\tau}\longrightarrow\widetilde{\mathrm{DF}}_{\tau\circ\mathrm{Frob}_{L_0}^{-i}},\ (e_{\tau,1},\ldots,e_{\tau,n})\longmapsto \left(e_{\tau\circ\mathrm{Frob}_{L_0}^{-i},1},\ldots,e_{\tau\circ\mathrm{Frob}_{L_0}^{-i},n}\right)M_i(\widetilde{A}M_0)^{-i}.$$

For  $i = 1, \ldots, [L'_0 : \mathbb{Q}_p]$ , writing  $i = i_0[L_0 : \mathbb{Q}_p] + j$  with  $0 \le j < [L_0 : \mathbb{Q}_p]$ , we define

$$\varphi^{i}: \widetilde{\mathrm{DF}}_{\tau} \longrightarrow \widetilde{\mathrm{DF}}_{\tau \circ \mathrm{Frob}^{-i}}, \ (e_{\tau,1}, \dots, e_{\tau,n}) \longmapsto (e_{\tau \circ \mathrm{Frob}^{-i},1}, \dots, e_{\tau \circ \mathrm{Frob}^{-i},n}) M_{i_{0}}(\widetilde{A}M_{0})^{-i_{0}}.$$

Thus  $\widetilde{\mathrm{DF}}_x$  equipped with the  $(\mathrm{Gal}(L'/L), \varphi)$ -action is a Deligne-Fontaine module over  $L'_0 \otimes_{\mathbb{Q}_p} k(x)[\varepsilon]/\varepsilon^2$  and  $\widetilde{\mathrm{DF}}_x \cong x^* \mathrm{DF}_{\widetilde{U}} \equiv D_{\mathrm{pst}}(\rho_x) \pmod{\varepsilon}$ . Put:

$$\widetilde{\mathcal{D}}_x := (\widetilde{\mathrm{DF}}_x \otimes_{L'_0} L')^{\mathrm{Gal}(L'/L)} \cong \bigoplus_{\tau \in \Sigma_L} \widetilde{\mathcal{D}}_{x,\tau}.$$

For each  $\tau$ ,  $\underline{\mathfrak{e}}_{\tau}$  form a basis of  $\widetilde{\mathcal{D}}_{x,\tau}$  over  $k(x)[\varepsilon]/\varepsilon^2$ . Using the basis  $\underline{\mathfrak{e}}_{\tau}$ , we associate to the flag  $\widetilde{\mathscr{F}}_{x,\tau}$  a decreasing filtration  $\operatorname{Fil}_{\tau}^i$  on  $\widetilde{\mathcal{D}}_{x,\tau}$  satisfying

$$\operatorname{rk}_{k(x)[\varepsilon]/\varepsilon^2}(\operatorname{Fil}_{\tau}^i \widetilde{\mathcal{D}}_{x,\tau}) = n - j + 1 \text{ for } -h_{j-1,\tau} < i \leq -h_{j,\tau}.$$

We obtain a filtered  $(\varphi, \operatorname{Gal}(L'/L))$ -module  $(\widetilde{\operatorname{DF}}_x, \widetilde{\mathcal{D}}_x)$  which is a deformation of  $(D_{\mathrm{pst}}(\rho_x), D_{\mathrm{dR}}(\rho_x))$  over  $k(x)[\varepsilon]/\varepsilon^2$ . Thus  $(\widetilde{\operatorname{DF}}_x, \widetilde{\mathcal{D}}_x)$  is weakly admissible, and by [32] is isomorphic to  $(D_{\mathrm{pst}}(\widetilde{\rho}_x), D_{\mathrm{dR}}(\widetilde{\rho}_x))$  for a certain deformation  $\widetilde{\rho}_x$  of  $\rho_x$  over  $k(x)[\varepsilon]/\varepsilon^2$ . By choosing an appropriate basis of  $\widetilde{\rho}_x$  over  $k(x)[\varepsilon]/\varepsilon^2$ , we see that

$$(\widetilde{\rho}_x, V_0 \otimes_E k(x)[\varepsilon]/\varepsilon^2 \xrightarrow{\sim} \mathrm{DF}(\widetilde{\rho}_x)_0 \cong \widetilde{\mathrm{DF}}_{x,0})$$

gives an element in the tangent space of  $\widetilde{U}$  at x which is sent to u.

The conditions in Lemma 4.3.4 cut out a smooth Zariski-open and Zariski-dense subspace of  $\prod_{\substack{j=1,...,s \ k \in \mathbb{Z}/d_j\mathbb{Z}}} \operatorname{GL}_{m_j}$  which, by taking fibre product with  $\operatorname{Res}_{\mathbb{Q}_p}^L(\operatorname{GL}_n/B)$ , gives a smooth Zariski-open and Zariski-dense subspace of  $\left(\prod_{\substack{j=1,...,s \ k \in \mathbb{Z}/d_j\mathbb{Z}}} \operatorname{GL}_{m_j}\right) \times \operatorname{Res}_{\mathbb{Q}_p}^L(\operatorname{GL}_n/B)$ . The proposition then follows by the same argument as in the last paragraph of the proof of [18, Lemma 2.4].

**Corollary 4.3.6.** The set  $\widetilde{U}_{\overline{\rho}}^{\text{pcr}}(\xi_0, h)$  is Zariski-open and Zariski-dense in  $\widetilde{\mathfrak{X}}_{\overline{\rho}}^{\text{pcr}}(\xi_0, h)$ .

Proof. Let  $U, \tilde{U}$  be as in the proof of Proposition 4.3.5. For each point  $x = (B_{j,k})$  of  $\prod_{\substack{j=1,\dots,s\\k\in\mathbb{Z}/d_j\mathbb{Z}}} \operatorname{GL}_{m_j}$  we can associate a Weil-Deligne representation  $\mathbf{r}_x$  of inertial type  $\xi_0$  with N = 0 as in the discussion below (4.30). By the same argument as in [26, Prop. 4.3], there exists a unique morphism  $f: \prod_{\substack{j=1,\dots,s\\k\in\mathbb{Z}/d_j\mathbb{Z}}} \operatorname{GL}_{m_j} \to (\operatorname{Spec} \mathcal{Z}_{\Omega_0})^{\operatorname{rig}}$  such that  $\mathbf{r}_{f(x)} \cong \mathbf{r}_x^{\operatorname{ss}}$  and the composition

 $\widetilde{U} \to U \to \mathfrak{X}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h}) \to (\operatorname{Spec} \mathcal{Z}_{\Omega_0})^{\mathrm{rig}}$  factors through  $\prod_{\substack{j=1,\dots,s\\k\in\mathbb{Z}/d_j\mathbb{Z}}} \operatorname{GL}_{m_j}$ . Hence (4.36) induces a smooth morphism

$$\widetilde{U} \times_{(\operatorname{Spec} \mathcal{Z}_{\Omega_0})^{\operatorname{rig}}} (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \longrightarrow \left( \left( \prod_{\substack{j=1,\dots,s\\k \in \mathbb{Z}/d_j \mathbb{Z}}} \operatorname{GL}_{m_j} \right) \times_{(\operatorname{Spec} \mathcal{Z}_{\Omega_0})^{\operatorname{rig}}} (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \right) \times \operatorname{Res}_{\mathbb{Q}_p}^L (\operatorname{GL}_n / B).$$

It follows from Lemma 4.3.4 (see also the last paragraph in the proof of Proposition 4.3.5) that the image of the generic points of  $\widetilde{U} \times_{(\operatorname{Spec} Z_{\Omega_0})^{\operatorname{rig}}} (\operatorname{Spec} Z_{\Omega})^{\operatorname{rig}}$  is Zariski-dense and Zariski-open in  $(\prod_{\substack{j=1,\ldots,s\\k\in\mathbb{Z}/d_j\mathbb{Z}}} \operatorname{GL}_{m_j}) \times_{(\operatorname{Spec} Z_{\Omega_0})^{\operatorname{rig}}} (\operatorname{Spec} Z_{\Omega})^{\operatorname{rig}}$ . Since

$$\widetilde{U} \times_{(\operatorname{Spec} \mathcal{Z}_{\Omega_0})^{\operatorname{rig}}} (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \longrightarrow U \times_{(\operatorname{Spec} \mathcal{Z}_{\Omega_0})^{\operatorname{rig}}} (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}}$$

is smooth and surjective, we deduce that the generic points are Zariski-dense and Zariski-open in  $U \times_{(\operatorname{Spec} \mathcal{Z}_{\Omega_0})^{\operatorname{rig}}} (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}}$ .

By the same argument as for [18, Lemma 2.2], we have

**Proposition 4.3.7.** The rigid space  $\widetilde{\mathfrak{X}}_{\overline{\rho}}^{\text{per}}(\xi_0, h)$  is reduced.

Let  $x = (\rho_x, (\mathbf{r}_{i,x})) \in \widetilde{U}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$ . The  $\Omega$ -filtration associated to  $(\mathbf{r}_{i,x})$  induces an  $\Omega$ -filtration on  $\mathrm{DF}(\rho_x)$ , to which we can associate  $w_x \in \mathscr{W}_{\max,L}^P$  as in § 2.3 (where  $w_x$  is denoted  $w_{\mathscr{F}}$ ). Recall that  $w_x$  measures the relative position of the Hodge filtration and the  $\Omega$ -filtration on  $D_{\mathrm{dR}}(\rho_x)$ . For  $w \in \mathscr{W}_{\max,L}^P$ , let  $V_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})_w$  be the set of points  $x = (\rho_x, (\mathbf{r}_{i,x}))$  in  $\widetilde{U}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$  satisfying  $w_x = w$ .

**Proposition 4.3.8.** (1) The set  $V_{\overline{\rho}}^{\text{pcr}}(\xi_0, \boldsymbol{h})_{w_0}$  is Zariski-open and Zariski-dense in  $\widetilde{U}_{\overline{\rho}}^{\text{pcr}}(\xi_0, \boldsymbol{h})$  (hence in  $\widetilde{\mathfrak{X}}_{\overline{\rho}}^{\text{pcr}}(\xi_0, \boldsymbol{h})$ ).

(2) For  $w \in \mathscr{W}_{\max,L}^P$ ,  $V_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})_w$  is locally Zariski-closed in  $\widetilde{U}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$ . Moreover, if  $w' \in \mathscr{W}_{\max,L}^P$  and  $w' \leq w$ , then  $V_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})_{w'}$  lies in the Zariski-closure  $V_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})_w$  of  $V_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})_w$  in  $\widetilde{U}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$ .

Proof. Let  $U \subset U_{\overline{\rho}}^{\text{pcr}}(\xi_0, \mathbf{h})$  be a non-empty affinoid open subset as in the proof of Proposition 4.3.5. We freely use the notation of *loc. cit.* For  $j = 1, \ldots, s$ , let  $B_{U,j} := \prod_{i \in \mathbb{Z}/[L'_0:L_0]} B_{U,j,i}$ , and  $P_j(X) \in \mathcal{O}_U[X]$  be the characteristic polynomial of  $B_{U,j}$ . Let  $\mathscr{U}$  be an étale covering of U such that  $P_j(X) = \prod_{i=1}^{m_j} (X - \alpha_{j,i})$  (with  $\alpha_{j,i} \neq \alpha_{j,i'}$  for  $i \neq i'$  since any point in U is generic), and such that, for each j, i, there exists  $\beta_{j,i} \in \mathcal{O}_{\mathscr{U}}$  such that  $\beta_{j,i}^{d_j} = \alpha_{j,i}$ . Using the same argument as in the proof of Lemma 4.3.4, we have an isomorphism of Weil-Deligne representations over  $\mathcal{O}_{\mathscr{U}}$ :

$$\mathbf{r}_{\mathscr{U}} \cong \bigoplus_{j=1}^{s} \bigoplus_{i=1}^{m_j} \widetilde{\mathbf{r}}_j \otimes_E \operatorname{unr}(\beta_{j,i}).$$

Let  $\widetilde{\mathscr{U}} := \mathscr{U} \times_{(\operatorname{Spec} \mathcal{Z}_{\Omega_0})^{\operatorname{rig}}} (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}}$  and  $r_{\widetilde{\mathscr{U}}} := r_{\mathscr{U}} \otimes_{\mathcal{O}_{\mathscr{U}}} \mathcal{O}_{\widetilde{\mathscr{U}}}$ . By the universal property of  $\widetilde{\mathscr{U}} \to \mathscr{U}$ , we have an (ordered) *r*-tuple  $(\mathbf{r}_{\widetilde{\mathscr{U}},i})_{i=1,\ldots,r}$  of Weil-Deligne subrepresentations of  $\mathbf{r}_{\widetilde{\mathscr{U}}}$  over  $\widetilde{\mathscr{U}}$  such that  $\oplus_{i=1}^r \mathbf{r}_{\widetilde{\mathscr{U}},i} \cong \oplus_{j=1}^s \oplus_{i=1}^{m_j} \widetilde{\mathbf{r}}_j \otimes_E \operatorname{unr}(\beta_{j,i})$  and  $\mathbf{r}_{\widetilde{\mathscr{U}},i}$  is of inertial type  $\xi_i$ .

The Deligne-Fontaine module  $\mathrm{DF}_{\widetilde{\mathscr{U}}} := \mathrm{DF}_U \otimes_{\mathcal{O}_U} \mathcal{O}_{\widetilde{\mathscr{U}}}$  over  $\widetilde{\mathscr{U}}$  is isomorphic to the Deligne-Fontaine module associated to  $\mathbf{r}_{\widetilde{\mathscr{U}},i}$ . Let  $\mathrm{DF}_{\widetilde{\mathscr{U}},i}$  be the Deligne-Fontaine module associated to  $\mathbf{r}_{\widetilde{\mathscr{U}},i}$ , we have thus  $\mathrm{DF}_{\widetilde{\mathscr{U}}} \cong \bigoplus_{i=1}^r \mathrm{DF}_{\widetilde{\mathscr{U}},i}$ . We let  $\widetilde{\mathscr{U}}^r$  be the  $(\mathbb{G}_m^{\mathrm{rig}})^r$ -torsor over  $\widetilde{\mathscr{U}}$  of isomorphisms of Deligne-Fontaine modules:

$$\oplus_{i=1}^r \mathrm{DF}_{\widetilde{\mathscr{U}},i} \xrightarrow{\sim} \mathrm{DF}_{\widetilde{\mathscr{U}}}.$$

By the universal property of  $\widetilde{\mathscr{U}}^r$ , we have a universal isomorphism

$$\oplus_{i=1}^{r} \mathrm{DF}_{\widetilde{\mathscr{U}}^{r},i} \xrightarrow{\sim} \mathrm{DF}_{\widetilde{\mathscr{U}}^{r}},$$

hence universal embeddings  $\bigoplus_{i=1}^{j} \mathrm{DF}_{\widetilde{\mathscr{U}}^{r},i} \hookrightarrow \mathrm{DF}_{\widetilde{\mathscr{U}}^{r}}$  for  $j = 1, \ldots, r$ , where  $*_{\widetilde{\mathscr{U}}^{r}}$  denotes the base change of the corresponding object to  $\mathcal{O}_{\widetilde{\mathscr{U}}^{r}}$ . Let  $W := (L \otimes_{\mathbb{Q}_{p}} \mathcal{O}_{\widetilde{\mathscr{U}}^{r}})^{n}$ , and we equip W with a filtration  $\mathrm{Fil}_{W}^{\bullet}$  consisting of free  $L \otimes_{\mathbb{Q}_{p}} \mathcal{O}_{\widetilde{\mathscr{U}}^{r}}$ -submodules which are direct summands of W such that  $\mathrm{rk}_{L \otimes_{\mathbb{Q}_{p}} \mathcal{O}_{\widetilde{\mathscr{U}}^{r}}} \mathrm{Fil}_{W}^{i} W = s_{i}$  for  $i = 1, \ldots, r$ . Let  $\widetilde{\mathscr{U}_{P}^{r}}$  be the  $\mathrm{Res}_{\mathbb{Q}_{p}}^{L} P$ -torsor over  $\widetilde{\mathscr{U}^{r}}$  of isomorphisms

$$(L \otimes_{\mathbb{Q}_p} \mathcal{O}_{\widetilde{\mathscr{U}}^r})^n \xrightarrow{\sim} (\mathrm{DF}_{\widetilde{\mathscr{U}}^r} \otimes_{L'_0} L')^{\mathrm{Gal}(L'/L)}$$

which send  $\operatorname{Fil}_W^i W \cong (L \otimes_{\mathbb{Q}_p} \mathcal{O}_{\widetilde{\mathscr{U}_r}})^{s_i}$  onto  $(\bigoplus_{i'=1}^i \operatorname{DF}_{\widetilde{\mathscr{U}_i},i'} \otimes_{L'_0} L')^{\operatorname{Gal}(L'/L)}$ . Via the universal isomorphism  $(L \otimes_{\mathbb{Q}_p} \mathcal{O}_{\widetilde{\mathscr{U}_p}})^n \xrightarrow{\sim} (\operatorname{DF}_{\widetilde{\mathscr{U}_p}} \otimes_{L'_0} L')^{\operatorname{Gal}(L'/L)}$ , the Hodge filtration on  $(\operatorname{DF}_{\widetilde{\mathscr{U}_p}} \otimes_{L'_0} L')^{\operatorname{Gal}(L'/L)}$  (which comes from the Hodge filtration on  $(\operatorname{DF}_U \otimes_{L'_0} L')^{\operatorname{Gal}(L'/L)}$  by base change) induces a morphism

$$\widetilde{\mathscr{U}}_P^r \longrightarrow \operatorname{Res}_{\mathbb{Q}_p}^L(\operatorname{GL}_n/B).$$
 (4.37)

By similar arguments as for (4.36), one can show that this morphism is smooth.

For  $w = (w_{\tau}) \in \mathscr{W}_{\max,L}^{P}$ , let  $\widetilde{\mathscr{U}}_{P,w}^{r}$  be the set of points which are sent to  $V_{\overline{\rho}}^{\mathrm{pcr}}(\xi_{0}, \mathbf{h})_{w}$  via the (smooth surjective) morphism  $\widetilde{\mathscr{U}}_{P,w}^{r} \to U$ . We see that  $x \in \widetilde{\mathscr{U}}_{P,w}^{r}$  if and only if the image of x under (4.37) is contained in the (generalized) Schubert cell  $\prod_{\tau \in \Sigma_{L}} (Pw_{\tau}B)/B$ . Equivalently  $\widetilde{\mathscr{U}}_{P,w}^{r}$  is the inverse image of  $\prod_{\tau \in \Sigma_{L}} (Pw_{\tau}B)/B$  in  $\widetilde{\mathscr{U}}_{P}^{r}$ . The proposition then follows from the corresponding facts on Schubert cells by the same argument as in the last paragraph of the proof of [18, Lemma 2.4].

Let 
$$w \in \mathscr{W}_{\max,L}^P$$
, then  $ww_{0,L} \in \mathscr{W}_{\min,L}^P$  hence  $ww_{0,L}(\mathbf{h})$  is strictly *P*-dominant. Define

$$\iota: \widetilde{\mathfrak{X}}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h}) \longrightarrow \mathfrak{X}_{\overline{\rho}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\mathrm{rig}} \times \widehat{\mathcal{Z}_{0,L}}, \ (\rho, (\mathbf{r}_i)) \mapsto (\rho, (\mathbf{r}_i), 1).$$
(4.38)

We consider  $\iota^{-1}(X_{\Omega,ww_{0,L}(\mathbf{h})}(\overline{\rho}))$ , which is a Zariski-closed subspace of  $\widetilde{\mathfrak{X}}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$ . By the discussion in § 2.3 (in particular (2.4)) and using that, for any  $x = (\rho, (\mathbf{r}_i)) \in V_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})_w$ , we have by definition  $w = w_x$  (which is denoted  $w_{\mathscr{F}}$  in § 2.3), one can check that

$$V_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})_w \subseteq \iota^{-1}(U_{\Omega, ww_{0,L}(\mathbf{h})}(\overline{\rho})) \subseteq \iota^{-1}(X_{\Omega, ww_{0,L}(\mathbf{h})}(\overline{\rho}))$$

which implies

$$\overline{V^{\mathrm{pcr}}_{\overline{\rho}}(\xi_0, \mathbf{h})_w} \subset \iota^{-1}(X_{\Omega, ww_{0,L}(\mathbf{h})}(\overline{\rho})).$$

By Proposition 4.3.8 (2), we deduce that, for  $w' \in \mathscr{W}_{\max,L}^P$  and  $w' \leq w$ , we have  $\iota(V_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})_{w'}) \subseteq X_{\Omega, ww_{0,L}(\mathbf{h})}(\overline{\rho})$ , and thus:

**Corollary 4.3.9.** Let  $x = (\rho_x, (\mathbf{r}_{i,x})) \in \widetilde{U}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$ . For all  $w \in \mathscr{W}_{\max,L}^P$  such that  $w \ge w_x$ , we have

$$\iota(x) = (\rho_x, (\mathbf{r}_{1,x}, \dots, \mathbf{r}_{r,x}), 1) \in X_{\Omega, ww_{0,L}(\mathbf{h})}(\overline{\rho}).$$

**Remark 4.3.10.** (1) The point  $\iota(x) \in X_{\Omega,ww_{0,L}(h)}(\overline{\rho})$  for  $w \in \mathscr{W}_{\max,L}^P$ ,  $w \ge w_x$ , is called a local companion point of  $\iota(x) \in X_{\Omega,h}(\overline{\rho})$  (see also Corollary 6.4.12 below).

(2) The case P = B (and  $\rho_x$  crystalline) was contained in [19, Thm. 4.2.3]. Indeed, in this case, as discussed in Remark 4.2.4 (1), we have the isomorphism

$$\iota_{\Omega,ww_{0,L}(h)}: X_{\Omega,ww_{0,L}(h)}(\overline{\rho}) \xrightarrow{\sim} X_{\mathrm{tri}}^{\Box}(\overline{\rho}) \ (\hookrightarrow (\mathrm{Spf} \ R_{\overline{\rho}})^{\mathrm{rig}} \times \widehat{T(L)})$$

where  $X_{\text{tri}}^{\Box}(\overline{\rho})$  is the trianguline variety of [17, § 2.2]. And the isomorphism  $\iota_{\Omega,ww_{0,L}(\mathbf{h})}$  sends  $\iota(x)$  in Corollary 4.3.9 exactly to the point  $x_w$  in [19, Thm. 4.2.3]. Note that the resulting points  $\{x_w\} \subset X_{\text{tri}}^{\Box}(\overline{\rho})$  are distinct.

(3) Note that, if  $w \neq w_x$ , then  $((\mathbf{r}_{i,x}), \delta = \boxtimes_{i=1}^r z^{ww_0(\mathbf{h})_i}) \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{L_P}(L)}$  is not a parameter of the  $\Omega$ -filtration induced by  $(\mathbf{r}_{i,x})$  on  $D_{\operatorname{rig}}(\rho_x)$  (compare with Remark 4.2.4 (2) and see Definition 4.1.6 (2), (3) for a parameter of an  $\Omega$ -filtration).

# 5 The geometry of some schemes related to generalized Springer resolutions

In this section, we show some results of geometric representation theory concerning algebraic varieties which are a "parabolic" generalization of Grothendieck's and Springer's resolution of singularities. These results will be crucially used in § 6 to describe the local rings of the Bernstein paraboline varieties at certain points.

### 5.1 Preliminaries

We let G/E be a connected split reductive algebraic group. We recall/introduce certain schemes  $X_P$  (where  $P \subseteq G$  is a parabolic subgroup) which are related to parabolic versions of Grothendieck' simultaneous resolution of singularities for G.

We fix a Borel subgroup B of G, and let  $T \subset B$  be a maximal torus and  $N \subset B$  the unipotent radical of B. Let  $P \supseteq B$  be a parabolic subgroup of G,  $L_P$  be the Levi subgroup of P containing T and  $N_P$  the unipotent radical of P. Let  $\mathfrak{p}$  be the Lie algebra of P over E,  $\mathfrak{n}_P$  its nilpotent radical,  $\mathfrak{r}_P$  the full radical of  $\mathfrak{p}$ ,  $\mathfrak{l}_P$  the Lie algebra of  $L_P$  over E, and  $\mathfrak{z}_{L_P}$  the centre of  $\mathfrak{l}_P$ . We have  $\mathfrak{r}_P \cong \mathfrak{n}_P \rtimes \mathfrak{z}_{L_P}$ . Let  $\mathfrak{g}$ ,  $\mathfrak{b}$ ,  $\mathfrak{n}$ ,  $\mathfrak{t}$  be the Lie algebra over E of G, B, N, T respectively.

Let  $\mathscr{W}$  be the Weyl group of G and  $w_0 \in \mathscr{W}$  the element of maximal length. For  $w \in \mathscr{W}$ , we also use w to denote some lift<sup>4</sup> of w in  $N_G(T) \subset G(E)$ . As in § 2.1, denote by  $\mathscr{W}_{\min}^P$  (resp.  $\mathscr{W}_{\max}^P$ , resp.  $\underset{\min}{P}\mathscr{W}$ , resp.  $\underset{\max}{P}\mathscr{W}$ ) the set of minimal (resp. maximal, resp. minimal, resp. maximal) length representatives of  $\mathscr{W}_{L_P} \setminus \mathscr{W}$  (resp.  $\mathscr{W}_{L_P} \setminus \mathscr{W}$ , resp.  $\mathscr{W} / \mathscr{W}_{L_P}$ , resp.  $\mathscr{W} / \mathscr{W}_{L_P}$ ). Note that  $w \in \mathscr{W}_{\min}^P$ if and only if  $ww_0 \in \mathscr{W}_{\max}^P$ . Also  $w \in \mathscr{W}_{\min}^P$  (resp.  $w \in \mathscr{W}_{\max}^P$ ) if and only if  $w^{-1} \in \underset{\min}{P}\mathscr{W}$ 

<sup>&</sup>lt;sup>4</sup>When we apply this notation for certain group operators, we always mean first applying group operations then taking a certain lift, e.g. we use  $w_1w_2 \in N_G(T)$  to denote a lift of  $w_1w_2 \in \mathcal{W}$  rather than the multiplication of a lift of  $w_1$  and a lift of  $w_2$ .

(resp.  $w^{-1} \in {}^{P}_{\max} \mathcal{W}$ ). For  $w \in \mathcal{W}$  or  $\mathcal{W}_{L_P} \setminus \mathcal{W}$ , denote by  $w^{\min} \in \mathcal{W}_{\min}^P$  (resp.  $w^{\max} \in \mathcal{W}_{\max}^P$ ) the corresponding representative of  $\mathcal{W}$ . We use "·" to denote the *dot* action of  $\mathcal{W}$  on the weights of t (cf. [57, Def. 1.8]).

If  $f: X \to Y$  is a morphism of schemes and Z a locally closed subscheme of Y, we denote  $f^{-1}(Z) := X \times_Y Z$ , which is a locally closed subscheme of X. If  $X_1$ ,  $X_2$  are locally closed subschemes of X, we denote  $X_1 \cap X_2 := X_1 \times_X X_2$ .

Let  $\tilde{\mathfrak{g}}_P$  be the closed *E*-subscheme of  $G/P \times \mathfrak{g}$  defined by

$$\{(gP,\psi)\in G/P\times\mathfrak{g}\mid \operatorname{Ad}(g^{-1})\psi\in\mathfrak{r}_P\}$$

where as usual Ad(h) means the adjoint action of  $h \in G$  (i.e. conjugation by h). We have an isomorphism of *E*-schemes (using standard notation, see e.g. [61, § VI.8]):

$$G \times^{P} \mathfrak{r}_{P} \xrightarrow{\sim} \tilde{\mathfrak{g}}_{P}, \ (g, \psi) \mapsto (gP, \operatorname{Ad}(g)\psi).$$
 (5.1)

We see that  $\tilde{\mathfrak{g}}_P$  is a vector bundle over G/P, hence is smooth and irreducible. We also have  $\dim \tilde{\mathfrak{g}}_P = \dim G/P + \dim \mathfrak{r}_P = \dim \mathfrak{n}_P + \dim \mathfrak{r}_P$ . There are natural morphisms:

$$\begin{array}{ll} \kappa_P: \quad \tilde{\mathfrak{g}}_P \longrightarrow \mathfrak{z}_{L_P} & (gP,\psi) \longmapsto \overline{\mathrm{Ad}(g^{-1})\psi} \\ q_P: \quad \tilde{\mathfrak{g}}_P \longrightarrow \mathfrak{g} & (gP,\psi) \longmapsto \psi \\ \pi_P: \quad \tilde{\mathfrak{g}}_P \longrightarrow G/P & (gP,\psi) \longmapsto gP \end{array}$$

where  $\overline{\operatorname{Ad}(g^{-1})}\psi$  is the image of  $\operatorname{Ad}(g^{-1})\psi \in \mathfrak{r}_P$  via the canonical surjection  $\mathfrak{r}_P \twoheadrightarrow \mathfrak{z}_{L_P}$ . Alternatively we can see  $q_P$  as the morphism  $G \times^P \mathfrak{r}_P \to \mathfrak{g}, (g, \psi) \longmapsto \operatorname{Ad}(g)\psi$ . We denote  $\tilde{\mathfrak{g}} := \tilde{\mathfrak{g}}_B \simeq G \times^B \mathfrak{b}$  and define  $q_B : \tilde{\mathfrak{g}} \to \mathfrak{g}$  similarly to  $q_P$ . We put:

$$X_P := \tilde{\mathfrak{g}} \times_{\mathfrak{g}} \tilde{\mathfrak{g}}_P,$$

which is also the closed subscheme of  $G/B \times G/P \times \mathfrak{g}$  defined by

$$X_P \simeq \{ (g_1 B, g_2 P, \psi) \in G/B \times G/P \times \mathfrak{g} \mid \operatorname{Ad}(g_1^{-1})\psi \in \mathfrak{b}, \operatorname{Ad}(g_2^{-1})\psi \in \mathfrak{r}_P \}.$$

It is easy to check that we have isomorphisms of E-schemes

$$\begin{array}{rccc} X_P & \xrightarrow{\sim} & G \times^P q_B^{-1}(\mathfrak{r}_P) & (g_1 B, g_2 P, \psi) & \longmapsto & \left(g_2, (g_2^{-1} g_1 B, \operatorname{Ad}(g_2^{-1})\psi)\right) \\ X_P & \xrightarrow{\sim} & G \times^B q_P^{-1}(\mathfrak{b}) & (g_1 B, g_2 P, \psi) & \longmapsto & \left(g_1, (g_1^{-1} g_2 P, \operatorname{Ad}(g_1^{-1})\psi)\right). \end{array}$$
(5.2)

#### 5.2 Analysis of the global geometry

We prove useful statements on the geometry of the *E*-scheme  $X_P$ .

Let  $\pi$  be the composition

$$\pi: X_P \hookrightarrow G/B \times G/P \times \mathfrak{g} \twoheadrightarrow G/B \times G/P.$$
(5.3)

We equip  $G/B \times G/P$  with an action of G by diagonal left multiplication. For  $w \in \mathcal{W}$ , write

$$U_w := G(w, 1)(B \times P) = G(1, w^{-1})(B \times P) \subset G/B \times G/P$$

Note that  $U_w$  only depends on the the right coset  $\mathscr{W}_{L_P}w$  (or equivalently, the left coset  $w^{-1}\mathscr{W}_{L_P}$ ). In fact, we have an isomorphism

$$G \times^B (G/P) \xrightarrow{\sim} G/B \times G/P, \ (g_1, g_2P) \longmapsto (g_1B, g_1g_2P)$$

which induces an isomorphism

$$G \times^B (Bw^{-1}P/P) \xrightarrow{\sim} U_w$$

Likewise, we have an isomorphism

$$G \times^P (G/B) \xrightarrow{\sim} G/B \times G/P, \ (g_1, g_2B) \longmapsto (g_1g_2B, g_1P)$$

which induces an isomorphism

$$G \times^P (PwB/B) \xrightarrow{\sim} U_w.$$

By the standard Bruhat decomposition of G/P and G/B, we deduce

$$G/B\times G/P=\sqcup_{w\in \mathscr{W}_{\min}^P}U_w=\sqcup_{w\in \mathscr{W}_{\max}^P}U_w.$$

We also deduce that  $U_w$  is a locally closed subscheme of  $G/B \times G/P$  which is smooth of dimension

$$\dim G - \dim B + \dim(Bw^{-1}P/P) = \dim G - \dim B + \lg(w^{\min})$$
$$= \dim G - \dim P + \lg(w^{\max}) = \dim G - \dim P + \dim(PwB/B) \quad (5.4)$$

where, for the first equality, we use  $\lg((w^{\min})^{-1}) = \lg(w^{\min})$  and  $(w^{\min})^{-1} \in {}^{P}_{\min} \mathscr{W}$ .

We let 
$$V_w := \pi^{-1}(U_w)$$
.

**Proposition 5.2.1.** The surjection  $V_w \rightarrow U_w$  induced by  $\pi$  is a (geometric) vector bundle of relative dimension dim  $\mathfrak{r}_P - \lg(w^{\min})$ .

*Proof.* Let  $y = (gB, gw^{-1}P) = (gB, g(w^{\min})^{-1}P) \in U_w \subset G/B \times G/P$ . One can check that

$$\pi^{-1}(y) = y \times \operatorname{Ad}(g) \Big( \operatorname{Ad}((w^{\min})^{-1}) \mathfrak{z}_{L_P} \oplus \big( \mathfrak{n} \cap \operatorname{Ad}((w^{\min})^{-1}) \mathfrak{n}_P \big) \Big).$$
(5.5)

Since  $(w^{\min})^{-1} \in \underset{\min}{\overset{P}{\mathscr{W}}}$ , one has

$$\dim \left( \mathfrak{n} \cap \operatorname{Ad}((w^{\min})^{-1})\mathfrak{n}_P \right) = \dim \left( \mathfrak{n} \cap \operatorname{Ad}((w^{\min})^{-1})\mathfrak{n} \right) - \dim \mathfrak{n}_{L_P}$$
$$= \dim \mathfrak{n} - \lg((w^{\min})^{-1}) - \dim \mathfrak{n}_{L_P} = \dim \mathfrak{n}_P - \lg(w^{\min}).$$

The proposition then follows by the same argument as for [19, Prop. 2.2.1].

From (5.4) and Proposition 5.2.1 we obtain that  $V_w$  is equidimensional of dimension dim G – dim B + dim  $\mathfrak{r}_P$ . Let  $X_w$  be the closed subscheme of  $X_P$  defined as the *reduced* Zariski-closure of  $V_w$  in  $X_P$ . By the same argument as in the first part of the proof of [19, Prop. 2.2.5], we have:

**Corollary 5.2.2.** The scheme  $X_P$  is equidimensional of dimension dim G – dim B + dim  $\mathfrak{r}_P$ . The irreducible components of  $X_P$  are  $\{X_w\}_{w \in \mathscr{W}_{\min}^P} = \{X_w\}_{w \in \mathscr{W}_{\max}^P}$  and  $V_w$  is open in  $X_w$ .

**Remark 5.2.3.** (1) We could equip the underlying closed subset  $X_w$  with another scheme structure, namely the scheme theoretic image of the open subscheme  $X_P \setminus \bigcup_{W_{L_P}w' \neq W_{L_P}w} X_{w'}$  of  $X_P$ . Let us denote it by  $X'_w$ . When P = B, by [19, Thm. 2.2.6],  $X_P$  is reduced, so  $X_w = X'_w$  as closed subschemes of  $X_P$ . However, in general, it is not clear to the authors if  $X_P$  is reduced, or if  $X_P$ is Cohen-Macaulay (for instance one can easily check that the last paragraph of the proof of [19, Prop. 2.2.5] does not extend to  $P \neq B$ ). (2) For  $w \in \mathcal{W}$  let us define  $q_B^{-1}(\mathfrak{r}_P)_w^0$  (resp.  $q_P^{-1}(\mathfrak{b})_w^0$ ) as the preimage of PwB/B (resp. of BwP/P) via the composition  $q_B^{-1}(\mathfrak{r}_P) \hookrightarrow G/B \times \mathfrak{g} \twoheadrightarrow G/B$  (resp.  $q_P^{-1}(\mathfrak{b}) \hookrightarrow G/P \times \mathfrak{g} \twoheadrightarrow G/P$ ). By similar arguments as in Proposition 5.2.1, the map  $q_B^{-1}(\mathfrak{r}_P)_w^0 \to PwB/B$  (resp.  $q_P^{-1}(\mathfrak{b})_w^0 \to BwP/P$ ) is a vector bundle of relative dimension  $\dim \mathfrak{r}_P - \lg(w^{\min})$ . We define  $q_B^{-1}(\mathfrak{r}_P)_w$  (resp.  $q_P^{-1}(\mathfrak{b})_w^0 \to BwP/P$ ) is a vector bundle of relative dimension  $\dim \mathfrak{r}_P - \lg(w^{\min})$ . We define  $q_B^{-1}(\mathfrak{r}_P)_w$  (resp.  $q_P^{-1}(\mathfrak{b})_w$ ) to be the reduced Zariski-closure of  $q_B^{-1}(\mathfrak{r}_P)_w^0$  (resp.  $q_P^{-1}(\mathfrak{b})_w^0$ ) in  $q_B^{-1}(\mathfrak{r}_P)$  (resp. in  $q_P^{-1}(\mathfrak{b})$ ). The scheme  $q_B^{-1}(\mathfrak{r}_P)$  (resp.  $q_P^{-1}(\mathfrak{b})$ ) is equidimensional of dimension  $\dim P - \dim B + \dim \mathfrak{r}_P$  (resp.  $\dim \mathfrak{r}_P)$  with irreducible components given by  $\{q_B^{-1}(\mathfrak{r}_P)_w\}_{w\in \mathcal{W}_{\min}^P}$  (resp.  $\{q_P^{-1}(\mathfrak{b})_w\}_{w\in \min} \mathcal{W}$ ). Moreover  $q_B^{-1}(\mathfrak{r}_P)_w^0$  (resp.  $q_P^{-1}(\mathfrak{b})_w^0$ ) is open in  $q_B^{-1}(\mathfrak{r}_P)_w$  (resp.  $q_P^{-1}(\mathfrak{b})_w$ ). From (5.2) we also have isomorphisms of reduced E-schemes for  $w \in \mathcal{W}_{\min}^P$ 

$$X_w \cong G \times^P q_B^{-1}(\mathfrak{r}_P)_w \cong G \times^B q_P^{-1}(\mathfrak{b})_{w^{-1}}.$$

**Lemma 5.2.4.** For  $w, w' \in \mathcal{W}$  we have that  $X_w \cap V_{w'} \neq \emptyset$  implies  $w'^{\min} \leq w^{\min}$  ( $\Leftrightarrow w'^{\max} \leq w^{\max} \Leftrightarrow w'^{\min}w_0 \geq w^{\min}w_0$ ).

*Proof.* The lemma follows by the same argument as in the proof of [19, Lemma 2.2.4], noting that in  $G \times^{P} (G/B)$  we have  $\overline{U_{w}} \cap U_{w'} \neq \emptyset \Rightarrow \overline{PwB/B} \supset Pw'B/B \Rightarrow w^{\max} \geq w'^{\max} \Rightarrow w^{\min} \geq w'^{\min}$  (where  $\overline{(-)}$  means Zariski-closure).

Denote by  $\kappa_B : X_P \to \mathfrak{t}$  (resp.  $\kappa_P : X_P \to \mathfrak{t}$ ) the morphism

$$(g_1B, g_2P, \psi) \longmapsto \overline{\operatorname{Ad}(g_1^{-1})\psi} \quad (\text{resp. } (g_1B, g_2P, \psi) \longmapsto \overline{\operatorname{Ad}(g_2^{-1})\psi})$$
(5.6)

where  $\operatorname{Ad}(g_1^{-1})\psi$  is the image of  $\operatorname{Ad}(g_1^{-1})\psi \in \mathfrak{b}$  via  $\mathfrak{b} \twoheadrightarrow \mathfrak{t}$  (and see § 5.1 for  $\operatorname{Ad}(g_2^{-1})\psi$ ). Note that  $\kappa_P$  factors through  $\mathfrak{z}_{L_P} \hookrightarrow \mathfrak{t}$ . For \* = B, P denote by  $\kappa_{*,w}$  the restriction of  $\kappa_*$  to  $X_w$ .

**Lemma 5.2.5.** For  $w \in \mathcal{W}$  we have  $\kappa_{P,w} = \operatorname{Ad}(w) \circ \kappa_{B,w}$ , where  $\operatorname{Ad}(w) : \mathfrak{t} \to \mathfrak{t}$  denotes the morphism induced by the adjoint action of  $\mathcal{W}$  on  $\mathfrak{t}$ . In particular, the following diagram commutes

$$\begin{array}{cccc} X_w & \xrightarrow{\kappa_{B,w}} & \mathfrak{t} \\ & & & \downarrow \\ & & & \downarrow \\ & \mathfrak{t} & \longrightarrow & \mathfrak{t}/\mathscr{W} \end{array}$$

where  $\mathfrak{t}/\mathscr{W} := \operatorname{Spec} R^{\mathscr{W}}_{\mathfrak{t}}$  (with  $\mathfrak{t} := \operatorname{Spec} R_{\mathfrak{t}}$ ) and the two morphisms  $\mathfrak{t} \to \mathfrak{t}/\mathscr{W}$  are both the canonical surjection.

*Proof.* This is the argument for [19, Lemma 2.3.4], let us recall it. As  $\mathfrak{t}/\mathscr{W}$  is affine, it is separated, hence the diagonal embedding  $\mathfrak{t}/\mathscr{W} \to \mathfrak{t}/\mathscr{W} \times_E \mathfrak{t}/\mathscr{W}$  is a closed immersion, and so is  $\mathfrak{t} \times_{\mathfrak{t}/\mathscr{W}} \mathfrak{t} \to \mathfrak{t} \times_E \mathfrak{t}$  by base change along  $\mathfrak{t} \times_E \mathfrak{t} \to \mathfrak{t}/\mathscr{W} \times_E \mathfrak{t}/\mathscr{W}$ . Since the diagram clearly commutes with  $V_w$  instead of  $X_w$  and  $V_w$  is Zariski-dense in  $X_w$ , the lemma follows.

In particular,  $\operatorname{Ad}(w) \circ \kappa_{B,w}$  only depends on the coset  $\mathscr{W}_{L_P}w$  as the same holds for  $\kappa_{P,w}$ . Consider the affine *E*-scheme  $\mathscr{T}_P := \mathfrak{t} \times_{\mathfrak{t}/\mathscr{W}} \mathfrak{z}_{L_P}$ . We have a morphism of *E*-schemes

$$(\kappa_B, \kappa_P) : X_P \longrightarrow \mathscr{T}_P.$$

**Lemma 5.2.6.** The irreducible components of  $\mathscr{T}_P$  are  $\{\mathscr{T}_w\}_{w \in \mathscr{W}_{\min}^P} = \{\mathscr{T}_w\}_{w \in \mathscr{W}_{\max}^P}$  where  $\mathscr{T}_w$  is the reduced *E*-scheme:

$$\mathscr{T}_w := \{ (\mathrm{Ad}(w^{-1})z, z), \ z \in \mathfrak{z}_{L_P} \}$$

(so  $\mathscr{T}_w$  only depends on the coset  $\mathscr{W}_{L_P}w$ ). Moreover  $X_w$  is the unique irreducible component of  $X_P$  such that  $(\kappa_B, \kappa_P)(X_w) = \mathscr{T}_w$ .

Proof. Since  $\mathfrak{t} \to \mathfrak{t}/\mathscr{W}$  is finite, we deduce dim  $\mathscr{T}_P = \dim \mathfrak{z}_{L_P} = \dim \mathscr{T}_w$  for any  $w \in \mathscr{W}$ . It is also clear that  $\mathscr{T}_w \cong \mathfrak{z}_{L_P}$  is irreducible. The first part of the lemma easily follows. By Lemma 5.2.5,  $(\kappa_B, \kappa_P)(X_w) \subset \mathscr{T}_w$ . From (5.5), we see that the restriction  $(\kappa_B, \kappa_P)|_{V_w} : V_w \to \mathscr{T}_w$  is surjective. The second part of the lemma follows.

For a scheme Y and a point  $y \in Y$ , recall that we denote  $\mathcal{O}_{Y,y}$  the local ring of Y at y,  $\widehat{\mathcal{O}}_{Y,y}$  the completion of  $\mathcal{O}_{Y,y}$  along its maximal ideal, and  $\widehat{Y}_y := \operatorname{Spf} \widehat{\mathcal{O}}_{Y,y}$  the associated formal scheme (whose underlying topological space is one point).

**Lemma 5.2.7.** Let x be a closed point of  $X_P$ ,  $w, w' \in \mathcal{W}$ . Assume  $x \in X_w$ . Then the composition  $\widehat{X}_{w,x} \hookrightarrow \widehat{X}_{P,x} \to \widehat{\mathscr{T}}_{P,(\kappa_B,\kappa_P)(x)}$  factors through  $\widehat{\mathscr{T}}_{w',(\kappa_B,\kappa_P)(x)} \hookrightarrow \widehat{\mathscr{T}}_{P,(\kappa_B,\kappa_P)(x)}$  if and only if  $\mathscr{W}_{L_P}w' = \mathscr{W}_{L_P}w$ .

*Proof.* Let  $y := (\kappa_B, \kappa_P)(x)$ , a closed point of  $\mathscr{T}_P$ . We have a commutative diagram of local rings

where the vertical maps are injective by Krull's intersection theorem. By assumption, the bottom composition factors through  $\hat{\mathcal{O}}_{\mathcal{T}_{u'},y}$ . Using the commutative diagram

$$\begin{array}{cccc} \mathcal{O}_{\mathscr{T}_{P},y} & \longrightarrow & \mathcal{O}_{\mathscr{T}_{w'},y} \\ & & & \downarrow \\ & & & \downarrow \\ \widehat{\mathcal{O}}_{\mathscr{T}_{P},y} & \longrightarrow & \widehat{\mathcal{O}}_{\mathscr{T}_{w'},y} \end{array}$$

and the injectivity of all the vertical maps in (5.7), we deduce that the upper composition in (5.7) factors through  $\mathcal{O}_{\mathscr{T}w',y}$ . In particular the map  $X_P \to \mathscr{T}_P$  sends the generic point of  $X_w$  to the generic point of  $\mathscr{T}_{w'}$ . By Lemma 5.2.6, we must have w' = w.

**Remark 5.2.8.** Lemma 5.2.7 is the analogue of [19, Lemma 2.5.2] where the normality of  $X_w$  there was used in the proof. However, this normality is in fact useless there, arguing as in the above proof.

Recall that  $\psi \in \mathfrak{r}_P$  is called *regular* if the subgroup  $C_G(\psi) := \{g \in G, \operatorname{Ad}(g)\psi = \psi\}$  satisfies dim  $C_G(\psi) \leq \dim C_G(\psi')$  for all  $\psi' \in \mathfrak{r}_P$ . We write  $\mathfrak{r}_P^{\operatorname{reg}}$  for the subset of regular elements in  $\mathfrak{r}_P$ , which is preserved under the *P*-action, and  $\mathfrak{z}_{L_P}^{\operatorname{reg}} := \mathfrak{z}_{L_P} \cap \mathfrak{r}_P^{\operatorname{reg}}$ . When  $\psi \in \mathfrak{z}_{L_P}$  we have  $\psi \in \mathfrak{z}_{L_P}^{\operatorname{reg}}$ exactly when  $C_G(\psi) = L_P$ . We say that  $\psi \in \mathfrak{r}_P$  is *regular semi-simple* if  $\psi \in P\mathfrak{z}_{L_P}^{\operatorname{reg}}$  (for the adjoint action of *P* on  $\mathfrak{r}_P$ ) and we write  $\mathfrak{r}_P^{\operatorname{reg}-ss} \subset \mathfrak{r}_P^{\operatorname{reg}}$  for the subset of regular semi-simple elements in  $\mathfrak{r}_P$ . It is well-known that both  $\mathfrak{r}_P^{\operatorname{reg}-ss}$  and  $\mathfrak{r}_P^{\operatorname{reg}}$  are Zariski-open (Zariski-dense) in  $\mathfrak{r}_P$  and that for each  $\psi \in \mathfrak{r}_P^{\text{reg}}$  we have dim  $C_G(x) = \dim L_P$ . Moreover the product map gives an isomorphism of (smooth irreducible) *E*-schemes:

$$P/L_P \times \mathfrak{z}_{L_P} \xrightarrow{\sim} \mathfrak{r}_P^{\operatorname{reg}-\operatorname{ss}}.$$
 (5.8)

Let  $\tilde{\mathfrak{g}}_P^{\operatorname{reg}-\operatorname{ss}}$  be the Zariski-open (Zariski-dense) subset of  $\tilde{\mathfrak{g}}_P$  corresponding to  $G \times^P \mathfrak{r}_P^{\operatorname{reg}-\operatorname{ss}}$  via the isomorphism (5.1), and  $X_P^{\operatorname{reg}-\operatorname{ss}} := \tilde{\mathfrak{g}}_P^{\operatorname{reg}-\operatorname{ss}} \times_{\mathfrak{g}} \tilde{\mathfrak{g}}$ , which is Zariski-open in  $X_P \cong \tilde{\mathfrak{g}}_P \times_{\mathfrak{g}} \tilde{\mathfrak{g}}$ . As  $V_w \cap X_P^{\operatorname{reg}-\operatorname{ss}} \neq \emptyset$  for any  $w \in \mathscr{W}$  (use (5.5) for instance),  $X_P^{\operatorname{reg}-\operatorname{ss}}$  is Zariski-dense in  $X_P$ .

**Proposition 5.2.9.** The scheme  $X_P^{\text{reg}-\text{ss}}$  is smooth over E. Moreover, the composition  $X_P^{\text{reg}-\text{ss}} \hookrightarrow X_P \xrightarrow{\kappa_P} \mathfrak{z}_{L_P}$  is smooth.

*Proof.* Since  $\mathfrak{z}_{L_P}$  is smooth over E, it is enough to prove the second statement. Let  $q_P^{-1}(\mathfrak{b})^{\operatorname{reg}-\operatorname{ss}} := \tilde{\mathfrak{g}}_P^{\operatorname{reg}-\operatorname{ss}} \cap q_P^{-1}(\mathfrak{b}) = \tilde{\mathfrak{g}}_P^{\operatorname{reg}-\operatorname{ss}} \times_{\tilde{\mathfrak{g}}_P} q_P^{-1}(\mathfrak{b})$ , which is Zariski-open in  $q_P^{-1}(\mathfrak{b})$  and Zariski-closed in  $\tilde{\mathfrak{g}}_P^{\operatorname{reg}-\operatorname{ss}}$ . An argument similar to (5.2) gives an isomorphism of E-schemes

$$X_P^{\operatorname{reg-ss}} \xrightarrow{\sim} G \times^B q_P^{-1}(\mathfrak{b})^{\operatorname{reg-ss}}.$$
 (5.9)

Seeing  $q_P^{-1}(\mathfrak{b})^{\operatorname{reg}-\operatorname{ss}}$  inside  $G \times^P \mathfrak{r}_P^{\operatorname{reg}-\operatorname{ss}}$  via (5.1), it is enough to prove that the composition

$$\kappa_P^{\operatorname{reg-ss}} : q_P^{-1}(\mathfrak{b})^{\operatorname{reg-ss}} \hookrightarrow G \times^P \mathfrak{r}_P^{\operatorname{reg-ss}} \longrightarrow \mathfrak{z}_{L_P}, \tag{5.10}$$

is smooth, where the second map is  $(g, \psi) \mapsto \overline{\psi}$  (note that  $\kappa_P^{\text{reg}-\text{ss}}$  is *B*-equivariant with the trivial action of *B* on the target). Indeed, under (5.9), the composition in the statement is the composition

$$G \times^B q_P^{-1}(\mathfrak{b})^{\operatorname{reg}-\operatorname{ss}} \xrightarrow{\operatorname{id} \times \kappa_P^{\operatorname{reg}-\operatorname{ss}}} G \times^B \mathfrak{z}_{L_P} \simeq G/B \times \mathfrak{z}_{L_P} \twoheadrightarrow \mathfrak{z}_{L_P}$$

(where the last surjection is the canonical projection), which is smooth as both maps are.

By (5.8), we have  $G \times^P \mathfrak{r}_P^{\text{reg}-\text{ss}} \simeq G/L_P \times \mathfrak{z}_{L_P}^{\text{reg}}$ . Under this isomorphism, the last map in (5.10) is  $(gL_P, \psi) \mapsto \psi$  (where  $\psi \in \mathfrak{z}_{L_P}^{\text{reg}}$ ), and  $q_P^{-1}(\mathfrak{b})^{\text{reg}-\text{ss}}$  is the closed subscheme of  $G/L_P \times \mathfrak{z}_{L_P}^{\text{reg}}$  defined by  $\{(gL_P, \psi) \mid \operatorname{Ad}(g)\psi \in \mathfrak{b}\}$ . Denote by Y the inverse image of  $q_P^{-1}(\mathfrak{b})^{\text{reg}-\text{ss}}$  under the smooth surjective map  $G \times \mathfrak{z}_{L_P}^{\text{reg}} \twoheadrightarrow G/L_P \times \mathfrak{z}_{L_P}^{\text{reg}}$ , then by base change  $Y \to q_P^{-1}(\mathfrak{b})^{\text{reg}-\text{ss}}$  is also smooth surjective, and using [1, Lemma 02K5] it is enough to prove that the composition  $Y \hookrightarrow G \times \mathfrak{z}_{L_P}^{\text{reg}} \twoheadrightarrow \mathfrak{z}_{L_P}^{\text{reg}}$  is smooth. It is enough to prove this Zariski-locally on Y, and since  $(Bw_0Nw)_{w\in\mathscr{W}}$  is a Zariski-open covering of G, one can replace  $G \times \mathfrak{z}_{L_P}^{\text{reg}}$  by  $Bw_0Nw \times \mathfrak{z}_{L_P}^{\text{reg}}$  (for an arbitrary  $w \in \mathscr{W}$ ) and Y by  $Y \cap (Bw_0Nw \times \mathfrak{z}_{L_P}^{\text{reg}})$ . Since the multiplication induces an isomorphism of schemes  $B \times w_0Nw \xrightarrow{\sim} Bw_0Nw$  and since Y is B-invariant, we have  $Y \simeq B \times Z$  for the closed subscheme  $Z := \{(w_0nw,\psi) \mid \operatorname{Ad}(w_0nw)\psi \in \mathfrak{b}\}$  of  $w_0Nw \times \mathfrak{z}_{L_P}^{\text{reg}}$ . As the projection  $B \times Z \twoheadrightarrow Z$  is smooth, it is enough to prove that  $Z \to \mathfrak{z}_{L_P}^{\text{reg}}, (w_0nw,\psi) \mapsto \psi$  is smooth.

Since  $\operatorname{Ad}(nw)\psi \in \mathfrak{b}$  and  $\operatorname{Ad}(w_0)\mathfrak{b} \cap \mathfrak{b} = \mathfrak{t}$ , it follows that

$$Z = \{ (w_0 n w, \psi) \mid \operatorname{Ad}(w_0 n w) \psi \in \mathfrak{t} \} \simeq \{ (n, \psi) \in N \times \mathfrak{z}_{L_P}^{\operatorname{reg}} \mid \operatorname{Ad}(n w) \psi \in \mathfrak{t} \}.$$

Since  $\operatorname{Ad}(w)\psi \in \mathfrak{t}$  and the adjoint action of unipotent elements on  $\mathfrak{t}$  doesn't change the diagonal entries, we have

$$Z \simeq \{(n,\psi) \in N \times \mathfrak{z}_{L_P}^{\mathrm{reg}} \mid \operatorname{Ad}(nw)\psi = \operatorname{Ad}(w)\psi\} \simeq \{(n,\psi) \in N \times \mathfrak{z}_{L_P}^{\mathrm{reg}} \mid \operatorname{Ad}(w^{-1}nw)\psi = \psi\}$$
$$\simeq w(N \cap L_P)w^{-1} \times \mathfrak{z}_{L_P}^{\mathrm{reg}}$$

where the last isomorphism follows from  $C_G(\psi) = L_P$ . As  $w(N \cap L_P)w^{-1}$  is an (affine) smooth scheme, so is the projection  $w(N \cap L_P)w^{-1} \times \mathfrak{z}_{L_P}^{\text{reg}} \twoheadrightarrow \mathfrak{z}_{L_P}^{\text{reg}}$ , which finishes the proof.  $\Box$ 

# 5.3 Analysis of the local geometry

In this paragraph we study the local geometry of  $X_w$  at certain points.

**Theorem 5.3.1.** Let  $w \in \mathcal{W}$  and  $x = (g_1B, g_2P, 0) \in X_w$ , then the scheme  $X_w$  is unibranch at x.

*Proof.* By Remark 5.2.3 (2), it is enough to prove the same statement with the irreducible component  $Y_w := q_P^{-1}(\mathfrak{b})_w$  of the closed *E*-subscheme  $q_P^{-1}(\mathfrak{b})$  of  $G \times^P \mathfrak{r}_P$ . We want to prove that the normalization of the (reduced) local ring of  $Y_w$  at a point (g, 0)  $(g \in G)$  is still a local ring. The argument below is strongly inspired by the proof of [65, Lemma 3.4.8] and we give full details.

We see the *E*-scheme  $\mathbb{A}^1$  as an algebraic multiplicative monoid. The scheme  $G \times \mathfrak{r}_P$  is endowed with a left action of  $\mathbb{A}^1$  by  $a(g, \psi) := (g, a\psi)$  (where  $g \in G, \psi \in \mathfrak{r}_P$ ). As the adjoint action of *P* on  $\mathfrak{r}_P$  is linear, this action of  $\mathbb{A}^1$  descends to a left action on  $G \times^P \mathfrak{r}_P$ . It also preserves  $(Bw, \mathfrak{r}_P \cap \operatorname{Ad}(w^{-1})\mathfrak{b}) \subset G \times \mathfrak{r}_P$ , hence its image  $q_P^{-1}(\mathfrak{b})^0_w$  in  $G \times^P \mathfrak{r}_P$  (see Remark 5.2.3 (2)), hence its Zariski-closure  $Y_w$ . Let  $Y^{\mathbb{G}_m}_w \subseteq Y_w$  be the closed subscheme (with its reduced structure) of fixed points by  $\mathbb{G}_m$ , where  $\mathbb{G}_m$  is seen as a Zariski-open subgroup of the monoid  $\mathbb{A}^1$ . Since  $a\psi = \psi \ \forall a \in \mathbb{G}_m \Leftrightarrow \psi = 0$ , we have  $(G \times^P \mathfrak{r}_P)^{\mathbb{G}_m} \simeq G \times^P 0 \simeq G/P \times \{0\}$ , and we easily deduce that  $Y^{\mathbb{G}_m}_w \simeq C_w \times \{0\} \subseteq G/P \times \{0\}$  where  $C_w$  is the Zariski closure of BwP/P in G/P. In particular we see that the action of  $\mathbb{A}^1$  on  $Y^{\mathbb{G}_m}_w$  is also trivial.

Consider the normalization map  $f: \tilde{Y}_w \to Y_w$ , which is a finite surjective birational morphism between two noetherian irreducible *E*-schemes (see for instance [1, § 035E] and [1, § 0BXQ]). Note that *f* is an isomorphism above  $BwP/P \times \{0\} \subseteq Y_w^{\mathbb{G}_m}$  as  $BwP \times^P \{0\}$  is contained in the image of  $(Bw, \mathfrak{r}_P \cap \operatorname{Ad}(w^{-1})\mathfrak{b})$  in  $Y_w$ , which is the smooth open set  $q_P^{-1}(\mathfrak{b})_w^0$  of  $Y_w$  (see Remark 5.2.3 (2)). By [65, Rem. 3.4.3], it is enough to prove that the (geometric) fiber  $f^{-1}(g, 0)$  is a connected scheme for  $g \in G$ . This is a fortiori true if  $f^{-1}(g, 0)$  is geometrically connected, hence we can extend scalars from *E* to an algebraic closure of *E*, which we do from now on in this proof (still using the notation *E*).

The composition  $\mathbb{A}^1 \times \widetilde{Y}_w \longrightarrow \mathbb{A}^1 \times Y_w \longrightarrow Y_w$  is surjective (as both maps are surjective, where the last map is the action of  $\mathbb{A}^1$  on  $Y_w$ ). By [1, Lemma 035J], since  $\mathbb{A}^1 \times \widetilde{Y}_w$  is normal this composition factors as  $\mathbb{A}^1 \times \widetilde{Y}_w \longrightarrow \widetilde{Y}_w \longrightarrow Y_w$ , which induces a canonical action of  $\mathbb{A}^1$ on  $\widetilde{Y}_w$  such that the map f is  $\mathbb{A}^1$ -equivariant. By exactly the same argument, we also have an action of the Borel B on  $\widetilde{Y}_w$  such that f is B-equivariant. Moreover these two actions of  $\mathbb{A}^1$  and Bcommute on  $\widetilde{Y}_w$  as they do on  $Y_w$ . The action of  $\mathbb{A}^1$  on  $\widetilde{Y}_w$  is again trivial on the closed subscheme  $\widetilde{Y}_w^{\mathbb{G}_m} \subseteq \widetilde{Y}_w$  (with its reduced structure), but we need another argument than for  $Y_w^{\mathbb{G}_m}$ . Consider the morphism of E-schemes:

$$m: \mathbb{A}^1 \times Y_w \longrightarrow Y_w \times Y_w, \ (a, \widetilde{y}) \longmapsto (\widetilde{y}, a\widetilde{y})$$

and see  $\widetilde{Y}_{w}^{\mathbb{G}_{m}}$  as a closed subscheme of  $\widetilde{Y}_{w} \times \widetilde{Y}_{w}$  via the diagonal embedding. Then  $m^{-1}(\widetilde{Y}_{w}^{\mathbb{G}_{m}})$  is a closed subscheme of  $\mathbb{A}^{1} \times \widetilde{Y}_{w}$  which obviously contains the locally closed subscheme  $\mathbb{G}_{m} \times \widetilde{Y}_{w}^{\mathbb{G}_{m}}$ . Hence it also contains its Zariski-closure, which is  $\mathbb{A}^{1} \times \widetilde{Y}_{w}^{\mathbb{G}_{m}}$ . In particular the action of  $\mathbb{A}^{1}$  on  $\widetilde{Y}_{w}$  is trivial on  $\widetilde{Y}_{w}^{\mathbb{G}_{m}}$ .

We prove that we have an isomorphism  $\widetilde{Y}_w^{\mathbb{G}_m} \xrightarrow{\sim} f^{-1}(Y_w^{\mathbb{G}_m})^{\mathrm{red}}$  of (reduced) closed subschemes of  $\widetilde{Y}_w$ . Since f is  $\mathbb{G}_m$ -equivariant we have a closed embedding  $\widetilde{Y}_w^{\mathbb{G}_m} \hookrightarrow f^{-1}(Y_w^{\mathbb{G}_m})^{\mathrm{red}}$ , hence it is enough to prove that  $\mathbb{G}_m$  acts trivially on  $f^{-1}(Y_w^{\mathbb{G}_m})^{\mathrm{red}}$  inside  $\widetilde{Y}_w$ . Since we are over an algebraically closed field E, the action of  $\mathbb{G}_m$  on  $f^{-1}(Y_w^{\mathbb{G}_m})^{\mathrm{red}}$  is trivial if and only if the action of  $E^{\times}$  is trivial on the set of *E*-points  $f^{-1}(Y_w^{\mathbb{G}_m})(E)$  (see for instance Remark 4 in [43, p. 38]). Since any such *E*-point is in  $f^{-1}(g,0)(E)$  for some  $(g,0) \in Y_w^{\mathbb{G}_m}$ , it is enough to prove that  $E^{\times}$  acts trivially on  $f^{-1}(g,0)(E)$ . Since *f* is a finite  $\mathbb{G}_m$ -equivariant morphism,  $f^{-1}(g,0)(E)$  is a finite set stable under the action of  $E^{\times}$ . Hence there is an integer  $n \gg 0$  such that  $x^n$  acts trivially on  $f^{-1}(g,0)(E)$  for any  $x \in E^{\times}$  (for instance  $n = |f^{-1}(g,0)(E)|!$ ). But as *E* is assumed algebraically closed, any element in  $E^{\times}$  is of the form  $x^n$ , hence  $E^{\times}$  acts trivially on  $f^{-1}(g,0)(E)$ .

We now consider the morphism  $0: \widetilde{Y}_w \longrightarrow \widetilde{Y}_w^{\mathbb{G}_m}, x \longmapsto 0.x$  where  $0 \in \mathbb{A}^1$ . Since  $\mathbb{A}^1$  acts trivially on  $\widetilde{Y}_w^{\mathbb{G}_m} \subset \widetilde{Y}_w$ , the morphism 0 is surjective. As  $\widetilde{Y}_w$  is irreducible and the image of an irreducible set is irreducible, it follows that the closed subset  $\widetilde{Y}_w^{\mathbb{G}_m}$  of  $\widetilde{Y}_w$  is irreducible. Since  $\widetilde{Y}_w^{\mathbb{G}_m} \xrightarrow{\sim} f^{-1}(Y_w^{\mathbb{G}_m})^{\mathrm{red}}$ , we deduce that f induces a finite birational surjective morphism of noetherian irreducible E-schemes  $f^{-1}(Y_w^{\mathbb{G}_m})^{\mathrm{red}} \to Y_w^{\mathbb{G}_m}$  (it is birational since it is an isomorphism above the open subset  $BwP/P \times \{0\}$  of  $Y_w^{\mathbb{G}_m} \simeq C_w \times \{0\}$ ). But  $C_w$ , and hence  $Y_w^{\mathbb{G}_m}$ , are normal schemes by [70]. One then uses Zariski's connectedness theorem (see e.g. [72, § III.9.V]) applied to the morphism  $f^{-1}(Y_w^{\mathbb{G}_m})^{\mathrm{red}} \to Y_w^{\mathbb{G}_m}$ , which implies that all the fibers  $f^{-1}(g, 0)$  are connected schemes and finishes the proof.

**Remark 5.3.2.** The proof of Theorem 5.3.1 only works for the points (g, 0) of  $Y_w$ . In particular, when  $P \neq B$ , we do not know the unibranch property of  $Y_w$  at points  $(g, \psi) \in Y_w$  where  $\psi \in \mathfrak{r}_P \setminus \{0\}$ is not regular semi-simple (the regular semi-simple case being a consequence of Proposition 5.2.9). Recall that, when P = B, this is known since  $Y_w$  is normal at every point ([19, Thm. 2.3.6]).

If  $x \in X_w$  is a closed point, recall that the tangent space  $T_x X_w$  of  $X_w$  at x can be identified with the  $k(x)[\varepsilon]/(\varepsilon^2)$ -points of  $X_w$  mapping to x via  $k(x)[\varepsilon]/(\varepsilon^2) \twoheadrightarrow k(x), \varepsilon \mapsto 0$ . Recall also that  $\dim_{k(x)} T_x X_w \ge \dim X_w = \dim(G/B) + \dim \mathfrak{r}_P$ .

**Proposition 5.3.3.** Assume  $x = (g_1B, g_2P, 0) \in X_w \cap V_{w'}$  (which implies  $w'^{\max} \leq w^{\max}$  by Lemma 5.2.4), then

$$\dim_{k(x)} T_x X_w \leq \dim T_{\pi(x)} \overline{U_w} + \dim \mathfrak{z}_{L_P}^{w(w')^{-1}} + \lg((w')^{\max} w_0)$$
  
where  $\mathfrak{z}_{L_P}^{w(w')^{-1}} := \{ z \in \mathfrak{z}_{L_P} \mid \operatorname{Ad}(w(w')^{-1}) z = z \}.$ 

*Proof.* The proposition follows by similar arguments as in the proof of [19, Prop. 2.5.3], to which we refer the reader for more details (e.g. on the notation). The closed embedding  $X_P \hookrightarrow G/B \times G/P \times \mathfrak{g}$  induces a closed embedding  $X_w \hookrightarrow \overline{U_w} \times \mathfrak{g}$ . Let  $\hat{x} = (\hat{g}_1 B, \hat{g}_2 P, \psi \varepsilon) \in T_x X_w$ , where we see  $(\hat{g}_1 B, \hat{g}_2 P) \in T_{\pi(x)} \overline{U_w}$ . As  $\hat{x} \in X_P(k(x)[\varepsilon]/(\varepsilon^2))$ , we have  $\operatorname{Ad}(g_1^{-1})\psi \in \mathfrak{b}$  and  $\operatorname{Ad}(g_2^{-1})\psi \in \mathfrak{r}_P$ . As  $x \in V_{w'}$ , there exists  $g \in G$  such that  $(g_1 B, g_2 P) = (gw'B, gP)$ . Replacing  $g_1$  by gw', and  $g_2$ by g, we have thus

$$\operatorname{Ad}(g^{-1})\psi \in \mathfrak{r}_P \cap \operatorname{Ad}(w')\mathfrak{b}.$$
(5.11)

By Lemma 5.2.5, we have  $\kappa_P(\hat{x}) = \operatorname{Ad}(w)\kappa_B(\hat{x})$  and hence

$$\overline{\mathrm{Ad}(g^{-1})\psi} = \mathrm{Ad}(w)\overline{\mathrm{Ad}((w')^{-1}g^{-1})\psi} \in \mathfrak{z}_{L_P}.$$
(5.12)

(n-1)

Writing  $\operatorname{Ad}(g^{-1})\psi = \lambda + \eta$  with  $\lambda \in \mathfrak{z}_{L_P}$  and  $\eta \in \mathfrak{n}_P$ , we deduce from (5.11) and (5.12):

$$\eta \in \mathfrak{n}_P \cap \mathrm{Ad}(w')\mathfrak{n} \text{ and } \lambda \in \mathfrak{z}_{L_P}^{w(w')^{-1}}.$$
 (5.13)

We have  $\dim(\mathfrak{n}_P \cap \operatorname{Ad}(w')\mathfrak{n}) = \dim(\mathfrak{n} \cap \operatorname{Ad}(w'^{\max})\mathfrak{n}) = \lg(w'^{\max}w_0)$  (where the first equality follows from  $\mathfrak{n}_{L_P} \cap \operatorname{Ad}(w'^{\max})\mathfrak{n} = 0$ ). Together with (5.13), the proposition follows.

**Corollary 5.3.4.** With the notation of Proposition 5.3.3, assume that  $\overline{U_w}$  is smooth at the point  $\pi(x)$  and that

$$\dim \mathfrak{z}_{L_P}^{w(w')^{-1}} + \lg(w^{\max}) - \lg(w'^{\max}) = \dim \mathfrak{z}_{L_P}.$$
(5.14)

Then  $X_w$  is smooth at x.

*Proof.* Recall dim  $\overline{U_w} = \dim G - \dim B + \lg(w^{\min})$ . Under the assumptions in the statement, we have by Proposition 5.3.3 (and using  $\lg(w^{\max}) - \lg(w^{\min}) = \dim \mathfrak{n}_{L_P}$ ):

 $\dim_{k(x)} T_x X_w$  $\leq \dim G - \dim B + \lg(w^{\min}) + \dim \mathfrak{z}_{L_P} - \left(\lg(w^{\max}) - \lg((w')^{\max})\right) + \lg(w_0) - \lg((w')^{\max})$  $= \dim G - \dim B - \dim \mathfrak{n}_{L_P} + \dim \mathfrak{z}_{L_P} + \lg(w_0) = \dim G - \dim B + \dim \mathfrak{n}_P + \dim \mathfrak{z}_{L_P}$  $= \dim X_w.$ 

The corollary follows.

**Remark 5.3.5.** (1) By [10, Thm. 6.0.4] and [10, Cor. 6.2.11], if  $\lg(w'^{\max}) \ge \lg(w^{\max}) - 2$ , then  $\pi(x)$  is a smooth point of  $\overline{U_w}$ . Indeed, by loc. cit., under this assumption,  $\overline{PwB/B} = \overline{Bw^{\max}B/B}$  is smooth at any point of  $Bw'^{\max}B/B$ . Then using the P-action, we deduce that  $\overline{PwB/B}$  is smooth at any point of Pw'B/B, hence  $\overline{U_w}$  is smooth at any point of  $U'_w$ . If moreover  $\lg(w'^{\max}) \ge \lg(w^{\max}) - 1$ , it is clear that (5.14) holds, so  $X_w$  is smooth at x. When  $\lg(w'^{\max}) = \lg(w^{\max}) - 2$  however, (5.14) does not necessarily hold when  $P \neq B$ : for example, when P is a maximal parabolic subgroup, then  $\dim \mathfrak{Z}_{L_P}^{w(w')^{-1}} = 1 = \dim \mathfrak{Z}_{L_P} - 1 < \lg(w^{\max}) - \lg(w'^{\max})$ . See Remark A.3.5 for related discussions.

(2) The proof of Proposition 5.3.3 actually shows that its statement and the one of Corollary 5.3.4 hold with  $X_w$  replaced by  $X'_w$  in Remark 5.2.3 (1). In particular, the closed immersion  $X_w \hookrightarrow X'_w$  is an isomorphism on local rings at points satisfying the assumptions in Corollary 5.3.4.

## 5.4 Characteristic cycles

We study the fibres  $\kappa_{P,w}^{-1}(\{0\}) \hookrightarrow X_w$ , and show that they are closely related to the characteristic cycles of certain *G*-equivariant regular holonomic *D*-modules over  $G/B \times G/P$ .

Let  $Z_P := \kappa_B^{-1}(\{0\})^{\text{red}} = \kappa_P^{-1}(\{0\})^{\text{red}} \hookrightarrow X_P$  (see (5.6), one may view  $Z_P$  as a generalized Steinberg variety). Let  $\mathcal{N} \subset \mathfrak{g}$  denote the nilpotent cone and put  $\widetilde{\mathcal{N}}_P := \{(gP, \psi) \in G/P \times \mathcal{N} \mid \operatorname{Ad}(g^{-1})\psi \in \mathfrak{n}_P\}$ . We have  $\widetilde{\mathcal{N}}_P \cong G \times^P \mathfrak{n}_P$ ,  $(gP, \psi) \mapsto (g, \operatorname{Ad}(g^{-1})\psi)$ . We write  $\widetilde{\mathcal{N}} := \widetilde{\mathcal{N}}_B$ (defined as  $\widetilde{\mathcal{N}}_P$  with B instead of P). The morphism  $q_B$  (resp.  $q_P$ ) in § 5.1 restricts to the so-called Springer resolution (resp. generalized Springer resolution):  $q_B : G \times^B \mathfrak{n} \to \mathcal{N}$ ,  $(g, \psi) \mapsto \operatorname{Ad}(g)\psi$ (resp.  $q_P : G \times^P \mathfrak{n}_P \to \mathcal{N}$ ,  $(g, \psi) \mapsto \operatorname{Ad}(g)\psi$ ). We have

$$Z_P \cong G \times^P q_B^{-1}(\mathfrak{n}_P)^{\mathrm{red}} \cong G \times^B q_P^{-1}(\mathfrak{n})^{\mathrm{red}}.$$

The morphism  $\pi$  (see (5.3)) restricts to a natural morphism  $\pi_Z : Z_P \to G/B \times G/P$ . Similarly to § 5.2, for  $w \in \mathscr{W}$  we put  $V'_w := \pi_Z^{-1}(U_w) = V_w \cap Z_P$ , and let  $Z_w$  be the Zariski-closure of  $V'_w$  in  $Z_P$  with the reduced scheme structure. We put  $q_B^{-1}(\mathfrak{n}_P)^0_w := q_B^{-1}(\mathfrak{r}_P)^0_w \cap q_B^{-1}(\mathfrak{n}_P)$  and  $q_P^{-1}(\mathfrak{n})^0_w := q_P^{-1}(\mathfrak{b})^0_w \cap q_P^{-1}(\mathfrak{n})$  (see Remark 5.2.3 (2) for  $q_B^{-1}(\mathfrak{r}_P)^0_w$  and  $q_P^{-1}(\mathfrak{b})^0_w$ ). We define  $q_B^{-1}(\mathfrak{n}_P)_w$ (resp.  $q_P^{-1}(\mathfrak{n})_w$ ) as the Zariski-closure of  $q_B^{-1}(\mathfrak{n}_P)^0_w$  (resp. of  $q_P^{-1}(\mathfrak{n})^0_w$ ) in  $q_B^{-1}(\mathfrak{n}_P)$  (resp. in  $q_P^{-1}(\mathfrak{n})$ ). By similar arguments as in § 5.2, we have **Proposition 5.4.1.** (1) The scheme  $Z_P$  (resp.  $q_P^{-1}(\mathfrak{n})$ , resp.  $q_B^{-1}(\mathfrak{n}_P)$ ) is equidimensional of dimension dim  $\mathfrak{n}$  + dim  $\mathfrak{n}_P$  (resp. dim  $\mathfrak{n}_P$ , resp. dim  $\mathfrak{n}$ ) with set of irreducible components given by  $\{Z_w\}_{w\in \mathcal{W}/\mathcal{W}_{L_P}}$  (resp.  $\{q_P^{-1}(\mathfrak{n})_w\}_{w\in \mathcal{W}/\mathcal{W}_{L_P}}$ , resp.  $\{q_B^{-1}(\mathfrak{n}_P)_w\}_{w\in \mathcal{W}_{L_P}\setminus\mathcal{W}}$ ).

(2) We have  $Z_w \cong G \times^P q_B^{-1}(\mathfrak{n}_P)_w^{\mathrm{red}} \cong G \times^B q_P^{-1}(\mathfrak{n})_{w^{-1}}^{\mathrm{red}}$ .

**Remark 5.4.2.** One can show that  $q_B^{-1}(\mathfrak{n})$  is also equidimensional of dimension dim  $\mathfrak{n}$ , hence the subscheme  $q_B^{-1}(\mathfrak{n}_P)^{\mathrm{red}}$  of  $q_B^{-1}(\mathfrak{n})$  is isomorphic to a union of irreducible components of  $q_B^{-1}(\mathfrak{n})$ . The irreducible components of  $q_B^{-1}(\mathfrak{n})$  are  $\{q_B^{-1}(\mathfrak{n})_w\}_{w\in\mathscr{W}}$ , where  $q_B^{-1}(\mathfrak{n})_w$  is the Zariski closure of the preimage  $q_B^{-1}(\mathfrak{n})_w^0$  of BwB/B via  $q_B^{-1}(\mathfrak{n}) \to G/B$ . Using similar argument as in the proof of [19, Prop. 2.2.1] (see in particular [19, (2.6)]), one can show that  $q_B^{-1}(\mathfrak{n})_w^0 \subseteq q_B^{-1}(\mathfrak{n}_P)$  if and only if  $w \in \mathscr{W}_{\mathrm{max}}^P$ . If so we have  $q_B^{-1}(\mathfrak{n})_w^0 = q_B^{-1}(\mathfrak{n}_P)_w^0$ . Hence we deduce  $q_B^{-1}(\mathfrak{n}_P)_w = q_B^{-1}(\mathfrak{n})_{w^{\mathrm{max}}}$  for  $w \in \mathscr{W}$ .

By exactly the same argument as in the proof of Theorem 5.3.1 replacing everywhere  $\mathfrak{r}_P$  by  $\mathfrak{n}_P$ , we have the following theorem which is interesting in its own right (and which is new even for P = B, see [19, Rem. 2.4.2]):

#### **Theorem 5.4.3.** Let $w \in \mathcal{W}$ and $x = (g_1B, g_2P, 0) \in Z_w$ , then the scheme $Z_w$ is unibranch at x.

For a finite type *E*-scheme *Z*, denote by  $Z^0(Z)$  the free abelian group generated by the irreducible components  $\{Z_i\}$  of *Z*. Given a scheme *Y* whose underlying topological space is a union of irreducible components of *Z*, put

$$[Y] := \sum_{i} m(Z_i, Y)[Z_i]$$
(5.15)

where  $m(Z_i, Y)$  is the length of the  $\mathcal{O}_{Y,\eta_i}$ -module  $\mathcal{O}_{Y,\eta_i}$  at the generic point  $\eta_i$ .

For  $w \in \mathcal{W}$ , let  $\overline{X}_w := \kappa_{P,w}^{-1}(\{0\}) = X_w \times_{\kappa_{P,w},\mathfrak{z}_{L_P}} \{0\}$ , where " $\times_{\kappa_{P,w},\mathfrak{z}_{L_P}}$ " means taking the fibre product over  $\mathfrak{z}_{L_P}$  via the morphism  $\kappa_{P,w} : X_w \to \mathfrak{z}_{L_P}$ . Note that we do *not* take the reduced associated scheme. By the same argument as in page 320 of [19],  $\overline{X}_w$  is equidimensional of dimension dim Z. So each irreducible component of  $\overline{X}_w$  is  $Z_{w'}$  for some  $w' \in \mathcal{W}$ . We have the following easy fact.

**Lemma 5.4.4.** We have  $m(Z_w, \overline{X}_w) = 1$ , and  $m(Z_{w'}, \overline{X}_w) \geq 1$  implying  $w'^{\max} \leq w^{\max}$ .

*Proof.* By the proof of Proposition 5.2.1, we have  $V_w \times_{\kappa_{P,w},\mathfrak{z}_P} \{0\} \cong V_w \cap Z_P = V'_w$ . The first part follows. The second part follows easily from Corollary 5.2.4.

We construct some other cycles which are closely related to (the characteristic cycles of) parabolic Verma modules (see Proposition 5.4.8 below). Let  $\mathfrak{z}_{L_P}^{\mathrm{sd}} \subset \mathfrak{z}_{L_P}$  be the set of strictly dominant integral weight  $\lambda = (\lambda_1, \ldots, \lambda_r) \in \mathfrak{z}_{L_P}$ , i.e.  $\lambda_i \in \mathbb{Z}$  and  $\lambda_i > \lambda_{i'}$  for i > i'. In particular  $\mathfrak{z}_{L_P}^{\mathrm{sd}} \subset \mathfrak{z}_{L_P}^{\mathrm{reg}}$  (see the end of § 5.2 for  $\mathfrak{z}_{L_P}^{\mathrm{reg}}$ ). This assumption on  $\lambda$  will be used in the proof of Proposition 5.4.7 in Appendix A.2. Let  $\mathfrak{z}_{\lambda} := E\lambda \hookrightarrow \mathfrak{z}_{L_P}$  (a one-dimensional vector subspace of  $\mathfrak{z}_{L_P}$ ) and  $\mathfrak{r}_{P,\lambda} := \mathfrak{z}_{\lambda} + \mathfrak{n}_P \hookrightarrow \mathfrak{r}_P$ . Define:

$$\begin{split} \tilde{\mathfrak{g}}_{P,\lambda} &:= \quad G \times^P \mathfrak{r}_{P,\lambda} \hookrightarrow G \times^P \mathfrak{r}_P, \\ X_{P,\lambda} &:= \quad \tilde{\mathfrak{g}}_{P,\lambda} \times_{\mathfrak{g}} \tilde{\mathfrak{g}} \hookrightarrow X_P. \end{split}$$

Let  $\pi_{\lambda}$  be the restriction of  $\pi$  to  $X_{P,\lambda}$ . For  $w \in \mathscr{W}$  put  $V_{w,\lambda} := \pi_{\lambda}^{-1}(U_w)$ , which is a vector bundle of relative dimension dim  $\mathfrak{r}_{P,\lambda} - \lg(w^{\min})$  over  $U_w$ . We let  $X_{w,\lambda}$  be the reduced Zariski closure of  $V_{w,\lambda}$  in  $X_{P,\lambda}$ . We have that  $X_{P,\lambda}$  is equidimensional of dimension dim G – dim P + dim  $\mathfrak{r}_{P,\lambda}$  with irreducible components given by  $\{X_{w,\lambda}\}$ . Let  $\overline{X}_{w,\lambda} := X_{w,\lambda} \times_{\kappa_{P,w},\mathfrak{z}_{\lambda}} \{0\}$ .

**Lemma 5.4.5.** (1) We have  $m(Z_w, \overline{X}_{w,\lambda}) = 1$ , and  $m(Z_{w'}, \overline{X}_{w,\lambda}) \geq 1$  implies  $w'^{\max} \leq w^{\max}$ .

(2) We have  $X_{w,\lambda} \cong (X_w \times_{\kappa_{P,w},\mathfrak{z}_{L_P}} \mathfrak{z}_{\lambda})^{\text{red}}$ . Consequently, for  $w'^{\max} \leq w^{\max}$ ,  $m(Z_{w'}, \overline{X}_w) \geq m(Z_{w'}, \overline{X}_{w,\lambda})$ . Moreover  $m(Z_{w'}, \overline{X}_w) > 0$  if and only if  $m(Z_{w'}, \overline{X}_{w,\lambda}) > 0$ .

Proof. (1) follows by the same argument as for Lemma 5.4.4. We have a natural closed immersion  $X_{w,\lambda} \hookrightarrow (X_w \times_{\kappa_{P,w},\mathfrak{z}_{L_P}} \mathfrak{z}_{\lambda})^{\mathrm{red}}$ . By the same argument as on Page 320 of [19], each irreducible component of  $X_w \times_{\kappa_{P,w},\mathfrak{z}_{L_P}} \mathfrak{z}_{\lambda}$  has dimension  $\dim X_w - (\dim \mathfrak{z}_{L_P} - \dim \mathfrak{z}_{\lambda}) = \dim X_{w,\lambda}$  and is thus some  $X_{w',\lambda}$  for  $w' \in \mathscr{W}$ . However, using Lemma 5.2.6, it is easy to see that, if  $\mathscr{W}_{L_P}w' \neq \mathscr{W}_{L_P}w$ , then  $V_{w',\lambda}$  can not be contained in  $(X_w \times_{\kappa_{P,w},\mathfrak{z}_{L_P}} \mathfrak{z}_{\lambda})^{\mathrm{red}}$ . The first part of (2) follows. Together with the isomorphism  $\overline{X}_w \cong (X_w \times_{\kappa_{P,w},\mathfrak{z}_{L_P}} \mathfrak{z}_{\lambda}) \times_{\kappa_{P,w},\mathfrak{z}_{\lambda}} \{0\}$ , the other parts also follows.

**Remark 5.4.6.** If P is maximal, then  $\mathfrak{z}_{\lambda} + \mathfrak{z} = \mathfrak{z}_{L_P}$ . We have in this case  $X_w \cong \mathfrak{z} \times X_{w,\lambda}$ , and hence  $\overline{X}_w \cong \overline{X}_{w,\lambda}$ . If P = B, then  $X_w$  is Cohen-Macaulay by [19, Prop. 2.3.3]. As  $X_w \times_{\kappa_{P,w},\mathfrak{z}_{L_P}} \mathfrak{z}_{\lambda}$ is cut out by  $(\dim X_w - \dim X_w \times_{\kappa_{P,w},\mathfrak{z}_{L_P}} \mathfrak{z}_{\lambda})$ -equations in  $X_w$ , we can deduce that  $X_w \times_{\kappa_{P,w},\mathfrak{z}_{L_P}} \mathfrak{z}_{\lambda}$ is also Cohen-Macaulay. Using Proposition 5.2.9 and similar arguments in the proof of [19, Thm. 2.2.6], we can show that  $X_w \times_{\kappa_{P,w},\mathfrak{z}_{L_P}} \mathfrak{z}_{\lambda}$  is reduced, hence equal to  $X_{w,\lambda}$ . By the proof of Lemma 5.4.5, we have in this case  $\overline{X}_{w,\lambda} \cong \overline{X}_w$ .

Let  $\mathcal{O}^{\mathfrak{p}}$  be the parabolic BGG category  $\mathcal{O}$  associated to  $\mathfrak{p}$  (cf. [57, § 9.3]) and  $\mathcal{O}^{\mathfrak{p}}(0)$  the full subcategory of  $\mathcal{O}^{\mathfrak{p}}$  consisting of objects with trivial infinitesimal character. Let  $\operatorname{Mod}_{\operatorname{rh}}(D_{G/B}, P)$ be the category of regular holonomic *P*-equivariant *D*-modules over G/B, which is the same as the category of coherent *P*-equivariant *D*-modules over G/B by [54, Thm. 11.6.1]. By the Beilinson-Bernstein localization, we have an equivalence of categories with inverse given by taking global sections (see for example [54, Thm. 11.5.3]):

$$\operatorname{Loc}_{\operatorname{BB}} : \mathcal{O}^{\mathfrak{p}}(0) \xrightarrow{\sim} \operatorname{Mod}_{\operatorname{rh}}(D_{G/B}, P).$$

Let  $\operatorname{Mod}_{\operatorname{rh}}(D_{G/P}, B)$  (resp.  $\operatorname{Mod}_{\operatorname{rh}}(D_{G/B \times G/P}, G)$ ) be the category of regular holonomic *B*-equivariant (resp. *G*-equivariant for the diagonal *G*-action) *D*-modules over G/P (resp. over  $G/B \times G/P$ ). Put

$$i_B: G/B \hookrightarrow G/B \times G/P, \ gB \mapsto (gB, P),$$
  
 $i_P: G/P \hookrightarrow G/B \times G/P, \ gP \mapsto (B, gP).$ 

As in [54, Prop. 13.1.1] (see also [82, Lemma 1.4]), we have  $R^j i_B^* \mathfrak{M} = 0$  (resp.  $R^j i_P^* \mathfrak{M} = 0$ ) for  $\mathfrak{M} \in \operatorname{Mod}_{\operatorname{rh}}(D_{G/B}, P)$  (resp. for  $\mathfrak{M} \in \operatorname{Mod}_{\operatorname{rh}}(D_{G/P}, B)$ ) and j > 0. Moreover the functor  $i_B^*$  (resp.  $i_P^*$ ) induces an equivalence of categories  $\operatorname{Mod}_{\operatorname{rh}}(D_{G/B \times G/P}, G) \xrightarrow{\sim} \operatorname{Mod}_{\operatorname{rh}}(D_{G/B}, P)$  (resp.  $\operatorname{Mod}_{\operatorname{rh}}(D_{G/B \times G/P}, G) \xrightarrow{\sim} \operatorname{Mod}_{\operatorname{rh}}(D_{G/P}, B)$ ).

For a smooth algebraic variety Y, let  $T^*Y$  be the cotangent bundle. For a regular holonomic D-module  $\mathfrak{M}$  over Y, denote by  $\operatorname{Ch}(\mathfrak{M}) \subset T^*Y$  the associated characteristic variety (cf. [54, § 2.2]). We have  $T^*G/B \cong G \times^B (\mathfrak{g/b})^{\vee} \cong G \times^B \mathfrak{n}$ , and  $T^*G/P \cong G \times^P (\mathfrak{g/p})^{\vee} \cong G \times^P \mathfrak{n}_P$  (see for example [31, Lemma 1.4.9] where we identify  $\mathfrak{g}$  with  $\mathfrak{g}^{\vee}$  using the Killing form, cf. [58, § 5]). By the same argument as in the proof of [19, Prop. 2.4.4], we have:

**Proposition 5.4.7.** For  $\mathfrak{M} \in Mod_{rh}(D_{G/B \times G/P}, G)$ 

$$\operatorname{Ch}(\mathfrak{M}) \cong G \times^B \operatorname{Ch}(i_P^*\mathfrak{M}) \cong G \times^P \operatorname{Ch}(i_B^*\mathfrak{M}) \subseteq T^*(G/B \times G/P)$$

is equidimensional of dimension  $\dim Z_P = \dim G/B + \dim G/P$ . In particular, the underlying topological space of Ch( $\mathfrak{M}$ ) is a union of irreducible components of  $Z_P$ .

For  $\mathfrak{M} \in \operatorname{Mod}_{\operatorname{rh}}(D_{G/B \times G/P}, G)$ ,  $[\operatorname{Ch}(\mathfrak{M})] \in Z^0(Z_P)$  is thus well-defined. For  $w \in \mathcal{W}$ , let

 $M_P(w^{\max}w_0 \cdot 0) := \mathcal{U}(\mathfrak{g}) \otimes_{\mathcal{U}(\mathfrak{p})} (w^{\max}w_0 \cdot 0) \in \mathcal{O}^{\mathfrak{p}}(0)$ 

be the parabolic Verma module, and  $L(w^{\max}w_0 \cdot 0)$  be the (unique) simple quotient of  $M_P(w^{\max}w_0 \cdot 0)$ 0) in  $\mathcal{O}^{\mathfrak{p}}(0)$ . Denote by  $\mathfrak{M}_P(w^{\max}w_0 \cdot 0)$  (resp.  $\mathfrak{L}(w^{\max}w_0 \cdot 0)$ ) the *D*-module over G/B associated to  $M_P(w^{\max}w_0 \cdot 0)$  (resp.  $L(w^{\max}w_0 \cdot 0)$ ) via  $i_B^* \circ \operatorname{Loc}_{BB}$ . The following proposition will be proved in Appendix A.2.

**Proposition 5.4.8.** Let  $\lambda \in \mathfrak{z}_{L_P}^{\mathrm{sd}}$ , then we have  $[\overline{X}_{w,\lambda}] = [\mathrm{Ch}(\mathfrak{M}_P(w^{\max}w_0 \cdot 0))]$  for all  $w \in \mathscr{W}$ .

Conjecture 5.4.9. For  $w \in \mathcal{W}$ , we have  $[\overline{X}_w] = [Ch(\mathfrak{M}_P(w^{\max}w_0 \cdot 0))].$ 

**Remark 5.4.10.** Conjecture 5.4.9 holds for P = B by [45, (6.2.3)] or [9, Prop. 2.14.2] (see Remark 5.4.6). It also holds in the case P is maximal by Proposition 5.4.8 and Remark 5.4.6.

**Theorem 5.4.11.** (1) The classes  $\{Z_w\}$ ,  $\{[Ch(\mathfrak{M}_P(w^{\max}w_0 \cdot 0))]\}$ ,  $\{[Ch(\mathfrak{L}(w^{\max}w_0 \cdot 0)]\}, \{[\overline{X}_w]\}$  for  $w \in \mathscr{W}_{L_P} \setminus \mathscr{W}$  are a basis of the finite free  $\mathbb{Z}$ -module  $Z^0(Z_P)$ .

(2) For  $\lambda \in \mathfrak{z}_{L_P}^{\mathrm{sd}}$  and  $w \in \mathscr{W}$ , we have

$$[\overline{X}_{w,\lambda}] = \sum_{w' \in \mathscr{W}_{L_P} \setminus \mathscr{W}} b_{w,w'} [\operatorname{Ch}(\mathfrak{L}(w'^{\max}w_0 \cdot 0))]$$
(5.16)

where  $b_{w,w'} \in \mathbb{Z}_{\geq 0}$  is the multiplicity of  $L(w'^{\max}w_0 \cdot 0)$  in  $M_P(w^{\max}w_0 \cdot 0)$  (hence  $b_{w,w'}$  only depends on the cosets  $\mathscr{W}_{L_P}w$  and  $\mathscr{W}_{L_P}w'$ ,  $b_{w,w} = 1$  and  $b_{w,w'} = 0$  unless  $w'^{\max} \leq w^{\max}$ ).

(3) Let  $w, w' \in \mathcal{W}$ , there are integers  $a_{w,w'} \in \mathbb{Z}_{\geq 0}$  only depending on the cosets  $\mathcal{W}_{L_P}w, \mathcal{W}_{L_P}w'$  such that

$$[\operatorname{Ch}(\mathfrak{L}(w^{\max}w_0\cdot 0))] = \sum_{w'\in\mathscr{W}_{L_P}\setminus\mathscr{W}} a_{w,w'}[Z_{w'}] \in Z^0(Z_P)$$

where  $a_{w,w} = 1$  and  $a_{w,w'} = 0$  unless  $w'^{\max} \le w^{\max}$ . Moreover, if  $w'^{\max} < w^{\max}$  and  $Bw'^{\max}B/B$  is contained in the smooth locus of the Zariski-closure of  $Bw^{\max}B/B$ , then  $a_{w,w'} = 0$ .

*Proof.* (1) follows from the same argument as in the proof of [19, Thm. 2.4.7] (with [19, Prop. 2.4.6] replaced by Proposition 5.4.8 and using Lemma 5.4.5 (2)). (2) is a direct consequence of Proposition 5.4.8. By Proposition 5.4.7, (3) will follow from a parallel statement for  $[Ch(i_B^*\mathfrak{L}(w^{\max}w_0 \cdot 0))]$ . Let  $a_{w,w'} \in \mathbb{Z}_{\geq 0}$  such that (see Remark 5.4.2 for the second equality):

$$[\operatorname{Ch}(i_B^*\mathfrak{L}(w^{\max}w_0\cdot 0))] = \sum_{w'\in\mathscr{W}_{L_P}\setminus\mathscr{W}} a_{w,w'}[q_B^{-1}(\mathfrak{n}_P)_{w'}] = \sum_{w'\in\mathscr{W}_{L_P}\setminus\mathscr{W}} a_{w,w'}[q_B^{-1}(\mathfrak{n})_{w'^{\max}}].$$

(3) follows then from [19, Thm. 2.4.7 (iii)].

**Remark 5.4.12.** By [27, Conj. 3.27] and [27, Thm. 3.28] (see also [34], [39]), the coefficients  $b_{w,w'}$  in Theorem 5.4.11 may be described using certain relative Kazhdan-Lusztig polynomials. However, we couldn't find a precise statement in the literature and we don't need such description in the paper. We remark that, by [27, Thm. 3.28],  $b_{w,w'}$  is also equal to the multiplicity of  $L(w'^{\max}w_0 \cdot \lambda)$  in  $M_P(w^{\max}w_0 \cdot \lambda)$  for any integral dominant weight  $\lambda$ . Finally, using [57, Thm. 9.4 (b)] and [57, Ex. 8.3 (a)], we easily deduce  $b_{w,w'} = 1$  when  $w'^{\max} < w^{\max}$  and  $\lg(w'^{\max}) = \lg(w^{\max}) - 1$ .

Let  $\overline{X}_P := X_P \times_{\kappa_P, \mathfrak{f}_{L_P}} \{0\}$ . If M is a coherent  $\mathcal{O}_{\overline{X}_P}$ -module, we define its class  $[M] \in Z^0(Z_P)$ as in (5.15) with  $m(Z_w, Y)$  replaced by the length  $m(Z_w, M)$  of the  $\mathcal{O}_{\overline{X}_P, \eta_{Z_w}}$ -module  $M_{\eta_{Z_w}}$ . Let x be a closed point in  $\overline{X}_P$  (or equivalently in  $Z_P$ ), the complete local rings  $\widehat{\mathcal{O}}_{Z_P, x}$ ,  $\widehat{\mathcal{O}}_{Z_w, x}$ are equidimenisonal, and the set of irreducible components of  $\operatorname{Spec} \widehat{\mathcal{O}}_{Z_P, x}$  is the (disjoint) union for all  $w \in \mathscr{W}_{L_P} \setminus \mathscr{W}$  of the sets of irreducible components of  $\operatorname{Spec} \widehat{\mathcal{O}}_{Z_w, x}$ . Note that in general we don't know whether  $\operatorname{Spec} \widehat{\mathcal{O}}_{Z_w, x}$  is irreducible (see the discussion above [19, Lemma 2.5.5]). However if the  $\mathfrak{n}_P$ -coordinate of x is zero, then  $\operatorname{Spec} \widehat{\mathcal{O}}_{Z_w, x}$  is irreducible by Theorem 5.4.3. Put  $\widehat{M}_x := M \otimes_{\mathcal{O}_{\overline{X}}} \widehat{\mathcal{O}}_{\overline{X}_P, x}$ , and define  $[\widehat{M}_x] \in Z^0(\operatorname{Spec} \widehat{\mathcal{O}}_{Z_P, x})$  similarly to [M] above. By the same argument as in the proof of [19, Lemma 2.5.5], we have

$$[\widehat{M}_x] = \sum_{w \in \mathscr{W}_{L_P} \setminus \mathscr{W}} m(Z_w, M)[\operatorname{Spec} \widehat{\mathcal{O}}_{Z_w, x}] \in Z^0(\operatorname{Spec} \widehat{\mathcal{O}}_{Z_P, x}),$$
(5.17)

where  $[\operatorname{Spec} \widehat{\mathcal{O}}_{Z_w,x}] \in Z^0(\operatorname{Spec} \widehat{\mathcal{O}}_{Z_P,x})$  is defined similarly as in (5.15). For  $w \in \mathscr{W}_{L_P} \setminus \mathscr{W}$ , put (see Theorem 5.4.11 (3) for  $a_{w,w'} \in \mathbb{Z}_{\geq 0}$ ):

$$[\widehat{\mathfrak{L}}(w^{\max}w_0 \cdot 0)_x] := \sum_{w' \in \mathscr{W}_{L_P} \setminus \mathscr{W}} a_{w,w'}[\operatorname{Spec} \widehat{\mathcal{O}}_{Z_{w'},x}] \in Z^0(\operatorname{Spec} \widehat{\mathcal{O}}_{Z_P,x}).$$

Lemma 5.4.13. Let  $w \in \mathcal{W}$ .

(1) Let  $\lambda \in \mathfrak{z}_{L_P}^{\mathrm{sd}}$ , we have

$$[\widehat{\mathcal{O}}_{\overline{X}_{w,\lambda},x}] = \sum_{w' \in \mathscr{W}_{L_P} \setminus \mathscr{W}} b_{w,w'} [\widehat{\mathfrak{L}}(w'^{\max}w_0 \cdot 0)_x] \in Z^0(\operatorname{Spec} \widehat{\mathcal{O}}_{Z_P,x}).$$
(5.18)

(2) For  $w' \in \mathcal{W}$  there are integers  $c_{w,w'}, c'_{w,w'} \in \mathbb{Z}_{\geq 0}$ , depending only on the cosets  $\mathcal{W}_{L_P}w$  and  $\mathcal{W}_{L_P}w'$ , satisfying the following conditions:

- $c_{w,w} = c'_{w,w} = 1;$
- $c'_{w,w'} \ge c_{w,w'};$
- $c_{w,w'} > 0 \iff c'_{w,w'} > 0;$
- $c_{w,w'} = c'_{w,w'} = 0$  except when  $w'^{\max} \le w^{\max}$ ;
- we have the equalities in  $Z^0(\operatorname{Spec} \widehat{\mathcal{O}}_{Z_P,x})$ :

$$[\widehat{\mathcal{O}}_{\overline{X}_{w,\lambda},x}] = \sum_{w' \in \mathscr{W}_{L_P} \setminus \mathscr{W}} c_{w,w'}[\operatorname{Spec} \widehat{\mathcal{O}}_{Z_{w'},x}] \quad and \quad [\widehat{\mathcal{O}}_{\overline{X}_{w},x}] = \sum_{w' \in \mathscr{W}_{L_P} \setminus \mathscr{W}} c'_{w,w'}[\operatorname{Spec} \widehat{\mathcal{O}}_{Z_{w'},x}].$$

(3) If  $x \in \overline{X}_w$  is a smooth point of  $X_w$ , then  $\widehat{\mathcal{O}}_{\overline{X}_w,x} \cong \widehat{\mathcal{O}}_{\overline{X}_{w,\lambda},x}$ , in particular (5.18) holds with  $\widehat{\mathcal{O}}_{\overline{X}_{w,\lambda},x}$  replaced by  $\widehat{\mathcal{O}}_{\overline{X}_w,x}$ .

*Proof.* (1) follows from (5.16) using (5.17). (2) follows from Lemma 5.4.5 by putting  $c_{w,w'} := m(Z_{w'}, \overline{X}_{w,\lambda})$  and  $c'_{w,w'} := m(Z_{w'}, \overline{X}_w)$ . For (3), it suffices to show  $\mathcal{O}_{\overline{X}_{w,\lambda},x} \cong \mathcal{O}_{\overline{X}_{w,\lambda}}$  if  $X_w$  is smooth at x. Using the description of  $\overline{X}_{w,\lambda}$  and  $\overline{X}_w$  in the proof of Lemma 5.4.5, we only need to show that  $X'_{w,\lambda} := X_w \times_{\kappa_{P,w},\mathfrak{s}_{L_P}} \mathfrak{z}_{\lambda}$  is reduced at the point x. It is easy to see that  $X_{w,\lambda}$  is cut out by  $(\dim \mathfrak{z}_{L_P} - \dim \mathfrak{z}_{\lambda})$ -equations in  $X_w$ . As  $X_w$  is smooth at x and  $\dim X_{w,\lambda} = \dim X_w - (\dim \mathfrak{z}_{L_P} - \dim \mathfrak{z}_{\lambda})$ , there exists an open neighbourhood  $U \subset X'_{w,\lambda}$  of x such that U is a local complete intersection, hence Cohen-Macaulay. It is thus sufficient to show that U is generically reduced (cf. [47, Prop. 5.8.5]). By Proposition 5.2.9,  $X_P^{\text{reg}-\text{ss}}$  is smooth. It is also Zariski-open and Zariski-dense in  $X_P$ . We deduce  $X_P^{\text{reg}-\text{ss}} = \sqcup_{w' \in \mathscr{W}/\mathscr{W}_{L_P}}(X_P^{\text{reg}-\text{ss}} \cap X_{w'})$  where  $\{X_P^{\text{reg}-\text{ss}} \cap X_{w'}\}$  are the irreducible components of  $X_P^{\text{reg}-\text{ss}}$ . In particular, we have that  $X_P^{\text{reg}-\text{ss}} \cap X_w$  is Zariski-open and Zariski-dense in  $X_w$ . Moreover  $\kappa_P : X_P^{\text{reg}-\text{ss}} \cap X_w \to \mathfrak{z}_{L_P}$  is smooth by Proposition 5.2.9. We deduce that  $X_{w,\lambda}^{\text{reg}-\text{ss}} \cap X_w \to \mathfrak{z}_{L_P}$  is smooth by Proposition 5.2.9. We deduce that  $X_{w,\lambda}^{\text{reg}-\text{ss}} \in (X_P^{\text{reg}-\text{ss}} \cap X_w) \to \mathfrak{z}_{L_P}$  is smooth by Proposition 5.2.9. We deduce that  $X_{w,\lambda}^{\text{reg}-\text{ss}} := (X_P^{\text{reg}-\text{ss}} \cap X_w) \times_{\kappa_P,\mathfrak{z}_{L_P}} \mathfrak{z}_{\lambda}$  is smooth and is Zariski-open in  $X'_w$ . One easily checks  $X_P^{\text{reg}-\text{ss}} \cap W_{w,\lambda} \neq \emptyset$  (e.g. using (5.5)), so  $X_P^{\text{reg}-\text{ss}} \cap V_{w,\lambda}$  is non-empty Zariski-open, hence Zariski-dense, in  $V_{w,\lambda}$ , which implies that  $X_{w,\lambda}^{\text{reg}-\text{ss}}$  is Zariski-dense in  $X_{w,\lambda}$ . As  $X_{w,\lambda}^{\text{reg}-\text{ss}}$  is smooth, we see that U is generically smooth hence generically reduced. The lemma follows.

# 6 Local models for the Bernstein paraboline varieties

Generalizing [19, § 3], we show that the geometry of the Bernstein paraboline varieties of § 4 is closely related to the schemes considered in § 5. We use the notation of § 4 and § 5 applied to  $G = \operatorname{GL}_n$ . When applied to  $G = (\operatorname{Res}_{\mathbb{Q}_p}^L \operatorname{GL}_n) \times_{\operatorname{Spec}}_{\mathbb{Q}_p} \operatorname{Spec} E \simeq \prod_{\Sigma_L} \operatorname{GL}_n$ , we modify the notation of § 5 by adding a subscript "L" to each scheme considered in § 5 (to stress the field "L" and to distinguish from the case  $G = \operatorname{GL}_n$ ), for instance  $\mathfrak{g}_L$  is  $\mathfrak{g} \otimes_{\mathbb{Q}_p} L$ ,  $X_{P,L} \cong \prod_{\tau \in \Sigma_L} X_P$  and we have  $\tilde{\mathfrak{g}}_{P,L}$ , etc.

### 6.1 Almost de Rham $B_{dR}$ -representations

We define and study certain groupoids of deformations of an almost de Rham  $B_{dR}$ -representation of  $Gal_L$ .

Let  $B_{pdR}^+$  be the algebra  $B_{dR}^+[\log(t)]$  defined in [44, § 4.3] and  $B_{pdR} = B_{pdR}^+ \otimes_{B_{dR}^+} B_{dR}$ . Recall  $B_{pdR}^+$  (resp.  $B_{pdR}$ ) is equipped with a natural action of  $Gal_L$  extending the usual  $Gal_L$ -action on  $B_{dR}^+$  (resp.  $B_{dR}$ ) such that  $g(\log(t)) = \log(t) + \log(\chi_{cyc}(g))$ . Moreover there is a unique  $B_{dR}$ -derivation  $\nu_{B_{pdR}}$  of  $B_{pdR}$  such that  $\nu_{pdR}(\log(t)) = -1$ . It is clear that  $\nu_{B_{pdR}}$  preserves  $B_{pdR}^+$  and commutes with  $Gal_L$ .

We fix an almost de Rham representation W of  $\operatorname{Gal}_L$  over  $B_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} E$ , i.e. W is a free  $B_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} E$ module equipped with a semi-linear  $\operatorname{Gal}_L$ -action such that  $\dim_L(B_{\mathrm{pdR}} \otimes_{B_{\mathrm{dR}}} W)^{\operatorname{Gal}_L} = \dim_{B_{\mathrm{dR}}} W$ (cf. [44, Thm. 4.1 (2)]). Let  $P \subseteq \operatorname{GL}_n$  be a parabolic subgroup as in § 2.1. Let  $\mathcal{F}_{\bullet} = (\mathcal{F}_i)_{0 \leq i \leq r}$  be a P-filtration on W, i.e.  $0 = \mathcal{F}_0 \subseteq \mathcal{F}_1 \subseteq \cdots \subseteq \mathcal{F}_r$  are  $B_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} E$ -subrepresentations of W such that  $\mathcal{F}_i/\mathcal{F}_{i-1}$  is free of rank  $n_i$  over  $B_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} E$  for  $1 \leq i \leq r$ . We assume that  $\mathcal{F}_i/\mathcal{F}_{i-1}$  is de Rham for  $1 \leq i \leq r$ , so  $\mathcal{F}_i/\mathcal{F}_{i-1} \cong (B_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} E)^{\oplus n_i}$  as  $\mathrm{Gal}_L$ -representation.

For A in Art(E), we call  $B_{dR} \otimes_{\mathbb{Q}_p} A$ -representation of  $\operatorname{Gal}_L$  a free  $B_{dR} \otimes_{\mathbb{Q}_p} A$ -module of finite rank endowed with a semi-linear action of  $\operatorname{Gal}_L$  (so  $\operatorname{Gal}_L$  acts trivially on A). We define  $X_{W,\mathcal{F}_{\bullet}}$ as the following groupoid over  $\operatorname{Art}(E)$ :

(1) The objects of  $X_{W,\mathcal{F}_{\bullet}}$  are the quadruples  $(A, W_A, \mathcal{F}_{A,\bullet}, \iota_A)$  where

- $A \in \operatorname{Art}(E)$  and  $W_A$  is a  $B_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} A$ -representation of  $\operatorname{Gal}_L$ ;
- $\mathcal{F}_{A,\bullet} = (\mathcal{F}_{A,i})_{0 \leq i \leq r}$  is a *P*-filtration on  $W_A$  by  $B_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} A$ -subrepresentations of  $\mathrm{Gal}_L$ such that  $\mathcal{F}_{A,0} = 0$  and  $\mathcal{F}_{A,i}/\mathcal{F}_{A,i-1}$ ,  $1 \leq i \leq r$ , is free of rank  $n_i$  over  $B_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} A$  and isomorphic to  $(\mathcal{F}_i/\mathcal{F}_{i-1}) \otimes_{B_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} E} \varepsilon_{A,i}$  for some rank one  $B_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} A$ -representation  $\varepsilon_{A,i}$ ;
- $\iota_A : W_A \otimes_A E \xrightarrow{\sim} W$  is an isomorphism of  $B_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} E$ -representations which induces isomorphisms  $\iota_A : \mathcal{F}_{A,i} \otimes_A E \xrightarrow{\sim} \mathcal{F}_i$  for  $i \in \{0, \ldots, r\}$ .
- (2) A morphism  $(A, W_A, \mathcal{F}_{A, \bullet}, \iota_A) \to (A', W_{A'}, \mathcal{F}_{A', \bullet}, \iota_{A'})$  is a morphism  $A \to A'$  in Art(E) and an isomorphism  $W_A \otimes_A A' \xrightarrow{\sim} W_{A'}$  of  $B_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} A'$ -representations which is compatible with  $\iota_A$  and  $\iota_{A'}$  and induces isomorphisms  $\mathcal{F}_{A,i} \otimes_A A' \xrightarrow{\sim} \mathcal{F}_{A',i}$  for  $i \in \{0, \ldots, r\}$ .

Note that the rank one  $\varepsilon_{A,i}$  in (1) and the  $\mathcal{F}_{A,i}$  are automatically almost de Rham since extensions of almost de Rham representations are always almost de Rham by [44, § 3.7].

We fix an isomorphism of  $L \otimes_{\mathbb{Q}_p} E$ -modules:

$$\alpha: (L \otimes_{\mathbb{Q}_p} E)^n \xrightarrow{\sim} D_{pdR}(W) := (B_{pdR} \otimes_{B_{dR}} W)^{\operatorname{Gal}_L},$$

and we let  $X_W, X_W^{\Box}$  be the groupoids over  $\operatorname{Art}(E)$  defined as in [19, § 3.1] where " $(-)^{\Box}$ " is with respect to  $\alpha$ . We have a natural morphism of groupoids  $X_{W,\mathcal{F}_{\bullet}} \to X_W$  sending  $(A, W_A, \mathcal{F}_{A,\bullet}, \iota_A)$ to  $(A, W_A, \iota_A)$ . We put  $X_{W,\mathcal{F}_{\bullet}}^{\Box} := X_{W,\mathcal{F}_{\bullet}} \times_{X_W} X_W^{\Box}$ . The objects of  $X_{W,\mathcal{F}_{\bullet}}^{\Box}$  are the 5-tuples  $(A, W_A, \mathcal{F}_{A,\bullet}, \iota_A, \alpha_A)$  where  $(A, W_A, \mathcal{F}_{A,\bullet}, \iota_A)$  is an object in  $X_{W,\mathcal{F}_{\bullet}}$  and  $\alpha_A$  is an isomorphism  $\alpha_A : (L \otimes_{\mathbb{Q}_p} A)^n \xrightarrow{\sim} D_{pdR}(W_A)$  such that  $\alpha_A \equiv \alpha \pmod{\mathfrak{m}_A}$ . A morphism  $(A, W_A, \mathcal{F}_{A,\bullet}, \iota_A, \alpha_A) \to$  $(A', W_{A'}, \mathcal{F}_{A',\bullet}, \iota_{A'}, \alpha_{A'})$  is a morphism  $(A, W_A, \mathcal{F}_{A,\bullet}, \iota_A) \to (A', W_{A'}, \mathcal{F}_{A',\bullet}, \iota_{A'})$  in  $X_{W,\mathcal{F}_{\bullet}}$  such that the following diagram commutes

Let  $(A, W_A, \mathcal{F}_{A, \bullet}, \iota_A, \alpha_A)$  be an object in  $X_{W, \mathcal{F}_{\bullet}}^{\sqcup}$ . Recall that the  $B_{dR}$ -derivation  $\nu_{B_{pdR}}$  on  $B_{pdR}$ induces an  $L \otimes_{\mathbb{Q}_p} A$ -linear nilpotent operator  $\nu_{W_A}$  on  $D_{pdR}(W_A)$ . We denote the matrix  $\alpha_A^{-1} \circ$  $\nu_{W_A} \circ \alpha_A$  by  $N_{W_A} \in M_n(L \otimes_{\mathbb{Q}_p} A) = \mathfrak{g}_L(A)$ . We let  $\mathcal{D}_{A, \bullet} := (\mathcal{D}_{A, i})_i$  with  $\mathcal{D}_{A, i} := D_{pdR}(\mathcal{F}_{A, i})$ .

**Lemma 6.1.1.** With the above notation  $(\alpha_A^{-1}(\mathcal{D}_{A,\bullet}), N_{W_A}) \in \widetilde{\mathfrak{g}}_{P,L}(A)$ .

*Proof.* The *P*-filtration  $\mathcal{D}_{A,\bullet}$  is stable by  $\nu_{W_A}$ . It is then sufficient to show that the induced action of  $\nu_{W_A}$  on  $\mathcal{D}_{A,i}/\mathcal{D}_{A,i-1}$  is a scalar (in  $L \otimes_{\mathbb{Q}_p} A$ ). Since the  $\mathcal{F}_{A,i}$  are almost de Rham and the  $\mathcal{F}_i/\mathcal{F}_{i-1}$ 

are de Rham, we have an isomorphism of *E*-vector spaces  $\mathcal{D}_{A,i}/\mathcal{D}_{A,i-1} \cong D_{\mathrm{dR}}(\mathcal{F}_i/\mathcal{F}_{i-1}) \otimes_{L \otimes_{\mathbb{Q}_p} E} D_{\mathrm{pdR}}(\varepsilon_{A,i})$  which is compatible with  $\nu_{W_A}$ . Hence  $\nu_{W_A}$  is given on  $\mathcal{D}_{A,i}/\mathcal{D}_{A,i-1}$  by  $\nu_{\mathcal{F}_i/\mathcal{F}_{i-1}} \otimes$ id + id  $\otimes \nu_{\varepsilon_{A,i}} = \mathrm{id} \otimes \nu_{\varepsilon_{A,i}}$ , which is the multiplication by the scalar  $\nu_{\varepsilon_{A,i}} \in L \otimes_{\mathbb{Q}_p} A$ .

We denote by  $\tilde{\mathfrak{g}}_{P,L}$  the completion of  $\tilde{\mathfrak{g}}_{P,L}$  at the point  $(\alpha^{-1}(\mathcal{D}_{\bullet}), N_W) \in \tilde{\mathfrak{g}}_{P,L}(E)$ , that we also see in the obvious way as a functor from  $\operatorname{Art}(E)$  to sets. If X is a groupoid over  $\operatorname{Art}(E)$ , we denote by |X| the functor on  $\operatorname{Art}(E)$  such that |X|(A) is the set of isomorphism classes of the category X(A) (see [62, Appendix] and [19, § 3.1] for more details). The following proposition easily follows from [19, Lemma 3.1.4].

**Proposition 6.1.2.** The groupoid  $X_{W,\mathcal{F}_{\bullet}}^{\Box}$  over  $\operatorname{Art}(E)$  is pro-representable. The functor

$$(A, W_A, \mathcal{F}_{A, \bullet}, \iota_A, \alpha_A) \longmapsto (\alpha_A^{-1}(\mathcal{D}_{A, \bullet}), N_{W_A})$$

induces an isomorphism of functors between  $|X_{W,\mathcal{F}_{\bullet}}^{\Box}|$  and  $\hat{\mathfrak{g}}_{P,L}$ . In particular  $|X_{W,\mathcal{F}_{\bullet}}^{\Box}|$  is prorepresented by a formally smooth noetherian complete local ring of residue field E and dimension  $[L:\mathbb{Q}_p](\dim \mathfrak{n}_P + \dim \mathfrak{r}_P)$ .

**Remark 6.1.3.** As discussed in [19, § 3.1], the morphism  $X_W^{\Box} \to |X_W^{\Box}|$  is actually an equivalence. We then easily deduce  $X_{W,\mathcal{F}_{\bullet}}^{\Box} \xrightarrow{\sim} |X_{W,\mathcal{F}_{\bullet}}^{\Box}|$ .

Since  $\nu_{\mathcal{F}_i/\mathcal{F}_{i-1}} = 0$ , the morphism  $\widetilde{\mathfrak{g}}_{P,L} \xrightarrow{\kappa_P} \mathfrak{z}_{L_P,L}$  (cf. § 5.1) induces a morphism  $\widehat{\mathfrak{g}}_{P,L} \xrightarrow{\kappa_P} \widehat{\mathfrak{z}}_{L_P,L}$ where  $\widehat{\mathfrak{z}}_{L_P}$  denotes the completion of  $\mathfrak{z}_{L_P}$  at 0. Consider the following composition of groupoids over  $\operatorname{Art}(E)$ 

$$\kappa_{W,\mathcal{F}_{\bullet}}: X_{W,\mathcal{F}_{\bullet}}^{\Box} \longrightarrow |X_{W,\mathcal{F}_{\bullet}}^{\Box}| \xrightarrow{\sim} \widehat{\tilde{\mathfrak{g}}}_{P,L} \xrightarrow{\kappa_{P}} \widehat{\mathfrak{z}}_{L_{P},L}.$$
(6.1)

One checks that (6.1) actually factors through

$$\kappa_{W,\mathcal{F}_{\bullet}}: X_{W,\mathcal{F}_{\bullet}} \longrightarrow \widehat{\mathfrak{z}}_{L_{P},L}.$$

The morphism (6.1) has the following functorial interpretation. Let  $x_A := (W_A, \mathcal{F}_{A,\bullet}, \iota_A) \in X_{W,\mathcal{F}_{\bullet}}$ , then the endomorphism  $\nu_{W_A}$  on  $D_{pdR}(W_A)$  induces an endomorphism  $\nu_{A,i}$  on every  $D_{pdR}(\mathcal{F}_{A,i})/D_{pdR}(\mathcal{F}_{A,i-1}) \cong D_{pdR}(\mathcal{F}_{A,i-1})$ . As in the proof of Lemma 6.1.1,  $\nu_{A,i}$  is a scalar in  $L \otimes_{\mathbb{Q}_p} A$  which is 0 modulo  $\mathfrak{m}_A$ . It follows that

$$\kappa_{W,\mathcal{F}_{\bullet}}(x_A) = (\nu_{A,1}, \dots, \nu_{A,r}) \in \widehat{\mathfrak{z}}_{L_P,L}(A).$$
(6.2)

# 6.2 $(\varphi, \Gamma)$ -modules of type $\Omega$ over $\mathcal{R}_{E,L}[\frac{1}{t}]$

We study certain groupoids of deformations of a  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{E,L}[\frac{1}{t}]$  equipped with an  $\Omega$ -filtration.

Let  $\Omega = \prod_{i=1}^{r} \Omega_i$  be a cuspidal component of  $L_P(L)$ . Let  $A \in \operatorname{Art}(E)$  and  $\mathcal{M}$  be a  $(\varphi, \Gamma)$ module of rank *n* over  $\mathcal{R}_{A,L}[1/t]$ . For  $i = 1, \ldots, r$ , let  $x_i$  be a closed point of  $\operatorname{Spec} \mathcal{Z}_{\Omega_i}$  and let  $\Delta_{x_i}$  be the associated *p*-adic differential equation. We call  $\mathcal{M}$  of type  $\Omega$  if there exists a filtration  $\mathcal{M}_{\bullet} = (\mathcal{M}_i)_{0 \leq i \leq r}$  by  $(\varphi, \Gamma)$ -submodules of  $\mathcal{M}$  over  $\mathcal{R}_{A,L}[1/t]$  such that  $\mathcal{M}_0 = 0$  and  $\mathcal{M}_i/\mathcal{M}_{i-1} \cong \Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta_i)[\frac{1}{t}]$  for some continuous character  $\delta_i : L^{\times} \to A^{\times}$ . Such a filtration  $\mathcal{M}_{\bullet}$  is called an  $\Omega$ -filtration of  $\mathcal{M}$ , and  $(\underline{x}, \delta) = ((x_i), \boxtimes_{i=1}^r \delta_i) \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{L_P}(L)}$  is called a parameter of  $\mathcal{M}_{\bullet}$  (compare with Definition 4.1.6). **Lemma 6.2.1.** Let  $\mathcal{M}$  be a  $(\varphi, \Gamma)$ -module of type  $\Omega$  over  $\mathcal{R}_{E,L}[1/t]$  and  $\mathcal{M}_{\bullet}$  an  $\Omega$ -filtration of  $\mathcal{M}$ of parameter  $(\underline{x}, \delta)$ . Then all parameters of  $\mathcal{M}_{\bullet}$  are of the form  $(\underline{x}', \delta')$  such that, for  $i = 1, \ldots, r$ ,  $\Delta_{x'_i} \cong \Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\psi_i)$  and  $\delta'_i = \delta_i \psi_i^{-1} \eta_i z^k$  for some unramified character  $\psi_i$  of  $L^{\times}$ ,  $\eta_i \in \mu_{\Omega_i}$ and  $\mathbf{k} \in \mathbb{Z}^{|\Sigma_L|}$ .

*Proof.* We have  $\Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i)[1/t] \cong \Delta_{x'_i} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta'_i)[1/t]$  if and only if for sufficiently large N:

$$\operatorname{Hom}_{(\varphi,\Gamma)}\left(\Delta_{x_{i}}, t^{-N}\Delta_{x_{i}'} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_{i}'\delta_{i}^{-1})\right) \neq 0$$

The lemma then follows by the same argument as in the proof of Lemma 4.1.9.

We now fix a  $(\varphi, \Gamma)$ -module  $\mathcal{M}$  over  $\mathcal{R}_{E,L}[\frac{1}{t}]$  of type  $\Omega$  and an  $\Omega$ -filtration  $\mathcal{M}_{\bullet}$ . We define the groupoid  $X_{\mathcal{M},\mathcal{M}_{\bullet}}$  over  $\operatorname{Art}(E)$  as follows:

- (1) The objects of  $X_{\mathcal{M},\mathcal{M}_{\bullet}}$  are the quadruples  $(A, \mathcal{M}_A, \mathcal{M}_{A,\bullet}, j_A)$  where  $A \in \operatorname{Art}(E)$ ,  $\mathcal{M}_A$ is a  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{A,L}[\frac{1}{t}]$  of type  $\Omega$ ,  $\mathcal{M}_{A,\bullet}$  is an  $\Omega$ -filtration of  $\mathcal{M}_A$ , and  $j_A$  is an isomorphism  $\mathcal{M}_A \otimes_A E \xrightarrow{\sim} \mathcal{M}$  which induces isomorphisms  $\mathcal{M}_{A,i} \otimes_A E \xrightarrow{\sim} \mathcal{M}_i$ .
- (2) A morphism  $(A, \mathcal{M}_A, \mathcal{M}_{A,\bullet}, j_A) \to (A', \mathcal{M}_{A'}, \mathcal{M}_{A',\bullet}, j_{A'})$  is a morphism  $A \to A'$  in  $\operatorname{Art}(E)$ and an isomorphism  $\mathcal{M}_A \otimes_A A' \xrightarrow{\sim} \mathcal{M}_{A'}$  which is compatible with the morphisms  $j_A, j_{A'}$ and induces isomorphisms  $\mathcal{M}_{A,i} \otimes_A A' \xrightarrow{\sim} \mathcal{M}_{A',i}$  for  $i = 1, \ldots, r$ .

**Lemma 6.2.2.** Let  $(\underline{x}, \delta)$  be a parameter of  $\mathcal{M}_{\bullet}$  and  $(A, \mathcal{M}_A, \mathcal{M}_{A, \bullet}, j_A) \in X_{\mathcal{M}, \mathcal{M}_{\bullet}}$ . There exists a unique character  $\delta_A = \boxtimes_{i=1}^r \delta_{A,i} : Z_{L_P}(L) \to A^{\times}$  such that  $\delta_{A,i} \equiv \delta_i \pmod{\mathfrak{m}_A}$  and  $(\underline{x}, \delta_A)$  is a parameter of  $\mathcal{M}_{A, \bullet}$ .

Proof. Let  $\delta_A = \boxtimes_{i=1}^r \delta_{A,i} : Z_{L_P}(L) \to A^{\times}$  be a continuous character such that  $\mathcal{M}_{A,i}/\mathcal{M}_{A,i-1} \cong \Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta_{A,i})[\frac{1}{t}]$ . Denote by  $\overline{\delta}_{A,i} : \delta_{A,i} \to A^{\times} \to E^{\times}$  the reduction of  $\delta_{A,i}$  modulo  $\mathfrak{m}_A$ . We have  $\Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\overline{\delta}_{A,i})[\frac{1}{t}] \cong \Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i)[\frac{1}{t}]$ . It follows that

$$\operatorname{Hom}_{(\varphi,\Gamma)}\left(\Delta_{x_{i}}, t^{-m}\Delta_{x_{i}}\otimes_{\mathcal{R}_{E,L}}\mathcal{R}_{E,L}(\delta_{i}^{-1}\overline{\delta}_{A,i})\right) \cong H^{0}_{(\varphi,\Gamma)}\left(t^{-m}\Delta_{x_{i}}\otimes_{\mathcal{R}_{E,L}}\Delta_{x_{i}}^{\vee}\otimes_{\mathcal{R}_{E,L}}\mathcal{R}_{E,L}(\delta_{i}^{-1}\overline{\delta}_{A,i})\right) \cong E^{-1}$$

for  $m \gg 0$ . By the same argument as in the proof of Lemma 4.1.9, we see that there exists an algebraic character  $\chi_i$  of  $L^{\times}$  and  $\psi_i \in \eta_{\Omega_i}$  such that  $\overline{\delta}_{A,i} = \delta_i \psi_i \chi_i$ . Since  $\Delta_{x_i}[\frac{1}{t}] \cong \Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\psi_i^{-1}\chi_i^{-1})[\frac{1}{t}]$ , replacing  $\delta_{A,i}$  by  $\delta_{A,i}\psi_i^{-1}\chi_i^{-1}$ , the existence in the lemma follows.

Assume  $\delta_A$ ,  $\delta'_A$  are two characters satisfying the properties in the lemma. We have an injection

$$A \hookrightarrow H^0_{(\varphi,\Gamma)}\Big(\Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \Delta_{x_i}^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta_{A,i}^{-1}\delta_{A,i}')\Big[\frac{1}{t}\Big]\Big),$$

where by definition  $H^0_{(\varphi,\Gamma)}(\mathcal{N}) := \operatorname{Hom}_{(\varphi,\Gamma)}(\mathcal{R}_A[1/t], \mathcal{N})$  for a  $(\varphi, \Gamma)$ -module  $\mathcal{N}$  over  $\mathcal{R}_{A,L}[1/t]$ (see [19, § 3.3] for the cohomology of  $(\varphi, \Gamma)$ -modules over  $\mathcal{R}_{E,L}[1/t]$ ). We write  $\Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \Delta_{x_i}^{\vee} \cong$  $\mathcal{R}_{E,L} \oplus (\Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \Delta_{x_i}^{\vee})^0$ , and we have  $H^0_{(\varphi,\Gamma)}((\Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \Delta_{x_i}^{\vee})^0[\frac{1}{t}]) = 0$  (using [19, (3.11)]). By an easy dévissage on A, we deduce that

$$H^{0}_{(\varphi,\Gamma)}\Big(\Delta_{x_{i}}\otimes_{\mathcal{R}_{E,L}}\Delta_{x_{i}}^{\vee}\otimes_{\mathcal{R}_{E,L}}\mathcal{R}_{A,L}(\delta_{A,i}^{-1}\delta_{A,i}')\Big[\frac{1}{t}\Big]\Big)\cong H^{0}_{(\varphi,\Gamma)}\Big(\mathcal{R}_{A,L}(\delta_{A,i}^{-1}\delta_{A,i}')\Big[\frac{1}{t}\Big]\Big).$$

By [19, Lemma 3.3.4] and  $\delta_{A,i} \equiv \delta'_{A,i} \pmod{\mathfrak{m}_A}$ , we must have  $\delta_{A,i} = \delta'_{A,i}$ .

Let  $\delta$  be a continuous character of  $Z_{L_P}(L)$ , that we also view as a point of  $\widehat{Z_{L_P}(L)}$ . We denote by  $\widehat{Z_{L_P}(L)}_{\delta}$  the completion of  $\widehat{Z_{L_P}(L)}$  at  $\delta$ . It is easy to see that the functor

$$A \in \operatorname{Art}(E) \longmapsto \{\delta_A = \boxtimes_{i=1}^r \delta_{A,i} : Z_{L_P}(L) \to A^{\times}, \ \delta_{A,i} \equiv \delta_i \pmod{\mathfrak{m}_A}\}$$

is pro-represented by  $\widehat{Z_{L_P}(L)}_{\delta}$ . By Lemma 6.2.2, we have a morphism of groupoids over  $\operatorname{Art}(E)$ :

$$\omega_{\delta}: X_{\mathcal{M}, \mathcal{M}_{\bullet}} \longrightarrow \widehat{Z_{L_{P}}(L)}_{\delta}, \ (A, \mathcal{M}_{A}, \mathcal{M}_{A, \bullet}, j_{A}) \longmapsto \delta_{A}.$$

Recall we have a functor  $W_{dR}$  from the category of  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{E,L}[\frac{1}{t}]$  to the category of  $B_{dR} \otimes_{\mathbb{Q}_p} E$ -representations of  $\operatorname{Gal}_L$  (cf. [19, Lemma 3.3.5 (ii)]). Moreover, by *loc. cit.*, for  $A \in \operatorname{Art}(E)$ ,  $W_{dR}$  sends a  $(\varphi, \Gamma)$ -module of rank n over  $\mathcal{R}_{A,L}[\frac{1}{t}]$  to a  $B_{dR} \otimes_{\mathbb{Q}_p} A$ -representation of  $\operatorname{Gal}_L$  of rank n. Let  $W := W_{dR}(\mathcal{M})$  and  $\mathcal{F}_i := W_{dR}(\mathcal{M}_i)$ . Assume that for one parameter (equivalently all parameters)  $(\underline{x}, \delta)$  of  $\mathcal{M}_{\bullet}$ , we have that  $\delta_i$  is locally algebraic for all i. Then it is easy to see that  $W_{dR}(\mathcal{M}_i/\mathcal{M}_{i-1}) \cong W_{dR}(\mathcal{M}_i)/W_{dR}(\mathcal{M}_{i-1})$  is de Rham (hence  $\cong (B_{dR} \otimes_{\mathbb{Q}_p} E)^{\oplus n_i})$ . For  $(A, \mathcal{M}_A, \mathcal{M}_{A, \bullet}, j_A) \in X_{\mathcal{M}, \mathcal{M}_{\bullet}}$ , let  $W_A := W_{dR}(\mathcal{M}_A)$  and  $\mathcal{F}_{\bullet,i} := W_{dR}(\mathcal{M}_{A,i})$ . If  $(\underline{x}, \delta_A)$  is a parameter of  $\mathcal{M}_{A, \bullet}$ , then we have

$$\mathcal{F}_{A,i}/\mathcal{F}_{A,i-1} \cong W_{\mathrm{dR}}(\mathcal{M}_{A,i}/\mathcal{M}_{A,i-1}) \cong W_{\mathrm{dR}}\left(\Delta_{x_i}\left[\frac{1}{t}\right]\right) \otimes_{B_{\mathrm{dR}}\otimes_{\mathbb{Q}_p} E} W_{\mathrm{dR}}\left(\mathcal{R}_{A,L}(\delta_{A,i})\left[\frac{1}{t}\right]\right), \quad (6.3)$$

where, for the last isomorphism, we use [5, Prop. 2.2.6 (2)] and the fact that  $W_{dR}(D[\frac{1}{t}]) = W_{dR}^+(D)[\frac{1}{t}]$  for a  $(\varphi, \Gamma)$ -module D over  $\mathcal{R}_{E,L}$ . Let  $\iota_A$  be the composition

$$\iota_A: W_{\mathrm{dR}}(\mathcal{M}_A) \otimes_A E \xrightarrow{\sim} W_{\mathrm{dR}}(\mathcal{M}_A \otimes_A E) \xrightarrow{\sim} W_{\mathrm{dR}}(\mathcal{M})$$

where the last isomorphism is induced by  $j_A$ . By (6.3), we see that  $(W_A, \mathcal{F}_{A,\bullet}, \iota_A) \in X_{W,\mathcal{F}_{\bullet}}$ , so  $W_{\mathrm{dR}}$  defines a morphism  $X_{\mathcal{M},\mathcal{M}^{\bullet}} \to X_{W,\mathcal{F}_{\bullet}}$ . Let  $X_{\mathcal{M}}$  be the groupoid over  $\operatorname{Art}(E)$  defined as  $X_{\mathcal{M},\mathcal{M}_{\bullet}}$  but forgetting everywhere the  $\Omega$ -filtrations. It is easy to see that  $W_{\mathrm{dR}}$  defines a morphism of groupoids  $X_{\mathcal{M}} \to X_W$ , and that the following diagram of groupoids commutes

$$\begin{array}{cccc} X_{\mathcal{M},\mathcal{M}_{\bullet}} & \xrightarrow{W_{\mathrm{dR}}} & X_{W,\mathcal{F}_{\bullet}} \\ & & & \downarrow \\ & & & \downarrow \\ & & & X_{\mathcal{M}} & \xrightarrow{W_{\mathrm{dR}}} & X_{W}. \end{array}$$

We fix an isomorphism  $\alpha : (L \otimes_{\mathbb{Q}_p} E)^n \xrightarrow{\sim} D_{pdR}(W)$ , so we have the groupoids  $X_W^{\square}, X_{W,\mathcal{F}_{\bullet}}^{\square}$  over Art(E) (cf. § 6.1). We put

$$X_{\mathcal{M}}^{\square} := X_{\mathcal{M}} \times_{X_{W}} X_{W}^{\square}, X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\square} := X_{\mathcal{M},\mathcal{M}_{\bullet}} \times_{X_{W,\mathcal{F}_{\bullet}}} X_{W,\mathcal{F}_{\bullet}}^{\square} \cong X_{\mathcal{M},\mathcal{M}_{\bullet}} \times_{X_{W}} X_{W}^{\square}$$

$$(6.4)$$

and note that  $X_{\mathcal{M}}^{\Box} \to X_{\mathcal{M}}, X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box} \to X_{\mathcal{M},\mathcal{M}_{\bullet}}$  are formally smooth of relative dimension  $[L : \mathbb{Q}_p]n^2$  by base change. We fix a parameter  $(\underline{x}, \delta)$  of  $\mathcal{M}_{\bullet}$ . For  $A \in \operatorname{Art}(E)$ , the natural map

$$\delta_A = \boxtimes_{i=1}^r \delta_{A,i} \longmapsto (\operatorname{wt}(\delta_{A,i}) - \operatorname{wt}(\delta_i)) \in (L \otimes_{\mathbb{Q}_p} A)^r \cong \widehat{\mathfrak{z}}_{L_P,L}(A)$$

induces a morphism of formal schemes wt  $- \operatorname{wt}(\delta) : \widehat{Z_{L_P}(L)}_{\delta} \to \widehat{\mathfrak{z}}_{L_P,L}.$ 

**Proposition 6.2.3.** The following diagram of groupoids over Art(E) is commutative:

$$\begin{array}{cccc} X_{\mathcal{M},\mathcal{M}_{\bullet}} & \xrightarrow{W_{\mathrm{dR}}} & X_{W,\mathcal{F}_{\bullet}} \\ & & & \\ \omega_{\delta} & & & \\ \widehat{Z_{L_P}(L)}_{\delta} & \xrightarrow{\mathrm{wt} - \mathrm{wt}(\delta)} & \widehat{\mathfrak{z}}_{L_P,L}. \end{array}$$

*Proof.* The proposition follows from (6.2) and [19, Lemma 3.3.6 (ii)] (which trivially generalizes to the case  $\mathcal{M} = \mathcal{R}_{A,L}(\delta)[\frac{1}{t}]^{\oplus m}$ ).

We call a parameter  $(\underline{x}, \delta)$  of  $\mathcal{M}_{\bullet}$  generic if the following condition is satisfied:

for 
$$i \neq j$$
, if  $\Delta_{x_i} \cong \Delta_{x_j} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\psi)$  for some smooth character  $\psi$  of  $L^{\times}$ , then  
 $\delta_i \delta_j^{-1} \psi \neq z^{\mathbf{k}}$  and  $\delta_i \delta_j^{-1} \psi \neq \operatorname{unr}(q_L^{-1}) z^{\mathbf{k}}$  for any  $\mathbf{k} \in \mathbb{Z}^{|\Sigma_L|}$ .
$$(6.5)$$

By Lemma 6.2.1, if  $\mathcal{M}_{\bullet}$  admits a generic parameter, then any parameter of  $\mathcal{M}_{\bullet}$  is generic, and in this case we call  $\mathcal{M}_{\bullet}$  (or even  $\mathcal{M}$  if  $\mathcal{M}_{\bullet}$  is understated) generic.

**Remark 6.2.4.** (1) Let D be a  $(\varphi, \Gamma)$ -module of rank n over  $\mathcal{R}_{E,L}$  such that  $D[\frac{1}{t}] \cong \mathcal{M}$ . The  $\Omega$ -filtration  $\mathcal{M}_{\bullet}$  on  $\mathcal{M}$  induces then an  $\Omega$ -filtration  $\mathscr{F}$  on D. It is straightforward to check that if  $\mathcal{M}_{\bullet}$  is generic then  $\mathscr{F}$  is generic in the sense of (4.13).

(2) Let  $\rho$  as in § 2.3 and use the notation of loc. cit. The  $\Omega$ -filtration  $\mathscr{F}$  on  $\mathcal{M}(\rho) \cong D_{\mathrm{rig}}(\rho)[\frac{1}{t}]$  is generic if and only if  $\rho$  is generic.

For 
$$1 \leq i, j \leq r, i \neq j$$
, denote by  $\mathcal{N}_{i,j}^0 := \Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \Delta_{x_j}^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i \delta_j^{-1})$  and  $\mathcal{N}_{i,j} := \mathcal{N}_{i,j}^0[\frac{1}{t}]$ .

**Lemma 6.2.5.** Assume  $(\underline{x}, \delta)$  is a generic parameter of  $\mathcal{M}_{\bullet}$  and let  $1 \leq i, j \leq r, i \neq j$ .

(1) We have  $H^0_{(\varphi,\Gamma)}(\mathcal{N}_{i,j}) = H^2_{(\varphi,\Gamma)}(\mathcal{N}_{i,j}) = 0$  and

$$\dim_E H^1_{(\varphi,\Gamma)}(\mathcal{N}_{i,j}) = [L:\mathbb{Q}_p]n_in_j.$$

(2) Suppose that  $\delta$  is locally algebraic, then the natural morphism

$$H^{1}_{(\varphi,\Gamma)}(\mathcal{N}_{i,j}) \longrightarrow H^{1}(\operatorname{Gal}_{L}, W_{\mathrm{dR}}(\mathcal{N}_{i,j}))$$

$$(6.6)$$

is an isomorphism.

(3) Suppose that  $\delta$  is locally algebraic and let  $A \in \operatorname{Art}(E)$ ,  $\delta_{A,i}, \delta_{A,j} : L^{\times} \to A^{\times}$  be continuous characters such that  $\delta_{A,i} \equiv \delta_i$ ,  $\delta_{A,j} \equiv \delta_j \pmod{\mathfrak{m}_A}$ , and  $\mathcal{N}_{i,j,A} := \Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \Delta_{x_j}^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_{A,i}\delta_{A,j}^{-1})[\frac{1}{t}]$ . Then the natural morphism

$$H^1_{(\varphi,\Gamma)}(\mathcal{N}_{i,j,A}) \longrightarrow H^1(\operatorname{Gal}_L, W_{\mathrm{dR}}(\mathcal{N}_{i,j,A}))$$

is surjective.

*Proof.* (1) We claim that for  $s = \{0, 1, 2\}$ ,

$$H^{s}_{(\varphi,\Gamma)}(\mathcal{N}_{i,j}) \cong H^{s}_{(\varphi,\Gamma)}(t^{-k}\mathcal{N}^{0}_{i,j}) \text{ for } k \text{ sufficiently large.}$$
(6.7)

Indeed, identifying the cohomology of  $(\varphi, \Gamma)$ -modules and the Galois cohomology of *B*-pairs (see for example [73, § 3]), we deduce from the morphism  $t^{-k}\mathcal{N}^0_{i,j} \hookrightarrow t^{-k-1}\mathcal{N}^0_{i,j}$  a long exact sequence

$$0 \to H^{0}_{(\varphi,\Gamma)}(t^{-k}\mathcal{N}^{0}_{i,j}) \to H^{0}_{(\varphi,\Gamma)}(t^{-k-1}\mathcal{N}^{0}_{i,j}) \to H^{0}_{(\varphi,\Gamma)}(t^{-k-1}\mathcal{N}^{0}_{i,j}/t^{-k}\mathcal{N}^{0}_{i,j}) \to H^{1}_{(\varphi,\Gamma)}(t^{-k}\mathcal{N}^{0}_{i,j}) \to H^{1}_{(\varphi,\Gamma)}(t^{-k-1}\mathcal{N}^{0}_{i,j}) \to H^{1}_{(\varphi,\Gamma)}(t^{-k-1}\mathcal{N}^{0}_{i,j}/t^{-k}\mathcal{N}^{0}_{i,j}) \to H^{2}_{(\varphi,\Gamma)}(t^{-k}\mathcal{N}^{0}_{i,j}) \to H^{2}_{(\varphi,\Gamma)}(t^{-k-1}\mathcal{N}^{0}_{i,j}) \to H^{2}_{(\varphi,\Gamma)}(t^{-k-1}\mathcal{N}^{0}_{i,j}/t^{-k}\mathcal{N}^{0}_{i,j}).$$
(6.8)

By [66, Thm. 4.7], we have  $H^2_{(\varphi,\Gamma)}(t^{-k-1}\mathcal{N}^0_{i,j}/t^{-k}\mathcal{N}^0_{i,j}) = 0$  and

$$\dim_E H^0_{(\varphi,\Gamma)}(t^{-k-1}\mathcal{N}^0_{i,j}/t^{-k}\mathcal{N}^0_{i,j}) = \dim_E H^1_{(\varphi,\Gamma)}(t^{-k-1}\mathcal{N}^0_{i,j}/t^{-k}\mathcal{N}^0_{i,j}).$$

By [15, Lemma 5.1.1], we have

$$H^{0}_{(\varphi,\Gamma)}(t^{-k-1}\mathcal{N}^{0}_{i,j}/t^{-k}\mathcal{N}^{0}_{i,j}) \cong H^{0}(\operatorname{Gal}_{L}, t^{-k-1}W^{+}_{\mathrm{dR}}(\mathcal{N}^{0}_{i,j})/t^{-k}W^{+}_{\mathrm{dR}}(\mathcal{N}^{0}_{i,j})).$$

As  $\Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \Delta_{x_j}^{\vee}$  is de Rham of constant Hodge-Tate weight 0, it follows that  $W_{\mathrm{dR}}^+(\mathcal{N}_{i,j}^0) \cong W_{\mathrm{dR}}^+(\mathcal{R}_{E,L}(\delta_i\delta_i^{-1}))^{\oplus n_i n_j}$ , hence

$$H^{0}\big(\operatorname{Gal}_{L}, t^{-k-1}W^{+}_{\mathrm{dR}}(\mathcal{N}^{0}_{i,j})/t^{-k}W^{+}_{\mathrm{dR}}(\mathcal{N}^{0}_{i,j})\big)$$
$$\cong H^{0}\big(\operatorname{Gal}_{L}, t^{-k-1}W^{+}_{\mathrm{dR}}\big(\mathcal{R}_{E,L}(\delta_{i}\delta_{j}^{-1})\big)/t^{-k}W^{+}_{\mathrm{dR}}\big(\mathcal{R}_{E,L}(\delta_{i}\delta_{j}^{-1})\big)\big)^{\oplus n_{i}n_{j}}$$

By [73, Lemma 2.16], the latter is zero when k is sufficiently large. We conclude that  $H^s_{(\varphi,\Gamma)}(t^{-k-1}\mathcal{N}^0_{i,j}/t^{-k}\mathcal{N}^0_{i,j}) = 0$  for  $s = \{0, 1, 2\}$  and k sufficiently large. By (6.8) and [19, (3.11)], (6.7) follows. Then (1) follows easily from the proof of Lemma 4.1.12 (see also Remark 6.2.4) and [66, Thm. 1.2 (1)].

(2) Since both sides of (6.6) have dimension  $[L : \mathbb{Q}_p]n_in_j$  over E, it suffices to show that the map is injective. By (6.7), it is enough to show that  $H^1_g(t^{-k}\mathcal{N}^0_{i,j}) := \operatorname{Ker}[H^1(t^{-k}\mathcal{N}^0_{i,j}) \to H^1(\operatorname{Gal}_L, W_{\mathrm{dR}}(\mathcal{N}_{i,j}))]$  is zero when k is sufficiently large. Put

$$H^{1}_{e}(t^{k}(\mathcal{N}^{0}_{i,j})^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\chi_{cyc}))$$
  
:= Ker  $\left[H^{1}_{(\varphi,\Gamma)}(t^{k}(\mathcal{N}^{0}_{i,j})^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\chi_{cyc})) \xrightarrow{f} H^{1}(\operatorname{Gal}_{L}, W_{e}(\mathcal{N}_{i,j} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\chi_{cyc})))\right]$ 

where  $W_e(D)$  denotes the  $B_e$ -module associated to a  $(\varphi, \Gamma)$ -module D, see [5, Prop. 2.2.6 (1)]. By [73, Prop. 2.11], we only need to show f is zero for k sufficiently large. Let s be an integer small enough so that  $t^s(\mathcal{N}_{i,j}^0)^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\chi_{cyc})$  has only negative Hodge-Tate weights. For  $k \geq s$ , the morphism f factors through (see the discussion below [19, (3.22)])

$$H^{1}_{(\varphi,\Gamma)}(t^{k}(\mathcal{N}^{0}_{i,j})^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\chi_{cyc})) \longrightarrow H^{1}_{(\varphi,\Gamma)}(t^{s}(\mathcal{N}^{0}_{i,j})^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\chi_{cyc})).$$
(6.9)

By (an easy generalization of) Lemma 4.1.12, and [66, Thm. 1.2 (1)],  $\dim_E H^1_{(\varphi,\Gamma)}(t^k(\mathcal{N}^0_{i,j})^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\chi_{cyc})) = [L:\mathbb{Q}_p]n_in_j$ . Using the cohomology of *B*-pairs (cf. [73, § 3]), we see that the map in (6.9) lies in the following long exact sequence, where  $W^+ := W^+_{dR}((\mathcal{N}^0_{i,j})^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\chi_{cyc}))$ :

$$0 \longrightarrow H^{0}_{(\varphi,\Gamma)} \left( t^{k}(\mathcal{N}^{0}_{i,j})^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\chi_{cyc}) \right) \longrightarrow H^{0}_{(\varphi,\Gamma)} \left( t^{s}(\mathcal{N}^{0}_{i,j})^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\chi_{cyc}) \right) \longrightarrow H^{0}(\operatorname{Gal}_{L}, t^{s}W^{+}/t^{k}W^{+}) \longrightarrow H^{1}_{(\varphi,\Gamma)} \left( t^{k}(\mathcal{N}^{0}_{i,j})^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\chi_{cyc}) \right) \longrightarrow H^{1}_{(\varphi,\Gamma)} \left( t^{s}(\mathcal{N}^{0}_{i,j})^{\vee} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\chi_{cyc}) \right).$$

By the same argument as in the proof of Lemma 4.1.12, it follows that the terms  $H^0_{(\varphi,\Gamma)}(\bullet)$  are both zero. When  $k \gg s$ , it is easy to see that  $\dim_E H^0(\operatorname{Gal}_L, t^s W^+/t^k W^+) = [L:\mathbb{Q}_p]n_i n_j$ , hence (6.9) and the map f are both zero. This concludes the proof of (2).

By dévissage, (3) follows from (2) together with the fact the functor  $W \mapsto H^1(\operatorname{Gal}_L, W)$ , on almost de Rham representations of  $\operatorname{Gal}_L$  over  $B_{\mathrm{dR}} \otimes_{\mathbb{Q}_p} E$ , is right exact (see for example the discussion below [19, (3.19)]).

**Theorem 6.2.6.** Assume  $(\underline{x}, \delta)$  is a generic parameter of  $\mathcal{M}_{\bullet}$  such that  $\delta$  is locally algebraic. The induced morphism of groupoids over  $\operatorname{Art}(E)$ 

$$X_{\mathcal{M},\mathcal{M}_{\bullet}} \longrightarrow \widehat{Z_{L_P}(L)}_{\delta} \times_{\widehat{\mathfrak{z}}_{L_P,L}} X_{W,\mathcal{F}_{\bullet}}$$

is formally smooth.

*Proof.* The theorem follows by the same argument as in the proof of [19, Thm. 3.4.4] with Lemma 3.4.2 and Lemma 3.4.3 of *loc. cit.* replaced by Lemma 6.2.5.  $\Box$ 

**Corollary 6.2.7.** With the assumption of Theorem 6.2.6, the morphisms  $X_{\mathcal{M},\mathcal{M}_{\bullet}} \to X_{W,\mathcal{F}_{\bullet}}$  and  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box} \to X_{W,\mathcal{F}_{\bullet}}^{\Box}$  are formally smooth.

*Proof.* By [19, Lemma 3.5.5], the morphism wt  $- \text{wt}(\delta) : Z_{L_P}(L)_{\delta} \to \hat{\mathfrak{z}}_{L_P,L}$  is formally smooth of relative dimension r. Together with Theorem 6.2.6, the first part of the corollary follows. The second part follows from the first part by base change.

The following proposition is analogous to [19, Prop. 3.4.6] (for a closed immersion of groupoids over  $\operatorname{Art}(E)$ , see the discussion before [19, Prop. 3.4.6]).

**Proposition 6.2.8.** Assume that  $\mathcal{M}_{\bullet}$  is generic, then the morphism  $X_{\mathcal{M},\mathcal{M}_{\bullet}} \to X_{\mathcal{M}}$  of groupoids over  $\operatorname{Art}(E)$  is relatively representable and is a closed immersion.

*Proof.* Since  $\mathcal{M}_{\bullet}$  is generic, using Lemma 6.2.5 (1) and an argument analogous to [3, Prop. 2.3.6], an  $\Omega$ -filtration  $\mathcal{M}_{A,\bullet}$  on a deformation  $\mathcal{M}_A$  of  $\mathcal{M}$  is unique if it exists. We deduce that  $|X_{\mathcal{M},\mathcal{M}_{\bullet}}|$ is a subfunctor of  $|X_{\mathcal{M}}|$  and that we have an equivalence of groupoids over  $\operatorname{Art}(E)$ :

$$X_{\mathcal{M},\mathcal{M}_{\bullet}} \cong X_{\mathcal{M}} \times_{|X_{\mathcal{M}}|} |X_{\mathcal{M},\mathcal{M}_{\bullet}}|.$$

Hence we only need to show that  $|X_{\mathcal{M},\mathcal{M}_{\bullet}}| \hookrightarrow |X_{\mathcal{M}}|$  is relatively representable. By [69, § 23], it is enough to check the following three conditions:

- (1) If  $A \to A'$  is a morphism in  $\operatorname{Art}(E)$  and  $(\mathcal{M}_A, \mathcal{M}_{A,\bullet}, j_A) \in |X_{\mathcal{M},\mathcal{M}_{\bullet}}|(A)$ , then  $(\mathcal{M}_A \otimes_A A', \mathcal{M}_{A,\bullet} \otimes_A A', j_A \otimes_A A') \in |X_{\mathcal{M},\mathcal{M}_{\bullet}}|(A')$ .
- (2) If  $A \to A'$  is an injective morphism in  $\operatorname{Art}(E)$ ,  $(\mathcal{M}_A, j_A) \in |X_{\mathcal{M}}|(A)$ , and if  $(\mathcal{M}_A \otimes_A A', j_A \otimes_A A') \in |X_{\mathcal{M},\mathcal{M}_{\bullet}}|(A') \hookrightarrow |X_{\mathcal{M}}|(A')$ , then  $(\mathcal{M}_A, j_A) \in |X_{\mathcal{M},\mathcal{M}_{\bullet}}|(A)$ .
- (3) If  $A, A' \in \operatorname{Art}(E)$ ,  $(\mathcal{M}_A, j_A) \in |X_{\mathcal{M}, \mathcal{M}_{\bullet}}|(A)$  and  $(\mathcal{M}_{A'}, j_{A'}) \in |X_{\mathcal{M}, \mathcal{M}_{\bullet}}|(A')$ , then  $(\mathcal{M}_A \times_{\mathcal{M}} \mathcal{M}_{A'}, j_{A \times_E A'} := j_A \circ \operatorname{pr}_1 = j_{A'} \circ \operatorname{pr}_2) \in |X_{\mathcal{M}, \mathcal{M}_{\bullet}}|(A \times_E A')$ .

(1) is clear. For (3), we just note that  $\mathcal{M}_{A \times_E A'} := \mathcal{M}_A \times_{\mathcal{M}} \mathcal{M}_{A'}$  admits the  $\Omega$ -filtration  $\mathcal{M}_{A \times_E A',i} := \mathcal{M}_{A,i} \times_{\mathcal{M}_i} \mathcal{M}_{A',i}$ . We prove (2). Let  $\mathcal{M}_{A,1} := \mathcal{M}_{A',1} \cap \mathcal{M}_A$ : this is a  $(\varphi, \Gamma)$ module over  $\mathcal{R}_{E,L}[\frac{1}{t}]$  (as it is a submodule of  $\mathcal{M}_A$ ) which is equipped with an action of A. By
an easy variation of the proof of Proposition 4.1.3, one can show that there exists a continuous
character  $\delta_{A,1} : L^{\times} \to A^{\times}$  such that  $\delta_{A,1} \equiv \delta_1 \pmod{\mathfrak{m}_A}$  and  $\mathcal{M}_{A,1} \cong \Delta_{x_1} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta_{A,1})[1/t]$ .
As  $\mathcal{M}_A/\mathcal{M}_{A,1} \hookrightarrow \mathcal{M}_{A'}/\mathcal{M}_{A',1}$  and  $\mathcal{M}_{A'}/\mathcal{M}_{A',1}$  is free over  $\mathcal{R}_{E,L}[1/t]$ , so is  $\mathcal{M}_A/\mathcal{M}_{A,1}$ . By [3,
Lemma 2.2.3 (i)],  $\mathcal{M}_A/\mathcal{M}_{A,1}$  is free over A. It follows by [3, Lemma 2.2.3 (ii)] that  $\mathcal{M}_A/\mathcal{M}_{A,1}$ ( $\hookrightarrow \mathcal{M}_{A'}/\mathcal{M}_{A',1}$ ) is free over  $\mathcal{R}_{A,L}[\frac{1}{t}]$ . Using  $\mathcal{M}_A/\mathcal{M}_{A,1} \hookrightarrow \mathcal{M}_{A'}/\mathcal{M}_{A',1}$ , by an induction argument, we can construct the desired filtration  $\mathcal{M}_{A,i}$  on  $\mathcal{M}_A$ , and hence  $(\mathcal{M}_A, j_A) \in |X_{\mathcal{M},\mathcal{M}_{\bullet}|(A) \hookrightarrow |X_{\mathcal{M}|(A)}$ . This concludes the proof.

By [19, Lemma 3.5.3 (i)] and the same argument as in the proof of *loc. cit.*, we have (where the third relative representability follows from the second one by base change):

**Lemma 6.2.9.** Assume that  $\mathcal{M}_{\bullet}$  is generic, the morphisms  $X_{\mathcal{M}} \to X_W, X_{\mathcal{M},\mathcal{M}_{\bullet}} \to X_{W,\mathcal{F}_{\bullet}}$  and  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box} \to X_{W,\mathcal{F}_{\bullet}}^{\Box}$  are relatively representable.

**Proposition 6.2.10.** Assume that  $\mathcal{M}_{\bullet}$  is a generic, then the groupoid  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box}$  over  $\operatorname{Art}(E)$  is prorepresentable. The functor  $|X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box}|$  is pro-represented by a formally smooth noetherian complete local ring of residue field E and dimension  $[L:\mathbb{Q}_p](n^2 + \dim_E \mathfrak{r}_P)$ .

*Proof.* The first statement follows from Lemma 6.2.9 and Proposition 6.1.2. By Proposition 6.1.2 and Corollary 6.2.7,  $|X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box}|$  is pro-represented by a formally smooth noetherian complete local ring of residue field E. We only need to calculate  $|X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box}|(E[\varepsilon]/\varepsilon^2)$ .

Assume that  $(\underline{x}, \delta)$  is a parameter of  $\mathcal{M}_{\bullet}$ . For each  $i = 1, \ldots, r$ , we fix an isomorphism  $\beta_i : \Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i)[\frac{1}{t}] \xrightarrow{\sim} \mathcal{M}_i/\mathcal{M}_{i-1}$ . As in the proof of [19, Prop. 3.5.7], we introduce a new groupoid  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\text{ver}}$  over  $\operatorname{Art}(E)$ :

- (1) The objects of  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\text{ver}}$  are the 5-tuples  $(A, \mathcal{M}_A, \mathcal{M}_{A,\bullet}, j_A, \underline{\beta}_A)$ , where  $(A, \mathcal{M}_A, \mathcal{M}_{A,\bullet}, j_A)$ is an object of  $X_{\mathcal{M},\mathcal{M}_{\bullet}}$  and  $\underline{\beta}_A = (\beta_{A,i})$  is a collection of isomorphisms  $\beta_{A,i} : \Delta_{x_i} \otimes_{\mathcal{R}_{E,L}}$  $\mathcal{R}_{A,L}(\delta_{A,i})[\frac{1}{t}] \xrightarrow{\sim} \mathcal{M}_{A,i}/\mathcal{M}_{A,i-1}$  (where  $\delta_{A,i}$  is as in Lemma 6.2.2) which are compatible with  $j_A$  and  $\underline{\beta} = (\beta_i)$ .
- (2) A morphism in  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\text{ver}}$  is a morphism in  $X_{\mathcal{M},\mathcal{M}_{\bullet}}$  which is compatible with the liftings of  $\beta$ .

We have  $|X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\text{ver}}| \cong X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\text{ver}}$ . For each  $i \in \{1, \ldots, r\}$ , we also use  $\mathcal{M}_{\bullet}$  to denote the induced filtration on  $\mathcal{M}_i$ , and we define  $X_{\mathcal{M}_i,\mathcal{M}_{\bullet}}^{\text{ver}} \cong |X_{\mathcal{M}_i,\mathcal{M}_{\bullet}}^{\text{ver}}|$  similarly to  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\text{ver}} \cong |X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\text{ver}}|$ . We first use an induction argument on i (inspired by the proof of [30, Thm. 3.3]) to show that the functor  $|X_{\mathcal{M}_i,\mathcal{M}_{\bullet}}^{\text{ver}}|$  is pro-represented by a formally smooth noetherian complete local ring of residue field E. It is easy to see that  $|X_{\mathcal{M}_1,\mathcal{M}_{\bullet}}^{\text{ver}}|$  is pro-represented by a formally smooth noetherian complete local ring  $R_{i-1}$ ]. Assume that  $|X_{\mathcal{M}_{i-1},\mathcal{M}_{\bullet}}^{\text{ver}}|$  is pro-represented by a formally smooth noetherian complete local ring  $R_{i-1}$  of residue field E and dimension

$$i - 1 + [L:\mathbb{Q}_p] \left( i - 1 + \sum_{1 \le j < j' \le i-1} n_j n'_j \right).$$

Let  $S_i$  denote the completion of  $R_{i-1} \otimes_E \widehat{\mathcal{O}}_{\widehat{L^{\times}},\delta_i}$  with respect to the maximal ideal generated by the maximal ideal of  $R_{i-1}$  and the one of  $\widehat{\mathcal{O}}_{\widehat{L^{\times}},\delta_i}$ . So  $S_i$  is a noetherian complete local ring which is formally smooth over E of dimension  $i + [L : \mathbb{Q}_p](i - 1 + \sum_{1 \le j < j' \le i-1} n_j n'_j)$ . For any morphism  $S_i \to A$  with  $A \in \operatorname{Art}(E)$ , let  $\mathcal{M}_{A,i-1}$  be the  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{A,L}[\frac{1}{t}]$  given by the pull-back along  $R_{i-1} \to S_i \to A$  of the universal  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{R_{i-1},L}[\frac{1}{t}]$ , and let  $\delta_{A,i}$  be the character  $L^{\times} \to \widehat{\mathcal{O}}_{\widehat{L^{\times}}, \delta_i}^{\times} \to S_i^{\times} \to A^{\times}$ . Let

$$N_{i} := \lim_{\substack{A \in \operatorname{Art}(E) \\ S_{i} \to A}} \operatorname{Ext}^{1}_{(\varphi, \Gamma)} \left( \Delta_{x_{i}} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta_{A,i}) \left[ \frac{1}{t} \right], \mathcal{M}_{A,i-1} \right).$$

Since  $\mathcal{M}_{\bullet}$  is generic, we deduce by Lemma 6.2.5 (1) (and a dévissage) that  $N_i$  is a free  $S_i$ -module of rank  $[L : \mathbb{Q}_p](n_i \sum_{j=1}^{i-1} n_j)$ . By definition,  $[\mathcal{M}_i] \in \operatorname{Ext}^1_{(\varphi, \Gamma)}(\Delta_{x_i} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{E,L}(\delta_i)[1/t], \mathcal{M}_{i-1})$ , which corresponds then to a maximal ideal  $\mathfrak{m}_i$  with residue field E of the polynomial  $S_i$ -algebra  $\operatorname{Symm}_{S_i} N_i^{\lor}$ . We let  $R_i$  be the completion of  $\operatorname{Symm}_{S_i} N_i^{\lor}$  at  $\mathfrak{m}_i$ . Thus

$$R_i \cong E[[x_1, \dots, x_{i+[L:\mathbb{Q}_p](i+\sum_{1 \le j \le j' \le i} n_j n'_j)}]]$$

and one can directly check that  $|X_{\mathcal{M}_i,\mathcal{M}_{\bullet}}^{\text{ver}}|$  is pro-represented by  $R_i$ . In particular  $|X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\text{ver}}|$  is prorepresented by a formally smooth noetherian complete local ring of dimension  $r + [L:\mathbb{Q}_p] \dim \mathfrak{r}_p$ .

Now we define  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box,\text{ver}}$  as  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box} \times_{X_{\mathcal{M},\mathcal{M}_{\bullet}}} X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\text{ver}}$ . Since  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\text{ver}} \cong |X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\text{ver}}|$  is pro-representable, it is easy to see that  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box,\text{ver}}$  is pro-representable (by adding formal variables corresponding to the framing  $\alpha$ ). The morphism  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box,\text{ver}} \to X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\text{ver}}$  is formally smooth of relative dimension  $n^{2}[L:\mathbb{Q}_{p}]$ . As  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box,\text{ver}}$  can be constructed from  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box}$  by adding frames (with respect to  $\underline{\beta}$ ) and  $\text{End}_{(\varphi,\Gamma)}(\Delta_{x_{i}} \otimes_{\mathcal{R}_{E,L}} \mathcal{R}_{A,L}(\delta_{A,i})[1/t]) \cong A$  (which follows from the proof of Lemma 6.2.2), the morphism  $X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box,\text{ver}} \to X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box}$  is formally smooth of relative dimension r. We then compute:

$$\dim_E |X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\square}|(E[\varepsilon]/\varepsilon^2) = n^2[L:\mathbb{Q}_p] + (r + [L:\mathbb{Q}_p]\mathfrak{r}_P) - r = [L:\mathbb{Q}_p](n^2 + \dim\mathfrak{r}_P)$$

which concludes the proof.

# **6.3** $(\varphi, \Gamma)$ -modules of type $\Omega$ over $\mathcal{R}_{E,L}$

We study certain groupoids of deformations of a  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{E,L}$  equipped with an  $\Omega$ -filtration. We keep the notation of § 6.2.

Let D be a  $(\varphi, \Gamma)$ -module of rank n over  $\mathcal{R}_{E,L}$  and  $\mathcal{M} := D[\frac{1}{t}]$ . Let  $W^+ := W^+_{dR}(D)$  be the associated  $B^+_{dR} \otimes_{\mathbb{Q}_p} E$ -representation of  $\operatorname{Gal}_L$  and  $W := W_{dR}(\mathcal{M}) \cong W^+[\frac{1}{t}]$ . Assume that W is almost de Rham. We define the groupoids  $X_D, X_{W^+}$  over  $\operatorname{Art}(E)$  of deformations of (respectively)  $D, W^+$  as in [19, § 3.5]. Recall that we have natural morphisms  $X_D \to X_{W^+}$  (induced by the functor  $W^+_{dR}(-)$ ) and  $X_D \to X_{\mathcal{M}}, X_{W^+} \to X_W$  (inverting t), and that the following diagram commutes:



By [19, Prop. 3.5.1], the induced morphism  $X_D \to X_{\mathcal{M}} \times_{X_W} X_{W^+}$  is an equivalence. Fix an isomorphism  $\alpha : (L \otimes_{\mathbb{Q}_p} E)^n \xrightarrow{\sim} D_{pdR}(W)$  so we have the groupoid  $X_W^{\Box}$  over  $\operatorname{Art}(E)$  (§ 6.1 or [19, § 3.1]). We put

$$X_{W^+}^{\square} := X_{W^+} \times_{X_W} X_W^{\square}, \ X_D^{\square} := X_D \times_{X_W} X_W^{\square}$$

We assume that D has distinct Sen weights  $(h_{1,\tau} > h_{1,\tau} > \cdots > h_{n,\tau})_{\tau \in \Sigma_L}$ . Then  $W^+$  is regular in the sense of [19, Def. 3.2.4]. Let  $(A, W_A^+, \iota_A, \alpha_A) \in X_{W^+}^{\Box}$  and  $W_A := W_A^+[1/t]$ , then the  $B_{\mathrm{dR}}^+ \otimes_{\mathbb{Q}_p} A$ -lattice  $W_A^+$  of  $W_A$  induces a complete flag  $\operatorname{Fil}_{W_A^+, \bullet} = \operatorname{Fil}_{W_A^+, \bullet}(D_{\mathrm{pdR}}(W_A)) :=$  $(\operatorname{Fil}_{W_A^+, i}(D_{\mathrm{pdR}}(W_A)))_{i=1,\ldots,n}$  of  $D_{\mathrm{pdR}}(W_A)$  by the formula

$$\operatorname{Fil}_{W_A^+,i}(D_{\operatorname{pdR}}(W_A)) := \bigoplus_{\tau \in \Sigma_L} \operatorname{Fil}_{W_A^+}^{-h_{n+1-i,\tau}}(D_{\operatorname{pdR},\tau}(W_A)) := \bigoplus_{\tau \in \Sigma_L} (t^{-h_{n+1-i,\tau}}W_A^+)_{\tau}^{\operatorname{Gal}_L},$$

where

$$D_{\mathrm{pdR},\tau}(W_A) := D_{\mathrm{pdR}}(W_A) \otimes_{L \otimes_{\mathbb{Q}_p} E} (L \otimes_{L,\tau} E)$$
$$(t^{-h_{n+1-i,\tau}} W_A^+)_{\tau} := t^{-h_{n+1-i,\tau}} (W_A^+ \otimes_{L \otimes_{\mathbb{Q}_p} E} (L \otimes_{L,\tau} E))$$

Since the flag  $\operatorname{Fil}_{W_A^+, \bullet}$  is stable under the endomorphism  $\nu_{W_A}$  of  $D_{pdR}(W_A)$  (see § 6.1), it follows that we have:

$$\left(\alpha_A^{-1}(\operatorname{Fil}_{W_A^+, \bullet}), N_{W_A} = \alpha_A^{-1} \circ \nu_{W_A} \circ \alpha_A\right) \in \tilde{\mathfrak{g}}_L(A).$$

Denote by  $\widehat{\mathfrak{g}}_L$  the completion of  $\mathfrak{g}_L$  at  $(\alpha^{-1}(\operatorname{Fil}_{W^+,\bullet}), N_W)$ . By [19, Thm. 3.2.5],  $X_{W^+}^{\square}$  is prorepresentable and we have an isomorphism of functors

$$|X_{W^+}^{\Box}| \xrightarrow{\sim} \widehat{\tilde{\mathfrak{g}}}_L, \ (W_A^+, \iota_A, \alpha_A) \longmapsto \left(\alpha_A^{-1}(\operatorname{Fil}_{W_A^+, \bullet}), N_{W_A}\right).$$

Consider the composition

$$\kappa_{W^+}: X_{W^+}^{\square} \longrightarrow |X_{W^+}^{\square}| \xrightarrow{\sim} \widehat{\mathfrak{g}}_L \xrightarrow{\kappa_B} \widehat{\mathfrak{t}}_L$$

where  $\hat{\mathfrak{t}}_L$  denotes the completion of  $\mathfrak{t}_L$  at 0 (and  $\kappa_B$  is defined in (5.6)). The morphism  $\kappa_{W^+}$  factors through a map still denoted by  $\kappa_{W^+}: X_{W^+} \longrightarrow \hat{\mathfrak{t}}_L$ .

We call D of type  $\Omega$  if D admits an  $\Omega$ -filtration  $D_{\bullet}$  (see Definition 4.1.6 (1)). The  $D_i$  for  $i = 1, \ldots, r$  are saturated  $(\varphi, \Gamma)$ -submodules of D and  $\mathcal{M}_{\bullet} := (\mathcal{M}_i)_{0 \leq i \leq r} := (D_i[\frac{1}{t}])_{0 \leq i \leq r}$  is an  $\Omega$  filtration of  $\mathcal{M}$  as in § 6.2. We assume that  $\mathcal{M}_{\bullet}$  is generic (which implies that  $D_{\bullet}$  is generic, see Remark 6.2.4 (1)). Assume D is of type  $\Omega$  and let  $D_{\bullet}$  be an  $\Omega$ -filtration of D. We put  $\mathcal{F}_{\bullet} := W_{\mathrm{dR}}(\mathcal{M}_{\bullet}) = W_{\mathrm{dR}}^+(D_{\bullet})[\frac{1}{t}]$  and define the following groupoids over  $\mathcal{R}_{E,L}$ 

$$\begin{aligned} X_{W^+,\mathcal{F}_{\bullet}} &:= X_{W^+} \times_{X_W} X_{W,\mathcal{F}_{\bullet}}, \qquad X_{W^+,\mathcal{F}_{\bullet}}^{\square} &:= X_{W^+,\mathcal{F}_{\bullet}} \times_{X_W} X_W^{\square} = X_{W^+} \times_{X_W} X_{W,\mathcal{F}_{\bullet}}^{\square}, \\ X_{D,\mathcal{M}_{\bullet}} &:= X_D \times_{X_\mathcal{M}} X_{\mathcal{M},\mathcal{M}_{\bullet}}, \qquad X_{D,\mathcal{M}_{\bullet}}^{\square} &:= X_{D,\mathcal{M}_{\bullet}} \times_{X_D} X_D^{\square} = X_{D,\mathcal{M}_{\bullet}} \times_{X_W} X_W^{\square}, \end{aligned}$$

where we have used  $X_D^{\square} = X_D \times_{X_W} X_W^{\square}$ .

**Proposition 6.3.1.** The morphisms of groupoids  $X_{D,\mathcal{M}_{\bullet}} \to X_{W^+,\mathcal{F}_{\bullet}}$  and  $X_{D,\mathcal{M}_{\bullet}}^{\Box} \to X_{W^+,\mathcal{F}}^{\Box}$  are formally smooth and relatively representable.

Proof. Since  $X_D \cong X_{\mathcal{M}} \times_{X_W} X_{W^+}$ , we have  $X_{D,\mathcal{M}_{\bullet}} \cong (X_{W^+} \times_{X_W} X_{\mathcal{M}}) \times_{X_{\mathcal{M}}} X_{\mathcal{M},\mathcal{M}_{\bullet}} \cong X_{W^+} \times_{X_W} X_{\mathcal{M},\mathcal{M}_{\bullet}}$ . The first part then follows by base change from Lemma 6.2.9 and Corollary 6.2.7. The second part follows from the first again by base change.

We define:

$$\mathcal{D}_{\bullet} := (\mathcal{D}_i)_{1 \le i \le r} := (D_{\mathrm{pdR}}(\mathcal{F}_i))_{1 \le i \le r} = (D_{\mathrm{pdR}}(W_{\mathrm{dR}}(\mathcal{M}_i)))_{1 \le i \le r}$$

Using  $X_{P,L} \cong \tilde{\mathfrak{g}}_{P,L} \times_{\mathfrak{g}_L} \tilde{\mathfrak{g}}_L$ , we have:

$$y := \left(\alpha^{-1}(\mathcal{D}_{\bullet}), \alpha^{-1}(\operatorname{Fil}_{W^+, \bullet}), N_W\right) \in X_{P,L}(E).$$
(6.10)

**Proposition 6.3.2.** (1) The groupoid  $X_{W^+,\mathcal{F}_{\bullet}}^{\Box}$  over  $\operatorname{Art}(E)$  is pro-representable. The functor  $|X_{W^+,\mathcal{F}_{\bullet}}^{\Box}|$  is pro-represented by the formal scheme  $\widehat{X}_{P,L,y}$ .

(2) The groupoid  $X_{D,\mathcal{M}_{\bullet}}^{\Box}$  over  $\operatorname{Art}(E)$  is pro-representable. The functor  $|X_{D,\mathcal{M}_{\bullet}}^{\Box}|$  is prorepresented by a formal scheme which is formally smooth of relative dimension  $[L:\mathbb{Q}_p]\dim \mathfrak{p}$ over  $\widehat{X}_{P,L,y}$ .

*Proof.* (1) follows by the same argument as in the proof of [19, Cor. 3.5.8] (with [19, Cor. 3.1.9] replaced by Proposition 6.1.2). Using  $X_{D,\mathcal{M}_{\bullet}} \cong X_{W^+} \times_{X_W} X_{\mathcal{M},\mathcal{M}_{\bullet}}$  and  $X_{W^+,\mathcal{F}_{\bullet}} = X_{W^+} \times_{X_W} X_{W,\mathcal{F}_{\bullet}}$ , we deduce  $X_{D,\mathcal{M}_{\bullet}} \cong X_{W^+,\mathcal{F}_{\bullet}} \times_{X_W,\mathcal{F}_{\bullet}} X_{\mathcal{M},\mathcal{M}_{\bullet}}$  and thus:

$$X_{D,\mathcal{M}_{\bullet}}^{\Box} \cong X_{W^{+},\mathcal{F}_{\bullet}}^{\Box} \times_{X_{W,\mathcal{F}_{\bullet}}} X_{\mathcal{M},\mathcal{M}_{\bullet}} \cong X_{W^{+},\mathcal{F}_{\bullet}}^{\Box} \times_{X_{W,\mathcal{F}_{\bullet}}^{\Box}} X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box}.$$
(6.11)

The first part of (2) follows from (1) and Proposition 6.3.1. From (6.11) and the fact that, for each groupoid Y in the fibre product on the right hand side of (6.11), we have  $Y \cong |Y|$  (see Remark 6.1.3 for  $X_{W,\mathcal{F}_{\bullet}}^{\Box}$ , the others being similar), we deduce  $|X_{D,\mathcal{M}_{\bullet}}^{\Box}| \cong |X_{W^+,\mathcal{F}_{\bullet}}^{\Box}| \times_{|X_{W,\mathcal{F}_{\bullet}}^{\Box}|} |X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box}|$ . By Corollary 6.2.7, Proposition 6.1.2 and Proposition 6.2.10, the functor  $|X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box}|$  is formally smooth over  $|X_{W,\mathcal{F}_{\bullet}}^{\Box}|$  of relative dimension

$$[L:\mathbb{Q}_p](n^2 + \dim \mathfrak{r}_P) - [L:\mathbb{Q}_p](\dim \mathfrak{n}_P + \dim \mathfrak{r}_P) = [L:\mathbb{Q}_p]\dim \mathfrak{p}.$$

The second part of (2) then follows by base change from the above fiber product.

For  $w = (w_{\tau})_{\tau \in \Sigma_L} \in \mathscr{W}_L \cong \mathscr{W}^{|\Sigma_L|}$ , let  $X_w := \prod_{\tau \in \Sigma_L} X_{w_{\tau}} \hookrightarrow X_{P,L}$ , where  $X_{w_{\tau}} \hookrightarrow X_P$  is defined as in § 5.2. This is an irreducible component of  $X_{P,L}$  which only depends on the coset  $\mathscr{W}_{L_P,L}w$ . We put (using Proposition 6.3.2 (1) and with  $\widehat{X}_{w,y}$  empty if  $y \notin X_w$ ):

$$X_{W^+,\mathcal{F}_{\bullet}}^{\Box,w} := X_{W^+,\mathcal{F}_{\bullet}}^{\Box} \times_{|X_{W^+,\mathcal{F}_{\bullet}}^{\Box}|} \widehat{X}_{w,y}.$$
(6.12)

**Corollary 6.3.3.** The groupoid  $X_{W^+,\mathcal{F}_{\bullet}}^{\Box,w}$  over  $\operatorname{Art}(E)$  is pro-representable. The functor  $|X_{W^+,\mathcal{F}_{\bullet}}^{\Box,w}|$  is pro-represented by the formal scheme  $\widehat{X}_{w,y}$ .

*Proof.* This follows from (6.12) and the equivalence of groupoids  $X_{W^+,\mathcal{F}_{\bullet}}^{\Box} \xrightarrow{\sim} |X_{W^+,\mathcal{F}_{\bullet}}^{\Box}|$ .

We define the groupoid  $X_{W^+,\mathcal{F}_{\bullet}}^w$  over  $\operatorname{Art}(E)$  as the subgroupoid of  $X_{W^+,\mathcal{F}_{\bullet}}$  which is the image of  $X_{W^+,\mathcal{F}_{\bullet}}^{\Box,w}$  by the forgetful morphism  $X_{W^+,\mathcal{F}_{\bullet}}^{\Box} \to X_{W^+,\mathcal{F}_{\bullet}}$ . Thus, for  $A \in \operatorname{Art}(E)$ , the objects of  $X_{W^+,\mathcal{F}_{\bullet}}^w(A)$  are the quadruples  $(W_A^+,\mathcal{F}_{A,\bullet},\iota_A,j_A)$  in  $X_{W^+,\mathcal{F}_{\bullet}}(A)$  such that

$$\left(\alpha_A^{-1}(\operatorname{Fil}_{W_A^+, \bullet}), \alpha_A^{-1}(\mathcal{D}_{A, \bullet}), \alpha_A^{-1} \circ \nu_{W_A} \circ \alpha_A\right) \in X_w(A)$$

for one (equivalently any) isomorphism  $\alpha_A : (L \otimes_{\mathbb{Q}_p} A) \xrightarrow{\sim} D_{pdR}(W_A)$ . As in [19, (3.26)] there is an equivalence of groupoids  $X_{W^+,\mathcal{F}_{\bullet}}^{\Box,w} \xrightarrow{\sim} X_{W^+,\mathcal{F}_{\bullet}}^w \times_{X_{W^+,\mathcal{F}_{\bullet}}} X_{W^+,\mathcal{F}_{\bullet}}^{\Box}$ . We define then

$$X_{D,\mathcal{M}_{\bullet}}^{\square,w} := X_{D,\mathcal{M}_{\bullet}}^{\square} \times_{X_{W^{+},\mathcal{F}_{\bullet}}^{\square}} X_{W^{+},\mathcal{F}_{\bullet}}^{\square,w} \text{ and } X_{D,\mathcal{M}_{\bullet}}^{w} := X_{D,\mathcal{M}_{\bullet}} \times_{X_{W^{+},\mathcal{F}_{\bullet}}} X_{W^{+},\mathcal{F}_{\bullet}}^{w}.$$
(6.13)

**Proposition 6.3.4.** The morphisms of groupoids  $X_{W^+,\mathcal{F}_{\bullet}}^w \to X_{W^+,\mathcal{F}_{\bullet}}^{\Box,w}$ ,  $X_{W^+,\mathcal{F}_{\bullet}}^{\Box,w} \to X_{W^+,\mathcal{F}_{\bullet}}^{\Box,w}$ ,  $X_{D,\mathcal{M}_{\bullet}}^{w} \to X_{D,\mathcal{M}_{\bullet}}^{\Box,w} \to X_{D,\mathcal{M}_{\bullet}}^{\Box,w}$  are relatively representable and are closed immersions.

*Proof.* The proposition follows by the same argument as in the proof of [19, Prop. 3.5.10] with [19, (3.25)] replaced by (6.12).

Define

$$\begin{aligned} \mathcal{S}(y) &:= \{ w \in \mathscr{W}_L \mid y \in X_w(E) \} = \{ w \in \mathscr{W}_L \mid \hat{X}_{w,y} \neq \emptyset \} \\ &= \{ w \in \mathscr{W}_L \mid X^w_{W^+,\mathcal{F}_{\bullet}} \neq 0 \} = \{ w \in \mathscr{W}_L \mid X^w_{D,\mathcal{M}_{\bullet}} \neq 0 \}. \end{aligned}$$
(6.14)

Each groupoid Y in the isomorphism  $X_{D,\mathcal{M}_{\bullet}}^{\Box,w} \cong X_{D,\mathcal{M}_{\bullet}}^{\Box} \times_{X_{W^+,\mathcal{F}_{\bullet}}^{\Box}} X_{W^+,\mathcal{F}_{\bullet}}^{\Box,w}$  is equivalent to the associated functor |Y| (as all the automorphisms of an object in the groupoid are trivial). Hence  $|X_{D,\mathcal{M}_{\bullet}}^{\Box,w}| \cong |X_{D,\mathcal{M}_{\bullet}}^{\Box}| \times_{|X_{W^+,\mathcal{F}_{\bullet}}^{\Box}|} |X_{W^+,\mathcal{F}_{\bullet}}^{\Box,w}|$ . From Proposition 6.3.2 and Corollary 6.3.3, we deduce:

**Corollary 6.3.5.** If  $w \in \mathcal{S}(y)$ , the functor  $|X_{D,\mathcal{M}_{\bullet}}^{\Box,w}|$  is pro-representable by a noetherian complete local *E*-algebra which is formally smooth of relative dimension  $[L:\mathbb{Q}_p]\dim\mathfrak{p}$  over  $\widehat{X}_{w,y}$ .

The map  $(\kappa_B, \kappa_P) : X_{P,L} \to \mathfrak{t}_L \times_{\mathfrak{t}_L/\mathscr{W}_L} \mathfrak{z}_{L_P,L} = \mathscr{T}_{P,L}$  induces a morphism  $\widehat{X}_{P,L,y} \to \widehat{\mathscr{T}}_{P,L,(0,0)}$ . Denote by  $\Theta$  the composition

$$X_{D,\mathcal{M}_{\bullet}}^{\Box} \longrightarrow X_{W^{+},\mathcal{F}_{\bullet}}^{\Box} \xrightarrow{\sim} |X_{W^{+},\mathcal{F}_{\bullet}}^{\Box}| \xrightarrow{\sim} \widehat{X}_{P,L,y} \longrightarrow \widehat{\mathscr{T}}_{P,L,(0,0)},$$

which factors through a morphism still denoted by  $\Theta: X_{D,\mathcal{M}_{\bullet}} \to \widehat{\mathscr{T}}_{P,L,(0,0)}$ . By Lemma 5.2.7, we have

**Corollary 6.3.6.** Let  $w \in \mathcal{S}(y)$  and  $w' \in \mathscr{W}_L$ , then the morphisms  $X_{D,\mathcal{M}_{\bullet}}^{\Box,w} \hookrightarrow X_{D,\mathcal{M}_{\bullet}}^{\Box} \to \widehat{\mathscr{T}}_{P,L,(0,0)}$ and  $X_{D,\mathcal{M}_{\bullet}}^w \hookrightarrow X_{D,\mathcal{M}_{\bullet}} \to \widehat{\mathscr{T}}_{P,L,(0,0)}$  of groupoids over  $\operatorname{Art}(E)$  induced by  $\Theta$  factor through the embedding  $\widehat{\mathscr{T}}_{w',(0,0)} \hookrightarrow \widehat{\mathscr{T}}_{P,L,(0,0)}$  if and only if  $\mathscr{W}_{L_P,L}w' = \mathscr{W}_{L_P,L}w$ .

### 6.4 Galois representations and Bernstein paraboline varieties

We show that the completed local rings of the Bernstein paraboline varieties at generic points of distinct integral weights can be described (up to formally smooth morphisms) by completed local rings on the variety  $X_{P,L}$ .

Let  $\rho : \operatorname{Gal}_L \to \operatorname{GL}_n(E)$  be a continuous group morphism and let V be the associated representation of  $\operatorname{Gal}_L$ . Let  $X_\rho$  be the groupoid over  $\operatorname{Art}(E)$  of deformations of the group morphism  $\rho$ , and  $X_V$  be the groupoid over  $\operatorname{Art}(E)$  of deformations of the representation V (so  $X_\rho$  can be viewed as the groupoid of framed deformations of V). The natural morphism  $X_\rho \to X_V$  is relatively representable and formally smooth of relative dimension  $n^2$ . Let  $D := D_{\operatorname{rig}}(V)$ , we have then an equivalence  $X_V \xrightarrow{\sim} X_D$ . The morphism  $X_\rho \to |X_\rho|$  is an equivalence. In fact, this holds for any groupoid with " $\rho$ " in subscript in this section.

Assume that D is almost de Rham with distinct Sen weights and that D admits a generic  $\Omega$ -filtration. Let  $\mathcal{M}_{\bullet}$  be a generic  $\Omega$ -filtration on  $\mathcal{M} = D[\frac{1}{t}]$  (recall  $\Omega$  is fixed in § 6.2) and put  $X_{V,\mathcal{M}_{\bullet}} := X_V \times_{X_D} X_{D,\mathcal{M}_{\bullet}}, X_{\rho,\mathcal{M}_{\bullet}} := X_\rho \times_{X_V} X_{V,\mathcal{M}_{\bullet}} \cong X_\rho \times_{X_{\mathcal{M}}} X_{\mathcal{M},\mathcal{M}_{\bullet}}$ . Note that  $X_{\rho,\mathcal{M}_{\bullet}} \to X_\rho$  is a closed immersion by Proposition 6.2.8 and base change. For  $w \in \mathscr{W}_L$ , we put

$$X_{V,\mathcal{M}_{\bullet}}^{w} := X_{V} \times_{X_{D}} X_{D,\mathcal{M}_{\bullet}}^{w} \text{ and } X_{\rho,\mathcal{M}_{\bullet}}^{w} := X_{\rho} \times_{X_{V}} X_{V,\mathcal{M}_{\bullet}}^{w}$$

(where we use  $W^+ := W^+_{dR}(D) \cong B^+_{dR} \otimes_{\mathbb{Q}_p} V$  in the definition of  $X^w_{D,\mathcal{M}_{\bullet}}$ , see (6.13)). Let  $y \in X_{P,L}$  be as in (6.10) and  $\mathcal{S}(y)$  as in (6.14).

**Theorem 6.4.1.** (1) The functor  $|X_{\rho,\mathcal{M}_{\bullet}}|$  (resp.  $|X_{\rho,\mathcal{M}_{\bullet}}^w|$  for  $w \in \mathcal{S}(y)$ ) is pro-representable by an equidimensional noetherian complete local ring  $R_{\rho,\mathcal{M}_{\bullet}}$  (resp.  $R_{\rho,\mathcal{M}_{\bullet}}^w$ ) of residue field E and dimension  $n^2 + [L:\mathbb{Q}_p](\frac{n(n-1)}{2} + r)$ . If  $\rho$  is moreover de Rham<sup>5</sup>, then  $R_{\rho,\mathcal{M}_{\bullet}}^w \cong R_{\rho,\mathcal{M}_{\bullet}}/\mathfrak{p}_w$  for a minimal prime ideal  $\mathfrak{p}_w$  of  $R_{\rho,\mathcal{M}_{\bullet}}$ . Finally, in this case, the map  $w \mapsto \mathfrak{p}_w$  is a bijection between  $\mathcal{S}(y)$  and the set of minimal prime ideals of  $R_{\rho,\mathcal{M}_{\bullet}}$ .

(2) The morphism  $|X_{\rho,\mathcal{M}_{\bullet}}^{w'}| \to |X_{V,\mathcal{M}_{\bullet}}^{w'}| \hookrightarrow |X_{V,\mathcal{M}_{\bullet}}| \cong |X_{D,\mathcal{M}_{\bullet}}| \xrightarrow{\Theta} \widehat{\mathscr{T}}_{P,L,(0,0)}$  of groupoids over Art(E) factors through  $\widehat{\mathscr{T}}_{w,(0,0)} \hookrightarrow \widehat{\mathscr{T}}_{P,L,(0,0)}$  if and only if  $\mathscr{W}_{L_{P},L}w' = \mathscr{W}_{L_{P},L}w$ .

*Proof.* (1) We have  $X_{\rho,\mathcal{M}_{\bullet}} \cong X_{\rho} \times_{X_{\mathcal{M}}} X_{\mathcal{M},\mathcal{M}_{\bullet}}$ . By Proposition 6.2.8 and the fact that  $X_{\rho}$  is pro-representable, we deduce that  $X_{\rho,\mathcal{M}_{\bullet}}$  is pro-representable. We have  $X_{\rho,\mathcal{M}_{\bullet}} \xrightarrow{\sim} |X_{\rho,\mathcal{M}_{\bullet}}|$  and we let  $R_{\rho,\mathcal{M}_{\bullet}}$  be the noetherian complete local ring which pro-represents  $|X_{\rho,\mathcal{M}_{\bullet}}|$ . Define

$$X_{\rho,\mathcal{M}\bullet}^{\Box} := X_{\rho,\mathcal{M}\bullet} \times_{X_{\mathcal{M},\mathcal{M}\bullet}} X_{\mathcal{M},\mathcal{M}\bullet}^{\Box} \cong X_{\rho} \times_{X_{\mathcal{M}}} X_{\mathcal{M},\mathcal{M}\bullet}^{\Box}$$

which is formally smooth of relative dimension  $[L:\mathbb{Q}_p]n^2$  over  $X_{\rho,\mathcal{M}_{\bullet}}$  by base change (see (6.4)). Since  $X_{\rho} \to X_V \cong X_D$  is relatively representable and formally smooth of relative dimension  $n^2$ , so is the morphism  $X_{\rho,\mathcal{M}_{\bullet}}^{\Box} \to X_D \times_{X_{\mathcal{M}}} X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box} \cong X_{D,\mathcal{M}_{\bullet}}^{\Box}$ . Together with Proposition 6.3.2 and Corollary 5.2.2, we deduce that  $R_{\rho,\mathcal{M}_{\bullet}}$  is equidimensional and

$$\dim R_{\rho,\mathcal{M}_{\bullet}} = n^2 + \left( [L:\mathbb{Q}_p]\dim\mathfrak{p} \right) + \left( [L:\mathbb{Q}_p] \left( \frac{n(n-1)}{2} + \dim\mathfrak{r}_P \right) \right) - n^2 [L:\mathbb{Q}_p]$$
  
$$= n^2 + [L:\mathbb{Q}_p] \left( \frac{n(n-1)}{2} + r \right).$$
(6.15)

Let  $X_{\rho,\mathcal{M}_{\bullet}}^{\Box,w} := X_{\rho,\mathcal{M}_{\bullet}}^w \times_{X_{\mathcal{M},\mathcal{M}_{\bullet}}} X_{\mathcal{M},\mathcal{M}_{\bullet}}^{\Box} \cong X_{D,\mathcal{M}_{\bullet}}^{\Box,w} \times_{X_D} X_{\rho}$ . As  $X_{\rho}$  is relatively representable over  $X_D$ , we deduce by Corollary 6.3.5 (and the fact  $X_{D,\mathcal{M}_{\bullet}}^{\Box,w} \xrightarrow{\sim} |X_{D,\mathcal{M}_{\bullet}}^{\Box,w}|$ ) that  $X_{\rho,\mathcal{M}_{\bullet}}^{\Box,w}$  is prorepresentable. It is also easy to see  $X_{\rho,\mathcal{M}_{\bullet}}^{\Box,w} \xrightarrow{\sim} |X_{\rho,\mathcal{M}_{\bullet}}^{\Box,w}|$ . As  $X_{\rho}$  is formally smooth of relative dimension  $n^2$  over  $X_D$ , using Corollary 6.3.5 again we have formally smooth morphisms (the first of relative dimension  $[L:\mathbb{Q}_p]n^2$ , the second of relative dimension  $n^2 + [L:\mathbb{Q}_p]\dim\mathfrak{p}$ )

$$|X^{w}_{\rho,\mathcal{M}_{\bullet}}| \longleftarrow |X^{\Box,w}_{\rho,\mathcal{M}_{\bullet}}| \longrightarrow \widehat{X}_{w,y}.$$
(6.16)

As  $X_{\rho,\mathcal{M}_{\bullet}}^{w} \cong X_{D,\mathcal{M}_{\bullet}}^{w} \times_{X_{D,\mathcal{M}_{\bullet}}} X_{\rho,\mathcal{M}_{\bullet}}$ , by (6.15), Proposition 6.3.4 and (6.16) (and the fact  $X_{\rho,\mathcal{M}_{\bullet}}^{w} \xrightarrow{\sim} |X_{\rho,\mathcal{M}_{\bullet}}^{w}|$ ), it follows that  $|X_{\rho,\mathcal{M}_{\bullet}}^{w}|$  is pro-representable by a (reduced) local complete noetherian ring of residue field E and dimension  $n^{2} + [L:\mathbb{Q}_{p}](\frac{n(n-1)}{2} + r)$ . When  $\rho$  is de Rham, the parameter  $N_{W}$  in y is zero, hence  $\widehat{\mathcal{O}}_{X_{w},y}$  is a domain by Theorem 5.3.1. The second part of (1) follows. Using Corollary 5.2.2, the last part of (1) also follows. Part (2) follows easily from Corollary 6.3.6.  $\Box$ 

**Corollary 6.4.2.** For  $w \in \mathcal{S}(y)$ , we have

$$\dim_E |X_{\rho,\mathcal{M}_{\bullet}}^w|(E[\varepsilon]/\varepsilon^2) = [L:\mathbb{Q}_p]\dim\mathfrak{p} + n^2 - n^2[L:\mathbb{Q}_p] + \dim_E \widehat{X}_{w,y}(E[\varepsilon]/\varepsilon^2).$$

*Proof.* The groupoid  $X_{\rho,\mathcal{M}_{\bullet}}^{\Box,w}$  is formally smooth of relative dimension  $n^2$  over  $X_{V,\mathcal{M}_{\bullet}}^{\Box,w} \cong X_{D,\mathcal{M}_{\bullet}}^{\Box,w}$ , and is formally smooth of relative dimension  $[L:\mathbb{Q}_p]n^2$  over  $X_{\rho,\mathcal{M}_{\bullet}}^w$ . We have  $Y \xrightarrow{\sim} |Y|$  for

<sup>&</sup>lt;sup>5</sup>Or equivalently potentially crystalline as D admits a generic  $\Omega$ -filtration.

 $Y \in \{X_{\rho,\mathcal{M}_{\bullet}}^{\Box,w}, X_{\rho,\mathcal{M}_{\bullet}}^{w}, X_{D,\mathcal{M}_{\bullet}}^{\Box,w}\}, \text{ and thus}$ 

$$\dim_{E} |X_{\rho,\mathcal{M}_{\bullet}}^{w}|(E[\varepsilon]/\varepsilon^{2}) = \dim_{E} |X_{D,\mathcal{M}_{\bullet}}^{\Box,w}|(E[\varepsilon]/\varepsilon^{2}) + n^{2} - [L:\mathbb{Q}_{p}]n^{2}$$
$$= [L:\mathbb{Q}_{p}]\dim\mathfrak{p} + n^{2} - n^{2}[L:\mathbb{Q}_{p}] + \dim_{E}\widehat{X}_{w,y}(E[\varepsilon]/\varepsilon^{2})$$

where the second equality follows from Corollary 6.3.5.

Let  $w_y \in \mathscr{W}_{\max,L}^P$  such that  $\pi(y) = (\alpha^{-1}(\mathcal{D}_{\bullet}), \alpha^{-1}(\operatorname{Fil}_{W^+, \bullet}))$  lies in the  $G_L$ -orbit of  $(w_y, 1)$  in  $G_L/B_L \times G_L/P_L$ . By Lemma 5.2.4 and the equalities in (6.14), we have

**Proposition 6.4.3.** Let  $w \in \mathscr{W}_L$ , if  $X^w_{\rho,\mathcal{M}_{\bullet}} \neq \emptyset$ , then  $w^{\max} \ge w_y$ .

Now fix a group morphism  $\overline{\rho}$ :  $\operatorname{Gal}_L \to \operatorname{GL}_n(k_E)$  and a strictly *P*-dominant weight  $\mathbf{h} \in \mathbb{Z}^{n|\Sigma_L|}$ of  $G_L$  as in § 4.2. Let  $x = (\rho, \underline{x}, \chi)$  be a point in  $X_{\Omega, \mathbf{h}}(\overline{\rho}) \hookrightarrow \mathfrak{X}_{\overline{\rho}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0, L}$ . Assume that  $\rho$  is almost de Rham (hence  $\chi$  is locally algebraic by Proposition 4.2.10) and has distinct Sen weights.

**Corollary 6.4.4.** We have that  $\mathcal{M} := D_{\mathrm{rig}}(\rho)[\frac{1}{t}]$  has an  $\Omega$ -filtration  $\mathcal{M}_{\bullet}$  of parameter  $(\underline{x}, \chi_{\overline{\omega}_L})$ . Moreover, if the parameter  $(\underline{x}, \chi_{\overline{\omega}_L})$  is generic, then  $\mathcal{M}_{\bullet}$  is the unique  $\Omega$ -filtration of parameter  $(\underline{x}, \chi_{\overline{\omega}_L})$  on  $\mathcal{M}$ .

*Proof.* The existence follows from Corollary 4.2.6. By Lemma 6.2.5 (1) and the same argument as in the proof of Corollary 4.1.13, the uniqueness follows.  $\Box$ 

Recall that  $X_{\rho} \cong |X_{\rho}|$  is equivalent to  $(\widehat{\mathfrak{X}_{\rho}})_{\rho}$  (cf. [62, § 2.3]). We have a natural morphism of formal schemes

$$\widehat{X_{\Omega,\mathbf{h}}(\overline{\rho})}_x \longrightarrow (\widehat{\mathfrak{X}_{\overline{\rho}}})_\rho \cong X_\rho.$$

**Proposition 6.4.5.** (1) The canonical morphism  $\widehat{X_{\Omega,h}(\overline{\rho})}_x \longrightarrow X_{\rho}$  factors through a morphism

$$\widehat{X_{\Omega,h}(\overline{\rho})}_x \longrightarrow X_{\rho,\mathcal{M}_{\bullet}}.$$

(2) The morphisms  $\widehat{X_{\Omega,h}(\overline{\rho})}_x \to X_{\rho}$  and  $\widehat{X_{\Omega,h}(\overline{\rho})}_x \to X_{\rho,\mathcal{M}_{\bullet}}$  are closed immersions of groupoids over  $\operatorname{Art}(E)$ .

*Proof.* (1) follows from the same argument as in the proof of [19, Prop. 3.7.2] with [60, Cor. 6.3.10] replaced by Corollary A.1.2 and Corollary A.1.3 (2). (2) follows from the same argument as in the proof of [19, Prop. 3.7.3].

Consider the composition

$$\Theta_x: \widehat{X_{\Omega,\mathbf{h}}(\overline{\rho})}_x \hookrightarrow X_{\rho,\mathcal{M}_{\bullet}} \to X_{V,\mathcal{M}_{\bullet}} \cong X_{D,\mathcal{M}_{\bullet}} \xrightarrow{\Theta} \widehat{\mathscr{T}}_{P,L,(0,0)}.$$

Let  $\mathbf{h}' = (h'_{1,\tau} > h'_{2,\tau} > \cdots > h'_{n,\tau})_{\substack{i=1,\dots,n\\ \tau \in \Sigma_L}}$  be the Sen weights of  $\rho$ . Then by Proposition 4.2.10, there exists  $w_x = (w_{x,\tau})_{\tau \in \Sigma_L} \in \mathscr{W}_{\min,L}^P$  such that, for  $j = 1, \dots, n$ ,  $h'_{w_{x,\tau}^{-1}(j),\tau} = \operatorname{wt}(\chi_i)_{\tau} + h_{j,\tau}$ where i is the integer such that  $s_{i-1} < j \leq s_i$ .

**Proposition 6.4.6.** The morphism  $\Theta_x$  factors through  $\widehat{\mathscr{T}}_{w_x w_{0,L},(0,0)}$ .

*Proof.* The proposition follows by the same argument as for [19, Lemma 3.7.4] using Proposition 4.2.10.

**Corollary 6.4.7.** Assume moreover that  $\rho$  is de Rham. The closed immersion  $X_{\Omega,h}(\overline{\rho})_x \hookrightarrow X_{\rho,\mathcal{M}_{\bullet}}$ factors through an isomorphism

$$\widehat{X_{\Omega,h}(\bar{\rho})}_x \xrightarrow{\sim} X^{w_x w_{0,L}}_{\rho,\mathcal{M}_{\bullet}}.$$
(6.17)

In particular,  $X_{\Omega,h}(\overline{\rho})$  is unibranch, hence irreducible, at x.

Proof. By Theorem 4.2.5 (1) and Theorem 6.4.1 (1),  $\dim X_{\Omega,\mathbf{h}}(\overline{\rho})_x = \dim X_{\rho,\mathcal{M}_{\bullet}}$ . We also know that  $X_{\Omega,\mathbf{h}}(\overline{\rho})$  is reduced and equidimensional by Proposition 4.2.5 (1). Hence  $\widehat{X_{\Omega,\mathbf{h}}(\overline{\rho})}_x$  is isomorphic to a union of irreducible components  $\operatorname{Spec} R^w_{\rho,\mathcal{M}_{\bullet}}$  of  $\operatorname{Spec} R_{\rho,\mathcal{M}_{\bullet}}$  (see Theorem 6.4.1 (1)). However, it follows from Proposition 6.4.6 and Theorem 6.4.1 (2) that  $\operatorname{Spec} R^w_{\rho,\mathcal{M}_{\bullet}}$  can not be contained in  $\widehat{X_{\Omega,\mathbf{h}}(\overline{\rho})}_x$  if  $\mathscr{W}_{L_P,L}w \neq \mathscr{W}_{L_P,L}(w_x w_{0,L})$ . We then obtain the isomorphism in (6.17).

Let  $y \in X_{P,L}(k(x))$  be the point in (6.10) associated to  $(\rho, \mathcal{M}_{\bullet})$ . By Corollary 6.4.7 and Proposition 6.4.3, we have

**Corollary 6.4.8.** Assume moreover that  $\rho$  is de Rham, then  $w_x w_{0,L} \ge w_y$ .

By corollary 5.3.4, we deduce:

**Corollary 6.4.9.** Assume that  $\rho$  is de Rham, that  $\pi(y) \in G_L/P_L \times G_L/B_L$  lies in the smooth locus of the closure of  $G_L(1, w_x w_{0,L})P_L \times B_L$ , and that

$$\dim \mathfrak{z}_{L_P,L}^{w_x w_{0,L} w_y^{-1}} + \lg(w_x w_{0,L}) - \lg(w_y) = \dim \mathfrak{z}_{L_P,L}.$$

Then  $X_{\Omega,h}(\overline{\rho})$  is smooth at the point  $x = (\rho, \underline{x}, \chi)$ .

By the discussion in Remark 5.3.5 (1), we obtain the following special case that will be frequently used:

**Corollary 6.4.10.** Assume that  $\rho$  is de Rham, that  $\lg(w_y) \geq \lg(w_x w_{0,L}) - 2$  and that

$$\dim \mathfrak{z}_{L_P,L} - \dim \mathfrak{z}_{L_P,L}^{w_x w_{0,L} w_y^{-1}} = 2 \quad if \ \lg(w_y) = \lg(w_x w_{0,L}) - 2.$$

Then  $X_{\Omega,h}(\overline{\rho})$  is smooth at the point  $x = (\rho, \underline{x}, \chi)$ .

**Remark 6.4.11.** When  $P \neq B$ , it could happen that  $\dim \mathfrak{z}_{L_P,L} - \dim \mathfrak{z}_{L_P,L}^{w_x w_{0,L} w_y^{-1}} = 1$  while  $\lg(w_y) = \lg(w_x w_{0,L}) - 2$  (see Remark A.3.5 below, and compare with [19, Rem. 4.1.6]). In this case, we don't know if  $X_{\Omega,h}(\overline{\rho})$  is smooth at the point x.

As a quick application, we obtain the following full description of local companion points which completes Corollary 4.3.9. Changing notation, let  $z := (\rho, (\mathbf{r}_1, \ldots, \mathbf{r}_r)) \in \widetilde{U}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$  be as in § 4.3 (where it was denoted x (see above Proposition 4.3.8)):  $\rho$  is now a generic potentially crystalline representation with distinct Hodge-Tate weights  $\mathbf{h}, \xi_0 = \bigoplus_{i=1}^r \xi_i$  with  $\xi_i$  the inertial type of  $\Omega_i$ , and  $(\mathbf{r}_1, \ldots, \mathbf{r}_r) \in (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}}$  is an (ordered) r-tuple of absolutely irreducible Weil-Deligne representations such that  $\mathbf{r}(\rho) \cong \bigoplus_{i=1}^r \mathbf{r}_i$ . Recall we have attached to the point z an element  $w_z \in \mathscr{W}_{\max,L}^P$  (see before Proposition 4.3.8 where  $w_z$  was denoted  $w_x$ ). In fact, if we let y be the point in (6.10) associated to z (for  $D = D_{\operatorname{rig}}(\rho)$  and  $\mathcal{M}_{\bullet}$  the  $\Omega$ -filtration on  $\mathcal{M} = D[\frac{1}{t}]$ associated to the filtration  $\{\bigoplus_{i=1}^j \mathbf{r}_i\}_{i=1}^r$  on  $\mathbf{r}(\rho)$  as in § 2.3), then by definition  $w_z = w_y$ .
**Corollary 6.4.12.** Let  $w \in \mathscr{W}_{\max L}^{P}$ , with the above notation the point

 $(\rho, (\mathbf{r}_1, \ldots, \mathbf{r}_r), 1) \in \mathfrak{X}_{\overline{\rho}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}_{0,L}}$ 

lies in  $X_{\Omega,ww_{0,L}(h)}(\overline{\rho})$  if and only if  $w \ge w_z$ .

*Proof.* The "if" part is Corollary 4.3.9. The "only if" part follows from Corollary 6.4.8.

**Remark 6.4.13.** The case for P = B (and  $\rho$  crystalline) was proved in [19, Thm. 4.2.3], see Remark 4.3.10 for related discussions.

# 6.5 Galois cycles

We construct certain cycles on the deformation space  $X_{\rho} \cong (\mathfrak{X}_{\overline{\rho}})_{\rho}$  of a characteristic zero representation  $\rho$  of  $\operatorname{Gal}_L$ .

Let  $\overline{\rho}$  be as in § 6.4 and  $\rho \in \mathfrak{X}_{\overline{\rho}}(E)$ . Recall that the local complete noetherian *E*-algebra  $\widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho},\rho}}$  is equidimensional of dimension  $n^2 + [L : \mathbb{Q}_p]n^2$  and pro-represents the functor  $|X_{\rho}|$  of (framed) deformations of  $\rho$  over  $\operatorname{Art}(E)$ . Denote by  $Z(\operatorname{Spec} \widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho},\rho}})$  (resp.  $Z^d(\operatorname{Spec} \widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho},\rho}})$  for  $d \in \mathbb{Z}_{\geq 0}$ ) the free abelian group generated by the irreducible closed subschemes (resp. the irreducible closed subschemes of codimension d) in  $\operatorname{Spec} \widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho},\rho}}$ . If A is a noetherian complete local ring which is a quotient of  $\widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho},\rho}}$ , we set

$$[\operatorname{Spec} A] := \sum_{\mathfrak{p}} m(\mathfrak{p}, A) [\operatorname{Spec} A/\mathfrak{p}] \in Z(\operatorname{Spec} \widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho}, \rho}})$$

where the sum is over the minimal prime ideals  $\mathfrak{p}$  of A,  $m(\mathfrak{p}, A) \in \mathbb{Z}_{\geq 0}$  is the finite length of  $A_{\mathfrak{p}}$  as a module over itself and  $[\operatorname{Spec} A/\mathfrak{p}]$  is the irreducible component  $\operatorname{Spec} A/\mathfrak{p}$  seen in  $Z(\operatorname{Spec} \widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho},\rho}})$ .

Assume that  $\rho$  has integral distinct  $\tau$ -Sen weights for each  $\tau \in \Sigma_L$ , and that  $D_{\mathrm{rig}}(\rho)[1/t]$  admits an  $\Omega$ -filtration  $\mathcal{M}_{\bullet}$  of generic parameter  $(\underline{x}, \delta) \in \operatorname{Spec} \mathcal{Z}_{\Omega} \times \widehat{\mathcal{Z}_{L_P}(L)}$  (so  $\delta$  is locally algebraic). We refer to § 5.4 (and the very beginning of § 6) for the schemes  $\overline{X}_{P,L}$  and  $\overline{X}_w$ . Let  $y \in \overline{X}_{P,L}(E) \subseteq$  $X_{P,L}(E)$  be the point associated to  $(\rho, \mathcal{M}_{\bullet})$  in (6.10) (depending on a choice of framing  $\alpha$ ). Let  $w \in \mathcal{W}_L$  such that  $y \in \overline{X}_w(E) \subset X_w(E)$  (which implies  $w^{\min}w_{0,L} \geq w_y$  by Proposition 6.4.3). Similarly as in [19, § 4.3], by Proposition 6.2.3 and Theorem 6.4.1 (and its proof), we have a commutative diagram of groupoids over  $\operatorname{Art}(E)$ :



where we still denote by  $\omega_{\delta}$  the composition  $X_{\rho,\mathcal{M}_{\bullet}} \to X_{\mathcal{M},\mathcal{M}_{\bullet}} \xrightarrow{\omega_{\delta}} Z_{L_{P}}(\widehat{L})_{\delta}$ . Taking everywhere (except for  $X_{\rho}$ ) the fibres over  $0 \in \widehat{\mathfrak{z}}_{L_{P},L}$ , we obtain the commutative diagram of affine schemes

over E:



For  $w = (w_{\tau}) \in \mathscr{W}_L$ , we denote  $Z_w := \prod_{\tau \in \Sigma_L} Z_{w_{\tau}} \hookrightarrow Z_{P,L}$  (see § 5.4 for  $Z_{w_{\tau}}$ ). The irreducible components of Spec  $\widehat{\mathcal{O}}_{\overline{X}_{P,L},y}$  are the union of the irreducible components of Spec  $\widehat{\mathcal{O}}_{Z_w,y}$  for w such that  $y \in Z_w(E)$  (the last condition does not depend on the choice of the framing  $\alpha$ ). By pull-back and smooth descent, we obtain a bijection between the irreducible components of Spec  $\widehat{\mathcal{O}}_{\overline{X}_{P,L},y}$ and the irreducible components of Spec  $\overline{R}_{\rho,\mathcal{M}_{\bullet}}$ . In particular, Spec  $\overline{R}_{\rho,\mathcal{M}_{\bullet}}$  is equidimensional of dimension  $n^2 + [L : \mathbb{Q}_p] \frac{n(n-1)}{2}$  (equivalently of codimension  $[L : \mathbb{Q}_p] \frac{n(n+1)}{2}$  in Spec  $\widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho},\rho}}$ ). For  $w \in \mathscr{W}_L$ , denote by

$$\mathfrak{Z}_w \in Z^{[L:\mathbb{Q}_p]\frac{n(n+1)}{2}}(\operatorname{Spec}\widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho},\rho}}) \tag{6.18}$$

the cycle corresponding via the embedding  $\operatorname{Spec} \overline{R}_{\rho,\mathcal{M}_{\bullet}} \hookrightarrow \operatorname{Spec} \widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho},\rho}}$  to the cycle  $[\operatorname{Spec} \widehat{\mathcal{O}}_{Z_{w},y}]$ under this bijection. By Theorem 5.4.3, we have:

**Lemma 6.5.1.** Assume that  $\rho$  is de Rham. Then the cycle  $\mathfrak{Z}_w$  is irreducible for  $w \in \mathscr{W}_L$ .

**Remark 6.5.2.** It is easy to see that  $Z_{w_{0,L}} = G_L/B_L \times G_L/P_L \times \{0\}$ . Indeed, we have a natural embedding  $G_L/B_L \times G_L/P_L \times \{0\} \hookrightarrow Z_{w_{0,L}}$ , which has to be an isomorphism since both are irreducible schemes of the same dimension. Thus if  $\mathfrak{Z}_{w_0} \neq 0$ , the nilpotent operator  $N_W$  associated to  $\rho$  is zero, hence  $\rho$  is de Rham. As  $\rho$  admits a generic  $\Omega$ -filtration  $\mathcal{M}_{\bullet}$ ,  $\rho$  is potentially crystalline. Let  $\mathbf{h}$  be the Hodge-Tate weights of  $\rho$  and  $\xi_0$  the inertial type of  $\rho$  (thus  $\xi_0$  has the form  $\bigoplus_{i=1}^r \xi_i$ ). By the same argument as in [19, Rem. 4.3.1] (using [63]), it follows that  $\mathfrak{Z}_{w_0} = [\widehat{\mathcal{O}}_{\mathfrak{X}_{2}^{\mathrm{pcr}}(\xi_0, \mathbf{h}), \rho]$ .

For  $w = (w_{\tau})$  and  $w' = (w'_{\tau}) \in \mathscr{W}_L$ , put  $a_{w,w'} := \prod_{\tau \in \Sigma_L} a_{w_{\tau},w'_{\tau}}$  and  $b_{w,w'} := \prod_{\tau \in \Sigma_L} b_{w_{\tau},w'_{\tau}}$ where  $a_{w_{\tau},w'_{\tau}}$ ,  $b_{w_{\tau},w'_{\tau}}$  are given as in Theorem 5.4.11 (applied to  $G = \operatorname{GL}_n$ ). Put

$$\mathfrak{C}_{w} := \sum_{w' \in \mathscr{W}_{L_{P}, L} \setminus \mathscr{W}_{L}} a_{w, w'} \mathfrak{Z}_{w'} \in Z^{[L:\mathbb{Q}_{p}]\frac{n(n+1)}{2}}(\operatorname{Spec}\widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho}, \rho}}).$$
(6.19)

Note that  $\mathfrak{Z}_w$  and  $\mathfrak{C}_w$  are independent of the choice of the representative w in its associated class in  $\mathscr{W}_{L_{P,L}} \setminus \mathscr{W}_{L}$ .

**Lemma 6.5.3.** Assume that  $\rho$  is de Rham, and let  $w \in \mathscr{W}_L$ . The followings are equivalent: (1)  $\mathfrak{C}_w \neq 0$ , (2)  $\mathfrak{Z}_w \neq 0$ , (3)  $w^{\max} \geq w_y$ .

*Proof.*  $(2) \Rightarrow (1)$  is clear.

(1)  $\Rightarrow$  (3): If  $\mathfrak{C}_w \neq 0$ , then  $\mathfrak{Z}_{w'} \neq 0$  ( $\Rightarrow y \in X_{w'}$ ) for some w' with  $w'^{\max} \leq w^{\max}$ . By Proposition 6.4.3,  $w'^{\max} \geq w_y$  hence  $w^{\max} \geq w_y$ .

(3)  $\Rightarrow$  (2): As  $\rho$  is de Rham (i.e. the entry  $N_W$  in y is zero), if  $w^{\max} \ge w_y$ , then y is contained in the Zariski-closure of  $(G_L(w^{\max}, 1)B_L \times P_L) \times \{0\}$  in  $G_L/B_L \times G_L/P_L \times \{0\} \hookrightarrow Z_{P,L}$ , thus  $y \in Z_w \Rightarrow \mathfrak{Z}_w \neq 0$ . Now let  $x = (\rho, \underline{x}, \chi) \in X_{\Omega, \mathbf{h}}(\overline{\rho})(E)$  such that  $\rho$  is de Rham and  $(\underline{x}, \chi)$  is generic. Let  $\mathcal{M}_{\bullet}$  be the unique  $\Omega$ -filtration on  $D_{\mathrm{rig}}(\rho)[1/t]$  of parameter  $(\underline{x}, \chi_{\overline{\omega}_L})$  (cf. Corollary 6.4.4) and y be the point (6.10) of  $X_{P,L}$  associated to  $(\rho, \mathcal{M}_{\bullet})$ . Recall that we have defined elements  $w_y \in \mathcal{W}_{\mathrm{max},L}^P$ and  $w_x \in \mathcal{W}_{\mathrm{min},L}^P$  (see above Proposition 6.4.3 for  $w_y$  and above Proposition 6.4.6 for  $w_x$ ). Let  $X_{\Omega,\mathbf{h}}(\overline{\rho})_{\mathrm{wt}(\chi)}$  denote the fibre of  $X_{\Omega,\mathbf{h}}(\overline{\rho})$  at  $\mathrm{wt}(\chi)$  (via the morphism  $X_{\Omega,\mathbf{h}}(\overline{\rho}) \to \widehat{\mathcal{Z}_{0,L}} \xrightarrow{\mathrm{wt}} \mathfrak{z}_{L_P,L}$ ). The following conjecture is a consequence of Conjecture 5.4.9 and Lemma 6.5.3:

#### Conjecture 6.5.4. We have

$$[\operatorname{Spec}\widehat{\mathcal{O}}_{X_{\Omega,h}(\overline{\rho})_{\operatorname{wt}(\chi)},x}] = \sum_{\substack{w \in \mathscr{W}_{L_{P,L}} \setminus \mathscr{W}_{L} \\ w_{y} \leq w^{\max} \leq w_{x}w_{0,L}}} b_{w_{x}w_{0,L},w} \mathfrak{C}_{w} \in Z^{[L:\mathbb{Q}_{p}]\frac{n(n+1)}{2}}(\operatorname{Spec}\widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho},\rho}})$$

**Remark 6.5.5.** It follows from Lemma 5.4.13 that Conjecture 6.5.4 holds if x is moreover a smooth point of  $X_{\Omega,h}(\overline{\rho})_{wt(\chi)}$ .

# 7 Applications

Under the Taylor-Wiles hypothesis, we show several (global) results on p-adic automorphic representations including a classicality result, and the existence of all expected companion constituents for certain parabolic subgroup P.

### 7.1 Automorphy cycles

We use locally analytic representation theory to construct certain cycles on patched Bernstein eigenvarieties.

# 7.1.1 Representation theoretic preliminaries

We give some preliminaries on locally analytic representations which we will use in § 7.1.2. We use the notation of § 3.1.

Denote by  $\mathcal{O}_{alg}^{\mathfrak{p}} \subset \mathcal{O}^{\mathfrak{p}}$  the full subcategory of objects with integral weights (see [74]). Let V be an admissible locally analytic representation of  $G_p$ . Let  $M \in \mathcal{O}_{alg}^{\mathfrak{p}}$ , the  $\mathfrak{p}$ -action on M canonically extends to a  $P(\mathbb{Q}_p)$ -action ([74, Lemma 3.2]). We equip  $\operatorname{Hom}_{U(\mathfrak{g})}(M, V)$  with the left action of  $P(\mathbb{Q}_p)$  given by  $(pf)(v) := pf(p^{-1}v)$  for  $p \in P(\mathbb{Q}_p)$ ,  $v \in V$ . Note that this action does preserve  $\operatorname{Hom}_{U(\mathfrak{g})}(M, V)$  because we have for  $p \in P(\mathbb{Q}_p)$ ,  $X \in U(\mathfrak{g})$  and  $v \in V$ :

$$(pf)(Xv) = pf(p^{-1}Xv) = pf(\operatorname{Ad}(p^{-1})(X)p^{-1}v) = p\operatorname{Ad}(p^{-1})(X)f(p^{-1}v) = Xpf(p^{-1}v) = X(pf)(v).$$

We also see that the derived  $\mathfrak{p}$ -action is trivial, so the  $P(\mathbb{Q}_p)$ -action on  $\operatorname{Hom}_{U(\mathfrak{g})}(M, V)$  is smooth. We equip  $\operatorname{Hom}_{U(\mathfrak{g})}(M, \Pi)^{N_P^0}$  with a natural Hecke action of  $L_P(\mathbb{Q}_p)^+$  defined as in (3.11).

For any  $M \in \mathcal{O}^{\mathfrak{b}}$  (in particular  $M \in \mathcal{O}^{\mathfrak{p}}_{alg}$ ), we endow  $\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M, V)$  with a canonical topology of space of compact type as follows. Choose finitely many weights  $\lambda^{i}$  of the Lie algebra of  $T_{\mathscr{I}}$ such that  $\oplus_{i} M_{B_{\mathscr{I}}}(\lambda^{i}) \twoheadrightarrow M$  where  $M_{B_{\mathscr{I}}}(\lambda^{i}) := \operatorname{U}(\mathfrak{g}) \otimes_{\operatorname{U}(\mathfrak{b}_{\mathscr{I}})} \lambda^{i}$ . Then  $\operatorname{Hom}_{\operatorname{U}(\mathfrak{g})}(M, V)$  is naturally a closed subspace of  $\bigoplus_i V[\mathfrak{b}_{\mathscr{I}} = \lambda^i]$  that we endow with the induced topology. Using that a continuous bijection of vector spaces of compact type is a topological isomorphism, one easily checks that this doesn't depend on the choice of the  $\lambda^i$  (for two choices  $\lambda^i$  and  $\mu^j$ , consider  $(\bigoplus_i M_{B_{\mathscr{I}}}(\lambda^i)) \oplus (\bigoplus_j M_{B_{\mathscr{I}}}(\mu^j)) \twoheadrightarrow M)$  and that for a morphism  $M \to N$  in  $\mathcal{O}^{\mathfrak{b}}$ , the induced morphism  $\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(N, V) \to \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M, V)$  is continuous. We denote by  $\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M, V)_{\mathrm{fs}}^{N_P^0}$  the finite slope part of  $\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M, V)_{P}^{N_P^0}$  defined as in [40, § 3.2]. The following lemma follows by the same argument as in the proof of [19, Lemma 5.2.1] (see [40, Prop. 3.2.4 (ii)] for the second isomorphism).

**Lemma 7.1.1.** Assume that V is very strongly admissible ([41, Def. 0.12]), let  $\pi$  be a finite length smooth representation of  $L_P(\mathbb{Q}_p)$  over E and  $M \in \mathcal{O}_{alg}^{\mathfrak{p}}$ . There are natural bijections

$$\operatorname{Hom}_{G_p} \left( \mathcal{F}_{P^-(\mathbb{Q}_p)}^{G_p} \left( \operatorname{Hom}_E(M, E)^{\mathfrak{n}_{P^-}^{\infty}}, \pi(\delta_P^{-1}) \right), V \right) \xrightarrow{\sim} \operatorname{Hom}_{L_P(\mathbb{Q}_p)^+} \left( \pi, \operatorname{Hom}_{U(\mathfrak{g})}(M, V)^{N_P^0}_{P} \right) \\ \cong \operatorname{Hom}_{L_P(\mathbb{Q}_p)} \left( \pi, \operatorname{Hom}_{U(\mathfrak{g})}(M, V)^{N_P^0}_{\mathrm{fs}} \right)$$

where  $\operatorname{Hom}_E(M, E)^{\mathfrak{n}_{P^-}^{\infty}} \subset \operatorname{Hom}_E(M, E)$  is the object in  $\mathcal{O}_{\operatorname{alg}}^{\mathfrak{p}^-}$  defined in [14, § 3] and  $\delta_P$  is the modulus character of  $P(\mathbb{Q}_p)$ .

Fix  $\mathfrak{d}$  an integral weight of  $\mathfrak{z}_{L_P}$  and  $\sigma$  a cuspidal type of  $L_P(\mathbb{Q}_p)$  as in § 3.1.1. Consider

$$V(M,\mathfrak{d},\sigma) := \left( \left( \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M,V)_{\mathrm{fs}}^{N_{P}^{0}} \otimes_{E} (\delta_{\mathfrak{d}} \circ \det_{L_{P}}) \right) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z_{L_{P}}^{0},E) \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}} \\ \cong \operatorname{Hom}_{L_{P}^{0}} \left( \sigma, \left( \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M,V)_{\mathrm{fs}}^{N_{P}^{0}} \otimes_{E} (\delta_{\mathfrak{d}} \circ \det_{L_{P}}) \right) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z_{L_{P}}^{0},E) \right)$$

where  $L_P(\mathbb{Q}_p)$  acts on  $(\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M,V)_{\mathrm{fs}}^{N_P^0} \otimes_E (\delta_{\mathfrak{d}} \circ \det_{L_P})) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z_{L_P}^0, E)$  via the diagonal action with  $L_P(\mathbb{Q}_p)$  acting on  $\mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z_{L_P}^0, E)$  via (3.2). As  $V(M,\mathfrak{d},\sigma)$  is a closed subspace of  $(\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M,V)_{\mathrm{fs}}^{N_P^0} \otimes_E (\delta_{\mathfrak{d}} \circ \det_{L_P})) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z_{L_P}^0, E) \otimes_E \sigma^{\vee}$  (where  $\sigma^{\vee}$  is equipped with the finest locally convex topology), it is also a space of compact type. We equip  $V(M,\mathfrak{d},\sigma)$  with an action of  $\mathcal{Z}_0 \times \Delta_0 \times \mathcal{Z}_\Omega$  similarly as for  $B_{\sigma,\lambda}(V)$  in (3.3):  $\mathcal{Z}_0 \cong Z_{L_P}^0$  acts via the regular action on  $\mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z_{L_P}^0, E), \Delta_0 \cong Z_{L_P}(\mathbb{Q}_p)$  acts via the diagonal action on  $\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M,V)_{\mathrm{fs}}^{N_P^0} \otimes_E$  $(\delta_{\mathfrak{d}} \circ \det_{L_P}) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z_{L_P}^0, E)$ , and the  $\mathcal{Z}_\Omega$ -action comes from the isomorphism (recalling that  $\mathcal{Z}_\Omega \cong \operatorname{End}_{L_P(\mathbb{Q}_p)}(\operatorname{c-ind}_{L_P^0}^{L_P(\mathbb{Q}_p)}\sigma)$ ):

$$V(M,\mathfrak{d},\sigma) \cong \operatorname{Hom}_{L_P(\mathbb{Q}_p)}\left(\operatorname{c-ind}_{L_P^0}^{L_P(\mathbb{Q}_p)}\sigma, \left(\operatorname{Hom}_{U(\mathfrak{g})}(M,V)_{\operatorname{fs}}^{N_P^0} \otimes_E (\delta_{\mathfrak{d}} \circ \det_{L_P})\right) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\operatorname{la}}(Z_{L_P}^0,E)\right).$$

Similarly as in the discussion above Lemma 3.1.2, the  $\Delta_0$ -action is equal to the action induced from  $\mathcal{Z}_{\Omega}$  via  $Z_{L_P}(\mathbb{Q}_p) \to \mathcal{Z}_{\Omega}$ . By considering the action of  $\mathfrak{l}_P$  on  $V(M, \mathfrak{d}, \sigma)$ , we have also

$$V(M,\mathfrak{d},\sigma) \cong \left( \left( \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M,V)_{\mathrm{fs}}^{N_P^0} \otimes_E (\delta_{\mathfrak{d}} \circ \det_{L_P}) \right) \widehat{\otimes}_E \left( \mathcal{C}^{\infty}(Z_{L_P}^0,E) \otimes_E \delta_{\mathfrak{d}}^0 \right) \otimes_E \sigma^{\vee} \right)^{L_P^0} \\ \cong \left( \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M,V)_{\mathrm{fs}}^{N_P^0} \otimes_E (\delta_{\mathfrak{d}} \circ \det_{L_P}) \otimes_E \left( \mathcal{C}^{\infty}(Z_{L_P}^0,E) \otimes_E \delta_{\mathfrak{d}}^0 \right) \otimes_E \sigma^{\vee} \right)^{L_P^0}$$

where  $\mathcal{C}^{\infty}(Z_{L_P}^0, E)$  denotes the space of smooth *E*-valued functions on  $Z_{L_P}^0$ , where  $\mathcal{C}^{\infty}(Z_{L_P}^0, E) \otimes_E \delta_0^0$  embeds into  $\mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z_{L_P}^0, E)$  by  $f \otimes 1 \mapsto [z \mapsto f(z)\delta_0^0(z)]$ , and where the second isomorphism follows from [64, Prop. 1.2] and the fact that  $\mathcal{C}^{\infty}(Z_{L_P}^0, E)$  is topologically isomorphic to a direct limit of finite dimensional *E*-vector spaces (each equipped with the finest locally convex topology).

**Lemma 7.1.2.** Let  $\lambda$  be an integral P-dominant weight of  $T_{\mathscr{I}}$ , we have an isomorphism of vector spaces of compact type (cf. Notation 3.1.5 and the discussion after Lemma 3.1.7)

$$\iota: V(M_P(\lambda), \mathfrak{d}, \sigma) \xrightarrow{\sim} B_{\Omega, \lambda}(V)[\mathfrak{z}_0 = 0]$$

such that for  $(\alpha, \beta) \in \mathbb{Z}_{\Omega} \times \mathbb{Z}_0$  and  $v \in V(M_P(\lambda), \mathfrak{d}, \sigma)$ :

$$\iota((\alpha,\beta)\cdot v) = \delta^0_{\mathfrak{d}}(\beta) \Big( \big(\iota_{(\delta^{\mathrm{unr}}_{\mathfrak{d},\underline{\varpi}})^{-1}}(\alpha),\beta\big)\cdot\iota(v) \Big).$$

*Proof.* We have isomorphisms

$$\operatorname{Hom}_{\operatorname{U}(\mathfrak{g})}(M_P(\lambda), V) \cong \operatorname{Hom}_{\mathfrak{l}_P}(L(\lambda)_P, V^{\mathfrak{n}_P}) \cong (V^{\mathfrak{n}_P} \otimes_E L(\lambda)_P^{\vee})^{\mathfrak{l}_F}$$

which are moreover topological isomorphisms if  $\operatorname{Hom}_{\mathfrak{l}_P}(L(\lambda)_P, V^{\mathfrak{n}_P}) \cong (V^{\mathfrak{n}_P} \otimes_E L(\lambda)_P^{\vee})^{\mathfrak{l}_P}$  is equipped with the induced topology as closed subspace of  $V \otimes_E L(\lambda)_P^{\vee}$ . We deduce then an  $L_P(\mathbb{Q}_p)^+$ -equivariant isomorphism (where  $L_P(\mathbb{Q}_p)^+$  acts on the right hand side by  $(zf)(v) = zf(z^{-1}v)$ ):

$$\operatorname{Hom}_{\operatorname{U}(\mathfrak{g})}(M_P(\lambda), V)^{N_P^0} \cong \operatorname{Hom}_{\mathfrak{l}_P}(L(\lambda)_P, V^{N_P^0}).$$

$$(7.1)$$

This isomorphism induces  $L_P(\mathbb{Q}_p)$ -equivariant isomorphisms by the universal property of the finite slope part functor [40, Prop. 3.2.4 (2)] (for the first) and Lemma 3.1.12 (for the second):

$$\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M_P(\lambda), V)_{\mathrm{fs}}^{N_P^0} \cong \operatorname{Hom}_{\mathfrak{l}_P}(L(\lambda)_P, V_{\mathrm{fs}}^{N_P^0}) \cong J_P(V)_{\lambda}[\mathfrak{z}_{L_P} = 0].$$
(7.2)

From (7.2) we deduce topological isomorphisms

$$\begin{split} V(M_{P}(\lambda),\mathfrak{d},\sigma) \\ &\cong \left( \left( J_{P}(V)_{\lambda} [\mathfrak{z}_{L_{P}} = 0] \otimes_{E} \left( \left( \delta^{\mathrm{unr}}_{\mathfrak{d},\underline{\varpi}} \delta^{0}_{\mathfrak{d},\underline{\varpi}} \right) \circ \det_{L_{P}} \right) \right) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E) \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}} \\ &\cong \left( \left( J_{P}(V)_{\lambda} \otimes_{E} \left( \delta^{\mathrm{unr}}_{\mathfrak{d},\underline{\varpi}} \circ \det_{L_{P}} \right) \right) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E) [\mathfrak{z}_{L_{P}} = 0] \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}} \\ &\cong \mathrm{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \mathrm{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma, \left( J_{P}(V)_{\lambda} \otimes_{E} \left( \delta^{\mathrm{unr}}_{\mathfrak{d},\underline{\varpi}} \circ \det_{L_{P}} \right) \right) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E) [\mathfrak{z}_{L_{P}} = 0] \right) \\ &\cong \mathrm{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \left( (\delta^{\mathrm{unr}}_{\mathfrak{d},\underline{\varpi}})^{-1} \circ \det_{L_{P}} \right) \otimes_{E} \mathrm{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma, J_{P}(V)_{\lambda} \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E) [\mathfrak{z}_{L_{P}} = 0] \right) \\ &\cong \mathrm{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \mathrm{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma, J_{P}(V)_{\lambda} \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E) [\mathfrak{z}_{L_{P}} = 0] \right) \\ &\cong \mathrm{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \mathrm{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma, J_{P}(V)_{\lambda} \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E) [\mathfrak{z}_{L_{P}} = 0] \right) \\ &\cong \mathrm{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \mathrm{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma, J_{P}(V)_{\lambda} \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E) [\mathfrak{z}_{L_{P}} = 0] \right) \\ &\cong \mathrm{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \mathrm{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma, J_{P}(V)_{\lambda} \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E) [\mathfrak{z}_{L_{P}} = 0] \right) \\ &\cong \mathrm{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \mathrm{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma, J_{P}(V)_{\lambda} \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E) [\mathfrak{z}_{L_{P}} = 0] \right) \\ &\cong \mathrm{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \mathrm{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma, J_{P}(V)_{\lambda} \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E) [\mathfrak{z}_{L_{P}} = 0] \right) \\ &\cong \mathrm{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \mathrm{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \sigma, J_{P}(V)_{\lambda} \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E) [\mathfrak{z}_{L_{P}} = 0] \right) \\ &\cong \mathrm{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \mathrm{c-ind}_{L_{P}^{0}}^{L_{P}(\mathbb{Q}_{p})} \right) \\ &\cong \mathrm{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \mathrm{c-ind}_{L_{P}$$

where the second isomorphism uses  $\mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z^0_{L_P}, E) \otimes_E \delta^0_{\mathfrak{d},\underline{\infty}} \cong \mathcal{C}^{\mathbb{Q}_p-\text{la}}(Z^0_{L_P}, E)$  and the fifth uses (3.8). The last part of the statement on the  $\mathcal{Z}_{\Omega} \times \mathcal{Z}_0$ -action follows by the same argument as in the proof of Lemma 3.1.6.

**Lemma 7.1.3.** For  $M \in \mathcal{O}_{alg}^{\mathfrak{p}}$ , we have

$$V(M,\mathfrak{d},\sigma)\cong \bigoplus_{\delta,\chi} V(M,\mathfrak{d},\sigma)[\mathfrak{m}_{\chi}][\mathfrak{m}_{\delta}^{\infty}] = \bigoplus_{\mathfrak{m}\in \operatorname{Spm} \mathcal{Z}_{\Omega},\chi} V(M,\mathfrak{d},\sigma)[\mathfrak{m}_{\chi}][\mathfrak{m}^{\infty}]$$

where  $\delta$  (resp.  $\chi$ ) runs through the smooth characters of  $\Delta_0 \cong Z_{L_P}(\mathbb{Q}_p)$  (resp. through the locally algebraic characters of  $\mathcal{Z}_0 \cong Z_{L_P}^0$  of weight  $\mathfrak{d}$ ), and  $\mathfrak{m}_{\delta} \subset E[\Delta_0]$  (resp.  $\mathfrak{m}_{\chi} \subset E[\mathcal{Z}_0]$ ) is the maximal ideal associated to the character  $\delta$  (resp.  $\chi$ ). Moreover, each term in the direct sums is finite dimensional over E. *Proof.* As  $M \in \mathcal{O}_{alg}^{\mathfrak{p}}$ , there exist finitely many *P*-dominant integral weights  $\lambda_i$  such that  $\oplus_i M_P(\lambda_i) \twoheadrightarrow M$ . Using successively the following facts:

- (1)  $\operatorname{Hom}_{\mathrm{U}(\mathfrak{q})}(-, V)$  is left exact;
- (2) taking  $N_P^0$ -invariant vectors is exact on smooth representations of  $P(\mathbb{Q}_p)$ ;
- (3) taking  $(-)_{\rm fs}$  preserves injectivity (cf. [40, Prop. 3.2.6 (ii)]);
- (4)  $(-)^{L_P^0}$  is left exact,

we deduce an injection:

$$V(M, \mathfrak{d}, \sigma) \hookrightarrow \bigoplus_i V(M_P(\lambda_i), \mathfrak{d}, \sigma).$$

It is sufficient to prove the statement for each  $V(M_P(\lambda_i), \mathfrak{d}, \sigma)$ . Indeed, if this holds, then any vector  $v \in V(M, \mathfrak{d}, \sigma)$  generates, under the action of  $\mathcal{Z}_{\Omega} \times \mathcal{Z}_{0}$ , a finite dimensional *E*-vector space, and from this we easily deduce the decompositions in the lemma for  $V(M, \mathfrak{d}, \sigma)$ . Moreover for each  $\mathfrak{m} \in \operatorname{Spec} \mathcal{Z}_{\Omega}$  and each  $\chi$ , the vector space  $V(M, \mathfrak{d}, \sigma)[\mathfrak{m}_{\chi}][\mathfrak{m}^{\infty}]$  is finite dimensional as it is a subspace of  $\bigoplus_i V(M_P(\lambda_i), \mathfrak{d}, \sigma)[\mathfrak{m}_{\chi}][\mathfrak{m}^{\infty}]$ . However, the statement in the lemma for  $V(M_P(\lambda_i), \mathfrak{d}, \sigma)$  follows from Lemma 7.1.2 and Lemma 3.1.4.

**Lemma 7.1.4.** Let  $\mathfrak{m} \subset \mathcal{Z}_{\Omega}$  be a maximal ideal and  $\chi$  a locally algebraic character of  $\mathcal{Z}_{0}$  of weight  $\mathfrak{d}$ . Let V be a continuous representation of  $G_{p}$  on a Banach space over E such that  $V^{\vee}$ is a finitely generated projective  $\mathcal{O}_{E}[[K_{p}]][1/p]$ -module where  $K_{p} := G(\mathbb{Z}_{p})$ . Denote by  $V^{\mathrm{an}} \subseteq V$ the subspace of locally analytic vectors ([42, Def. 3.5.3]), which is an admissible locally analytic representation of  $G_{p}$  on a E-vector space of compact type ([42, Prop. 6.2.4]). Then the functor  $V^{\mathrm{an}}(-,\mathfrak{d},\sigma)[\mathfrak{m}_{\chi}][\mathfrak{m}^{\infty}]$  is exact on  $\mathcal{O}_{\mathrm{alg}}^{\mathfrak{p}}$ .

Proof. Let  $\pi_{\mathfrak{m}}$  be the smooth  $L_{P}(\mathbb{Q}_{p})$ -representation associated to  $\mathfrak{m}$  and  $\omega$  its central character. ter. Let  $\mathfrak{m}_{\omega} \subset E[\Delta_{0}]$  be the maximal ideal associated to  $\omega$ . Then  $V^{\mathrm{an}}(M,\mathfrak{d},\sigma)[\mathfrak{m}_{\chi}][\mathfrak{m}^{\infty}] = V^{\mathrm{an}}(M,\mathfrak{d},\sigma)[\mathfrak{m}_{\chi}][\mathfrak{m}_{\chi}]$  is a direct summand of  $V^{\mathrm{an}}(M,\mathfrak{d},\sigma)[\mathfrak{m}_{\omega}^{\infty}][\mathfrak{m}_{\chi}]$ , and it is enough to prove the statement with  $[\mathfrak{m}^{\infty}]$  replaced by  $[\mathfrak{m}_{\omega}^{\infty}]$ . Let

$$\delta := \left( \left( (\chi(\delta_{\mathfrak{d}}^{0})^{-1})_{\underline{\varpi}} (\delta_{\mathfrak{d},\underline{\varpi}}^{\mathrm{unr}})^{-1} \right) \circ \det_{L_{P}} \right) \omega = \left( (\chi\delta_{\mathfrak{d}}^{-1}) \circ \det_{L_{P}} \right) \omega$$

which is a smooth character of  $Z_{L_P}(\mathbb{Q}_p)$ . Let  $\mathfrak{m}_{\delta} \subset E[Z_{L_P}(\mathbb{Q}_p)]$  (resp.  $\mathfrak{m}_{\delta,+} \subset E[Z_{L_P}(\mathbb{Q}_p)^+]$ ) be the maximal ideal associated to  $\delta$ . For  $M \in \mathcal{O}_{alg}^{\mathfrak{p}}$ , we have (by unwinding the  $\Delta_0$ -action, and noting that  $L_P(\mathbb{Q}_p)$  acts on  $\chi$  via  $\chi_{\varpi}^{-1} \circ \det_{L_P}$ , cf. (3.2))

$$V^{\mathrm{an}}(M,\mathfrak{d},\sigma)[\mathfrak{m}_{\chi}][\mathfrak{m}_{\omega}^{\infty}] \cong \left(\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M,V^{\mathrm{an}})_{\mathrm{fs}}^{N_{P}^{0}}[\mathfrak{m}_{\delta}^{\infty}] \otimes_{E} (\delta_{\mathfrak{d}} \circ \det_{L_{P}}) \otimes_{E} \chi \otimes_{E} \sigma^{\vee}\right)^{L_{P}^{0}} \\ \cong \left(\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M,V^{\mathrm{an}})^{N_{P}^{0}}[\mathfrak{m}_{\delta,+}^{\infty}] \otimes_{E} (\delta_{\mathfrak{d}} \circ \det_{L_{P}}) \otimes_{E} \chi \otimes_{E} \sigma^{\vee}\right)^{L_{P}^{0}} (7.3)$$

where the second isomorphism follows from [40, Prop. 3.2.11] and [40, Lemma 3.2.8]. In particular, the right hand side of (7.3) is finite dimensional by Lemma 7.1.3. By [19, Lemma 5.2.5], the functor  $\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(-, V^{\mathrm{an}})$  is exact, hence so is the functor  $\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(-, V^{\mathrm{an}})^{N_P^0}$  as  $\pi \mapsto \pi^{N_P^0}$  is exact on smooth representations  $\pi$  of  $P(\mathbb{Q}_p)$ . We deduce that, for an exact sequence  $0 \to M_1 \to M_2 \to M_3 \to 0$  in  $\mathcal{O}_{\mathrm{alg}}^{\mathfrak{p}}$ , the following sequence is exact:

$$0 \to \left( \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M_{3}, V^{\mathrm{an}})^{N_{P}^{0}} \otimes_{E} (\delta_{\mathfrak{d}} \circ \det_{L_{P}}) \otimes_{E} \chi \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}} \\ \to \left( \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M_{2}, V^{\mathrm{an}})^{N_{P}^{0}} \otimes_{E} (\delta_{\mathfrak{d}} \circ \det_{L_{P}}) \otimes_{E} \chi \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}} \\ \to \left( \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M_{1}, V^{\mathrm{an}})^{N_{P}^{0}} \otimes_{E} (\delta_{\mathfrak{d}} \circ \det_{L_{P}}) \otimes_{E} \chi \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}} \to 0.$$
(7.4)

By (7.3) it is enough to show that the sequence (7.4) stays exact after applying  $[\mathfrak{m}_{\delta+}^{\infty}]$ .

We now explain how to prove this by a generalization of the arguments in the last paragraph of the proof of [20, Prop. 4.1]. We let H be as in *loc. cit.* and replace  $N_P^0$  by  $H \cap N_P(\mathbb{Q}_p)$  (which won't cause any problem). Using [40, Prop. 3.3.2], we choose finitely many  $z_i \in Z_{L_P}(\mathbb{Q}_p)^+$  such that  $z_i N_P^0 z_i^{-1} \subset (N_P^0)^p$  for  $N_P^0 = N_P(\mathbb{Q}_p) \cap H$ , and such that  $Z_{L_P}^0$  and the  $z_i$  generate  $Z_{L_P}(\mathbb{Q}_p)$ as group. For  $M = M_P(\lambda) \in \mathcal{O}_{alg}^p$ , we have by (7.1) a topological isomorphism

$$\left( \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M, V^{\mathrm{an}})^{N_{P}^{0}} \otimes_{E} (\delta_{\mathfrak{d}} \circ \det_{L_{P}}) \otimes_{E} \chi \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}}$$
$$\cong \left( (V^{\mathrm{an}})^{N_{P}^{0}} \otimes_{E} L(\lambda)_{P}^{\vee} \otimes_{E} (\delta_{\mathfrak{d}} \circ \det_{L_{P}}) \otimes_{E} \chi \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}}.$$
(7.5)

By an easy generalization of [18, Lemma 5.3] (see also the proof of Proposition 3.1.15), we can write the right hand side (hence the left hand side) of (7.5) as an increasing union over j of BH-subspaces  $V_j$  such that each  $z_i$  preserves  $V_j$  and acts on (the underlying Banach space)  $V_j$  via a compact operator. The same holds for a general  $M \in \mathcal{O}_{alg}^{\mathfrak{p}}$  as  $(\operatorname{Hom}_{U(\mathfrak{g})}(M, V^{\operatorname{an}})^{N_P^0} \otimes_E (\delta_{\mathfrak{d}} \circ \det_{L_P}) \otimes_E \chi \otimes_E \sigma^{\vee})^{L_P^0}$  is a closed subspace of a finite direct sum  $\bigoplus_i (\operatorname{Hom}_{U(\mathfrak{g})}(M_P(\lambda_i), V^{\operatorname{an}})^{N_P^0} \otimes_E (\delta_{\mathfrak{d}} \circ \det_{L_P}) \otimes_E \chi \otimes_E \sigma^{\vee})^{L_P^0}$  for some integral P-dominant weights  $\lambda_i$ . We can now apply the argument in the last paragraph of the proof of [20, Prop. 4.1] with the BH-subspaces  $\Pi_h$  of *loc. cit.* replaced by the BH-subspaces  $V_j$  (for general M) discussed above to conclude.

# 7.1.2 Cycles on patched Bernstein eigenvarieties

We construct certain cycles in the completed local rings (at some specific points) of patched Bernstein eigenvarieties.

We assume that we are in the setting of § 3.3 and we use without comment the notation and constructions of *loc. cit.* We fix  $\lambda$  an integral *P*-dominant weight,  $\Omega$  a cuspidal Bernstein component of  $L_P(\mathbb{Q}_p)$  and  $\overline{\rho}$  a  $U^p$ -modular continuous representation of Gal<sub>F</sub> over  $k_E$  (where  $U^p$  is a "prime-to-*p* level"). We have the associated patched Bernstein eigenvariety  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$ , see (3.29).

Let  $\mathfrak{d}$  be an integral weight of  $\mathfrak{z}_{L_P}$  and put  $\mu := \lambda + \mathfrak{d} \circ \det_{L_P}$ . Let  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})_{\mathfrak{d}}$  be the fibre of  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  at  $\mathfrak{d}$  via  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho}) \to \widehat{\mathcal{Z}}_0 \to \mathfrak{z}_0^{\vee} \cong \mathfrak{z}_{L_P}^{\vee}$ . Note that using [28, Lemma 6.2.5] and [28, Lemma. 6.2.10], we can deduce from Corollary 3.3.3 (1), (2) and Proposition 3.3.2 that  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})_{\mathfrak{d}}$  is equidimensional of dimension

$$g + |S|n^2 + \sum_{v \in S_p} \left( [F_{\widetilde{v}} : \mathbb{Q}_p] \frac{n(n-1)}{2} \right).$$

Let  $\mathcal{M}_{\mu,\lambda} := \mathcal{M}_{\Omega,\lambda}^{\infty} \otimes_{\mathcal{O}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})}} \mathcal{O}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})_{\vartheta}}$ . Using (3.28) the vector space of compact type  $\Gamma(\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})_{\vartheta}, \mathcal{M}_{\mu,\lambda})^{\vee}$  is topologically isomorphic to the following vector spaces of compact type (which are of compact type by an obvious generalization to the patched case of the discussion

above Lemma 7.1.1):

$$\begin{split} \left(J_{P}(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\lambda}\widehat{\otimes}_{E}\mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z_{L_{P}}^{0},E)\otimes_{E}\sigma^{\vee}\right)^{L_{P}^{0}}[\mathfrak{z}_{0}=\mathfrak{d}] \\ &\cong \left(J_{P}(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\lambda}\widehat{\otimes}_{E}\mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z_{L_{P}}^{0},E)[\mathfrak{z}_{L_{P}}=\mathfrak{d}]\otimes_{E}\sigma^{\vee}\right)^{L_{P}^{0}} \\ &\cong \left(J_{P}(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\lambda}[\mathfrak{z}_{L_{P}}=\mathfrak{d}\circ\det_{L_{P}}]\widehat{\otimes}_{E}\mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z_{L_{P}}^{0},E)[\mathfrak{z}_{L_{P}}=\mathfrak{d}]\otimes_{E}\sigma^{\vee}\right)^{L_{P}^{0}} \\ &\cong \left(\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M_{P}(\mu),\Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\mathrm{fs}}^{N_{P}^{0}}\otimes_{E}(\delta_{\mathfrak{d}}\circ\det_{L_{P}})\widehat{\otimes}_{E}\mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z_{L_{P}}^{0},E)[\mathfrak{z}_{L_{P}}=\mathfrak{d}]\otimes_{E}\sigma^{\vee}\right)^{L_{P}^{0}} \\ &\cong \left(\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M_{P}(\mu),\Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\mathrm{fs}}^{N_{P}^{0}}\otimes_{E}(\delta_{\mathfrak{d}}\circ\det_{L_{P}})\widehat{\otimes}_{E}\mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z_{L_{P}}^{0},E)(\mathfrak{z}_{L_{P}}=\mathfrak{d})\otimes_{E}\sigma^{\vee}\right)^{L_{P}^{0}} \end{split}$$

where the second and fourth isomorphism follow by considering the action of  $l_P$  (see also (3.6)), the third isomorphism follows from (7.2) and the natural isomorphism

$$J_P(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\mu}[\mathfrak{z}_{L_P}=0] \cong J_P(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\lambda}[\mathfrak{z}_{L_P}=\mathfrak{d}\circ\mathrm{det}_{L_P}]\otimes_E(\delta_{\mathfrak{d}}^{-1}\circ\mathrm{det}_{L_P}).$$
(7.6)

The natural surjection  $M_P(\mu) \rightarrow L(\mu)$  induces an injection

$$\left( \left( \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(L(\mu), \Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\mathrm{fs}}^{N_{P}^{0}} \otimes_{E} (\delta_{\mathfrak{d}} \circ \det_{L_{P}}) \right) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z_{L_{P}}^{0}, E) \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}}$$

$$\longleftrightarrow \left( \left( \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(M_{P}(\mu), \Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\mathrm{fs}}^{N_{P}^{0}} \otimes_{E} (\delta_{\mathfrak{d}} \circ \det_{L_{P}}) \right) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z_{L_{P}}^{0}, E) \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}}.$$
(7.7)

As in the discussion above Lemma 7.1.1, we equip the right hand side of (7.7) with the topology induced by the one on  $\Pi_{\infty}^{R_{\infty}-\mathrm{an}} \otimes_E (\delta_{\mathfrak{d}} \circ \det_{L_P}) \widehat{\otimes}_E \mathcal{C}^{\mathbb{Q}_p-\mathrm{la}}(Z_{L_P}^0, E) \otimes_E \sigma^{\vee}$ , and the left hand side of (7.7) with the topology induced by the one on the right hand side, which identifies it with a closed subspace (in particular, all spaces are of compact type). It is also clear that the morphism in (7.7) is equivariant under the action of  $R_{\infty} \times \mathcal{Z}_0 \times \mathcal{Z}_{\Omega}$ . In particular the left hand side of (7.7) is preserved by  $\mathcal{O}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\bar{p})_{\mathfrak{d}}}$ . Let  $\mathcal{L}_{\mu,\lambda}$  be the quotient of the  $\mathcal{O}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\bar{p})_{\mathfrak{d}}}$ -module  $\mathcal{M}_{\mu,\lambda}$  such that

$$\Gamma\left(\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})_{\mathfrak{d}}, \mathcal{L}_{\mu,\lambda}\right)^{\vee} \\ \cong \left(\left(\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(L(\mu), \Pi^{R_{\infty}-\mathrm{an}}_{\infty})^{N^{0}_{P}}_{\mathrm{fs}} \otimes_{E} (\delta_{\mathfrak{d}} \circ \det_{L_{P}})\right) \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}}, E) \otimes_{E} \sigma^{\vee}\right)^{L^{0}_{P}}.$$
(7.8)

The  $\mathcal{O}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})_{\mathfrak{d}}}$ -module  $\mathcal{L}_{\mu,\lambda}$  is finitely generated, and its schematic support defines a (possibly empty) Zariski-closed rigid subspace  $\mathfrak{Z}_{\mu,\lambda}^{0}$  of  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})_{\mathfrak{d}}$ . We denote by  $\mathfrak{Z}_{\mu,\lambda} \subseteq \mathfrak{Z}_{\mu,\lambda}^{0,\mathrm{red}}$  the union of its irreducible components of dimension dim  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})_{\mathfrak{d}}$ , which is still Zariski-closed in  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})_{\mathfrak{d}}$ .

Next we move to a similar discussion for the completion of the patched Bernstein eigenvarieties at some specific points. Let  $x := (x^p, x_p, \underline{x}, \chi_x) \in (\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}} \times (\operatorname{Spf} R^{\Box}_{\overline{\rho}_p})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_0$ . We write  $y := (x^p, x_p) \in (\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \cong (\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}} \times (\operatorname{Spf} R^{\Box}_{\overline{\rho}_p})^{\operatorname{rig}}$ . Let  $\mathfrak{m}_y, \mathfrak{m}_{\underline{\chi}}, \mathfrak{m}_{\underline{\chi}x}$  be the associated maximal ideals of  $R_{\infty}[1/p], \mathcal{Z}_{\Omega}, E[\mathcal{Z}_0]$  respectively. Let  $\mathfrak{d}_x := \operatorname{wt}(\chi_x)$  and  $\lambda^x := \lambda + \mathfrak{d}_x \circ \operatorname{det}_{L_P}$ . By definition, we have an isomorphism of finitely generated  $\widehat{\mathcal{O}}_{\mathcal{E}^\infty_{\Omega,\lambda}(\overline{\rho}),x}$ -modules (for such sheaves of modules, we identify them with their global sections with no ambiguity):

$$\mathcal{M}^{\infty}_{\Omega,\lambda} \otimes_{\mathcal{O}_{\mathcal{E}^{\infty}_{\Omega,\lambda}(\bar{\rho})}} \widehat{\mathcal{O}}_{\mathcal{E}^{\infty}_{\Omega,\lambda}(\bar{\rho}),x} \cong \left( \left( J_P(\Pi^{R_{\infty}-\mathrm{an}}_{\infty})_{\lambda}[\mathfrak{m}^{\infty}_{y}] \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z^{0}_{L_{P}},E)[\mathfrak{m}^{\infty}_{\chi_{x}}] \otimes_{E} \sigma^{\vee} \right)^{L^{0}_{P}}[\mathfrak{m}^{\infty}_{\underline{x}}] \right)^{\vee}$$

and an isomorphism of finitely generated  $\widehat{\mathcal{O}}_{\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})\mathfrak{d}_{x},x}$ -modules:

$$\mathcal{M}_{\Omega,\lambda}^{\infty} \otimes_{\mathcal{O}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})}} \widehat{\mathcal{O}}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})\mathfrak{d}_{x},x} \\ \cong \left( \left( J_{P}(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\lambda}[\mathfrak{m}_{y}^{\infty}] \widehat{\otimes}_{E} \mathcal{C}^{\mathbb{Q}_{p}-\mathrm{la}}(Z_{L_{P}}^{0},E)[\mathfrak{z}_{L_{P}}=\mathfrak{d}_{x}][\mathfrak{m}_{\chi_{x}}^{\infty}] \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}}[\mathfrak{m}_{\underline{x}}^{\infty}] \right)^{\vee} \\ \cong \left( \left( J_{P}(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\lambda}[\mathfrak{m}_{y}^{\infty}] \otimes_{E} \chi_{x} \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}}[\mathfrak{m}_{\underline{x}}^{\infty}] \right)^{\vee}$$
(7.9)

which are both non-zero if and only if  $x \in \mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$ . Assume in the sequel  $\mathfrak{d}_x \in \mathbb{Z}^{[F^+:\mathbb{Q}]}$ . By modifying the Bernstein centre  $\Omega$  and using Proposition 3.2.3 (which obviously generalizes to the patched case), we can and do assume that the character  $\chi_x$  is algebraic. We have

$$\begin{pmatrix} J_P(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\lambda}[\mathfrak{m}_{y}^{\infty}] \otimes_{E} \chi_{x} \otimes_{E} \sigma^{\vee} \end{pmatrix}^{L_{P}^{0}}[\mathfrak{m}_{\underline{x}}^{\infty}] \\ \cong \begin{pmatrix} J_P(\Pi_{\infty}^{R_{\infty}-\mathrm{an}})_{\lambda}[\mathfrak{m}_{y}^{\infty}][\mathfrak{z}_{L_{P}}=\mathfrak{d}_{x}\circ\det_{L_{P}}] \otimes_{E} \chi_{x} \otimes_{E} \sigma^{\vee} \end{pmatrix}^{L_{P}^{0}}[\mathfrak{m}_{\underline{x}}^{\infty}] \\ \cong \begin{pmatrix} \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})} \left( M_{P}(\lambda^{x}), \Pi_{\infty}^{R_{\infty}-\mathrm{an}} \right)_{\mathrm{fs}}^{N_{P}^{0}}[\mathfrak{m}_{y}^{\infty}] \otimes_{E} \left( \delta_{\mathfrak{d}_{x}}\circ\det_{L_{P}} \right) \otimes_{E} \chi_{x} \otimes_{E} \sigma^{\vee} \end{pmatrix}^{L_{P}^{0}}[\mathfrak{m}_{\underline{x}}^{\infty}]$$
(7.10)

where the second isomorphism follows from (7.2) and (7.6) (be careful with the action of  $\Delta_0$ ).

Lemma 7.1.5. With the above notation, the functor

$$M \longmapsto \left( \operatorname{Hom}_{U(\mathfrak{g})} \left( M, \Pi_{\infty}^{R_{\infty} - \operatorname{an}} \right)_{\mathrm{fs}}^{N_{P}^{0}} [\mathfrak{m}_{y}^{\infty}] \otimes_{E} \left( \delta_{\mathfrak{d}_{x}} \circ \operatorname{det}_{L_{P}} \right) \otimes_{E} \chi_{x} \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}} [\mathfrak{m}_{\underline{x}}^{\infty}]$$

on the category  $\mathcal{O}_{alg}^{\mathfrak{p}}$  is exact.

Proof. For each  $t \in \mathbb{Z}_{\geq 1}$ , let  $\mathcal{I}_t$  be the kernel of the composition  $S_{\infty}[1/p] \to R_{\infty}[1/p] \twoheadrightarrow R_{\infty}[1/p]/\mathfrak{m}_y^t$ . Then  $\Pi_{\infty}[\mathcal{I}_t]^{\vee}$  is a finite projective  $\mathcal{O}_E[[K_p]][1/p]$ -module (see § 3.3) and we have  $\Pi_{\infty}[\mathcal{I}_t]^{\mathrm{an}} = \Pi_{\infty}^{R_{\infty}-\mathrm{an}}[\mathcal{I}_t]$  (which easily follows from [17, (3.2)]). By Lemma 7.1.4 and Lemma 7.1.3 applied to  $V := \Pi_{\infty}[\mathcal{I}_t]$ , the functor  $M \mapsto (\operatorname{Hom}_{U(\mathfrak{g})}(M, V)_{\mathrm{fs}}^{N_P^0} \otimes_E (\delta_{\mathfrak{d}_x} \circ \det_{L_P}) \otimes_E \chi_x \otimes_E \sigma^{\vee})^{L_P^0}[\mathfrak{m}_x^\infty]$  is exact. We then argue as at the end of the proof of [18, Thm. 5.5].

Define

$$\mathcal{M}_{\lambda^{x},\lambda,y} := \mathcal{M}_{\Omega,\lambda}^{\infty} \otimes_{\mathcal{O}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})}} \widehat{\mathcal{O}}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})\mathfrak{d}_{x},x} \cong \mathcal{M}_{\lambda^{x},\lambda} \otimes_{\mathcal{O}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})\mathfrak{d}_{x}}} \widehat{\mathcal{O}}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})\mathfrak{d}_{x},x}$$
(7.11)

which by (7.9) is isomorphic to the strong dual of the space in (7.10). Let  $L(\mu)$  be an irreducible constituent of  $M_P(\lambda^x)$  and define  $\mathcal{N}_{\mu,\lambda,y}$  as the dual of the following space of compact type:

$$\left(\operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}\left(L(\mu), \Pi_{\infty}^{R_{\infty}-\mathrm{an}}\right)_{\mathrm{fs}}^{N_{P}^{0}}[\mathfrak{m}_{y}^{\infty}] \otimes_{E} \left(\delta_{\mathfrak{d}_{x}} \circ \det_{L_{P}}\right) \otimes_{E} \chi_{x} \otimes_{E} \sigma^{\vee}\right)^{L_{P}^{0}}[\mathfrak{m}_{\underline{x}}^{\infty}].$$
(7.12)

By Lemma 7.1.5,  $\mathcal{N}_{\mu,\lambda,y}$  is a subquotient of the  $\widehat{\mathcal{O}}_{\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})\mathfrak{d}_x,x}$ -module  $\mathcal{M}_{\lambda^x,\lambda,y}$ . One can show moreover that  $\mathcal{N}_{\mu,\lambda,y}$  is preserved by the action of  $R_{\infty} \times \mathcal{Z}_0 \times \mathcal{Z}_{\Omega}$ , hence is also a finitely generated  $\widehat{\mathcal{O}}_{\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})\mathfrak{d}_x,x}$ -module.

**Lemma 7.1.6.** We have  $\mathcal{N}_{\mu,\lambda,y} \neq 0$  if and only if

$$\operatorname{Hom}_{G_p}\left(\mathcal{F}_{P^{-}(\mathbb{Q}_p)}^{G_p}\left(L^{-}(-\mu), \pi_{\underline{x}}\otimes_E\left(\left(\delta_{\mathfrak{d}_x}^{-1}\chi_{x,\underline{\varpi}}\right)\circ\det_{L_P}\right)\otimes_E\delta_P^{-1}\right), \Pi_{\infty}^{R_{\infty}-\operatorname{an}}[\mathfrak{m}_y]\right)\neq 0.$$

*Proof.* By definition  $\mathcal{N}_{\mu,\lambda,y} \neq 0$  if and only if

$$\left(\operatorname{Hom}_{\operatorname{U}(\mathfrak{g})}\left(L(\mu), \Pi_{\infty}^{R_{\infty}-\operatorname{an}}\right)_{\operatorname{fs}}^{N_{P}^{0}}[\mathfrak{m}_{y}] \otimes_{E} \left(\delta_{\mathfrak{d}_{x}} \circ \operatorname{det}_{L_{P}}\right) \otimes_{E} \chi_{x} \otimes_{E} \sigma^{\vee}\right)^{L_{P}^{0}}[\mathfrak{m}_{\underline{x}}] \neq 0.$$

Using Lemma 3.1.2 (applied to  $M = \mathcal{Z}_{\Omega}/\mathfrak{m}_{\underline{x}}$ ) and Lemma 7.1.1, the above vector space is isomorphic to (be careful with the  $L_P(\mathbb{Q}_p)$ -action on the factor  $\chi_x$ ):

$$\operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \pi_{\underline{x}}, \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})} \left( L(\mu), \Pi_{\infty}^{R_{\infty}-\operatorname{an}}[\mathfrak{m}_{y}] \right)_{\mathrm{fs}}^{N_{P}^{0}} \otimes_{E} \left( \delta_{\mathfrak{d}_{x}} \circ \operatorname{det}_{L_{P}} \right) \otimes_{E} \left( \chi_{x,\underline{\varpi}} \right)^{-1} \circ \operatorname{det}_{L_{P}} \right)$$

$$\cong \operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \pi_{\underline{x}} \otimes_{E} \left( \left( \delta_{\mathfrak{d}_{x}}(\chi_{x,\underline{\varpi}})^{-1} \right) \circ \operatorname{det}_{L_{P}} \right), \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})} \left( L(\mu), \Pi_{\infty}^{R_{\infty}-\operatorname{an}}[\mathfrak{m}_{y}] \right)_{\mathrm{fs}}^{N_{P}^{0}} \right)$$

$$\cong \operatorname{Hom}_{G_{p}} \left( \mathcal{F}_{P^{-}(\mathbb{Q}_{p})}^{G_{p}} \left( L^{-}(-\mu), \pi_{\underline{x}} \otimes_{E} \left( \left( \delta_{\mathfrak{d}_{x}}^{-1} \chi_{x,\underline{\varpi}} \right) \circ \operatorname{det}_{L_{P}} \right) \otimes_{E} \delta_{P}^{-1} \right), \Pi_{\infty}^{R_{\infty}-\operatorname{an}}[\mathfrak{m}_{y}] \right).$$

The lemma follows.

(

It follows from Theorem 3.3.9 and Proposition 6.4.5 (2) that the natural morphism  $\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y} \to \widehat{\mathcal{O}}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})\mathfrak{d}_{x},x}$  is surjective. Consequently  $\mathcal{N}_{\mu,\lambda,y}$  is a finitely generated  $\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y}$ -module. Let  $\lambda'$  be another integral *P*-dominant weight and  $\mathfrak{d}$  an integral weight of  $\mathfrak{z}_{L_{P}}$  such that  $L(\mu)$  is an irreducible constituent of  $M_{P}(\lambda' + \mathfrak{d} \circ \det_{L_{P}})$ . Let  $\underline{x'} \in \operatorname{Spec} \mathcal{Z}_{\Omega}$  such that  $\pi_{\underline{x'}} \cong \pi_{\underline{x}} \otimes_{E} (\delta_{\mathfrak{d}}^{\operatorname{unr}} \circ \det_{L_{P}})$  and  $\chi' := \delta_{\mathfrak{d}}^{0}$  (recall  $\chi_{x} = \delta_{\mathfrak{d}_{x}}^{0}$ ). Let  $x' := (y, \underline{x'}, \chi') \in (\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$ , hence  $\lambda^{x'} = \lambda' + (\operatorname{wt}(\chi') \circ \det_{L_{P}}) = \lambda' + \mathfrak{d} \circ \det_{L_{P}}$ . Using similar arguments as in the proof of Lemma 3.1.6 (especially in the last paragraph), we have a topological isomorphism which commutes with  $R_{\infty}$ :

$$\begin{pmatrix} \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})} \left( L(\mu), \Pi_{\infty}^{R_{\infty}-\mathrm{an}} \right)_{\mathrm{fs}}^{N_{P}^{0}}[\mathfrak{m}_{y}^{\infty}] \otimes_{E} \left( \delta_{\mathfrak{d}_{x}} \circ \det_{L_{P}} \right) \otimes_{E} \chi_{x} \otimes_{E} \sigma^{\vee} \end{pmatrix}^{L_{P}^{0}}[\mathfrak{m}_{\underline{x}}^{\infty}] \\ \xrightarrow{\sim} \left( \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})} \left( L(\mu), \Pi_{\infty}^{R_{\infty}-\mathrm{an}} \right)_{\mathrm{fs}}^{N_{P}^{0}}[\mathfrak{m}_{y}^{\infty}] \otimes_{E} \left( \delta_{\mathfrak{d}} \circ \det_{L_{P}} \right) \otimes_{E} \chi' \otimes_{E} \sigma^{\vee} \right)^{L_{P}^{0}}[\mathfrak{m}_{\underline{x}'}^{\infty}].$$

Note however that the  $Z_{\Omega} \times Z_0$ -actions on both sides differ by a twist.<sup>6</sup> We then obtain (using Lemma 7.1.5 for the second part):

**Lemma 7.1.7.** We have an isomorphism of  $\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y}$ -modules  $\mathcal{N}_{\mu,\lambda,y} \cong \mathcal{N}_{\mu,\lambda',y}$ . Consequently, if  $\mathcal{N}_{\mu,\lambda,y} \neq 0$ , then  $\mathcal{M}_{\lambda^{x'},\lambda',y} \neq 0$ , hence  $x' = (y, \underline{x}', \chi') \in \mathcal{E}_{\Omega,\lambda'}^{\infty}(\overline{\rho})$ .

Keep the notation of Lemma 7.1.7 and suppose moreover  $\lambda^{x'} = \lambda^x$ , then by similar arguments (with  $L(\mu)$  replaced by  $M_P(\lambda^x)$ ), we obtain an isomorphism of  $\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y}$ -modules  $\mathcal{M}_{\lambda^x,\lambda,y} \cong \mathcal{M}_{\lambda^x,\lambda',y}$ . We will use the notation  $\mathcal{N}_{\mu,y} := \mathcal{N}_{\mu,\lambda,y}$ ,  $\mathcal{M}_{\lambda^x,y} := \mathcal{M}_{\lambda^x,\lambda,y}$  in the sequel when we are only concerned with the  $R_{\infty}$ -action. As  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})_{\mathfrak{d}_x}$  is equidimensional of dimension  $g + |S|n^2 + \sum_{v \in S_p} ([F_{\widetilde{v}} : \mathbb{Q}_p] \frac{n(n-1)}{2})$ , so is  $\widehat{\mathcal{O}}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})\mathfrak{d}_x,x}$  (see for example the discussion on page 309 of [19]). For a finitely generated  $\widehat{\mathcal{O}}_{\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})\mathfrak{d}_x,x}$ -module  $\mathcal{N}$  we set (where  $Z^i(-)$  is defined in the same way as in § 6.5):

$$[\mathcal{N}] := \sum_{\mathfrak{Z}} m(\mathfrak{Z}, \mathcal{N})[\mathfrak{Z}] \in Z^{0}(\operatorname{Spec} \widehat{\mathcal{O}}_{\mathcal{E}^{\infty}_{\Omega, \lambda}(\overline{\rho})_{\mathfrak{d}_{x}}, x}) \subset Z^{[F^{+}:\mathbb{Q}]\frac{n(n+1)}{2}}(\operatorname{Spec} \widehat{\mathcal{O}}_{\mathfrak{X}_{\infty}, y})$$

<sup>&</sup>lt;sup>6</sup>As we will not use this result, we leave the curious readers work out the precise twist.

where  $\mathfrak{Z} = \operatorname{Spec} \mathcal{O}_{\eta_3}$  runs through the irreducible components of  $\operatorname{Spec} \widehat{\mathcal{O}}_{\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})\mathfrak{d}_x,x}$  ( $\eta_3$  is the associated generic point and  $\mathcal{O}_{\eta_3}$  the localization at  $\eta_3$ ), and  $m(\mathfrak{Z},\mathcal{N})$  is the length of the  $\mathcal{O}_{\eta_3}$ -module  $\mathcal{N}_{\eta_3}$ . We have by Lemma 7.1.5:

$$[\mathcal{M}_{\lambda^x,y}] = \sum_{\mu} b_{\lambda^x,\mu} [\mathcal{N}_{\mu,y}]$$
(7.13)

where  $b_{\lambda^x,\mu}$  denotes the multiplicity of  $L(\mu)$  as an irreducible constituent in  $M_P(\lambda^x)$ . The following lemma is straightforward (comparing (7.8) with (7.12)):

**Lemma 7.1.8.** We have  $[\mathcal{N}_{\mu,y}] \neq 0$  (resp.  $\mathcal{N}_{\mu,y} \neq 0$ ) if and only if  $x \in \mathfrak{Z}_{\mu,\lambda}$  (resp.  $x \in \mathfrak{Z}_{\mu,\lambda}^0$ ).

Assume that  $\rho_{x,\tilde{v}}$  is generic potentially crystalline with distinct Hodge-Tate weights  $\mathbf{h}_{\tilde{v}}$  for all v|p, and that  $x^p$  lies in the smooth locus of  $\mathfrak{X}^p_{\infty} := (\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}}$ . By Corollary 6.4.7,  $\mathfrak{X}^p_{\infty} \times X_{\Omega,\mathbf{h}}(\overline{\rho}_p)$  is irreducible at the point x (see Theorem 3.3.9 for the notation). We deduce:

**Corollary 7.1.9.** The embedding in Theorem 3.3.9 induces a local isomorphism at the point x, and  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  is irreducible at x.

Let  $\mathbf{h} = (\mathbf{h}_{\widetilde{v}})$  and  $\mu = (\mu_{\widetilde{v}}) = (\mu_{\widetilde{v},i,\tau})_{\substack{i=1,\dots,n\\v\in S_p,\tau\in\Sigma_{\widetilde{v}}}}$  with  $\mu_{\widetilde{v},i,\tau} = h_{\widetilde{v},i,\tau} + i - 1$ . By Proposition 3.3.4, there exists  $w_x = (w_{x,\widetilde{v}}) \in \mathscr{W}_{\min}^P \cong \prod_{v\in S_p} \mathscr{W}_{\min,F_{\widetilde{v}}}^{P_{\widetilde{v}}}$  such that  $\lambda^x = w_x \cdot \mu$ . For each  $v \in S_p$ and  $w_{\widetilde{v}} \in \mathscr{W}_{F_{\widetilde{v}}}$ , let  $\mathfrak{C}_{\widetilde{v},w_{\widetilde{v}}} \in Z^{[F_{\widetilde{v}}:\mathbb{Q}_p]\frac{n(n+1)}{2}}$  (Spec  $\widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho_{\widetilde{v}}},\rho_{\widetilde{v}}}}$ ) be the cycle defined in (6.19) (applied to  $L = F_{\widetilde{v}}, \rho = \rho_{\widetilde{v}}$  and so on). For each  $w = (w_{\widetilde{v}}) \in \mathscr{W}_F$ , we put

$$\mathfrak{C}_{w,F} := [\operatorname{Spec} \widehat{\mathcal{O}}_{\mathfrak{X}^{p}_{\infty}, x^{p}}] \times \prod_{v \in S_{p}} \mathfrak{C}_{\widetilde{v}, w_{\widetilde{v}}} \in Z^{[F^{+}:\mathbb{Q}]\frac{n(n+1)}{2}}(\operatorname{Spec} \widehat{\mathcal{O}}_{\mathfrak{X}_{\infty}, y}).$$
(7.14)

Similarly, let  $\mathfrak{Z}_{w,F} := [\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty}^{p},x^{p}}] \times \prod_{v \in S_{p}} \mathfrak{Z}_{\tilde{v},w_{\tilde{v}}} \in Z^{[F^{+}:\mathbb{Q}]\frac{n(n+1)}{2}}(\operatorname{Spec} \widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y})$ , where  $\mathfrak{Z}_{\tilde{v},w_{\tilde{v}}}$  is the cycle defined in (6.18) (applied to  $L = F_{\tilde{v}}, \rho = \rho_{\tilde{v}}$ ). Note that both  $\mathfrak{C}_{w,F}$  and  $\mathfrak{Z}_{w,F}$  are independent of the choice of the representative w in its associated class in  $\mathscr{W}_{L_{P}} \setminus \mathscr{W}_{F}$ . By Lemma 6.5.1, we have:

**Lemma 7.1.10.** For  $w \in \mathscr{W}_F$ , the cycle  $\mathfrak{Z}_{w,F}$  is irreducible.

For  $w, w' \in \mathscr{W}_F$ , let

$$a_{w,w'} := \prod_{v \in S_p} a_{w_{\widetilde{v}},w'_{\widetilde{v}}}, \ c_{w,w'} := \prod_{v \in S_p} c_{w_{\widetilde{v}},w'_{\widetilde{v}}}, \ c'_{w,w'} := \prod_{v \in S_p} c'_{w_{\widetilde{v}},w'_{\widetilde{v}}}$$

where  $a_{w_{\widetilde{v}},w'_{\widetilde{v}}} \in \mathbb{Z}_{\geq 0}$  are those  $a_{w,w'}$  defined above (6.19) (applied to  $L = F_{\widetilde{v}}$  and so on),  $c_{w_{\widetilde{v}},w'_{\widetilde{v}}}, c'_{w_{\widetilde{v}},w'_{\widetilde{v}}} \in \mathbb{Z}_{\geq 0}$  are the integers  $c_{w,w'}, c'_{w,w'}$  respectively in Lemma 5.4.13 (2) (adapted to the setting  $G = \operatorname{Res}_{\mathbb{Q}_p}^{F_{\widetilde{v}}} \operatorname{GL}_n$  and so on). By (6.19) and Theorem 5.4.11 (3), we have

$$\mathfrak{C}_{w,F} = \sum_{w' \in \mathscr{W}_{L_P} \setminus \mathscr{W}_F} a_{w,w'} \mathfrak{Z}_{w',F} = \sum_{\substack{w' \in \mathscr{W}_{L_P} \setminus \mathscr{W}_F \\ w' \max \le w^{\max}}} a_{w,w'} \mathfrak{Z}_{w',F} \in Z^{[F^+:\mathbb{Q}_P]\frac{n(n+1)}{2}}(\operatorname{Spec} \widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y}).$$
(7.15)

We let  $w_{y,\widetilde{v}} \in \mathscr{W}_{\max,F_{\widetilde{v}}}^{P_{\widetilde{v}}}$  be the element associated to  $\rho_{\widetilde{v}}$  (and  $\Omega$ ) as in § 2.3 (denoted by  $w_{\mathscr{F}}$  there) and

$$w_y := (w_{y,\widetilde{v}}) \in \mathscr{W}_{\max}^P = \prod_{v \in S_p} \mathscr{W}_{\max,F_{\widetilde{v}}}^{P_{\widetilde{v}}}$$
(7.16)

which is exactly the product for  $v \in S_p$  of the  $w_y$  in § 6.5 for  $L = F_{\tilde{v}}$ . By Lemma 6.5.3 we have:

**Corollary 7.1.11.** For  $w = (w_{\widetilde{v}}) \in \mathscr{W}_F$ , we have  $\mathfrak{C}_{w,F} \neq 0 \iff \mathfrak{Z}_{w,F} \neq 0 \iff w^{\max} \geq w_y$ .

Corollary 7.1.11 allows to further refine (7.15) as:

$$\mathfrak{C}_{w,F} = \sum_{\substack{w' \in \mathscr{W}_{L_P} \setminus \mathscr{W}_F \\ w_y \le w' \max \le w^{\max} \le w^{\max}}} a_{w,w'} \mathfrak{Z}_{w',F} \in Z^{[F^+:\mathbb{Q}_P]\frac{n(n+1)}{2}}(\operatorname{Spec}\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y}).$$
(7.17)

Corollary 7.1.12. (1) We have

$$[\operatorname{Spec} \widehat{\mathcal{O}}_{\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})\mathfrak{d}_{x},x}] = \sum_{\substack{w \in \mathscr{W}_{L_{P}} \setminus \mathscr{W}_{F} \\ w_{y} \leq w^{\max} \leq w_{x}w_{0,F}}} c'_{w_{x}w_{0,F},w} \mathfrak{Z}_{w,F} \in Z^{[F^{+}:\mathbb{Q}]\frac{n(n+1)}{2}}(\operatorname{Spec} \widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y}).$$

(2) Assume x is a smooth point of  $\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})$ , then we have:

$$[\operatorname{Spec} \widehat{\mathcal{O}}_{\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho})\mathfrak{d}_{x},x}] = \sum_{\substack{w \in \mathscr{W}_{L_{P}} \setminus \mathscr{W}_{F} \\ w_{y} \leq w^{\max} \leq w_{x}w_{0,F}}} b_{w_{x}w_{0,F},w}\mathfrak{C}_{w,F} \in Z^{[F^{+}:\mathbb{Q}]\frac{n(n+1)}{2}}(\operatorname{Spec} \widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y})$$

where  $b_{w,w'} := \prod_{v \in S_p} b_{w_{\widetilde{v}},w'_{\widetilde{v}}}$  for  $w = (w_{\widetilde{v}}), w' = (w'_{\widetilde{v}}) \in \mathscr{W}_F$ , which is equal to the multiplicity of  $L(w'^{\max}w_{0,F} \cdot 0)$  in  $M_P(w^{\max}w_{0,F} \cdot 0)$ .

*Proof.* (1) (resp. (2)) is a consequence of Corollary 7.1.9, Lemma 5.4.13 (2) (resp. Lemma 5.4.13 (1), (3)) and Corollary 7.1.11.  $\Box$ 

# 7.2 Companion constituents

In this section, we prove our main (global) results on *p*-adic automorphic representations.

For  $v \in S_p$ , we let  $\rho_{\widetilde{v}}$  be an *n*-dimensional generic potentially crystalline representation of  $\operatorname{Gal}_{F_{\widetilde{v}}}$ over E with distinct Hodge-Tate weights  $\mathbf{h}_{\widetilde{v}} = (h_{\widetilde{v},i,\tau})_{\substack{i=1,\dots,n\\\tau\in\Sigma_{\widetilde{v}}}}$ . Let  $\lambda := (\lambda_{\widetilde{v}}) := (\lambda_{\widetilde{v},i,\tau})_{\substack{i=1,\dots,n\\v\in S_p,\tau\in\Sigma_{\widetilde{v}}}}$ with  $\lambda_{\widetilde{v},i,\tau} = h_{\widetilde{v},i,\tau} + i - 1$  (so  $\lambda$  is integral P-dominant). Let  $x := (y,\underline{x},1) = (x^p, x_p,\underline{x},1) \in$  $(\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_0$  (so  $y := (x^p, x_p) \in \operatorname{Spf} R_{\infty})^{\operatorname{rig}}$ ). Let  $w = (w_{\widetilde{v}}) \in \mathscr{W}_{\min}^P$ , and assume x lies in  $\mathcal{E}^{\infty}_{\Omega,w\cdot\lambda}(\overline{\rho})$ . Assume moreover that  $x^p$  lies in the smooth locus of  $\mathfrak{X}^p_{\infty}$ . We let  $w_y \in \mathscr{W}_{\max}^P$ be as in (7.16).

**Lemma 7.2.1.** Let  $w' \in \mathscr{W}_{\min}^P$ , if  $[\mathcal{N}_{w' \cdot \lambda, y}] \neq 0$  then  $w'w_{0,F} \geq w_y$ .

Proof. If  $[\mathcal{N}_{w'\cdot\lambda,y}] \neq 0$ , we deduce by Lemma 7.1.7 (applied to  $\lambda' = w' \cdot \mathbf{h}$ ,  $\mathfrak{d} = \mathfrak{d}_x = 0$ ) that  $(y, \underline{x}, 1) \in \mathcal{E}^{\infty}_{\Omega, w'\cdot\lambda}(\overline{\rho})$ . Then using Theorem 3.3.9 and Corollary 6.4.12, we get  $w'w_{0,F} \geq w_y$ .  $\Box$ 

The following lemma is a consequence of (7.13), Remark 5.4.12 and Lemma 7.2.1 (compare with Corollary 7.1.12 (2)):

**Lemma 7.2.2.** We have an equality of cycles in  $Z^{[F^+:\mathbb{Q}]\frac{n(n+1)}{2}}(\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y})$ :

$$\left[\mathcal{M}_{w\cdot\lambda}, y\right] = \sum_{\substack{w' \in \mathscr{W}_{\min}^P \\ w_y \le w'w_{0,F} \le ww_{0,F}}} b_{ww_{0,F},w'w_{0,F}} [\mathcal{N}_{w'\cdot\lambda,y}].$$
(7.18)

# **Proposition 7.2.3.** We have $[\mathcal{N}_{w \cdot \lambda, y}] \neq 0$ .

*Proof.* By Theorem 3.3.9 and Corollary 6.4.12, we have  $ww_{0,F} \ge w_y$ .

If  $ww_{0,F} = w_y$ , then by Corollary 7.1.9 and Corollary 6.4.10, we know that  $\mathcal{E}^{\infty}_{\Omega,w\cdot\lambda}(\overline{\rho})$  is smooth at the point x. By Corollary 3.3.3 (3), it follows that  $\mathcal{M}^{\infty}_{\Omega,w\cdot\lambda}$  is (non-zero) locally free at the point x, hence  $[\mathcal{M}_{w\cdot\lambda,y}] \neq 0$ . We easily deduce from (7.18)  $[\mathcal{M}_{w\cdot\lambda,y}] = [\mathcal{N}_{w\cdot\lambda,y}]$  hence  $[\mathcal{N}_{w\cdot\lambda,y}] \neq 0$ .

Now assume  $ww_{0,F} > w_y$ . We let  $\mathfrak{X}_{\overline{\rho}_p}^{\mathrm{pcr}}(\xi_0, \mathbf{h}) := \prod_{v \in S_p} \mathfrak{X}_{\overline{\rho}_{\widetilde{v}}}^{\mathrm{pcr}}(\xi_{0,\widetilde{v}}, \mathbf{h}_{\widetilde{v}})$  where  $\xi_0 = (\xi_{0,\widetilde{v}})$  is the inertial type of  $\rho_p = (\rho_{\widetilde{v}})$ . For  $w' = (w'_{\widetilde{v}}) \in \mathscr{W}_{\mathrm{max}}^P$ , we let

$$V_{\overline{\rho}_p}^{\mathrm{pcr}}(\xi_0, \mathbf{h})_{w'} := \prod_{v \in S_p} V_{\overline{\rho}_{\widetilde{v}}}^{\mathrm{pcr}}(\xi_{0,\widetilde{v}}, \mathbf{h}_{\widetilde{v}})_{w'_{\widetilde{v}}} \subset \mathfrak{X}_{\overline{\rho}_p}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$$

where each term in the product is defined as in Proposition 4.3.8. As  $ww_{0,F} > w_y^{\text{max}}$ , by Proposition 4.3.8, we see that  $\rho_p = (\rho_{\tilde{v}})$  lies in the Zariski-closure  $\overline{V_{\bar{\rho}_p}^{\text{pcr}}(\xi_0, \mathbf{h})_{ww_{0,F}}}$  of  $V_{\bar{\rho}_p}^{\text{pcr}}(\xi_0, \mathbf{h})_{ww_{0,F}}$  in  $\mathfrak{X}_{\bar{\rho}_p}^{\text{pcr}}(\xi_0, \mathbf{h})$ . Let  $\mathcal{C}_{ww_{0,F}}$  be the connected component of  $\overline{V_{\bar{\rho}_p}^{\text{pcr}}}(\xi_0, \mathbf{h})_{ww_{0,F}}$  at the point  $\rho_p$ , and let  $\mathcal{U}^p$  be the smooth locus of the irreducible component of  $\mathfrak{X}_{\infty}^p$  at the point  $x^p$ . The embedding in (4.38) induces a closed embedding

$$\iota: \overline{V_{\overline{\rho}_p}^{\mathrm{pcr}}(\xi_0, \mathbf{h})_{ww_{0,F}}} \times \mathcal{U}^p \longrightarrow X_{\Omega, w \cdot \lambda}(\overline{\rho}_p) \times \mathcal{U}^p.$$

We have  $\iota(\rho_p, x^p) = x \in \mathcal{E}^{\infty}_{\Omega, w \cdot \lambda}(\overline{\rho})$ . Moreover, by Corollary 6.4.7, it follows that  $X_{\Omega, w \cdot \lambda}(\overline{\rho}_p) \times \mathcal{U}^p$  is irreducible at any point in the image of  $\iota$ . We then deduce that  $\iota$  induces a closed embedding

$$\iota: \mathcal{C}_{ww_{0,F}} \times \mathcal{U}^p \hookrightarrow \mathcal{E}^{\infty}_{\Omega, w \cdot \lambda}(\overline{\rho}).$$
(7.19)

Let  $\mathcal{V}_{ww_{0,F}} := \mathcal{C}_{ww_{0,F}} \cap V_{\overline{\rho}_p}^{\mathrm{pcr}}(\xi_0, \mathbf{h})_{ww_{0,F}}$ , which is Zariski dense in  $\mathcal{C}_{ww_{0,F}}$ ,  $(x'_p, x'^p)$  be any point in  $\mathcal{V}_{ww_{0,F}} \times \mathcal{U}^p$ , and put  $y' := (x'_p, x'^p) \in \mathfrak{X}_{\infty}$ . Since  $ww_{0,F} = w_{y'}$  (where, similarly as for  $w_y, w_{y'}$ is associated to  $x'_p$  and  $\Omega$ , noting that the framing in  $x'_p$  does not cause problems), by the same argument as for the case  $ww_{0,F} = w_y$  above (but now applied with  $\iota(y')$  instead of x), we have  $[\mathcal{N}_{w\cdot\lambda,y'}] \neq 0$ . By Lemma 7.1.8, this implies  $\iota(y') \in \mathfrak{Z}_{w\cdot\lambda,w\cdot\lambda}$ . As  $\iota(y)$  lies in the Zariski closure in  $\mathcal{E}^{\infty}_{\Omega,w\cdot\lambda}(\overline{\rho})$  of the set of all points  $\iota(y')$  as above, and  $\mathfrak{Z}_{w\cdot\lambda,w\cdot\lambda}$  is by definition Zariski-closed in  $\mathcal{E}^{\infty}_{\Omega,w\cdot\lambda}(\overline{\rho})$ , we deduce  $\iota(y) \in \mathfrak{Z}_{w\cdot\lambda,w\cdot\lambda}$ , hence (by Lemma 7.1.8 again)  $[\mathcal{N}_{w\cdot\lambda,y}] \neq 0$ .

**Corollary 7.2.4.** We have  $[\mathcal{N}_{w'\cdot\lambda,y}] \neq 0$  for all  $w' \in \mathscr{W}_{\min}^P$  such that  $w'w_{0,F} \geq ww_{0,F}$  (equivalently  $w' \leq w$ ).

Proof. Assume first  $\lg(w') = \lg(w) + 1$ . As  $[\mathcal{N}_{w \cdot \lambda, y}] \neq 0$ , by (7.18),  $[\mathcal{M}_{w' \cdot \lambda, y}] \neq 0$  (note that  $b_{ww_{0,F},w'w_{0,F}} = 1$ , cf. Remark 5.4.12) hence  $(x^p, x_p, \underline{x}, 1) \in \mathcal{E}^{\infty}_{\Omega,w' \cdot \lambda}(\overline{\rho})$  (cf. (7.11)). Applying Proposition 7.2.3 with w replaced by w', we obtain  $[\mathcal{N}_{w' \cdot \lambda, y}] \neq 0$ . We can then start again this argument with a w'' such that  $\lg(w'') = \lg(w') + 1$ . Using [11, Thm. 2.5.5], the corollary follows from an obvious induction.

**Remark 7.2.5.** In particular, if  $x \in \mathcal{E}^{\infty}_{\Omega,w\cdot\lambda}(\overline{\rho})$ , then  $x \in \mathcal{E}^{\infty}_{\Omega,w'\cdot\lambda}(\overline{\rho})$  for all  $w' \in \mathscr{W}^P_{\min}$ ,  $w' \leq w$ . In the trianguline case (i.e. P = B), it was proved in [18, Thm. 5.5] that this holds without assuming  $\rho_{\overline{\nu}}$  to be generic potentially crystalline with distinct Hodge-Tate weights (see also Remark 4.3.10 (2)). The proof was quite different and based on the fact that, for any  $w' \leq w$  (in  $\mathscr{W}_F$ ),  $L(w \cdot \lambda)$  is an irreducible constituent of the Verma module  $M(w' \cdot \lambda)$ , which does not hold in general if  $w', w \in \mathscr{W}^P_{\min}$  and  $M(w' \cdot \lambda)$  is replaced by  $M_P(w' \cdot \lambda)$ . When  $P \neq B$ , the authors do not know if this statement holds without the potentially crystalline assumption.

**Corollary 7.2.6.** Assume that we are in the same situation as Conjecture 3.3.11 (we use the notation of loc. cit.) Let  $x^p$  be the point of  $(\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}}$  associated to  $\mathfrak{m}$ , and assume moreover that  $x^p$  is a smooth point of  $(\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}}$ . Then Conjecture 3.3.11 (1) is equivalent to Conjecture 3.3.11 (2).

Proof. By Lemma 3.3.13, we are left to show that, if  $x = (\mathfrak{m}, \boxtimes_{v \in S_p} \pi_{L_{P_{\widetilde{v}}}}, 1) \in \mathcal{E}_{\Omega, w \cdot \lambda}^{\infty}(\overline{\rho})$ , then  $\widehat{\otimes}_{v \in S_p} C(w_{\widetilde{v}}, \mathscr{F}_{\widetilde{v}}) \hookrightarrow \Pi_{\infty}^{R_{\infty}-\mathrm{an}}[\mathfrak{m}]$ . However, this follows directly from Proposition 7.2.3 and Lemma 7.1.6 (applied to  $\mu = w \cdot \lambda$ ,  $\mathfrak{m}_y = \mathfrak{m}$  etc.).

**Corollary 7.2.7** (Classicality). Let  $\overline{\rho}$  be an  $U^p$ -modular continuous representation of  $\operatorname{Gal}_F$  over  $k_E$ . Let  $\mathcal{E}_{\Omega,\lambda}(U^p,\overline{\rho})$  be as in (the end of) § 3.2.4 and  $x = (\eta, \pi_{L_P}, \chi) \in \mathcal{E}_{\Omega,\lambda}(U^p,\overline{\rho})$ . Assume

- (1) Hypothesis 3.3.1;
- (2)  $\rho_{x,\tilde{v}}$  is generic potentially crystalline with distinct Hodge-Tate weights for all  $v \in S_p$ ;
- (3)  $\rho_{x,\tilde{v}}$  is a smooth point of  $(\operatorname{Spf} R_{\overline{\rho}_{\tilde{v}}})^{\operatorname{rig}}$  for  $v \in S \setminus S_p$ .

Then  $\widehat{S}(U^p, E)^{\text{lalg}}[\mathbb{T}^S = \eta] \neq 0$ , i.e.  $\rho_x$  is associated to a classical automorphic representation of  $G(\mathbb{A}_{F^+})$ .

Proof. By Proposition 3.2.3, we can and do assume  $\chi = 1$  (by modifying  $\lambda$  and  $\Omega$ ). Then by Proposition 3.2.17, there exist a dominant weight  $\mu$  and  $w \in \mathscr{W}_{\min}^{P}$  such that  $\lambda = w \cdot \mu$ . By Hypothesis 3.3.1 and Proposition 3.3.8, we can associate to x a point, still denoted by x, of the form  $(y = (x^{p}, x_{p}), \pi_{L_{P}}, 1)$  in  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  satisfying  $\widehat{S}(U^{p}, E)[\mathbb{T}^{S} = \eta] \cong \Pi_{\infty}[\mathfrak{m}_{y}]$ . Moreover, by the assumption (3),  $x^{p}$  is a smooth point of  $(\operatorname{Spf} R_{\infty}^{p})^{\operatorname{rig}}$ . By Corollary 7.2.4, we see  $\mathcal{N}_{\mu,y} \neq 0$  hence by Lemma 7.1.6 (noting that, as  $\mu$  is dominant, the Orlik-Strauch representation  $\mathcal{F}_{P^{-}(\mathbb{Q}_{p})}^{G_{p}}(-)$  there is locally algebraic)  $\widehat{S}(U^{p}, E)[\mathbb{T}^{S} = \eta]^{\operatorname{lalg}} \cong \Pi_{\infty}[\mathfrak{m}_{y}]^{\operatorname{lalg}} \neq 0$ . The corollary follows.

**Remark 7.2.8.** (1) Using similar arguments as in the proof of [18, Thm. 3.9], one may remove the assumption (3) in Corollary 7.2.7 but assuming the weight  $\lambda + \operatorname{wt}(\chi) \circ \operatorname{det}_{L_P}$  is dominant.

(2) If  $S \setminus S_p = \{v_1\}$  where  $v_1$  is a finite place of  $F^+$  as in [26, § 2.3] and  $U^p$  is chosen as in loc. cit., then  $(\operatorname{Spf} R^p_{\infty})^{\operatorname{rig}}$  is smooth (so the assumption (3) is automatically satisfied).

**Theorem 7.2.9.** Assume that we are in the setting of § 3.2.1, and let  $\rho : \operatorname{Gal}_F \to \operatorname{GL}_n(E)$  be a continuous representation unramified outside S such that the modulo p reduction  $\overline{\rho}$  is irreducible and  $U^p$ -modular. Let  $\mathfrak{m}_{\rho}$  be the associated maximal ideal of  $\mathbb{T}^S[1/p]$  (via (3.19)). Assume

- (1) Hypothesis 3.3.1;
- (2)  $\rho_{\tilde{v}}$  is generic potentially crystalline with distinct Hodge-Tate weights for all  $v \in S_p$ ;
- (3)  $\rho_{\tilde{v}}$  is a smooth point of  $(\operatorname{Spf} R_{\overline{\rho_{\tilde{v}}}})^{\operatorname{rig}}$  for  $v \in S \setminus S_p$ ;
- (4) there exists a parabolic subgroup  $P \supset \prod_{v \in S_p} \operatorname{Res}_{\mathbb{Q}_p}^{F_{\widetilde{v}}} B$  such that  $J_P(\widehat{S}(U^p, E)^{\operatorname{an}}[\mathfrak{m}_{\rho}])$  has non-zero locally algebraic vectors for  $L_P^D(\mathbb{Q}_p)$ .

Then  $\widehat{S}(U^p, E)[\mathfrak{m}_{\rho}]^{\text{lalg}} \neq 0$ , i.e.  $\rho$  is associated to a classical automorphic representation of  $G(\mathbb{A}_{F^+})$ .

Proof. Let P be a minimal parabolic subgroup such that the assumption (4) holds. Thus there exists an integral P-dominant weight  $\lambda$  such that  $J_P(\widehat{S}(U^p, E)^{\mathrm{an}}[\mathfrak{m}_\rho])_{\lambda} \neq 0$  (see (3.1)). As  $J_P(\widehat{S}(U^p, E)^{\mathrm{an}}[\mathfrak{m}_\rho])_{\lambda}$  is an essentially admissible locally analytic representation of  $L_P(\mathbb{Q}_p)$  and is smooth for the  $L_P^D(\mathbb{Q}_p)$ -action, there exists a compact open subgroup  $H \subset L_P^D(\mathbb{Z}_p)$  such that  $J_P(\widehat{S}(U^p, E)^{\mathrm{an}}[\mathfrak{m}_\rho])_{\lambda}^H$  is a non-zero essentially admissible locally analytic representation of  $Z_{L_P}(\mathbb{Q}_p)$ . There exists thus a continuous character  $\delta$  of  $Z_{L_P}(\mathbb{Q}_p)$  such that

$$J_P(\widehat{S}(U^p, E)^{\mathrm{an}}[\mathfrak{m}_\rho])^H_{\lambda}[Z_{L_P}(\mathbb{Q}_p) = \delta] \neq 0.$$

We let  $\chi$  be a continuous character of  $Z_{L_P}^0$  such that  $\operatorname{wt}(\chi) \circ \operatorname{det}_{L_P} = \operatorname{wt}(\delta)$  and put  $\delta^{\infty} := \delta(\chi_{\underline{\omega}}^{-1} \circ \operatorname{det}_{L_P})$ , which is a smooth character of  $Z_{L_P}(\mathbb{Q}_p)$ . Consider the (non-zero)  $L_P(\mathbb{Q}_p)$ -representation:

$$(J_P(\widehat{S}(U^p, E)^{\mathrm{an}}[\mathfrak{m}_{\rho}])_{\lambda} \otimes_E \chi_{\underline{\varpi}}^{-1} \circ \mathrm{det}_{L_P})[Z_{L_P}(\mathbb{Q}_p) = \delta^{\infty}]$$

which is smooth and admissible by [42, Cor. 6.4.14]. Let  $\pi_{L_P}$  be an irreducible subrepresentation of  $(J_P(\widehat{S}(U^p, E)^{\mathrm{an}}[\mathfrak{m}_{\rho}])_{\lambda} \otimes_E \chi_{\underline{\varpi}}^{-1} \circ \det_{L_P})[Z_{L_P}(\mathbb{Q}_p) = \delta^{\infty}]$ , so we have

$$\pi_{L_P} \otimes_E (\chi_{\underline{\varpi}} \circ \det_{L_P}) \otimes_E L(\lambda)_P \longrightarrow J_P(S(U^p, E)^{\mathrm{an}}[\mathfrak{m}_{\rho}]).$$
(7.20)

Let  $\Omega$  be the Bernstein component of  $\pi_{L_P}$ . As P is minimal, using (7.20), it is not difficult to see that  $\Omega$  is cuspidal. By (3.13) and (7.20), it follows that  $(\mathfrak{m}_{\rho}, \pi_{L_P}, \chi) \in \mathcal{E}_{\Omega,\lambda}(U^p, \overline{\rho})$ . The theorem then follows from Corollary 7.2.7.

**Remark 7.2.10.** The case where  $P = \prod_{v \in S_p} \operatorname{Res}_{\mathbb{Q}_p}^{F_{\overline{v}}} B$  was essentially proved (without the assumption (3)) in [19, Thm. 5.1.3] (see also [19, Rem. 5.1.5]).

Next we prove results towards the socle conjecture in [13]. Let  $\lambda = (\lambda_{\tilde{\nu},i,\tau})$  be an integral dominant weight of  $G_p$ ,  $\mathbf{h} = (h_{\tilde{\nu},i,\tau})$  with  $h_{\tilde{\nu},i,\tau} = \lambda_{\tilde{\nu},i,\tau} - i + 1$  (hence  $\mathbf{h}$  is strictly dominant). For  $w \in \mathscr{W}_{\min}^P$ , let  $\mathscr{S}_w$  be the set of points  $x = (y = (x^p, x_p), \underline{x}, 1) \in \mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})_0$  (recall that  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})_0$  is the fibre of  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  at the weight 0 of  $\mathfrak{z}_{L_P}$ ) such that:

- (1) for  $v \in S \setminus S_p$ ,  $\rho_{x,\tilde{v}}$  is a smooth point in  $(\operatorname{Spf} R_{\overline{\rho_{\tilde{v}}}})^{\operatorname{rig}}$ ;
- (2) for  $v \in S_p$ ,  $\rho_{x,\tilde{v}}$  is generic potentially crystalline of Hodge-Tate weights  $\mathbf{h}_{\tilde{v}}$  and of inertial type  $\xi_{0,\tilde{v}}$  (which is determined by  $\Omega$ );
- (3)  $w_y = ww_{0,F}$ .

Put  $\mathscr{S} := \bigcup_{w \in \mathscr{W}_{\min}^{P}} \mathscr{S}_{w}$ . Note that  $\mathscr{S}_{1}$  consists of non-critical points in  $\mathscr{S}$ .

**Proposition 7.2.11.** Let  $x \in \mathscr{S}$ , then we have  $[\mathcal{N}_{\lambda,y}] \in \mathbb{Z}\mathfrak{C}_{w_{0,F},F} \in Z^{\frac{n(n+1)}{2}[F^+:\mathbb{Q}]}(\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y})$  (cf. (7.14)).

Proof. Let  $\sigma$  be a cuspidal  $L_P(\mathbb{Z}_p)$ -type for  $\Omega$ . For any  $s \in \mathbb{Z}_{\geq 1}$ , put  $\pi_{\underline{x},s} := (\operatorname{c-ind}_{L_P^0}^{L_P(\mathbb{Q}_p)} \sigma)^{\infty} \otimes_{\mathcal{Z}_{\Omega}} \mathcal{Z}_{\Omega}/\mathfrak{m}_x^s$ . By Lemma 3.1.2 and Lemma 7.1.1, we have (noting that  $\mathcal{N}_{\lambda,y}^{\vee}$  is isomorphic to the direct

limit of the following *E*-vector spaces over s, cf. (7.12))

$$\left( \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(L(\lambda), \Pi_{\infty}^{R_{\infty}-\mathrm{an}}[\mathfrak{m}_{y}^{s}])_{\mathrm{fs}}^{N_{p}^{0}} \otimes_{E} \sigma^{\vee} \right)^{L_{p}^{0}} [\mathfrak{m}_{\underline{x}}^{s}] \xrightarrow{\sim} \operatorname{Hom}_{L_{P}(\mathbb{Q}_{p})} \left( \pi_{\underline{x},s}, \operatorname{Hom}_{\mathrm{U}(\mathfrak{g})}(L(\lambda), \Pi_{\infty}^{R_{\infty}-\mathrm{an}}[\mathfrak{m}_{y}^{s}])_{\mathrm{fs}}^{N_{p}^{0}} \right) \xrightarrow{\sim} \operatorname{Hom}_{G_{p}} \left( \mathcal{F}_{P^{-}(\mathbb{Q}_{p})}^{G_{p}} \left( L^{-}(-\lambda), \pi_{\underline{x},s} \otimes_{E} \delta_{P}^{-1} \right), \Pi_{\infty}^{R_{\infty}-\mathrm{an}}[\mathfrak{m}_{y}^{s}] \right) \cong \operatorname{Hom}_{G_{p}} \left( L(\lambda) \otimes_{E} \left( \operatorname{Ind}_{P^{-}(\mathbb{Q}_{p})}^{G_{p}} \pi_{\underline{x},s} \otimes_{E} \otimes_{E} \delta_{P}^{-1} \right)^{\infty}, \Pi_{\infty}^{R_{\infty}-\mathrm{an}}[\mathfrak{m}_{y}^{s}] \right) \hookrightarrow \operatorname{Hom}_{G_{p}} \left( L(\lambda) \otimes_{E} \left( \operatorname{c-ind}_{K_{p}}^{G_{p}} \widetilde{\sigma} \right), \Pi_{\infty}^{R_{\infty}-\mathrm{an}}[\mathfrak{m}_{y}^{s}] \right) \cong \operatorname{Hom}_{K_{p}} (L(\lambda) \otimes_{E} \widetilde{\sigma}, \Pi_{\infty}^{R_{\infty}-\mathrm{an}})[\mathfrak{m}_{y}^{s}]$$

$$(7.21)$$

where  $\tilde{\sigma}$  is a  $K_p$ -type of  $G_p$  for  $(\operatorname{Ind}_{P^-(\mathbb{Q}_p)}^{G_p} \pi_{\underline{x}} \otimes_E \delta_P^{-1})^{\infty}$  (note that  $(\operatorname{Ind}_{P^-(\mathbb{Q}_p)}^{G_p} \pi_{\underline{x},s} \otimes_E \delta_P^{-1})^{\infty}$ ) is isomorphic to a successive extension of  $(\operatorname{Ind}_{P^-(\mathbb{Q}_p)}^{G_p} \pi_{\underline{x}} \otimes_E \delta_P^{-1})^{\infty})$ . By [26, Lemma 4.17], the  $R_{\infty}$ -action on the last term in (7.21) factors through  $R_{\infty}^p \widehat{\otimes}_{\mathcal{O}_E} R_{\overline{\rho}_p}^{pcr}(\xi_0, \mathbf{h})$  (where we recall that  $R_{\overline{\rho}_p}^{pcr}(\xi_0, \mathbf{h}) = \widehat{\otimes}_{v \in S_p} R_{\overline{\rho}_v}^{pcr}(\xi_{0,\widetilde{v}}, \mathbf{h}_{\widetilde{v}}))$ . We deduce that the  $R_{\infty}$ -action on  $\mathcal{N}_{\lambda,y}$  also factors through  $R_{\infty}^p \widehat{\otimes}_{\mathcal{O}_E} R_{\overline{\rho}_p}^{pcr}(\xi_0, \mathbf{h})$ . By Remark 6.5.2, we have  $\mathfrak{C}_{w_{0,F},F} = [\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty}^p,x^p} \widehat{\otimes} \widehat{\mathcal{O}}_{\mathfrak{X}_{\overline{\rho}_p}^{pcr}(\xi_0, \mathbf{h}),\rho_p}]$  (here the completion in  $\widehat{\otimes}$  is with respect to the maximal ideal associated to  $(x^p, \rho_p)$ ). The proposition follows.  $\Box$ 

Let  $x \in \mathscr{S}$  and recall that, by Corollary 7.1.9,  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  is irreducible at the point x. Let  $m_y \in \mathbb{Z}_{\geq 1}$  such that  $\mathcal{M}_{\Omega,\lambda}^{\infty}$  is locally free of rank  $m_y$  in the smooth locus of a sufficiently small neighbourhood of x in  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$ . We view x as a point of  $(\operatorname{Spf} R_{\infty})^{\operatorname{rig}} \times (\operatorname{Spec} \mathcal{Z}_{\Omega})^{\operatorname{rig}} \times \widehat{\mathcal{Z}}_{0}$ . For  $w \in \mathscr{W}_{\min}^{P}$ , we want to understand when  $x \in \mathcal{E}_{\Omega,w\cdot\lambda}^{\infty}(\overline{\rho})$ .

**Lemma 7.2.12.** If  $\lg(w_y) \ge \lg(w_{0,F}) - 1$ , then for any  $w \in \mathscr{W}_{\min}^P$  such that  $ww_{0,F} \ge w_y$ , we have  $[\mathcal{N}_{w\cdot\lambda,y}] \ne 0$ , hence  $x \in \mathcal{E}_{\Omega,w\cdot\lambda}^{\infty}(\overline{\rho})$ . Moreover,  $\mathcal{M}_{\Omega,w\cdot\lambda}^{\infty}$  is locally free of rank  $m_y$  at x.

Proof. Since  $\lg(w_y) \ge \lg(w_{0,F}) - 1$ , by Corollary 7.1.9 and Corollary 6.4.10,  $\mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  is smooth at the point x. Hence by Corollary 3.3.3 (3),  $\mathcal{M}_{\Omega,\lambda}^{\infty}$  is locally free of rank  $m_y$  at the point x. By Corollary 7.1.12 (2) and (7.18) (noting that, as  $\lg(w_y) \ge \lg(w_{0,F}) - 1$ , for  $w \in \mathscr{W}_{\min,F}^P$  we have

 $b_{w_{0,F},ww_{0,F}} = \begin{cases} 1 & ww_{0,F} \ge w_y \\ 0 & \text{otherwise} \end{cases}$ , see Remark 5.4.12 and Lemma 7.2.15 (1) below), we have:

$$\sum_{w_y \le ww_{0,F} \le w_{0,F}} [\mathcal{N}_{w \cdot \lambda, y}] = [\mathcal{M}_{\lambda, y}] = m_y \sum_{w_y \le ww_{0,F} \le w_{0,F}} \mathfrak{C}_{ww_{0,F}, y} \in Z^{[F^+:\mathbb{Q}]\frac{n(n+1)}{2}}(\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty}, y}).$$
(7.22)

By Lemma 7.2.3,  $[\mathcal{N}_{\lambda,y}] \neq 0$  and the case where  $w_y = w_{0,F}$  (which implies w = 1) follows. Assume now  $\lg(w_y) = \lg(w_{0,F}) - 1$  and  $ww_{0,F} = w_y$ . By (7.22) and Proposition 7.2.11, we have

$$[\mathcal{N}_{w\cdot\lambda,y}] = m_y \mathfrak{C}_{w_{0,F},y} + m_y \mathfrak{C}_{ww_{0,F},y} - [\mathcal{N}_{\lambda,y}] \in m_y \mathfrak{C}_{ww_{0,F},y} + \mathbb{Z}\mathfrak{C}_{w_{0,F},y}.$$
(7.23)

By (7.17) and Lemma 7.1.11, it is easy to see  $\mathfrak{C}_{ww_{0,F},y} \notin \mathbb{Z}\mathfrak{C}_{w_{0,F},y}$ . We deduce  $[\mathcal{N}_{w\cdot\lambda,y}] \neq 0$ , hence  $x \in \mathcal{E}^{\infty}_{\Omega,w\cdot\lambda}(\overline{\rho})$ . Again by Corollary 7.1.9 and Corollary 6.4.10,  $\mathcal{E}^{\infty}_{\Omega,w\cdot\lambda}(\overline{\rho})$  is smooth at xhence  $\mathcal{M}^{\infty}_{\Omega,w\cdot\lambda}$  is locally free at x, say of rank  $m'_{y}$ . Hence  $[\mathcal{N}_{w\cdot\lambda,y}] = [\mathcal{M}_{w\cdot\lambda,y}] = m'_{y}\mathfrak{C}_{ww_{0,F},y} \in \mathbb{Z}_{\geq 0}\mathfrak{C}_{ww_{0,F},y}$ . By (7.23) and the fact  $\mathfrak{C}_{ww_{0,F},y} \notin \mathbb{Z}\mathfrak{C}_{w_{0,F},y}$  (again), we deduce  $[\mathcal{N}_{w\cdot\lambda,y}] = m_{y}\mathfrak{C}_{ww_{0,F},y}$ (i.e.  $m'_{y} = m_{y}$ ) and  $[\mathcal{N}_{\lambda,y}] = m_{y}\mathfrak{C}_{w,y}$ . This concludes the proof. Lemma 7.2.13. Let  $w \in \mathscr{W}_{\min}^P$ .

(1) Assume that, for any  $x' = (y', \underline{x}', 1) \in \mathscr{S}_w$ , we have  $[\mathcal{N}_{w \cdot \lambda, y'}] \neq 0$ . Then for any  $x = (y, \underline{x}, 1) \in \mathscr{S}$  with  $w_y \leq ww_{0,F}$ , we have  $[\mathcal{N}_{w \cdot \lambda, y}] \neq 0$ .

(2) Keep the assumption in (1), and assume moreover that, for all  $x' \in \mathscr{S}_w$ ,  $\mathcal{M}_{\Omega,w\cdot\lambda}^{\infty}$  is locally free of rank  $m_{y'}$  in the smooth locus of a sufficiently small neighbourhood of x'. Then, for any  $x = (y, \underline{x}, 1) \in \mathscr{S}$  with  $w_y \leq ww_{0,F}$ ,  $\mathcal{M}_{\Omega,w\cdot\lambda}^{\infty}$  is locally free of rank  $m_y$  in the smooth locus of a sufficiently small neighbourhood of x.

*Proof.* The lemma follows by the same arguments as in Step 9 of the proof of [19, Thm. 5.3.3]. We include a proof for the convenience of the reader.

(1) By Lemma 7.1.8, we only need to show  $x \in \mathfrak{Z}_{w \cdot \lambda, w \cdot \lambda}$ . We use the notation in the proof of Proposition 7.2.3. Denote by  $\mathcal{C} \subset \mathfrak{X}_{\overline{\rho}_p}^{\mathrm{pcr}}(\xi_0, \mathbf{h})$  the irreducible component containing  $\rho_{x,p}$ . The embedding in (4.38) induces an embedding

$$\iota: \mathcal{C} \times \mathcal{U}^p \longrightarrow X_{\Omega, \mathbf{h}}(\overline{\rho}_p) \times \mathcal{U}^p$$

which sends y to  $x \in \mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho}) \hookrightarrow X_{\Omega,\mathbf{h}}(\overline{\rho}_p) \times \mathcal{U}^p$ . By Corollary 7.1.9,  $X_{\Omega,\lambda}(\overline{\rho}_p) \times \mathcal{U}^p$  is irreducible at any point in the image of  $\iota$ . We then deduce  $\iota(\mathcal{C} \times \mathcal{U}^p) \subset \mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  (this is similar to (7.19)). It is clear that  $\mathcal{V}_{ww_{0,F}} \subset \mathcal{C}_{ww_{0,F}} \subset \mathcal{C}$  (recalling that  $\mathcal{C}$  is irreducible and smooth, cf. [63, Thm. 3.3.8]), and that any point in  $\iota(\mathcal{V}_{ww_{0,F}} \times \mathcal{U}^p) \subset \mathcal{E}_{\Omega,\lambda}^{\infty}(\overline{\rho})$  lies in  $\mathscr{S}_w$ . Recall the morphism in (4.38) induces

$$\iota: \mathcal{C}_{ww_{0,F}} \times \mathcal{U}^p \longrightarrow X_{\Omega,w(\mathbf{h})}(\overline{\rho}_p) \times \mathcal{U}^p.$$

For each  $y' \in \mathcal{V}_{ww_{0,F}} \times \mathcal{U}^p$  with  $x' = \iota(y') \in \mathscr{S}_w$ , we have by assumption  $[\mathcal{N}_{w\cdot\lambda}] \neq 0$ , hence by Lemma 7.1.8  $x' \in \mathfrak{Z}_{w\cdot\lambda,w\cdot\lambda} \subset \mathcal{E}^{\infty}_{\Omega,w\cdot\lambda}(\overline{\rho}) \hookrightarrow X_{\Omega,w(\mathbf{h})}(\overline{\rho}_p) \times \mathcal{U}^p$ . As x lies in the Zariski-closure of such x', it follows  $x = \iota(y) \in \mathfrak{Z}_{w\cdot\lambda,w\cdot\lambda}$ . (1) is then a consequence of Lemma 7.1.8.

(2) By (1), we have  $x \in \mathfrak{Z}_{w \cdot \lambda, w \cdot \lambda} \hookrightarrow \mathcal{E}_{\Omega, w \cdot \lambda}^{\infty}(\overline{\rho})$ . Let  $\mathcal{D}_w$  (resp.  $\mathcal{D}_1$ ) be the irreducible component of  $\mathcal{E}_{\Omega, w \cdot \lambda}^{\infty}(\overline{\rho})$  (resp.  $\mathcal{E}_{\Omega, \lambda}^{\infty}(\overline{\rho})$ ) containing x. As in the proof of (1), we have  $\iota(\mathcal{V}_{ww_{0,F}}) \subset \mathcal{D}_1$  and  $\iota(\mathcal{V}_{ww_{0,F}}) \subset \mathcal{D}_w$ . Let  $U_w$  be a Zariski-open neighbourhood of x in  $\mathcal{E}_{\Omega, w \cdot \lambda}^{\infty}(\overline{\rho})$ . Since  $\mathcal{E}_{\Omega, w \cdot \lambda}^{\infty}(\overline{\rho})$  is irreducible at x (Corollary 7.1.9), shrinking  $U_w$ , we can and do assume that  $U_w$  and the smooth locus  $U_w^{\mathrm{sm}}$  of  $U_w$  are both irreducible (note that  $U_w^{\mathrm{sm}}$  is Zariski-open Zariski-dense in  $U_w$ , cf. [17, Prop. 2.3]). There exists m such that  $\mathcal{M}_{\Omega, w \cdot \lambda}^{\infty}$  is locally free on  $U_w^{\mathrm{sm}}$  of constant rank m. Similarly, we let  $U_1$  be an irreducible Zariski-open neighbourhood of x in  $\mathcal{E}_{\Omega, \lambda}^{\infty}(\overline{\rho})$  such that  $U_1^{\mathrm{sm}}$  is also irreducible. Then  $\mathcal{M}_{\Omega, \lambda}^{\infty}$  is locally free on  $U_1^{\mathrm{sm}}$  of rank  $m_y$ . We have  $(x^p, x_p) \in \iota^{-1}(U_1) \cap \iota^{-1}(U_w) \cap$  $(\mathcal{V}_{ww_{0,F}} \times \mathcal{U}^p)$ . Since  $\mathcal{V}_{ww_{0,F}} \times \mathcal{U}^p$  is Zariski-dense in  $\mathcal{C}_{ww_{0,F}} \times \mathcal{U}^p$ , we deduce  $\iota^{-1}(U_1) \cap \iota^{-1}(U_w) \cap$  $(\mathcal{V}_{ww_{0,F}} \times \mathcal{U}^p) \neq \emptyset$ , and we let  $y' = (x'^p, x'_p)$  be a point in the intersection. By Corollary 7.1.9 and Corollary 6.4.10, the point  $x' := \iota(y') \in U_w$  is in fact a smooth point of  $U_w$ . Hence by assumption  $m = m_{y'} = m_y$ . This finishes the proof.

**Lemma 7.2.14.** Let  $x = (y, \underline{x}, 1) \in \mathscr{S}$  and assume  $\lg(w_y) \leq \lg(w_{0,F}) - 2$ . Assume that, for all  $w \in \mathscr{W}_{\min}^P$ , with  $ww_{0,F} > w_y$ , we have

- $[\mathcal{N}_{w \cdot \lambda, y}] \neq 0$  (so  $x \in \mathcal{E}^{\infty}_{\Omega, w \cdot \lambda}(\overline{\rho})$ );
- $\mathcal{M}^{\infty}_{\Omega.w\cdot\lambda}$  is locally free of rank  $m_y$  in the smooth locus of a neighbourhood of  $x \in \mathcal{E}^{\infty}_{\Omega.w\cdot\lambda}(\overline{\rho})$ .

We assume that one of the following two conditions holds:

- (1) there exists  $w \in \mathscr{W}_{\max}^P$  such that
  - $w > w_y$  and  $\lg(w) = \lg(w_y) + 2;$
  - dim  $\mathfrak{z}_{L_P}^{ww_y^{-1}}$  = dim  $\mathfrak{z}_{L_P} 2;$
  - the Bruhat interval  $[w_y, w] = \{w' | w_y < w' < w\}$  is full in  $\mathscr{W}_{\max}^P$ , i.e. there exist distinct  $w_1, w_2 \in \mathscr{W}_{\max}^P$  such that  $w_y < w_1, w_2 < w$ ,  $\lg(w_1) = \lg(w_2) = \lg(w_y) + 1$  (e.g. see [19, Lemma 5.2.7]);
- (2) there exists  $w \in \mathscr{W}_{\max}^P$  such that
  - $w > w_y$  and  $\lg(w) = \lg(w_y) + 2;$
  - the Bruhat interval  $[w_y, w]$  is not full in  $\mathscr{W}_{\max}^P$ .

Then  $[\mathcal{N}_{w_y w_{0,F} \cdot \lambda, y}] \neq 0$  (which implies  $x \in \mathcal{E}^{\infty}_{\Omega, w_y w_{0,F} \cdot \lambda}(\overline{\rho})$ ), and  $\mathcal{M}^{\infty}_{\Omega, w_y w_{0,F} \cdot \lambda}$  is locally free of rank  $m_y$  at the point x.

Before proving Lemma 7.2.14, we first give an easy lemma on the multiplicities  $b_{w,w'}$  in Theorem 5.4.11 (2).

**Lemma 7.2.15.** Let  $w, w' \in \mathscr{W}_{\max}^P$  such that w > w'.

(1) If 
$$\lg(w') = \lg(w) - 1$$
, then  $b_{w,w'} = 1$ .  
(2) If  $\lg(w') = \lg(w) - 2$ , then  $b_{w,w'} = \begin{cases} 1 & [w',w] \text{ is full in } \mathscr{W}_{\max}^P \\ 0 & \text{otherwise.} \end{cases}$ 

*Proof.* We only prove (2), (1) following from similar (and easier) arguments. We only need to prove a similar statement for a single  $\operatorname{GL}_n(E)$  (rather than  $G_p$ ). We let hence  $P \subset \operatorname{GL}_n$  be a parabolic subgroup,  $w, w' \in \mathscr{W}_{\max}^P \subset \mathscr{W}$ , and  $b_{w,w'}$  be the multiplicity of  $L(w'w_0 \cdot 0)$  in  $M_P(ww_0 \cdot 0)$ . Recall first that  $L(w'w_0 \cdot 0)$  has multiplicity one in the Verma module  $M(ww_0 \cdot 0)$  (see for instance [57, Ex. 8.3(a)]). By [57, Thm. 9.4(b)], we have an exact sequence

$$\bigoplus_{\alpha \in S_P} M(s_\alpha w w_0 \cdot 0) \to M(w w_0 \cdot 0) \to M_P(w w_0 \cdot 0) \to 0$$
(7.24)

where  $S_P$  denotes the set of simple roots of  $L_P$  and  $s_\alpha$  denotes the corresponding simple reflection. Note that for each  $\alpha \in S_P$  we have  $s_\alpha w < w$  and  $\lg(s_\alpha w) = \lg(w) - 1$  (e.g. see the proof of Proposition A.2.1).

We claim that there exists a unique  $\alpha \in S_P$  such that  $w'w_0 > s_\alpha ww_0$  if and only if [w', w]is not full in  $\mathscr{W}_{\max}^P$ . Indeed, we know there exist two, and only two, elements  $w_i \in \mathscr{W}$  such that  $w' < w_i < w$  (see [19, Lemma 5.2.7]) and by [11, Thm. 2.5.5], at least one of  $w_i$ , say  $w_1$ , lies in  $\mathscr{W}_{\max}^P$ . If  $w'w_0 > s_\alpha ww_0$  for some  $\alpha \in S_P$ , we then deduce  $w' < s_\alpha w < w$ , hence  $s_\alpha w \in [w', w]$ . However,  $s_\alpha w \notin \mathscr{W}_{\max}^P$  so [w', w] is not full in  $\mathscr{W}_{\max}^P$ . If [w', w] is not full in  $\mathscr{W}_{\max}^P$ , there then exists a unique element  $w'' \notin \mathscr{W}_{\max}^P$  such that w' < w'' < w. This implies

$$w' = w'^{\max} < w'' < w''^{\max} \le w^{\max} = w.$$

As  $\lg(w) = \lg(w') + 2$ , this implies  $w''^{\max} = w^{\max} = w$ . Using  $\lg(w) = \lg(w'') + 1$ , it follows that there exists  $s_{\alpha} \in S_P$  such that  $w'' = s_{\alpha}w$ .

So if [w', w] is full,  $L(w'w_0 \cdot 0)$  does not appear in  $M(s_\alpha ww_0 \cdot 0)$  for all  $s_\alpha \in S_P$ . By (7.24) and the discussion above it, we obtain  $b_{w,w'} = 1$ . If [w', w] is not full, let  $\alpha \in S_P$  such that  $w' < s_\alpha w < w$ . Then  $L(w'w_0 \cdot 0)$  has multiplicity one in  $M(s_\alpha ww_0 \cdot 0)$  (and zero in  $M(s_{\alpha'} ww_0 \cdot 0)$ for  $\alpha' \in S_P$ ,  $\alpha' \neq \alpha$ ). As  $M(s_\alpha ww_0 \cdot 0) \hookrightarrow M(ww_0 \cdot 0)$ , we deduce by counting the multiplicities of  $L(w'w_0 \cdot 0)$  in the first two terms of (7.24) that  $b_{w,w'} = 0$ .

Proof of Lemma 7.2.14. Assume condition (1) holds. The proof is the same as Step 10 of the proof of [19, Thm. 5.3.3]. As  $w, w_1, w_2 > w_y$ , by assumption,  $x \in \mathcal{E}^{\infty}_{\Omega, w_i w_{0,F} \cdot \lambda}(\overline{\rho})$  for  $i \in \{1, 2, \emptyset\}$ . Moreover, by Corollary 7.1.9 and Corollary 6.4.10, for  $i \in \{1, 2, \emptyset\}$ ,  $\mathcal{E}^{\infty}_{\Omega, w_i w_{0,F} \cdot \lambda}(\overline{\rho})$  is smooth at x. By Corollary 3.3.3 (3),  $\mathcal{M}^{\infty}_{\Omega, w_i w_{0,F} \cdot \lambda}$  is locally free of rank  $m_y$  at x. Similarly as for (7.22), using Corollary 7.1.12 (2) (applied to the completion of  $\mathcal{E}^{\infty}_{\Omega, w_i w_{0,F} \cdot \lambda}(\overline{\rho})_0$  at x and to the completion of  $\mathcal{E}^{\infty}_{\Omega, w w_{0,F} \cdot \lambda}(\overline{\rho})_0$  at x), (7.18) and Lemma 7.2.15, we obtain equations of cycles in  $Z^{[F^+:\mathbb{Q}]\frac{n(n+1)}{2}}(\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y})$  for i = 1, 2:

$$\left[\mathcal{N}_{w_y w_{0,F} \cdot \lambda, y}\right] + \left[\mathcal{N}_{w_i w_{0,F} \cdot \lambda, y}\right] = m_y \mathfrak{C}_{w_y, y} + m_y \mathfrak{C}_{w_i, y},\tag{7.25}$$

$$\left[\mathcal{N}_{w_y w_{0,F} \cdot \lambda, y}\right] + \left[\mathcal{N}_{w_1 w_{0,F} \cdot \lambda, y}\right] + \left[\mathcal{N}_{w_2 w_{0,F} \cdot \lambda, y}\right] + \left[\mathcal{N}_{w w_{0,F} \cdot \lambda, y}\right] = m_y(\mathfrak{C}_{w_y, y} + \mathfrak{C}_{w_1, y} + \mathfrak{C}_{w_2, y} + \mathfrak{C}_{w, y}).$$
(7.26)

If  $[\mathcal{N}_{w_y w_{0,F} \cdot \lambda, y}] \neq 0$ , then x is a smooth point of  $\mathcal{E}_{\Omega, w_y w_{0,F} \cdot \lambda}^{\infty}(\overline{\rho})$  (by Corollary 7.1.9 and Corollary 6.4.10) and  $\mathcal{M}_{\Omega, w_y w_{0,F} \cdot \lambda}^{\infty}$  is locally free at x (by Corollary 3.3.3 (3)). By Corollary 7.1.12 (2) and (7.18), it follows  $[\mathcal{N}_{w_y w_{0,F} \cdot \lambda, y}] = [\mathcal{M}_{w_y w_{0,F} \cdot \lambda, y}] \in \mathbb{Z}\mathfrak{C}_{w_y, y}$ . Note that  $[\mathcal{N}_{w_y w_{0,F} \cdot \lambda, y}] \in \mathbb{Z}\mathfrak{C}_{w_y, y}$  obviously holds if  $[\mathcal{N}_{w_y w_{0,F} \cdot \lambda, y}] = 0$ . In summary, there exists  $m'_y \in \mathbb{Z}_{\geq 0}$  such that  $[\mathcal{N}_{w_y w_{0,F} \cdot \lambda, y}] = m'_y \mathfrak{C}_{w_y, y}$ . We then deduce from (7.25):  $[\mathcal{N}_{w_i w_{0,F} \cdot \lambda, y}] = m_y \mathfrak{C}_{w_i, y} + (m_y - m'_y)\mathfrak{C}_{w_y, y}$ . By Theorem 5.4.11 (3) and Remark 5.3.5 (1), for  $w', w'' \in \mathscr{W}_{\max}^P$ ,  $a_{w', w''} = 0$  if  $w' \neq w''$  and  $\lg(w'') \geq \lg(w') - 2$ . Combined with (7.17) and using Lemma 7.1.10, it follows that, for  $w' \in \{w_y, w_1, w_2, w\}$ , we have  $\mathfrak{C}_{w', y} = \mathfrak{Z}_{w', y} \neq 0$ , and thus  $\mathfrak{C}_{w', y}$  is irreducible. The equations (7.25) and (7.26) then imply:

$$\begin{aligned} \left[ \mathcal{N}_{w_i w_{0,F} \cdot \lambda, y} \right] &= m_y \mathfrak{Z}_{w_i, y} + (m'_y - m_y) \mathfrak{Z}_{w_y, y}, \\ \left[ \mathcal{N}_{w w_{0,F} \cdot \lambda, y} \right] &= m_y \mathfrak{Z}_{w, y} + (m_y - m'_y) \mathfrak{Z}_{w_y, y}. \end{aligned}$$

As  $\mathfrak{Z}_{w_y,y}$  can only have non-negative coefficients in  $[\mathcal{N}_{ww_{0,F}\cdot\lambda,y}]$  and  $[\mathcal{N}_{w_iw_{0,F}\cdot\lambda,y}]$ , we must have  $m'_y = m_y$ . The proposition in this case follows.

Assume condition (2) holds. Let  $w_1$  be the unique element in  $\mathscr{W}_{\max}^P$  such that  $w_y < w_1 < w$ . Similarly as above, we have that  $\mathfrak{C}_{w',y} = \mathfrak{Z}_{w',y} \neq 0$  (and  $\mathfrak{C}_{w',y}$  is irreducible) for  $w' \in \{w_y, w_1, w\}$ , that  $\mathcal{E}^{\infty}_{\Omega,w_1w_{0,F}\cdot\lambda}(\overline{\rho})$  is smooth at the point x, and that  $\mathcal{M}^{\infty}_{\Omega,w_1w_{0,F}\cdot\lambda}$  is locally free of rank  $m_y$  at x. We have as in (7.25):

$$[\mathcal{N}_{w_y w_{0,F} \cdot \lambda, y}] + [\mathcal{N}_{w_1 w_{0,F} \cdot \lambda, y}] = m_y(\mathfrak{C}_{w_y, y} + \mathfrak{C}_{w_1, y}) = m_y(\mathfrak{Z}_{w_y, y} + \mathfrak{Z}_{w_1, y}).$$

By the same argument as in the last paragraph, there exists  $m'_y \in \mathbb{Z}_{\geq 0}$  such that  $[\mathcal{N}_{w_y w_{0,F} \cdot \lambda, y}] = m'_y \mathfrak{Z}_{w_y,y}$  and  $[\mathcal{N}_{w_1 w_{0,F} \cdot \lambda, y}] = m_y \mathfrak{Z}_{w_1,y} + (m_y - m'_y) \mathfrak{Z}_{w_y,y}$  (so  $m_y \geq m'_y$ ). By Corollary 7.1.12 (1) (applied with " $\lambda = w w_{0,F} \cdot \lambda$ " and  $\mathfrak{d}_x = 0$ ), we have

$$[\operatorname{Spec}\widehat{\mathcal{O}}_{\mathcal{E}^{\infty}_{\Omega,ww_{0,F}\cdot\lambda}(\overline{\rho})_{0},x}] = \sum_{\substack{w'\in\mathscr{W}_{L_{P}}\setminus\mathscr{W}_{F}\\w_{y}\leq w'\max\leq w}} c'_{w,w'}\mathfrak{Z}_{w',F} \in Z^{[F^{+}:\mathbb{Q}_{p}]\frac{n(n+1)}{2}}(\operatorname{Spec}\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y}).$$
(7.27)

By the discussion on  $a_{w',w''}$  in the last paragraph, we can deduce from Corollary 5.4.13 (1), (2) that  $c_{w,w'} = b_{w,w'}$  for w' as in (the sum of) (7.27). Together with Lemma 7.2.15 (2), we see that  $c_{w,w_y} = 0$ , hence that  $c'_{w,w_y} = 0$  by Lemma 5.4.13 (2). We can thus refine (7.27) as

$$[\operatorname{Spec}\widehat{\mathcal{O}}_{\mathcal{E}^{\infty}_{\Omega,ww_{0,F}\cdot\lambda}(\overline{\rho})_{0,x}}] = \mathfrak{Z}_{w,F} + c'_{w,w_{1}}\mathfrak{Z}_{w_{1},F} \in Z^{[F^{+}:\mathbb{Q}_{p}]\frac{n(n+1)}{2}}(\operatorname{Spec}\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y}).$$
(7.28)

As the  $\widehat{\mathcal{O}}_{\mathfrak{X}_{\infty},y}$ -action on  $\mathcal{M}_{ww_{0,F}\cdot\lambda,y}$  factors through  $\widehat{\mathcal{O}}_{\mathcal{E}^{\infty}_{\Omega,ww_{0,F}\cdot\lambda}(\overline{\rho})_{0},x}$ , using (7.28) it follows that

$$[\mathcal{M}_{ww_{0,F}\cdot\lambda,y}] = a_0\mathfrak{Z}_{w,F} + a_1\mathfrak{Z}_{w_1,F}$$

for some  $a_0, a_1 \in \mathbb{Z}_{>0}$ . Using (7.18) and Lemma 5.4.13 (2), we get

$$[\mathcal{N}_{w_1w_{0,F}\cdot\lambda,y}] + [\mathcal{N}_{ww_{0,F}\cdot\lambda,y}] = [\mathcal{M}_{ww_{0,F}\cdot\lambda,y}] = a_0\mathfrak{Z}_{w,F} + a_1\mathfrak{Z}_{w_1,F},$$

hence

$$[\mathcal{N}_{ww_{0,F}\cdot\lambda,y}] + m_y \mathfrak{Z}_{w_1,y} + (m_y - m'_y)\mathfrak{Z}_{w_y,y} = a_0 \mathfrak{Z}_{w,F} + a_1 \mathfrak{Z}_{w_1,F}.$$
(7.29)

As  $\mathfrak{Z}_{w,F}$ ,  $\mathfrak{Z}_{w_1,F}$  are distinct from  $\mathfrak{Z}_{w_y,y}$ , and as  $\mathfrak{Z}_{w_y,y}$  can only have non-negative coefficients in  $[\mathcal{N}_{ww_{0,F}\cdot\lambda,y}]$ , (7.29) implies (using  $m_y \geq m'_y$ )  $m_y - m'_y = 0$ , hence  $[\mathcal{N}_{w_yw_{0,F}\cdot\lambda,y}] = m_y\mathfrak{Z}_{w_y,y}$ . This concludes the proof.

**Remark 7.2.16.** If none of the conditions (1) or (2) in Lemma 7.2.14 hold, then for any  $w \in \mathscr{W}_{\max}^P$  satisfying  $w > w_y$  and  $\lg(w) = \lg(w_y) + 2$ , we have that  $[w_y, w]$  is full in  $\mathscr{W}_{\max}^P$  and  $\dim \mathfrak{z}_{L_P}^{Ww_y^{-1}} > \dim \mathfrak{z}_{L_P} - 2$ . We use the notation in the proof of Lemma 7.2.14 assuming condition (1). The main difference now is that we don't know if  $\mathcal{E}_{\Omega,ww_0,F}^{\infty}(\overline{\rho})$  is smooth at x (see Remark 6.4.11). Consequently (by Corollary 7.1.12 (1) and Lemma 7.2.15, and using similar arguments as in the proof of Lemma 7.2.14 in the case of condition (2)), the equation (7.26) has to be replaced by an equation of the form (the other equations being unchanged):

$$[\mathcal{N}_{w_{y}w_{0,F}\cdot\lambda,y}] + [\mathcal{N}_{w_{1}w_{0,F}\cdot\lambda,y}] + [\mathcal{N}_{w_{2}w_{0,F}\cdot\lambda,y}] + [\mathcal{N}_{ww_{0,F}\cdot\lambda,y}] = a_{0}\mathfrak{Z}_{w,y} + a_{1}\mathfrak{Z}_{w_{1},y} + a_{2}\mathfrak{Z}_{w_{2},y} + a_{3}\mathfrak{Z}_{w_{y},y} \quad (7.30)$$

with  $a_i \in \mathbb{Z}_{\geq 0}$ . Unfortunately, we do not have more control on these coefficients  $a_i$ . The equations (7.25) and (7.30) seem not enough to deduce  $[\mathcal{N}_{w_y w_{0,F} \cdot \lambda, y}] = m_y \mathfrak{Z}_{w_y, y}$  or even  $[\mathcal{N}_{w_y w_{0,F} \cdot \lambda, y}] \neq 0$ . Note that, though we also don't have much control on the coefficients  $a_0$ ,  $a_1$  in the proof of Lemma 7.2.14 when condition (2) holds, the argument here can work as there are fewer terms.

**Proposition 7.2.17.** Let  $x = (y, \underline{x}, 1) \in \mathscr{S}$  and  $w \in \mathscr{W}_{\min}^P$  such that  $ww_{0,F} \ge w_y$ . Assume that, for all  $w' \in \mathscr{W}_{\min}^P$  such that  $w'w_{0,F} \ge ww_{0,F}$ , one of the following properties holds

- (1)  $\lg(w'w_{0,F}) \ge \lg(w_{0,F}) 1;$
- (2) one of the conditions (1), (2) in Lemma 7.2.14 holds with  $w_y$  replaced by  $w'w_{0,F}$ .

Then  $[\mathcal{N}_{w\cdot\lambda,y}] \neq 0$  (which implies  $x \in \mathcal{E}^{\infty}_{\Omega,w\cdot\lambda}(\overline{\rho})$ ), and  $\mathcal{M}^{\infty}_{\Omega,w\cdot\lambda}$  is locally free of rank  $m_y$  in the smooth locus of a sufficiently small neighbourhood of the point x.

*Proof.* By Lemma 7.2.13, we are reduced to the case  $x \in \mathscr{S}_w$  (i.e.  $w_y = ww_{0,F}$ ). The case where  $\lg(ww_{0,F}) \ge \lg(w_{0,F}) - 1$  already follows from Lemma 7.2.12. Assume  $\lg(ww_{0,F}) \le \lg(w_{0,F}) - 2$ , by Lemma 7.2.14 and the assumption in the proposition, we only need to prove

 $[\mathcal{N}_{w'\cdot\lambda,y}] \neq 0$  and  $\mathcal{M}_{\Omega,w'\cdot\lambda}^{\infty}$  is locally free of rank  $m_y$  in the smooth locus of a sufficiently small neighbourhood of the point x (7.31)

for all  $w' \in \mathscr{W}_{\min}^P$  with  $w'w_{0,F} > ww_{0,F}$ . To prove (7.31), again by Lemma 7.2.13, we only need to show (7.31) for  $x \in \mathscr{S}_{w'}$ . By an obvious induction, we are finally reduced to the case  $\lg(ww_{0,F}) \ge \lg(w_{0,F}) - 1$ , which follows from Lemma 7.2.12.

**Corollary 7.2.18.** Let  $x = (y, \underline{x}, 1) \in \mathscr{S}$ . Assume that, for any  $v \in S_p$ , any two factors  $\operatorname{GL}_{n_{\overline{v},i}}$ in  $L_{P_{\overline{v}}}$  with  $n_{\overline{v},i} > 1$  (if they exist) are not adjacent in the product  $L_{P_{\overline{v}}} \cong \prod_{i=1}^{r_{\overline{v}}} \operatorname{GL}_{n_{\overline{v},i}}$ . Then  $[\mathcal{N}_{w\cdot\lambda,y}] \neq 0$  for  $w \in \mathscr{W}_{\min}^P$  if and only if  $ww_{0,F} \geq w_y$ .

*Proof.* The "only if" part follows from Lemma 7.2.1 (which actually holds without the assumption on the adjacent  $\operatorname{GL}_{n_{\tilde{v},i}}$  in the statement). By Proposition A.3.3 in the appendix, under the assumption in the statement, the condition in Corollary 7.2.17 holds for all  $w \in \mathscr{W}_{\min}^P$ . The "if" part then follows by Corollary 7.2.17.

**Remark 7.2.19.** Without the assumption on the adjacent  $\operatorname{GL}_{n_{\tilde{v},i}}$  in Corollary 7.2.18 (for example, when  $P_{\tilde{v}}$  is maximal with  $L_{P_{\tilde{v}}} \cong \operatorname{GL}_{n_{\tilde{v},1}} \times \operatorname{GL}_{n_{\tilde{v},2}}$  and  $n_{\tilde{v},1}, n_{\tilde{v},2} > 1$ ), there may exist  $w_y$  with  $\lg(w_y) \leq \lg(w_{0,F}) - 2$  such that none of the conditions (1) or (2) in Lemma 7.2.14 hold (see Remark A.3.5). As discussed in Remark 7.2.16, in this case we don't know how to deduce  $[\mathcal{N}_{w_yw_{0,F}\cdot\lambda,y}] \neq 0$ from  $[\mathcal{N}_{w\cdot\lambda,y}] \neq 0$  for  $w \in \mathscr{W}_{\min}^P$  such that  $ww_{0,F} > w_y$ . Using the method of [85], it may be possible to obtain some cases where the  $P_{\tilde{v}}$  don't satisfy the assumption in Corollary 7.2.18, but it seems that the maximal parabolic case mentioned above is still resisting.

**Theorem 7.2.20.** Assume we are in the setting of Conjecture 3.2.18. Assume moreover:

- (1) Hypothesis 3.3.1;
- (2)  $\widehat{S}(U^p, E)^{\text{lalg}}[\mathfrak{m}_{\rho}] \neq 0;$
- (3) for all  $v \in S_p$ , the condition in Corollary 7.2.18 holds for the parabolic subgroup  $P_{\tilde{v}}$  associated to  $\mathscr{F}_{\tilde{v}}$ .

Then Conjecture 3.2.18 holds.

Proof. We use the notation of § 3.2.5. Note that, by local-global compatibility in the classical local Langlands correspondence ([25]), the assumption (2) in the statement is equivalent to the existence of an embedding  $\widehat{\otimes}_{v \in S_p} C(1, \mathscr{F}_{\widetilde{v}}) \hookrightarrow \widehat{S}(U^p, E)^{\mathrm{lalg}}[\mathfrak{m}_{\rho}]$ . We let  $\Omega := \Omega_{\mathscr{F}}$  and  $\underline{x} \in (\mathrm{Spec}\,\mathcal{Z}_{\Omega})^{\mathrm{rig}}$  be the point associated to  $\pi_{L_P}$  ( $\pi_{L_P}$  defined as in § 3.2.5). By the assumption (2) and the second sentence, we have  $z = (\eta_{\rho}, \underline{x}, 1) \in \mathcal{E}_{\Omega,\lambda}(U^p, \overline{\rho})$ . By Proposition 3.3.8, z corresponds to a point  $x = (y = (x^p, x_p), \underline{x}, 1) \in \mathcal{E}_{\Omega,\lambda}(\overline{\rho})$ . We have  $\Pi_{\infty}^{R_{\infty}-\mathrm{an}}[\mathfrak{m}_y] \cong \widehat{S}(U^p, E)^{\mathrm{an}}[\mathfrak{m}_{\rho}]$ . By similar arguments as above [17, Lemma 4.6],  $x^p$  is a smooth point of  $\mathfrak{X}_{\infty}^p$ . By Corollary 7.2.18 and Lemma 7.1.6, it follows that there is an embedding

$$\widehat{\otimes}_{v \in S_p} C(w_{\widetilde{v}}, \mathscr{F}_{\widetilde{v}}) \longleftrightarrow \Pi_{\infty}^{R_{\infty} - \operatorname{an}}[\mathfrak{m}_y] \cong \widehat{S}(U^p, E)^{\operatorname{an}}[\mathfrak{m}_{\rho}]$$

if and only if  $w_{\tilde{v}} \leq w_{\mathscr{F}_{\tilde{v}}} w_{0,F_{\tilde{v}}}$  for all  $v \in S_p$  (noting that  $(w_{\mathscr{F}_{\tilde{v}}}) = w_y$ ).

**Remark 7.2.21.** (1) By (the proof of) Theorem 7.2.9, the assumption (2) in Theorem 7.2.20 can also be replaced by the assumptions (2), (3), (4) in Theorem 7.2.9, the assumption (4) being for  $P = \prod_{v \in S_p} P_{\tilde{v}}$  associated to  $\{\mathscr{F}_{\tilde{v}}\}$ .

(2) By the same argument but using Corollary 7.2.17 instead of Corollary 7.2.18, we can prove

 $\widehat{\otimes}_{v \in S_p} C(w_{\widetilde{v}}, \mathscr{F}_{\widetilde{v}}) \longleftrightarrow \Pi_{\infty}^{R_{\infty} - \mathrm{an}}[\mathfrak{m}_y] \cong \widehat{S}(U^p, E)^{\mathrm{an}}[\mathfrak{m}_\rho]$ 

if  $w_{\widetilde{v}} \leq w_{\mathscr{F}_{\widetilde{v}}} w_{0,F_{\widetilde{v}}}$  for all  $v \in S_p$  and  $w = (w_{\widetilde{v}})$  satisfies the conditions in Corollary 7.2.17.

In particular, we have the following special case of Theorem 7.2.20:

Corollary 7.2.22. Assume that we are in the setting of Conjecture 3.2.18, and that:

- (1) Hypothesis 3.3.1 holds;
- (2)  $\widehat{S}(U^p, E)^{\text{lalg}}[\mathfrak{m}_{\rho}] \neq 0;$
- (3) for all  $v \in S_p$ , each  $r(\rho_{\tilde{v}})$  has at most one irreducible constituent of dimension > 1.

For  $v \in S_p$ , let  $\mathscr{F}_{\widetilde{v}}$  be a minimal parabolic filtration of  $\mathbf{r}(\rho_{\widetilde{v}})$ . Then Conjecture 3.2.18 holds for  $\mathscr{F} = (\mathscr{F}_{\widetilde{v}})$ .

**Remark 7.2.23.** The case where all irreducible constituents of  $\mathbf{r}(\rho_{\tilde{v}})$  are 1-dimensional (i.e. the (generic) crystabelline case) was proved in [19, Thm. 5.3.3].

# A Appendix

### A.1 $\Omega$ -filtration in families

Generalizing [60, § 6.3] we show that  $\Omega$ -filtrations on  $(\varphi, \Gamma)$ -modules (cf. § 4.1.2) can interpolate in *p*-adic families. The results in this section may be viewed as a parabolic analogue of the (so-called) global triangulations.

The following theorem generalizes [60, Thm. 6.3.9].

**Theorem A.1.1.** Let  $n \ge r \in \mathbb{Z}_{\ge 1}$ . Let X be a reduced rigid analytic space over E, M a  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{X,L}$  of rank n, and  $\Delta$  a  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{X,L}$  of rank r such that, for any point x of X,  $\Delta_x$  is irreducible and de Rham of constant Hodge-Tate weight 0. Let  $\mathbf{h} = (\mathbf{h}_{\tau})_{\tau \in \Sigma_L} = (h_{i,\tau})_{\substack{i=1,...,r\\ \tau \in \Sigma_L}}$  be an integral positive strictly dominant weight (i.e.  $h_{1,\tau} > h_{2,\tau} > \cdots > h_{r,\tau} \ge 0$  for all  $\tau \in \Sigma_L$ ). Assume that there exists a Zariski-dense subset  $X_{alg}$  of points of X such that for all  $x \in X_{alg}$ :

- (1)  $H^0_{(\omega,\Gamma)}(\Delta_x \otimes_{\mathcal{R}_{k(x),L}} M_x^{\vee})$  is one-dimensional over k(x);
- (2) Im  $\eta_x$  is de Rham of Hodge-Tate weights  $\mathbf{h}$  for  $0 \neq \eta_x \in \operatorname{Hom}_{(\varphi,\Gamma)}(M_x, \Delta_x) \cong H^0_{(\varphi,\Gamma)}(\Delta_x \otimes_{\mathcal{R}_{k(x),L}} M_x^{\vee}).$

Then there exist

- a proper birational surjective morphism  $f: X' \to X$ ;
- a unique homomorphism  $\eta: f^*M \to f^*\Delta$  (up to multiplication by  $\mathcal{O}_{X'}^{\times}$ )

such that the following properties are satisfied:

- (1) the set Z of closed points  $x \in X'$  failing to have properties (a) and (b) below is Zariski-closed and disjoint from  $f^{-1}(X_{alg})$ :
  - (a)  $\operatorname{Hom}_{(\varphi,\Gamma)}(M_x, \Delta_x)$  is one-dimensional and generated by the pull-back  $\eta_x : M_x \to \Delta_x$  of  $\eta_j$ ;
  - (b)  $\operatorname{Im}(\eta_x)$  is de Rham of Hodge-Tate weights h;
- (2) the kernel of  $\eta$  is a  $(\varphi, \Gamma)$ -module over  $\mathcal{R}_{X',L}$  of rank n-r;
- (3) the cokernel of  $\eta$  is locally t-torsion;
- (4) Im  $\eta|_{X'\setminus Z}$  is a  $(\varphi, \Gamma)$ -module of rank r, and for all  $x \in X' \setminus Z$ , the induced morphism  $(\operatorname{Im} \eta)_x \to \Delta_{f(x)}$  is injective.

*Proof.* The theorem follows by a variation of the proof of [60, Thm. 6.3.9]. We include a proof for the convenience of the reader. One main difference from *loc. cit.* is that in our case, for any  $x \in X_{\text{alg}}$ ,  $\text{Im}(\eta_x)$  is not saturated in  $\Delta_x$  except when r = 1. As in the first paragraph of *loc. cit.*, we are reduced to the case where X is reduced, normal and connected (thus any coherent sheaf over X has constant generic rank).

Step 1. By [60, Thm. 4.4.5 (1)] and [60, Thm.], the complex  $\mathcal{C}^{\bullet}_{(\varphi,\Gamma)}(\Delta \otimes_{\mathcal{R}_{X',L}} M^{\vee})$  is perfect and concentrated in degree [0, 2], i.e. is quasi-isomorphic to  $[P_0 \to P_1 \to P_2]$  where the  $P_i$  are locally free  $\mathcal{O}_X$ -modules of finite ranks. By [60, Cor. 6.3.6 (2)], we can obtain a proper birational morphism  $f_0: X'_0 \to X$  with  $X'_0$  reduced and normal (and irreducible) such that

- $H^0_{(\varphi,\Gamma)}(f^*_0(\Delta \otimes_{\mathcal{R}_{X,L}} M^{\vee}))$  is flat;
- $H^i_{(\omega,\Gamma)}(f^*_0(\Delta \otimes_{\mathcal{R}_{X,L}} M^{\vee}))$  has Tor-dimension less than or equal to 1 for i = 1, 2.

Note that  $f_0^{-1}(X_{\text{alg}})$  is Zariski-dense in  $X'_0$ . Using the condition (1) and the above flatness, we see that  $\mathcal{L} := H^0_{(\varphi,\Gamma)}(f_0^*(\Delta \otimes_{\mathcal{R}_{X,L}} M^{\vee}))$  is locally free of rank 1 over  $\mathcal{O}_{X'_0}$ . We have thus a tautological morphism of  $(\varphi, \Gamma)$ -modules over  $\mathcal{R}_{X'_0,L}$ :

$$\eta: f_0^* M \otimes_{\mathcal{O}_{X_0'}} \mathcal{L} \longrightarrow f_0^* \Delta. \tag{A.1}$$

It is clear that, for any  $x \in X'_0$ , the induced morphism  $\eta_x : f_0^* M_x \to f_0^* \Delta_x$  is non-zero. We have actually an exact sequence as in [60, (6.3.9.1)]:

$$\begin{split} 0 &\to H^0_{(\varphi,\Gamma)}(f_0^*(\Delta \otimes_{\mathcal{R}_{X',L}} M^{\vee})) \otimes_{\mathcal{O}_{X'}} k(x) \to H^0_{(\varphi,\Gamma)}(f_0^*(\Delta \otimes_{\mathcal{R}_{X',L}} M^{\vee}) \otimes_{\mathcal{O}_{X'}} k(x)) \\ &\to \operatorname{Tor}_1^{X'_0}\left(H^1_{(\varphi,\Gamma)}(f_0^*(\Delta \otimes_{\mathcal{R}_{X',L}} M^{\vee})), k(x)\right) \to 0 \end{split}$$

We deduce that the set  $Z_0$  of points  $x \in X'_0$  such that  $H^0_{(\varphi,\Gamma)}(f^*_0(\Delta \otimes_{\mathcal{R}_{X',L}} M^{\vee})) \otimes_{\mathcal{O}_{X'}} k(x)$  is not isomorphic to  $H^0_{(\varphi,\Gamma)}(f^*_0(\Delta \otimes_{\mathcal{R}_{X',L}} M^{\vee}) \otimes_{\mathcal{O}_{X'}} k(x))$  is Zariski-closed in  $X'_0$  (with its complement given by the flat locus of the coherent sheaf  $H^1_{(\varphi,\Gamma)}(f^*_0(\Delta \otimes_{\mathcal{R}_{X',L}} M^{\vee}))$ ). Step 2. Let Q be the cokernel of  $\eta$ . For any point  $x \in X'_0$ , since  $\Delta_x$  is irreducible and  $\eta_x$  is non-zero, we deduce that  $Q_x = \operatorname{Coker} \eta_x$  is t-torsion. Let  $\operatorname{Spm} A$  be an affinoid open of X'. By [60, Lemma 2.2.9], the base change of the morphism  $\eta$  via  $\operatorname{Spm} A \to \mathcal{R}_{X',L}$  admits a model  $\epsilon_A^r : M_A^r \to \Delta_A^r$  and the cokernel  $Q_A^r$  is a model of  $Q_A$  (the restriction of Q to  $\mathcal{R}_{A,L}$ ) (where we use the notation of [60, § 2] for  $M_A^r$  etc.). We have that  $Q_A^{(r/p,r]}$  is a finitely presented module over  $\mathcal{R}_{A,L}^{(r/p,r]}$ . Since  $Q_x = \operatorname{Coker} \eta_x$  is t-torsion for all  $x \in X'$ , we see  $Q_A^{(r/p,r]}$  is supported on the zero locus of t in  $\operatorname{Spm} \mathcal{R}_{A,L}^{(r/p,r]}$ , which is a rigid analytic space finite over A. We deduce that  $Q_A^{(r/p,r]}$  is annihilated by  $t^N$  for some  $N \in \mathbb{Z}_{>0}$ . Since  $Q_A^{(r/p^m,r/p^{m-1}]} \cong (\varphi^{m-1})^* Q_A^{(r/p,r]}$ , we deduce that  $Q_A^r \cong \prod_{m \in \mathbb{Z}_{\geq 1}} Q_A^{(r/p^m,r/p^{m-1}]}$  is also annihilated by  $t^N$  (note that  $Q_A^r$  is coadmissible by [60, Lemma 2.1.4 (5)]). Applying [60, Cor. 6.3.6] to the finitely generated A-module  $Q_A^{(r/p,r]}$ , we obtain  $g : \operatorname{Spm} A' \to \operatorname{Spm} A$  such that  $g^* Q_A^{(r/p,r)}$  has Tor-dimension less than 1. Using the  $\varphi$ -action as above, we deduce that  $g^* Q_A^{(r/p^m,r/p^{m-1}]}$  also has Tor-dimension less than 1, and hence that  $Q_{A'}^r := g^* Q_A^r \cong \prod_{m \in \mathbb{Z}_{\geq 1}} g^* Q_A^{(r/p^m,r/p^{m-1}]}$  has Tor-dimension less than 1. The morphisms  $g : A' \to A$  (with A varying) glue to a birational projective morphism  $g : X' \to X'_0$  such that  $Q_{X'}^r := g^* Q$  has Tor-dimension less than 1, and we finally obtain  $f : X' \xrightarrow{g} X'_0 \to X$ . By the above discussion,  $Q_{X'}^r$  is locally t-torsion.

Step 3. We have an exact sequence (by pulling-back (A.1) via g, and where we use  $\mathcal{L}$  to denote the pull-back of  $\mathcal{L}$  in (A.1)):

$$f^*M \otimes_{\mathcal{O}_{X'}} \mathcal{L} \xrightarrow{\eta} f^*\Delta \to Q_{X'} \to 0.$$

Specializing to a point  $x \in X'$  and using the fact that  $Q_{X'}$  has Tor-dimension less than 1, we get

$$0 \to \operatorname{Tor}_{1}^{X'}(Q_{X'}, k(x)) \to \operatorname{Im}(\eta)_{x} \to (f^{*}\Delta)_{x} \to Q_{X',x} \to 0,$$
(A.2)

and  $\operatorname{Tor}_{i}^{X'}(\operatorname{Im}(\eta), k(x)) = 0$  for i > 0. Then, specializing  $0 \to \operatorname{Ker}(\eta) \to (f^*M \otimes_{\mathcal{O}_{X'}} \mathcal{L}) \to \operatorname{Im}(\eta) \to 0$  to x, we get

$$0 \to \operatorname{Ker}(\eta)_x \to (f^*M)_x \to \operatorname{Im}(\eta)_x \to 0.$$
(A.3)

Since  $Q_{X'}$  is locally t-torsion, both  $\operatorname{Tor}_1^{X'}(Q_{X'}, k(x))$  and  $Q_{X',x}$  are annihilated by a certain power of t. We deduce that  $\operatorname{Im}(\eta_x)$  is a  $(\varphi, \Gamma)$ -module of rank r over  $\mathcal{R}_{k(x),L}$ , and  $\operatorname{Ker}(\eta_x)$ is a  $(\varphi, \Gamma)$ -module of rank n - r over  $\mathcal{R}_{k(x),L}$  (recall that  $\eta_x$  denotes the induced morphism  $(f^*M)_x \to (f^*\Delta)_x$ ). We also deduce from (A.2) and (A.3):

$$0 \to \operatorname{Ker}(\eta)_x \to \operatorname{Ker}(\eta_x) \to \operatorname{Tor}_1^{X'}(Q_{X'}, k(x)) \to 0.$$

Since  $\operatorname{Tor}_{1}^{X'}(Q_{X'}, k(x))$  is t-torsion, it follows that  $\operatorname{Ker}(\eta)_{x}$  is a  $(\varphi, \Gamma)$ -module of rank n - r for all  $x \in X'$ . By the same argument as in the last paragraph of the proof of [60, Thm. 6.3.9], we deduce that  $\operatorname{Ker}(\eta)$  is a  $(\varphi, \Gamma)$ -module of rank n - r (in the sense of [60, Def. 2.2.12]).

Step 4. We prove that  $U := \{x \in X' \mid \operatorname{Tor}_1^{X'}(Q_{X'}, k(x)) = 0\}$  is Zariski-open and contains  $f^{-1}(X_{\text{alg}})$ . Let Spm A be an affinoid in X' and  $Q_A^r$  be a model of  $Q_A := Q_{X'}|_{\text{Spm}A}$ . For  $x \in \operatorname{Spm} A$ , by the same argument as in Step 2, we see that the following statements are equivalent:

- $Q_A^{(r/p,r]}$  is flat over A at x;
- $Q_A^r$  is flat over A at x.

Since  $Q_{X'}$  has Tor-dimension less than 1 (by Step 2),  $U \cap \text{Spm} A$  is exactly the flat locus of  $Q_A^{(r/p,r]}$ . We deduce that U is Zariski-open. Now assume  $f^{-1}(X_{\text{alg}}) \cap \text{Spm} A$  is Zariski-dense in Spm A (noting that such Spm A can cover X'). For each  $x \in f^{-1}(X_{\text{alg}}) \cap \text{Spm} A$ , by the assumption (2) in the theorem,  $\text{Im}(\eta_x) \subset \Delta_{f(x)}$  is de Rham of Hodge-Tate weights **h**. We deduce that  $Q_A^{(r/p,r]} \otimes_A k(x)$  has constant dimension (determined by **h**). Indeed, using [6, Thm. A], we can deduce that  $Q_A^r \otimes_A k(x) = \Delta_{f(x)}^r / \text{Im}(\eta_x)^r$  admits a filtration by  $\mathcal{R}_{k(x),L}^r$ -submodules of graded pieces given by  $\{\mathcal{R}_{k(x),L}^r / \prod_{\tau \in \Sigma_L} t_{\tau}^{h_{i,\tau} - h_{i+1,\tau}}\}_{i=1,\dots,r}$  where  $t_{\tau} \in \mathcal{R}_{k(x),L}$  is the element defined in [60, Not. 6.2.7], and  $h_{r+1,\tau} := 0$  for all  $\tau$ . Since  $\mathcal{R}_{k(x),L}^r / \prod_{\tau \in \Sigma_L} t_{\tau}^{h_{i,\tau} - h_{i+1,\tau}}\}_{i=1,\dots,r}$  where  $t_{\tau} \in \mathcal{R}_{k(x),L}$ , we see that  $Q^{(r/p,r]} \otimes_A k(x)$  admits a filtration with graded pieces given by  $\{\mathcal{R}_{k(x),L}^r / \prod_{\tau \in \Sigma_L} t_{\tau}^{h_{i,\tau} - h_{i+1,\tau}}\}_{i=1,\dots,r}$ . Hence  $\dim_{k(x)} Q + A^{(r/p,r]} \otimes_A k(x)$  is constant for  $x \in f^{-1}(X_{\text{alg}}) \cap \text{Spm} A$ . Together with the fact  $f^{-1}(X_{\text{alg}}) \cap \text{Spm} A$  is Zariski-dense in Spm A, we deduce that  $Q_A^{(r/p,r]}$  is locally free at points in  $f^{-1}(X_{\text{alg}}) \cap \text{Spm} A$ . In particular,  $f^{-1}(X_{\text{alg}}) \cap \text{Spm} A$ .

Step 5. Consider the restriction D of  $\operatorname{Im} \eta$  on U (see Step 4). Specializing the exact sequence

$$0 \to D \to (f^*\Delta)_U \to Q_U \to 0$$

at each point  $x \in U$ , we get

$$0 \to D_x \to (f^*\Delta)_x \to Q_x \to 0.$$

In particular,  $D_x$  is a  $(\varphi, \Gamma)$ -module of rank r. By the same argument as in the last paragraph of the proof of [60, Thm. 6.3.9], we deduce that D is a  $(\varphi, \Gamma)$ -module ([60, Def. 2.2.12]). For  $x \in U$ , since  $(f^*\Delta)_x$  is de Rham,  $D_x$  is de Rham as well. Since  $f^{-1}(X_{\text{alg}})$  is Zariski-dense in U, by interpolating the Sen weights (see for example [60, Def. 6.2.11]),  $D_x$  has Sen weights **h**. In summary,  $D_x$  is de Rham of Hodge-Tate weights **h** for all  $x \in U$ . The theorem follows by taking Z to be the union of the complement of U and  $g^{-1}(Z_0)$  (see in particular Step 3 for the properties (2) and (3), Step 1 for (1)(a), and Step 4 & 5 for (1)(b) and (4)).

Note that in Theorem A.1.1  $f^{-1}(X_{alg})$  is Zariski-dense in  $X' \setminus Z$ .

**Corollary A.1.2.** Let X be a reduced rigid analytic space over E. Let  $n_i \in \mathbb{Z}_{\geq 1}$  for  $1 \leq i \leq r$ and  $n := \sum_{i=1}^{r} n_i$ . Assume we have the following data:

- (1) a  $(\varphi, \Gamma)$ -module M over  $\mathcal{R}_{X,L}$  of rank n;
- (2) for each i, a  $(\varphi, \Gamma)$ -module  $\Delta_i$  over  $\mathcal{R}_{X,L}$  of rank  $n_i$  such that for all  $x \in X$ ,  $\Delta_{i,x}$  is irreducible and de Rham of constant Hodge-Tate weight 0;
- (3) for each i, a continuous character  $\delta_i$  of  $L^{\times}$  over X;
- (4) for each *i*, an integral positive strictly dominant weight  $\mathbf{h}_i = (\mathbf{h}_i)_{\tau \in \Sigma_L} = (h_{i,j,\tau})_{\substack{j=1,...,n_i \\ \tau \in \Sigma_L}}$  (i.e.  $h_{i,1,\tau} > \cdots > h_{i,n_i,\tau} \ge 0$  for all  $\tau \in \Sigma_L$ );
- (5) a Zariski-dense subset  $X_{alg}$  of closed points of X such that for all  $x \in X_{alg}$ ,  $M_x$  admits a filtration  $0 = \operatorname{Fil}_0 M_x \subsetneq \operatorname{Fil}_1 M_x \subsetneq \cdots \subsetneq \operatorname{Fil}_r M_x = M_x$  satisfying
  - $\dim_{k(x)} \operatorname{Hom}_{(\varphi,\Gamma)}(\operatorname{Fil}_{i} M_{x}, \Delta_{i,x} \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x),L}(\delta_{i,x})) = 1;$
  - for any non-zero morphism of  $(\varphi, \Gamma)$ -modules  $\eta_{i,x}$ : Fil<sub>i</sub>  $M_x \to \Delta_x \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x),L}(\delta_{i,x})$ , Im  $\eta_x \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x),L}(\delta_{i,x}^{-1})$  is de Rham of Hodge-Tate weights  $h_i$ .

Then there exist

- a proper birational surjective morphism  $f: X' \to X$  of reduced rigid analytic spaces;
- a filtration  $0 = \operatorname{Fil}_0(f^*M) \subsetneq \cdots \subsetneq \operatorname{Fil}_r(f^*M) = f^*M$  on  $f^*M$  by  $(\varphi, \Gamma)$ -submodules over  $\mathcal{R}_{X',L}$ ;
- line bundles  $\mathcal{L}_i$  over X' for  $1 \leq i \leq r$ ;
- unique morphisms  $\eta_i$ : Fil<sub>i</sub> $(f^*M) \otimes_{\mathcal{O}_{X'}} \mathcal{L}_i \to f^*(\Delta_i \otimes_{\mathcal{R}_{X,L}} \mathcal{R}_{X,L}(\delta_i))$  for  $1 \leq i \leq r$ ;
- a Zariski-closed subset Z of X disjoint from  $X_{alg}$

such that the following properties are satisfied

- (1) for each *i*, the cohernel of  $\eta_i$  is locally t-torsion and  $\operatorname{Ker} \eta_i = \operatorname{Fil}_{i-1}(f^*M) \otimes_{\mathcal{O}_{X'}} \mathcal{L}_i$ ;
- (2) the  $(\operatorname{Fil}^{i} f^{*}M)|_{X'\setminus Z}$  are direct summands of  $f^{*}M|_{X'\setminus Z}$  as  $\mathcal{R}_{X'\setminus Z,L}$ -modules;
- (3) for all  $x \in X' \setminus Z$ , the k(x)-vector space

$$\operatorname{Hom}_{(\varphi,\Gamma)}\left((\operatorname{Fil}_{i}f^{*}M)_{x},\Delta_{i,f(x)}\otimes_{\mathcal{R}_{k(x),L}}\mathcal{R}_{k(x),L}(\delta_{i,f(x)})\right)$$

is one dimensional generated by the pull-back of  $\eta_i$ , and the  $(\varphi, \Gamma)$ -module  $\operatorname{Im}(\eta_{i,x}) \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x),L}(\delta_{i,k(x)}^{-1})$  is de Rham of Hodge-Tate weights  $h_i$  for  $1 \leq i \leq r$ .

*Proof.* The corollary follows from Theorem A.1.1 by induction. We first apply Theorem A.1.1 to  $(M \otimes_{\mathcal{R}_{X,L}} \mathcal{R}_{X,L}(\delta_r^{-1}), \Delta_r, X)$  (= the data  $(M, \Delta, X)$  in the theorem) to obtain the data:

$$(f_r: X'_r \to X, \ Z_r \subset X'_r, \ \eta_r: f_r^*(M \otimes_{\mathcal{R}_{X,L}} \mathcal{R}_{X,L}(\delta_r^{-1})) \to f_r^*\Delta_r)$$

corresponding respectively to  $f, Z, \eta$  in Theorem A.1.1. Next we apply Theorem A.1.1 to  $(\text{Ker } \eta_r \otimes_{\mathcal{R}_{X'_{r,L}}} \mathcal{R}_{X'_{r,L}}(\delta_r \delta_{r-1}^{-1}), f_r^* \Delta_{r-1}, X'_r)$  to obtain  $(f_{r-1} : X'_{r-1} \to X'_r, Z_{r-1} \subset X'_{r-1}, \eta_{r-1})$ . By iterating this argument, we finally get  $f : X' := X'_1 \xrightarrow{f_1} X'_2 \to \cdots \to X'_r \xrightarrow{f_r} X$  and we put Z to be the union of the pull-backs of the  $Z_i$ 's. The corollary follows.

**Corollary A.1.3.** Keep the setting of Corollary A.1.2 and let  $x \in X'$ .

(1) We have that  $(f^*M)_x \cong M_{f(x)}$  admits a filtration  $0 = \operatorname{Fil}_0 M_{f(x)} \subsetneq \operatorname{Fil}_1 M_{f(x)} \subsetneq \cdots \subsetneq$  $\operatorname{Fil}_r M_{f(x)} = M_{f(x)}$  by saturated  $(\varphi, \Gamma)$ -submodules of  $M_{f(x)}$  such that  $\operatorname{gr}_i M_{f(x)}[\frac{1}{t}] \cong \Delta_{i,x} \otimes_{k(x)} \mathcal{R}_{k(x),L}(\delta_{i,x})[\frac{1}{t}]$ .

(2) Let  $A \in \operatorname{Art}(E)$  and  $\operatorname{Spec} A \to X'$  be a morphism sending the only point of  $\operatorname{Spec} A$  to x. Let  $M_A$  denote the pullback of  $f^*M$  along  $\operatorname{Spec} A \to X'$ . Then  $M_A := M_A[\frac{1}{t}]$  admits a filtration  $0 = \operatorname{Fil}_0 M_A \subsetneq \operatorname{Fil}_1 M_A \subsetneq \cdots \subsetneq \operatorname{Fil}_r M_A = M_A$  by  $(\varphi, \Gamma)$ -modules over  $\mathcal{R}_{A,L}[\frac{1}{t}]$  such that  $\operatorname{gr}_i M_A \cong \Delta_{i,A} \otimes_{\mathcal{R}_{A,L}} \mathcal{R}_{A,L}(\delta_{i,A})[\frac{1}{t}]$ .

*Proof.* Specializing the objects in Corollary A.1.2 at the point x, we have  $(\varphi, \Gamma)$ -modules over  $\mathcal{R}_{k(x),L}$ :

 $\{(\operatorname{Fil}_i f^* M)_x = (\operatorname{Ker} \eta_{i+1})_x\}_{i=1,\dots,r-1}, \ \{\operatorname{Ker} \eta_{i,x}\}_{i=1,\dots,r}, \ \{\Delta_{i,f(x)}\}_{i=1,\dots,r}$ 

where (using the fact that Coker  $\eta_{i,x}$  is t-torsion):

$$\operatorname{rk}_{\mathcal{R}_{k(x),L}}(\operatorname{Fil}_{i}f^{*}M)_{x} = \operatorname{rk}_{\mathcal{R}_{k(x),L}}(\operatorname{Ker}\eta_{i+1})_{x} = \sum_{j=1}^{i} n_{j} \text{ and } \operatorname{rk}_{\mathcal{R}_{k(x),L}}\operatorname{Ker}(\eta_{i,x}) = \sum_{j=1}^{i-1} n_{j}$$

We also have morphisms of  $(\varphi, \Gamma)$ -modules over  $\mathcal{R}_{k(x),L}$ :

$$\begin{cases} \operatorname{Ker}(\eta_i)_x \longrightarrow \operatorname{Ker}(\eta_{i,x}) & i = 1, \dots, r\\ \operatorname{Ker}(\eta_{i,x}) \hookrightarrow (\operatorname{Fil}_i f^* M)_x \cong \operatorname{Ker}(\eta_{i+1})_x & i = 1, \dots, r-1. \end{cases}$$

We have that  $\operatorname{Ker}(\eta_{i,x})$  is saturated in  $(\operatorname{Fil}_i f^*M)_x$  for all *i*. Since  $\operatorname{Coker} \eta_i$  is locally *t*-torsion, by similar arguments as in Step 3 of the proof of Theorem A.1.1, we see that the cokernel of the morphism  $\operatorname{Ker}(\eta_i)_x \to \operatorname{Ker}(\eta_{i,x})$  is *t*-torsion, and hence this morphism is injective (as both source and target have the same rank over  $\mathcal{R}_{k(x),L}$ ). We take thus  $\operatorname{Fil}_r M_{f(x)} := M_{f(x)}$  and let  $\operatorname{Fil}_i M_{f(x)}$ be the saturation of the image of the composition (note that some of the injections may not be saturated, for example the first one):

$$(\operatorname{Fil}_{i} f^{*}M)_{x} \cong \operatorname{Ker}(\eta_{i+1})_{x} \hookrightarrow \operatorname{Ker}(\eta_{i+1,x}) \hookrightarrow (\operatorname{Fil}_{i+1} f^{*}M)_{x}$$
$$\hookrightarrow \operatorname{Ker}(\eta_{i+2,x}) \cdots \hookrightarrow \operatorname{Ker}(\eta_{r,x}) \hookrightarrow M_{f(x)}.$$

Since  $\operatorname{Ker}(\eta_{i+1})_x/\operatorname{Ker}(\eta_{i,x}) = (\operatorname{Fil}_i f^*M)_x/\operatorname{Ker}(\eta_{i,x}) \hookrightarrow \Delta_{i,x} \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x),L}(\delta_{i,x})$  with *t*-torsion cokernel (both source and target having the same rank), we deduce

$$\operatorname{gr}_{i} M_{f(x)}\left[\frac{1}{t}\right] \cong \operatorname{Ker}(\eta_{i+1})_{x}\left[\frac{1}{t}\right] / \operatorname{Ker}(\eta_{i,x})\left[\frac{1}{t}\right] \cong \Delta_{i,x} \otimes_{\mathcal{R}_{k(x),L}} \mathcal{R}_{k(x),L}(\delta_{i,x})\left[\frac{1}{t}\right].$$

Part (1) follows. Part (2) follows by similar arguments.

### A.2 Characteristic cycles of parabolic Verma modules

We prove Proposition 5.4.8 by a generalization of Ginzburg's method ( $[45, \S 6.3]$ ).

We let  $B \subset P \subset G$  be as in § 5, and we use without comment the notation there. For a smooth algebraic variety X over E, we denote by  $D_X$  the sheaf of differential operators on X, see for instance [54, § 1.1]. Recall that  $\mathcal{O}_X$  is equipped with a canonical left  $D_X$ -module structure. We will use below the notation of [54].

For  $w \in \mathcal{W}$ , we define  $C_w := BwB/B$  and  $C_w^P := PwB/B$ , which are locally closed (smooth) subschemes of G/B, and  $C_{P,w} := BwP/P$ , which is a locally closed (smooth) subscheme of G/P. So  $C_w^P$  (resp.  $C_{P,w}$ ) only depends on the image of  $\mathcal{W}$  in  $\mathcal{W}_{L_P} \setminus \mathcal{W}$  (resp. in  $\mathcal{W}/\mathcal{W}_{L_P}$ ). Denote by  $\overline{C_w}$ , (resp.  $\overline{C_w^P}$ , resp.  $\overline{C_{P,w}}$ ) the Zariski closure of  $C_w$  (resp.  $C_w^P$ , resp.  $C_{P,w}$ ) in G/B (resp. G/B, resp. G/P). We have

$$G/B = \sqcup_{w \in \mathscr{W}} C_w = \sqcup_{w \in \mathscr{W}_{L_P} \setminus \mathscr{W}} C_w^P \text{ and } G/P = \sqcup_{w \in \mathscr{W}/\mathscr{W}_{L_P}} C_{P,u}$$

and we remark that all these schemes are over  $\operatorname{Spec} E$ .

For  $w \in \mathscr{W}$ , consider  $j_w : C_w^P \hookrightarrow G/B$ . By an easy variation of [81, Lemma 5.1] and [81, Rem. 5.2] to the case G/B, the open immersion  $C_w^P \hookrightarrow \overline{C_w^P}$  is affine, hence so is  $j_w$ . Consider the direct image  $\int_{j_w} \mathcal{O}_w^P = R(j_w)_* (D_{G/B \leftarrow C_w^P} \otimes_{D_{C_w^P}}^L \mathcal{O}_w^P) \in D_{qc}^b(D_{G/B})$  (cf. [54, § 1.5]). As  $j_w$  is a

locally closed immersion (i.e. the composition of an open immersion and a closed immersion), by [54, Ex. 1.3.2] and [54, Ex. 1.5.12], we deduce that  $D_{G/B \leftarrow C_w^P}$  is locally free over  $D_{C_w^P}$ . Together with the fact that  $j_w$  is affine, we have  $\int_{j_w} \mathcal{O}_w^P \cong (j_w)_* (D_{G/B \leftarrow C_w^P} \otimes_{D_{C_w^P}} \mathcal{O}_{C_w^P}) =: \mathfrak{N}_w^P$ , which is a *P*-equivariant coherent  $D_{G/B}$ -module, hence by [54, Thm. 11.6.1 (i)] a *P*-equivariant holonomic regular  $D_{G/B}$ -module. The following proposition is may-be well-known to experts, but we couldn't find a reference.

**Proposition A.2.1.** We have  $\Gamma(G/B, \mathfrak{N}_w^P) \cong M_P(w^{\max}w_0 \cdot 0)^{\vee}$  where  $(-)^{\vee}$  denotes the dual in the BGG category  $\mathcal{O}^{\mathfrak{b}}$  (cf. [57, § 3.2]).

*Proof.* We have  $C_w^P = \bigsqcup_{u \in \mathscr{W}_{L_P}} C_{uw^{\max}}$ . For  $u \in \mathscr{W}_{L_P}$ , we have  $\lg(uw^{\max}) = \lg(w^{\max}) - \lg(u)$ . Indeed, let  $\Phi^+$  (resp.  $\Phi^-$ ) be the set of positive (resp. negative) roots with respect B, then for  $w' \in \mathscr{W}$ ,  $\lg(w') = |\{\alpha \in \Phi^+ \mid w'(\alpha) \in \Phi^-\}|$ . As  $w^{\max}$  has maximal length in  $\mathscr{W}_{L_P}w$ ,  $w^{\max}(\Phi^+) \cap \Phi^+_{L_P} = \emptyset$  hence  $\lg(w^{\max}) = \left|w^{\max}(\Phi^+) \cap (\Phi^- \setminus \Phi^-_{L_P})\right| + \left|\Phi^-_{L_P}\right|$ . For  $u \in \mathscr{W}_{L_P}$ ,  $u \in \mathscr{W}_{L_P}$ ,  $u \in \mathscr{W}_{L_P}$ preserves the sets  $\Phi^+ \setminus \Phi^+_{L_P}$  and  $\Phi^- \setminus \Phi^-_{L_P}$ , hence

$$\begin{aligned} \lg(uw^{\max}) &= |uw^{\max}(\Phi^+) \cap \Phi^-| &= |uw^{\max}(\Phi^+) \cap (\Phi^- \setminus \Phi^-_{L_P})| + |u(\Phi^-_{L_P}) \cap \Phi^-_{L_P}| \\ &= |w^{\max}(\Phi^+) \cap (\Phi^- \setminus \Phi^-_{L_P})| + |\Phi^-_{L_P}| - |u(\Phi^+_{L_P}) \cap \Phi^-_{L_P}| &= \lg(w^{\max}) - \lg(u). \end{aligned}$$

Let  $d := |\mathscr{W}_{L_P}|$ . For  $k = 0, \ldots, d$ , denote by  $C_w^k := \bigsqcup_{u \in \mathscr{W}_{L_P}} C_{uw^{\max}}$  and  $Z_w^k := \bigsqcup_{u \in \mathscr{W}_{L_P}} C_{uw^{\max}}$ . By  $\underset{lg(u) \leq k}{\underset{lg(u) = k}{\underset$ of *loc. cit.*, for instance see page 33 of [54] for  $i_k^{\dagger}$ ,  $j_k^{\dagger}$ ):

$$\int_{i_k} i_k^{\dagger} \mathcal{O}_{C_w^k} \to \mathcal{O}_{C_w^k} \to \int_{j_k} j_k^{\dagger} \mathcal{O}_{C_w^k} \xrightarrow{+1} .$$
(A.4)

Applying  $\int_j = Rj_*$  to (A.4) (for the open immersion  $j: C_w^k \hookrightarrow C_w^P$ , cf. [54, Ex. 1.5.22]), we obtain distinguished triangles in  $D^b_{\mathrm{qc}}(D_{C^P_w})$ :

$$\int_{j} \int_{i_{k}} i_{k}^{\dagger} \mathcal{O}_{C_{w}^{k}} \to \int_{j} \mathcal{O}_{C_{w}^{k}} \to \int_{j} \int_{j_{k}} j_{k}^{\dagger} \mathcal{O}_{C_{w}^{k}} \xrightarrow{+1} .$$

We have by definition  $i_k^{\dagger} \mathcal{O}_{C_w^k} \cong \mathcal{O}_{Z_w^k}[k]$ . Indeed, by the discussion below [54, Def. 1.3.1], we can deduce that the derived inverse image of  $\mathcal{O}_{C_w^k}$  via  $i_k$  as  $D_{C_w^k}$ -module is the same as the derived inverse image of  $\mathcal{O}_{C_k^k}$  via  $i_k$  as  $\mathcal{O}_{C_k^k}$ -module (which is  $\mathcal{O}_{Z_k^k}$  as  $\mathcal{O}_{C_k^k}$  is obviously flat over  $\mathcal{O}_{C_k^k}$ ). Using [54, Prop. 1.5.21] and [54, Prop. 1.5.24], we deduce

$$\int_{j} \int_{i_{k}} i_{k}^{\dagger} \mathcal{O}_{C_{w}^{k}} \cong \int_{i} \mathcal{O}_{Z_{w}^{k}}[k] \cong i_{*}(D_{C_{w}^{P} \leftarrow Z_{w}^{k}} \otimes_{D_{Z_{w}^{k}}} \mathcal{O}_{Z_{w}^{k}})[k].$$

We also have  $\int_j \int_{j_k} j_k^{\dagger} \mathcal{O}_{C_w^k} \cong \int_j j_k^{\dagger} \mathcal{O}_{C_w^k} \cong \int_j \mathcal{O}_{C_w^{k-1}} \cong R j_* \mathcal{O}_{C_w^{k-1}}$  (cf. [54, Ex. 1.5.22] for the last equality and note that there is an abuse of notation here: j in the first term is the embedding  $C_w^k \hookrightarrow C_w^P$ , while j in the other terms is the embedding  $C_w^{k-1} \hookrightarrow C_w^P$ ). In summary, we obtain distinguished triangles in  $D^b_{\rm qc}(D_{C^P_{\rm uv}})$ :

$$i_*(D_{C_w^P \leftarrow Z_w^k} \otimes_{D_{Z_w^k}} \mathcal{O}_{Z_w^k})[k] \to Rj_*\mathcal{O}_{C_w^k} \to Rj_*\mathcal{O}_{C_w^{k-1}} \xrightarrow{+1} .$$
(A.5)

By taking the long exact cohomology sequence of (A.5) in the case k = d (recall  $C_w^d = C_w^P$ ), it follows:

$$R^{l}j_{*}\mathcal{O}_{C_{w}^{d-1}} \cong \begin{cases} \mathcal{O}_{C_{w}^{d}} & l = 0\\ \int_{i} \mathcal{O}_{Z_{w}^{d}} & l = d-1\\ 0 & \text{otherwise.} \end{cases}$$

Using induction and (A.5) with k decreasing, we can show for  $k \ge 2$ :

$$R^{l} j_{*} \mathcal{O}_{C_{w}^{k-1}} \cong \begin{cases} j_{*} \mathcal{O}_{C_{w}^{k}} & l = 0\\ 0 & l = 1, \dots, k-2 \text{ (if } k > 2)\\ R^{l} j_{*} \mathcal{O}_{C_{w}^{k}} & l \ge k+1, \end{cases}$$

and we have an exact sequence for  $k \geq 2$ 

$$0 \to R^{k-1}j_*\mathcal{O}_{C_w^{k-1}} \to \int_i \mathcal{O}_{Z_w^k} \to R^k j_*\mathcal{O}_{C_w^k} \to R^k j_*\mathcal{O}_{C_w^{k-1}} \to 0.$$
(A.6)

Since  $j: C_w^0 \to C_w^P$  is affine,  $R^l j_* \mathcal{O}_{C_w^0} = 0$  for all l > 0. By (A.5) for k = 1, we deduce

$$0 \to j_* \mathcal{O}_{C^1_w} (\cong \mathcal{O}_{C^P_w}) \to j_* \mathcal{O}_{C^0_w} \to \int_i \mathcal{O}_{Z^1_w} \to R^1 j_* \mathcal{O}_{C^1_w} \to 0$$
(A.7)

and  $R^l j_* \mathcal{O}_{C_w^1} = 0$  for  $l \ge 2$ . This last fact together with  $R^l j_* \mathcal{O}_{C_w^k} \cong R^l j_* \mathcal{O}_{C_w^{k-1}}$  for  $l \ge k+1$  and  $k \ge 2$  imply  $R^l j_* \mathcal{O}_{C_w^k} = 0$  for  $l \ge k+1$  and  $k \in \{0, \ldots, d\}$ . Hence (A.6) becomes (for  $k \ge 2$ )

$$0 \to R^{k-1}j_*\mathcal{O}_{C_w^{k-1}} \to \int_i \mathcal{O}_{Z_w^k} \to R^k j_*\mathcal{O}_{C_w^k} \to 0.$$

These exact sequences together with (A.7) form a long exact sequence (noting that  $C_w^0 = Z_w^0$ ):

$$0 \to \mathcal{O}_{C_w^P} \to \int_i \mathcal{O}_{Z_w^0} \to \int_i \mathcal{O}_{Z_w^1} \to \dots \to \int_i \mathcal{O}_{Z_w^d} \to 0$$

Applying  $\int_{j_w}$  (which is exact, since  $j_w$  is an affine immersion, see the discussion above Proposition A.2.1), we finally obtain a long exact sequence (where we also use  $j_w$  to denote the affine embeddings  $Z_w^i \hookrightarrow G/B$  for all i)

$$0 \to \int_{j_w} \mathcal{O}_{C_w^P} \to \int_{j_w} \mathcal{O}_{Z_w^0} \to \int_{j_w} \mathcal{O}_{Z_w^1} \to \dots \to \int_{j_w} \mathcal{O}_{Z_w^d} \to 0.$$

Taking global sections (which is exact, cf. [54, Thm. 11.2.3]), and using [54, Prop. 12.3.2 (ii)] (where  $\mathbb{D}$  of *loc. cit.* is also referred to as the Verdier dual for coherent left *D*-modules) and [45, Thm. 2.4 (ii)], we obtain a long exact sequence of  $\mathfrak{g}$ -modules

$$0 \to \Gamma(G/B, \mathfrak{N}^P_w) \to M(w^{\max}w_0 \cdot 0)^{\vee} \to \bigoplus_{\substack{u \in \mathscr{W}_{L_P} \\ \lg(u)=1}} M(uw^{\max}w_0 \cdot 0)^{\vee} \to \cdots$$
$$\lim_{\substack{w \to 0 \\ w \to 0}} M(w^{\min}w_0 \cdot 0)^{\vee} \to 0. \quad (A.8)$$

By [57, Prop. 9.6],  $\Gamma(G/B, \mathfrak{N}_w^P)$  has the same formal character as  $M_P(w^{\max}w_0 \cdot 0)^{\vee}$ . By [54, Thm. 11.5.3],  $\Gamma(G/B, \mathfrak{N}_w^P) \in \mathcal{O}^{\mathfrak{p}}(0)$ . Taking duals, we deduce from (A.8) a surjective morphism  $M(w^{\max}w_0 \cdot 0) \twoheadrightarrow \Gamma(G/B, \mathfrak{N}_w^P)^{\vee}$  which, by [57, Thm. 9.4 (c)], has to factor through  $M_P(w^{\max}w_0 \cdot 0)$ 

 $0) \to \Gamma(G/B, \mathfrak{N}_w^P)^{\vee}$ . As both have the same formal character, It follows  $M_P(w^{\max}w_0 \cdot 0) \cong \Gamma(G/B, \mathfrak{N}_w^P)^{\vee}$ , which concludes the proof. (Note that we cannot apply directly [57, Thm. 9.4 (b)] to the dual of (A.8) to deduce this result as it is *a priori* not clear if the map  $\bigoplus_{u \in \mathscr{W}_{L_P}} M(uw^{\max}w_0 \cdot \log(u)) = 1$ 

 $0) \rightarrow M(w^{\max}w_0 \cdot 0)$  is the same as in *loc. cit.*).

Recall we have equivalences of categories (see the discussion above Proposition 5.4.7):

$$\operatorname{Mod}_{\operatorname{rh}}(D_{G/P}, B) \xleftarrow{i_P^*}_{\sim} \operatorname{Mod}_{\operatorname{rh}}(D_{G/B \times G/P}, G) \xrightarrow{i_B^*} \operatorname{Mod}_{\operatorname{rh}}(D_{G/B}, P).$$

We remark that the equivalence of categories  $\iota := i_P^* \circ (i_B^*)^{-1}$  in general does not induce isomorphisms of  $U(\mathfrak{g})$ -modules when taking global sections. For example, for P = B, we have  $\Gamma(G/B, \iota(\mathfrak{L}(w \cdot 0))) \cong L(w^{-1} \cdot 0)$ .

For  $w \in \mathscr{W}$  and  $U_w = G(w, 1)B \times P$  (seen in  $G/B \times G/P$ ), we have  $i_B^{-1}(U_w) = C_w^P$  (resp.  $i_P^{-1}(U_w) = C_{P,w^{-1}}$ ) and  $U_w \cong G \times^P C_w^P$  (resp.  $U_w \cong G \times^B C_{P,w^{-1}}$ ). We use  $j_w$  to denote the embeddings  $C_w^P \hookrightarrow G/B$ ,  $U_w \hookrightarrow G/B \times G/P$  and  $C_{P,w} \hookrightarrow G/P$ . Let  $\widetilde{\mathfrak{N}}_w := \int_{j_w} \mathcal{O}_{U_w} \in Mod_{\mathrm{rh}}(D_{G/B \times G/P}, G)$ , and  $\mathfrak{N}_{P,w} := \int_{j_w} \mathcal{O}_{C_{P,w}} \cong (j_w)_*(D_{G/P \leftarrow C_{P,w}} \otimes_{D_{C_{P,w}}} \mathcal{O}_{C_{P,w}})$  (where the isomorphism follows from the fact that  $j_w : C_{P,w} \to G/P$  is an affine locally closed immersion, see the discussion below (A.9), see also [22, Prop. 1.4.5]). Similarly as in [54, (13.1.7)] (with one B replaced by P, which does not cause any problem), we have  $i_B^* \widetilde{\mathfrak{N}}_w \cong \mathfrak{N}_w^P$  and  $i_P^* \widetilde{\mathfrak{N}}_w \cong \mathfrak{N}_{P,w^{-1}}$ .

Recall from § 5.4 that  $\mathfrak{M}_P(w^{\max}w_0 \cdot 0) \in \mathrm{Mod}_{\mathrm{rh}}(D_{G/B \times G/P}, G)$  satisfies

$$i_B^*\mathfrak{M}_P(w^{\max}w_0 \cdot 0) \cong \operatorname{Loc}_{\operatorname{BB}}\left(M_P(w^{\max}w_0 \cdot 0)\right) \in \operatorname{Mod}_{\operatorname{rh}}(D_{G/B}, P)$$

By Proposition A.2.1,  $i_B^* \mathfrak{M}_P(w^{\max}w_0 \cdot 0)$  is isomorphic to the Verdier dual of  $\mathfrak{N}_w^P \cong i_B^* \mathfrak{\widetilde{N}}_w$  (for example see [45, Thm. 2.4] and see [54, § 2.6] for the Verdier dual of coherent left *D*-modules). From this, together with [54, Thm. 2.7.1] and the fact that  $i_B^*$  induces an equivalence of categories, we deduce that  $\mathfrak{M}_P(w^{\max}w_0 \cdot 0)$  is isomorphic to the Verdier dual of  $\mathfrak{\widetilde{N}}_w$ .

We can now prove Proposition 5.4.8 by generalizing the proof of [45, Thm. 6.2].

Proof of Proposition 5.4.8. As  $\mathfrak{M}_P(w^{\max}w_0\cdot 0)$  is isomorphic to the Verdier dual of  $\widetilde{\mathfrak{N}}_w$ , both have the same characteristic cycle (cf. [54, Prop. 2.6.12]). It is sufficient to show  $[\overline{X}_{w,\lambda}] = [Ch(\widetilde{\mathfrak{N}}_w)]$ . Consider  $q_{P,\lambda}: G \times^P \mathfrak{r}_{P,\lambda} \to \mathfrak{g}$ . We have as in (5.2)  $X_{P,\lambda} \cong G \times^B q_{P,\lambda}^{-1}(\mathfrak{b})$ . Similarly as in Remark 5.2.3 (2), we see that  $q_{P,\lambda}^{-1}(\mathfrak{b})$  is equidimensional of dimension dim  $\mathfrak{r}_{P,\lambda}$  with irreducible components given by  $\{q_{P,\lambda}^{-1}(\mathfrak{b})_w\}_{w \in \mathcal{W}/\mathcal{W}_{L_P}}$ , where  $q_{P,\lambda}^{-1}(\mathfrak{b})_w$  denotes the Zariski closure of the preimage  $q_{P,\lambda}^{-1}(\mathfrak{b})_w^0$ of  $C_{P,w}$  in  $q_{P,\lambda}^{-1}(\mathfrak{b})$  (with the reduced subscheme structure) via the composition  $q_{P,\lambda}^{-1}(\mathfrak{b}) \to G/P \times$  $\mathfrak{g} \to G/P$ . We also have  $X_{w,\lambda} \cong G \times^B q_{P,\lambda}^{-1}(\mathfrak{b})_w$ . Let  $\kappa_{P,\lambda}$  denote the morphism  $G \times^P \mathfrak{r}_{P,\lambda} \to \mathfrak{z}_{\lambda}$ ,  $(g,\psi) \mapsto \overline{\psi}$  and  $\overline{q_{P,\lambda}^{-1}(\mathfrak{b})_w} := q_{P,\lambda}^{-1}(\mathfrak{b})_w \times_{\kappa_{P,\lambda},\mathfrak{z}_{\lambda}} \{0\}$  (with the canonical scheme structure). Then we have an isomorphism of schemes

$$\overline{X}_{w,\lambda} \cong G \times^B \overline{q_{P,\lambda}^{-1}(\mathfrak{b})_{w^{-1}}}.$$

By Proposition 5.4.7, it is sufficient to show (as cycles in  $T^*G/P$ )

$$\left[\overline{q_{P,\lambda}^{-1}}(\mathfrak{b})_{w^{-1}}\right] \cong \left[\operatorname{Ch}(i_P^* \widetilde{\mathfrak{N}}_w)\right] = \left[\operatorname{Ch}(\mathfrak{N}_{P,w^{-1}})\right] = \left[\operatorname{Ch}\left(\int_{j_{w^{-1}}} \mathcal{O}_{C_{P,w^{-1}}}\right)\right]$$

By abuse of notation, we still denote  $\lambda \circ \det_{L_P}$  by  $\lambda$ , which is now a dominant weight of  $\mathfrak{t}$  (with respect to B). Let  $\chi_{-\lambda}$  be the character of P (which factors through  $L_P$ ) of weight  $-\lambda$  over E. Put  $\mathcal{L}_{-\lambda} := G \times^P \chi_{-\lambda}$ , which is a line bundle over G/P. By the Borel-Weil-Bott theorem,  $H^0(G/P, \mathcal{L}_{-\lambda}) \cong L^-(-\lambda)$  (:= the finite dimensional algebraic representation of G of lowest weight  $-\lambda$ , i.e.  $L^-(-\lambda)^{N^-} = E(-\lambda)$  for the unipotent radical  $N^-$  of the Borel subgroup  $B^-$  opposite to B). Indeed, we have by definition

$$H^{0}(G/P, \mathcal{L}_{-\lambda}) \cong \{ f : G \to E \text{ algebraic functions } | f(gp) = \chi_{-\lambda}^{-1}(p)f(g), \forall p \in P, g \in G \}$$
(A.9)

with the G-action given by  $(gf)(g') = f(g^{-1}g')$ . Let  $L_P^{\lambda} \subset L_P$  be the kernel of  $\chi_{-\lambda} : L_P \to E^{\times}$ . Consider  $\kappa : Y := G/(L_P^{\lambda}N_P) \twoheadrightarrow G/P$ . Each element in  $H^0(G/P, \mathcal{L}_{-\lambda})$  can be viewed as an algebraic function on Y. Let  $\mathfrak{e}$  be the highest weight vector of  $L(\lambda) \cong L^-(-\lambda)^{\vee}$  (which is the finite dimensional algebraic representation of G of highest weight  $\lambda$  with respect to B), i.e.  $N\mathfrak{e} = \mathfrak{e}$  and  $\beta\mathfrak{e} = \chi_{\lambda}(\beta)\mathfrak{e} = \chi_{-\lambda}^{-1}(\beta)\mathfrak{e}$  for all  $\beta \in P$  where  $N \subset B$  is the unipotent radical. Let  $\mathfrak{e}^*$  be the lowest weight vector of  $L^-(-\lambda)$ . Put  $f_1 : G \to E, g \mapsto \langle \mathfrak{e}^*, g\mathfrak{e} \rangle$  (with  $\langle -, - \rangle$  the natural pairing between  $L(\lambda)$  and  $L^-(-\lambda)$ ), which corresponds to  $\mathfrak{e}^*$  via (A.9). Let

$$f_{w^{-1}} := w^{-1} f_1 = [g \mapsto \langle \mathfrak{e}^*, wg \mathfrak{e} \rangle]$$

which corresponds to  $w^{-1}\mathfrak{e}^*$  via (A.9) and which we can and do view as an algebraic function on Y. Applying [22, Prop. 1.4.5] (that easily generalizes to our connected split reductive group G) and using the natural surjection  $G/B \twoheadrightarrow G/P^7$ , we can deduce that  $\kappa^{-1}(\partial \overline{C_{P,w^{-1}}})$  (where  $\partial \overline{C_{P,w^{-1}}} = \overline{C_{P,w^{-1}}} \setminus C_{P,w^{-1}}$ ) is exactly the zero locus of  $f_{w^{-1}}|_{\kappa^{-1}(\overline{C_{P,w^{-1}}})}$ .

Let  $\mathcal{U} \subset G/P$  be (Zariski-open) complement of the zero locus of the (global) section  $f_{w^{-1}}$  of the line bundle  $\mathcal{L}_{-\lambda}$  over G/P. So  $\kappa^{-1}(\mathcal{U}) = Y \setminus f_{w^{-1}}^{-1}(0)$  and  $C_{P,w^{-1}} = \overline{C_{P,w^{-1}}} \cap \mathcal{U}$ , in particular  $C_{P,w^{-1}}$  is Zariski-closed in  $\mathcal{U}$ . Denote by  $\iota : C_{P,w^{-1}} \hookrightarrow \mathcal{U}$  the closed embedding, then we have (see [54, Ex. 2.3.8] for the first isomorphism, the last isomorphism is induced by the Killing form)

$$\operatorname{Ch}\left(\int_{\iota} \mathcal{O}_{C_{P,w^{-1}}}\right) \cong T^*_{C_{P,w^{-1}}} \mathcal{U} \cong T^*_{C_{P,w^{-1}}} G/P \hookrightarrow T^*G/P \cong G \times^P (\mathfrak{g}/\mathfrak{p})^{\vee} \cong G \times^P \mathfrak{n}_P.$$

where  $T^*_{C_{P,w^{-1}}}G/P$  denotes the conormal bundle of  $C_{P,w^{-1}}$  in G/P. Recall we have  $q_P: T^*G/P \cong G \times^P \mathfrak{n}_P \to \mathfrak{g}, (g, \psi) \mapsto \operatorname{Ad}(g)\psi$ , and  $q_P^{-1}(\mathfrak{n})^{\operatorname{red}} = \bigcup_{u \in \mathscr{W}/\mathscr{W}_{L_P}} T^*_{C_{P,u}}G/P \hookrightarrow T^*G/P$  (for example by the same argument as in the proof of [31, Prop. 3.3.4]).

We now apply Ginzburg's method in [45, § 6.3] to  $\Lambda := \operatorname{Ch}(\int_{\iota} \mathcal{O}_{C_{P,w^{-1}}})$  and  $f := f_{w^{-1}}$  in order to calculate  $\operatorname{Ch}(\int_{j_{w^{-1}}} \mathcal{O}_{C_{P,w^{-1}}}) = \operatorname{Ch}(\mathfrak{N}_{P,w^{-1}})$  (don't confuse  $j_{w^{-1}}$  with the above  $\iota$ , and note that  $j_{w^{-1}}$  is not closed in general). We have  $T^*G/(L_P^{\lambda}N_P) \cong G \times^{L_P^{\lambda}N_P} \mathfrak{r}_{P,\lambda}$  (identifying  $\mathfrak{g}$  to  $\mathfrak{g}^{\vee}$  via the Killing form). For  $x \in Bw^{-1}P$ , let  $df_x \in E\lambda + \mathfrak{n}_P$  such that  $(x, df_x) \in df \subset$  $G \times^{L_P^{\lambda}N_P} (E\lambda + \mathfrak{n}_P) \cong T^*G/(L_P^{\lambda}N_P)$ . As the map  $q : G \times^{L_P^{\lambda}N_P} (E\lambda + \mathfrak{n}_P) \to \mathfrak{g}, (g, \psi) \mapsto \operatorname{Ad}(g)\psi$ coincides with the moment map  $T^*G/(L_P^{\lambda}N_P) \to \mathfrak{g}^{\vee} (\cong \mathfrak{g})$  (cf. [31, § 1.4]), we have for  $X \in \mathfrak{g}$  the equality  $(q(x, df_x))(X) = \langle \mathfrak{e}^*, w(-X)x\mathfrak{e} \rangle$ . By multiplying x on the right by an element of P, we can and do assume that x has the form  $uw^{-1} \in C_{P,w^{-1}}$  with  $u \in B$  satisfying  $wuw^{-1} \in B^-$ . Let  $K(\cdot, \cdot)$  denote the Killing form on  $\mathfrak{g}$ , we can calculate:

$$q(uw^{-1}, df_{uw^{-1}})(X) = -\chi_{w^{-1}(\lambda)}(u)K(\operatorname{Ad}(uw^{-1})(\lambda), X).$$
(A.10)

<sup>&</sup>lt;sup>7</sup>This surjection induces a surjection  $G \times^B \chi_{-\lambda} \twoheadrightarrow G \times^P \chi_{-\lambda}$ , and we have  $H^0(G/B, G \times^B \chi_{-\lambda}) \cong L^-(-\lambda)$ . We can then deduce the desired results for Schubert cells in G/P from those for Schubert cells in G/B given in [22, Prop. 1.4.5].

Hence  $q(uw^{-1}, df_{uw^{-1}}) \in E^{\times} \operatorname{Ad}(uw^{-1})(\lambda)$ . By unwinding the definition of q(-, -), this implies  $df_{uw^{-1}} \in E^{\times} \lambda$ .

It is clear that  $G/(L_P^{\lambda}N_P)$  is a principal  $\mathbb{G}_m$ -bundle over G/P, and there is a natural induced  $\mathbb{G}_m$ -action on  $T^*G/(L_P^{\lambda}N_P)$  such that  $(T^*G/(L_P^{\lambda}N_P))/\mathbb{G}_m \cong G \times^P \mathfrak{r}_{P,\lambda}$ . For  $a \in E^{\times}$ , we claim that we have by (A.10) an isomorphism of schemes:

$$(\kappa^*\Lambda + adf)/\mathbb{G}_m \cong q_{P,\lambda}^{-1}(-aw^{-1}(\lambda) + \mathfrak{n}) \cap \kappa_{P,\lambda}^{-1}(-a\lambda).$$
(A.11)

Indeed, by checking the formula at each closed point of  $C_{P,w^{-1}}$ , we can obtain the equality after taking the reduced subscheme structure on both sides. However, as in the proof of [45, Thm. 6.3], the left hand side is locally isomorphic to a translation of  $\Lambda \cong T^*_{C_{P,w^{-1}}}G/P \subset T^*G/P \cong$  $G \times^P \mathfrak{n}_P \subset G \times^P \mathfrak{r}_{P,\lambda}$  in  $G \times^P \mathfrak{r}_{P,\lambda}$ , hence is reduced. On the other hand, we have a closed immersion

$$q_{P,\lambda}^{-1}(-aw^{-1}(\lambda) + \mathfrak{n}) \times_{G \times {}^{P}\mathfrak{r}_{P,\lambda}} \kappa_{P,\lambda}^{-1}(-a\lambda) \hookrightarrow q_{P}^{-1}(\mathfrak{b}) \times_{G \times {}^{P}\mathfrak{r}_{P}} \kappa_{P}^{-1}(-a\lambda),$$
(A.12)

which is an equality on closed points. By Proposition 5.2.9 and its proof,  $q_P^{-1}(\mathfrak{b}) \times_{G \times^P \mathfrak{r}_P} \kappa_P^{-1}(-a\lambda)$  is smooth and Zariski closed in  $q_P^{-1}(\mathfrak{b})$ . It is easy to see that the closed subschemes  $\kappa_P^{-1}(-a\lambda) \times_{G \times^P \mathfrak{r}_P} q_P^{-1}(\mathfrak{b})_{w'}$  (for  $w' \in \mathcal{W}/\mathcal{W}_{L_P}$ ) of  $q_P^{-1}(\mathfrak{b}) \times_{G \times^P \mathfrak{r}_P} \kappa_P^{-1}(-a\lambda)$  are disjoint, hence each is open and smooth (and reduced). We deduce

$$q_P^{-1}(\mathfrak{b}) \times_{G \times^P \mathfrak{r}_P} \kappa_P^{-1}(-a\lambda) = \sqcup_{w' \in \mathscr{W}/\mathscr{W}_{L_P}} \kappa_P^{-1}(-a\lambda) \times_{G \times^P \mathfrak{r}_P} q_P^{-1}(\mathfrak{b})_{w'}.$$

Then one can check that (A.12) factors through a closed immersion

$$q_{P,\lambda}^{-1}(-aw^{-1}(\lambda) + \mathfrak{n}) \times_{G \times^{P} \mathfrak{r}_{P,\lambda}} \kappa_{P,\lambda}^{-1}(-a\lambda) \longleftrightarrow q_{P}^{-1}(\mathfrak{b})_{w^{-1}} \times_{G \times^{P} \mathfrak{r}_{P}} \kappa_{P}^{-1}(-a\lambda),$$

which is bijective on closed points hence is an isomorphism since the right hand side is reduced. In particular, the right hand side of (A.11) is also reduced so (A.11) holds.

Let  $\Lambda^{\sharp} := q_{P,\lambda}^{-1}(E^{\times}w^{-1}(\lambda) + \mathfrak{n}) \cap \kappa_{P,\lambda}^{-1}(E^{\times}\lambda)$ . By similar arguments as above, one can show isomorphisms of reduced schemes

$$\Lambda^{\sharp} \xrightarrow{\sim} q_{P,\lambda}^{-1}(\mathfrak{b})_{w^{-1}} \times_{G \times^{P} \mathfrak{r}_{P,\lambda}} \kappa_{P,\lambda}^{-1}(E^{\times}\lambda) \xrightarrow{\sim} q_{P}^{-1}(\mathfrak{b})_{w^{-1}} \times_{G \times^{P} \mathfrak{r}_{P}} \kappa_{P}^{-1}(E^{\times}\lambda).$$

We also have  $\Lambda^{\sharp} \cong q_{P,\lambda}^{-1}(\mathfrak{b})_{w^{-1}} \times_{G \times^{P} \mathfrak{r}_{P}} \kappa_{P}^{-1}(\mathfrak{z}^{\text{reg}-\text{ss}})$  (e.g. using the fact that both have the same  $\overline{E}$ -points and are reduced), hence  $\Lambda^{\sharp}$  is Zariski-open (and Zariski-dense) in  $q_{P,\lambda}^{-1}(\mathfrak{b})_{w^{-1}}$ . Thus the scheme theoretic image of  $\Lambda^{\sharp}$  in  $(T^*G/(L_P^{\lambda}N_P))/\mathbb{G}_m \cong G \times^{P} \mathfrak{r}_{P,\lambda}$  is just  $q_{P,\lambda}^{-1}(\mathfrak{b})_{w^{-1}}$ . By Ginzburg's formula ([45, Thm. 6.3], see also [45, Thm. 3.2], and note that the theorem is "algebraic" so it can be applied with  $\mathbb{C}$  replaced by E) applied to the case where j is the open immersion  $\mathcal{U} \hookrightarrow G/P$  and  $\mathscr{M}$  is  $\int_{\iota} \mathcal{O}_{C_{P,w^{-1}}}$  (so gr  $\mathscr{M}$  is just  $\mathrm{Ch}(\int_{\iota} \mathcal{O}_{C_{P,w^{-1}}})$  and  $\lim_{s\to 0} (\mathrm{gr} \mathscr{M})^{s}$  in *loc. cit.* is the fibre of the scheme theoretic image of  $\Lambda^{\sharp}$  at 0 via  $\kappa_{P}$ ), we obtain:

$$\left[\operatorname{Ch}\left(\int_{j_{w^{-1}}}\mathcal{O}_{C_{P,w^{-1}}}\right)\right] = \left[q_{P,\lambda}^{-1}(\mathfrak{b})_{w^{-1}} \times_{\kappa_{P,\lambda},\mathfrak{z}_{\lambda}} \{0\}\right].$$

The proposition follows.

#### A.3Bruhat intervals of length 2

We show some properties of Bruhat intervals of length 2 in certain parabolic quotient of  $S_n$ , that are used in Corollary 7.2.18.

We write 
$$w \in S_n$$
 in the form  $w = \begin{pmatrix} 1 & 2 & \cdots & n \\ w^{-1}(1) & w^{-1}(2) & \cdots & w^{-1}(n) \end{pmatrix}$ . For  $i \neq j \in \{1, \dots, n\}$ ,

we denote by  $t_{ij}$  the transposition exchanging i and j. If  $w = \begin{pmatrix} 1 & 2 & \cdots & n \\ a_1 & a_2 & \cdots & a_n \end{pmatrix}$  and i < j, then  $t_{ij}w = \begin{pmatrix} 1 & \cdots & i & \cdots & j & \cdots & n \\ a_1 & \cdots & a_j & \cdots & a_i & \cdots & a_n \end{pmatrix}$ . Let  $r \ge 1$ , and  $n_i \in \mathbb{Z}_{\ge 1}$  for  $1 \le i \le r$  such that  $\sum_{i=1}^r n_i = n$ . Put  $s_i := \sum_{j=1}^i n_i$  and  $J_i := \{s_{i-1} + 1, \dots, s_i\}$ . We have thus  $\{1, \dots, n\} = \bigsqcup_{i=1}^r J_i$ . Let  $\mathscr{W}_J$  be the subgroup of  $S_n$  generated by the  $t_{jk}$  for  $j, k \in J_i, i = 1, \dots, r$  (so  $\mathscr{W}_J \cong \prod_{i=1}^r S_{n_i}$ ). Denote by  $\mathscr{W}_{\max}^J \subseteq S_n$  the set of maximal length representatives of  $\mathscr{W}_J \setminus S_n$ .

**Lemma A.3.1.** We have  $w = \begin{pmatrix} 1 & 2 & \cdots & n \\ a_1 & a_2 & \cdots & a_n \end{pmatrix} \in \mathscr{W}_{\max}^J$  if and only if for any  $j = 1, \dots, r$  the sequence  $(a_j)_{j \in J_i}$  is decreasing.

Let  $[w_1, w_2]$  be a Bruhat interval of length 2, i.e.  $w_1 < w_2$ ,  $\lg(w_2) = \lg(w_1) + 2$ , and  $[w_1, w_2] = \lg(w_1) + 2$ .  $\{w' \in S_n \mid w_1 < w' < w_2\}$ . Recall  $|[w_1, w_2]| = 2$  (cf. [19, Lemma 5.2.7]). Assume  $w_1, w_2 \in \mathscr{W}^J_{\max}$ , and denote by  $[w_1, w_2]_J := \{w' \in W^J_{\max} | w_1 < w' < w_2\}$ . Recall  $[w_1, w_2]_J$  is called *full* if  $[w_1, w_2]_J = [w_1, w_2]$ . As  $\lg(w_2) = \lg(w_1) + 2$ , there exist reflections  $t_{ab}$ ,  $t_{cd}$  such that  $w_2 = t_1 t_2 w_1$ . We call a full interval  $[w_1, w_2]_J$  (of length 2) nice if the integers a, b, c, d cannot be contained in two  $J_i$ .

**Remark A.3.2.** Identifying  $S_n$  with the Weyl group  $\mathscr{W}$  of  $GL_n$ , J corresponds to a parabolic subgroup P of  $GL_n$  containing B such that  $\mathscr{W}_{L_P} = \mathscr{W}_J$ . It is easy to see that, if  $[w_1, w_2]_J$  is full and nice, then  $\dim \mathfrak{z}_{L_P}^{w_2 w_1^{-1}} = \dim \mathfrak{z}_{L_P} - 2$  (see Proposition 5.3.3 for the notation).

**Proposition A.3.3.** Let  $w = \begin{pmatrix} 1 & 2 & \cdots & n \\ a_1 & a_2 & \cdots & a_n \end{pmatrix} \in \mathscr{W}_{\max}^J$  with  $\lg(w) \le \lg(w_0) - 2$ . Assume that the partition  $\{J_i\}$  satisfies: if  $|J_i| > 1$  then  $|J_{i-1}| = 1$  (if  $i \ge 2$ ) and  $|J_{i+1}| = 1$  (if  $i \le r-1$ ). Then there exists  $u \in \mathscr{W}_{\max}^J$  such that u > w,  $\lg(u) = \lg(w) + 2$  and one of the following two properties is satisfied:

- (1)  $[w, u]_J$  is full and nice;
- (2)  $[w, u]_J$  is not full.

We will frequently use the following easy lemma (see for example [11, Lemma 2.1.4]):

**Lemma A.3.4.** Let  $w = \begin{pmatrix} 1 & 2 & \cdots & n \\ a_1 & a_2 & \cdots & a_n \end{pmatrix} \in S_n$  and i < j. Then  $t_{ij}w$  is a cover of w (i.e.  $t_{ij}w > w$  and  $\lg(t_{ij}w) = \lg(w) + 1$  if and only if  $a_i < a_j$  and there does not exist i, j, k with i < k < j such that  $a_i < a_k < a_j$ .

Proof of Proposition A.3.3. Let  $I := \{j \mid a_j < a_{j+1}\}$ . Since  $w \in \mathcal{W}_{\max}^J$ , for any  $j = 1, \ldots, r$ , the sequence  $(a_j)_{j \in J_i}$  is decreasing. We have thus  $I \subset \{s_j \mid j = 1, \ldots, r-1\}$ . Since  $w \neq w_0, I \neq \emptyset$ . We prove the proposition by a (somewhat tedious) case-by-case discussion.

**Case (1):** Assume there exist  $s_{k_1}, s_{k_2} \in I$  such that  $k_2 > k_1 + 1$ . In this case, we don't need the assumption on  $\{J_i\}$ . For  $k_i$ , consider the set  $I_i := \{(j_1, j_2) \mid j_1 \in J_{k_i}, j_2 \in J_{k_i+1}, a_{j_1} < a_{j_2}\}$ . We have  $(s_i, s_i+1) \in I_i$ , so  $I_i \neq \emptyset$ . Let  $(j_{i,1}, j_{i,2}) \in I_i$  such that  $j_{i,1} - j_{i,2}$  is maximal. We have thus  $a_{j_{i,1}} < a_{j_{i,2}}$  and  $\begin{cases} a_{j_{i,1}-1} > a_{j_{i,2}} \text{ if } j_{i,1} - 1 \in J_{k_i} \\ a_{j_{i,1}} > a_{j_{i,2}+1} \text{ if } j_{2}+1 \in J_{k_i+1}. \end{cases}$  By Lemma A.3.4,  $t_{j_{i,1}j_{i,2}}w$  is a cover of w (in fact, for any  $(j_1, j_2) \in I_i, t_{j_1j_2}w$  is a cover of w). By the choice of  $(j_{i,1}, j_{i,2})$  and Lemma A.3.1, we see that  $t_{j_{i,1}j_{i,2}}w \in \mathscr{W}_{\max}^J$ . As  $k_2 > k_1 + 1, t_{j_{1,1}j_{1,2}}$  and  $t_{j_{2,1}j_{2,2}}$  commute. Put  $u := t_{j_{1,1}j_{1,2}}t_{j_{2,1}j_{2,2}}w$ , then it is easy to see that  $[w, u]_J = [w, u] = w < t_{j_{1,1}j_{1,2}}w, t_{j_{2,1}j_{2,2}}w < u$  is full and nice.

**Case (2)**: There exists k such that  $I = \{s_k\}$ . We have either  $n_k = 1$  or  $n_{k+1} = 1$ . Note that we cannot have  $n_k = n_{k+1} = 1$  since if so  $\lg(w) = \lg(w_0) - 1$ .

(2.1) Assume  $n_k = 1$ , let  $j \in J_{k+1}$  be maximal such that  $a_{s_k} < a_j$ . We have  $j \ge s_k + 2$  since otherwise  $\lg(w) = \lg(w_0) - 1$ . Put  $u := t_{s_k(j-1)}t_{s_kj}w$ , then  $\lg(u) = \lg(w) + 2$  and  $[w, u] = w < t_{s_kj}w, t_{s_k(j-1)}w < u$ . However  $t_{s_k(j-1)} \notin \mathscr{W}_{\max}^J$ . So  $[w, u]_J$  is not full.

(2.2) The case  $n_{k+1} = 1$  is parallel to (2.1).

**Case (3)**: There exists k such that  $I = \{s_k, s_{k+1}\}$ . By our assumption on  $\{J_i\}$ , we can further divide this case into two cases:  $n_{k+1} = 1$  or  $n_k = n_{k+2} = 1$ .

(3.1)  $n_{k+1} = 1$ : let  $j \in J_k$  be minimal such that  $a_j < a_{s_k+1}$  (note that  $s_k + 1 = s_{k+1}$  in this case).

(3.1.1) If  $j < s_k$ , put  $u := t_{j+1s_{k+1}}t_{js_{k+1}}w \in \mathscr{W}_{\max}^J$ . In  $S_n$ , the interval [w, u] is given by  $w < t_{js_{k+1}}w, t_{(j+1)s_{k+1}}w < u$ . However,  $t_{(j+1)s_{k+1}}w \notin \mathscr{W}_{\max}^J$ . So  $[w, u]_J$  is not full.

(3.1.2) If  $j = s_k$ , let  $j_1 \in J_{k+2}$  (resp.  $j_2 \in J_{k+2}$ ) be maximal such that  $a_{j_1} > a_{s_{k+1}}$  (resp.  $a_{j_2} > a_{s_k}$ ) (recalling that  $a_{s_{k+1}+1} > a_{s_{k+1}} > a_{s_k}$ ). Note that  $j_2 \ge j_1$ .

(3.1.2.1) If  $a_{s_{k+1}} > a_{j_2}$ , then put  $u := t_{s_{k+1}j_2}t_{s_ks_{k+1}}w \in \mathscr{W}_{\max}^J$ . Then  $[w, u]_J = [w, u] = w < t_{s_ks_{k+1}}w, t_{s_kj_2}w < u$ . Indeed, as  $a_{s_{k+1}} > a_{j_2}, t_{s_kj_2}w$  is a cover of w. It is also easy to see that  $[w, u]_J$  is nice.

(3.1.2.2) If  $a_{s_{k+1}} < a_{j_2}$  (so  $j_1 = j_2$ ), put (again)  $u := t_{s_{k+1}j_2}t_{s_ks_{k+1}}w \in \mathscr{W}_{\max}^J$ . In this case we have  $[w, u]_J = [w, u] = w < t_{s_ks_{k+1}}w, t_{s_{k+1}j_2}w < u$ , and  $[w, u]_J$  is nice (so the only difference with (3.1.2.1) is that  $t_{s_kj_2}w$  is replaced by  $t_{s_{k+1}j_2}w$ ).

(3.2)  $n_k = n_{k+2} = 1$ . Let  $j_1 \in J_{k+1}$  (resp.  $j_2 \in J_{k+1}$ ) be maximal (resp. minimal) such that  $a_{j_1} > a_{s_k}$  (resp.  $a_{j_2} < a_{s_{k+2}}$ ).

(3.2.1) If  $j_1 \ge s_k + 1$  (resp.  $j_2 \le s_{k+1} - 1$ ), which implies  $n_{k+1} > 1$ , then one can use the same argument as in (2.1) (resp. (2.2)) to find  $u \in \mathcal{W}_{\max}^J$  such that  $[w, u]_J$  is not full.

(3.2.2) If  $j_1 = s_k + 1$  and  $j_2 = s_{k+1}$ , and if  $n_{k+1} > 1$  (so  $j_2 > j_1$ ), then put  $u := t_{s_k(s_k+1)}t_{s_{k+1}s_{k+2}}w = t_{s_{k+1}s_{k+2}}t_{s_k(s_k+1)}w$ . It is easy to see  $[w, u]_J = w < t_{s_k(s_k+1)}w, t_{s_{k+1}s_{k+2}}w < u$  is full and nice.

(3.2.3) If  $n_{k+1} = 1$ , this is a special case of (3.1.2.2).

This concludes the proof.

**Remark A.3.5.** Without the assumption on  $\{J_i\}$  in Proposition A.3.3, it could happen that for any  $u \in \mathscr{W}_{\max}^J$ , u > w and  $\lg(u) = \lg(w) + 2$ ,  $[w, u]_J$  is full but not nice. For example, let n = 4, r = 2,  $J_1 = \{1, 2\}$ ,  $J_2 = \{3, 4\}$ , and  $w = t_{12}t_{34}t_{23} = \begin{pmatrix} 1 & 2 & 3 & 4 \\ 3 & 1 & 4 & 2 \end{pmatrix} \in \mathscr{W}_{\max}^J$ .

Then  $u = t_{34}t_{23}t_{12}t_{23}t_{34} = \begin{pmatrix} 1 & 2 & 3 & 4 \\ 4 & 2 & 3 & 1 \end{pmatrix}$  is the only element in  $\mathscr{W}_{\max}^J$  such that u > w and  $\lg(u) = \lg(w) + 2$ . One can check  $[w, u]_J = [w, u] = \left\{ \begin{pmatrix} 1 & 2 & 3 & 4 \\ 4 & 1 & 3 & 2 \end{pmatrix}, \begin{pmatrix} 1 & 2 & 3 & 4 \\ 3 & 2 & 4 & 1 \end{pmatrix} \right\}$  hence is full. However, in this case, any full interval (of length 2) is not nice.

# A.4 Errata to [19]

• It turns out that the conjecture [19, Conj. 2.3.7] (describing  $X_w \cap V_{w'}$  for  $w' \leq w$ ) is wrong, some counter-examples were found by Seginus Mowlavi. It is however true for most pairs  $w' \leq w$ , see his forthcoming PhD. thesis [71].

• In the equality following the definition of the scheme Z in [19, (2.11)],  $\widetilde{\mathcal{N}} \times_{\mathcal{N}} \widetilde{\mathcal{N}}$  should be replaced by  $(\widetilde{\mathcal{N}} \times_{\mathcal{N}} \widetilde{\mathcal{N}})^{\text{red}}$  (in fact, it is possible that  $\widetilde{\mathcal{N}} \times_{\mathcal{N}} \widetilde{\mathcal{N}} = (\widetilde{\mathcal{N}} \times_{\mathcal{N}} \widetilde{\mathcal{N}})^{\text{red}}$ , but we don't need it).

• The following erratum is due to Benjamin Schraen and Zhixiang Wu and corrects a mistake in the proof of [19, Prop. 3.7.2]. We use below the notation of *loc. cit.* It is stated there that the map  $f: \hat{V}_{\tilde{x}} \to \hat{U}_x$  induces an injection on global sections, which is false in general. Fortunately this can be corrected.

Let us recall the context of the proof of [19, Prop. 3.7.2]. We have a reduced affinoid space Uover  $L, x \in U$  a point and a morphism of formal schemes  $\widehat{U}_x \to X_r$  where  $X_r = \operatorname{Spf} R_r$  and  $R_r$ is a (framed) deformation ring. We construct a proper birational morphism  $f: \widetilde{U} \to U$  which is obtained as a composition of normalizations and blowing up. It is proved in the first part of the proof of [19, Prop. 3.7.2] that if  $\widetilde{x} \in \widetilde{U}$  maps to  $x \in U$ , the composite morphism of formal schemes

$$\widehat{\widetilde{U}}_{\widetilde{x}} \longrightarrow \widehat{U}_x \longrightarrow X_r$$

factors through a closed formal subscheme  $X_{r,\mathcal{M}_{\bullet}} = \operatorname{Spf} R_{r,\mathcal{M}_{\bullet}} \subset X_r$ . Now, the actual proof diverges from [19, Prop. 3.7.2].

Note that, f being a composition of normalizations and blowing up, is projective locally on U. Up to refining U, we can assume that  $\widetilde{U}$  is a closed rigid analytic subspace of  $\mathbb{P}^n_U$  for some  $n \geq 1$ . By [12, Cor. 5.4 (c)], up to further refining U, we can assume that there exists a p-adically complete topologically finitely generated  $\mathcal{O}_L$ -algebra  $A_0$  and a closed formal scheme  $\mathfrak{W}_0 \subset \mathbb{P}^n_{\mathrm{Spf} A_0}$  such that  $U = (\mathrm{Spf} A_0)^{\mathrm{rig}}$  and  $\widetilde{U} = \mathfrak{W}_0^{\mathrm{rig}}$ . As  $A_0$  is a p-adically complete ring, it follows from [46, Cor. 5.1.8] that there exists a projective scheme  $W_0$  over  $\mathrm{Spec} A_0$  whose p-adic completion is isomorphic to  $\mathfrak{W}_0$ . Let  $W := W_0[p^{-1}]$ , this is a projective scheme over  $\mathrm{Spec} A$  where  $A := A_0[p^{-1}]$ . We can identify the closed points of U and  $\mathrm{Spec} A$  and the closed points of  $\widetilde{U}$  and W. As the completion of  $\mathrm{Spec} A$  (resp. of W) and U (resp.  $\widetilde{U}$ ) at x (resp.  $\widetilde{x}$ ) are isomorphic, it is sufficient to prove the statement with U replaced by  $\mathrm{Spec} A$  and  $\widetilde{U}$  replaced by W.

Note that the map  $W \to \operatorname{Spec} A$  is proper dominant. It is proper since it is projective. The dominance follows from the fact that the map  $f: \widetilde{U} \to U$  is surjective so that all the closed points of  $\operatorname{Spec} A$  are in the image of  $W \to \operatorname{Spec} A$ .

Let I be an ideal of A such that  $\sqrt{I} = \mathfrak{m}_x$ . Let  $W_I := W \times_{\operatorname{Spec} A} \operatorname{Spec}(A/I)$  and  $\widetilde{U}_I := \widetilde{U} \times_{(\operatorname{Spf} A_0)^{\operatorname{rig}}} (\operatorname{Spf}(A_0/I \cap A_0))^{\operatorname{rig}}$  so that  $\widetilde{U}_I$  is isomorphic to the rigid analytic space associated to the projective k(x)-scheme  $W_I$ .

The map  $\operatorname{Spec}(A/I) \hookrightarrow \operatorname{Spec} A$  gives rise to an A/I-point of the formal scheme  $\operatorname{Spf} \widehat{A_{\mathfrak{m}_x}}$ and to a morphism of formal schemes  $\operatorname{Spf}(A/I) \hookrightarrow \operatorname{Spf} R_r$ . Let us prove that the composition  $R_r \to A/I \to H^0(W_I, \mathcal{O})$  factors through  $R_{r,\mathcal{M}_{\bullet}}$ . Namely we have a map

$$H^0(W_I, \mathcal{O}) \hookrightarrow \prod_{y \in |W_I|} \mathcal{O}_{W_I, y} \hookrightarrow \prod_{y \in |W_I|} \widehat{\mathcal{O}}_{W_I, y}$$

where  $|W_I|$  is the set of closed points of  $W_I$ , and the assertion follows from the fact that each  $R_r \to \widehat{\mathcal{O}}_{W_I,y}$  factors through  $R_{r,\mathcal{M}_{\bullet}}$ . This being true for all I, the map

$$R_r \longrightarrow \varprojlim_I H^0(W_I, \mathcal{O})$$

factors through  $R_{r,\mathcal{M}_{\bullet}}$ .

Since the map  $W \to \operatorname{Spec} A$  is dominant and A is reduced, the map  $A \to H^0(W, \mathcal{O})$  is injective. As  $\varprojlim_I H^0(W_I, \mathcal{O})$  is the completion of  $H^0(W, \mathcal{O})$  for the  $\mathfrak{m}_x$ -adic topology (see [46, Thm. 4.1.5]) and  $\widehat{A}_{\mathfrak{m}_x}$  is a flat A-module, the map  $\widehat{A}_{\mathfrak{m}_x} \to \varprojlim_I H^0(\widetilde{U}_I, \mathcal{O})$  is injective. This proves that the map  $R_r \to \widehat{A}_{\mathfrak{m}_x}$  factors through  $R_{r,\mathcal{M}_{\bullet}}$ , which gives the desired statement in [19, Prop. 3.7.2].

• In [19, § 4.3], all  $[K : \mathbb{Q}_p] \frac{n(n+3)}{2}$  should be replaced by  $n + [K : \mathbb{Q}_p] \frac{n(n+1)}{2}$ .

• In [19, § 5], the BGG category  $\mathcal{O}$  should be replaced by its full subcategory  $\mathcal{O}_{alg}$  of objects with integral weights (see [74]).
## References

- [1] The Stacks Project Authors. The stacks project. https://stacks.math.columbia.edu.
- [2] Daniel Barrera Salazar and Chris Williams. Parabolic eigenvarieties via overconvergent cohomology. *Mathematische Zeitschrift*, 2021.
- [3] Joël Bellaïche and Gaëtan Chenevier. Families of Galois representations and Selmer groups. Astérisque, 324:1–314, 2009.
- [4] John Bergdall. Paraboline variation over *p*-adic families of  $(\varphi, \Gamma)$ -modules. Compositio Mathematica, 153(1):132–174, 2017.
- [5] Laurent Berger. Construction de  $(\varphi, \Gamma)$ -modules: représentations *p*-adiques et *B*-paires. Algebra & Number Theory, 2(1):91–120, 2008.
- [6] Laurent Berger. Équations différentielles *p*-adiques et  $(\varphi, N)$ -modules filtrés. Astérisque, 319:13–38, 2008.
- [7] Laurent Berger and Pierre Colmez. Familles de représentations de de Rham et monodromie p-adique. Astérisque, 319:303–337, 2008.
- [8] Joseph Bernstein (written by Pierre Deligne). Le centre de Bernstein. Représentations des groupes réductifs sur un corps local. *Travaux en Cours*, 1–32, Hermann, 1984.
- [9] Roman Bezrukavnikov and Simon Riche. Affine braid group actions on derived categories of Springer resolutions. Annales Scientifiques de l'École Normale Supérieure, 45:535–599, 2012.
- [10] Sara Billey and V. Lakshmibai. Singular loci of Schubert varieties. Progress in Mathematics, 182, 2000.
- [11] Anders Björner and Francesco Brenti. Combinatorics of Coxeter groups. Graduate Texts in Mathematics, 231, 2005.
- [12] Siegfried Bosch and Werner Lütkebohmert. Formal and rigid analytic geometry II Flattening techniques. *Mathematische Annalen*, 296:403–429, 1993.
- [13] Christophe Breuil. Vers le socle localement analytique pour  $GL_n$  I. Annales de l'Institut Fourier, 66:633–685, 2016.
- [14] Christophe Breuil. Vers le socle localement analytique pour  $GL_n$  II. Mathematische Annalen, 361:741–785, 2015.
- [15] Christophe Breuil and Yiwen Ding. Sur un problème de compatibilité local-global localement analytique. *Memoirs of the Amer. Math. Soc.*, to appear.
- [16] Christophe Breuil and Yiwen Ding. Higher  $\mathcal{L}$ -invariants for  $GL_3(\mathbb{Q}_p)$  and local-global compatibility. *Cambridge Journal of Mathematics*, 8(4):775–951, 2020.
- [17] Christophe Breuil, Eugen Hellmann, and Benjamin Schraen. Une interprétation modulaire de la variété trianguline. *Mathematische Annalen*, 367(3-4):1587–1645, 2017.
- [18] Christophe Breuil, Eugen Hellmann, and Benjamin Schraen. Smoothness and classicality on eigenvarieties. *Inventiones Mathematicae*, 209(1):197–274, 2017.

- [19] Christophe Breuil, Eugen Hellmann, and Benjamin Schraen. A local model for the trianguline variety and applications. *Publications Mathématiques de l'IHÉS*, 130:299–412, 2019.
- [20] Christophe Breuil and Florian Herzig. Towards the finite slope part for  $GL_n$ . International Mathematics Research Notices, 10495–10552, 2020.
- [21] Christophe Breuil and Peter Schneider. First steps towards p-adic Langlands functoriality. Journal für die reine und angewandte Mathematik, 2007(610):149–180, 2007.
- [22] Michel Brion. Lectures on the geometry of flag varieties. Topics in Cohomological Studies of Algebraic Varieties: Impanga Lecture Notes, 33–85, 2005.
- [23] Colin J Bushnell and Philip C Kutzko. The admissible dual of GL<sub>N</sub> via restriction to compact open subgroups. *Harmonic analysis on reductive groups*, 89–99, 1991.
- [24] Kevin Buzzard. Eigenvarieties. London Math. Soc. Lecture Note Series, 320:59–120, 2007.
- [25] Ana Caraiani. Monodromy and local-global compatibility for  $\ell = p$ . Algebra & Number Theory, 8:1597–1646, 2014.
- [26] Ana Caraiani, Matthew Emerton, Toby Gee, David Geraghty, Vytautas Paškūnas, and Sug Woo Shin. Patching and the *p*-adic local Langlands correspondence. *Cambridge Journal* of Mathematics, 4(2):197–287, 2016.
- [27] Luis G Casian and David H Collingwood. The Kazhdan-Lusztig conjecture for generalized Verma modules. *Mathematische Zeitschrift*, 195(4):581–600, 1987.
- [28] Gaëtan Chenevier. Familles *p*-adiques de formes automorphes pour  $GL_n$ . Journal für die reine und angewandte Mathematik, 570:143–217, 2004.
- [29] Gaëtan Chenevier. On the infinite fern of Galois representations of unitary type. Annales Scientifiques de l'École Normale Supérieure, 44(6):963–1019, 2011.
- [30] Gaëtan Chenevier. Sur la densité des représentations cristallines du groupe de Galois absolu de Q<sub>p</sub>. Mathematische Annalen, 335:1469−1525, 2013.
- [31] Neil Chriss and Victor Ginzburg. Representation theory and complex geometry. *Modern Birkhäuser Classics*, 1997.
- [32] Pierre Colmez and Jean-Marc Fontaine. Construction des représentations *p*-adiques semistables. *Inventiones Mathematicae*, 140:1–43, 2000.
- [33] J-F Dat. Caractères à valeurs dans le centre de Bernstein. Journal für die reine und angewandte Mathematik, 508:61–83, 1999.
- [34] Vinay V Deodhar. On some geometric aspects of bruhat orderings ii. the parabolic analogue of Kazhdan-Lusztig polynomials. *Journal of Algebra*, 111(2):483–506, 1987.
- [35] Yiwen Ding. Some results on the locally analytic socle for  $\operatorname{GL}_n(\mathbb{Q}_p)$ . International Mathematics Research Notices, 5989–6035, 2018.
- [36] Yiwen Ding. Companion points and locally analytic socle for  $GL_2(L)$ . Israel Journal of Mathematics, 231:47–122, 2019.

- [37] Yiwen Ding. Simple  $\mathcal{L}$ -invariants for  $GL_n$ . Transactions of the Amer. Math. Soc., 372(11):7993–8042, 2019.
- [38] Yiwen Ding.  $GL_2(\mathbb{Q}_p)$ -ordinary families and automorphy lifting. https://arxiv.org/abs/ 1907.07372.
- [39] J Matthew Douglass. An inversion formula for relative Kazhdan-Lusztig polynomials. Communications in Algebra, 18(2):371–387, 1990.
- [40] Matthew Emerton. Jacquet modules of locally analytic representations of p-adic reductive groups I. Construction and first properties. Annales Scientifiques de l'École Normale Supérieure, 39(5):775–839, 2006.
- [41] Matthew Emerton. Jacquet modules of locally analytic representations of *p*-adic reductive groups II. The relation to parabolic induction. to appear in *J. Institut Math. Jussieu*.
- [42] Matthew Emerton. Locally analytic vectors in representations of locally p-adic analytic groups. Memoirs of the Amer. Math. Soc., 248(1175), 2017.
- [43] John Fogarty. Fixed point schemes. American Journal of Mathematics, 95(1):35–51, 1973.
- [44] Jean-Marc Fontaine. Arithmétique des représentations galoisiennes *p*-adiques. *Astérisque*, 295:1–115, 2004.
- [45] Victor Ginzburg. &-modules, Springer's representations and bivariant Chern classes. Advances in Mathematics, 61:1–48, 1986.
- [46] Alexander Grothendieck and Jean Dieudonné. Éléments de Géométrie Algébrique 3: Étude cohomologique des faisceaux cohérents (première partie). Publications Mathématiques de l'IHÉS, 11, 1961.
- [47] Alexander Grothendieck and Jean Dieudonné. Éléments de Géométrie Algébrique 4: Étude locale des schémas et des morphismes de schémas (première partie). Publications Mathématiques de l'IHÉS, 20:5–259, 1964.
- [48] Lucio Guerberoff. Modularity lifting theorems for Galois representations of unitary type. Compositio Mathematica, 147(4):1022–1058, 2011.
- [49] Michael Harris and Richard Taylor. The geometry and cohomology of some simple Shimura varieties. *Annals of Math. Studies*, 151, 2001.
- [50] Yiqin He. Bernstein simple  $\mathcal{L}$ -invariants. In progress, 2021.
- [51] Eugen Hellmann. Families of *p*-adic Galois representations and  $(\varphi, \Gamma)$ -modules. Commentarii Mathematici Helvetici, 91(4):721–749, 2016.
- [52] Eugen Hellmann and Benjamin Schraen. Density of potentially crystalline representations of fixed weight. *Compositio Mathematica*, 152(8):1609–1647, 2016.
- [53] Richard Hill and David Loeffler. Emerton's Jacquet functors for non-Borel parabolic subgroups. *Documenta Mathematica*, 16:1–31, 2011.
- [54] Ryoshi Hotta, Kiyoshi Takeuchi, and Toshiyuki Tanisaki. *D*-modules, perverse sheaves, and representation theory. *Progress in Mathematics*, 236, 2008.

- [55] Shanxiao Huang. Thesis. In progress, 2021.
- [56] Roland Huber. Étale cohomology of rigid analytic varieties and adic spaces. Aspects of Mathematics, 30, 2013.
- [57] James E Humphreys. Representations of semisimple Lie algebras in the BGG category O. Graduate Studies in Mathematics, 94, 2008.
- [58] James E Humphreys. Introduction to Lie algebras and representation theory, Graduate Texts in Mathematics, 9, 1972.
- [59] Kiran Kedlaya and Ruochuan Liu. On families of  $(\varphi, \Gamma)$ -modules. Algebra & Number Theory, 4(7):943–967, 2011.
- [60] Kiran S Kedlaya, Jonathan Pottharst, and Liang Xiao. Cohomology of arithmetic families of  $(\varphi, \Gamma)$ -modules. Journal of the American Mathematical Society, 27(4):1043–1115, 2014.
- [61] Reinhardt Kiehl and Rainer Weissauer. Weil conjectures, perverse sheaves and l-adic Fourier transform. Ergebnisse der Mathematik und ihrer Grenzgebiete, 42, 2001.
- [62] Mark Kisin. Moduli of finite flat group schemes, and modularity. Annals of Math. (2), 170(3):1085–1180. 2009.
- [63] Mark Kisin. Potentially semi-stable deformation rings. Journal of the American Mathematical Society, 21(2):513–546, 2008.
- [64] Jan Kohlhaase. The cohomology of locally analytic representations. Journal für die reine und angewandte Mathematik, 651:187–240, 2011.
- [65] Daniel Le, Bao Le Hung, Brandon Levin, and Stefano Morra. Local models for Galois deformation rings and applications. https://arxiv.org/abs/2007.05398.
- [66] Ruochuan Liu. Cohomology and duality for  $(\varphi, \Gamma)$ -modules over the Robba ring. International Mathematics Research Notices, 2007.
- [67] Ruochuan Liu. Triangulation of refined families. *Commentarii Mathematici Helvetici*, 90(4):831–904, 2015.
- [68] David Loeffler. p-adic L-functions in universal deformation families. https://arxiv.org/ abs/2003.13738.
- [69] Barry Mazur. An introduction to the deformation theory of Galois representations. Galois groups over Q, Springer, 243–311, 1997.
- [70] Vikram Mehta and Vasudevan Srinivas. Normality of Schubert varieties. American Journal of Mathematics, 109(5):987–989, 1987.
- [71] Seginus Mowlavi. Thesis. In progress, 2021.
- [72] David Mumford. The red book of varieties and schemes (second edition). Lecture Notes in Maths, 1358, 1999.
- [73] Kentaro Nakamura. Classification of two-dimensional split trianguline representations of p-adic fields. Compositio Mathematica, 145(04):865–914, 2009.

- [74] Sascha Orlik and Matthias Strauch. On Jordan-Hölder series of some locally analytic representations. Journal of the American Mathematical Society, 28(1):99–157, 2015.
- [75] Alexandre Pyvovarov. The endomorphism ring of projectives and the Bernstein centre. Journal de Théorie des Nombres de Bordeaux, 32(1):49–71, 2020.
- [76] Peter Schneider. Nonarchimedean Functional Analysis. Springer Monographs in Math., 2002.
- [77] Peter Schneider and Jeremy Teitelbaum. Locally analytic distributions and *p*-adic representation theory, with applications to GL<sub>2</sub>. Journal of the American Mathematical Society, 15(2):443–468, 2002.
- [78] Peter Schneider and Jeremy Teitelbaum. Algebras of *p*-adic distributions and admissible representations. *Inventiones Mathematicae*, 153(1):145–196, 2003.
- [79] Peter Schneider and Jeremy Teitelbaum. Duality for admissible locally analytic representations. Representation Theory, 9:297–326, 2005.
- [80] Peter Scholze. The local Langlands correspondence for  $GL_n$  over *p*-adic fields. *Inventiones* Mathematicae, 192(3):663–715, 2013.
- [81] Peng Shan. Graded decomposition matrices of v-Schur algebras via Jantzen filtration. Representation Theory, 16:212–269, 2012.
- [82] Toshiyuki Tanisaki. Characteristic varieties of highest weight modules and primitive quotients. Advanced Studies in Pure Maths., 14:14–38, 1988.
- [83] Richard Taylor. Galois representations associated to Siegel modular forms of low weight. Duke Mathematical Journal, 63(2):281–332, 1991.
- [84] Jack Thorne. On the automorphy of ℓ-adic Galois representations with small residual image. J. Inst. Math. Jussieu, 11:855–920, 2012.
- [85] Zhixiang Wu. Local models for the trianguline variety and partially classical families. https://arxiv.org/abs/2103.03823.

C. Breuil, CNRS, BÂTIMENT 307, FACULTÉ D'ORSAY, UNIVERSITÉ PARIS-SACLAY, 91405 ORSAY CEDEX, FRANCE

*E-mail address*, C. Breuil: christophe.breuil@universite-paris-saclay.fr

Y. Ding, B.I.C.M.R., PEKING UNIVERSITY, NO.5 YIHEYUAN ROAD HAIDIAN DISTRICT, BEIJING, P.R. CHINA 100871

E-mail address, Y. Ding: yiwen.ding@bicmr.pku.edu.cn

## Index

 $B_{\Omega,\lambda}(V), 27$  $B_{\sigma,\lambda}(V), 21$  $F_D^0, 51$  $F_{D,\mathscr{F}}, 58$  $F_{D,\mathscr{F}}^0, 59$  $F^{0}_{\rho,\mathscr{F}}, 60$  $L(\lambda), 20$  $L(\lambda)_P, 20$  $M[\mathfrak{m}^{\infty}], 23$  $R_{\rho,\mathcal{M}_{\bullet}}, 106$  $R^w_{\rho,\mathcal{M}_{\bullet}}, 106$  $U_{\Omega,\mathbf{h}}(\overline{\rho}), 62$  $U_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0, \mathbf{h}), 74$  $V_w, 82$  $V_{\overline{\rho}}^{
m pcr}(\xi_0, {f h})_w, \, 78 \ X_P, \, 81$  $X_w, 82$  $X_{D,\mathcal{M}_{\bullet}}, 103$  $X_{D,\mathcal{M}_{\bullet}}^{\square,w}, 104$  $X_{D,\mathcal{M}\bullet}^{\Box}, 103$  $X_{D,\mathcal{M}_{\bullet}}^{w}, 104$  $X_{W,\mathcal{F}_{\bullet}}, 94$  $X_{W,\mathcal{F}_{\bullet}}^{\sqcup}, 94$  $\begin{array}{c} X_{W^+,\mathcal{F}_{\bullet}}, \ 103 \\ X_{W^+,\mathcal{F}_{\bullet}}^{\Box,w}, \ 104 \end{array}$  $X_{W^+,\mathcal{F}_{\bullet}}^{\Box}, 103$  $X^w_{W^+,\mathcal{F}_{\bullet}}, 104$  $X_{\Omega,\mathbf{h}}(\overline{\rho}), \, 62$  $X_{\Omega,\mathbf{h}}(\overline{\rho}_p), \, 48$  $X_{\mathcal{M},\mathcal{M}_{\bullet}}, 97$  $X^{w}_{\rho,\mathcal{M}_{\bullet}}, 105$  $Z^{0}(Z), 89$  $Z^{i}(-), 109$  $Z_P, 88$ [M], 92[Y], 89 $\operatorname{Art}(E), 51$  $Ch(\mathfrak{M}), 90$  $\mathbb{T}^S, 32$  $\mathbb{T}_v, 32$  $\mathcal{E}^{\infty}_{\Omega,\lambda}(\overline{\rho}), 45$ 

 $\mathcal{E}_{\Omega,\lambda}(U^p), 33$  $\mathcal{M}_{\Omega,\lambda}(U^p), 33$  $\mathcal{M}^{\infty}_{\Omega,\lambda}, 45$  $\mathcal{Z}_{\Omega}, 20$  $\chi_{\rm cyc}, 14$  $\chi_{\underline{\varpi}}, 24$  $\chi_{\varpi_L}, 14$  $\chi_{\varpi_i}, 24$  $\mathcal{O}(X), 15$  $\mathcal{O}_{X,x}, 15$  $\delta^0, 14$  $\delta_{\lambda}, 19$  $\delta_{\overline{\omega}}^{\chi}, 10$   $\delta_{\overline{\omega}}^{0}, 24$   $\delta_{\overline{\omega}}^{unr}, 24$   $\delta_{\overline{\omega}i}^{0}, 24$   $\delta_{\overline{\omega}i}^{unr}, 24$   $\delta_{\overline{\omega}i}^{unr}, 24$  $\mathfrak{X}_{\overline{\rho}}^{\mathrm{per}}(\xi_0, \mathbf{h}), 74$  $\mathfrak{z}_0, 23$  $\mathfrak{z}_{L_{P}}^{w(w')^{-1}}, 87$  $\mu_{\Omega}, 14, 28$  $\mu_{\Omega}^{\text{unr}}, 14$  $\mathcal{T}_P, 83$  $\mathcal{T}_w, 84$  $\mathscr{W}, 15$  $\mathscr{W}_{\max}^P, 15$  $\mathscr{W}_{\min}^P, 15$  $\mathscr{W}_F, 36$  $\mathscr{W}_L, 15$  $\mathscr{Z}, 61$  $r(\rho), 16$  $X_{x}, 15$  $Z_{L_P}(L)_{\delta}, 97$  $\widehat{\mathcal{O}}_{X,x}, 15$  $\widetilde{U}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0,\mathbf{h}),\ 74$  $\widetilde{\mathfrak{X}}_{\overline{\rho}}^{\mathrm{pcr}}(\xi_0,\mathbf{h}), \, 74$  $a_{w,w'}, 91$  $b_{w,w'}, 91$ k(x), 15 $w_0, \, 15$  $w_{0,F}, 43$  $w_{0,L}, 15$